

The Incidence and Economic Burden of Hospital Acquired Infection occurring in Surgical Patients

Thesis submitted to the University of London for the degree of Doctor of Philosophy, August 2003

Rosalind Mary Plowman

Health Services Research Unit Department of Public Health and Policy London School of Hygiene & Tropical Medicine Keppel Street, London, WC1E 7HT, UK

BEST COPY

AVAILABLE

Variable print quality

ACKNOWLEDGEMENTS

Many people have helped and supported me whilst undertaking this study and I am grateful to them all. A very special thank you goes to my supervisor Professor Jenny Roberts. Jenny encouraged me to embark on this work and has supported me throughout. Her dedication and enthusiasm have been an inspiration to me.

Thanks must also go to all those involved in the Socio-economic Burden of HAI study. I would like to thank the members of the Steering Committee and Advisory Group, for their thoughtful and much appreciated insights. I would especially like to thank Dr Tony Swan, and Dr Barry Cookson for their much valued advice and their support and encouragement throughout.

I would also like to thank the members of the research team especially Dr Nick Graves and Mark Griffin who worked so very hard on the underlying study. Thanks must also go to the research assistants who were responsible for recruiting patients into the study and subsequently collecting the vast amount of data: Rachel Dunk, Alison Franklin, Janette Trevarthen, Maggie Waters, Jennifer White, and Linda Wright. A big thank you also to Barbara Ayres the project team secretary who always had a smile on her face and kept us all going when things were difficult.

Warm thanks must also go to the Board and clinical and non-clinical staff at the study hospital for their willingness to take part in this study and their help and support throughout. A special thank you must go to the Director of Nursing Practice who offered support and encouragement throughout and assisted greatly with practical issues at the study site. A big thank you also to the Information Officer who extracted data from the hospital data set often at very short notice, the Laboratory Manager who gave up considerable time to assist in costing laboratory tests, the Medical Coding Manager, the Medical Records Supervisor, and the ward nursing staff and other health care professionals who

2

made the research assistants so welcome. Thanks must also go to the patients that participated so willingly in this research. Without them there would be no study. This research is indebted to them.

Thanks must also go to the Department of Health who funded the original study and North Thames Regional Health Authority for awarding me a training fellowship. A special thank you to Professor Liz Meerabeau, Dr Elizabeth Tebbs and Dr William Maton-Howarth who at various stages were involved in the study at the level of the Department of Health. I very much appreciate the support they all offered.

Warm thanks also go to Andre Charlet who has always been there to offer statistical advice and my family, friends, and colleagues especially Valerie Ward, Dr Sara Schroter, Dr Pauline Allen and my dear sister P, who have supported and encouraged me throughout.

Finally a very special thank you to Chris for his support throughout, and my wonderful children - Ellie, Anthony and Max - who have only ever known me working on this project and are very much looking forward to the day when I am not!

ABSTRACT

Background: Approximately 9% of patients in hospital have a hospital acquired infection (HAI). These infections place a burden on the health sector, patients and carers.

Objectives: To assess the incidence of, and independent risk factors for HAIs occurring in adult surgical patients; to assess the impact of these infections on the hospital sector; and to show how this information may be used to assess the potential benefits of prevention.

Design: A prospective survey of the incidence of HAI was conducted. Resources used by both infected and uninfected patients were recorded and costed. Generalised linear modelling techniques were used to estimate the impact of HAI on the observed variation in costs. Logistic regression analysis was used to determine independent risk factors for HAI.

Setting: A district general hospital in England

Subjects: 2469 adult patients admitted to five surgical specialties between April 1994 and May 1995.

Results: 7.5% (95% CI: 6.4, 8.6) acquired one or more HAIs that presented during the in-patient period. The incidence, economic impact and independent risk factors varied with site of infection. On average HAIs increased hospital costs by a factor of 2.3 (95% CI: 2.0, 3.0), equivalent to an additional £2,254 (95% CI: £1,738, £2,770) per case and increased length of stay by a factor of 2.1 (95% CI: 1.8, 2.5), equivalent to an extra 7.8 days (95% CI: 5.7, 10.0) per case. The estimates represent the average gross benefits of prevention. Net benefits depend on the cost and effectiveness of prevention activities. Estimates of the gross benefits of a 15% reduction in infection rates and a framework for assessing the net benefits of prevention are presented.

Conclusion: The study provides an estimate of HAI by specialty and site for surgical patients. It calculates the burden on the hospital sector and shows the benefits that might accrue if HAI rates were reduced.

TABLE OF CONTENTS

	Page
Title Page	1
Acknowledgements	2
Abstract	4
Table of Contents	6
List of Tables	22
List of Figures	29
List of Appendices	32
Abbreviations	34

CHAPTER 1: INTRODUCTION

1.1	Introduction	35
1.2	The scale of the problem	36
1.3	The impact of HAI on the individual	37
1.4	The impact on the health sector	37
1.5	Why patients acquire an infection	38
1.6	The scope for prevention	38
1.7	The potential benefits of improved prevention	40
1.8	The rationale for this thesis and the anticipated contribution it will make	40
1.9	My role in the socio-economic burden study	44
	1.9.1 Literature review	44
	1.9.2 Establishing links with staff at the study hospital	44
	1.9.3 Further development and implementation of the study protocol	45

1.9.4	Selection and implementation of an appropriate data entry system45
1.9.5	Obtaining the approval of the relevant ethical committees prior to the pilot study46
1.9.6	Conducting a pilot study and revising the methods in response to the findings of the pilot study46
1.9.7	Re-submitting the study protocol and data collection forms for ethical approval prior to the main study46
1.9.8	Recruiting and training six research assistants46
1.9.9	Managing the main data collection and entry period47
1.9.10	Appointing a secretary48
1.9.11	Liaising closely with the research economist and assisting in the development of unit costs of resources used
1.9.12	Liaising closely with the project statisticians
1.9.13	Ensuring that the work is completed within an appropriate time frame
1.9.14	Writing the final report and submitting it to the Department of Health49
1.9.15	Preparation of the report for publication
1.9.16	Liaising with the Department of Health50
1.9.17	Liaising with the National Audit Office in relation to their work on HAI
1.9.18	Dissemination of the findings51

1.12	The structure of the thesis	53
	1.11.2 Objectives	52
	1.11.1 Aim	52
1.11	Aims and objectives of this thesis	52
1.10	My role in this thesis	51

CHAPTER 2: THE EPIDEMIOLOGY OF HAI

2.1	Introdu	uction	••••••	55
2.2	The fre	equency ar	nd distribution of HAI	57
2.3	Preval	ence of HA	N	57
	2.3.1	The over	all prevalence of HAI	61
		2.3.1.1	Case definitions	62
		2.3.1.2	Identification methods	63
		2.3.1.3	Sites of infections included	64
		2.3.1.4	Rates reported	65
		2.3.1.5	Case mix included	65
	2.3.2	Prevalen	ce of specific types of HAI	67
	2.3.3	The preva	alence of HAI in patients admitted to	
		surgical s	pecialties	69
	2.3.4	Frequenc	y of specific types of infections occurring in	
		patients a	idmitted to surgical specialties	70
2.4	Incider	nce of HAI.		71
	2.4.1	Estimates	of the incidence of HAI	72
	2.4.2	The relati	ve incidence of specific types of infection	74

	2.4.3	The incid surgical s	ence of HAIs occurring in patients admitted to specialties	75
	2.4.4	The incid admitted	ence of specific types of infection in patients to surgical specialties	76
		2.4.4.1	The incidence of SWIs in patients admitted to surgical specialties	76
		2.4.4.2	The incidence of UTIs in surgical patients	80
		2.4.4.3	The incidence of LRTIs in surgical patients	80
	2.4.4	Evidence	of inter-hospital variation	80
	2.4.5	Infections	s presenting post-discharge	81
2.5	Nation	al estimate	es of the number of HAIs occurring per annum	81
2.6	Impact	t of HAI on	mortality	82
	2.6.1	Methods	used to assess the impact of HAI on	
		mortality		84
		2.6.1.1	Review of case notes	84
		2.6.1.2	Autopsy examination	84
		2.6.1.3	Review of death certificates	84
		2.6.1.4	Case control approach	85
		2.6.1.5	Data stratified by underlying risk	86
		2.6.1.6	Regression analysis	86
	2.6.2	Specialty	specific mortality rates	87
	2.6.3	Site-spec	ific mortality rates	87
	2.6.4	Specialty	and site-specific mortality estimates	90
	2.6.5	Number o	of deaths associated with HAIs per annum	92

2.7	Aetiolo	gy of HAI	92
	2.7.1	Micro-organism/Pathogen	93
	2.7.2	Route of Transmission	94
	2.7.3	Patient	94
	2.7.4	Environment	95
2.8	Risk fa	ctors for HAI	96
	2.8.1	Risk factors for HAIs at any site	97
	2.8.2	Risk factors for surgical wound infections	98
	2.8.3	Risk factors for urinary tract infections	.100
	2.8.4	Risk factors for lower respiratory tract infections	.102
	2.8.5	Risk factors for bloodstream infections	.103
2.9	Risk in	dices	.103
2.10	The pro	oblem of microbial resistance to antibiotics and	
	other a	ntimicrobials	.106
	2.10.1	The development of antibiotic resistance	.107
2.11	Conclu	sion	.108
CHAP	TER 3:	THE ECONOMIC EVALUATION OF HAI	
3.1	Introdu	ction	.110
3.2	Econor	mic evaluation	.113
	3.2.1	Partial economic evaluations	.113
	3.2.2	Full economic evaluations	.115
		3.2.2.1 Cost-effectiveness analysis	. 115
		3.2.2.2 Cost-utility analysis	. 115
		3.2.2.3 Cost benefit analysis	. 116

`

3.3	The ec	onomic evaluation of HAI116
3.4	Estimat	tes of the economic burden of HAI117
	3.4.1	The range of costs resulting from HAI126
	3.4.2	Attributing resources to HAI128
	3.4.3	Estimating the cost of resources attributable to HAI
	3.4.4	Alternative measures of the impact of HAI133
3.5	Estimat	tes of the economic burden of HAI to the hospital sector133
	3.5.1	Estimates of the cost of specific types of HAI
	3.5.2	Estimates of the cost of HAIs occurring in selected patient groups
	3.5.3	The distribution of the additional hospital costs resulting from treating HAIs
	3.5.4	The national burden of HAIs135
3.6	Assess	ment of the benefits of prevention136
	3.6.1	Estimates of the gross benefits of prevention
	3.6.2	Estimates of the net benefits of prevention139
		3.6.2.1 Studies that have assessed the benefits of an
		effective infection control programme
		 3.6.2.2 Studies that have assessed the costs and benefits of antimicrobial prophylaxis
	3.6.3	Estimates of cost savings resulting from not carrying
		out 'prevention' activities which have little or no positive
		effect145
3.7	Conclus	sion145

CHA	PTER 4	: THE STUDY OF THE SOCIO-ECONOMIC BURDEN OF H	AI
4.1	Introdu	uction	147
4.2	Aims a	and objectives of the socio-economic burden of HAI study	148
	4.2.1	Aim	148
	4.2.1	Objectives	148
4.3	Overvi	iew of study design	148
4.4	Study	site	149
	4.4.1	Infection control arrangements at the study hospital	151
4.5	Sampl	e Size	151
4.6	Subjec	ts1	153
4.7	Recrui	tment1	154
4.8	Data c	ollected1	156
	4.8.1	Data on patient characteristics1	57
	4.8.2	Data on HAIs presenting during the hospital phase1	58
	4.8.3	Data on health sector resource use during the hospital phase	59
4.9	Method	ds used to derive cost estimates of the resources used1	59
	4.9.1	Methods used to derive cost estimates of hospital resources used during the in-patient phase1	59
4.10	Data M	lanagement1	63
4.11	Data a	nalysis1	64
	4.11.1	Checking for recruitment/selection bias1	64
	4.11.2	Incidence of HAIs presenting during the in-patient period1	64
	4.11.3	Incidence of HAIs presenting post discharge1	64

	4.11.4	Attribution of costs to HAI164
		4.11.4.1 In-patient analysis164
		4.11.4.2 Post-discharge analysis165
	4.11.5	Impact of HAI on health status166
	4.11.6	Deriving national estimates of the number of patients
		who acquire one or more HAI
	4.11.7	Deriving national estimates of the cost of HAI to the
		hospital sector166
4.12	Project	Management167
4.13	Project	Timetable168
4.14	Conclu	sion169
CHAPI	FER 5:	METHODS
5.1	Introdu	ction170
5.2	The ain	ns and objectives of this thesis170
	5.2.1	Aim170
	5.2.2	Objectives
5.3	Overvie	ew of study design171
5.4	Selection	on of surgical patients and relevant data for the purposes
	of the F	PhD thesis
5.5	Data ar	nalysis undertaken for PhD thesis177

5.5.1

5.5.2

5.5.3

	5.5.3.2 Multivariable analysis - overview
	5.5.3.3 Normal linear regression – no transformation
	of dependent variable180
	5.5.3.4 Linear regression – log transformation of
	dependent variable181
	5.5.3.5 Generalised linear modelling assuming a
	gamma distribution
	5.5.4 Distribution of costs incurred
5.6	Data management
5.7	Study timetable
5.8	Conclusion188
CHAF	PTER 6: SAMPLE CHARACTERISTICS
6.1	Introduction189
6.2	Sample size
6.3	Day recruited191
6.4	Age and sex distribution192
6.5	Specialty distribution193
6.6	Type of admissions194
6.7	Re-admissions195
	6.7.1 Number of patients classified as 're-admissions'
	6.7.2 Characteristics of patients classified as 're-admissions' 195
6.8	Distribution of patients by primary discharge diagnosis
6.9	Number of co-morbidities

6.10	Charac proced	cteristics of patients who had one or more operative lures
	6.10.1	Number of patients who had one or more operations
	6.10.2	Day of surgery199
	6.10.3	Characteristics of patients who underwent one or more operations
	6.10.4	Types of surgery performed201
6.11	In-patie	ent deaths203
6.12	Discha	rge destination and follow up care203
6.13	Service	es organized on discharge from hospital
	6.13.1	Transport
	6.13.2	Follow up care
6 1 4	Conclu	206
0.14	Conciu	SIO[1200
CHAP	TER 7:	INCIDENCE OFAND RISK FACTORS FOR HAI:RESULTS
CHAP 7.1	TER 7:	INCIDENCE OFAND RISK FACTORS FOR HAI:RESULTS ction
6.14 CHAP 7.1 7.2	TER 7: Introdu The inc	INCIDENCE OFAND RISK FACTORS FOR HAI:RESULTS ction
CHAP 7.1 7.2	TER 7: Introdu The inc 7.2.1	INCIDENCE OFAND RISK FACTORS FOR HAI:RESULTS ction
CHAP 7.1 7.2	TER 7: Introdu The inc 7.2.1	INCIDENCE OFAND RISK FACTORS FOR HAI:RESULTS ction
6. 14 CHAP 7.1 7.2	TER 7: Introdu The inc 7.2.1	INCIDENCE OFAND RISK FACTORS FOR HAI:RESULTS ction
6. 14 CHAP 7.1 7.2	TER 7: Introdu The inc 7.2.1	INCIDENCE OFAND RISK FACTORS FOR HAI:RESULTS ction
6. 14 CHAP 7.1 7.2 7.3	TER 7: Introdu The inc 7.2.1 7.2.2 Day of	INCIDENCE OFAND RISK FACTORS FOR HAI:RESULTS ction
 CHAP 7.1 7.2 7.3 7.4 	TER 7: Introdu The inc 7.2.1 7.2.2 Day of Indepen	INCIDENCE OFAND RISK FACTORS FOR HAI:RESULTS ction
 CHAP 7.1 7.2 7.3 7.4 7.5 	TER 7: Introdu The inc 7.2.1 7.2.2 Day of Indepen	INCIDENCE OFAND RISK FACTORS FOR HAI:RESULTS ction

7.7	The inc	idence of UTIs and identification of risk factors22	1
	7.7.1	The incidence of UTIs22	1
	7.7.2	Independent risk factors for UTIs22	3
7.8	Inciden	ce of SWIs and the identification of risk factors22	6
	7.8.1	Overall incidence of SWIs22	6
	7.8.2	The incidence of SWIs occurring in patients who had one	
		or more operative procedures	6
	7.8.3	Independent risk factors for SWI	6
7.9		The incidence of lower respiratory tract infections and	
		identification of independent risk factors23	1
	7.9.1	The incidence of LRTIs23	1
	7.9.2	Independent risk factors for LRTIs23	1
7.10	Conclu	sion23	5
СНАРТ	'ER 8:	THE IMPACT OF HAIS ON HOSPITAL IN-PATIENT COSTS: RESULTS	
8.1	Introdu	ction23	6
8.2	Results	of the single variable analysis23	6
8.3	Results	of the multivariable regression analysis24	1
	8.3.1	Results of the generalised linear model that used an identity link	2
		8.3.1.1 Estimates of the cost of HAI242	2
		8.3.1.2 Testing for interactions	3
		8.3.1.3 Specialty specific estimates of the impact of HAI	
			-

2011.2

		8.3.1.4 Site specific estimates of the impact of HAI on hospital costs
	8.3.2	Results of the generalised linear model that used a loglink 245
		8.3.2.1 Estimates of the ratio of costs incurred by infected and uninfected patients
		8.3.2.2 Testing for interactions246
		8.3.2.3 Specialty specific estimates of the impact of HAI on hospital costs
		8.3.2.4 Site specific estimates of the impact of HAI on hospital costs
8.4	The uni	e distribution of hospital costs incurred by infected and nfected patients
8.5	Conclu	sion252
СНАРТ	'ER 9:	THE IMPACT OF HAIS ON LENGTH OF HOSPITAL STAY: RESULTS
9.1	Introdu	ction253
9.2	Results	of the single variable analysis253
9.3	Results	of the multivariable analysis257
	9.3.1	Results of the generalised linear model that used an identity link
		9.3.1.1 Estimates of the impact of HAI on length of hospital stay258
		9.3.1.2 Testing for interactions259
		9.3.1.3 Specialty specific estimates of the impact of HAI on length of hospital stay

4.000 (MC

(1) A set of the se

		9.3.1.4 Site specific estimates of the impact of HAI on length of hospital stay260
	9.3.2	Results of the generalised linear model that used a loglink 262
		9.3.2.1 Estimates of the ratio of LOS incurred by infected and uninfected patients
		9.3.2.2 Testing for interactions
		9.3.2.3 Specialty specific estimates of the impact of HAI on LOS
		9.3.2.4 Site specific estimates of the impact of HAI on LOS 264
9.4	Conclu	usion

CHAPTER 10: THE POTENTIAL BENEFITS OF INVESTMENT IN INFECTION PREVENTION AND CONTROL

10.1	Introduction20		266
10.2	The potential benefits of prevention2		266
10.3	The gr	The gross benefits of prevention	
	10.3.1	Methods used to derive estimates of the potential gross benefits of a 15% reduction in infection rates.	267
	10.3.2	The potential gross benefits of a 15% reduction in infection rates at the level of the study hospital.	269
	10.3.3	National estimates of the potential gross benefits of a 15% reduction in infection rates.	279

10.4	The net benefits of investment in infection control		
	10.4.1	An economic model to assess the cost and benefits of the routine use of silver alloy coated urinary catheters to reduce the risk of urinary tract infections in catheterised patients	
	10.4.2	Aims and objectives of the economic model	
	10.4.3	The structure of the economic model	
	10.4.4	Illustrative model	
	10.4.5	Results of the illustrative model	
	10.4.6	Sensitivity analysis	
	10.4.7	Validity of the model	
	10.4.8	Interpreting the results of the model	
	10.4.9	Potential applications	
10.5	Conclu	sion297	

CHAPTER 11: DISCUSSION AND CONCLUSIONS

11.1	Introduction		298
11.2	Overvi	ew of results	298
	11.2.1	Incidence of HAI	298
	11.2.2	Risk factors for HAI	300
		11.2.2.1 Urinary tract infections	300
		11.2.2.2 Surgical wound infections	301
		11.2.2.3 Lower respiratory tract infections	301

	11.2.3	Economic burden
		11.2.3.1 Estimated cost to the hospital sector as a result of additional in-patient care
		11.2.3.2 Estimated number of additional days infected patients remain in hospital
		11.2.3.3 Distribution of additional costs
11.3	Strengt	ths of study
	11.3.1	Inclusion of a number of surgical specialties
	11.3.2	Use of previously validated definitions of infection and surveillance methods
	11.3.3	A detailed approach to costing resource use
	11.3.4	Attribution methods
	11.3.5	Generalisability of results to other health care settings
11.4	Limitati	ions
	11.4.1	The study was limited to one hospital
	11.4.2	Limited to adult non-day case patients admitted to selected surgical specialties
	11.4.3	Potential patient selection bias
	11.4.4	Limited to the assessment of the incidence of HAI presenting during the in-patient period
	11.4.5	Strict criteria for identifying surgical wound infections
	11.4.6	Incomplete data on the pathogens involved
	11.4.7	Incomplete data on risk factors for HAI
	11.4.8	The use of average costs

	11.4.9	Attribution of resources used and costs to HAI	324
	11.4.10) The study was limited to the estimation of the costs to the hospital sector	; 325
	11.4.11	I The study was limited to the estimation of the economic burden of HAI	326
11.5	Method	dological considerations	327
	11.5.1	Recruitment	327
	11.5.2	Difficulties associated with conducting research in a clinical context	328
	11.5.3	Quantity of data required	329
11.6	Implica	tions of results for policy	329
11.7	Areas 1	for future research	333
11.8	Conclu	sion	334
Refere	nces		36-367
Appen	dices		68-481

LIST OF TABLES

Table 2.1	Prevalence of HAIs and of infected patients as found in
	studies conducted in Europe, the USA, Canada and
	Australia since 197559
Table 3.1	An overview of studies that have assessed the impact of
	HAIs (all sites) on hospital length of stay and costs (1975-2002)
Table 3.2	An overview of studies that have assessed the impact of
	hospital acquired UTIs on hospital length of stay and costs
	(1975-2002)
Table 3.3	An overview of studies that have assessed the impact of
	hospital acquired SWIs on hospital length of stay and costs
	(1975-2002)
Table 3.4	An overview of studies that have assessed the impact of
	hospital acquired LRTIs on hospital length of stay and costs
	(1975-2002)
Table 3.5	An overview of studies that have assessed the impact of
	hospital acquired BSIs on hospital length of stay and costs
	(1975-2002)
Table 3.6	An overview of studies that have assessed the impact of
	multiple HAIs on hospital length of stay and costs
	(1975-2002)125
Table 4.1	An overview of the data collected157
Table 5.1	Shapiro- Francia W' test for normal data

Table 6.1	Age and sex distribution192
Table 6.2	Sex and specialty distribution193
Table 6.3	Age and specialty distribution
Table 6.4	Admission type distribution194
Table 6.5	Admission type distribution by admission specialty195
Table 6.6	Characteristics of patients admitted within month of a previous admission
Table 6.7	The distribution of patients admitted within a month of a previous admission to hospital by presence or absence of a HAI and admission specialty
Table 6.8	The number and percentage of patients by primary discharge diagnosis grouped into 14 diseases categories
Table 6.9	The number and percentage of patients by number of co-morbidities and admission specialty
Table 6.10	The number and percentage of patients who had one or more operations
Table 6.11	The number and percentage of patients undergoing primary procedures classified according to the Office of Population Censuses and Surveys listed categories
Table 6.12	Distribution of patients by discharge destination

Table 6.13	The number and percentage of patients requiring NHS transport home by age group204
Table 6.14	The number and percentage of patients requiring NHS transport home by admission specialty
Table 6.15	The number and percentage of patients for whom community based follow-up care was organized by type of care and specialty
Table 6.16	The number and percentage of patients for whom hospital based follow-up care was organized by type of care and specialty
Table 7.1	The incidence of HAIs presenting in surgical patients during the hospitalised phase by selected characteristics
Table 7.2	Incidence of HAI by primary operative procedure category211
Table 7.3	The number of patients presenting with one or more HAIs in hospital per 1000 patient days at risk
Table 7.4	The day the primary HAI presented214
Table 7.5	The odds of acquiring a HAI by key patient characteristics (single variable analysis)
Table 7.6	The odds of acquiring a HAI by key patient characteristics (multivariable analysis)
Table 7.7	The number, percentage and incidence of HAIs by site of infection

Table 7.8	The day HAIs presented by site of infection219
Table 7.9	The incidence of UTIs in surgical patients
Table 7.10	The odds of acquiring a UTI by key patient characteristics (single variable analysis)
Table 7.11	The odds of acquiring a UTI by selected factors after controlling for all other factors listed (multivariable analysis)225
Table 7.12	The incidence of SWIs in surgical patients
Table 7.13	The incidence of SWIs occurring in patients who had one or more operative procedures
Table 7.14	The odds of acquiring a SWI by key patient characteristics (single variable analysis)
Table 7.15	The odds of acquiring a SWI (multi-variable analysis)230
Table 7.16	The incidence of LRTIs in surgical patients
Table 7.17	The odds of acquiring a LRTI by key patient characteristics (single variable analysis)
Table 7.18	The odds of acquiring a LRTI (multi-variable analysis)234
Table 8.1	Hospital costs incurred during the in-patient hospital stay by key patient characteristics
Table 8.2	Mean in-patient hospital costs by HAI status and key patient characteristics

Table 8.3	Hospital in-patient costs by type of HAI241
Table 8.4	Results of the generalised linear model which assessed the impact of HAI on hospital costs using an identity link242
Table 8.5	Specialty specific estimates of the additional costs incurred by infected patients
Table 8.6	Estimates of the additional costs incurred by infected patients by site of infection
Table 8.7	Results of the generalised linear model which assessed the impact of HAI on hospital costs using a log link
Table 8.8	Specialty specific estimates of the ratio of costs incurred by infected compared to uninfected patients
Table 8.9	Estimates of the ratio of costs incurred by infected compared to uninfected patients by site of infection
Table 8.10	The distribution of hospital costs incurred by infected and uninfected patients
Table 9.1	Length of hospital stay by key patient characteristics
Table 9.2	The mean length of stay by HAI status and key patient characteristics
Table 9.3	Length of hospital stay by infection status
Table 9.4	Results of the generalised linear model which assessed the impact of HAI on length of hospital stay using an identity link258

Table 9.5	Specialty specific estimates of the impact of HAI on length of hospital stay
Table 9.6	Estimates of the impact of specific types of HAI on length of hospital stay
Table 9.7	Results of the generalised linear model which assessed the impact of HAI on length of hospital stay using a log link
Table 9.8	Specialty specific estimates of the ratio of costs incurred by infected and uninfected patients
Table 9.9	Estimates of the ratio of LOS incurred by infected patients compared to uninfected patients by site of infection
Table 10.1	Estimates of the number of adult non-day case patients admitted to five surgical specialties at the study hospital in 1994/5 who acquired an infection by site of infection and specialty
Table 10.2	Estimates of the cost of HAIs occurring in adult, non-day case patients admitted to five surgical specialties at the study hospital in 1994/5 and the potential benefits of a 15% reduction in rates.
Table 10.3	Estimates of the number of additional days adult, non-day case patients, admitted to five surgical specialties at the study hospital in 1994/5 and who acquired an HAI remained in hospital and the number of bed days released if rates were 15% lower

Table 10.4	Estimates of the number of adult non-day case patients	
	admitted to five surgical specialties at NHS hospitals in	
	England who acquired an HAI IN 1994/5 by site of infection	
	and specialty27	'5

LIST OF FIGURES

•

Figure 2.1	Prevalence rate formula58
Figure 2.2	Cumulative incidence formula71
Figure 2.3	Incidence density formula71
Figure 5.1	A plot of the quantiles of LOS against the quantiles of the normal distribution
Figure 5.2	A plot of the quantiles of hospital costs against the quantiles of the normal distribution
Figure 5.3	A plot of the quantiles of LOS transformed onto the log scale (base10) against the quantiles of the normal distribution182
Figure 5.4	A plot of the quantiles of hospital costs transformed onto the log scale (base10) against the quantiles of the normal distribution
Figure 5.5	A plot of the mean length of stay against the standard deviation for length of stay184
Figure 5.6	A plot of the mean hospital costs against the standard deviation for hospital costs
Figure 5.7	A plot of the coefficient of variation of LOS against the categories involved
Figure 5.8	A plot of the coefficient of variation of hospital costs against the categories involved

Figure 6.1	The number of patients eligible for recruitment and the number recruited
Figure 6.2	The cumulative percentage of patients recruited by day of admission (n= 2462)*
Figure 6.3	Distribution of patients by admission specialty193
Figure 7.1	The day the primary HAI presented215
Figure 10.1	The structure of an economic model to assess the costs and benefits of an intervention to prevent catheter associated urinary tract infections
Figure 10.2	An economic model of the costs and benefits of the routine use of silver alloy coated catheters in adult non-day case patients admitted to five surgical specialties of a hospital in England in 1994/5
Figure 10.3	Estimated costs and benefits of the routine use of silver alloy coated catheters in adult, non-day case patients admitted to the general surgical specialty of an NHS hospitals in England
Figure 10.4	Estimated costs and benefits of the routine use of silver alloy coated catheters in adult, non-day case patients admitted to the orthopaedic specialty of an NHS hospitals in England290
Figure 10.5	Estimated costs and benefits of the routine use of silver alloy coated catheters in adult, non-day case patients admitted to the urology specialty of an NHS hospitals in England

- Figure 10.6 Estimated costs and benefits of the routine use of silver alloy coated catheters in adult, non-day case patients admitted to the gynaecology specialty of an NHS hospitals in England291

LIST OF APPENDICES

Appendix 1	Research assistants' training and seminar programme368
Appendix 2	Standards for Records and Record Keeping -Socio-economic
	Burden of Hospital Acquired Infection
Appendix 3	Presentations concerning the Socio-economic Burden of
	Hospital Acquired Infection Study
Appendix 4	Wards involved in the Socio-economic burden of Hospital
	Acquired Infection Study379
Appendix 5	Patient information sheet, consent and decline to
	participate forms
Appendix 6	Definitions of infections presenting during the patient's
	hospital stay386
Appendix 7	Examples of data collection forms used
Appendix 8	Members of the Socio-economic Burden of Hospital
	Acquired Infection Project Team, Steering Committee and
	Advisory Group431
Appendix 9	Data variables included in the data analysis434
Appendix 10	The age, sex, admission type and specialty distribution of
	study participants and eligible patients who were not
	recruited into the study438
Appendix 11	The incidence of HAIs by primary operative procedure444

Appendix 12	Results of the analysis that assessed how hospital costs varied with site of infection and selected patient characteristics
Appendix 13	The distribution of hospital costs incurred by patients with specific types of infection
Appendix 14	Results of the analysis that assessed how length of hospital stay varied with site of infection and selected patient characteristics

MATERIAL NOT BOUND IN THESIS:

Plowman R, Graves N, Griffin M, Roberts JA, Swan AV, Cookson B, et al. The socio-economic burden of hospital acquired infection. London: Public Health Laboratory Service, 1999. Executive summary.

Plowman R, Graves N, Griffin M, Roberts JA, Swan AV, Cookson B, *et al.* The rate and cost of hospital acquired infections occurring in patients admitted to selected specialties of a district general hospital in England and the national burden imposed. *Journal of Hospital Infection* 2001;47(3):198-209.

Plowman R, Graves N, Esquivel N, Roberts J. An economic model to assess the cost and benefits of an intervention to reduce the risk of catheter related nosocomial urinary tract infections. *Journal of Hospital Infection* 2001;4(1):33-42.

ABBREVIATIONS

ASA	American Society of Anaesthesiologists scores
BMI	Body mass index
BSI	Bloodstream infection
CDC	Centre for Disease Control
CI	Confidence interval
COI	Cost of illness
DGH	District general hospital
HAI	Hospital acquired infection
HICC	Hospital Infection Control Committee
ICU	Intensive care unit
ICN	Infection control nurse
ICT	Infection control team
IR	Infection rate
LOS	Length of stay
LRTI	Lower respiratory tract infection
OPCS	Office of population censuses and surveys
OR	Odds ratio
PAS	Patient administration system
RCT	Randomised controlled trial
SENIC	Study of the efficacy of nosocomial infection control
SWI	Surgical wound infection
UTI	Urinary tract infection

.

CHAPTER 1 INTRODUCTION

1.1 Introduction

Hospital acquired infections (HAI) are infections that are acquired during a patient's hospital stay. They are not a new problem; however, in recent years there has perhaps been greater awareness of the scale of the problem and the health and economic consequences that result from HAI.

In the UK, the issue of HAI has been the subject of a number of research studies conducted over the past few years, including the research on which this thesis is based.¹⁻³

In 1992 the Department of Health commissioned the Public Health Laboratory Service to conduct an audit of infection rates, and infection control polices and practices in 19 hospitals in England and Wales, the results of which were published in 1997.¹ The audit drew attention to the problem of HAI, and highlighted considerable variation in infection rates occurring in similar patients treated in similar settings, and variations in infection control policies and practices.

At the same time the Department of Health commissioned the Public Health Laboratory Service and the London School of Hygiene and Tropical Medicine to conduct a study of the socio-economic burden of HAI. The results of this work were published in January 2000 and are reported in some detail in Chapter 4 of this thesis. The research drew attention to the substantial burden these infections place on scarce health sector resources and on patients and their carers.²
This work was used a little later in 1998 by the National Audit Office in their study of HAI. The study examined a number of issues including the scale of the problem, the rise of antibiotic resistance, the level of resources allocated to infection control within NHS Trusts in England, the resources available to infection control teams, and the potential benefits of investment in infection control practices. The results were published in February 2000.³ The results were important in that they again served to highlight the very real problem of infection in hospitals, and the apparent lack of resources available to infection control teams to deal with the problem. The report was presented to the Parliamentary Select Committee of Public Accounts and the subject of a hearing in March 2000, following which a report was issued.⁴ This served to further raise the profile of the problem of HAI, bringing it to the attention of the government, policy makers, health care professionals and the general public. In response the Department of Health has introduced a number of policy initiatives aimed at strengthening infection prevention and control activities within the NHS.⁵

1.2 The scale of the problem

Prevalence studies conducted in England and Wales in 1980⁶ and again in 1993/94⁷ found that at any one time an estimated 9% of hospital in-patients had an infection that they had acquired after admission. At the same time, it is likely that many more patients discharged from hospital had an infection related to a recent hospital admission. More recently the study of the socio-economic burden of HAI estimated that at least 321,000 patients admitted to NHS hospitals in England acquired one or more HAIs in 1994/5.⁸ This estimate is based on the incidence of HAI observed in the study and relates to the number of infections occurring in adult non-day case patients admitted to the medical and surgical specialties covered in the study at NHS hospitals in England (an estimated 70% of adult non-day case admissions), and is further limited to infections which present during the in-patient period. Thus the actual number of patients acquiring one or more HAIs is likely to be considerably higher.

36

1.3 The impact of HAI on the individual

The acquisition of an infection in hospital may have health and financial consequences for affected patients. The impact on health status and recovery will vary considerably from one patient to another depending on the type of infection acquired, the underlying health status of the patient, the effectiveness of treatment given and how these various factors interact. For example, in some cases the impact may be limited to minor discomfort, but in other cases the acquisition of an infection may prolong recovery, cause temporary or permanent disability and may directly cause or substantially contribute to a patient's death. An estimated 5000 deaths every year are thought to be directly caused by an infection acquired in hospital, and HAIs are thought to contribute to a further 15,000 deaths.⁹

In terms of the economic consequences for the individual, acquiring an infection in hospital may result in an increase in out of pocket expenditure on health related items, and a reduction in income due to a delayed return to employment or in some cases the failure to return to employment. Few studies have estimated the magnitude of these costs. Some notable exceptions include a study by Farbry *et al* (1982)¹⁰ that indicated surgical wound infections (SWIs) delayed return to work, and the study on which this thesis is based which assessed the impact HAIs had on both patients and their carers. ²

1.4 The impact on the health sector

The acquisition of an infection in hospital may result in the additional use of hospital and primary health care resources. Additional hospital costs may result from an increased need for investigations and procedures, increased dispensary demands, additional nursing and medical care and a prolonged inpatient stay. Following discharge, patients who have acquired an infection in hospital may require a greater number of hospital appointments, and/or appointments with primary health professionals, including general practitioners, practice nurses and district nurses, than they would have required in the absence of an infection.

37

A number of studies have assessed the economic burden HAIs place on the health sector. These studies vary in scope. The majority concentrate on the cost to the hospital sector resulting from increased in-patient care, and relatively few look at costs to the health sector post-discharge. The types of infections included vary, some focussing on all types of HAI occurring in selected patient groups, while others focus on a specific type of infection. The methods used to identify, measure and value resources used and attribute them to HAIs also vary, as do the estimates derived. However, despite variations in scope and the methods employed all point to the substantial burden these infections place on health sector resources. The socio-economic burden of HAI study estimated that HAIs cost the health sector in England at least £1 billion in 1994/5.²

1.5 Why patients acquire an infection

Infection occurs as a result of complex interactions between potential pathogens (bacteria, virus, fungus or protozoan) and the host, in this case the patient. Within the hospital environment patients are at particular risk of infections. Patients whose primary illness compromises their immune system, such as AIDS and various forms of haematological malignancy, are particularly susceptible, as are the young and the elderly. Treatment regimens, such as chemotherapy, will also render a patient susceptible to infection and invasive procedures including operative procedures, and the insertion of intravenous and intra-arterial catheters, urinary catheters and endo-tracheal tubes offer direct access for micro-organisms. In addition the patients normal flora is frequently replaced with hospital flora that is often resistant to antibiotics and so may set up a potential source of infection.

1.6 The scope for prevention

Whilst not all HAIs are preventable the evidence indicates that a proportion of infections can be prevented through improvements in infection prevention and control activities. The early results from the National Nosocomial Infection Surveillance Scheme provided strong evidence that this is the case. Infection rates occurring in patients who had specific surgical procedures were found to

vary markedly from one hospital to another and the observed variation in rates continued to be present after adjustment for case mix differences had been made.⁵ These results suggest that the variation observed to some extent reflected differences in clinical practice and serve to highlight the potential for a reduction in infection rates.

Quantifying the level of infections that can be prevented is difficult: it is difficult to quantify an event that does not occur. The most frequently quoted estimate, taken from the results of a comprehensive study conducted in the US (the Study of the Efficacy of Nosocomial Infection Control - SENIC), suggests that 30% of HAIs could be prevented.¹¹ However, caution should be exercised before applying this estimate to infections occurring in the UK. The estimate was derived from a study conducted over 20 years ago in the US. Since that time there have been many changes in the treatments and care options offered, and the case mix of patients treated. As such there may be important differences in the risk profile of patients treated today, compared to that of over 20 years ago. Nevertheless the results demonstrated that there is scope for a reduction in rates.

The results of a National Audit Office survey of NHS hospitals in England, documented in their report 'The Management and Control of Hospital Acquired Infections in Acute NHS Trusts in England' published in 2000, further highlight the fact that there is scope for a reduction in infection rates.³ As part of their survey, infection control teams (ICTs) were asked what proportion of infections they believed could be prevented. The responses ranged from 5% to 40%. The responses were subsequently weighted by the number of beds in the responders' hospitals and a 'bed weighted' average of 15% derived. Whilst this is somewhat less than the US estimate, it does serve to highlight the belief that preventable remains unknown.

1.7 The potential benefits of improved prevention

Quantifying the benefits of prevention is difficult. Difficulties are associated with both identifying and valuing infections averted. It is difficult to measure the benefits of an event that did not take place. However, estimates derived from the results of the socio-economic study suggest that the benefits of prevention are likely to be substantial.² This point was highlighted recently in the NAO report, which estimated that if a 15% reduction in rates could be achieved this could lead to the release of health sector resources valued at £150 million per annum.³ It should be stressed that this estimate reflects the value of resources, which might be released for alternative use. It does not represent potential cash savings. However, given that the NHS is working above capacity in most sectors HAIs have an opportunity cost. The estimates represent the gross benefits of prevention. Net benefits will depend on the cost of interventions introduced to achieve such a reduction.

1.8 The rationale for this thesis and the anticipated contribution it will make

At the time of undertaking this thesis relatively little was known about the cost of HAIs to the health sector in England. It was clear from the results of the 1980 National prevalence study that HAIs affected a large number of patients every year: 9.2% of patients at any one time were found to be affected in 1980.6 However, data on the magnitude of the economic burden these infections imposed on the health sector, and indeed on patients and carers were lacking. Data from a number of international studies, and in particular the results of work stemming from the SENIC study conducted in the US indicated that HAIs were likely to be placing a substantial burden on limited NHS resources and on patients and carers. Estimates suggested that HAI cost the health sector in the US at least \$4 billion per annum in 1985.12 Whilst studies had been conducted in the UK,13 14 these were either relatively small, or rather specific in nature relating to particular patient groups, and were undertaken some years ago. A more comprehensive estimate of the magnitude of the burden of HAI was needed. The only national estimates available at the time were an estimate derived by the Department of Health which, based on a number of broad

40

assumptions, estimated that HAIs cost the health sector £111 million per annum,¹⁵ and an estimate derived by Coello *et al*¹⁴ which indicated that HAIs occurring in surgical patients cost the health sector in England an estimated £170 million in 1993.

It was against this background, and in the knowledge that a proportion of HAIs were preventable, that the Department of Health in October 1992 commissioned a study of the socio-economic burden of HAI. This thesis is based on this study. The socio-economic burden study aimed to provide a comprehensive assessment of the costs resulting from HAIs occurring in adult, non-day case patients admitted to specialties that are common to most hospitals. This was to include costs to the health sector as a result of additional in-patient care, costs to the health sector arising post-discharge and costs to patients and carers. It was anticipated that the results would raise awareness of the magnitude of the burden of HAIs, and the potential gross savings that might result from improved prevention and control. Through this process it was hoped that a greater understanding of the economic burden resulting from these infections would be gained, and that the information generated would assist in policy formation and inform clinical practice in relation to infection prevention and control. Further details about this study are presented in Chapter 4.

This thesis is closely linked with the socio-economic burden of HAI study. It focuses on a sub-section of data relating to adult non-day case surgical patients. The thesis assesses the incidence and economic impact that HAIs that present during the patient's hospital stay have on the hospital sector as a result of additional in-patient care. It also explores possible risk factors for these infections and considers how the information derived may be used to demonstrate the potential benefits of investment in prevention and inform prevention strategies.

Surgical patients were selected as the population of interest as it was felt that it would be interesting and beneficial to explore in greater detail the incidence and impact of HAIs occurring in this patient group. The underlying study assessed the incidence and economic burden of HAIs occurring in patients admitted to selected medical and surgical specialties taken together, and also the incidence and burden of infections occurring in patients admitted to each medical and surgical specialty. No attempt was made to assess the incidence and economic burden of HAIs occurring in patients admitted to surgical specialties taken together. A further reason for selecting surgical patients was an interest in exploring risk factors for infections in this patient group.

As indicated above this thesis focuses on the incidence of HAIs presenting during the in-patient period and the economic impact that HAIs had on the hospital sector as a result of additional in-patient care. The decision to focus on infections that presented during the in-patient period, and their impact on the hospital sector was informed by the results and inherent limitations of the earlier study.

Estimates of the incidence of infections presenting post-discharge, derived in the earlier study, were based on the responses given to specific questions within a detailed questionnaire sent to a proportion (approximately one third) of patients one month post-discharge. The estimates derived indicated that a proportion of patients experienced symptoms suggestive of a surgical wound, urinary and/or respiratory tract infection. However, it was not possible to say whether the symptoms reported represented actual infections, or whether the infection, if present, was hospital acquired. Furthermore, since only a small proportion of patients were followed up there was considerable uncertainty surrounding some of the estimates derived and it was clear that there were insufficient data available to allow for further analysis limited to a subset of data relating to surgical patients from which estimates could be derived with a degree of certainty. Similarly, whilst the underlying study's estimates of the impact HAIs had on hospital costs as a result of additional in-patient care were based on data on the hospital costs incurred by all study participants, estimates of the costs falling on the primary health care sector, patients and carers were only derived for a proportion of patients. The results of the analysis of the impact HAIs had on these other areas of costs clearly showed that HAIs imposed a burden on the primary sector, patients and carers. However, as with the estimates of the incidence of HAI presenting post-discharge, there was considerable uncertainty surrounding some of the estimates derived and it was again clear from the results derived that there were insufficient data available to allow for further analysis limited to surgical patients from which estimates could be derived with a degree of certainty.

Similar reasons informed the decision not to examine the impact HAIs occurring in surgical patients had on health status. The underlying study's assessment of the impact HAIs had on health status was based on an analysis of responses given to the SF-36, included as part of the questionnaire, administered to a proportion of patients post-discharge. The results of this analysis indicated that HAIs had a negative impact on health status as measured by this instrument. However, again there was uncertainty surrounding the estimates derived and it was clear that there were insufficient data available to justify further analysis limited to a subset of data relating to surgical patients from which robust estimates could be derived.

The thesis aims to add to the findings of the socio-economic burden study by providing a more detailed account of the incidence of HAIs occurring in this patient group than presented in the underlying study report and, through the use of slightly different methodology, explore the economic burden these infections place on the hospital sector in more detail. It is hoped that this piece of work will further contribute to our understanding of the economic burden these infections impose, how costs can be attributed to infection and possible risk factors for infection. The thesis will subsequently show how information on the economic burden imposed by these infections might be used to demonstrate the benefits of investment in infection prevention and control and how information on costs and possible risk factors can be combined and used to inform policy and practice.

1.9 My role in the socio-economic burden study

In September 1993 I was appointed project leader for the socio-economic burden of HAI study. As project leader I was responsible for the day to day management of the project, developing and implementing the outline protocol, and writing up and disseminating the results of the study. During the course of the study my role included a wide range of activities summarised below.

1.9.1 Literature review

On appointment I conducted a review of the literature on the epidemiology and economic burden of HAI. This review was updated at regular intervals throughout the course of the study.

1.9.2 Establishing links with staff at the study hospital

At an early stage I established links with staff at the study hospital including the chief executive, director of clinical practice, director of nursing practice, clinical directors, business managers, senior nurse managers, nursing staff on each of the study wards, consultants and their medical teams, laboratory staff, members of the infection control team and infection control committee, general practitioners, primary health care managers, district nurse patch managers and staff from the medical records, medical coding, finance and information technology departments. This involved writing to selected individuals, setting up a meeting and subsequently meeting with staff to discuss the aims and objectives of the study and the methods to be employed. The majority of these meetings were on a one to one basis. However, when both appropriate and possible, group meetings and seminars were held. This latter approach was particularly useful when informing ward staff of the project.

1.9.3 Further development and implementation of the study protocol

As project leader I was responsible for the further development and implementation of the study protocol. This included the development of patient and relative information sheets, consent forms, procedures for recruiting patients into the study, and appropriate data collection methods. The latter included the development of data collection sheets and patient questionnaires. This was done in consultation with members of the steering and advisory groups, other researchers and where appropriate members of the study hospital staff. Procedures for recruitment and data collection were written up and circulated to steering group and advisory group members for comment and subsequent approval, together with patient and relative information sheets and consent forms and all other data collection forms.

1.9.4 Selection and implementation of an appropriate data entry system

It was clear from the outset that the data requirements of this study were considerable. As project leader I was responsible for investigating the various data entry systems available and presenting the options to the steering committee, together with recommendations as to which would be most appropriate and why. A variety of systems were explored including paper questionnaires, and subsequent data entry either by project staff or a data entry company; the use of hand held computers; and the use of scanning software developed by Formic Ltd. After a thorough investigation of a number of options, the latter system was selected and purchased. I was subsequently responsible for setting up the scanning data entry system. This involved creating questionnaires and data collection surveys with the scanning software, setting up each individual question within the questionnaire ready for scanning and subsequently testing the scanning process. Testing involved checking that the forms were being read correctly and that the process was reliable. This proved to be a very time consuming process. There were a number of problems with the early versions of the software that had to be corrected by the software

manufacturer. This process led to delays in the development of the data collection forms and used a considerable amount of my time.

1.9.5 Obtaining the approval of the relevant ethical committees prior to the pilot study

As project leader I was responsible for submitting all relevant documentation to the ethical committee at the study hospital for their consideration, and subsequently responding to any queries that they had.

1.9.6 Conducting a pilot study and revising the methods in response to the findings of the pilot study

As project leader I was responsible for piloting the proposed methods. A small pilot study was conducted. This identified a number of problems with the draft data collection sheets and the proposed methods. A report was drafted and presented to the steering committee together with recommendations for changes and revisions to the data collection methods and data collection tools. Changes were agreed and subsequently made to the relevant forms and procedures. A report summarising the outcome of the pilot study, was subsequently presented to the Advisory Committee.

1.9.7 Re-submitting the study protocol and data collection forms for ethical approval prior to the main study

Following the pilot study the protocol and data collection forms had to be resubmitted to the study hospital ethical review committee for their consideration prior to the main study. As project leader I was responsible for this process and for responding to any subsequent queries that they had.

1.9.8 Recruiting and training six research assistants

As project leader I was responsible for the recruitment and training of six research assistants. Five research assistants were appointed in January 1994 and a further research assistant was appointed in June 1994 when more

funding became available from the Department of Health. All six research assistants were qualified nurses.

It was my responsibility to ensure that the research assistants received adequate training prior to the main data collection period, and further training as and when required. This was achieved through a training programme developed and administered prior to the main data collection period, and additional training given as necessary. A copy of the training programme can be found in Appendix 1. In addition, a seminar course was developed for the research assistants (see Appendix 1). Seminars were held on alternate weeks throughout their period of employment and covered topics such as the economics of HAI, the epidemiology of HAI, study design, basic statistical methods and health policy. Lecturers included myself, other members of the project team and steering committee and external speakers.

1.9.9 Managing the main data collection and entry period

As project leader I was responsible for managing the data collection and data entry process. A variety of techniques were adopted to assist in this process. As discussed above protocols setting out the data collection and entry processes were developed and agreed by the steering committee. The research assistants received training in all aspects of the study. A document outlining the standards to be applied during the data collection period was drafted and approved by the steering committee (see Appendix 2). Each research assistant received a copy of this document which they were subsequently asked to read, discuss and sign. At the outset the research assistants received close supervision. As the project progressed input by myself was reduced. However, the research assistants were free to contact me at any time should they have a query, and I ensured that I met with the research assistants at least once a week at a team meeting.

1.9.10 Appointing a secretary

At the outset the funding available did not allow for the appointment of secretarial support to assist me as project leader, and the other members of the project team. As such, with the exception of some help with a couple of mail merges, all secretarial and administrative work was undertaken by myself. It was clear that some support was needed, and a document outlining the need for additional funds to cover the cost of secretarial support was submitted to the Department of Health. The Department agreed to make available additional funding for administrative support and a secretary was appointed in April 1994. Further support was subsequently employed on an ad hoc basis.

1.9.11 Liaising closely with the research economist and assisting in the development of unit costs for resources used

The project team included a research economist who was primarily responsible for costing resources used by infected and uninfected patients. As project leader, I was a member of the interview panel for the appointment of the research economist. I subsequently worked closely with the economist on all aspects of his work, and assisted in the development and implementation of methods used to cost various resources. For example, I worked closely with the economist to develop unit costs for drugs administered, and procedures performed and I was solely responsible for developing costs for nursing care administered to patients based on the amount of care patients received during the course of their hospital stay.

1.9.12 Liaising closely with the project statisticians

Whilst data analysis was primarily the responsibility of two statisticians, I worked closely with both throughout the study. My role involved assisting in the development of an appropriate strategy for data cleaning, actively engaging in the data cleaning process, and assisting in the development and implementation of an appropriate approach to the analysis of data. As the analysis was undertaken I reviewed and commented on all results, made

suggestions as to how the analysis could be modified, and contributed to the indepth discussion of the results that followed.

1.9.13 Ensuring that the work is completed within an appropriate time frame

At an early stage it was apparent that, given the scope of the project, it would not be feasible to complete the study within the two years funding available. As such I was responsible for drafting a report requesting an extension to the time frame and additional funding from the Department of Health. The report was well received and an extended time frame together with additional funding was successfully obtained from the Department of Health.

1.9.14 Writing the final report and submitting it to the Department of Health

The final report consisted of four documents: parts I and II, separate appendices, and two stand alone executive summaries that varied in length.² Part I included background information, details of the methods used, the results of the study, a discussion of the results, conclusions drawn and recommendations made. Part II included detailed information about the methods used to derive unit costs for resources used. As project leader I was responsible for drafting Part I and the two executive summaries. The research economist primarily drafted part II and the appendices were the responsibility of both the research economist and myself. A copy of the executive summary can be found at the back of this thesis.

Over the course of the project chapters were drafted and circulated to the project Steering Committee for comment and subsequent approval. Drafts were also submitted to the Advisory Committee for comment, and later the Department of Health. A final draft which had been approved by the project Steering Committee was submitted to the Department of Health in August 1997, and distributed for internal and external review. A full set of reviewers comments was received in November 1997. Informal discussion followed and

a formal response submitted to the Department in February 1998 following which there was a further period of discussion and debate. Further editing work and additional analysis was then undertaken. Amendments to the original text, together with the results of the additional analysis requested were submitted to the Department of Health in September 1998 and the report was released for publication in September 1999 and subsequently published in January 2000.

1.9.15 **Preparation of the report for publication**

The decision was taken that the full report should be published and made available to a wide audience. As such it was agreed that the input of an editor and publication team should be sought. I was subsequently responsible for working with an editor and the Public Health Laboratory production team to produce an edited version of the report.

1.9.16 Liaising with the Department of Health

Throughout the duration of the study I was responsible for liaising with the Department of Health and keeping them informed of any new developments and overall progress. This was done through telephone discussions, letters, reports and presentations made at the advisory committee meetings held regularly over the course of the project.

1.9.17 Liaising with the National Audit Office in relation to their work on HAI

Whilst the cost study was in progress the National Audit Office embarked on a study of HAI. As part of their initial work they approached myself, Jenny Roberts and Nick Graves to discuss the aims and objectives of the cost study. In several subsequent meetings we discussed the questionnaire they were planning to use, advised on a number of issues, updated the NAO about the progress with our study and endeavoured to ensure that any information relating to the cost study was presented appropriately in the NAO report – 'The management of HAI in acute NHS Trusts in England.'³

1.9.18 Dissemination of the findings

The final report was launched on February 17th 2000. I was actively involved in this process, working closely with the press officers from the three institutions involved in the research and the Department of Health. I assisted in drafting the press releases and the organisation of the press briefing. As mentioned above the report received considerable media attention, including wide coverage in all the major newspapers. Following the launch I gave a number of interviews for both national and local television and radio stations. The report was subsequently sent to all consultant microbiologists, infection control teams and public health physicians. I was responsible for drafting the covering letter and overseeing this process.

The main results were also reported in a paper published in the Journal of Hospital Infection.⁸ I was responsible for drafting this paper, circulating it to the authors for comment and subsequently revising it as necessary and submitting it to the relevant journals.

During the course of the study and since the report's publication I have made a number of presentations to conferences and seminars. Appendix 3 provides details of presentations made over the course of the study.

1.10 My role in this thesis

The preceding section has outlined the key activities undertaken by myself as project leader for the socio-economic burden of HAI study. All the activities listed are directly relevant to this thesis. Additional activities undertaken for the purposes of this thesis include, a more detailed literature review, further cleaning of the data set, further data analysis and the development of an economic model to asses the costs and benefits of investment in prevention. These activities were all undertaken by myself.

1.11 Aims and objectives of this thesis

1.11.1 Aim

To assess the incidence of, and independent risk factors for HAIs occurring in surgical patients admitted to a district general hospital and the impact these infections have on the secondary health care sector, and to examine how the information obtained may be used to assess the potential benefits of investment in the prevention and control of HAIs.

1.11.2 Objectives

The specific objectives were to:

- 1. Review the literature on the epidemiology of HAI, risk factors for HAI, and the economic evaluation of HAI.
- Determine the incidence of HAIs occurring in adult, non-day case patients admitted to selected surgical specialties of a district general hospital.
- 3. Explore how the incidence of HAI varies with selected patient characteristics and identify possible risk factors.
- 4. Determine the impact HAIs occurring in this patient group have on secondary health care sector resource use and costs.
- 5. Examine how information on the economic burden of HAIs may be used to assess the potential benefits of investment in the prevention and control of HAIs.

1.12 The structure of the thesis

The thesis begins with a review of the literature. The literature on the epidemiology of HAI is reviewed in Chapter 2. The chapter covers four broad areas: the frequency and distribution of HAI; the mortality risk associated with acquiring an infection in hospital and the number of patients estimated to die from HAIs per year; the aetiology of HAI and specific risk factors for infection, and the problem of antibiotic resistance.

The literature on the economic evaluation of HAI is reviewed and discussed in Chapter 3. The chapter includes a discussion of the methodological issues associated with the economic evaluation of HAI, followed by a review of studies that have estimated the burden of HAI and those studies that have assessed the benefits of prevention.

An overview of the study that examined the socio-economic burden of HAI on which this thesis is based is subsequently presented in Chapter 4, and the methods which are specifically related to this thesis are presented in Chapter 5.

The results of this work are presented in Chapters 6, 7, 8 and 9. Chapter 6 provides an overview of the sample characteristics. Chapter 7 presents the results of the analysis that examined the incidence of HAI and specific risk factors for infection. Chapter 8 presents the results of the economic analysis which examined the impact of HAI on hospital costs and Chapter 9 the results of the analysis that assessed the impact of HAI on length of hospital stay.

How the results of this work might be used to inform clinical practice, is then explored in Chapter 10. Estimates of the gross benefits of prevention at the level of the study hospital and at the national level are presented, together with a framework for assessing the potential net benefits of investment in infection prevention and control activities. A worked example of a simple model of the costs and potential benefits of investment in infection prevention and control activities is presented.

53

Finally, the results of this study, methodological considerations and the implications for infection prevention and control policy and clinical practice are discussed in Chapter 11 and conclusions drawn.

CHAPTER 2 THE EPIDEMIOLOGY OF HOSPITAL ACQUIRED INFECTION

2.1 Introduction

In this chapter the literature relating to the epidemiology of hospital-acquired infection (HAI) is reviewed. An overview of the epidemiology of HAI in terms of its frequency and distribution, the mortality risk associated with acquiring an infection, and the number of patients estimated to die from these infections every year is provided. The aetiology of HAI including a discussion of specific risk factors for HAI and the problem of antibiotic resistance are also examined. The focus of this chapter will be the problem of HAIs occurring in surgical patients. However, in order to put the problem in context, information relating to the overall problem of HAI will also be presented where appropriate.

The literature reviewed was identified through a series of consecutive searches, carried out during the period 1993 - March 2003, using the computerised bibliographic databases Medline, and PubMed. These two bibliographic databases were selected as they cover the major journals of relevance to this These searches were supplemented by reference follow up, hand thesis. searching of selected journals, and consultation with experts in the field. The search was limited to papers published in the English language during the period 1975 - March 2003, and further limited to studies conducted in developed European countries, the United States of America, Canada and Australia. This time period was selected, as in the mid 1970s the US study of the Efficacy of Nosocomial Infection Control (SENIC), was published representing a time of new understanding and interest in HAIs. As such it was considered appropriate to review literature published from the mid-1970s. The search strategy involved the use of selected key words (thesaurus terms) and combining these with a number of "free text" words. The key words used were: hospital acquired infection; nosocomial infections; healthcare associated infections; and hospital associated infections. These were combined with the following "free text" words: prevalence; incidence; risk factors; risk indices; antibiotic resistance; antimicrobial resistance; mortality; morbidity and deaths. The abstract of each paper was subsequently read and a decision made as to whether the paper was relevant to this review. Those papers that were concerned with HAIs occurring in adult non-day case patients cared for in specialties common to most hospitals (medicine, surgery, urology, care of the elderly, orthopaedics, gynaecology and obstetrics) were considered to be relevant, as were papers that included both day case and in-patients and papers that included both patients admitted to the specialties listed above as well as those admitted elsewhere. Papers that were limited to day case patients, children, or patients admitted to specialties not included in this study, were excluded from the review.

References cited in papers which had not been identified through the computerised search were subsequently followed up. Hand searches of the following journals were also carried out: Journal of Hospital Infection; Infection Control; American Journal of Infection Control, Epidemiology and Infection Control; Current Issues in Infection Control; and the British Medical Journal. Experts in the field including microbiologists and infection control specialists from the Central Public Health Laboratory Service, at Colindale in North London, and selected NHS Trusts were also consulted to identify grey literature such as project reports and policy documents of relevance to this review.

Some papers identified as part of the review of studies that assessed the economic burden of HAI are also included in this section, since they not only reported data on the cost of infection but also the incidence of HAIs and the number of deaths occurring in infected patients. The results of the socio-economic burden of HAI study, on which this thesis is based, are reported separately in Chapter 4.

2.2 The frequency and distribution of HAI

Data on the frequency and distribution of HAI are available from a variety of sources. A review of the literature indicates that a number of studies have assessed the frequency of HAI either in terms of its prevalence or incidence (see sections 2.3 and 2.4). Many hospitals also routinely collect data on the frequency of selected HAIs, and many participate in national surveillance schemes, the results of which, in some cases, have been published. A recent National Audit Office survey of infection control indicated that in 1998 ~94% of infection control teams surveyed carried out some form of surveillance.³ Epidemiological, clinical audit and hospital surveillance studies provide useful insights into the extent of the problem of HAI in terms of the frequency of the problem and, in many cases, also provide data on how infection rates vary with specialty, age and other patient characteristics. However, valid comparisons of the results of studies conducted in different settings and over time are difficult. The methods used to determine the incidence or prevalence of HAI employed in the different studies vary in important respects, which in turn impact on the results obtained. For example, studies vary in terms of the case definitions and case ascertainment methods used, the formula used to calculate rates, the types of infections included and the case mix studied. These differences are discussed in detail in sections 2.3 and 2.4, in the context of studies that have assessed the prevalence of HAI (section 2.3) and the incidence of HAI (section 2.4).

2.3 Prevalence of HAI

Prevalence figures are a measure of the proportion of individuals in a population with a specific disease at a single point in time (point prevalence rate) or over a period of time (period prevalence rate) (figure 2.1). Point prevalence figures are more common and have generally been used in studies of the prevalence of HAI. They are usually expressed as a percentage.

Figure 2.1 Prevalence rate formulae

1.	Point PR = no. persons with a disease at particular point in time	
	total population	
2.	Period PR = total no. persons with disease at some time during the specified period	
	total population at mid-point of interval	
PR	= prevalence rate	

Multicentre prevalence studies have been conducted in a number of countries. Table 2.1 provides an overview of prevalence studies that have been conducted in developed European countries, the US and Australia since 1975. These studies provide important information on both the overall problem of HAIs and the prevalence of infection among specific patient groups.

ā	
la	
Ist	
A	
put	
3	
ad	
an	
С С	
Š	
the	
e,	
ē	
ШС	
2	
Ţ	
cte	
ą	
Š	
ŏ	
ies	
Pn	
st	
2.	
P	
Ē	
4	
ä	
Ę	
tiel	
pat	
Ð	
ŝ	
fe	
P	
5	
A	
Ť	
0 0	
õ	14
le	Ť
ev:	0
Ā	
	1
Ξ	
6 7	
ğ	
Ļ	

Country	Study author and year of publication	Year of study	No. Hospitals	No. patients	Types patients	Definitions used	Prevalence of infections (%)	Prevalence of infected patients (%)
Sweden	Bernander <i>et al</i> (1978) ¹⁶	1975	S	4,246	All patients	Based on CDC 1972 definitions ²³	17.0	
Denmark	Jepsen et al (1980) ¹⁷	1978	55	1,363	All patients	Based on CDC 1972 definitions ²³		10.4
Denmark	Jepsen et al (1980) ¹⁷	1979	25	1,557	All patients	Based on CDC 1972 definitions ²³	•	12.1
England & Wales	Meers et al (1981) ⁶	1980	\$ 3	18,163	Al patients	Study definitions	9.2	•
Norway	Hovig et al (1981) ¹⁸	1980	15	7,833	A patients	Based on CDC 1972 definitions ²³	9.0	•
Italy	Moro et al (1986) ²⁴	1983	130	34,577	All patients	CDC 1972 ²³	•	6.8
Belgium	Mertens et al $(1987)^{26}$	1984	106	8,723	Surgical or ICU*	Based on CDC 1972 definitions ²³	10.3	9.3
Czechoslovakia	Sramova et al $(1988)^{22}$	1984	8	12,260	All patients	World Health Organisation 1981 26	6.1	•
Australia	McLaws et al (1988) ²⁷	1984	569	28,643	All patients	Based on CDC 1972 definitions ²³	8.1	•
Spain	EPINE Working Group (1992) ¹⁹	1990	123	38,389	All patients	CDC1988 28	9.9	8.5
Spain	Vaque et al (1999) ²¹	1990	123	38.489	Al patients	CDC 1988 26 + CDC modified SSI (1992) with the exception of asymptomatic bacteruria which was not included in the UTI category	0.0	8
Spain	Vaque et al (1999) ²¹	1991	135	42.185	Al patients	CDC 1988 ²⁶ + CDC modified SSI (1992) with the exception of asymptomatic bacteruria which was not included in the UTI category	6	7.8
Norway	Avitsland et al (1992)	1991	76	14,977	Al patients	CDC 1976	6.3	•
France	Sartor et al (1995) ²⁵	1992 (May)	Ø	1,220	All patients	CDC 1988 28	•	8.6"
France	Sartor et al (1995) ²⁵	1992 (Nov)	•••	1,389	Al patients	Conseil superieur d'Hygiene Publique de France (similar to CDC definitions)	1	7.1**
Spain	Vaque et al (1999) ²¹	1992	163	44.343	All patients	CDC 1988 + CDC modified SSI (1992) with the exception of asymptomatic bacteruria which was not included in the UTI category	8.5 2.5	7.3
UK and Republic of Ireland	Emmerson et al (1996) ⁷	1993/4	157	37,111	All patients	Study	•	9.0
Spain	Vaque et al (1999) ²¹	1993	171	46,983	Al patients	CDC 1988 + CDC modified SSI (1992) with the exception of asymptomatic bacteruria which was not included in the UTI category	80 4	7.1
SSI - survical site i	nfection. UTI - urinary tract infection							

*Surgical site, blood and urinary tract infections only ** Not clear if prevalence figure relates to prevalence of infections or patients - the latter has been assumed Table 2.1 continued: Prevalence of HAIs and infected patients as found in studies conducted in Europe, the US, Canada and Australia since 1975

Country	Study author and year of publication	Year of study	No. Hospitals	No. patients	Types patients	Definitions used	Prevalence of infections (%)	Prevalen ce of infected patients (%)
Crete	Gikas et al (1999) ³¹	1994	Ø	1,305	All patients	C CDC 1988 ²⁶ + CDC modified SSI (1992) ²⁶ with the exception of asymptomatic bacteruria which was not included in the UTI category	6.8 8	•
Spain	Vaque et al (1999) ²¹	1994	186	49,689	Al patients	CDC 1988 $\frac{36}{20}$ + CDC modified SSI (1992) $\frac{36}{20}$ with the exception of asymptomatic bacteruria which was not included in the UTI category	8.3	7.3
Germany	Gastmeier et al (1998) ²²	1994	22	14,996	All patients	CDC 1988 2	3.6	3.5
Crete	Gikas <i>et al</i> (1999) ³¹	1995	Ø	1,386	All patients	CDC 1988 $\frac{26}{3}$ + CDC modified SSI (1992) $\frac{26}{3}$ with the exception of asymptomatic bacteruria which was not included in the UTI category	S.S	•
Spain	Vaque et al (1999) ²¹	1995	201	51,339	All patients	CDC 1988 26 + CDC modified SSI (1992) 26 with the exception of asymptomatic bacteruria which was not included in the UTI category	8.0	6.9
Switzerland	Pittet et al (1999)	1996	4	1,349	Medical, surgical & ICU	CDC 1988 ²⁶ with the exception of bacteriuria which was not categorised as an HAI	13.0	11.6
Crete	Gikas <i>et al</i> (1999) ³¹	1996	ŵ	1,279	All patients	CDC 1988 $\frac{26}{3}$ + CDC modified SSI infection (1992) $\frac{26}{3}$ with the exception of asymptomatic bacteruria which was not included in the UTI category	5.0 1	•
Spain	Vaque et al (1999) ²¹	1996	206	51,961	All patients	CDC 1988 26 + CDC modified SSI infection (1992) 26 with the exception of asymptomatic bacteruria which was not included in the UTI category	8.4	72
France	Astagneau et al (1999) ³⁴	1996	ន	6,389	Al patients	Based on CDC 1988 ²⁸ + CDC modified SSI (1992) 28		9.3
France	The French Prevalence Survey Study Group (2000) ³⁶	1996	830	236,334	All patients	Study	7.6	6.7
Norway	Scheel et al (1999) ³⁶	1997	71	12,755	All patients	Based on CDC 1988 2	6.1	l
Spain	Vaque et al (1999) ²¹	1997	214	51,674	Al patients	CDC 1988 $\frac{26}{3}$ + CDC modified SSI (1992) $\frac{26}{3}$ with the exception of asymptomatic bactenuria which was not included in the UTI category	8.1	0 [.] 07
Italy	Pavia et al (2000) ³⁷	1999	9	888	All - except psychiatric	CDC 1988 2 + CDC modified SSI(1992) 2	1.7	ſ
Greece	Gikas et al (2002) ³⁶	1999	14	3,925	All patients	Based on CDC 1988 $\frac{3}{2}$ + CDC modified SSI (1992) $\frac{3}{2}$	9.3	8.6
SSI - surgical site	e infection, UTI – urinary tract infection	lion						

of - auguos are mouuri, or - umany sacrameteror. Strainal site Mood and uninary tract infactions only

*Surgical site, blood and urinary tract infections only ** Not clear if prevalence figure relates to prevalence of infections or patients - the latter has been assumed

2.3.1 The overall prevalence of HAI

Prevalence studies conducted during the late 1970s estimated the prevalence of HAIs to be between 10-17%. In 1975 the prevalence of HAIs occurring in 34,246 patients admitted to five Swedish hospitals was found to be 17%.¹⁶ In 1978 10.4% of 1363 patients admitted to 25 Danish hospitals had one or more HAI and in 1979 12.1% of 1,557 admitted to the same hospitals had one or more HAIs at the time of survey.¹⁷

Prevalence surveys conducted between 1980 and the mid 1990s generally found the prevalence of HAI to be slightly lower at around 9%.^{67 18-21} The most recent study conducted in the UK and Republic of Ireland found 9.0% of patients surveyed in 1993 had one or more HAIs at the time of survey.⁷ A few studies conducted during this time period found a lower prevalence. In 1984 Sramova *et al* (1988)²² found the prevalence of HAIs in 12,260 patients admitted to 23 hospitals in Czechoslovakia in 1984 to be 6.1%, and in 1992 a study by Sartor *et al* found that 7.1% of 1389 patients admitted to eight hospitals in France had one or more HAIs, somewhat lower than an earlier estimate for the same year of 8.6%.²⁰

The results of prevalence studies conducted since 1995 vary, ranging from a low prevalence of infection rate of just 1.7% observed in a study by Pavia *et al* (2000)³⁷ involving 888 patients admitted to six hospitals in Italy (Cantazano) to 13% observed by Pittet *et al* (1999) in a Swiss period prevalence study conducted in 1996.³³

Other studies conducted during this time period found the prevalence of infections to be between 5.9% and 9.3%.^{21 31 35 36 38 39} In 1995 a Greek study observed a rate of 5.5% increasing to 5.9% in 1996.³¹ In 1996 a French prevalence survey involving 236,334 patients found the prevalence of infection to be 7.6%: 6.7% of patients studied had one or more infections,³⁵ and a Spanish survey found the prevalence of infections to be 8.4%: 7.2% of the 51,961 patients studied had one or more HAIs.²¹ In 1997 a Norwegian prevalence survey found the prevalence of infection to 6.1%.³⁶ and in 1999 a

61

prevalence study conducted in Greece found the prevalence of HAI to be 9.3%.38

The higher prevalence rate observed in the study by Pittet *et al* (1999)³³ can in part be explained by a number of factors: (I) the study was a period prevalence study, not a point prevalence study; (2) it was conducted in four teaching hospitals where you would expect higher rates than in other types of hospitals; (3) the study was conducted shortly after a national holiday weekend resulting in a low occupancy rate in the hospitals surveyed and a greater likelihood that the population studied were a relatively high risk population compared to the 'normal' hospital population.

If the above results are taken at face value they would appear to suggest that the prevalence of HAI has decreased overtime. However, as indicated in section 2.2 valid comparisons between studies and overtime are difficult. The methods employed to detect HAIs vary, the case definitions used differ, some studies have limited the types of infections included and the types of patients surveyed varies with study. These and other factors will inevitably impact on the prevalence rates observed and are discussed below.

2.3.1.1 Case definitions

The majority of studies have used the Centre for Disease Control (CDC) definitions of HAI or a modified version. Depending on the timing of the study either the 1972 CDC²³ or the 1988 CDC criteria²⁸ with or without the 1992 modified CDC definition of surgical site infections have been used.²⁹ Some studies whilst using these definitions have elected to exclude bacteriuria as a diagnosis for a hospital acquired UTI.^{21 30 31 33} Others have used a variant of the CDC definitions, adapting them to suit the patient population being studied. For example, a Norwegian study made some slight modifications to the 1998 CDC criteria. Details of the changes made were not presented in the paper.³⁶ Other studies have used the WHO definitions⁴⁰ and a few, such as the UK prevalence study of 1980 and 1996, have developed their own working definitions of

infections.⁶⁷ However, the first UK prevalence study definitions were based on the CDC criteria and the second UK prevalence study definitions were based on the 1980 working definitions and CDC criteria. The two prevalence studies conducted in France in May and November 1992 used two different sets of definitions. The survey conducted in May used the 1988 CDC criteria, and the survey conducted in November used the definitions of the Conseil Superieur D'Hygiene Publique de France for the diagnosis of infections of the urinary tract, chest (pneumonia), surgical wounds and bloodstream infections, which are similar to the CDC definitions.²⁰

Whilst the commonly used definitions are very similar they do differ in a number of important respects that may have an impact on the prevalence rate observed. For example, the criteria for urinary tract infections may vary with respect to whether microbiology evidence is required. Consequently, it is possible that in hospitals where access to microbiology services is limited, or where the culture is such that few specimens are taken, the infection rate will be underestimated. Gastmeier *et al* (1998)³² found infections rates were significantly higher in hospitals with an on-site laboratory service, than in those who did not have such facilities. However, it is not clear whether the apparent lower infection rate observed in hospitals without on-site access to microbiology facilities reflected an underestimate of the 'true ' rate or simply reflected a different case-mix, which was at relatively low risk of infection, at hospitals where access to on-site facilities were not deemed necessary.

2.3 1.2 Identification methods

The prevalence studies listed in Table 2.1 also vary in the methods used to detect HAIs. The approach generally adopted involved the following. All relevant data sources including treatment charts, case notes and microbiology records were consulted to identify signs and symptoms of infection. If the evidence obtained met the criteria detailed in the definitions used, an infection was said to be present. The majority of studies involved hospital personnel in this process. However, a few utilised external assessors.⁴¹ The degree of

training these staff received varies amongst the studies ranging from just a brief training session,²⁸ to a relatively substantial training programme.³³ ⁴¹ The validity of the results obtained depend on the ability of the staff to comply with the protocol adopted, and assuming compliance, the sensitivity and specificity of the selected approach. There is some evidence that greater accuracy in diagnosing HAI is achieved when better qualified staff are involved and when there is a more substantial training programme.⁴²

Inevitably in some cases there will be a degree of uncertainty as to whether an HAI is present. To overcome this, some studies instructed the assessor to indicate the degree of certainty associated with each diagnosis. For example, in the UK prevalence study, researchers were instructed to classify infections as 'certain', 'probable' or 'possible'. In contrast a German study only recorded 'certain' infections.⁴² This contributed to the considerably lower overall rate of 3.5% observed in the German prevalence study. When Gastmeier *et al* (1998)⁴² reworked the UK estimates only including 'certain' infections the prevalence fell from 9.0% to 4.2%. Whilst this is still higher than the German estimate, it is a considerably closer estimate. Further analysis indicated that if the German estimates were limited to infections occurring in hospitals with a minimum of 600 beds, in line with the types of hospitals included in the UK study, the estimated prevalence increased from 3.5% to 4.4% in line with the UK modified estimate (i.e. limited to 'certain' and excluding 'probable' or 'possible' infections) of 4.2%.⁴²

2.3.1.3 Sites of infections included

The majority of prevalence studies aimed to include all types of HAIs. However, a few limited the infections included to the commoner infections. For example a Norwegian prevalence study limited the infections included to the four most frequent infections: urinary tract, surgical wound, lower respiratory tract and bloodstream infections.³⁶ Limiting the types of HAIs included will inevitably have an impact on the overall prevalence rate, with the result being an underestimate of the overall scale of the problem. Studies also vary depending

upon whether they include all HAIs present on the day of survey³⁵ or limit inclusion to those acquired during the current admission and exclude those acquired during a previous admission.

2.3.1.4 Rates reported

As indicated in Table 2.1 some studies present prevalence rates based on the number of infections per 100 patients discharged,⁶ ¹⁶ ¹⁸ ¹⁹ ²¹ ²² ³⁰ ³¹ ³³ ³⁵⁻⁴⁰ others the number of infected patients per 100 patients surveyed⁷ ¹⁷ ¹⁹⁻²¹ ²⁴ ²⁵ ³² ³⁴ ³⁵ ³⁸ ⁴⁰ with some presenting both data on the prevalence of HAI (all infections) and the prevalence of infected patients who may have more than one infection.¹⁹ ²¹ ³⁵ ⁴⁰ ³⁸ Studies vary with respect to the criteria used to categorise patients with more than one infection into a specific infection group based on the type of infections identified. Whilst Jepsen *et al* (1980) classified patients according to the primary infection as identified by the investigator,¹⁷ other papers did not provide details about how infections at more than one site were managed.

2.3.1.5 Case mix included

The risk of acquiring an HAI varies considerably with patient population depending on both the intrinsic and extrinsic risk profile (see section 2.8). This risk profile will inevitably vary with case mix surveyed, and this in turn will impact on the overall rates observed. The prevalence of HAI in patients at high risk of a HAI is likely to be considerably higher than in low risk patients. This is clearly illustrated in the sub-group analysis of many of the prevalence studies. For example, the prevalence of HAI in patients admitted to 157 hospitals in the UK and the Republic of Ireland ranged from a low of 0% for dental patients, to a high of 34.2% in patients within the intensive care unit.⁷ Other multi-centre studies have also found the prevalence of HAI to vary considerably with specialty group. Consistently these studies have found the prevalence of HAI to be highest in intensive care patients.^{19 20} 22 31 32 34 37 40 and in most cases higher in surgical than medical patients.^{17 19 22 31 32 34 37 40} Furthermore, within selected specialties the prevalence will vary depending on the risk profile of the patients studied. For example, the very low prevalence rate observed in the recent

Italian survey by Pavia *et al* (2000),³⁷ may in part be due to there being a higher proportion of low risk patients in this study compared to other studies. As the authors acknowledge, an earlier study had shown that many patients with complicated conditions migrate to Northern regions or other European countries for treatment.⁴³ It is likely that the risk of HAI and consequently the infection rate is higher in these migrating patients.

Prevalence studies vary with respect to the patient groups included. Some studies, for example have attempted to survey all patients ⁶ 7 ¹⁶⁻¹⁸ ²⁰⁻²² ²⁴, ²⁷ ³⁰ ³¹ ³⁴⁻³⁶ ⁴⁰ whereas others limited their survey to patients admitted to selected specialties.²⁵ ⁴² A number of studies have focussed on the prevalence of HAI amongst patients treated on intensive care unit⁴⁴ and a Belgium study was limited to patients treated in surgical and intensive care units.²⁵ Where all specialties are included, given that the prevalence of HAI varies considerably with specialty inevitably the proportion of patients within the various specialties will impact on the overall rate observed, again making cross survey comparisons difficult.

Prevalence studies also vary with respect to the types of hospital included. Some studies include all types of hospitals, whereas others are limited to the larger hospitals. For example, the first UK prevalence study was limited to patients treated in hospitals with a minimum of 500 acute beds.⁶ Limiting the survey to larger hospitals may result in a higher national prevalence rate than if all hospitals were included as larger hospitals tend to treat populations at higher risk. Participating hospitals may be selected on a representative basis or in other cases hospitals may be asked to volunteer to participate in the study.⁷ The latter may introduce selection bias. The prevalence rate also appears to vary with type of hospital, the prevalence being higher in teaching hospitals than non-teaching hospitals. Jepsen *et al* (1980)¹⁷ in a Danish prevalence study observed a prevalence rate of 13.2% in general surgical patients treated in post-graduate teaching hospitals compared to an 4.5% in general surgical patients treated in non-post-graduate teaching hospitals. The prevalence rate has also been found to vary with care setting (acute, subacute or chronic) in part reflecting the differing risk profiles of the patients within these different care settings. A period prevalence study by Sax *et al* (2001),⁴⁵ conducted in a hospital in Geneva in May 1998, found that whilst the overall period prevalence was 11.3% within acute settings it was 8.4%, sub-acute 11.4% and chronic settings 16.4%. The odds of acquiring an infection were greater in sub-acute and chronic care settings when compared to acute settings even after adjustment for case mix factors (odds ratios 2.59 and 2.34 respectively).

2.3.2 Prevalence of specific types of HAI

The prevalence of HAI varies considerably with site of infection. Prevalence studies have generally found that infections of the urinary tract (UTI) are the most common, with surgical wound infections (SWIs) and lower respiratory tract infections (LRTIs) consistently forming the other most prevalent sites.^{7 31 32 17 19} UTIs accounted for 23.2% of the infections identified in the UK and Republic of Ireland Prevalence Survey of 1993/4, SWIs 11.7%, and LRTIs 22.9%.7 An exception is a Swiss prevalence study that found SWIs to be the most frequent The type of infection (30%) followed by UTIs (22%), and LRTIs (15%).³³ authors argue that this in part could reflect the fact that their study excluded bacteriuria as a criterion for hospital acquired UTI. It should also be noted that the study was limited to medical, surgical and intensive care patients, which inevitably is likely to have had an impact on the types and frequency of specific types of infections observed. The Greek prevalence study also found a differing pattern.³⁸ The study involving 3,925 patients admitted to 14 hospitals conducted in1999 found LRTIs to be the most frequent type of infection: LRTIs accounted for 30% of infections identified; UTIs 22.7%; bloodstream infections (BSIs) 15.8% and SWIs 14.8%. The authors suggest that this might reflect the high number of patients in intensive care units on the day of study (6.7%), whilst also noting that the UK prevalence study conducted in 19907 also identified an increase in the prevalence of LRTIs.

The need to take into account differences in methodology and the case mix of the patients studied, may be further illustrated by reference to the two large multicentre studies conducted in the UK in 1980⁶ and 1993/4.⁷ The prevalence of HAI was found to be similar for the two time periods: 9.2% and 9.0% respectively. It cannot, however, be concluded from these two studies that the prevalence of HAI has remained stable overtime. The definitions of infection differed between the two studies, as did the methods used to detect HAI, and this may have had an impact on the prevalence rates observed. However, perhaps more importantly despite the two studies involving patients from similar specialties, important case mix differences were present in the two groups. For example, whereas in 1980 37.7% of males and 40.8% of females were 65 years or over, in 1993 these figures had risen to 48.8% and 50.7% respectively. The NHS was thus treating a significantly older population than previously had been the case, and consequently a population at greater risk of acquiring an infection in hospital. Thus whilst the results may at first appear to suggest that infection rates have remained stable over time, they may in fact reflect an improvement in quality of care, given the higher risk population. However, countering this is the change in discharge patterns that occurred over the intervening years. In 1993/4, patients were being discharged home at an earlier point in their recovery than was the case in 1980. It might be expected that this would have resulted in either an absolute reduction in the prevalence of HAI, and/or a reduction in the prevalence of HAI amongst in-patients and an increase in the prevalence of HAI in the community.

Given the above it is clear that if valid comparisons are to be made between prevalence studies, allowances must be made for differences in the intrinsic and extrinsic risks profile of the populations involved and for differences in the methods used to assess the prevalence of infection. A paper by Gastmeier *et al* (1998)⁴² compares the methods used in different prevalence studies, in terms of the definitions used, how hospitals were selected, training of investigators, the proportion of infections accompanied by a positive laboratory result and the availability of data for diagnosing infection (e.g. are laboratory facilities available

and what is the policy on specimen collection), and the date of study. The paper concludes that due to the many methodological differences comparisons of rates between countries should be avoided. However, even in the absence of such adjustment the results of multi-centre prevalence studies clearly demonstrate that HAIs affect a substantial number of patients.

2.3.3 The prevalence of HAI in patients admitted to surgical specialties

As indicated above prevalence studies frequently provide information on the prevalence of HAI in patients admitted to specific specialties. The results of these studies clearly indicate that the prevalence varies with specialty and with study. However, for all the reasons indicated above (sections 2.3.1.1 - 2.3.1.5) valid cross study comparisons are difficult. Furthermore, the specialty groupings adopted vary. Whilst a number of studies provide data on the prevalence rate amongst surgical patients, in some cases it would appear that this includes infections occurring in both general surgical patients and patients admitted to sub-surgical specialties such as gynaecology, urology and orthopaedics, whereas in other cases the HAI rates presented are limited to general surgical patients, with rates for the other sub-surgical specialties presented separately. For example, Emmerson et al (1996),7 in a paper that presented the main results of the UK prevalence study of 1993/4, reported prevalence rates for nine different surgical specialties, whereas a Greek prevalence conducted in 1999³¹ and an earlier Spanish survey conducted in 1990²¹ have sub-divided their sample into just two surgical categories: surgical and obstetric and gynaecology specialties combined. Varving specialty groupings together with the factors mentioned in sections 2.3.1.1 - 2.3.1.5 accounts at least in part for the wide variation in prevalence rates observed. For example, the reported prevalence of HAIs amongst surgical patients ranges from a low of just 1.2% observed in a study by Pavia et al (2000)³⁷ to the relatively high level of 13.9% observed in a study by Jepsen et al (1980);¹⁷ the reported HAI prevalence rates for obstetric and gynaecology patients range from no infections observed in a study by Gikas et al (1999)³¹ to 11.1% in a

study by Pavia *et al* (2000);³⁷ and the reported prevalence of HAI in urology patients range from a low of 7% in a study by Scheel *et al*³⁶ to 18.6% observed amongst patients admitted to Danish teaching hospital in 1979.¹⁷

Some papers present data on infections in surgical patients based on the type of surgery conducted (clean or dirty). For example Sartor *et al* (1980) provide rates for clean surgery and other surgery.²⁰ Others have focussed on the prevalence of SWI according to the type of wound as defined by the wound classification system: clean, clean/contaminated, contaminated or dirty.^{19 31} These studies have all found the prevalence to be highest in patients with dirty wounds. However, there is no overall clear trend for infection rates occurring in the other wound categories. For example, a Spanish¹⁹ prevalence study found that the prevalence of SWI increased with increasing contamination of the wound. However, a Greek prevalence study did not observe this trend and found that prevalence amongst the other groupings differed overtime.³¹

2.3.4 Frequency of specific types of infections occurring in patients admitted to surgical specialties

A number of prevalence studies provide data on the frequency of specific infections within selected specialties.⁷ ¹⁷ ²⁰ ³² The pattern observed varies with the study. The most recent UK prevalence study indicated that amongst general surgical patients LRTIs were the most frequent type of infection: 2.6% acquired a LRTI; 2.4% a SWI; 1.9% a UTI and 0.8% a skin infection. Amongst gynaecology patients a different pattern emerged with UTIs being the most frequent type of infection: 5.1% had a UTI, 2.4% a SWI, 1.0% a LRTI and 0.3% a skin infection. UTIs were also the most frequent type of infection amongst orthopaedic and trauma patients and urology patients. In orthopaedic and trauma patients 3.7% had a UTI, 2.8% a SWI, 2.0% a LRTI and 1.4% a skin infection. In urology and uro-surgical patients 6.1% had a UTI, 1.2% a LRTI, 0.8% a SWI and 0.5% a skin infection.

2.4 Incidence of HAI

Incidence measures quantify the number of new cases of a disease that develop in a population of individuals at risk over a specified period of time. There are two specific types of incidence: cumulative incidence and incidence density or force of morbidity.

Cumulative incidence is the proportion of people who develop a disease over a specified period of time (Figure 2.2):

Figure 2.2: Cumulative incidence formula

CI =	number of new cases of a disease during a given period of time
	total population at risk
CI = cumulativ	re incidence

Cumulative incidence therefore provides an estimate of the probability, or risk, that an individual will develop a disease during a specified period of time.

Incidence density, also referred to as the 'incidence rate' or the 'force of morbidity', measures the number of new cases of a disease during the period of survey and expresses this figure as a proportion of the time each individual remained at risk (Figure 2.3).

Figure 2.3: Incidence density formula

ID = number of new cases of a disease during a given period of time total period - time at risk ID = Incidence density
The incidence density of HAI is therefore a measure of the number of new cases of HAI occurring during the period of survey, expressed as a proportion of the time each patient remained in hospital free from infection during the same time period. That is the number of days from time of admission to either time of discharge in the absence of a HAI, or where an HAI is present, the day of onset of the HAI.

The incidence density can be interpreted as the risk of developing a disease per unit of time exposed, and as such it is a more precise measure of the impact of exposure in a population. In addition, determining the incidence density of HAI, as opposed to cumulative incidence, overcomes some of the confounding effects of length of stay. Hospital length of stay will vary, both within and between hospitals, depending on factors relating to case-mix and discharge policies, and this will inevitably have an impact on the cumulative incidence of HAI. If the mean length of stay is relatively short, the number of discharged patients will be relatively high, and as a result the cumulative incidence may appear low, whereas in situations where the mean length of stay is relatively long the opposite may occur.⁴⁶

2. 4.1 Estimates of the incidence of HAI

Studies of the incidence of HAI have tended to focus on specific infections, or infections occurring in specific patient groups, rather than consider the incidence of all types of HAI occurring in a broad case-mix of patients. However, two studies conducted in England do provide important data on the incidence of HAI occurring in a broad case mix of patients.

In 1992, a study involving 3,326 adult patients admitted to the surgical, medical, gynaecology, and orthopaedic specialties of a district general hospital in England found that 7.2% acquired one or more infections that presented during the in-patient stay.⁴⁷

In 1994, an audit of infection control policies and practices in 19 hospitals in England and Wales included an assessment of the incidence of urinary tract, respiratory tract and bloodstream infections (the three most common types of infection), occurring in adult patients, admitted to the medical, surgical, gynaecology and orthopaedic specialties of the selected hospitals and who had a minimum stay of 3 days. A total of 81,218 patient episodes involving 72,434 patients were observed, of which 80,752 episodes involving 72,013 patients could be allocated to specialty groups. Over the study period a total of 2,148 infections were observed, giving a rate of 3.0 per 100 patients and 2.7 per 100 episodes.¹

Data from the US provide further insight into the incidence of HAI. In 1975, Haley *et al*,⁴⁸ as part of a pilot study for the study of the Efficacy of Nosocomial Infection Control (SENIC), assessed the incidence of HAI occurring in 4,067 patients admitted over an 11 week period to a hospital in Atlanta, USA. Of these 5.1% of patients acquired one or more HAIs that presented during the inpatient period, and there were 6.0 HAIs per 100 admissions. UTIs were the most frequent type of infection accounting for 40% of infections observed, followed by SWIs (33%), and LRTIs (16%). Primary bacteraemias accounted for 2% of infections observed and infections at other sites the remaining 9% of infections observed.

Similar results were obtained from the SENIC study itself. The SENIC study investigated the incidence of HAIs occurring in patients admitted to 338 hospitals, representing the 6,449 acute care, general medical and surgical hospitals in the US, over a prolonged period from 1970 - 1975. The HAIs studied were limited to the four most frequent types of infections: urinary tract, surgical wound, respiratory tract and bloodstream. These four infections were thought to account for 80% of all HAIs.⁴⁹ A report based on the analysis of a random sample of 169,526 adult patients admitted to these hospitals over a 12 month period in 1975-1976, supplemented with data on the incidence of infections at sites other than those studied in the SENIC study taken from data

73

from the 75 hospitals participating in the US National Nosocomial Infections Study, and extrapolated to all adult patients admitted to the 6,449 acute sector hospitals, estimated that there were 5.7 infections (all sites) per 100 admissions and that 4.5% of all admissions acquired one or more HAIs which presented during their in-patient admission.⁴⁹ Based on the results of the SENIC pilot studies,⁴⁸ and the increasing use of medical technology and no corresponding evidence of increased safety or infection control input, the authors concluded that this estimate was likely to be an underestimate of the incidence of HAI.⁴⁹

As with prevalence studies, comparisons between the findings of these three studies should be made with caution. The definitions of infection and the methods used to identify infections, the case-mix of patients studied and the treatment patterns, including discharge patterns, differed with study. Meaningful comparisons are dependent on these and other factors being controlled for.^{46 50}

2.4.2 The relative incidence of specific types of infection

Studies that assessed the incidence of HAI occurring in a broad case mix of patients also provide important data on the incidence of specific types of infection. Consistently these studies find urinary tract, surgical wound and respiratory tract infections to be the most frequent types of infection. The study by Glenister *et al* (1992) found that UTIs were the most frequent type of infection, accounting for 27% of all infections observed; SWIs accounted for 23%, and pneumonia accounted for 15%.⁴⁷

Estimates from the US indicate that in 1976/6 urinary tract, surgical wound and respiratory tract infections accounted for 42%, 24% and 10% of the total number of infections estimated to have occurred in patients admitted to the acute sector hospitals that year. ⁴⁹

2.4.3 The incidence of HAIs occurring in patients admitted to surgical specialties

Studies that have assessed the incidence of HAIs occurring in patients admitted to surgical specialties include those that have assessed the incidence of all types of infection occurring in a broad case mix of patients (including surgical and medical patients), which have subsequently stratified their results by specialty; and studies that have assessed the incidence of all types of HAIs or selected types of HAIs occurring in selected surgical patient groups.

For example, a study by Glenister *et al* (1992) which assessed the incidence of HAIs occurring in medical and surgical adult patients admitted to a district general hospital (DGH) in England observed an overall incidence rate of 7.2%, increasing to 9.7% in patients admitted to the surgical specialties (general surgery, gynaecology and orthopaedics). The incidence amongst general surgical patients alone was 8.7%, in gynaecology patients 9.3%, and in orthopaedic patients 13.6%.⁴⁷

Studies that have limited their focus to the incidence of HAIs or selected types of HAIs occurring in surgical patients include those that have assessed the incidence of infections in patients who have undergone a particular operative procedure such as a caesarean section,^{51,52,53-56} or hysterectomy.⁵⁷ Others have assessed the incidence of a specific type of infection occurring in a particular patient group. For example, Costantini⁵⁸ assessed the incidence of infections occurring in patients cared for on the intensive care unit. Gravel-Tropper *et al*,(1995)⁵⁹ assessed the incidence of SWIs in gynaecology and obstetrics patients. Leigh *et al* (1981)⁶⁰ conducted an eight year study of post-operative wound infection in two district general hospitals in England between 1971 and 1978. Erbaydar *et al* (1995) ⁶¹ in a study primarily concerned with estimating the impact of SWIs on length of hospital stay, estimated the incidence of SWIs occurring in general surgical patients who underwent an operative procedure within a hospital in Turkey.

The estimates of infections within a selected group of patients vary. For example, Henderson in a review of studies that assessed the incidence of SWIs occurring in patients who had undergone caesarean sections found the rate to range from 0-24%.⁵³ Factors previously mentioned relating to the methodology employed and the case-mix and treatment regimens, may explain much of the variation observed. More recently, Nice *et al* (1996)⁵⁵ in an audit of SWIs following emergency and elective caesarean sections in five West Yorkshire Hospitals in the UK, found that of the 628 women who had a caesarean section, 7.2% acquired a SWI. The SWI rates varied between then hospitals ranging from 2.5-17.2%. Analysis of surveillance data collected between January 1997 and December 1998 in eight maternity hospitals in France, found the incidence of all types of infection occurring in patients who had a caesarean section to be 7.4% and just 1.2% for vaginal deliveries.⁵⁶

2.4.4 The incidence of specific types of infection in patients admitted to surgical specialties

2.4.4.1 The incidence of SWIs in patients admitted to surgical specialties

Studies that have assessed the incidence of SWIs occurring in surgical patients have found the incidence to vary. For example, Cruse and Foord (1980)⁶² in a ten year prospective study conducted in Canada between 1967 and 1977 and involving 62,939 wounds observed an incidence rate of 5.7%. In the US the National Nosocomial Infection Surveillance System (NNIS), between 1986 and 1996, observed a lower rate. Of the 593,344 operations observed 2.6% were complicated with a SWI⁶³ More recently, Abussaud and Meqdem (1986) in a study involving 504 surgical patients observed a slightly higher rate: 3.6% of patients acquired a SWI.⁶⁴ A similar rate was observed in a randomised controlled study by Lynch *et al*,⁶⁵ designed to assess the cost-effectiveness of chlorhexidine body wash as a means of reducing the risk of infection: 3.2% of 3,482 patients had a SWI as defined by purulent discharge from their wound.⁶⁵ However, when the definition of SWI was relaxed such that patients who did not have a purulent discharge, but had an ASEPSIS score of more than 10 were classified as infected, the SWI rate increased to 5.8%.⁶⁵ A study by Malone *et*

al (2002) also observed a similar rate: 3.2% of the 5031 non-cardiac surgical patients included in the study acquired a SWI.⁶⁶

A slightly lower SWI rate of 2.8% was observed by Culver *et al*⁶⁷ in a study conducted in the US. The estimate was derived from data on 84,691 operations performed at 44 NNIS hospitals in the US between January 1987 and December 1990. A smaller study conducted by Kirkland *et al* (1999)⁶⁸ in the US in 1999 observed a lower rate: 1.2% of 22,742 patients who underwent inpatient surgery acquired a SWI, and an earlier study by Rubenstein *et al* (1982)⁶⁹ conducted in Israel in 1979 observed a SWI rate of 1.9% in general surgical patients and 1.6% in orthopaedic patients.

An earlier study conducted in 1979 and limited to 1,346 patients who had undergone selected abdominal surgery observed a still lower infection rate of 0.97%⁷⁰ and a five year prospective study conducted by Krukowski *et al* (1984) involving 1,504 patients who underwent abdominal surgery at a hospital in the UK observed a rate of 2.8%, with over 50% of infections identified post-discharge.⁷¹ However, a study by Bremmelgaard *et al* (1989)⁷² involving 42,228 general surgical and orthopaedic patients found that 6.3% of patients acquired a SWI. Garibaldi *et al* (1991) in a study of 1,852 adult surgical patients admitted to university affiliated US hospitals over a four year period observed a similar rate: 6.5% of patients studied acquired a SWI.⁷³ A study by Mishriki *et al* (1990),⁷⁴ which assessed the incidence of SWIs in 702 adult, non-trauma patients of which 600 were in-patients and 102 day cases, observed a higher incidence rate of 7.3% and Erbaydar *et al* (1995)⁶¹ observed a considerably higher incidence rate: 15.2% of the 1482 general surgical patients studied acquired one or more SWIs

The variation in the estimates obtained can again in part be attributed to methodological differences: differing case definitions and detection methods and varying follow-up times. For example, the relatively high incidence rate observed by Mishriki *et al* (1990)⁷⁴ can in part be attributed to the definition of

77

SWI used which was based on clinical data and the fact that infections presenting post-discharge were included in the estimate with post-discharge follow up extending over a period of six weeks: 55% of infections identified presented after the patient was discharged from hospital. In contrast the relatively low SWI incidence rate reported by Culver *et al* (1991)⁶⁷ was limited to the incidence of SWI presenting during the in-patient period and relates to the incidence of SWI as defined by relatively strict criteria.

The impact of case definitions and case detection methods on the rates obtained was highlighted in a short report by Reilly *et al* (2001).⁷⁵ Over a 28 month period 1,772 patients admitted to a hospital in the UK were followed for a period of 30 days following clean surgery. The utilisation of a relatively strict definition for SWI that required the presence of pus from the wound resulted in a SWI rate of 5%. A wider definition resulted in a SWI rate of 8%. These estimates relate to SWIs presenting within 30 days of clean surgery. If however the time period of interest was limited to infections presenting during the inpatient period then the 8% rate fell to an artificially low rate of 1%.

A recent systematic review to assess the evidence of validity and reliability of the definitions and detection methods used in prospective studies of SWIs published between 1993-1999, highlighted the wide variation found in these studies with respect to these factors, and how this impacts on the estimates of infection rates derived and importantly limits valid comparisons being made.⁷⁶ Forty-one different definitions of SWI were used in the ninety studies included in the review, of which only five were definitions proposed by multi-disciplinary groups.^{28 77-80}

Whilst methodological differences clearly have an impact on the overall rates observed, case mix differences perhaps play a greater role. Data on the incidence of SWI and how this varies with wound class clearly demonstrate the impact that case mix differences have on rates. As might be expected studies consistently found the SWI rate to be considerably lower in clean surgery than in contaminated surgery. Cruse and Foord's (1980) ten year study of wound infections clearly showed this. Whilst Cruse and Foord's⁶² study observed an overall SWI rate of 4.7%, rates varied markedly with wound class. The SWI rate in clean surgery was just 1.5%, rising to 7.7% in clean-contaminated surgery, 15.2% in contaminated surgery and 40% in dirty surgery.

More recent studies have continued to observe variation in rates according to wound class. For example, whilst Abussaud and Megdem (1986)⁶⁴ in their study involving 504 surgical patients observed an overall incidence rate of 3.6%, the rate was found to vary from a low of just 1.1% for patients who underwent clean surgery to 18.5% amongst patients who underwent contaminated surgery. Similarly Bremmelgaard et al (1989)⁷² in their study involving 42,228 general surgical and orthopaedic patients found the SWI rate varied from a relatively low rate of 2.3% in patients who underwent clean surgery to a high rate of 27.1% in patients who had dirty wounds, with an overall rate of 6.3%. However, Malone et al (2002)66 observed a slightly different pattern. In their study of 5,031 non-cardiac surgical patients infection rates were lowest in the dirty wound category: just 1.8% of patients classified as having undergone 'dirty-infected' surgery acquired a SWI. This compared to rates of 2.4% in patients who underwent 'clean' surgery, 4.2% in patients who underwent 'clean-contaminated' surgery and 4.6% in patients who underwent 'contaminated surgery'. The authors attributed this finding to either the relatively low number (57) of dirty cases studied or differences in wound care practices.

This variation in SWI rates according to wound class category inevitably has an impact on the overall SWIs rate observed. The overall rate will clearly depend to some extent on the proportion of patients within each wound class category. The distribution of other intrinsic and extrinsic risk factors will also have an impact on the rates observed. These are discussed in section 2.8.

79

2.4.4.2 The incidence of UTIs in surgical patients

While studies that have assessed the incidence of UTIs in patients admitted to both medical and surgical specialties suggest that between 1 and 3% acquire a UTI;^{81,82} studies limited to surgical patients generally suggest that the incidence amongst surgical patients is considerably higher.¹⁴ Coello et al found that 6.3% of adult patients admitted to the surgical, urology, gynaecology and orthopaedic specialties of a DGH in England and who had an operative procedure acquired a UTI.¹⁴ Bueno-Cavanillas et al (1991)⁸³ in a study involving 449 surgical patients admitted to a hospital in Spain in 1986 observed an incidence of 8.7%. A study by Melatomaa et al (2000)⁵⁷ which assessed the incidence of HAIs occurring inpatients who had either an abdominal hysterectomy 516 (75%); vaginal hysterectomy 105 (15%); or laproscopic hystectomy 66 (10%), at a Finnish University Teaching Hospital during the period October 1993-September 1994, observed a higher UTI incidence rate of 13.5%. An exception is a study by Rubenstein et al (1982),⁶⁹ conducted in Israel in 1979: 3.1% of the 967 general surgical patients surveyed and 2.9% of the 968 orthopaedic patients surveyed acquired a UTI.

2. 4.4.3 The incidence of LRTIs in surgical patients

Incidence studies have generally found 0.8 – 0.9% of medical and surgical patients acquire a respiratory tract infection with the rates varying with type of patient.⁸⁴ ⁸⁵ A number of studies have focussed specifically on pneumonias occurring in ventilated patients (ventilator- associated pneumonias- VAP). The incidence of VAPs has been found to vary from 10% to 65%.^{86-92 93 94}

2.4.4 Evidence of inter-hospital variation

Evidence from national surveillance schemes indicates that infection rates vary between hospitals which cannot be explained by case mix differences.^{95 96} For example, data from the UK National Nosocomial Infection Surveillance Scheme indicate that the incidence of surgical site infection varied with both the category of procedures and hospital. Similarly the incidence of bacteraemia varied by specialty and within specialty between hospital sites. This inter-hospital variation remained after controlling for age structure and other case mix differences.⁵ Given that the hospitals involved in this national surveillance scheme all used the same data collection methods the results suggest that factors relating to policies and practice may account for the variation observed.

2.4.5 Infections presenting post-discharge

The problem of HAI is not restricted to the hospital stay: a proportion of infections do not present until after the patient has been discharged from hospital. Prevalence studies fail to take into account these infections and incidence studies, until recently, have generally focused on infections presenting during the in-patient phase. As a result these studies inevitably underestimate the scale of the problem of HAI.

Studies examining the incidence of SWIs presenting post-discharge indicate that anything from 20-86% of SWIs present post-discharge.^{57 65 71 74 97-105} The variation in the proportion of infections identified post-discharge may be explained, at least in part, by reference to the different methods used to identify infections, differences in the case-mix of the population studied and varied discharge patterns.^{104 106}

2.5 National estimates of the number of HAIs occurring per annum

National estimates of the number of HAIs occurring per annum and/or the number of patients who acquire one or more HAIs have been derived. For example, in the US Haley *et al*,⁴⁹ based on data from the SENIC study and the NNIS survey, estimated that there are at least 2.1 million HAIs per annum. The authors suggest that whilst this figure is substantial it is likely to be an underestimate of the scale of the problem. They argue that as a result of advances in medical technology, it is now possible to treat patients who in the past were untreatable. Many of these patients are at high risk of acquiring an infection. Whilst medical advances have facilitated treatment there is no substantial evidence of improved infection control. As such it is argued that the

actual figure is likely to be closer to 4 million. Estimates of the number of HAIs occurring in surgical patients alone were not presented

In the UK in 1997 Glynn *et al* ¹ estimated that were at least 100,000 HAIs every year. This estimate was derived from estimates of the incidence of chest, urinary tract, and bloodstream infections occurring in adult patients admitted for a minimum of three days to the medical, surgical, orthopaedic and gynaecology specialties of 19 district general hospitals in England and Wales in 1996. Extrapolating the results of their study to the national level the researchers estimated that there were 60,000 HAIs of the urinary tract, respiratory tract or bloodstream per year. When infections occurring in patients admitted to other specialties are taken into account, together with skin and SWIs then this figure was estimated to increase to 100,000 infections per annum. Estimates of the number of HAIs occurring in surgical patients alone were not presented

2.6 Impact of HAI on mortality

HAIs may directly cause death, substantially contribute to the terminal event, or have little or no role in a patient's death.¹⁰⁷ In any individual case it is difficult to tease out the contribution of the infection to the terminal event. The role that HAIs play in hospital deaths inevitably varies with type of infection, and the patients underlying disease and health status.

Studies that present information on mortality rates in infected and uninfected patients show that the mortality rate is significantly higher in infected than uninfected patients.¹⁰⁸⁻¹¹⁰ ⁸⁹ ⁹¹ ¹¹¹ ¹¹² ⁶⁸ ⁹² ¹¹³⁻¹¹⁸ An exception is a study by Freeman *et al* (1979). Eighty-five infected patients admitted to a US hospital in 1973 were matched with suitable controls. Little overall difference in mortality rates between infected and uninfected controls was found. However, as acknowledged by the authors the sample was relatively small and as such insufficient to pick up a small but statistically significant increased risk if present.¹¹¹

The higher mortality rate generally observed amongst infected patients may reflect both the increased mortality risk that HAIs present, and marked differences in the distribution of other mortality risk factors in the two groups (infected and uninfected patients). The case mix of the infected group may, even in the absence of an infection, render the infected patients at greater risk of in-patient death than the uninfected patients, with many of the risk factors for HAI also being risk factors for mortality.¹¹⁸

The results of a prospective study of 4,714 patients admitted to three Spanish hospitals in 1994/5 provide evidence that many of the risk factors for infection are simultaneously risk factors for mortality.¹¹⁸ The study assessed whether the SENIC and NNIS indices of intrinsic infection risk were also good predictors of in-hospital mortality. The results of the crude data analysis indicated that both indices were related to in-patient mortality. However, the results of the regression analysis which controlled for the potential confounding effects of age, sex, American Society of Anaesthesiologists scores (ASA score), cancer, renal failure, diabetes mellitus and a stay on the intensive care unit, found that the SENIC index ceased to demonstrate a significant trend with mortality (p = 0.025), while the NNIS index did demonstrates a significant trend (p=<0.001).

The higher mortality rates observed in infected patients may therefore to some extent reflect the underlying mortality risk. Consequently, deaths occurring in infected patients may not be directly attributed to the presence of an infection but attributable to other factors. The task therefore is to assess the specific contribution of the infection to death. Studies that have attempted to do this vary in their approach.

2.6.1 Methods used to assess the impact of HAI on mortality

2.6.1.1 Review of case notes

In some studies case notes of deceased patients were reviewed and a judgement made as to whether the infection was the primary cause of death or a substantial contributing factor. Gross *et al* (1980)¹¹⁹ present guidelines for determining whether the presence of an HAI directly caused the patients death or acted as a contributing factor. However, as Salemi *et al* (1995)¹¹⁵ point out even in the presence of guidelines categorising patient deaths in this way can be difficult. Salemi *et al* (1995) argue for one category that combines both deaths directly caused by HAI and those in which the presence of an infection was a contributing factor, suggesting this would improve the validity of the mortality data.¹¹⁵

2.6.1.2 Autopsy examination

Other studies rely on the results of autopsy examination for information on the number of deaths resulting from a HAI.¹¹⁰ For example, Daschner *et al* (1978) examined autopsy data from 1000 patient autopsies during the years 1975 – 1976. In their study, autopsy findings, complemented by a review of medical records and the clinical judgement of a pathologist, were used to determine the presence of an infection and subsequently categorise patients into those whose HAI directly led to death and those in which the HAI was a contributing factor to the terminal event. The criterion used for allocating patients into these two groups was not published. HAI was identified as the direct cause of death in 7.4% and a contributing factor in 6.3% of the 1000 autopsies examined, and found to be either the direct cause of death or a contributing factor in 80% of those autopsied patients who had an infection.

2.6.1.3 Review of death certificates

An alternative approach adopted by some studies has been to review death certificates to identify patients for whom HAI has been listed as the cause of death.¹¹⁵ This approach is thus reliant on the cause of death being accurately recorded on the death certificate. Failure to report deaths from HAI on the

death certificate will in some cases result in the death rate being underestimated. Salemi et al (1995)¹¹⁵ in their study of deaths occurring in patients admitted to a single US hospital who had acquired either a pneumonia, primary BSI or SWI, found the number of deaths attributable to an infection increased when the case notes of patients with pneumonia or BSI whose death certificate did not list infection as a cause of death were examined. Twenty-four per cent of patients who acquired a pneumonia and 25% of patients who acquired a BSI died. An examination of death certificates alone indicated that the infection was either the direct cause or a contributing factor in 15% of deaths occurring in patients with a pneumonia or BSI. Following an examination of case notes of those patients for whom the infection was not listed as the cause of death on the death certificate, the proportion of deaths attributable to the infection increased to 20% in patients who had a pneumonia and 19% in patients who had a BSI.¹¹⁵ Both of these increases were significant (p<0.05).

2.6.1.4 Case control approach

Other studies have adopted a case control approach to adjust for severity of underlying illness.¹⁰⁸ ¹⁰⁹ ¹¹³, ⁶⁸ ⁹¹ ¹¹¹ ¹²⁰ ¹²¹ For example, Rose *et al* (1977)¹⁰⁸ compared mortality rates amongst 40 patients with hospital acquired BSIs with that occurring in 40 uninfected matched controls treated at a US hospital between December 1972 and August 1974. The mortality rate in bacteraemic patients was 38% compared to just 10% in controls. During this time period a further 84 patients had a BSI, however medical notes were not available in 19 cases and suitable control patients could not be found in the remaining cases. The mortality rate amongst these unmatched bacteraemic patients and 109 additional bacteraemic patients admitted to the study hospital between January and October 1975 was 34%. The inability of this study to find suitable controls for all infected patients is a common problem associated with this approach,¹⁰⁹ and inevitably may introduce bias into the mortality estimates derived.

2.6.1.5 Data stratified by underlying risk

To enable more valid comparisons in death rates both over time, and between institutions, some studies have assessed the death rate in infected patients and then stratified the data by underlying risk. For example, a study by Salemi et al (1995), based on infection data from a single US hospital for the years 1987-1992, assessed the number of deaths that were either primarily due to the presence of an infection or in which the infection played a substantial role and subsequently stratified the data by underlying risk based on the patients severity of illness.¹¹⁵ The severity of illness classification system used consisted of five categories ranging from no risk of death expected during the patients hospital admission to almost certain death expected. For the purpose of their work the five categories were collapsed into an 'at risk of death during hospital admission group' and a 'not at risk of death during admission group'. Of those patients who had acquired a BSI and had been classified as 'not at risk of death' the infection either directly caused or contributed to death in 5% of cases, whereas in patients who had acquired a BSI and had been classified as 'at risk of death', the infection either directly caused or contributed to the patients death in 21% of cases. The association between pneumonia and death in the 'not at risk of death' and 'at risk of death' groups was 13% and 23% respectively. The authors concluded that if valid comparisons are to be made mortality rates should be stratified by risk.

2.6.1.6 Regression analysis

Regression analysis is another technique that has been used to control for factors other than the presence of an HAI which might contribute to the patient's death and delineate the role of the HAI.^{89 91 92 112 118} For example, a study by Fagon *et al*⁹² conducted a stepwise logistic regression analysis which assessed the independent effects of nosocomial pneumonia on mortality risk in intensive care patients. The analysis controlled for a range of factors known to be associated with mortality in intensive care patients. The results indicated that after controlling for a range of other factors, patients who acquired a pneumonia were 2.1 (95% CI: 1.8, 3.6) times more likely to die than uninfected patients.

2.6.2 Specialty specific mortality rates

Studies that have examined in-patient deaths and more specifically the contribution of an infection to death have generally found that infections occurring in surgical patients were more likely to cause or contribute to death than infections occurring in patients admitted to non-surgical specialties.¹¹⁰ ¹¹⁹ For example, Gross *et al* (1980) found that whilst more deaths occurred on the medical services only 25% were associated with the infection, whereas on the surgical specialties half of the deaths were associated with infection. A study by Daschner *et al* (1978) also found that most infections causing or contributing to death were acquired on surgical specialties.¹¹⁰

2.6.3 Site-specific mortality rates

The impact of HAI on mortality risk whilst depending on the patient's underlying condition and the interplay between the infection and the patient's condition, will also vary with type of infection. The results of studies which have assessed site specific death rate have generally found that death rates are higher in patients with bloodstream and lower respiratory tract infections than in patients with a UTI.¹¹⁹ For example, unpublished data from the NNIS system in the US which was presented by Hughes et al at the Twenty-second Interscience Conference on Antimicrobial Agents and Chemotherapy, and reported in an article by Martone et al. (1998),¹²² indicated that whilst overall 0.9% of patients with an HAI die as a direct result of the infection and in a further 2.7% of cases the HAI is a substantial contributing factor there was considerable variation in attributable risk by site. Pneumonia and BSIs were found to be greater risk factors for mortality than infections at other sites: 4.4% of deaths were directly attributable to bloodstream infections and 8.6% partially attributable; 3.1% were This directly attributable to pneumonia and 10,1% partially attributable. contrasts with 0.8% of deaths attributable to 'other sites' and 2.5% partially attributable to 'other sites'; 0.6% directly attributable to SWIs and 1.9% partially attributable to SWIs; and 0.1% directly attributable to UTIs and 0.7% partially attributable to UTIs.

Interestingly whilst the risk of death is generally found to be relatively low in patients who acquire a UTI, a study by Platt *et al* (1982) indicated that after controlling for a number of other factors including severity of illness, age and duration of catheterisation, the presence of a hospital acquired UTI was associated with a three-fold increase in risk of mortality.¹¹² While it was acknowledged that the presence of a significant association between infection and mortality is not necessarily indicative of a causal relationship, and that the possibility remains that the UTI was a marker for a factor not included in the logistic regression analysis, the later was not thought to be the case.

The study by Salemi *et al* (1995) referred to above, based on infection data from a single US hospital for the years 1987-1992 assessed the number of deaths that were either primarily due to the presence of an infection or in which the infection played a substantial role.¹¹⁵ This study found that 19% of patients who had a BSI and 20% of patients with pneumonia subsequently died as a result of infection compared to just 1% of patients with a SWI who died from infection. The authors subsequently compared this to CDC rates of 13%, 13% and 2.5%. There were no significant differences in the rates observed for both BSI and SWI but there were for pneumonia.

Delgado-Rodriguez *et al*(1999)¹¹⁸ in their study of 4,714 surgical patients admitted to three Spanish hospitals in 1994/5 mentioned above found that after controlling for a range of potential confounders (age, sex, presence or absence of a cancer, renal failure, or diabetes, stay in the ICU and the NNIS index), the presence of an organ/space surgical site infection was significantly related to mortality (Odds ratio 4.9: 95% CI: 1.5 - 15.6) as were BSIs (Odds ratio17.3: 95% CI: 3.5 - 87.0). The acquisition of a single infection at the following sites was not significantly associated with mortality: superficial incisional wound infection; deep incision wound infection, respiratory tract infection, urinary tract infection the combination of a surgical site and respiratory tract

infection or a surgical site and bloodstream infection was significantly associated with mortality, but other combinations were not.

The analysis of data from the French surgical site infection surveillance network (INCISO) for the years 1997-1999 indicated that the crude mortality rate amongst patients with a SSI was considerably higher than in uninfected patients: 5.8% and 1.35 respectively.¹²³ The attributable mortality rate was thus 4.5%. After adjustment for a range of risk factors including sex, age, NNIS risk index, length of hospital stay prior to surgery, type of surgery (elective or emergency) and the presence or absence of endoscopic surgery, it was estimated that 38% of deaths occurring in infected patients were directly attributable to the infection. Subjective surgeon assessment estimated that 25% were directly attributable to the infection, a rate that was also reported by the US NNIS system.¹²⁴

A case control study by Kirkland *et al* (1999)⁶⁸ examined surgical site infections and mortality. Over a four-year period (1991-1995) 272 patients who underwent surgical procedures at a hospital in the US acquired a surgical site infection. Of these 255 (94%) were successfully matched with uninfected controls. Mortality rates were found to be twice as high in infected than uninfected patients: 7.8% of infected patients died compared to just 3.5% of uninfected patients.

A study by Leu *et al* (1989) which focused on hospital acquired pneumonia in 115,921 patients admitted to a US hospital over a five year period from January 1st 1979 to December 31st 1983, observed a mortality rate of 30% amongst the 890 patients who acquired a pneumonia (initial episodes only). The attributable mortality, derived from a case-control analysis of a sub-sample of 74 patients, was estimated to be 33%.⁹¹

A case control study by Rose *et al*¹⁰⁸ focussed purely on BSIs occurring in patients admitted to a US hospital over a two year period from December 1972 to August 1974. Over this period 124 patients had a BSI. Medical records were

available for review in 105 cases. Adequate control patients were identified for just 40 of these cases. The crude mortality rate in patients with a BSI was 38% compared to 10% in uninfected controls. The authors acknowledged that the failure to find suitable matches for all cases may have introduced bias into the results. As such cases for which no controls could be found were also reviewed together with an additional 125 BSIs occurring in 113 patients admitted between January 1975 and October 1975. The medical records were available for 109 of these additional cases. The mortality rate amongst the 65 unmatched cases and the 109 additional cases was 34%. A study by Sprengler *et al* (1978)¹⁰⁹ observed a similar mortality rate: 30% of patients with bacteraemia died compared to 4% of uninfected matched controls giving an estimated attributable mortality rate of 26%. Estimates of the risk ratio for death indicated that infected patients were 14 times more likely to die than uninfected patients.

A case control study of hospital acquired BSIs occurring in 4,002 critically ill patients treated on the surgical intensive care unit indicated a lower risk ratio. The study observed a crude mortality rate of 50% in infected patients compared to just 15% in matched uninfected patients.¹¹³ The estimated attributable mortality rate was therefore 35% (95% CI: 25%, 45%), a higher rate than that observed in the study by Sprengler *et al.* However, the risk ratio for death was lower: 3.31 (95% CI: 1.78, 6.15).¹¹³

Other studies which have specifically examined the relationship between bacteraemias and mortality indicate that BSIs resulting from hospital acquired UTIs have a case fatality rate of between 13 and 30%.¹²⁵⁻¹²⁷ However, the majority of deaths occurred in patients with severe underlying disease.¹²⁵

2.6.4 Specialty and site-specific mortality estimates

Some studies have focussed on infections occurring in selected patient groups and/or specific infections occurring in selected patient groups. For example, a number of studies have assessed the impact of pneumonia on mortality in patients treated on intensive care units. A study by Fagon *et al* (1996) involving 1,978 patients admitted to an ICU, observed a death rate in patients who acquired a pneumonia of 52.4% compared to 22.4% in patients who had not acquired a pneumonia. Multiple stepwise logistic regression analysis was subsequently conducted to assess the independent effect of hospital acquired pneumonia on mortality risk in patients admitted to the ICU.⁹² The results of their multivariate analysis that controlled for a range of factors known to be strongly associated with death in ITU patients indicated that pneumonia was an independent risk factor for mortality in these patients (odds ratio: 2.1 (95% CI: 1.6, 2.8). BSIs were also associated with an increase risk of death (OR: 2.5; 95% CI 1.8, 3.6). The acquisition of a UTI was significantly associated with mortality in the univariate analysis but as in other studies was not significantly associated with mortality in the multivariate analysis.⁸⁹

In contrast to this, a study by Craven et al (1988) found that whilst pneumonia was associated with death, it was not among the seven variables remaining after multivariate analysis, thus highlighting some doubt about the direct effect of pneumonia on death.89 The rigorous approach to the diagnosis of Fagon et al (1996)⁹² pneumonia may in part explain the different findings. adopted strict criteria for the diagnosis of pneumonia. The diagnosis was dependent on microbiological evidence of infection. All patients who developed a new and persistent infiltrate and had purulent tracheal secretions underwent immediate fiberoptic bronchoscopy at which time specimens were taken using a protected specimen brush and bronchalveolar lavage. Diagnosis was then This strict approach confirmed from the results of the diagnostic tests. distinguishes their study from earlier studies which included clinically diagnosed pneumonia and thus probably includes patients with lung processes that mimic pneumonia. The authors offer this distinction as a possible explanation for the differences observed between their results and that of earlier studies that failed to identify pneumonia as a risk factor.^{87 89 128} However, it is of course possible that the very process of conducting a fiberoptic bronchoscopy may have increased the patient's mortality risk.

2.6.5 Number of deaths associated with HAIs per annum

Estimates of the number of deaths occurring annually within a given country are also presented in the literature. For example, in the US, Haley combined data from the SENIC study⁴⁹ with data from a concurrent assessment of mortality performed in the NNIS system¹²⁹ to derive estimates of the number of deaths either directly or partially attributable to infection. The results indicated that in 1982, an estimated 19,027 deaths were attributable to HAI and a further 58,092 deaths partially attributable to HAI.¹²² If these estimates are accepted, then in 1982 deaths from HAI alone were the 11th leading cause of death in the US, and deaths that were partly or solely attributable to HAI, were the 4th leading cause of death.¹¹

Equivalent data are not available for the UK. However, assuming a similar mortality rate, it has been estimated that 5,000 in-patient deaths per year might be primarily attributable to HAI and a further 15,000 in-patient deaths might be partially attributable to an infection acquired in hospital.⁹ These estimates are inevitably crude; however, they represent the best estimates available to date.

2.7 Aetiology of HAI

This section provides an overview of the aetiology of HAI drawing on a range of literature in recognised texts on issues relating to hospital infections and where appropriate relevant papers.¹³⁰⁻¹³²

HAIs occur when micro-organisms invade a patient and cause disease. The micro-organisms may be from an endogenous source that is from a site within the patient, or from an exogenous source e.g. from another patient or from the environment. If the causative micro-organisms are normally present in the patient (a commensal) an infection may develop as a result of a change in the relationship between the micro-organisms and the patient. If the micro-organisms are transported from an external source to the patient, an infection may develop if the balance between the agent and the patient's defences favours the micro-organisms. The interaction between micro-organisms, the

route of transmission and patient is called the chain of infection. The environment had a significant impact on all elements in this chain.

2.7.1 Micro-organism/Pathogen

A micro-organism that leads to a disease state in an individual is called a pathogen. It can be a bacterium, a protozoan, a virus or a fungus, although the vast majority of HAIs are caused by bacteria.⁶ ¹⁹ ²⁰ ⁷⁷ The ability of the organism to cause disease, its pathogenicity, varies considerably amongst the diverse members of the microbial world. For example, *Staphylococcus aureus* (*S. aureus*) is a major pathogen responsible for between 10 and 20% of HAIs,⁶ ¹⁹ ²⁰ ⁷⁷ and has relatively high pathogenicity compared to *Staphylococcus epidermidis* (*S. epidermidis*), a member of the same bacterial genus (Micrococcaceae). *S. epidermidis* is a normal skin commensal and rarely associated with significant infection in a non-susceptible host. The relatively high pathogenicity of *S. aureus* can be attributed to the presence of specific virulence factors that enhance its potential ability to cause disease.

Virulence refers to the degree of pathogenicity of an organism and may be described by reference to epidemiological factors including morbidity, mortality and communicability, or by clinical factors characterising the severity of the infection observed. Organisms such as S. aureus, Streptococcus pneumoniae and Mycobacterium tuberculosis can be regarded as principal pathogens, regularly causing disease in immunocompetent individuals. However, when considering the aetiological agents involved in HAI, a considerable number of non-principal organisms Pseudomonas aeruginosa, are implicated. Enterococcus faecalis and indeed S. epidermidis are major causes of HAI and yet rarely cause disease in people with intact host defences. Such organisms can be regarded as opportunists. This opportunism is a direct result of hospitalised individuals being more likely to lack an intact innate or acquired immune system, and therefore make up a more susceptible population.

2.7.2 Route of transmission

Micro-organisms can be transmitted to patients by a variety of routes including direct contact (e.g. hands of staff), indirect contact (e.g. via a piece of equipment), by the airborne route in droplet nuclei, by ingestion (e.g. contaminated hospital food) or by inoculation (e.g. via blood transfusion).

2.7.3 Patient

Individuals are protected from microbial invasion by non-specific and specific defence systems. The non-specific defence system includes the skin, mucous membranes, certain bodily secretions and the inflammatory response. The intact skin provides a tough outer layer which few microbes can penetrate. Antibacterial substances, present in the sweat and the secretions of the sebaceous glands, add further protection, and microbes normally found on the skin (e.g. S. epidermidis) protect against invasion by pathogens through competition for nutrients. The mucus membranes of the respiratory tract produce mucus that traps particles that enter the airway. The cilia then move the mucus upwards to the oropharynx where it is swallowed or expectorated. Lysozyme, present in tears and saliva, is capable of breaking down (lysing) bacterial cell walls especially those of Gram positive bacteria. Finally, the inflammatory response, classically characterised by redness, heat, swelling and pain at the site of invasion represents the initiation of the specific defence system. This system comprises the humoral and cellular arms of the immune system. Both may be acquired naturally through infection or artificially through vaccination.

Within the hospital environment individuals are exposed to greater microbial risk than in the community. On admission to hospital the normal skin flora are often replaced by strains of hospital bacteria that are more resistant to antibiotics and can cause serious infection if they enter the body. Northey *et al* (1974) in a study of intensive care patients found that within two weeks of admission nearly every patient was colonised with infective bacteria and this often involved resistant organisms.¹³³

Medical and surgical therapies often require therapeutic interventions that breach the natural defence mechanisms providing a route of entry for invading micro-organisms. For example, surgical procedures and intravenous therapy result in a break in the integrity of the skin, and urethral catheterisation provides a direct route of entry for micro-organisms to enter the urinary tract: microorganisms may enter during the process of catheterisation, or they may travel retrogressively through, or along, a urinary catheter. Similarly the insertion and presence of an endotracheal tube provides a direct route of access for micro organisms, bypassing the protective action of the mucous membranes and cilia of the lungs, which may also be inhibited by drugs administered.

Whilst the presence of a route of entry increases the risk of infection, it does not necessarily follow that an infection will result. The development of an infection is dependent on the pathogenicity of the invading agent, and the susceptibility of the host, in this case the patient. The very young are particularly susceptible since their immune system is in an immature state. The elderly are similarly at greater risk since their immune system, such as AIDS, leukaemia and other haematological malignancies are particularly vulnerable to infection. These patients who are generally immuno-suppressed and as such susceptible to infection, are made more vulnerable by the toxic effects of the drug therapy they receive. See section 2.8 for a more detailed discussion of risk factors for infection.

2.7.4 Environment

Environmental factors such as temperature, air movement and the presence of chemicals, gases and toxins may have an effect on any of the factors involved in the development of infection. Particular environmental factors may limit, inhibit or prevent the development of an infection. For example, environmental factors such as temperature and humidity may promote or inhibit the growth of micro-organisms in their reservoir. Movement and velocity of the air may affect transmission of micro-organisms from source to susceptible host.

2.8 Risk factors for HAI

As indicated above infections occur as a result of complex interactions between specific factors relating to the patient's condition, the treatment administered and the environment (see section 2.7). The risk of acquiring an infection in hospital therefore varies considerably from one patient to another. Studies that have assessed risk factors for HAI have identified a number of factors that are significantly associated with the presence of HAI. However, these factors are not necessarily causative factors – they may simply be closely related to the actual risk factors may occur simultaneously with other factors, exerting an additive or even synergistic effect: that is they are interrelated. Thus when examining risk factors for infection the problem is not only to find factors that are significantly associated with the presence of an infection.

The technique adopted to do this has generally been multiple logistic regression analysis. Those factors identified in the univariate analysis as being significantly associated with the presence of an infection are entered into a logistic regression analysis in an attempt to identify independently significant factors. However, the possibility still arises that the factors identified are markers for other factors. Thus the validity of the results obtained are to some extent dependent on the completeness of the original variables introduced – how comprehensive the list of variables included was.

Risk factors for infection can be classified into two broad groups: intrinsic risk factors and extrinsic factors. Intrinsic risk factors are specific factors relating to the patient's condition that place them at risk of acquiring an infection, they reflect the patient's susceptibility to infection. Patients with conditions that affect the immune system, such as leukaemia and HIV, are inevitably at greater risk of acquiring an infection than patients without these conditions. Other intrinsic risk factors include old age, diabetes, and obesity. Extrinsic risk factors relate to factors external to the patient and include factors that increase the risk of micro-

organisms entering the patient. For example, medical interventions that breach the intact protective skin barrier and urethral catheteristaion which provides a direct port of access for invading micro-organisms.¹⁰⁷

There are many examples in the literature of studies that have examined risk factors for HAIs at any site or specific sites.⁶² ¹³⁴ ¹³⁵ ¹³⁶ ¹ ¹⁹ ³³ ⁵⁷ ⁶⁶ ⁷¹⁻⁷⁴ ⁸³ ¹³⁷⁻¹⁴² The following sections examine this literature in more detail, focusing on studies that have assessed risk factors for HAIs at all sites and then studies that have focussed on risk factors for surgical wound, urinary tract, respiratory tract and bloodstream infections.

2.8.1 Risk factors for HAIs at any site

Studies that have examined risk factors for HAIs have identified a range of factors including age;⁸³ length of pre-operative stay;^{19 83} severity of underlying illness as defined by factors such as diagnostic group, number of co-morbidities, the American Society of Anaesthesiologists score (ASA score), specific co-morbidities, and blood loss during surgery; type of admission,^{37 57 83} practising surgeon,^{33 74} malnutrition,¹⁴⁰ presence of devices such as central venous or urinary catheters,³³, and wound class (clean, clean contaminated, contaminated or dirty).

Some of the risk factors identified in these studies appear to be universally applicable to most patient populations, whereas others appear to be unique to the sample studied. Whilst in some cases this may reflect a genuine difference in the risk profile of the patients studied, in others it may simply reflect differences in the approach taken to assess risk factors for HAI, and, in particular, the range of explanatory variables included in the analysis.

As indicated above, studies have generally identified more than one risk factor for infection within a patient population. For example, a logistic multiple regression analysis of data collected for a Spanish prevalence study identified the following risk factors: the number of intrinisc risk factors (coma, renal failure

97

diabetes, hypoalbuminaemia, pressure sores, alcoholism, smoking and drug addiction); number of extrinsic factors (urinary catheterisation, peripheral vascular catheterization, central catheterisation with peripheral insertion, parenteral nutrition, tracheostomy and mechanical ventilation); baseline risk (three categories of clinical prognosis: severe, moderate and mild); length of stay prior to infection; and number of diagnoses. These factors were all found to be significantly associated with infection (p<0.001).¹⁹ Interestingly whilst infections rates were found to increase with age, the results of the logistic regression indicated that age was not significantly associated with infection when other factors were taken in to account. In contrast an Italian prevalence study found that after controlling for a range of factors in a multiple logistic regression analysis, age was significantly associated with HAI, together with ward of stay, urinary catheterisation and receiving antibiotics.³⁷

Another example is a study by Meltomaa *et al* (2000)⁵⁷ that assessed the risk factors for HAIs occurring in 687 women who underwent a hysterectomy at a University Teaching Hospital in Finland between October 1993-September 1994. The results of the multivariate analysis indicated five significant risk factors for an infection at any site: lack of antibiotic prophylaxis, blood loss during surgery of over 100mls (blood loss greater than 100 mls and less than 300 mls was associated with greater risk than blood loss in excess of 300mls, a situation which may have been due to patients with over 300mls of blood loss in some cases receiving a blood transfusion); intermittent catheterisation; anaemia; and post-operative administration of laxatives and cholinergic agents for urinary or bowel problems. However, when UTIs were excluded from the analysis, intermittent catheterisation ceased to be a risk factor.

2.8.2 Risk factors for surgical wound infections

Studies have identified a number of intrinsic and extrinsic risk factors for SWIs. Intrinsic risk factors include age, severity of illness,⁶⁶ ⁷³ diabetes⁶⁶ wound class,⁷² ⁷³ infections at other sites prior to surgery,¹³⁹ malnutrition,¹⁴⁰ obesity¹⁴¹ and weight loss of greater than 10% during the six months prior to surgery.⁶⁶

Extrinsic risk factors include length of pre-operative stay, 62 74 83 the administration of a pre-operative shave. 62 74 length of surgery; 62 66 73 97 and the presence of prophylactic abdominal drains, which have been shown in both clinical and experimental studies to be associated with higher infection rates. Mishriki et al (1990)⁷⁴ in a study of 702 adult, non- trauma patients assessed how the incidence of SWI rate varied with 23 potential risk factors: age, sex, pre-operative stay (days), current antibiotic treatment, co-existing diabetes, current immunosuppression, nutritional state, coexisting metabolic disease, preexisting malignancy, preoperative shaving, prophylactic antibiotic treatment, type of skin preparation, duration of operation, operating surgeon, operating theatre, operation site, size of wound, type of wound closure, suture technique, use of deep tension sutures, type of wound dressing, use of surgical drains and main ward of stay (a modified data set was used for day cases). The authors selected these 23 factors as they had been shown in earlier studies to be related to SWIs. Of these 23 potential risk factors, only four were found to be significantly associated with SWIs: age> 55 in clean, clean-contaminated and dirty categories (p= <0.05); pre-operative stay > 3 days in clean-contaminated category (p = <0.05); pre-operative shaving in the contaminated group (p=<0.005); and individual surgeon in the clean category (p=<0.001). A forward stepwise regression analysis of ten independent variables was undertaken (age, preoperative stay, shaving, prophylactic antibiotics, skin preparation, surgeons, operating time, suture type, deep tension sutures, drains). It is not clear from the paper why these 10 were selected from the 23 possible variables. However, the results confirmed the univariate analysis although the overall contribution of these variables in the clean and clean-contaminated category was found to account for a small proportion of the overall variation. The The association between surgeon and SWI had greatest significance. elimination of a single surgeon's case load from the sample would have reduced the SWI rate by over 40%.

Garbaldi *et al* (1991)⁷³ assessed the risk factors for SWI in a study of 1852 adult surgical patients admitted to a university affiliated US hospital over a four year period from January 1982 to January 1986. Four independent variables which were highly predictive of SWI were identified from a stepwise logistic regression analysis of potential risk factors. These were wound class; ASA score; duration of procedure; and the results of an intra-operative culture However, the presence of intra-operative positive cultures had little explanatory power. The predictive power of a positive culture was low (32%) and the false positive rate was high (82%) and concordance with isolates from infected wounds low (41%).

A study by Kampf *et al*¹⁴² utilised data collected as part of the first national prevalence survey conducted in Germany in 1944. The results of the multiple logistic regression analysis which included a number of variables found that the department (surgery or intensive care), age (>45 years), diabetes mellitus, male sex and size of hospital (> 600 beds) were significantly associated with SSIs.

2.8.3 Risk factors for urinary tract infections

The presence of a urinary catheter and the duration of catheterisation are key risk factors for this type of infection.¹ ¹³⁴ ¹³⁷ ¹³⁸ ¹⁴³ ¹⁴⁴ While 1% of noncatheterised patients will develop bacteriuria, between 10 and 30% of catheterised patients will develop bacteriuria and between 10 and 30% of these patients will develop symptoms of a UTI.81 During the process of catheterisation bacteria may be introduced into the bladder, and following insertion, the bacteria may migrate from the peri-urethral area into the bladder, either through the catheter lumen or along the catheter - mucosal surface. Studies have indicated that the process of catheterisation is associated with a 1% risk of bacteriuria, and each subsequent day that the catheter is in situ there is a 3-10% risk of developing bacteriuria.¹⁴³ ¹⁴⁵ Following removal of the catheter, patients continue to remain at risk of acquiring an infection. For example, Harstein et al found that 11% of catheterised patients developed bacteriuria within 24 hours of the catheter being removed.¹⁴³

Urinary catheters are frequently used in the care and treatment of patients in both the acute and long term care setting. A study of indwelling catheterisation and related nursing practice in adult patients admitted to specialties other than mental handicap, psychiatry or obstetrics at five district general hospitals in England found the prevalence of catheterisation was 12.6%.¹⁴⁶ Data from the US suggests that between 15 and 25% of patients in the acute care setting were catheterised. ⁸¹

Catheterisation rates vary with specialty and from one care setting to another. The results of a recent audit of infection control policies and practices in 19 district general hospitals in England and Wales clearly demonstrates this.¹ The median catheterisation rate observed in adult patients admitted to selected specialties at 19 different hospitals, and who had a minimum in-patient stay of three days was considerably higher in patients admitted to the gynaecology specialty (40.4%: range 20.9% – 72.0%), when compared to the medical (11.6: range 5 –17%); surgical (34.4%: range 16.2-50.0%); and orthopaedic (17.3%: range 10.1 – 26.0%) specialties.

In addition to duration of catheterisation a number of other independent risk factors for UTIs in catheterised patients have been identified. A study of 1474 catheterised patients identified eight additional independent risk factors: absence of use of a urinemeter, microbial colonisation of the drainage bag, diabetes mellitus, absence of antibiotic use, female patient, indications for catheterisation other than drainage during surgery or output measurement, abnormal serum creatinine and catheter care violations.¹³⁸ More recently, Glynn *et al* (1997)¹ identified seven risk factors in addition to catheterisation: the risk was found to increase with increasing number of catheterisations, an ICD9 diagnosis other than endocrine diseases; female sex; transfer or emergency admission as opposed to elective admission, increasing age; and length of hospital stay. However, as pointed out by Glynn *et al* (1997) it was not clear whether longer length of stay was a risk factor for this type of infection or a result of the infection itself.

A study by Kampf *et al*¹⁴² utilised data collected as part of the first national prevalence survey conducted in Germany in 1944. The results of the multiple logistic regression analysis which included a number of variables found that unconsciousness, old age (>75 years) prior operation, female sex and the size of the hospital (>200 bed) were all significant risk factors for UTIs. The authors argued that the first four risk factors were likely to reflect need for urethral catheterisation, which as indicated above, and acknowledged by the authors has been shown to be a key risk factor for this type of infection. With regard to hospital size they suggest that this may reflect the lower use of urine cultures in smaller hospitals, which can be critical to a diagnosis of UTI when using the CDC definitions.

2.8.4 Risk factors for lower respiratory tract infections

Specific risk factors for lower respiratory tract infections (LRTIs) include the presence of a nasogastric tube, mechanical ventilation, aspiration, specific lung disorders such as chronic airway disease and depressed consciousness.¹ ¹⁴⁷ However, as with infections at other sites the severity of the patient's condition will also have an impact on the risk of infection. The precise range of factors identified in these studies again varies from one study to another, which, as discussed above, may reflect both genuine differences in the risk profile of the population studied and/or the comprehensiveness of the methods employed to assess risk. For example, a recent audit of infection control in 19 hospitals in England and Wales identified the following risk factors for LRTIs occurring in adult patients admitted to medical and surgical specialties: the presence of a naso-gastric tube with the risk increasing with the number of days the tube was in place (1 day; 2 or more days); the presence of other devices with the risk again increasing with the number of days these were in situ (1 device, 2+devices); an ICD9 grouping other than endocrine diseases; four or more discharge diagnoses; sex with males at greater risk than females; age over 50; and length of stay with the risk of infection increasing with increasing length of stay (6-10 days, 11-15 days, 16+ days). In contrast Kampf et al (1998)¹⁴⁷ in their analysis of data from a prevalence study conducted in Germany identified

the following factors as being significantly related to the presence of a LRTI: polytrauma, impaired consciousness, chronic airway diseases, prior surgery and cardiovascular diseases. The differing results may simply reflect the fact that different explanatory variables were included in the analysis. For example, the study by Kampf *et al*¹⁴⁷ failed to include data on the presence of a naso-gastric tube, but included far more detailed information on diagnosis than the study by Glynn *et al* (1997).¹

2.8.5 Risk factors for bloodstream infections

In addition to factors relating to the patient's underlying susceptibility to infection, a number of specific risk factors for BSIs have been identified in the literature including the presence of intravenous and central venous lines, invasive procedures, and the presence of infections at other sites – in particular the presence of a UTI.^{1 148} Glynn *et al* 1997¹ in a multivariate analysis of data on adult patients admitted to 19 hospitals in England and Wales identified the following risk factors: the number of central venous catheters; the presence of other infections, an ICD9 diagnosis other than endocrine diseases (the reference category); three or more discharge diagnoses, and length of hospital stay, the risk increasing with increasing length of stay. However, as indicated above it was not clear from the results of their analysis whether the longer length of stay in infected patients acted as a risk factor for the infection or was a consequence of an infection acquired in hospital.

2.9 Risk indices

Based on the information derived from studies of risk factors for infection, composite measures of risk, based on a combination of risk factors, have been developed. These measures, described as risk indices, aim to control for the underlying infection risk of a patient population and thus facilitate the valid comparison of rates of HAI from one population to another. Risk indices have primarily been developed to assess the risk of acquiring a surgical wound infection, although some have also proved to be good predictors of infections at other sites.

For example, based on data from the US Study of the Efficacy of Nosocomial Infection (SENIC) Haley *et al* (1985)¹⁴⁹ developed a risk index for SWI. Of the ten potential risk factors for SWI included in the logistic regression analysis, four were found to be significantly associated with SWI: an operation involving the abdomen; an operation lasting over two hours; an operation classified as contaminated or dirty-infected based on the traditional wound classification system and three or more diagnoses on discharge (a proxy for intrinsic patient risk). In the final model all four risk factors had nearly equal coefficients. As such the risk index developed weighted all four risk factors equally, and patients were allocated to one of five groups based on the number of risk factors present: no risk factors, one, two, three and four risk factors.

The US NNIS surgical wound infection risk index, is an adaptation of the SENIC index.⁶⁷ Each surgical patient undergoing an operative procedure is allocated to a risk group based on how many of the following risk factors the patient had.

- American Society of Anaesthesiologists (ASA) pre-operative score of 3, 4 or
 5.
- 2. An operation classified as either contaminated or dirty-infected according to the traditional wound classification system.
- An operation with duration of surgery more than T hours, where T depends on operative procedure performed.

The ASA score is an index designed to assess the pre-operative health status of patients. It ranges from 1 for an otherwise healthy patient to 5 for a patient not expected to survive the next two hours. Thus it is a slightly more detailed approach to assessing intrinsic risk than that used in the SENIC index of three or more discharge diagnoses. Furthermore it is assessing health status prior to surgery, whereas in the SENIC index assessment of health status is based on number of diagnoses at time of discharge from hospital, which in infected patients could be affected by the presence of an infection.

The NNIS index also differs from the SENIC index with respect to the information included regarding length of surgery. In the SENIC index there is simply a two hour cut off point: patients with surgery lasting more than two hours are considered to be at greater risk than patients whose surgery took less than two hours. In the NNIS index, only those patients whose surgery took longer than might be expected (T) are included. The 'T' value, as it is known, was determined from information on the distribution of duration of surgery for different operative procedures. The 75th percentile of each distribution was identified and rounded to the nearest whole hour and used as the cut off point for distinguishing between short and long duration. The T value identified in this way is thus a means of identifying surgery that is taking longer than usually expected. A study by Culver et al (1991) involving data on 84,691 operations performed on patients admitted to 44 NNIS hospitals during the period January 1987 through to December 1990, found the NNIS indices to be a better predictor of SWI risk than the traditional wound classification system.⁶⁷ It was also found to be a reasonable predictor of postoperative infections at other sites.⁶⁷ The application of this index elsewhere is widespread. However there are a number of important considerations. The T value was calculated from the distributions of duration of surgery in patients undergoing operative procedures within the NNIS scheme. It is questionable how accurately these reflect the distribution of duration of surgery conducted elsewhere and at different points in time. It is feasible that with the advancement of technology some procedures conducted today are carried out in less time. There is thus a need to question the validity of the T values prior to incorporating them in the index and if possible develop specific T values for the health care setting of interest.

The other issue to consider is that two of the risk factors, wound class and duration of surgery, are potentially also markers of quality of care. Consequently, adjustment for them in a comparative analysis of rates may mask rather than highlight a potential problem area.

2.10 The problem of microbial resistance to antibiotics and other antimicrobials

Microbial resistance to antibiotics is a problem which, at the time of their discovery in the late 1930s, was unimaginable. Antibiotics were viewed as the magic bullet designed to treat all infections, and they undoubtedly led to safer medical and surgical practice and increased life expectancy.¹⁵⁰ However, soon after their discovery problems relating to resistance developed as microorganisms developed mechanisms to circumvent these drugs, and there is now the very real prospect of a post-antibiotic era.¹⁵¹ Bacterial evolutionary responses to the selective pressure of antibiotics have resulted in microorganisms resistant to virtually every known antibiotic. In 1941 virtually all strains of S. aureus were sensitive to penicillin. Within three years of its introduction several strains of S. aureus became capable of β lactamase production, enabling hydrolysis of the β lactam structure present in penicillin, thereby removing the drugs clinical efficacy.¹⁵⁰ In an endeavour to overcome this specific resistance problem, semi-synthetic penicillins were produced. Methicillin was the first such synthetic penicillin, but was superseded by a less toxic derivative, flucloxacillin, although methicillin is still used for in-vitro sensitivity testing as it has identical resistant patterns to those of flucloxacillin.¹⁵² Soon after the drugs introduction methicillin resistant S. aureus (MRSA) developed,¹⁵³ and the frequency of isolation of this organism has increased steadily. MRSA strains are resistant to all penicillin derivatives and are in many cases resistant to other antibiotics. Vancomycin and teicoplanin are the only two consistently effective agents of use clinically.¹⁵⁰ Many other examples of antibiotic resistant micro-organisms can be cited. For example the global rise of vancomycin resistant multi-drug resistant enterococci (VRE) and Mycobacterium tuberculosis (MDRTB). A number of reports have documented the scale of the problem and situation.¹⁵⁴⁻¹⁶⁰

2.10.1 The development of antibiotic resistance

Antimicrobial resistance may be innate or acquired. Innate resistance refers to bacteria inherently resistant to one or more antibiotics. Many of these bacteria do not represent a threat to healthy humans, but may give rise to infection in hospitalised patients. Examples include the *Pseudomonas* species and some *Enterococci*.

Acquired resistance may develop as a result of mutations in a small proportion of a bacterial population. In the presence of the antibiotic to which it is resistant, the proportion of these altered bacteria multiply and become more dominant. For example, some mycobacterium *tuberculosis* are naturally resistant to the antibiotic streptomycin. In the presence of this antibiotic these bacteria soon become dominant in a population.

Acquired resistance may also develop through the transfer of genetic material encoding resistance from one bacterium to another. This can occur through the direct transfer of genetic material on plasmids, on a bacterial virus or a bacteriophage, or via the direct transfer of DNA. As before in the presence of antibiotics, the susceptible bacteria are killed, thereby selecting out resistant strains that subsequently become more dominant in the population.

More specifically antibiotic resistance develops through a variety of molecular mechanisms that pathogens have developed to circumvent antimicrobials. The House of Lords report on antibiotic resistance describes five broad mechanisms that give rise to antibiotic resistance: inactivation, where the bacteria can inactivate the drug before it reaches its target within the bacterial cell; impermeability, where the outer layers of the cell is impermeable preventing the drug from entering the cell; alteration of target site, where the target is altered so that it is no longer recognised by the antibiotic; efflux, where the drug enters the bacteria but is then pumped out; and by-pass where the bacteria acquire an alternative metabolic pathway resulting in the antibiotic's target being made redundant.¹⁵⁴
Inappropriate prescribing is a key factor in the development of resistance. This includes the inappropriate use of antibiotics for self-limiting or viral infections such as colds, and some sore throats, and where antibiotics are justified, an inappropriate drug and/or treatment regime selected. The House of Lords Report on antibiotic resistance comments that in the UK, whilst the present use of antibiotics is conservative compared to that in other countries, between 5-50% of antibiotic prescriptions are inappropriate, the proportion varying with geographical location.¹⁵⁴ In the US it has been estimated that over half of the antibiotics prescribed are prescribed inappropriately; that is antibiotics are not indicated or they are incorrectly prescribed.¹⁶¹ The availability of antibiotics over the counter in some countries also contributes to the situation, as does the failure of patients to complete a full course of antibiotics.

There is some evidence to suggest that the inappropriate use of antibiotics in animal husbandry is another contributing factor. Of particular concern is the use of antibiotics in low quantities as growth promoters. Antibiotic resistance may be transferred directly from animals to humans via the food chain, or resistance may be transferred from the communal bacterial populations of animals to the commensal bacterial populations of humans and this transferred resistance eventually evolving into human pathogenic forms.¹⁵⁴

2.11 Conclusion

This chapter has examined the extent of the problem of HAI in terms of its frequency and distribution as measured by prevalence and incidence studies, the mortality risk associated with acquiring an infection in hospital, the aetiology of and risk factors for HAIs, and the problem of antibiotic resistance. It is clear from the literature that HAIs affect a substantial number of patients every year, and that acquiring an infection in hospital increases an individual's mortality risk and in some cases can directly cause death or substantially contribute to death. It is also clear that the risk of acquiring an infection varies considerably from one patient to another, and a number of risk factors for these infections have

been identified. Some are directly related to the patient's condition (intrinsic risk factors), whilst others relate to the process of treatment and care (extrinsic risk factors). The recognition of these factors provides valuable information that can then be used to plan care and treatment which aims to reduce the impact of these risk factors. It is also clear from the literature that microbial resistance to antibiotics is a growing problem and one that needs to be tackled now.

This thesis aims to provide more timely data on the incidence of HAI occurring in adult patients admitted to five surgical specialties common to most general hospitals. It also aims to identify independent risk factors for these infections, and provide some data on the impact HAI has on mortality risk in this patient group. The thesis will not examine the important area of antibiotic resistance. This was considered beyond the scope of both the underlying study and this thesis. The following chapter will consider the economic burden of HAI and the potential economic benefits of prevention.

CHAPTER 3 THE ECONOMIC EVALUATION OF HAI

3.1 Introduction

Economic evaluation is concerned with the assessment of the costs and consequences of alternative practices,¹⁶² thus enabling the comparison of alternative interventions in order to obtain the maximum health gain for a given expenditure, or the lowest expenditure for a given health gain.

In the context of hospital acquired infection (HAI), it is clear from the preceding chapter that HAIs affect a substantial number of patients every year. Whilst it is unlikely that all infections can be prevented,¹⁶³ the evidence suggests that a proportion of these infections could be prevented through improvements in infection control practice.¹⁶⁴ Evidence from the study of the efficacy of nosocomial infection control (SENIC) conducted in the US indicated that hospitals with an organised hospital wide infection control programme which included intensive surveillance, the feedback of results to those who need to know, one infection control nurse per 250 beds and the involvement of an infection control physician or microbiologist, on average achieved a 32% reduction in infection rates over time.¹¹ This evidence informed guidance issued by the Department of Health in 1995,⁹ which stated that about 30% of HAIs could be prevented through improvements in infection control.

More recently, the National Audit Office (NAO) in England conducted a survey of NHS hospitals in which they asked infection control teams (ICTs) whether they believed a 30% reduction in rates could be achieved in their NHS Trust: 39% thought this could be achieved, 49% thought the estimate was too high, and the remaining 12% either did not know or did not answer the question. The NAO survey also asked ICTs to estimate what proportion of infections they considered to be preventable in their Trust. The responses varied from less than 5% to over 35%. The average percentage reduction across all NHS Trusts, adjusted for the number of beds in the individual hospitals that responded, was 15%.³ Whilst this is a subjective assessment of the proportion of HAIs that are preventable, it does demonstrate that there is a strong belief that a proportion of infections can be prevented. Evidence from studies which have assessed the incidence of HAI overtime in response to various interventions provides harder evidence that a proportion can be prevented (see section 3.6.2).

Whilst it is clear that a proportion of HAIs can be prevented, the prevention of HAIs is not cost free. Economic evaluation is thus concerned with estimating the costs associated with these infections, and the cost of activities that aim to prevent infection and their economic consequences, thus providing valuable data which can contribute to decision making regarding the allocation of resources to infection control and their use.

This chapter considers the literature relating to the economic evaluation of HAI. It begins with an overview of the processes and techniques of economic evaluation and discusses these in the context of HAI. Studies that have assessed the economic burden of HAI and the benefits of investment in prevention activities will then be reviewed.

The literature reviewed was identified through a series of consecutive searches, carried out during the period 1993 – March 2003, using the computerised bibliographic databases Medline and PubMed, supplemented by reference follow up, hand searching of selected journals, and consultation with experts in the field. The search was limited to papers published in the English language during the period 1975 – March 2003, and further limited to studies conducted in the developed countries of Europe, Canada, Australia and the USA. As with the literature review reported in Chapter 2, the decision to limit the review to literature published from the mid 1970s onwards, was based on the fact the Study of the Efficacy of Nosocomial Infection was published at this time, representing the beginning of a period of renewed interest in HAI.

The search strategy involved the use of selected key words (thesaurus terms) and combining these with a number of "free text" words. The key words used were: hospital acquired infection; nosocomial infections; health care associated infections; hospital-associated infections. These were combined with the following "free text" words: economic; burden; costs; resources; cost of illness; evaluation; length of stay. The abstract of each paper was subsequently read and a decision made as to whether the paper was relevant to this review. Those papers that were concerned with methodological approaches to assessing the economic burden of HAIs and studies that specifically assessed the economic burden resulting from HAIs occurring in adult non-day case patients cared for in specialties common to most hospitals (medicine, general surgery, urology, care of the elderly, orthopaedics, gynaecology and obstetrics) were considered to be relevant. Papers that were limited to day case patients, children, or patients admitted to specialities not included in this study were excluded from the review.

References cited in papers that had not been identified through the computerised search were subsequently followed up and hand searches of the following journals were carried out: Journal of Hospital Infection; Infection Control; Epidemiology and Infection Control; American Journal of Infection Control, Infection Control and Hospital Epidemiology, Current Issues in Infection Control; British Medical Journal. Experts in the field were also consulted to identify grey literature such as project reports and policy documents of relevance to this review. Experts included microbiologists, infection control doctors, infection control nurses and health economists from the Central Public Health Laboratory Service, the London School of Hygiene and Tropical Medicine, the National Audit Office, and NHS Trusts in England. The published results of the study of the socio-economic burden of HAI, the study on which thesis is based, are not presented in this chapter, but presented in Chapter 4.

3.2 Economic evaluation

Economic evaluations take a number of different forms including the following approaches: cost of illness, cost-effectiveness, cost utility, and cost-benefit analysis. Drummond (1987), when describing the various techniques available, classifies them into two distinct groups based on whether they enable a partial or full economic evaluation.¹⁶²

3.2.1 Partial economic evaluations

Economic evaluations which do not include a comparator and fail to include both costs and consequences may be considered partial evaluations. They may be limited to a description of the outcome, or the costs, or include both costs and outcomes but no comparator. Alternatively they may include a comparison of two or more alternatives but be limited to the analysis of the efficacy or effectiveness or alternatively the costs involved.

Cost of illness (burden of illness) studies are examples of partial economic evaluations. These studies are limited to an assessment of the costs associated with a particular illness. They aim to identify, measure and value the direct, indirect and intangible costs of a particular illness. The approach represents one of earliest forms of economic evaluation.¹⁶⁵ For example, Petty in the 17th century employed a human capital approach to assess policies such as moving people from inner London to Hampstead Heath to prevent them catching the Plague, and later in the mid-nineteenth century cost of illness studies were used to justify public health activities.¹⁶⁵

Cost of illness (COI) studies became increasingly popular in the 1950s and 1960s,¹⁶⁶ following which their popularity declined with many questioning their value. In the past COI studies have been conducted to demonstrate the economic burden of a disease and thus facilitate comparison of the burden of different diseases and assist in decision-making regarding prioritising health care and the subsequent allocation of resource. However, it has been argued that data limited to the burden of disease has little value and may be

misleading. By definition these studies are limited to the cost of the disease and do not attempt to look at the potential effectiveness or cost of interventions to prevent or treat the illness, and the benefits that might result. As such COI studies do not provide direct guidance on the allocation of resources.¹⁶⁷

Byford *et al* (2000) argue that without data on the costs and effectiveness of prevention activities informed decisions about the allocation of resources cannot be taken. The situation might arise when a decision is taken to allocate more resources to the treatment or prevention of a particular disease simply because it is more expensive than the others that are being considered. However, without knowledge of the cost of the activities being considered and their effectiveness the decision might result in additional resources being used for little health gain.¹⁶⁸

Similarly illnesses which impose a relatively small burden, but are easily preventable may be overlooked. Byford *et al* (2000)¹⁶⁸ cite the example of phenylketonuria, a disease which has a relatively low incidence and imposes a relatively small burden on society, but which is easily preventable at low cost, resulting in substantial health gains to the individuals concerned. Byford *et al* (2000)¹⁶⁸ thus argue that the results of some COI studies may divert the attention of policy decisions makers away from areas where improvements and health gains can be made at low cost

However, despite these limitations, COI studies do provide valuable data. Whilst the results do not represent the net benefits of investment in a particular activity, they do represent the resources that might become available if the disease did not exist. As such the results represent the potential benefits that might arise in terms of costs avoided, if the disease could be eliminated. These data can subsequently be used in conjunction with data on cost and effectiveness of activities which aim to prevent the illness and the results used to inform policy. The results of COI studies can also serve to highlight the magnitude of a problem and the items that are most costly, so facilitating

114

managers' attempts to improve the efficiency of provision. They may also be influential in setting the agenda for policy initiatives and for initiating further evaluative studies.¹⁶⁹ In the context of HAI a review of the literature indicates that a number of COI studies have been conducted; these are reviewed in section 3.4.

3.2.2 Full economic evaluations

Economic evaluations that assess both the costs and consequences of two or more alternatives may be classified as full economic evaluations.¹⁶² Cost-effectiveness analysis, cost utility analysis and cost-benefit analysis are all techniques that might be used in a full economic evaluation.

3.2.2.1 Cost–effectiveness analysis

This type of analysis involves the comparison of alternative activities, which produce health outcomes that can be measured in the same units. For example, it might involve the comparison of interventions whose outcome can be measured in terms of the cost per life year gained or in the context of HAI the cost per infection averted. Alternatively the results can be presented as life years gained (or the number of infections averted) per pound spent. This type of analysis therefore allows the comparison of interventions with differing costs and levels of effectiveness. Furthermore, providing the interventions considered have a common effect, this approach can be used to compare the costs and consequences of a range of different interventions

3.2.2.2 Cost–utility analysis

In this type of analysis the outcome is expressed in terms of utility thus reflecting the preferences individuals or society have for a particular set of health outcomes. It thus assesses outcomes in terms of quality of life with the frequently used measure being quality adjusted life years (QALY). Other alternatives include healthy years equivalent (HYE) and disability adjusted life years (DALY). The results of this type of analysis are expressed in terms of the cost per quality adjusted life year gained, cost per healthy year gained, or cost

per disability adjusted life year gained by undertaking one intervention instead of another. Providing utility measures can be developed this approach allows the comparison of a range of interventions, and their multiple effects. However, the approach is frequently hampered by difficulties in obtaining appropriate utility data.

3.2.2.3 Cost-benefit analysis

Cost-benefit analysis involves a monetary value of both the costs and the benefits of the interventions being considered. Thus this type of analysis can be used when outcomes can be measured in financial terms, and enables the comparison of alternatives whose outcome cannot be reduced to a single common unit of effect and interventions that produce multiple outcomes. In the context of HAI, the costs of a range of activities aimed at reducing the risk of infections could be examined and a monetary value applied to the outcome measure of the number of cases of HAI averted.

3.3 The economic evaluation of HAI

In 1992 Haley commented that the number of studies estimating the economic burden of HAI has increased sharply since the mid 1970s.¹⁷⁰ Since that time many more studies have been published. Many of these are partial evaluations adopting a cost of illness approach, whereas others have attempted a full economic evaluation adopting techniques such as cost effectiveness and cost benefit analysis. Economic modelling, utilising data from a variety of data sources, has also been conducted. Studies which have included an assessment of the economic burden of HAI are discussed in section 3.4 and those that have assessed the cost and benefits of investment in infection control are considered in section 3.5.

3.4 Estimates of the economic burden of HAI

As indicated above, a number of studies have assessed the economic burden of HAI. Tables 3.1 - 3.6 provide an overview of studies conducted since 1975 that have assessed the burden of HAI to the hospital sector and the estimates derived. Table 3.1 presents estimates of the economic impact of all types of HAI and Tables 3.2 - 3.6 present estimates of the economic impact of specific types of infection.

Table 3.1:

An overview of studies that have assessed the impact of HAIs (all sites) on hospital length of stay and costs (1975-2002)

								Additional	Additional
ar of Country Types of patients Type of No. hosps included in Method dy studied	Types of patients Type of No. hosps included in Method studied in Method	lype of No. hosps included in Metho nospital No. hosps included in Metho	No. hosps included in Method analysis	vo. HAIs ncluded in Methor analysis	Wethor	d of attribution	Characteristics used in natching	days in hospital	hospital costs
975 USA All admissions Not stated 1 120 Case	All admissions Not stated 1 120 Case	Not stated 1 120 Case	1 120 Case	120 Case	Case	control - exact match	Diagnosis code, operative procedure, hospital service, age	13.4	\$1018
875 USA All admissions Not stated 1 58 Case	All admissions Not stated 1 58 Case	Not stated 1 58 Case	1 58 Case	58 Case	Case	control - close match	Diagnosis code, operative procedure, hospital service, age	16.1	\$1224
975 USA All admissions Not stated 1 183 Comp	All admissions Not stated 1 183 Comp	Not stated 1 183 Comp	1 183 Comp	183 Comp	Comp	arative - unmatched	NA	17.0	\$1292
975 USA All admissions Not stated 1 183 Physi	All admissions Not stated 1 183 Physi	Not stated 1 183 Physic	1 183 Physic	183 Physi	Physi	cian assessment	NA	4.8	\$576
978 USA All admissions teaching 1 123 Author hospital	All admissions teaching 1 123 Autho hospital	Community 1 123 Author teaching 1 123 Author hospital	1 123 Autho	123 Autho	Autho	r assessment"	NIA	3.0	\$636
75-6 USA All admissions Mixed 3 256 Conc	All admissions Mixed 3 256 Conc	Mixed 3 256 Conc	3 256 Conc	256 Conc	Conc	urrent	NA	4.5	\$590
978 UK Orthopaedic District 1 29 Case house the house of the case house the case house the case of the	Orthopaedic District 1 29 Case horoital 1 29 Case	District District 1 29 Case househald	1 29 Case	29 Case	Case	e control	Sex, age, operation, diagnosis, Quetalet's index, smoking habits, social class	17.0	£775
979 Israel General surgical & University 1 90 Case orthopaedic	General surgical & University 1 90 Case orthopaedic affiliated	University 1 90 Case	1 90 Case	30 Case	Case	e control	Sex, age within 5 (sometimes 7) years, pre-operative diagnosis, operative procedure, month of hospitalisation, post-op stay at least 7 days	6.7	\$787
988 UK gynaecology general 1 67 Case patients who had a hospital 1 67 Case survival proceeding	General surgical, orthopaedic & District gynaecology general 1 67 Case patients who had a hospital surgical succedure	District District 1 67 Case hospital	1 67 Case	67 Case	Case	e control	Primary features: first operative procedure & primary diagnosis. Secondary features: sex, age (+or - 10 years) surgical service	8.2	£1,041

	ind costs (1975	-2002)						
ountry	Types of patients studied	Type of hospital	No. hosps	No. HAIs included in analysis	Method of attribution	Characteristics used in matching	Additional days in hospital	Additional hospital costs
in a start where the start whe	General and digestive surgical patients	Not stated	-	5	Case control	Primary features: primary diagnosis, operative procedure, surgical category (clean, clean-contaminated contaminated, dirty), age (+or – 5years and in one case 6 years). Secondary features; presence of neoplastic or endocrine disease, elective or emergency procedure, number of days (+or – 5 days) before infection with an invasive device in situ.	11.4	6 111
lurkey	General surgery	University tertiary hospital	-	223	Crude comparison	NIA	17.0	1
l'urkey	General surgical	Urriversity tertiary hospital	-	151	Case control	Age, length of pre-operative stay, presence of malignancy or diabetes, presence of Foley catheters or drains	10.6	1
. 중 큰 큰 큰	ain atry they they they they they they they the	Intry Types of patients studied General and digestive surgical patients rey General surgery key General surgical	Itypes of patients Type of hospital studied hospital ain General and digestive surgical nd university patients University key General surgery key General surgical key General surgical	Intry Type of studied No. hosps ain General and digestive surgical Not stated 1 ain digestive surgical Not stated 1 rev General surgery University 1 key General surgery tertiary 1 key General surgery University 1	Intry Type of studied No. hosps included in hospital ain General and digestive surgical Not stated 1 52 ain digestive surgical Not stated 1 52 key General surgery University 1 223 key General surgery University 1 223 key General surgery Liniversity 1 151	Intry Types of patients Type of hospital No. hosps Included in analysis Method of attribution ain getteral and digestive surgical Not stated 1 52 Case-control ain digestive surgical Not stated 1 52 Case-control riew General and digestive surgical University 1 23 Crude comparison riew General surgery tertiary 1 233 Crude comparison riew General surgical University 1 151 Case control	Itypes of patients Type of hospital No. hosps included in analysis Method of attribution Characteristics used in natching Analysis hospital No. hosps included in analysis Method of attribution Primary features: primary feagoesis, operative procedure, surgical cartegory (clean, clean-contaminated operients ain digestive surgical digestive surgical patients Not stated 1 5.2 Case-control Primary features: primary procedure, surgical cartegory (clean, clean-contaminated or strandinated dirty), age (+or - 5years). Secondary features; key General surgery 1 5.2 Case-control presence of moopastic or endoorine disease, elective or invasive device in situ, invasive device in situ, key General surgical University 1 2.23 Crude comparison NA key General surgical University 1 151 Case control presence of moopastic or endoorine disease, elective or endoorine disease, ele	mutry autry studied Type of bospital No. hosps analysis Method of attribution Characteristics used in matching days in hospital ain ain digestive performants No. hosps bospital No. hosps analysis Method of attribution Characteristics used in diagrossis days in hospital ain digestive patients Ceneral and digestive surgical Not stated 1 52 Case-control patients patients ceneral and contaminated, dirty, age (+or - 5 days) 11.4 key Ceneral surgical Not stated 1 52 Case-control paris- patients page (+or - 5 days) 11.4 key Ceneral surgical Not stated 1 52 Case-control parisments, discreted or presence of neoplastic or endocrine disease, elective or endocrin stut. 17.0

An overview of studies that have assessed the impact of HAIs (all sites) on hospital length of stay Table 3.1- continued:

An overview of studies that have assessed the impact of hospital acquired UTIs on hospital length of stay and costs (1975-2002) Table 3.2:

First Author and date of publication	Year of study	Country	Types of patients studied	Type of hospital	No. hospitals	No. HAIs included in the analysis	Method of attribution	Characteristics used in matching	Additional days in hospital	Addition al hospital costs
Haley (1986) ¹¹	1975-6	ASU	Al admissions	Mixed	e	177	Concurrent	NA	1.0	\$593**
Scheckler (1978) ¹⁷¹	1978	USA	Al admissions	Community hospital	-	35	Concurrent	N/A	0.6	\$146
Davies (1979) ¹³	1978	¥	Orthopaedic	DGH	4	G)	Case control	Sex, age, operation, diagnosis, Quetalet's index, smoking habits, social class	13.0	£617
Rubenstein (1982) ¹⁰	1979	stael	General surgical & orthopaedics	University affiliated	-	8	Case control	Sex, age within 5 (sometimes 7) years, pre-operative diagnosis, operative procedure, month of hospitalisation, post-op stay at least 7 days	5.1	\$510
Coello (1993) ¹⁴	1988	ž	General surgical, orthopaedic & gynaecology pts who had a surgical procedure	DGH	÷	36	Case control	Primary features: first operative procedure & primary diagnosis. Secondary features: sex, age (+or - 10 years) surgical service	3.6	£467
Hillan (1995) 🛀	Not stated	ž	Patients who had a caesarean sections	University teaching hospital	-	65	Crude comparison of infected pts with pts with no febrile morbidity or infection	NA	8. O	1
Tambyah (2002) ¹⁷⁴	1997-98	S	Adult patients scheduled to be catheterised for more than 24 hours taking part in a trial of different types of catheter	University teaching hospital	-	235	Concurrent	N A	i	\$288

*Number of HAIs was taken from an eartier paper¹⁰ where the actual number of HAIs was not clearly stated in the text ** Estimate based on the results of the study conducted in 1976 and adjusted by Haley for inflation to 1985 costs ***One patient developed a BSI secondary to the UTI had an extended LOS

Table 3.3:

An overview of studies that have assessed the impact of hospital acquired SWIs on hospital length of stay

	69	nud co	sts 1975-2002)	_						
First author and date of publication	Year of study	Country	Types of patients studied	Type of hospital	No. hosps	No. HAIs included in the analysis	Method of attribution	Characteristics used in matching	Additional days in hospital	Additional hospital costs
Haley (1986) ¹¹	1975-6	ASU	All admissions	Mixed	e	110*	Concurrent	NA	0.7	\$2734
Scheckler (1978) ¹⁷¹		NSA	All admissions	Communit y teaching hospital	-	3 8	Author assessment*	NA	6.5	\$1329
Rubenstein (1982) %	1979	Israel	General surgical & orthopaedic	Not stated	-	19	Case -control	Sex, age within 5 (sometimes 7) years, pre-operative diagnosis, operative procedure, month of hospitalisation, post-op stayat least 7 days	12.9	\$1290
Mugford (1989) ¹⁷⁵	1987	¥	Patients who had a caesarean sections	University teaching hospital	-	4	Crude comparison of infected pts with pts with no febrile morbidity or infection	NIA	1.3	£716
Coello (1993) ¹⁴	1988	ž	General surgical, orthopaedic & gynaecology pts who had a surgical procedure	НЭС	-	12	Case- control	Primary features: first operative procedure and primary diagnosis. Secondary features: sex, age(+or - 10 years) surgical service	10.2	£1,456
Poulsen (1994) ¹⁷⁶	1985-8	Denmark	Surgical	University teaching hospital	-	291	Case-control	Operation code, sex, age group	5.7	1
Hyryla (1994) ¹⁷⁷	1988-90	Finland	Surgical patients whose infection warranted compensation	Al types	2	1100	Regression modelling	NA	33.2	1
O'Donoghue (1992) ¹⁷⁸	1990	ž	Orthopaedic patients	НЭО	-		Case control - outbreak of 10 serious wound infections occurring with 1-2 weeks of surgery	Sex, age (within 5 years), pre- operative stay, orthopaedic procedure performed, general health status prior to surgery.	17.0	62,220

Table 3.3 continued: An overview of studies that have assessed the impact of hospital acquired SWIs on hospital length of stay

	- 4	and co	nsts (1975-2007	(2						
First author and date of publication	Year of study	Country	Types of patients studied	Type of hospital	No. hosps	No. HAIs included in the analysis	Method of attribution	Characteristics used in matching	Additional days in hospital	Additional hospital costs
Vegas(1993) ¹⁷³	1990	Spain	General and digestive surgical patients	I	-	R	Case-control	Primary features: primary diagnosis, operative procedure, surgical category (clean, clean- contaminated contaminated, dirty), age (+or – 5years and in one case 5 years). Secondary features; presence of neoplastic or endocrine disease, elective or endocrine disease, number of days (+or – 5 days) before infection with an invasive device in situ.	5.4 6.4	\$5,617
Hillan (1995)	not stated	ž	Pts who had a caesarean sections	University teaching hospital	-	43	Crude comparison of infected pts with no febrile morbidity or infection	NIA	3.6	1
Taylor(1995) ¹⁷⁸	1992	Canada	Surgical pts admitted to selected speciatities who underwent clean or contaminated surgenf**	University hospital	-	õ	Case-control	Risk-indexed procedure in the same surgical division	19.5	
Kirkland(1999)**	1991-95	NSA	Surgical	Community hospital	1	255	Case-control	Age, procedure, NNISS risk index, date of surgery, surgeon	6.5	\$3089
Whitehouse (2002) ¹⁸⁰	5000	Asu USA	Orthopaedic	University teaching hospital and a community hosoital	N	29	Case -control	Type of operative procedure, NNIS risk index, age within 5 years, date of surgery within the same year, surgeon.	1.0	1
•This was based	1 on physic	ian's opin	tion and data on expe	scted LOS give	in primary	and secondary	diagnosis, age, sex and service a	ind date of onset and duration of HA	7	

** General surgery, orthopaedic surgery, thoracic surgery, neuro surgery, department of obstetrics & gynaecology patients *** Initial period of hospitalisation.

>	ſ	-
of sta		ditions
gth o		tio A.
l len		Addit
spita		
o h n		
∏s o		
I LR		
uirec		
acq		
pital		
f hos		
act o		
impi		
I the		
ssec		
asse		
have		
that		
lies	02)	
i stuc	5-20	
Jo Me	(197	
rervia	costs	
An o	and	
-		
4		
ile 3.		
ab		

First Author and date of publication	Year of study	Country	Types of patients studied	Type of hospital	No. hosps	No. HAIs included in the analysis	Method of attribution	Characteristics used in matching	Additio nal days in hospital	Additional hospital costs
Scheckler (1978) ¹⁷¹	1978	ASU	All admissions	Community hospital	-	17	Author assessment *	N/A	3.8	\$878
Dixon (1978) ¹⁶¹	Not stated	USA	All admissions	Mixed	Not stated	Not stated	Not stated	Not stated	4.0	\$832
Haley (1986) ¹¹	1975-76	USA	All admissions	Mixed	3	64	Concurrent	NA	Q	\$4947
Hillan (1995) ⁵⁴	not stated	ž	Pts who had caesarean section	University teaching hospital	-	53	Crude comparison of infected patients with no febrile morbidity or infection	N/A	2.5	I
Kappstein (1992) ¹⁸²	1988-89	Germany	Ventitated patients - ICU	University hospital	-	ž	Case control	Reason for ventilation, age with 10 years; duration of ventilation therapy in controls and duration of ICU stay at least equal to the duration of ventilation therapy and ICU stay prior to the onset of infection	10.13 - ICU days	\$8,800
This was based on •• Estimate of the n	physicians of umber of ext	pinion and ra days pe	data on expected I atients with a LRTI	OS given pr remain in ho	imary and s ipital was ta	econdary diagn ken from a stud	osis, age, sex and service and ly reported in a paper publishe	l date of onset and duration of HAI d in 1978. ¹⁸¹		

	and	costs (1975-2002)						,	
First Author and date of publication	Year of study	Country	Types of patients studied	Type of hospital	No. hosps	No. HAIs included in the analysis	Method of attribution	/ Characteristics used in d matching	Additio nal days in hospital	Additional hospital costs
Wey (1988) ¹⁸³	88.	V SD	Al admissions	University teaching hospital	-	*	Case-control	Stage 1: Undertying disease, age, major surgical procedure, date of admission, sex. Stage 2: Scoring system based on characteristics used in stage one plus controls need to have he same period at risk. Stage 3: secondary diagnosis and surgical procedures. Stage four: Disease severity based on duration of underlying illness.	30.0	I
Pittet (1994) ¹¹⁵	1998-19901	VSN	Surgical intensive care	University teaching hospital	-	20	Case-control	Admission to surgical intensive care, primary diagnosis, age within 5 years, sex, LOS in controls equal to the number of days from admission to infection in cases (= or - 10%), number of discharge diagnosis.	14.0	£33,268
Pittet (1994) ¹¹³	1998-1990	VSN	Surgical intensive care	University teaching hospital	-	41 (survivors)	Case-control	Admission to surgical intensive care, primary diagnosis, age within 5 years, sex, length of stay in controls equal to the number of days from admission to infection in cases (= or- 10%), number of discharge diagnosis.	24.0	£40,890 (per survivor)
Orsi (2002) ¹²¹	1994 - 95	tały	Surgical	University teaching hospital	-	5	Case-control	Primary diagnosis, age within 5 years, sex, length of stay n controls equal to the number of days from admission to infection in cases (= or- 20%), number of discharge diagnosis, presence of central venous catheter.	19.1	16,356 Euro
Orsi (2002) ¹²¹	1994 - 95	taty	Surgical	University teaching hospital		105	Case-control	Primary diagnosis, age within 5 years, sex, number of discharge diagnosis, presence of central venous catheter.	19.9	

An overview of studies that have assessed the impact of hospital acquired BSIs on hospital length of stay Table 3.5: An overview of studies that have assessed the impact of multiple HAIs on hospital length of stay and costs Table 3.6:

	(1975- 1	2002)									
First Author and date of publication	Year of study	Country	Types of patients studied	Type of hospital	No. hospitals	No. HAIs included in the analysis	Method of attribution	Characteristics used in matching	Addītio nal days in hospit al	Additional hospital costs	
Coello (1993) ¹⁴	1988	ž	General surgery, orthopaedic & gymaecology who had undergone a surgical	НЭО	-	Gi	Case-control	Primary features: first operative procedure and primary diagnosis. Secondary features: sex, age(+or - 10 years) surgical service	56	£3,362	
Rubenstein (1982) ⁰⁸	1979	Israel	General surgery & orthopaedics	Not stated	-	æ	Case- control	Sex, age within 5 (sometimes 7) years, pre-operative diagnosis, operative procedure, month of hospitalisation, post-op stay at least 7 days	\$	\$1800	

This was based on physicians opinion and data on expected LOS given primary and secondary diagnosis, age, sex and service, date of onset and duration of HAI

The estimates presented in Tables 3.2-3.6 demonstrate that the estimated impact of HAI not only varies considerably with site of infection, but also with study. The observed variations can in part be attributed to methodological and case-mix differences. Methodological differences include variation in the range of costs included, the methods used to attribute resources to HAI, and the methods used to value resources. Other methodological differences relate to the definitions of HAI used, and the methods used to identify HAIs.

Variations in the case-mix studied include important differences in the types of patients studied and the treatment regimes received. For example, some studies include a broad case mix of patients whereas others focus on a particular patient group defined by operative procedure or specialty. Treatment patterns also vary with hospital and over time. For example, patients are discharged home at an earlier point in their recovery today than a few years ago. This changing treatment pattern will inevitably impact on the cost estimates derived and will prohibit valid direct comparisons between the results of studies conducted at different times.

Despite these important differences, some common themes emerge from the literature. Before considering these, some of the key methodological differences and issues that emerge from the literature will be discussed.

3.4.1 The range of costs resulting from HAI

The acquisition of an infection in hospital may have an impact on the costs incurred by the health care sector, community care services, affected patients, those who care for them, the economy and the environment. For example, HAIs may result in additional costs to the hospital sector, general practitioners, district nursing services, community midwifery services and a range of other health and community care services. Patients may experience an increase in personal expenditure on items such as drugs and dressings and incur financial losses due to a reduction in earning capacity as a result of delayed, or in some case a failure to return to work. Patients may also incur non-financial costs in the form of a temporary or long term reduction in health status. Informal carers may also lose work or leisure time. Society may be affected also. Production losses may result from a delayed or non-return to work. Environmental costs may ensue as a direct consequence of efforts to treat HAIs, e.g. the treatment of HAIs may result in an increase in the use of dressings, which subsequently have to be destroyed, at a cost to the environment.¹⁸⁴

The range of costs to be included in an evaluation of the burden of HAI will depend on the viewpoint to be taken. If a societal viewpoint is adopted than all costs should be included, whereas if the viewpoint of the health service is adopted then the range of costs may be limited to those that impact on the health service. Studies that have estimated the economic burden of HAI tend to limit the range of costs included to those that fall on the hospital sector as a result of additional in-patient care. Haley (1992)¹⁷⁰ suggests that this deficiency is perpetuated by the use of economic studies to persuade hospital managers of the financial importance of infection. There are however a few exceptions. Davies and Cottingham (1979)¹³ in their study of HAIs occurring in orthopaedic patients, included the impact of infection on primary health care services, and Elliston et al¹⁸⁵ in a small study involving 71 patients who had a surgical wound, examined the incidence of SWIs presenting after discharge and their impact on community health care services Kirkland et al (1999)⁶⁸ in their study of SWIs occurring in surgical patients admitted to a hospital in the US extended their cost analysis to include the costs associated with re-admission to hospital within 30 days of discharge. Hyryla et al (1994)¹⁷⁷ considered the implications of HAI for the Finnish social security system; Poulsen et al (1994)¹⁷⁶ examined the impact of HAI on the Danish social security system; Persson et al (1988)186 included a value for the loss of health suffered by patients as a result of an HAI; Fabry et al (1982)¹⁰ examined whether SWIs delayed the time of return to work; and Dashner (1989)¹⁸⁴ has examined the environmental consequence of selected treatment interventions.

3.4.2 Attributing resources to HAI

McGowan (1982) notes that attributing resources and their associated costs to HAI is difficult.¹⁸⁷ Studies have generally used one of three key methods to attribute costs to the presence of infection: the concurrent method, comparative method and comparative method with matching (see Haley *et al.*, 1980; McGowan, 1981; McGowan, 1982; and Haley, 1992 for a detailed discussion of these methods).⁴⁸ ¹⁷⁰ ¹⁸⁷ ¹⁸⁸

With the concurrent method a suitably qualified individual reviews patient records and identifies which resources were used as a result of an infection. Given the subjective nature of assessment of resources attributed to infection the validity and reliability of the approach has been questioned.¹⁸⁷ It has also been suggested that physician reviewers may be reticent about attributing the use of resources to HAI, and as such costs are likely to be underestimated.⁴⁸

To a degree some of these criticisms were overcome in a study by Wakefield *et al* (1987).¹⁸⁹ Trained personnel reviewed medical records, using a carefully prepared protocol -'the appropriateness evaluation protocol' - to assess whether each in-patient day was (a) attributable to the reason for admission, (b) jointly attributable to the reason for admission and the HAI, or (c) attributable to the HAI alone. This approach has been found to be both repeatable and valid.¹⁹⁰⁻¹⁹² However, the approach is dependent on detailed and accurate hospital records being available.

The comparative method involves assessing the cost of resources used by infected and uninfected patients, and then attributing the differences between the costs observed to the presence of an infection. This method therefore assumes that the two groups (infected and uninfected patients) are the same in all respects except for the presence or absence of an HAI. This is clearly not the case. There may be many factors, other than HAI, which differ between these two groups and which also have an impact on resources used.

The comparative method with matching attempts to control for factors other than HAI that might differ between the two patient groups. Infected patients are matched with one or more uninfected patients on the basis of factors thought to have an impact on resource use. Studies have generally matched patients using a combination of the following factors: age, sex, diagnosis, number of comorbidities and type of operation (Tables 3.1 - 3.6). The resources used by patients and controls are then compared, and the differences in costs attributed to the HAI. Haley (1991) notes that it is important that the differences between the cost incurred by infected and uninfected patients are first determined for individual patients, and then either summed (or averaged) to determine the total (or average) costs attributable to infection. Haley (1991)¹⁹³ notes that a common mistake in studies of this type is to break the matching, and simply compare the total costs incurred by infected patients, to cost incurred by uninfected patients. This approach is invalid and may lead to biased estimates of the costs attributable to HAI.

This comparative method with matching approach is hampered by practical difficulties associated with finding suitable control patients. Many studies have been unable to match all infected patients with uninfected controls. Haley et al (1980)⁴⁸ in a review of matched studies conducted between 1953 and 1975, found considerable variation in the percentage of infected patients successfully matched with uninfected patients. Successful matching of infected patients with uninfected patients ranged from a low of 32 per cent to 100 per cent. Scheckler et al (1978)¹⁷¹ in a study to assess the economic burden of HAIs occurring in 104 patients admitted to a community hospital in the US between January and March 1978, was unable to find a sufficient number of suitable controls and as such had to abandon this approach to attribution of costs. Rubenstein et al (1982)⁶⁹ in a study involving 152 infected general surgical and orthopaedic patients was only able to find suitable controls for 59% of patients. A similar proportion of infected patients were successfully matched in a study by Kappstein et al (1992)¹⁸² which assessed the excess costs and LOS resulting from ventilator associated pneumonia occurring in patients admitted to the

intensive care unit of a University Teaching Hospital in Germany. Kappstein *et al* found that after excluding cases that died during their admission, suitable controls could only be found for 34 of the 57 cases (60%). A more recent study, conducted in the UK, successfully matched 67 (85%) of the 79 infected surgical patients for whom medical records were available for review,¹⁴ whereas a study conducted in Turkey in 1994 involving general surgical patients successfully matched only 67% of their 225 infected patients.⁶¹

Freeman (1979)¹¹¹ notes that the ability to find matched uninfected patient depends on the size of the pool of uninfected patients and the number of matching characteristics. The absence of a suitable pool of uninfected patients may necessitate a reduction in the number of matching parameters, thus reducing the comparability of the infected and uninfected groups, and/or the exclusion of unmatched infected patients from the analysis, which in some cases, may limit the analysis to an un-representative subset of infected patients. Both responses may have an impact on the accuracy of the estimates of the costs attributable to HAI.¹⁹³

In 1980 Haley *et al* (1980) compared the concurrent and comparative method by using both to study the same population.⁴⁸ Haley found that the closer the matching the lower the estimate of the number of additional days attributable to infection. However, regardless of the level of matching the comparative method appeared to overestimate the number of days attributable to HAI compared to the results derived using the concurrent method. At the same time it was acknowledged that the concurrent method may underestimate the cost of HAI. The physician-epidemiologist tended to attribute extra days to HAI only if they were clearly the consequences of a HAI.

Haley (1991) suggests that one of the problems with studies utilising the comparative approach is that matching parameters such as age, sex, service, first diagnosis and first operation do not adequately control for differences between infected and uninfected patients which may have an impact on

resource use.¹⁹³ Matching should ensure that prior to the acquisition of an infection, infected and uninfected patients have the same predicted length of stay and level of resource use. Haley argues that diagnostic related groups are the best predictor of length of stay, and that this should be included in such studies, together with the number of discharge diagnoses. This latter measure increased the predictive power of the diagnostic group on length of stay and level of resources used. Together these factors have been found to explain 34% of the variance in length of stay.¹⁹³

3.4.3 Estimating the cost of resources attributable to HAI

Economists measure costs as the benefit forgone by using resources in one way rather than in another, more precisely the next best alternative use. In the context of HAI, it is probable that resources used to care for patients with an HAI would, in the absence of the infection, have had alternative uses. The benefits forgone represent the opportunity costs of HAI.

Deriving estimates of the opportunity costs of HAI presents a number of difficulties. The most common method to estimate the value of resources used in one way, as opposed to another, is by applying monetary prices. However, prices will only approximate the opportunity costs if markets are 'perfect'. For a number of reasons markets are not 'perfect.' For example there are problems associated with uncertainty, imperfect information, externalities and the number of firms given size of the market.¹⁹⁴ Furthermore, few health care systems generate prices and when they do, for example charges in the US health system, they may not reflect the cost of resources. For example, cost shifting may be present as a result of hospitals shifting their charges for underreimbursed costs to those payers with which they can recover more than their costs.¹⁷⁰ ¹⁹⁵ To overcome these difficulties some studies, particularly those conducted in the US, have applied a cost-to-charge ratio in an endeavour to convert charges to costs. This measure is generally the ratio of the sum of the total hospital expenditure per annum to the sum of patient charges per annum. The cost-to-charge ratio typically lies between 0.6 and 0.8.193 This approach

overcomes some of the bias associated with applying charges, however the estimate derived is a somewhat crude estimate of total costs attributable to HAI, and as a result of cost shifting, may not be accurate for individual patients and departments.¹⁹³ This latter criticism can be overcome to a degree by stratifying the resources and associated charges attributable to HAI by individual departments, and applying a department based cost-to-charge ratio to these charges.

Dawson (1994) argues that the costs derived from the UK health sector are usually the result of mechanistic accounting conventions designed to recover expenditure rather than reflect resource use.¹⁹⁶ For example, the allocation of overheads and capital charges are made by convention. These costs are a significant proportion of total costs.¹⁹⁷

In many studies average unit costs are applied. For example, the number of additional days a patient with an HAI remains in hospital is determined and an average cost per bed day applied. This approach may also introduce bias into the estimates derived. Average costs are a function of the total quantity produced. Thus if the cost per day in hospital is considered, the average cost per bed day is a function of the total number of bed days produced. Average cost estimates derived from hospitals operating at different levels of capacity will therefore differ. Furthermore daily costs will differ over the period of hospitalisation. For example, Hollingsworth *et al* (1993) found that patients admitted to hospital with a fractured neck of femur incurred relatively high daily costs during the first few days in hospital, after which they decreased.¹⁹⁸ This pattern of daily costs is likely to vary with type of patient and in patients who acquire an infection in hospital.

An alternative approach to those identified above is the cost accounting approach, or what Haley (1991) has described as `micro costing.'¹⁹³ This involves determining the actual cost of delivering the identified services. This

132

approach provides more accurate and valid estimates of the costs attributable to HAI but is time consuming, costly and difficult to conduct.

3.4.4 Alternative measures of the impact of HAI

An alternative measure of the impact of HAI on resource use is the number of additional days patients remain in hospital as a result of HAI. Haley argues that this is a harder measure of the cost of HAI as it is subject to less variation from year to year than charges which are subject to inflationary pressures.¹⁷⁰ ¹⁹³ Studies generally provide estimates of both the number of additional days in hospital and the costs incurred. A few studies have also presented data on the number of antibiotic days and the number of investigations.¹⁴

3.5 Estimates of the economic burden of HAI to the hospital sector

As indicated above the estimates of the economic burden of HAI presented in Tables 3.1 - 3.6 vary considerably with the site of infection and also within any site specific category. Whilst in part this can be attributed to methodological and case mix differences some common themes emerge from the literature and are discussed below.

3.5.1 Estimates of the cost of specific types of HAI

Infections of the urinary tract are generally the least expensive, whereas the more costly infections tend to be infections of the bloodstream, chest and infections at more than one site. This is clearly demonstrated in the few studies that have assessed the cost of different types of infection occurring in the same patient population. A US study which estimated the impact of HAIs at differing sites occurring in patients admitted to three hospitals in the US in 1975/6 found that on average UTIs increased the patients LOS by one day (additional cost \$594); pneumonia increased the patients LOS by six days (additional cost \$4,947); BSIs - seven days (additional cost \$3,061) and SWIs seven days (additional cost \$2,734).¹¹.

Data from a UK study of 67 surgical patients with an HAI estimated the cost per UTI to be £467, SWI £1,454 and multiple infections £3,362. On average UTIs extended the hospital stay by 4 days, SWIs 10 days and multiple infections 26 days.¹⁴

Overall the results presented in Tables 3.2-3.6 indicate that there are marked variations in the estimates of the impact of different types of infections on the hospital sector. Estimates of the number of days that may be attributable to a UTI vary from 1 to 13 days. In contrast estimates of the impact of SWIs on hospital LOS range from a low estimate of 1.3 additional days in a study limited to women who had a caesarean section¹⁷⁵ to 33.2 additional days in a study which involved patients whose infection warranted compensation from the Finnish Social Security system.¹⁷⁷ Estimates of the impact of BSIs on hospital LOS range from 7 to 14 extra days, and estimates of the impact of chest infections vary from 2.5 to 10.3 additional days. Finally estimates of the impact of HAIs at more than one site vary from 18 to 26 additional days.

3.5.2 Estimates of the cost of HAIs occurring in selected patient groups

Costs have also been found to vary considerably with patient group. This may reflect differences in the type of infection occurring in the selected patient group, which in turn have different resource implications, and/or the implications of the infection for resource use in the selected patient group. Infections occurring in intensive care patients have been found to be particularly resource intensive. For example, Pittet *et al* (1994) estimated that BSIs occurring in ITU patients cost \$40,000 per survivor.¹¹³ Amongst the surgical specialties, infections occurring in orthopaedic patients also appear to be particularly costly. Coello *et al* (1993) estimated the average cost per HAI occurring in orthopaedic patients to be £2,646. This compared to £1,365 per HAI occurring in general surgical and urology patients, and £404 per HAI occurring in gynaecology patients.¹⁴

3.5.3 The distribution of the additional hospital costs resulting from treating HAIs

Some studies have attempted to disaggregate the cost estimates derived and have examined the distribution of the in-patient costs. For example, Coello *et al* (1993) estimated the costs associated with an extended length of hospital stay, antibiotics, microbiology tests and radiological investigations: 93% of the total additional cost incurred by surgical patients with an HAI could be attributed to an extended length of stay.¹⁴ A number of other studies have also looked at the distribution of costs, but as with the study by Coello, they have generally limited the distribution of costs to those linked to time in hospital, antibiotics, microbiology tests and x-rays (e.g. Davies and Cottingham, 1979; ¹³ Wakefield *et al.*, 1987)¹⁸⁹

3.5.4 The national burden of HAIs

National estimates of the burden of HAI are also presented in the literature. In the US HAIs have been estimated to cost the hospital sector \$4 billion,¹² and in a paper by Losos *et al* (1984)¹⁹⁹ the direct and indirect costs of HAIs affecting patients in Canadian acute care hospitals were estimated to be between \$300 and \$1billion depending on the estimates of incidence and excess length of stay used.

In the UK, in 1981 a crude estimate of the additional costs of HAI was presented in an editorial in the Journal of Hospital Infection.²⁰⁰ Based on the assumption that HAIs extend LOS by at least three days; the average cost per day in hospital is £50; and 5% of patients acquire an infection in hospital, it was estimated that HAIs were costing the hospital sector in England alone at least £30 million per annum.

In 1988, a Joint DHSS/PHLS working group derived an estimate that was considerably higher. It was estimated that in England 950,000 bed days were lost per annum, at a cost to the NHS of £111 million.(DHSS/PHLS, 1988) This figure was derived using what Haley has termed the `back of the envelope

approach' or crude weighting.¹⁷⁰ It was assumed that 5% of patients acquire an infection in hospital and this, on average, extends the hospital stay by four days. These assumptions were then applied to the total number of admissions to NHS provider units in England and an estimate of the number of bed days lost, and costs incurred, based on an average cost per day, derived.

In 1993 Coello *et al*¹⁴ based on the results of a study examining the burden of HAI occurring in surgical patients admitted to one district general hospital estimated that HAIs occurring in surgical patients alone cost NHS hospitals in England, £170 million per annum.

As indicated above the overall cost to the hospital sector is a function of both the cost per case and the incidence of HAI. Data from the US indicates that although UTIs account for 45% of HAIs, they only account for 13% of the additional costs incurred, whereas pneumonia accounts for 19% of HAIs, but accounts for 39% of the additional costs and SWIs account for 29% of HAIs, but account for 42% of the additional costs. BSIs account for 2% of HAIs, but 4% of additional costs.¹¹

3.6 Assessment of the benefits of prevention

Studies that have assessed the benefits of prevention vary considerably in scope and study design. They can be broadly categorised into those that have assessed the gross benefits of prevention, that is they have not taken into account the costs of achieving a reduction in rates; those that have taken into account the costs of prevention and assessed the net benefits of prevention; and those that have looked at the cost of carrying out selected ineffective 'prevention' activities.

When evaluating an economic study Drummond (1987)¹⁶² suggests that the evaluator should consider ten key questions which aim to identify the presence or absence of ten methodological features of a well conducted economic study. However, in the context of studies that have assessed the costs and benefits of

infection control practices, Drummond²⁰¹ suggests that the following six areas are of particular relevance and should be considered in any assessment: the viewpoint selected; the alternatives selected for comparison; the range of costs and benefits considered; how costs and benefits were assessed; whether incremental analysis was conducted; and whether the evaluation included sensitivity analysis.

As in any economic evaluation a choice of viewpoints can be selected, ranging from a very broad viewpoint such as that of society, to a narrower viewpoint such as the health sector, the hospital sector or the infection control team itself. The majority of studies reported in the literature take the viewpoint of the hospital sector and neglect costs and benefits falling elsewhere.

The alternatives selected will clearly depend on the focus of the study. However, as pointed out by Drummond²⁰¹ it should be noted that one of the problems facing such economic evaluations is that infection control practices tend to be complementary – they do not work in isolation. As such it is difficult to tease out the costs and benefits of a particular infection control practice. An exception is those studies that have assessed the costs and benefits of selected prophylactic antibiotic regimes. These studies tend to be part of a randomised controlled study, designed to control for factors other than the presence or absence of the drug that might impact on infection rates.

The range of costs and benefits included to a greater extent will depend on the viewpoint adopted in any given study. As indicated above this tends to be the hospital sector and as such the range of costs and benefits tends to be limited to those experienced by the hospital sector

The methods by which costs and benefits are assessed are clearly of great importance. This includes how the costs associated with the intervention are identified and valued and how avoided costs, that is the costs of infections averted are identified and valued. The methods by which the costs are identified, attributed to HAI and valued will have implications for the reliability and validity of the results obtained.

Incremental analysis is also important. Incremental analysis provides important data on how much should be invested in infection control practice. It allows for the determination of the most economically optimal level of control, such that any additional investment in infection control will result in costs greater than the benefits that are likely to result, and any less investment will result in benefits greater than the costs of achieving a lower infection rate. Drummond²⁰¹ makes reference to a study by Persson *et al* (1988),¹⁸⁶ which assessed the costs and benefits of antibiotic prophylaxis in patients undergoing total hip replacements, and adopted this approach to the analysis of data.

Testing the sensitivity of study results to underlying assumptions is also important. Economic evaluations frequently include a number of assumptions, and the robustness of the results to changes in the parameters used in the study should be assessed. Drummond²⁰¹ makes reference to a study by Weinstein *et al* (1986)²⁰² in which assumptions about the range of clinical practices and costs were subjected to sensitivity analysis.

3.6.1 Estimates of the gross benefits of prevention

The results of the recent NAO survey of ICTs at acute NHS Trusts in England indicated that ICTs believed that a proportion of infections could be prevented. Estimates of the proportion of infections that could be prevented through improvements in infection control varied with NHS Trusts. After adjustment for the number of beds at each hospital, the results of the survey indicated that on average ICTs believed that a 15% reduction in infection rates could be achieved (bed weighted average). Applying this figure to the most recent estimate of the national burden these infections place on the health sector in England, the NAO estimated that a 15% reduction in infection rates would result in the release of resources valued at £150 million.³ This estimate represents the gross benefits of prevention. The net benefits of prevention would be dependent on the costs of achieving a reduction in rates.

3.6.2 Estimates of the net benefits of prevention

Studies that have estimated the net benefits of prevention can be broadly categorised into those that have assessed the benefits of an effective infection control programme, those that have focussed on the costs and benefits of prophylactic antibiotics; and those that have assessed the costs and benefits of selected infection control practices. The approach adopted in these studies varies, with some utilising economic modelling techniques and others, particularly those concerned with assessing the costs and benefits of alternative prophylactic antibiotic regimes, utilising a randomised controlled trial study design.

3.6.2.1 Studies that have assessed the benefits of an effective infection control programme

A number of US estimates of the potential benefits of investing in an infection control team and programme are presented in the literature. In 1975, the US Centre for Disease Control estimated the cost and benefits of an infection control programme implemented in a 250 bed hospital. These estimates were subsequently revised in 1979, and further adjusted to 1985 prices in 1986.¹¹ The cost of establishing and maintaining an infection control programme in 1985 prices was estimated to be \$60,000. This estimate includes the cost of employing an infection control nurse, a part time physician consultant, half-time clerical support, consumables and the cost of overheads. Earlier work indicated that HAIs cost the average 250 bed hospital \$1 million per year. Consequently a 6% reduction in the costs associated with infection would pay for the cost of the infection control programme and any further reductions would result in greater returns for the investment.

Dixon (1987)²⁰³ presents a similar hypothetical model of the costs and benefits of infection control in the US. The model relates to a hypothetical 250 bed hospital, with 12,000 adult and paediatric admissions per year. Based on the findings of the SENIC study Dixon estimated that in the absence of an effective infection control programme this hypothetical hospital would have an estimated

713 HAIs per year; whereas in the presence of an effective programme there would be an estimated 487 HAIs per year. Estimates of the costs of these infections were subsequently made, utilising data from an earlier study by Dixon in 1978.¹⁸¹ The results indicated that HAIs occurring in a hospital without an effective infection control programme would cost the hospital sector in terms of excess charges an estimated \$800,000 per year, whereas the cost in hospitals with an effective programme would be \$550,000: a cost saving of \$250,000 per year. Dixon subsequently estimated the cost of an effective infection control programme, to be \$60,000. When these costs are taken into account, the net savings to the hospital per year were estimated to be \$190,000. Dixon acknowledges that these estimates are crude. In reality the level of savings may vary considerably. However, he points out that the costs used in his model were conservative estimates of the economic burden of HAI and as such the actual savings could be more substantial. He also notes that the estimates used were average costs estimates. As such in a hospital that treats a greater proportion of high risk patients or provides high tech care, the costs savings may again be greater than estimated in his model. Finally he notes that the estimates of incidence and the level of HAIs that can be prevented were taken from the SENIC data derived in 1976. The SENIC study found that incidence rates were increasing in hospitals without effective programmes and falling in hospital with such programmes. As such again the benefits might be greater in some hospitals. However, in other cases the estimates may be over estimates. For example, the potential for reducing HAIs rate may be lower in some hospitals as a result of a low risk case mix.

In 1994, Wenzel presented a similar model of the costs and benefits of investment in infection control programmes as part of a lecture at the 3rd International Conference of the Hospital Infection Society.²⁰⁴ Based on a number of assumptions, about the cost of an effective infection control programme, the number of admissions per annum to a 250 bed hospital, the incidence of HAI and the impact these infections have on length of hospital stay and costs, Wenzel demonstrated that depending on whether the incidence of

140

HAI was 10% or 5%, the costs of the programme would be re-cooped if 12.5 to 25% or infections were prevented. It should be noted that the costs of infection included in this model were limited to the marginal costs estimated to amount to \$84,000. If the full costs of hospitalisation had been included in the cost estimates, estimated at \$2.52 million, the proportion of infections that would need to be prevented to cover the cost of the infection control programme would be substantially reduced. If a wider cost perspective was taken, such that costs incurred by the health service post-discharge and the cost to the patients and carers were included the proportion of infections that would need to be prevented to cover costs would be lower still.

Wenzel (1995)²⁰⁴ also presented estimates of the cost per life year gained as a result of effective prevention activities. These estimates were derived from a hypothetical model incorporating data on the cost of an infection control programme in 250-bed hospital, and incidence, mortality and attributable mortality rates. The results indicated that if it could be assumed that the quality of life of those patients who would have died from an infection if it had not been prevented was excellent, the cost per year of life saved compared favourably with the cost per life year saved as result of other preventative programmes, and the cost per quality adjusted life year of other programmes. Furthermore. Wenzel argued, that even if the quality of life year was only a small proportion of 100% or if the programme costs were considerably higher that those estimated in the model, infection control would continue to compare favourably to other programmes and as such was one of the most cost effective prevention activities. However, it should be noted that this model assumes that infections that directly cause mortality are preventable. It is not clear how preventable such infections are or how costly prevention of such infections would be. Further work is perhaps required to refine such a model and apply it to a specific situation such as a particular type of infection occurring in a defined patient group. However, despite this limitation, the models presented in the paper by Wenzel (1995) provide valuable insight into how cost and mortality

data can be used in conjunction with other data variables to demonstrate the potential benefits of investment in infection control.

An economic model developed by Miller *et al* (1989)²⁰⁵ also demonstrates the benefits of investing in an effective infection control programme. The model incorporated data on admission and infection rates at the University of Virginia Medical Centre in the US in 1985, hospital charges associated with infections and the likely level of effectiveness of an infection control programme as defined by the SENIC study. Based on this data, Miller *et al* (1989) estimated that in 1985, the infection control programme which, with the exception of reporting back surgeon specific rates, met the criteria for a very effective programme as defined by SENIC, generated income amounting to \$2,401,709.

In the UK Currie *et al* (1989),²⁰⁶ assuming the national burden of HAI amounted to £111 million (based on an earlier Department of Health estimate) estimated that a reduction in the incidence of HAI by 20%, 32% and 50%, after offsetting the costs of the infection control teams and their programmes, would produce annual savings of £15.6 million, £29.3 million and £50 million respectively to the NHS.

3.6.2.2 Studies that have assessed the costs and benefits of antimicrobial prophylaxis

A number of studies have assessed the costs and benefits of antimicrobial prophylaxis,¹⁷⁵ 186 207-212 and a review article by McGowan (1991) ²¹³ discusses issues relating to the assessment of the costs and benefits of this form of prevention. Many of these studies have found that the benefits, measured in terms of a reduction in the incidence of HAI and the associated costs, outweigh the costs of the intervention, although the level of benefits differs with alternative regimens.¹⁸⁶ ²¹¹ ²¹³ For example, Shapiro *et al* (1983)²⁰⁷ examined the costs and benefits of antimicrobial prophylaxis in abdominal and vaginal hysterectomy. The use of the prophylactic antibiotics was found to reduce costs by \$102 and \$492 per patient respectively. However, the authors note that

these savings would be lost if more expensive antibiotics had been used (unless they were more effective), and if the duration of administration was extended. Mugford *et al* (1989)¹⁷⁵ examined the costs and benefits of prophylactic antibiotics administered to patients undergoing caesarean section. Data on the effectiveness of prophylactic antibiotics were obtained from 58 controlled trials, and estimates of the costs associated with the alternative antibiotic regimens, and the costs of treating infections if they occurred were derived locally. The administration of prophylactic antibiotics was associated with a significant reduction in the incidence of SWI and a reduction in health care costs. Expenditure on prophylactic antibiotics was more than compensated for by the savings that resulted from a reduction in the incidence of HAIs and the associated treatment costs.

Persson *et al*(188)¹⁸⁶ examined the use of prophylaxis in total joint replacement surgery. Four approaches were considered, both in isolation and combination: systematic antibiotics, polymethylmethacrylate cement impregnated with gentamicin; surgical enclosure, exhaust ventilated suits. This study is interesting in that it did not simply look at the costs to the hospital sector, but also assigned a value to the effects of loss of health. The authors argue that the selection of an appropriate prophylactic regime should not be solely based on reducing costs to the hospital sector. The inclusion of a variable representing the value of loss of health, produced an economic optimum that allowed selection of a more costly regime and subsequently further reductions in infection rates and the need for re-operation.

3.6.2.3 Studies that assessed the costs and benefits of specific prevention activities

In addition to the above some studies have assessed the costs and benefits of specific infection control practices. For example, O'Donoghue *et al* (1992)¹⁷⁸ assessed the costs and benefits of procedures introduced to curtail an outbreak of ten serious surgical wound infections occurring in orthopaedic patients that occurred within 1-2 weeks of surgery. The cause of these infections was
thought to be five damaged mattresses, found to be colonised with *S.aureus*, *E.faecalis*, coliforms and *Pseudomonas* species, which were subsequently replaced. The costs of this intervention was assessed and compared to the cost of the infections, through a retrospective case control study. The ten SWIs were estimated to cost the hospital sector £22,199. The cost of replacing the mattresses was just £182.

A study by Lynch *et al* (1992) ⁶⁵ assessed the cost and effectiveness of using chlorhexidine detergent in pre-operative whole-body disinfection as a means of preventing SWIs. The results of their randomised controlled trial, conducted between April 1987 and December 1999, indicated that whilst the SWI infection rate was lower in patients who used the chlorhexidine body wash than that observed in patients who received the placebo, the difference was not significant at the 0.05 level of significance, and there was no significant difference in the costs of treating infected patients in the placebo or treatment group.

Slater *et al* (2001)²¹⁴ estimated the potential savings of employing a vascular catheter-care specialist nurse for the surgical ICU within a US hospital, as a means of tackling the problem of catheter associated BSIs. The costs of employing the nurse were compared to the potential savings that might accrue if one infection per month was prevented. The potential benefits in terms of the estimated value of resources released if one infection per month was prevented were found to be greater than the costs of employing the nurse. A specialist nurse was subsequently employed. Within nine months of employment 18 fewer BSIs than the previous year had been identified. Assuming that on average each bloodstream infections utilises hospital resources valued at \$6,000 per infection, this represented gross savings estimated at \$108,000 and estimated net savings of at least £58,000.

Plowman et al (2001)²¹⁵ developed a model for estimating the costs and benefits associated with the routine use of silver alloy coated urinary catheters

144

as a means of preventing a hospital acquired catheter related UTI. The results of their model indicated that in England a 14.6% reduction in UTIs in catheterised medical patients and a 11.4% reduction in surgical patients would cover the cost of this intervention and any further reductions would result in net benefits.

3.6.3 Estimates of cost savings resulting from not carrying out 'prevention' activities which have little or no positive effect

A number of studies have looked at the appropriateness of allocation of resources to infection control practices considered to have little or no positive preventative effect. For example, Lawrence *et al.*²¹⁶ assessed the costs of routine pre-operative urine testing and subsequent treatment of asymptomatic bacteriuria in patients admitted for elective non-prosthetic knee surgery. The results indicated that routine screening and treatment cost \$1.5 million per SWI prevented.

Daschner (1984) provides a summary account of a number of changes that were made to infection control policy and practice in a hospital in Germany and the estimated cost savings.²¹⁷ The aim was to move to more cost effective practise and away from infection control 'rituals' with little or no proven efficacy. For example, in the absence of an epidemic, routine environmental culturing and screening of staff for *staphylococci* was discontinued as was twice daily meatal care with PVP-iodine, changing of intravenous infusion sets every 24 hours and the use of in-line filters as a means of reducing urinary tract and bloodstream infections. These other changes were made in response to research findings. The estimated cost savings to the hospital sector as a result of these and other changes in practice over a six year period between 1977 and 1982 was 5,522,471 DM.

3.7 Conclusion

The economic evaluation of HAI presents a number of methodological difficulties. To date studies have tended to focus on the assessment of the

burden of HAI and relatively few have assessed the costs and benefits of investment in infection control activities. Studies that have assessed the economic burden of HAIs vary in scope and in terms of the methods used. However, despite these differences it is evident that the burden imposed is substantial. It is also evident that the economic burden varies with type of infection and admission specialty.

Evidence from a variety of sources indicates that whilst not all infections can be prevented some can, and studies that have directly assessed the costs and benefits of infection control activities, and the results of economic modelling exercises indicate that the benefits of investment in some cases may be considerable.

Many of the studies that have assessed the burden of HAI and the benefits of prevention have been conducted in countries other than England. Whilst the overall message that HAIs utilise considerable levels of health sector resource and impose a burden on the primary health sector, patients and carers is likely to be transferable to HAIs occurring in patients admitted to hospitals in England, the magnitude of resource use may vary. Similarly, whilst the overall message that investment in infection control activities may result in positive benefits, both in terms of health gains and the release of resources for alternative use is likely to be applicable to the situation in England, the magnitude of these benefits may also vary.

This thesis aims to provide more timely and relevant data on the costs associated with HAIs occurring in adult patients admitted to five surgical specialties common to most hospitals in England. These data may subsequently be used to demonstrate the burden these infections impose and, in conjunction with information on the cost and effectiveness of prevention activities, may be used to assist in demonstrating the benefits of prevention, and the results subsequently used to inform policy and practice.

THE STUDY OF THE SOCIO-ECONOMIC BURDEN OF HAI

4.1 Introduction

As indicated in chapter one of this thesis, this work has developed from a study that assessed the socio-economic burden of hospital-acquired infections (HAI) occurring in adult, non-day case patients admitted to selected medical and surgical specialties of a district general hospital (DGH) in England.² This thesis is concerned with the incidence of HAI occurring in a sub-set of patients admitted to the surgical specialties, and the costs incurred by the hospital sector during the hospitalised phase. The methods employed in the socio-economic burden of HAI study have been reported in detail elsewhere.² However, since the methods employed form part of this thesis an overview of the study is provided in this chapter. The aims and objectives of the study are presented, followed by an overview of the research methods employed. Special attention is given to methods of relevance to this thesis.

The chapter limits itself to work undertaken as part of the socio-economic burden of HAI study. Details of how a subset of data relevant to this thesis was selected, the statistical analysis undertaken for the purpose of the thesis and management issues relevant to the thesis alone are presented in the following chapter (Chapter 5). A summary of the main results of the Socio-economic Burden of HAI study can be found in the study's Executive Summary, at the back of this thesis. A copy of a peer-reviewed paper, which presents the main results of the in-patient analysis, can also be found at the back of this thesis.⁴

147

4.2 Aims and objectives of the socio-economic burden of HAI study

4.2.1 Aim

The socio-economic burden of HAI study aimed to assess the burden of HAI in terms of the costs to the public sector, patients, their families and society as a whole.

4.2.2 Objectives

The specific objectives were:

- I) To determine the overall burden of HAI in terms of:
 - a) the cost to the secondary and primary health care sectors and community care services;
 - b) the impact on the health status of patients;
 - c) the costs to patients and their families, and to the economy.
- II To establish the relative costs of different types of HAI.
- III To determine the type of patients that incur the highest costs for specific infections.
- IV To use the data obtained to construct models to predict the effects of HAI on the cost categories described above.

4.3 Overview of study design

The socio-economic burden of HAI study was designed to assess the impact of HAI on the secondary and primary health care sectors, community care services, informal carers and patients themselves. Adult, non-day case patients were recruited from selected wards of a DGH and daily profiles of the resources used by patients with and without a HAI were obtained. Those patients who

presented with a HAI during their in-patient stay and a sample of patients who did not, were followed up after discharge from hospital, using a structured questionnaire. This questionnaire aimed to alert the project team to possible infections experienced after discharge from hospital and provided information on care received from health and community care services, family and friends. It also provided information on costs incurred by patients and health status postdischarge. More detailed information about care received from health care services post-discharge was subsequently obtained from the patients' health care records. Estimates of the cost of the resources used were made. Statistical data analysis was subsequently conducted to determine the extent to which observed variations in the level of resources used, and the costs incurred, could be explained by the presence of a HAI. Ethical committee approval was obtained from the study hospital's Ethical Review Committee prior to both the pilot study and main study in September 1993 and March 1994 respectively.

4.4 Study site

Resources were available to explore the impact of HAIs occurring in patients admitted to selected specialties at one site. To enhance the applicability of the results to other health care settings in the UK, tertiary referral centres were excluded in favour of a DGH. Other selection criteria included similar service provision to other DGHs and easy access to the two institutions involved in the research: the Public Health Laboratory Service and the London School of Hygiene and Tropical Medicine.

The study hospital selected was part of an NHS Trust providing general acute, selected regional tertiary specialist and primary care services. It was the Trust's largest single provider of acute health care. At the time recruitment and data collection for the main study commenced (April 1994), the study hospital, which served a population of 260,000, had 579 beds, and an out-patient and an accident and emergency department. The in-patient caseload for 1994/95 was

39,898, of which 9,298 were elective cases, 20,358 emergency cases and 10242 day cases.

The budget for 1994/95 was £67 million. The aggregate resources employed by the Trust were organised into 32 directorates: 15 clinical directorates; 7 support directorates and 10 overhead directorates. Each directorate was responsible for its own budgets and was managed as quasi-independent firm producing a pre-defined range of intermediate goods and services. Of the total resources employed by the Trust, 48% were managed by the clinical directorates, 18% by the support directorates and 34% by the overhead directorates.

A retrospective assessment of how representative the study hospital was of others in England was undertaken in terms of the number of bed days produced, number of staff employed, the average cost per bed day, the average length of stay and the expected length of stay given the case mix of patients. Data on the number of bed days produced, number of staff employed and average cost per bed day at the study hospital and for hospitals throughout England were retrieved from the Chartered Institute of Public Finance Accountants/Healthcare Financial Managers database.²¹⁸ The study hospital values for these three variables were found to lie within the interquartile range of the distributions. The Health Services Indicators database²¹⁹ provided information on the average length of stay, and expected length of stay given the case mix of patients for some of the specialties involved in this study, at both the study site and at other hospitals throughout England. The study hospital values for these variables fell within the interguartile range of the distributions. These findings suggest that for the variables considered, the study hospital was not atypical of other hospitals throughout England.

150

4.4.1 Infection control arrangements at the study hospital

Infection control prevention and control policies and procedures operate at many different levels within any health care setting. This section describes the 'formal' infection control arrangements present at the study site.

The Infection Control Team (ICT) consisted of two Infection Control Nurses (ICNs) and a microbiologist who designated approximately three hours to infection control matters per week. The infection control team covered 579 acute sector beds and a further 60 non acute beds at two other hospitals. This is equivalent to one ICN per 289 acute beds and 30 non acute beds. This ratio of ICNs to number of beds compared favourably with the national average at the time of one ICN per 477 acute and 376 non acute beds in 1995.²²⁰

The responsibilities of the ICT were similar to those set out in the document entitled 'Hospital Infection Control: Guidance on the control of infection in hospitals' issued by the Department of Health in 1995.⁹ The ICT routinely carried out 'alert' organism and condition surveillance, and when time allowed, targeted surveillance. The ICT reported to the Chief Executive and to the Hospital Infection Control Committee(HICC).

The HICC consisted of the ICT, a Consultant in Communicable Disease Control, the Director of Nursing Practice (representing the Chief Executive), a surgeon, another consultant and an occupational health physician. The HICC responsibilities were similar to those set out in the above mentioned document. Link nurses were not present in the hospital at the time the data were collected but have since been established.

4.5 Sample Size

The primary aim of the Socio-economic Burden of HAI study design was to compare the level of resources used by infected patients (and the associated costs) with those used by uninfected, but otherwise similar patients. A large component of resource use depends on length of stay (LOS). Since relevant data on resource use were not readily available, the sample size calculations were based on a planned analysis of LOS comparing the mean values in patients with and without an HAI and using the difference between the means as an estimate of the true effect of HAI. The aim was not to test if there was a difference, but to derive adequate estimates of the magnitude of the effect. In order to do this there needed to be enough individuals in both the infected and uninfected groups so that, despite the large variation in individual LOS values, the confidence interval on the difference between the means would be acceptably narrow. The size of the smaller group primarily determines the precision of an estimate obtained as the difference between two group means. Since it was known that those infected would be only a small proportion of those in the cohort of admissions recruited into the study, it was clear that they would be the smaller group. The number recruited therefore had to be of sufficient size to ensure a large enough group of infected patients.

In order to derive sample size estimates information on the variability of LOS in both uninfected and infected patients was required. This information was obtained from a study by Coello et al (1993)¹⁴ where LOS differences had been combined to calculate the standard error of the difference between mean LOS infected and uninfected patients from three separate specialties in (gynaecology, orthopaedics and general surgery and urology combined). It was assumed that the estimate required was the difference between the mean stay given a HAI and the mean stay for similar patients without a HAI. The precision of such an estimate is measured by its standard error, usually as a percentage of the value of the estimate. As indicated above a small case control study¹⁴ provided data on the LOS differences between those with and without a HAI in three different specialties The standard errors of the LOS differences can be expressed in terms of the number of cases required, assuming at least an equal number in the non-HAI group.

It was assumed that a 10% precision would be adequate (i.e. that the standard error should be less than or equal to 10% of the estimate) and that stratified analysis would increase precision to some extent. On this basis it appeared that about 400 cases would be sufficient. Consequently if around 6% of admissions acquire one or more HAIs, 6800 admissions would be required to produce sufficient cases.

At an early stage in the socio-economic burden of HAI study it became apparent that given the resources available for data collection, it would not be feasible to recruit a sample of more than about 4700 patients. Assuming a HAI incidence rate of 6%, this sample size would yield about 282 cases of HAI. It was calculated that this reduction in the size of the smaller comparison group would reduce the precision of the analysis, defined as above, by approximately 4% so the standard error of the difference in the mean LOSs increased to 14% of the observed difference. It was concluded that this would still be sufficient to provide adequately precise results. In the event a smaller sample was recruited but a higher incidence rate was observed: 3980 patients were recruited, of which 7.8% (309 patients) acquired and presented with an HAI in hospital. At the end of the study the observed difference in mean LOS was actually found to be 13.8 days with a standard error of 1.3 or 9.4% of the difference and as such the attained precision with the reduced sample size was better than originally planned.

4.6 Subjects

The focus of the study was the socio-economic burden of HAIs occurring in adult in-patients admitted to specialties common to most hospitals. Adult patients (over 18 years of age) who had an in-patient stay of 30 hours or more, and who were admitted to the following specialties were therefore eligible for recruitment into the study: general surgery, general medicine, urology, gynaecology, orthopaedics, ear, nose and throat, elderly care and, if they had undergone a caesarean section, obstetrics. Nationally, adult, non-day case patients admitted to these specialties at other NHS hospitals in England accounted for 70% of all adult non-day case admissions in 1994/5.²²¹ Resource constraints prohibited the recruitment of all patients admitted to these specialties. The study was therefore limited to patients admitted to the selected specialties on designated wards. Details of the number of wards involved in the study can be found in Appendix 4

4.7 Recruitment

Recruitment commenced in April 1994 and continued until May 1995. The informed written consent of eligible patients was obtained by one of six research assistants, all of whom were experienced Registered Nurses. The research assistants were responsible for recruitment and data collection on specific study wards. Details of the wards each research assistant was responsible for can be found in Appendix 4.

The recruitment process involved the following. The aims and objectives of the study, together with details of the information required, the data sources to be accessed and the extent of the patient's active involvement in the study were explained. It was stressed that any information obtained would be kept strictly confidential. Patients were then given an opportunity to ask questions and an information sheet about the study was given to the patient to read in their own time. After a suitable period of time the research assistant returned and answered any further questions the patient might have. If the patient agreed to participate their written consent was obtained. Finally, the research assistants informed participating patients that if they had any further questions they could either ask them when they visited the ward each day, or contact them or the project co-ordinator by phone. The information sheet included the name and telephone number of the research assistant who recruited them into the study and the project co-ordinator's contact details. These numbers were also included on posters displayed on the ward which outlined the aims and objectives of the study, the project information book kept at the nurses station on each ward, and in the letter sent to patients followed up post-discharge.

As far as possible patients were recruited into the study on the day of admission to hospital. If on admission to hospital a patient was too ill to give consent, demographic and clinical data were collected, but not included in the study until the patient was able to give consent. Any data collected on patients who subsequently did not wish to participate in the study were destroyed.

Where a patient's condition precluded the research assistant obtaining their consent at any time during the hospital stay, the consent of a close relative was sought.

Every effort was made to ensure that patients who declined participation in the study felt comfortable with their decision. It was emphasised that their non-participation would not affect their treatment in any way. The sex, age group and admission specialty of patients who declined participation were recorded, enabling the representativeness of the sample to be checked.

Time constraints prohibited the recruitment of all eligible patients. Although recruitment itself did not take up a lot of time, the collection of detailed data on resource use throughout the study-participants' stay did. The research assistants were instructed to give priority to the collection of full data sets on all study participants rather than recruiting all eligible patients.

Selection bias was avoided through guidelines, training, supervision and on going monitoring of the recruitment process. Information on the reason for non-recruitment was recorded together with the sex, age group and admission specialty of eligible patients enabling the representativeness of the sample to be checked.

Examples of the data collection form used to record these data, the patient information sheet and patient and relative consent forms are given in Appendix 5. No attempt was made to recruit patients during periods of annual leave and baseline data were not recorded on these patients.

4.8 Data collected

The following data were collected: baseline data relating to the patient population, data on the presence or absence of HAIs presenting during the inpatient and post-discharge phase; information on health sector and community service resource use; the cost of resources used; costs incurred by the patient themselves and their family and friends and information on the impact of HAI on the health status of patients (Table 4.1).

This thesis explores the economic impact that HAIs occurring in surgical patients have on the secondary health care sector as a result of additional inpatient care. As such the following categories of data are relevant to this work: baseline data relating to the patient population, data on the presence or absence of HAIs presenting during the in-patient phase; information on health sector resources use (secondary sector only) and the cost of resources used

The following sections provide details of the data variables of relevance to this thesis, the data sources used, and an overview of the methods developed to derive appropriate cost estimates of the resources used. Further details of the methods used to derived cost estimates for resources used can be found in the detailed account provided in Part II of the report of this study.²

Categories of information	Examples of information sought
Patient characteristics:	Age, sex, reason for admission, diagnosis, co- morbidities, household size, social class and socio - economic group
	Care received from formal and informal carers prior to admission.
Hospital acquired infection data:	
	Site, date of onset and, if known, pathogen.
HAIs presenting during the hospital stay	
	Site and, if data are available, date of onset and
Infections presenting post-discharge	pathogen.
Health- sector resource use and costs:	
Secondary sector	Investigations, care and treatment received
Primary sector	Investigations, care and treatment received
Community care services	Care received
Costs to the patient and their family and friends:	
Patient	Expenses incurred by patients.
	Time of resuming normal daily activities
	Time of return to paid employment
Family and friends	Care received from family and friends.
Impact of HAIs on health status:	Effect of HAI on general health status.

Table 4.1: An overview of the data collected

4.8.1 Data on patient characteristics

Data on patient characteristics included age, sex, reason for admission, primary diagnosis and co-morbidities both on admission to hospital and discharge from hospital, route of admission (e.g. elective via pre-admission clinic, elective direct to the ward or emergency via the accident and emergency department), the specialty of the admitting consultant, dates of admission and discharge, and discharge destination. This information was obtained from the medical and nursing notes.

Information on the patient's current or most recent occupation, employment status prior to admission to hospital (e.g. unemployed, self employed, employed), and household size and composition was obtained from either the patient, a relative or the medical/nursing records. If obtained from the medical or nursing notes, as far as possible it was verified with the patient and/or relative or friend. Based on the information provided (current or most recent occupation and employment status prior to admission to hospital) patients were categorised into social class and socio-economic groups according to the OPCS classification system.²²²

Information on the care received prior to admission from the formal sector and family and friends was also obtained. Formal sector care included district nursing services, meals on wheels and home help services. Care from family and friends included assistance with daily activities of living such as washing, dressing and cooking and regular visits made by family and friends to check that the patient was managing. This information was obtained from the patients, their relatives or friends, or the medical or nursing notes. If obtained from the medical or nursing notes, as far as possible it was verified with the patient and/or relative or friend.

4.8.2 Data on HAIs presenting during the hospital phase

HAIs presenting during the in-patient phase were identified using the reference method of surveillance described in the Public Health Laboratory Service (PHLS) report `A Study of Surveillance Methods for Detecting Hospital Infections.'⁷⁷ This method aims to identify all patients with a HAI. It involves liaison with ward personnel and consulting all relevant data sources, such as laboratory reports and nursing and medical records, to identify signs and symptoms associated with HAIs. If these met the definitions of infections used in this study a HAI was recorded, together with the site of infection, date of onset and, if known, the pathogen(s) involved. The definitions used in this study were those developed as part of the aforementioned PHLS report.⁷⁷ Minor changes were made to the text to aid interpretation and use (Appendix 6).

4.8.3 Data on health sector resource use during the hospital phase

Data on the resources used during each patient's stay in hospital were collected. This included all investigations carried out (for example X-rays, laboratory tests, endoscopies and cardiac tests); procedures performed (for example operations, insertion of intra-vascular catheters, insertion of urinary catheters); care administered (for example care administered by nurses, physiotherapists, occupational therapists) and drugs and intravenous infusions administered. This information was obtained from a variety of data sources including medical and nursing notes, the notes of other relevant health care professionals, laboratory print outs and drug prescription charts. Information was also obtained about care organised for dependants whilst the patient was in hospital, for example respite care. This information was obtained from the patient or a relative or friend.

4.9 Methods used to derive cost estimates of the resources used

A detailed approach was taken to the estimation of the cost of resources used. This section provides an overview of the methods used to derive estimates of the cost of hospital resources used during the patient's in-patient stay. A detailed account is reported elsewhere, together with an account of the methods used to derive estimates of the costs of other resources used.²

4.9.1 Methods used to derive cost estimates of hospital resources used during the in-patient phase

As indicated in section 4.3 information on the resources used by patients during their admission was collected for each patient regardless of whether they had an infection. These resources may be broadly classified into two groups those associated with occupying a hospital bed (i.e. the cost of maintaining a hospital bed) and those relating to specific care and treatment administered to individual patients.

Estimates of the costs of occupying a hospital bed were derived as follows. Data on the costs of the overhead directorates were obtained from the Trust. Overhead directorates included the following: finance, estates, hotel services, personnel, planning, Trust management, technical services, education and training and the reserves directorate.

An allocation model was subsequently developed whereby specialty specific estimates of the average daily cost of occupying a bed were derived. The model took into account the proportion of individual overhead directorate costs used by each specialty and the proportion of individual overhead directorates costs used by the other overhead directorates. A detailed account of the allocation model and the assumptions that informed it can be found elsewhere.² The resulting specialty specific estimate of the cost per bed day was subsequently multiplied by the patient's length of hospital stay to derive individual patient estimates of the cost of occupying a bed.

Costs associated with care and treatment administered included the cost of medical time, nursing time, the time of other health care professionals such as physiotherapists, and the cost of diagnostic investigations, procedures carried out and consumables used.

Cost of medical care: The amount of medical care patients receive varies with the patient. In order to gain an accurate assessment of the cost of medical care received, it would be necessary to record and cost all contacts that patients had with members of the medical profession over the course of their hospital admission, and determine the time medical staff spent planning and organising care for individual patients. Unfortunately, resource constraints prohibited such an approach. The average daily cost of medical care was derived for each specialty and allocated to patients in the study in accordance with the length of hospital stay.

Cost of nursing care: Nursing costs were allocated to patients based on the amount of nursing care patients received each day. The method used to allocate nursing care costs was based on a system developed and extensively validated in Australia. It was subsequently introduced to the Hammersmith Hospital, London in 1988 and internally validated. The system involves allocating patients to one of seven care groups determined by the amount of care received within a nursing shift. The seven care groups represent a spectrum of care extending from patients that require minimal care (care group one) to those who require extensive nursing care (care group seven). An appropriate cost is then calculated for each care group based on the amount of available nursing time patients falling into each care group are thought to consume. The cost of nursing care for each patient can then be calculated by summing the nursing care costs for each nursing shift over the whole hospital stay.

Cost of contacts with professionals allied to medicine: Average costs of consultations with chiropodists, speech therapists, dieticians, occupational therapists and physiotherapists were derived from cost and activity data supplied by the study hospital and allocated in accordance with the number of contacts supplied to individual patients

Costs of investigations: The cost of radiological investigations were derived by applying Korner weights to cost and activity data supplied by the study hospital. The average unit costs of microbiology, chemistry, haematology and histopathology tests were derived from cost and activity data supplied by the trust.

Cost of surgery: The costs associated with surgery were derived by first calculating a cost per session for each specialty from which a cost per minute of theatre time for each specialty could be derived. The derived costs included the cost of running the theatre. This included the cost of nursing staff, technicians, management and administrative staff; the cost of consumables associated with

161

the provision of surgical interventions, including pharmacy inputs, medical gases, drapes, surgical equipment and dressings; and finally indirect and capital inputs. The cost per session did not include the cost of a surgeon's time which was included in the costs of medical time allocated separately to each patient on a daily basis

The second stage involved calculating the cost per procedure. This was achieved by applying the specialty specific cost estimates derived above (cost per minute of theatre time for each specialty) to data on the average time associated with procedures classified according to level of complexity (i.e. minor; intermediate and major and a number of intermediate levels). This information was supplied by the study hospital.

The third and final stage involved classifying the operative procedures undergone by individual patients by level of complexity using the classification system described above which is used by BUPA (1993),²²³ and subsequently applying the appropriate cost estimate derived as described above. Further details about the methods used, including how sub-procedures were costed, are reported elsewhere.²

Cost of pharmaceuticals: Estimates of the cost of pharmaceuticals supplied were derived by combining the price of the drugs (determined as a result of negotiation between the pharmacy at the study hospital and the pharmaceutical industry), with the costs of any consumables associated with the delivery of the drug. For example, the cost of an intra-muscular antibiotic would include the cost of the drug, a syringe, the fluid with which the drug is mixed, and two needles (one used to mix the drug and one to administer the drug).

The costs of consumables supplied to patients: The cost of consumables such as intravenous catheters, wound drains, and urinary catheters were derived from the NHS supplies catalogue (1994).²²⁴ The combination of resources used

to supply a product was based on information provided by an F-grade nurses familiar with clinical practice at the study hospital.

Once these unit costs were available they were multiplied by the data collected on resources used by the patients. This allowed the development of cost profiles for all study participants. These profiles provided information on the total cost of resources and the contribution of specific components of cost to the total costs incurred.

4.10 Data Management

Data were recorded on data collection forms designed using optical scanning software designed by Formic Ltd (see Appendix 7 for examples). The data were then scanned onto the scanning software database and exported in either Dbase or SPSS format for analysis at the Central Public Health Laboratory and London School of Hygiene and Tropical Medicine. This method of data entry has been shown to be 99.9% accurate in test-re-test studies.²²⁵ Prior to analysing the data a number of data checks were made to eliminate any substantial recording errors. The data cleaning process included checking that all surveys had been scanned completely and a number of range, categorical and logical checks. For example, checks were made that the date of discharge was after the admission date; that only females were recorded as having been admitted to the gynaecology and obstetric wards and specialties, that the ages recorded fell within an acceptable range (18-100), and the ICD9 diagnosis codes were valid and appropriate given the patient's sex and admission specialty.

At all times patient information was handled in ways that maintained patient confidentiality. Patients participating in the study were identified by a unique study number. This number was the only form of patient identification stored on the database. Personal details, such as the patient's name, address and hospital number, were stored on a separate computer whose use was restricted to named individuals. Those responsible for collecting and handling data

163

received appropriate training and were asked to sign a document outlining the standards for data handing, and record keeping. A copy of this document is at Appendix 2

4.11 Data analysis

4.11.1 Checking for recruitment/selection bias

The age, sex, admission type and admission specialty distributions of patients recruited into the study were compared to the distributions that would have been present if all eligible patients were recruited.

4.11.2 Incidence of HAIs presenting during the in-patient period

The number of patients with one or more hospital-acquired infections presenting during the in-patient period was expressed as a percentage of the number of patients discharged and 95% confidence intervals derived. Site and specialty specific estimates were also derived.

4.11.3 Incidence of HAIs presenting post-discharge

Estimates of the proportion of patients reporting symptoms and treatment which met the study criteria for infections presenting post-discharge were derived for four groups of patients: patients who did not have an HAI identified either during the in-patient or post-discharge phase; patients who had no in-patient HAI, but had evidence of a possible infection post-discharge; patients who had an HAI identified during the in-patient phase but no evidence of an infection postdischarge and finally patients who had an HAI identified during the in-patient phase and had evidence of a possible infection post-discharge.

4.11.4 Attribution of costs to HAIs

4.11.4.1 In-patient analysis

Attributing resource use and costs to the presence of an infection presents a number of difficulties. Factors other than infection may have an impact on resource use. In this study linear regression modelling techniques were used to control for a range of factors that could potentially influence the level of

resource use: age, sex, diagnosis, number of co-morbidities, admission specialty, and admission type. Since the cost and length of hospital stay distributions included a few very high values they were skewed to the right. For this reason the analysis was performed assuming that the underlying distribution was Gamma in form (this distribution, very similar to the log-Normal, is appropriate for skewed data).²²⁶ Estimates of the impact of one or more hospital-acquired infections, and of specific types of infection, adjusted for the effects of confounding variables were derived from this modelling process.

During the analysis the effects of social class, nursing dependency (defined as the average level of nursing care required from admission to the time the HAI was identified) and disease stage were also investigated. This last variable is a measure of severity of illness, derived from information on the patients' age, sex, diagnosis, co-morbidities, operation codes, admission type, length of inpatient stay and discharge destination. Previously validated algorithms²²⁷ were used to allocate patients to one of three disease stage groups (low, medium and high) depending on the severity of their illness using software developed by CHKS Ltd. The inclusion of these additional variables was found to have very little effect on the estimated impact of HAI on hospital sector costs once the other explanatory variables had been taken into account.

4.11.4.2 Post-discharge analysis

The primary outcome measures for the post-discharge analysis included the number and cost of general practitioner, district nurse, hospital doctor, health visitor and community midwife visits, and the costs to patients and their informal carers. As with the in-patient analysis, the variables had skewed distributions. Ideally the distribution used in the in-patient analysis (Gamma distribution) would have been used in the post-discharge analysis. However, the data had many zeros for these outcomes. Consequently the Gamma distribution was considered inappropriate and a log normal distribution was used for the regression modelling.

4.11.5 Impact of HAI on health status

This aspect of the analysis was limited to the impact of HAI on health status as measured by the responses given to the general health status questionnaire, the Short-form 36 (SF-36), administered within the post-discharge questionnaire. Mean scores for the eight dimensions of health covered by the SF-36, were derived using the standard SF-36 scoring algorithms²²⁸ and mental and physical summary scores subsequently derived, again using the standard scoring systems.²²⁹ These were compared to the norms derived in the Oxford Healthy Lifestyle Study²³⁰ using the two sample unpaired t-test. Mean scores for each dimension of health and for the two summary measures were then determined for patients within each of the four HAI groups and the impact of HAI on these measures determined. Regression analysis as described above was used to control for factors other than the presence of an infection that might influence the scores obtained.

4.11.6 Deriving national estimates of the number of patients who acquired one or more HAIs

National estimates of the number of adult patients who acquired one or more infections in hospital, which presented during the in-patient stay were derived by applying the observed in-patient incidence rate and 95% confidence interval to data on the number of adult patients (\geq 18), excluding day cases, admitted to similar specialties, in NHS provider units throughout England in 1994/5. The same approach was used to derive specialty specific estimates of the number of patients acquiring one or more infections and estimates of the number of patients acquiring specific types of infection.

4.11.7 Deriving national estimates of the cost of HAIs to the hospital sector

Estimates of the burden these infections imposed on the hospital sector were derived from data on the observed incidence of hospital-acquired infections presenting during the in-patient period; the estimated ratio of the hospital costs incurred by infected compared to uninfected patients obtained from the linear modelling analysis; the mean hospital costs incurred by uninfected patients; and data on the number of adult admissions. If N is the number of patients admitted nationally, C the baseline cost of treating uninfected patients, *i* the estimated incidence and *r* the estimated ratio of costs incurred by infected compared to uninfected patients, then N*i* C(r-1) provides an estimate of the national burden.

The variance of this estimate was derived from the standard deviations of the estimates of the incidence and ratio of costs, sd*i* and sd*r* respectively, as follow: $N^2 C^2 [i^2 s dr^2 + (r-1)^2 s dr^2]$

This estimated variance was subsequently used to obtain 95% confidence intervals for the estimates of the national burden of infection, it being assumed that the sampling error in such an estimate would be approximately normal. Estimates of the number of additional days patients remained in hospital and site and specialty specific estimates of the burden imposed were made using the same approach.

4.12 Project Management

The socio-economic burden of HAI study was a joint venture by the Central Public Health Laboratory and the London School of Hygiene and Tropical Medicine. The project team were responsible for the organisation and conduct of the study. Over the course of the study it comprised of the following members: a project co-ordinator, research economist, six research assistants, a statistician, a project secretary and a part-time administrative assistant.

The project team was advised by a multidisciplinary steering group consisting of experts in infection control, microbiology, epidemiology and economics. The project steering group met as and when was required. The project steering group and team were advised by a multi-disciplinary Advisory Committee comprising DH representatives, and experts in infection control, microbiology, economics and community nursing. Appendix 8 provides details of the members of the project team, steering group and advisory committee.

167

4.13 **Project Timetable**

The socio-economic burden of HAI study was commissioned by the DH to the Central Public Health Laboratory and London School of Hygiene and Tropical Medicine in October 1992 and a project co-ordinator (Rosalind Plowman) appointed in July 1993. The study comprised of four key phases listed below.

Phase 1: This was conducted over a period of nine months from July 1993 to March 1994 and involved a literature review, the development and piloting of the study tools and research methods, and their subsequent modification in the light of the findings of the pilot study.

Phase 2: This was conducted over a period of 16 months from April 1994 to July 1995 and constituted the main data collection period.

Phase 3: This was conducted over a period of 13 months from August 1995 to September 1996 and involved the validation, analysis and interpretation of the data set for the purposes of the costs study and final Department of Health Report.

Phase 4: This involved the preparation of the final cost study report and its submission to the Department of Health for internal and external review and subsequent modification. This was conducted over prolonged period of time. The report was submitted to the Department of Health in August 1997 and distributed for internal and external review. A full set of reviewer's comments was received in November 1997. A formal response was submitted to the Department of Health in February 1998 following which there was a period of discussion and debate. Further editing work and additional analysis was undertaken. Amendments to the original text together with the results the additional analysis requested were submitted to the Department of Health in September 1998 and the report accepted for publication in September 1999, and finally released in January 2000

4.14 Conclusion

This chapter has provided an account of the methods used in the socioeconomic burden of HAI study. The methods used in the socio-economic burden of HAI study, in particular those that relate to the impact of HAI on hospital resource use and costs, are common to this thesis. The following chapter builds on the information presented in this chapter. Details about how data on patients of interest to this thesis were selected from the socio-economic burden of HAI data set are presented, together with an account of the analysis undertaken.

5.1 Introduction

The preceding chapter has described in some detail the study to which this thesis is linked. As stated in the preceding chapter the methods employed in the socio-economic burden of HAI study are common to this thesis. In this chapter the aims and objectives of this thesis will be presented and discussed, followed by a detailed account of the methods which are particular to this work. The chapter describes how a subset of data was selected from the data set collected as part of the socio-economic burden of HAI study, and the statistical analysis undertaken. Details regarding data and project management are also presented.

5.2 The aims and objectives of this thesis

5.2.1 Aim

To assess the incidence of, and independent risk factors for HAIs occurring in surgical patients admitted to a district general hospital and the impact these infections have on the secondary health care sector, and to examine how the information obtained may be used to assess the potential benefits of investment in the prevention and control of HAIs.

5.2.2 Objectives

The specific objectives were to:

- 1. Review the literature on the epidemiology of HAI, risk factors for HAI, and the economic evaluation of HAI.
- 2. Determine the incidence of HAIs occurring in adult, non-day case patients admitted to selected surgical specialties of a district general hospital.

- 3. Explore how the incidence of HAI varies with various patient characteristics and identify possible risk factors.
- 4. Determine the impact HAIs occurring in this patient group had on secondary health care sector resource use and costs.
- 5. Examine how information on the economic burden of HAIs may be used to assess the potential benefits of investment in the prevention and control of HAIs.

5.3 Overview of study design

As detailed in the preceding chapter, the study to which this thesis is linked was designed to assess the impact that HAIs occurring in medical and surgical adult patients had on the secondary and primary health care sector, community care services, informal carers and patients themselves. This thesis is concerned with the incidence of HAIs occurring in patients admitted to the surgical specialties and presenting during the in-patient period, risk factors for these infections and the impact these infections had on secondary health care sector resource use and costs as a result of additional in-patient care. Thus data relating to the incidence of HAI occurring in surgical patients, risk factors for these infections and the resources used during the in-patient period, collected as part of the socio-economic burden of HAI study, are of relevance to this thesis. An overview of how these data were collected, and the analysis undertaken for the purposes of this work follows.

Adult, non- day-case patients were recruited from selected wards of a district general hospital. Background demographic data were collected on all patients together with information on selected risk factors. Information on infections presenting during the hospitalised phase was obtained through surveillance using validated surveillance methods and case definitions. Data on resources used by all patients were collected resulting in daily profiles of the resources used by patients with and without a HAI. Estimates of the cost of the resources

used were made. A sub-set of data relating to patients admitted to the surgical specialties was subsequently identified and statistical data analysis conducted to determine the incidence of HAI, independent risk factors for infection, and the extent to which observed variations in the level of resources used, and the costs incurred, could be explained by the presence of a HAI. How information on the economic burden of HAIs may be used to assess the potential benefits of investment in the prevention and control of HAIs was subsequently explored.

Details regarding the study setting, how patients were selected and recruited into the study, the data variables of interest and data collection methods employed can be found in section 4.4-4.8 of Chapter 4. The following section describes how the subset of data of relevance to this thesis was selected.

5.4 Selection of surgical patients and relevant data for the purposes of the PhD thesis

This thesis is concerned with the incidence of HAIs occurring in adult patients admitted to selected surgical specialties, risk factors for these infections, and the impact these infections have on hospital resource use and costs as a result of additional in-patient care. The wider socio economic burden of HAI study included data on patients admitted to five surgical specialties: general surgery, orthopaedics, urology, gynaecology, and if they had undergone a caesarean section the obstetric specialty. This section provides details of how patients admitted to these specialties were identified and how the subset of data relevant to this work was compiled.

Patients admitted to the five specialties of interest could be identified from the underlying data set on the basis of admitting ward, specialty or consultant. It was decided to select patients on the basis that their admitting consultant was not an ENT surgeon, physician or geriatrician, but a general surgeon, gynaecologist, urologist, obstetrician or orthopaedic consultant.

Prior to selecting patients on the basis of their consultant code the validity of this approach was checked. The final data set was renamed and the admission consultant, ward, specialty and diagnosis cross-checked to assess the face validity of this information: i.e. given the consultant's speciality would you expect the documented ward and admission specialty. Unusual records were highlighted and looked at in more detail. That is reports including primary diagnosis, co-morbidities, ward transfer and consultant transfers were generated and examined. Judgements were subsequently made about the data and the data file modified as necessary. For example, if the patient's consultant was recorded as being a gynaecologist, but the ward, specialty and diagnosis indicated the patient was a urology patient, the assumption was made that this was a urology patient and the consultant code changed to indicate that this assumption had been made. Or if the consultant, specialty and diagnosis indicated that the patient was a gynaecology patient but the ward was surgical, it was assumed that there were no beds available on the gynaecology wards and no changes were made. Once this process was complete patients were selected on the basis of admitting consultant code and reports generated and checked again. Having completed this second check the data set for this work was selected, and saved as a separate data file.

Within this limited data set there were eight patients who were transfers from another ward or hospital. These patients by definition will have incomplete inpatient data sets: their data sets will relate to varying proportions of their hospital stay. Consequently these eight patients were excluded from the final data set used.

Having identified the patients, data relating to their hospital stay were obtained from the relevant survey databases, and datasets specific to the needs of this thesis were created. For example, datasets limited to surgical patients were created for information on operations undertaken, procedures performed, investigations carried out and nursing care administered. Further data checking was also carried out, including range, categorical and logical checks.

173

Whilst carrying out this procedure it was noted that the number of surgical patients for whom there were data on operative procedures appeared rather low. The data relating to the first operation indicated that only 1936/2469 (78.4%) patients included in the surgical data set, had evidence of a first operation having been performed. In 533 cases (21.6%) there was no evidence of a first operation having been performed. This rather surprising observation was felt to warrant further investigation. Did the data accurately reflect the procedures performed? If the data set was a valid account of procedures performed in the wider population sample? Had important information not been entered and as such was the data set incomplete?

In order to investigate these important questions the hospitals Patients Administration System (PAS) data set for the year 1994/5 was analysed to determine what proportion of the wider hospital population, admitted to the specialties included in this study, had an operative procedure. The analysis revealed that during the year 1994/5 67% of the patients admitted to the surgical, gynaecology, orthopaedic and urology specialities had an operative procedure. This was in fact lower than the percentage of patients in the study sample who had undergone an operative procedure. At face value this was an encouraging finding. Furthermore, it was noted that not all the operation codes included in the PAS data related to actual operations. Some referred to procedures such as blood transfusions and endoscopies. Despite this it was considered necessary to explore the situation further.

As far as possible the PAS records of study patients who did not have any evidence within the study database of a surgical operation having taken place were checked. Of the 533 patient records checked, data were available for 524 patients. In 148 (28.2%) cases there was evidence that an operative procedure had taken place (i.e. an OPCS code included in the PAS database). However, in 43 cases the operation codes were for procedures, which in this study were not classified as an operation – e.g. blood transfusion, removal of catheter and

endoscopy. As such according to the PAS database it would appear that 105 (20%) of the 524 patients for whom no operation data were available in the study database and data was available in the PAS database, in fact had one or more operations. As such, the operation codes for these 105 patients were entered into a newly created data variable within the reduced operations dataset. The project economist subsequently derived cost estimates for all 105 cases and the information was entered onto the database within a newly created variable denoting revised operation. It was also accepted that the remaining 419 patients did not have an operation. It was also accepted that the 15 patients for whom there was no evidence of surgery having taken place in the study database and no evidence of admission in the PAS database, did not have surgery. As such an estimated 2041 (82.7%) of the 2469 patients included in this study had one or more operative procedures.

This process of checking the validity of the operation data revealed a second related problem. It was noted that of the 1936 patients for whom there were data within the study database on the surgical procedure performed, cost estimates were only available for 1830 patients; 106 of the 1936 patients for whom operation data were available did not have a cost derived for the operation performed. This situation was addressed as follows. In 57 of the 106 cases for whom there was evidence within the study database of an operation having been performed but no cost estimates, costs were derived and entered onto the database. Estimates were derived using the methods developed for the underlying study, reported in detail elsewhere² and briefly summarised in section 4.9.1. In the remaining 49 cases there was evidence of surgery having taken place, but procedure codes had not been entered onto the database. As far as possible these codes were obtained from the PAS database and cost estimates derived. In 37 cases procedure codes were obtained from PAS and subsequently used to derive cost estimates. However, in two cases there was no evidence of an operation in the PAS database and in the remaining ten cases there was no evidence of admission to hospital in the PAS database.

Once new operation costs had been derived and entered the final stage was to compute a new total cost estimate substituting the newly derived operation cost data for that originally used.

Having created a number of datasets limited to surgical patients, a number of new variables summarising specific aspects of resource use were subsequently computed from the raw data. For example, a variable was created to denote whether patients had received antibiotics prior to the onset of infection. In order to create this variable, it was first necessary to create an antibiotic data set from the many data collection forms that included antibiotic drugs. Antibiotic drug use had been recorded on six different types of data collection forms. These forms had been designed to collect data on oral drugs, intravenous and intramuscular drugs, intravenous infusions, operations performed, and drugs that had not been listed elsewhere. The latter included two different data collection forms one with space for recording data on up to six drugs not listed elsewhere and the other space for recording up to 12 drugs that had not been specifically listed elsewhere.

With the exception of the last three forms, the data collection forms related to two-week periods. That is for each type of data collection form there was a form relating to weeks 1-2, another for weeks 3-4 and so on. In order to create one antibiotic database, data collection forms relating to the different two-week periods were merged. These merged datasets, relating to different types of data collection forms, were then brought together into one antibiotic dataset.

Once an antibiotic database had been created it was then possible to create a new variable denoting whether the patient had received antibiotics prior to the onset of an infection, and whether the antibiotics received were administered via the oral, intravenous or intramuscular route, or by all three modes of administration. Using a similar approach three new variables were created to denote whether patients had a urinary catheter, one or more wound drains or a naso-gastric or endotracheal tube in situ prior to the onset of infection.

5.5 Data analysis undertaken for PhD thesis

The data were analysed using SPSS 10 and STATA version 7. The data analysis included a descriptive analysis of the data set, a more detailed exploration of the incidence of HAI and possible risk factors for these infections, and estimation of the costs attributable to HAI.

5.5.1 Descriptive analysis

A descriptive analysis was undertaken. This included an exploration of the age, sex, specialty, admission type, primary diagnosis, and number of co morbidities. Where appropriate cross tabulations were developed, and tests for significance were conducted using the Chi-Square test. In order to check the representativeness of the recuited sample the age, sex, admission type and admission specialty distributions of study participants and eligible patients who either declined or were not recruited due to practical reasons, such as insufficient time, were analysed and compared. Appendix 9 provides details of how the variables listed were categorised for the purpose of the analysis.

5.5.2 Incidence of HAI and identification of risk factors

The incidence of HAI was assessed in terms of the following two definitions:

- the number of patients with one or more HAIs expressed as percentage of the number of patients discharged
- ii) the number of primary HAIs/1000 patient days at risk

For each definition a single variable analysis was conducted to assess how the incidence varied with selected patient characteristics (age, sex, admission specialty, admission type, diagnosis group, number of co-morbidities, presence of absence of diabetes mellitus, and whether the patient had an operation).

Based on definition (i) above, a multivariable logistic regression analysis was conducted to assess which factors were independently associated with risk of acquiring an infection. The significance of the effects of each variable in the analysis was assessed using likelihood ratio χ^2 tests.

The above analysis was repeated for each type of infection: urinary tract, surgical wound, chest, bloodstream, skin and infections at sites not classified elsewhere. The variables included in the risk factor analysis varied slightly with site of infection. For example, the analysis examining the incidence of UTIs included a variable denoting presence or absence of a catheter prior to the onset of infection. Further details of the variables included are presented in the relevant result sections in Chapter 7.

5.5.3 Economic impact

The analysis considered the impact of HAI on two measures of resource use: hospital costs and length of hospital stay.

5.5.3.1 Single variable analysis

Single variable analysis was conducted to assess how hospital costs and length of hospital stay (LOS) varied with infected and uninfected patients and other patient characteristics including age, sex, admission specialty, admission type, primary discharge diagnosis group, and number of co-morbidities. The mean and median hospital costs incurred and length of hospital stay were calculated together with the 25th and 75th percentiles, minimum and maximum values and the standard deviation. As the data were skewed the significance of the observed differences was assessed using the non-parametric Mann-Whitney and Kruskal-Wallis tests.

5.5.3.2 Multivariable analysis - overview

The single variable analysis indicated that both LOS and hospital costs not only varied with HAI status but also with a number of factors including age, sex, admission type, and number of co-morbidities (see Chapters 8 and 9). In order to assess the independent effect of HAI on hospitals costs and LOS multivariable regression analysis was undertaken. The aim was to assess the impact of HAI on hospital costs and LOS after controlling for other factors that might influence the magnitude of the costs incurred and a patient's LOS. With the exception of primary discharge diagnosis group the regression analysis to assess the impact of HAI on LOS and costs included all the factors included in the single variable analysis (equations 1 and 2). Primary discharge diagnosis group was excluded as the number of patients in many of the subgroups was very small.

Equation 1:

Hospital costs regressed on sex, age, admission specialty, admission type, number of co-morbidities, HAI status.

Equation 2:

Length of stay regressed on sex, age, admission specialty, admission type, number of co-morbidities, HAI status.

Further details of the explanatory variables included in this analysis can be found in Appendix 9.
5.5.3.3 Normal linear regression – no transformation of dependent variable Multiple regression analysis is based on a number of assumptions: independence, linearity, normality, and constant variance:

- 1. The observations should be independent.
- 2. The relationship between the dependent and independent variables should be linear.
- 3. For each combination of values of the independent variables, the distribution of the dependent should be normal.
- 4. For each combination of values of the independent variables the distribution of the dependants should have a constant variance.

These conditions were not met in this data set. Both dependent variables (LOS and hospital costs) had a skewed distribution and the variance was not constant. Figure 5.1 plots the quantiles of LOS against the quantiles of the normal distribution and Figure 5.2. the quantiles of hospital costs against the quantiles of the normal distribution. If the data were normally distributed the plotted points would lie approximately on a straight line. It can be seen from figures 5.1 and 5.2 that this is not the case with respect to the LOS and cost data, indicating that both the LOS and cost data are not normally distributed.





Figure 5.2: A plot of the quantiles of hospital costs against the quantiles of the normal distribution



A quantitative assessment of the deviation from Normality was made using the Shapiro Francia W' test. The results presented in Table 5.1 indicate that both the LOS and cost data were not compatible with a Normal distribution (p= 0.00001).

Table 5.1 : 3	Shapiro-	Francia	W' to	est for	normal	data
---------------	----------	---------	-------	---------	--------	------

		Shapiro- Francia W' test for normal data						
Variable	Obs	W'	V	z	Prob>z			
Length of stay	2469	0.56361	499.748	7.979	0.00001			
Hospital costs	2469	0.50115	571.277	8.047	0.00001			

5.5.3.4 Linear regression – log transformation of dependent variable

A frequently used method to deal with skewed data is to transform the data onto a log scale. Transforming the LOS and hospital costs data on to the log scale (log LOS base 10 and log Cost base 10) produced a distribution that is closer to the normal distribution (figure 5.3 and 5.4).

Figure 5.3: A plot of the quantiles of LOS transformed onto the log scale (base10) against the quantiles of the normal distribution



Figure 5.4: A plot of the quantiles of hospital costs transformed onto the log scale (base10) against the quantiles of the normal distribution



However, it is clear from Figures 5.3 and 5.4 that even after transformation onto the Log (base 10) scale there remains positive skewness, with this being more marked in the case of hospital costs than hospital length of stay. Transformation onto the Log 10 scale does not pull the tail of the distribution in

quite enough. The higher observed values are still higher than expected given the mean and standard deviation. A shifted transformation may have produced a distribution closer to the normal distribution, but its use would make interpretation more difficult. Furthermore, even in the absence of a shifted transformation, interpretation problems arise. The estimates and confidence intervals derived from regression models in which the dependent variable is transformed onto the log scale are measured on the log scale. To obtain a clear interpretation the estimates would need to be transformed by exponentiation (anti- logarithms) back to the original scale. This would produce estimates of the ratio of hospital costs and LOS for each level of a variable category group compared to the categories base line. Thus for the HAI variable it would produce a ratio of hospital costs incurred by infected patients compared to uninfected patients, and a ratio of hospital LOS in infected compared to uninfected patients. Difficulties arise when estimating the magnitude of these increases in terms of additional costs incurred and the number of extra days patients remain in hospital. Whilst the ratio of costs or LOS could be applied to the arithmetic mean costs or LOS of uninfected patients, it is questionable whether this is appropriate given that the ratios were derived on the log scale.

5.5.3.5 Generalised linear modelling assuming a gamma distribution

An alternative approach to dealing with the problem of LOS and hospital costs having a skewed distribution is to assume that the skewed outcome variables have a Gamma distribution, which is very similar in form to the Log/Normal distribution, and construct generalised linear models which allow estimation of the effect of HAI on LOS and costs, allowing for the effects of the other factors included in the model, utilising a maximum likelihood approach. This approach was used in this analysis.

The Gamma distribution has a decreasing coefficient of variation (standard deviation/mean) as the mean increases. To assess whether it was appropriate to assume an underlying Gamma distribution the mean LOS and mean total costs were plotted against the standard deviation for LOS and total costs

(Figures 5.5 and 5.6) and the coefficients of variation (sd/mean) plotted against the categories involved for both LOS and costs (Figures 5.8 and 5.9). Figures 5.5 and 5.6 show that the standard deviation for LOS and total costs increases with increasing mean LOS and total costs and Figures 5.7 and 5.8 show that the coefficients of variation for both LOS and total costs decrease slightly with increasing mean. As such, for the purposes of the analysis it was considered appropriate to assume an underlying Gamma distribution.

Figure 5.5: A plot of the mean length of stay against the standard deviation for length of stay



Figure 5.6 A plot of the mean hospital costs against the standard deviation for hospital costs



Figure 5.7: A plot of the coefficient of variation of LOS against the categories involved



Figure 5.8: A plot of the coefficient of variation of hospital costs against the categories involved



The models developed used a) an identity link and b) a log link. Models incorporating an identity link assume additive effects and consequently enable estimates of the mean additional costs and LOS incurred by infected patients to be taken directly from the model. That is the fitted coefficients associated with HAI directly estimate the additional costs incurred by infected compared to

uninfected patients, or the number of extra days infected patients remain in hospital compared to uninfected patients, after controlling for the effects of all the other factors included in the model. In contrast models incorporating a log link assume multiplicative or proportional effects and as such provide estimates of the ratio of costs and LOS incurred by infected and uninfected patients.

Both simple models which assessed the main effects of the independent variables identified in equation one and two on costs and LOS and more detailed models which assessed two way interactions between HAI and the independent variables were fitted and compared. The statistical significance of each variable in the generalised linear model was determined using the likelihood ratio χ^2 test.

Models were also developed to assess the impact of specific types of infection on costs and LOS. The above approach was repeated substituting the HAI explanatory variable for a variable denoting infections at the following sites: urinary tract, surgical wound, lower respiratory tract, skin, blood, other single site infections not classified elsewhere and multiple sites. The analysis assessed how hospital costs and length of hospital stay varied in patients with these different types of infection compared to uninfected patients, having controlled for factors that may have had impact on length of hospital stay and costs.

5.5.4 Distribution of costs incurred

The distribution of the costs associated with infected and uninfected patients was assessed. Costs were classified into one of 16 categories: hospital overheads; directorate management; capital charges; medical time; nursing care; paramedics and specialist nurses; physiotherapy; surgical interventions; consumables used for specific procedures; antimicrobials; non-antimicrobial drugs; microbiology tests; other pathology tests; endoscopies; radiology; and other tests.

The mean costs incurred by infected and uninfected patients were determined for each category, and the additional costs incurred by infected patients calculated, together with the percentage contribution of each category to the overall additional costs incurred. Generalised linear modelling techniques using the maximum likelihood approach described above were subsequently conducted to assess the impact of HAI on the different cost categories after controlling for the effects of sex, age, specialty, admission type, number of comorbidities and HAI status.

5.6 Data management

Issues relating to data management have been discussed in the preceding chapter (see section 4.11). The data handling standards applied in the socio-economic burden of HAI study were maintained throughout this work.

5.7 Study timetable

As outlined in the preceding chapter the socio-economic burden of HAI study comprised four phases, all of relevance to this thesis.

Phase 1: This involved a literature review, the development and piloting of the study tools and research methods, and their subsequent modification in the light of the findings of the pilot study. This was conducted over a period of nine months from July 1993 to March 1994.

Phase 2: The main data collection period. This was conducted over a period of 16 months from April 1994 to July 1995.

Phase 3: The validation, analysis and interpretation of the data set for the purposes of the socio-economic burden of HAI study and final Department of Health Report. This was conducted over a period of 13 months from August 1995 to September 1996.

Phase 4: The preparation of the final socio-economic burden of HAI study report and its submission to the Department of Health for internal and external review and subsequent modification. This was conducted over prolonged period of time, culminating with the publication and subsequent release of the final report in January 2000.

These four phases of the project were followed by four further phases which were exclusively concerned with the surgical subset.

Phase 5: Surgical patients were identified and a separate PhD data set which included data variables of relevance to the planned analysis was created. Further data checks were made and the data cleaned as necessary. New data variables relating to the presence or absence of selected risk factors prior to the onset of infection were computed.

Phase 6: The subset of data was analysed. This included a descriptive analysis of the data set, and exploration of the incidence of and key risk factors for infection; and an assessment of the impact of HAI on hospital resources use during the in-patient hospital stay and associated costs.

Phase 7: This involved an exploration of how the costs estimates derived may be incorporated into models to assess the potential benefits of investment in the prevention and control of HAIs.

Phase 8. In parallel with the work described in phases 5-7, a more detailed and in depth review of the literature was conducted, and the thesis drafted.

5.8 Conclusion

This chapter has provided an account of the methods used in this study. The following four chapters present the results of this work. Issues relating to the validity and limitations of the approaches used are discussed in Chapter 11.

CHAPTER 6 SAMPLE CHARACTERISTICS

6.1 Introduction

This chapter presents the results of the descriptive analysis which assessed the key characteristics of the sample recruited and how representative they were of all eligible patients. Data relating to the sample size, and time of recruitment are presented first. This is followed by the results of the analysis that explored how a range of patient characteristics were distributed including: age, sex, specialty, admission type, discharge diagnostic group, number of co-morbidities, operations performed, discharge destination and care planned on discharge from hospital.

6.2 Sample size

Between April 1994 and May 1995 3534 adult, non-day case patients admitted to the surgical specialties covered in this study were eligible for recruitment. Of these 2477 (70.1%) patients agreed to participate in this study, 197 (5.6%) declined participation, 851 (24.1%) were eligible for recruitment but were not invited to participate in this study due to time and resource constraints, and in a further nine (0.3%) cases the reason for non-recruitment was not recorded. Of the 2477 surgical patients recruited into the study eight were excluded from the analysis as they were transfers from another ward or hospital and as such differed from the rest of the sample in important respects: 2469 patients were therefore included in the analysis (Figure 6.1).



The number of patients eligible for recruitment and the number recruited Figure 6.1:

190

The results of the analysis which compared the age, sex, admission type and admission specialty distributions of patients recruited into the study with eligible patients who were not recruited are reported in Appendix 10. The results indicated that the proportion of patients who were aged 60 years and over was similar for patients in the two cohorts. However there were statistically significant differences with respect to the sex, admission type and admission specialty distributions in the two cohorts of patients. There were fewer male patients, emergency admissions, and general surgical and urology patients in the recruited cohort, and more female patients, elective admissions, and orthopaedic, elderly care, ear nose and throat, obstetric and gynaecology patients. However, the results of the analysis which compared the age, sex, admission type and admission specialty distribution of patients recruited into the study, with that which would have been present if all eligible patients had been recruited, found that the differences between the cohort of recruited patients and the intended cohort of all eligible patients was small. Consequently, since the HAI and non-HAI comparisons controlled for these factors, it is reasonable to assume that the results obtained are generalisable to all eligible patients, despite the significant differences between those recruited and those not recruited.

6.3 Day recruited

Information on the day recruited was available for 2462 (99.7%) patients. The majority of patients were recruited during the early part of their admission: 94% were recruited within the first six days of admission with 31.2% recruited on day one; 13.7% on day two and 27.5% on day three. Figure 6.2 provides further details of when patients were recruited into the study. In those cases where patients were recruited quite late, this was either due to the patients condition precluding them being invited to participate until this time, or due to the research assistant being unable to find a suitable time to recruit the patient until quite late into their admission.



Figure 6.2: The cumulative percentage of patients recruited by day of admission (n= 2462)*

*Note: Information on day recruited was not available for seven patients. These patients were excluded from this analysis.

6.4 Age and sex distribution

Of the 2469 patients recruited into the study 1466 (59.4%) were female and 1003 (40.6%) were male. The minimum age was 18 years and the maximum 94 years, with a mean age of 54.5 years (SD \pm 18.4). Table 6.1 provides further details of the age and sex distribution.

Tunie V. T. Age and Sex distribution	Та	ble	6.1:	Age	and	sex	distribution
--------------------------------------	----	-----	------	-----	-----	-----	--------------

	Ma	Male		le	All patients	
Age group	n	%	n	%	n	%
18-34	126	24.51	388	75.49	514	20.82
35 – 54	217	32.24	456	67.76	673	27.26
55-74	514	54.05	437	45.95	951	38.52
75+	146	44.11	185	55.89	331	13.41
All patients	1003	40.62	1466	59.38	2469	100.00

6.5 Specialty distribution

Figure 6.3 shows the distribution of patients by admission specialty. Overall a greater proportion of patients (36%) were recruited from the general surgical specialty than from the other specialties covered.





Table 6.2 shows the specialty and sex distribution. It can be seen that the sample included a greater number of female than male patients. This is in part due to the inclusion of two exclusively female specialties (obstetrics and gynaecology). Within the surgical and orthopaedic specialties also more female than male patients were recruited. The only specialty with more male than female patients was urology.

Specialty	n	Male patients		Female patients		
		n	%	n	%	
General surgery	884	388	43.9	496	56.1	
Orthopaedics	501	219	43.7	282	56.3	
Urology	472	396	83.9	76	16.1	
Gynaecology	386	0	0.0	386	100.0	
Obstetrics	226	0	0.0	226	100.0	
All patients	469	1003	40.6	1466	59.4	

Table 6.2: Sex and specialty distribution

Table 6.3 shows the age and specialty distribution. Approximately half of the sample (48.1%) were aged 18-54 and the remainder (51.9%) aged over 55 years. As to be expected the gynaecology and obstetric specialties had a younger age profile than the other specialties.

		18-	34	35 -	- 54	55	5-74	7	5+
Specialty	n	n	%	n	%	n	%	n	%
General surgery	884	141	16.0	280	31.7	359	40.6	104	11.8
Orthopaedics	501	62	12.4	100	20.0	225	44.9	114	22.8
Urology	472	33	7.0	74	15.7	274	58.1	91	19.3
Gynaecology	386	88	22.8	184	47.7	92	23.8	22	5.7
Obstetrics	226	190	84.1	35	15.5	1	0.4	0	0.0
All patients	2469	514	20.8	673	27.3	951	38.5	331	13.4

Table	6.3:	Age	and	specialty	distribution
-------	------	-----	-----	-----------	--------------

6.6 Type of admissions

Table 6.4 shows the number and percentage of patients by admission type. These seven routes of admission were re-classified into two groups, elective and emergency admissions, as indicated in Table 6.4. Of the 2469 patients recruited into the study, 1629 (66%) were elective admissions and 840 (34%) were either urgent or emergency admissions. Table 6.5 shows how the number and percentage of patients admitted via these two different routes varied with specialty. It can be seen that, with the exception of the obstetric specialty, all admission specialties had more elective than emergency admissions.

Table 0.4: Admission type distribution	Table	6.4:	Admission	type	distribution
--	-------	------	-----------	------	--------------

Admission route	Number of patients	%	Grouped admission category
Elective via pre-admission clinic	365	14.8	Elective
Elective - direct to the ward	1264	51.2	Elective
Urgent direct to the ward	100	3.9	Emergency
Emergency via A+E	668	27.1	Emergency
Emergency via the OPD	55	2.2	Emergency
Emergency via GP	7	0.3	Emergency
Emergency via community midwife	10	0.4	Emergency
All patients	2469	100	

A+E = Accident and emergency department

OPD = Out-patients Department

Specialty	n		Admiss	sion type	
		Elec	tive	Emer	gency
		n	%	n	%
General surgery	884	588	66.5	296	33.5
Orthopaedics	501	281	56.1	220	43.9
Urology	472	347	73.5	125	26.5
Gynaecology	386	309	80.1	77	19.9
Obstetrics	226	104	46.0	122	54.0
All patients	2469	1629	66.0	840	34.0

Table 6.5: Admission type distribution by admission specialty

6.7 Re-admissions to hospital

6.7.1 Number of patients classified as 're-admissions'

Patients were classified as a 're-admission' if they had a hospital admission within the last month. Of the 2469 patients recruited into the study 263 patients (10.7%) were classified as re-admissions. Of the remaining 2206 patients, in 10 cases information on re-admission status was not recorded. These patients were assumed not to be re-admissions, the assumption being that if they were, this information would have been recorded. Of the 263 patients classified as re-admissions, in 228 cases (86.7%) the admission was linked to the earlier admission, in 17 cases (6.5%) the admissions were not linked and in 18 cases (6.8%) information on whether this was a linked admission was not recorded.

6.7.2 Characteristics of patients classified as 're-admissions'

Table 6.6 provides an overview of the patient characteristics of patients classified as re-admissions (i.e they were in-patients less than a month prior to the admission in which they were recruited). Of the 263 re-admissions 45 (17.1%) had a HAI, with eight presenting with a HAI on the day of admission, 31 presenting with an HAI at some point during their admission and in the remaining six cases the date of onset of infection was not recorded. Table 6.7 shows the number and percentage of re-admissions presenting with an HAI by admission specialty. Of the 263 patients re-admitted to hospital within a month of discharge three patients died in hospital. All three patients were male surgical patients, two aged 73 years and one 54 years. One patient had a HAI – a BSI acquired during their previous admission.

Patient characteristic	n	%
Sex		
Male	91	34.6
Female	172	65.4
Age group		
18-34	96	36.5
35 - 54	62	23.6
55-74	83	31.6
75+	22	8.4
Specialty		· · · · · · · · · · · · · · · · · · ·
General surgery	82	31.2
Orthopaedics	28	10.6
Urology	49	18.6
Gynaecology	34	12.9
Obstetrics	70	26.6
Discharge diagnosis group		
Infectious & parasitic diseases	1	0.4
Neoplasms	41	15.6
Endocrine, nutritional & metabolic diseases & immunity disorders	•	-
Diseases of blood & blood forming organs	-	•
Diseases of the nervous system & sense organs	-	-
Diseases of the circulatory system	18	6.8
Diseases of the respiratory system	•	-
Diseases of the digestive system	35	13.3
Diseases of the genitourinary system	42	16
Complications of pregnancy, childbirth & puerperium & certain	77	29.3
conditions originating in the perinatal period		
Diseases of the skin & subcutaneous tissue	3	1.1
Diseases of the musculoskletal system & connective tissue	14	5.3
Injury & Poisoning	17	6.5
Symptoms, signs & ill-defined conditions; mental disorders &	15	5.7
congenital abnormalities		
Co-morbidities		
None	147	55.9
One	82	31.2
Two	23	8.7
Three or more	11	4.2
HAI		· · · · · · · · · · · · · · · · · · ·
No	218	82.9
Yes	45	17.1
In-patient death		
No	256	97.3
Yes	7	2.7

Table 6.6: Characteristics of patients admitted within month of a previous admission

•

			H.	AI	
Specialty		No		Y	'es
	n	n	%	n	%
General surgery	82	72	27.4	10	3.8
Orthopaedics	28	21	8.0	7	2.7
Urology	49	45	17.1	4	1.5
Gynaecology	34	23	8.7	11	4.2
Obstetrics	70	57	21.7	13	4.9
Total	263	218	82.9	45	17.1

Table 6.7: The distribution of patients admitted within a month of a previousadmission to hospital by presence or absence of a HAI and admissionspecialty

Of the 263 patients who were classified as re-admissions, 70 (26.6%) had already participated in this study. Of these 70 patients seven (2.7%) had a HAI during their previous admission. Three had a SWI, two had UTIs; one had a LRTI; and one a skin infection.

6.8 Distribution of patients by primary discharge diagnosis

Table 6.8 shows the number and percentage of patients by primary discharge diagnosis grouped into 14 disease categories. Seventy-seven per cent of the primary discharge diagnoses fell within just five of the 14 ICD9 categories: diseases of the genitourinary system (23%); diseases of the digestive system (16.9%); neoplasms (14.5%); complications of pregnancy, childbirth and puerperium (10.3%), and diseases of the musculoskeletal system and connective tissue (12.5%).

197

Table 6.8:The number and percentage of patients by primary discharge diagnosis grouped into 14 diseases categories

Study disease categories 1-14	ICD9 disease categories included in study categories	Title of Categories	Number of patients within each of the 14 disease categories	Percentage of patients within each of the 14 disease category
1		Infectious and parasitic disease	13	0.5
2	11	Neoplasms	357	14.5
3	10	Endocrine, nutritional and metabolic diseases and Immunity disorders	24	1.0
4	IV	Diseases of blood and blood forming organs	3	0.1
5	VI	Diseases of the nervous system and sense organs	2	0.1
6	VII	Diseases of the circulatory system	150	6.1
7	VIII	Diseases of the respiratory system	6	0.2
8	IX	Diseases of the digestive system	418	16.9
9	X	Diseases of the genitourinary system	569	23.0
10	XI	Complications of pregnancy, childbirth and puerperium Certain conditions originating in the perinatal period	254	10.3
- 11	XII	Diseases of the skin and subcutaneous tissue	31	1.3
12	XIII	Diseases of the musculoskeletal system and connective tissue	309	12.5
13	XVII	Injury and Poisoning	187	7.6
14	XVI V XIV	Symptoms, signs and ill- defined conditions Mental disorders Congenital abnormalities	146	5.9
All groups			2469	100.0

6.9 Number of co-morbidities

•

The majority (66.5%) of patients had no co-morbidities; 23.4% had one; 7.0% two and 3.1% three or more co-existing conditions. Table 6.9 shows the number of co-morbidities by specialty.

		Admission specialty											
Specialty	General surgery		Orthopedic		Urology		Gynaecology		Obstetrics		All specialties		
·····	n	%	n	%	n	%	n	%	n	%	n	%	
None	492	55.7	365	72.2	361	76.5	322	83.4	102	45.1	1642	66.5	
One	263	29.8	84	16.8	77	16.3	54	14.0	100	44.2	578	23.4	
Two	79	8.9	41	8.2	27	5.7	8	2.1	18	8.0	173	7.0	
Three or more	50	5.7	11	2.2	7	1.5	2	0.5	6	2.7	76	3.1	

Table 6.9: The number and percentage of patients by number of co-morbidities and admission specialty

6.10 Characteristics of patients who had one or more operative procedures

6.10.1 Number of patients who had one or more operations

Of the 2469 patients included in this study 2041 (82.7%) had one or more operations. Of these 2041 patients, 1999 (97.9%) had one operation, 40 (2.0%) had two operations, and two (0.1%) had three operations.

6.10.2 Day of surgery

Data on day of operation were available for 1910 patients. Ninety-five per cent of patients for whom data were available had surgery within 5 days of admission to hospital: 11.8% had surgery on the day of admission; 77.1% on day two; 4.0% on day three; 2.1% on day 4 and 1.4% on day five.

6.10.3 Characteristics of patients who underwent one or more operations

Table 6.10 presents the results of the analysis that examined the percentage of patients who underwent one or more operations by age, sex, admission specialty, diagnosis, and number of co-morbidities.

	l .	Patients who	had one or		
Patient characteristic	n	more op	erations	P value	
		No.	%		
Sex				· · · · · · · · · · · · · · · · · · ·	
Male	1003	815	81.3	0.126	
Female	1466	1226	83.6		
Age group					
18-34	514	416	80.9	0.298	
35 - 54	673	570	84.7		
55-74	951	787	82.8		
75+	331	268	81.0		
Specialty					
General surgery	884	682	77.1	<0.001	
Orthopaedics	501	433	86.4		
Urology	472	373	79.0		
Gynaecology	386	327	84.7		
Obstetrics	226	226	100.0		
Admission type					
Elective	1629	1559	95.7	<0.001	
Emergency	840	482	57.4		
Discharge diagnosis group					
Infectious & parasitic diseases	13	3	23.1	<0.001	
Neoplasms	357	322	90.2		
Endocrine, nutritional & metabolic diseases & immunity disorders	24	24	100.0		
Diseases of blood & blood forming organs	3	1	33.3		
Diseases of the nervous system & sense organs	5	0	0.0		
Diseases of the circulatory system	150	131	87.3		
Diseases of the respiratory system	6	5	83.3		
Diseases of the digestive system	418	327	78.2		
Diseases of the genitourinary system	569	493	86.6		
Complications of pregnancy, childbirth & puerperium & certain	254	241	94.9		
conditions originating in the perinatal period					
Diseases of the skin & subcutaneous tissue	31	25	80.6		
Diseases of the musculoskletal system & connective tissue	309	287	92.9		
Injury & Poisoning	187	142	75.9		
Symptoms, signs & ill-defined conditions; mental disorders &	146	40	27.4		
congenital abnormalities					
Co-morbidities					
None	1642	1396	85.0	<0.001	
One	578	452	78.2		
Two or more	249	193	77.5		

Table 6.10:The number and percentage of patients who had one or more

operations

The proportion of patients who had one or more operations varied little with sex or age category: 81.3% of males and 83.6% of females had one or more operations and for all age categories at least 80% had one or more operations. There was some variation with admission specialty. The observed variation was significant (p=<0.001). As expected, 100% of patients admitted to the obstetric specialty had an operation. The study was limited to patients who had a caesarean section. In the other specialties the proportion of patients who had an operation varied from 77.1% (general surgery) to 86.4% (orthopaedics). A significantly higher proportion of elective admissions (95.7%) than emergency admissions (57.4%) had one or more operations varied with primary discharge diagnosis and varied slightly with number of co-morbidities.

American Society of Anaesthesiologists scores (ASA) scores were reported for 1556 patients. Of these patients 699 (25.6%) had a score of one; 667 (42.9%) had a score of two; 174 (11.2%) had a score of three, nine (0.6%) had a score of four and seven (0.4%)had a score of five.

6.10.4 Types of surgery performed

The analysis of types of procedure performed was based on the Office of Population Census and Survey classification of operative procedures system (fourth addition)²³¹ and limited to the frequency of the first procedure reported. It was assumed that the first reported procedure represented the primary operative procedure undertaken. The research assistants were instructed to report the primary procedure first, and in those cases where data on procedure group had been derived direct from the Patient Administration System database it was usual practice to report the primary procedure first. Of the 2025 initial operations for which data on procedure were available, 1414 (69.8%) involved one procedure, 388 (19.2%) involved two procedures, 163 (8.0%) involved three, 45 (2.2%) involved four, nine (0.2%) involved five, and six (0.3%) involved six procedures.

Table 6.11 provides details of the number and overall percentage of primary procedures, classified within the broad operation categories, performed. The results reported in Table 6.11 indicated that almost 80% of operations involved primary procedures classified within six broad operation categories: operations involving bones and joints (18.5%); urinary system (17.6%), upper female genital tract (12.0%), lower digestive tract (11.1%), female genital tract associated with pregnancy and childbirth (11.1%) and soft tissues (8.8%). Further details of the specific types of procedures performed are presented in Appendix 11.

Table	6.11:The	number	and	perc	centage	of	pat	tients	unde	rgoli	ng primary
	pro	cedures	class	ified	accord	ling	to	the	Office	of	Population
	Cei	nsuses ar	nd Sum	veys	listed ca	ateg	orie	S			

		No. & % of pat	tients whose
	OPCS operation category	primary proced	ure fell within
		each cat	egory
		No.	%
A	Nervous system	10	0.5
В	Endocrine system and breast	104	5.1
C	Eye	0	0.0
D	Ear	0	0.0
E	Respiratory tract	1	0.0
F	Mouth	7	0.3
G	Upper digestive tract	23	1.1
н	Lower digestive tract	225	11.1
J	Other abdominal organs - principally digestive	80	4.0
ĸ	Heart	0	0.0
L	Arteries and Veins	56	2.8
М	Urinary	356	17.6
N	Male genital organs	26	1.3
Ρ	Lower female genital	60	3.0
Q	Upper female genital	242	12.0
R	Female genital tract associated with pregnancy and childbirth	224	11.1
S	Skin	23	1.1
T	Soft tissue	179	8.8
V	Bones and joints of skull and spine	15	0.7
W	Other bones and joints	375	18.5
X	Miscellaneous operations	10	0.5
Y	Subsidiary classification of methods of operation	8	0.4
Z	Subsidiary classification of sites of operations	1	0.0

Note: analysis was based on 2025 patients for whom data on primary procedure was available.

6.11 In-patient deaths

Twenty-one patients (0.9%) died during the in-patient period. Eleven patients were female, ten patients male. Overall 14 (66.7%) were surgical patients and the remaining seven orthopaedic patients. Eighty-eight per cent of deaths occurred in patients who had one or more operations. The mean age was 74 years with a minimum age of 32 and maximum age of 93 years. Four (19%) had no co-morbidities; six (28.6%) had one co-morbidity; two (9.5%) had two co-morbidities and nine (42.9%) had three or more co-morbidities. Eight (38.1%) patients had an HAI identified during their hospital stay. Of these one had a UTI, two a LRTI, one a BSI, one an infection at a site not classified elsewhere, one a BSI and an infection at a site not specified elsewhere and one a LRTI and an infection at another site. Overall patients who acquired an infection while in hospital were 4.3 (95% CI: 1.6, 11.2) times more likely to die than uninfected patients.

6.12 Discharge destination and follow up care

The majority (97.6%) of patients were discharged home. Table 6.12 provides details of the discharge destination.

Discharge destination	n	%
Home	2410	97.6
Other ward>1week	7	0.3
Other ward <1 week	2	0.1
Other hospital> 1 week	13	0.5
Other hospital<1 week	2	0.1
Hospice	2	0.1
Convalescent home	1	0.0
Nursing home	2	0.1
Relatives	9	0.4
Died	21	0.9
Total	2469	100

Table 6.12:	Distribution of	patients by	v dischard	e destination
-------------	-----------------	-------------	------------	---------------

6.13 Services organized on discharge from hospital

6.13.1 Transport.

Of the 2469 patients recruited into the study 42 (1.7%) required an ambulance to take them home and 32 (1.3%) required a hospital car. The remaining 2395 (97%) patients did not require transport. Table 6.13 shows the age distribution of patients who utilised this service and table 6.14 the specialty distribution. The use of hospital transport (ambulance or car) was found to increase with increasing age and be highest for patients admitted to the orthopaedic specialty.

 Table 6.13: The number and percentage of patients requiring NHS

 transport home by age group

[Tran	sport	
		Amb	ulance	Hos	pital car
Age group		n	%	n	%
18-34	(n= 514)	0	0.0	1	0.2
35 - 54	(n= 673)	2	0.3	0	0.0
55-74	(n= 951)	17	1.8	18	1.9
75+	(n= 331)	23	7.0	13	3.9
All patients	(n= 2469)	42	1.7	32	1.3

 Table 6.14: The number and percentage of patients requiring NHS transport

 home by admission specialty

Spec	ialty	Transport								
	F	Ambu	lance	Hos	pital car					
		n	%	n	%					
General surgery	(n = 884)	9	1.0	6	0.7					
Orthopaedics	(n = 501)	26	5.2	23	4.6					
Urology	(n = 472)	7	1.8	3	0.8					
Gynaecolgy	(n = 386)	0	0.0	0	0.0					
Obstetrics	(n = 226)	0	0.0	0	0.0					
All specialties	(n = 2469)	42	1.7	32	1.3					

6.13.2 Follow up care

In a number of cases follow-up care was organized for patients prior to discharge. Tables 6.15 and 6.16 shows the number and percentage of patients for whom follow up care was arranged by type of care and admission specialty.

	General	surgery	Orthop	aedics	Urok	gy	Gyna	scology	Obst	etrics	All pa	itients
	(n=	884)	(n= (501)	(n=4)	72)	(n=	(n=386)		226)	(n=2	:469)
	n	%	n	%	n	%	n	%	n	%	n	%
District nurse	97	11.0	55	11.0	96	20. 3	5	1.3	0	0.0	253	10.2
General practitioner	25	2.8	11	2.2	11	2.3	5	1.3	0	0.0	52	2.1
Practice nurse	1	0.1	0	0.0	0	0.0	0	0.0	0	0.0	1	0.0
Elderly support team	2	0.2	12	2.4	1	0.2	0	0.0	0	0.0	15	0.6
СТТ	0	0.0	1	0.2	0	0.0	0	0.0	0	0.0	1	0.0
Continence advisor	2	0.2	2	0.4	6	1.3	0	0.0	0	0.0	10	0.4
Macmillan nurse	6	0.7	0	0.0	4	0.8	0	0.0	0	0.0	10	0.4
Stoma nurse	1	0.1	1	0.2	1	0.2	0	0.0	0	0.0	3	0.1
Diabetic nurse	1	0.1	0	0.0	0	0.0	0	0.0	0	0.0	1	0.0
Hospital at home team	0	0.0	17	3.4	0	0.0	0	0.0	0	0.0	17	0.7
Plastic surgery dressing clinic	0	0.0	1	0.2	0	0.0	0	0.0	0	0.0	1	0.0
Physiotherapy	0	0.0	12	2.4	0	0.0	0	0.0	0	0.0	12	0.5
Occupational therapy	0	0.0	3	0.6	0	0.0	1	0.3	0	0.0	4	0.2
Social worker	10	1.1	13	2.6	7	1.5	2	0.5	0	0.0	32	1.3
Home help	5	0.6	3	0.6	6	1.3	1	0.3	0	0.0	15	0.6
Meals on wheels	5	0.6	1	0.2	10	2.1	0	0.0	0	0.0	16	0.6

Table 6.15: The number and percentage of patients for whom community based follow-up care was organized by type of care and specialty

CTT - community therapy team

	General surgery (n= 884)		Orthopedics (n= 501)		Urology (n=472)		Gynecology (n=386)		Obstetrics (n = 226)		All patients (n=2469)	
	n	%	n	%	n	%	n	%	n	%	n	%
Out-patient appointment	588	66	439	87. 6	347	73.5	274	71.0	1	0.4	1649	66.8
ECG	2	0.2	2	0.4	2	0.4	1	0.3	0	0.0	6	0.2
Endoscopy	1	0.1	0	0.0	1	0.2	0	0.0	0	0.0	2	0.1
TWOC	2	0.2	0	0.0	49	10.4	13	3.4	0	0.0	64	2.6
Lithotripsy	0	0.0	0	0.0	1	0.2	0	0.0	0	0.0	1	0.1
Day hospital	8	0.9	3	0.6	9	1.9	0	0.0	0	0.0	9	0.4
Other	9	1.0	3	0.6	13	2.8	0	0.0	1	0.0	30	1.2

 Table 6.16: The number and percentage of patients for whom hospital based follow-up care was organized by type of care and specialty

ECG - electro-cardiogram; TWOCC - Trial without catheter

6.14 Conclusion

In this chapter the results of the analysis that explored the general characteristic of the data set have been presented. The distribution of patients by age, sex, specialty, admission type, admission specialty, primary discharge diagnosis group and number of co-morbidities have been presented, as has the mortality rate observed and the follow up care organised for these patients. The following chapter will examine the incidence of HAIs occurring in this sample of patients.

CHAPTER 7 THE INCIDENCE OF AND RISKS FACTORS FOR HAI: RESULTS

7.1 Introduction

This chapter presents the results of the descriptive analysis of the incidence of HAIs occurring in surgical patients and the results of the multivariable analysis that identified independent risk factors for HAI. The analysis is limited to those infections that presented during the in-patient period. The chapter begins with an exploration of the overall incidence of HAI as defined by two alternative definitions of incidence, and how the incidence varies with selected patient characteristics (section 7.2). Data on the day the primary infection presented are then presented in section 7.3. The results of the analysis that explored independent risk factors for HAI are presented in section 7.4 and sections 7.5 -7.6 present the results of the analysis that explored the incidence of specific types of HAI, and the day different types of infections presented. Finally sections 7.7-7.9 present the results of the analysis that examined how the incidence of the three most frequent types of infections, urinary tract, surgical wound and lower respiratory tract infections, varied with selected patient characteristics and the results of the analysis that explored independent risk factors for these infections.

7.2 The incidence of HAI

The incidence of HAI was assessed according to two alternative definitions:

- the number of patients who presented with one or more HAIs expressed as a percentage of the number of patients discharged
- the number of primary HAIs that presented during the in-patient period/1000 patient days at risk

The estimates of the incidence used only HAIs which were both acquired and identified during the hospital stay. Patients admitted with an HAI who did not acquire a second infection were excluded from the numerator and included in the denominator. There were nine patients in this category. In a further seven cases data on the day of presentation were not available. In these seven cases it was assumed that the patient acquired their infection at some point during their admission and that they were not admitted with the infection. This assumption was based on the fact that it is likely that if the patient had been admitted with an infection related to a previous admission the research assistant would have entered this information on the data collection sheet at the time of recruitment.

7.2.1 The number of patients who presented with one or more HAIs expressed as a percentage of the number of patients discharged

One hundred and eighty-four patients presented with one or more HAIs during the in-patient period: an incidence rate of 7.5% (95% CI: 6.4, 8.6). Table 7.1 shows how the incidence of HAI varied with key patient characteristics. The incidence of HAIs presenting during the in-patient period was higher in female patients, increased with increasing age, was higher in patients classified as emergency admissions when compared to elective admissions, and was highest in patients admitted to the gynaecology specialty, followed by patients admitted to the obstetric, orthopaedic, urology and surgical specialties. The incidence of HAI also varied with primary discharge diagnosis, although the confidence intervals around many of the disease categories were wide, and was higher in patients with co-morbidities than in those patients with no co-morbidities. The incidence of HAI was higher in patients who had diabetes mellitus listed as a co-morbidity although this was not found to be significant (p= 0.620), and was significantly higher in patients who had one or more operations.

heresterietic	[IAI	IR	P value
Characteristic	n	No	Yes	(95% Cl)	
Sex				E 4 (4 0 4+ 7 0)	40.004
	1003	952	51	5.1 (4.3 to 7.3)	<0.001
remaie	1465	1333	133	9.1 (7.6 to 10.6)	<u> </u>
Age				E 4 (0.0 k) D (0)	0.017
18-34	514	486	28	5.4 (3.9 to 8.0)	0.017
35-54	673	631	42	6.2 (4.5 to 8.3)	
55-74	951	872	79	8.3 (16.8 to 10.5)	ļ
75+	331	296	35	10.6 (8.0 to 15.1)	
Admission type					
Elective	1629	1520	109	6.7 (5.9 to 8.1)	0.044
Emergency	840	765	75	8.9 (7.1 to 11.1)	
Specialty	ļ				
General surgery	868	815	53	6.1 (4.6 to 7.9)	<0.001
Orthopaedics	501	464	37	7.4 (5.2 to 10.4)	
Urology	472	448	24	5.1 (3.9 to 8.4)	
Gynaecology	386	337	49	12.7 (9.9 to 16.8)	
Obstetrics	226	205	21	9.3 (6.2 to 14.4)	
Primary diagnosis					
Infectious & parasitic	13	12	1	7.7 (0.2 to 36.0)	0.009
Neoplasms	357	329	28	7.8 (5.3 to 11.1)	
Endocrine, nutritional & metabolic diseases & immunity disorders	24	20	4	16.7 (4.7 to 37.4)	
Diseases of blood & blood forming organs	3	3	0	0 (0.0 to 0.0)	
Diseases of the nervous system & sense organs	2	2	0	0 (0.0 to 0.0)	
Diseases of the circulatory system	150	142	8	5.3 (2.3 to 10.2)	
Diseases of the respiratory system	6	4	2	33.3 (4.3 to 77.7)	
Diseases of the digestive system	418	397	21	5.0 (3.1 to 7.6)	
Diseases of the genitourinary system	569	521	48	8.4 (6.6 to 11.4)	
Complications of pregnancy, childbirth & puerperium & certain conditions originating in the perinatal period	254	231	23	9.1 (6.1 to 13.7)	
Diseases of the skin & subcutaneous tissue	31	28	3	9.7 (2.5 to 26.9)	
Diseases of the musculoskletal system & connective tissue	309	297	12	4.2 (2.6 to 7.1)	
Injury & poisoning	187	164	23	12.3 (9.7 to 20.3)	
Symptoms, signs & ill- defined conditions; mental disorders & congenital abnormalities	146	135	11	7.5 (3.8 to 13.1)	
Number of co- morbidities					
None	1642	1538	104	6.3 (5.2 to 7.6)	<0.001
One	578	530	48	8.3 (6.4 to 11.1)	
Two	173	159	14	8.1 (4.4 to 13.1)	
Infee of more	76	58	18	23.7 (13.8 to 33.8)	
Diabetes menitus insted as a co-morbidity					0.600
No	2378	2202	176	7.4 (6.4 to 8.5)	0.620
188	91	83	8	8.8 (3.9 to 16.6)	
Une or more operations					
NO	428	409	19	4.4 (2.7 to 6.8)	0.009
Yes	2041	1876	165	8.1 (6.9 to 9.3)	

Table 7.1: The incidence of HAIs presenting in surgical patients during the hospitalised phase by selected characteristics

IR - incidence rate; CI -confidence interval

The results presented in Table 7.2 show how the incidence of HAI varied with primary operative procedure classified according to the Office of Population Census and Surveys Operation Classification System (fourth edition).²³¹ The results presented show how the incidence varied with primary procedure classified according to the broad body system groups. The variation in incidence rates observed for the different procedure categories was highly significant (p=<0.001). The analysis was limited to 2024 patients for whom data on primary operative procedure were available. The number of patients in some of the subgroups was small and as such it is difficult to draw strong conclusions. However, if categories which include more than 20 patients are considered, it can be seen that the highest incidence rates were observed in patients who had operations involving the upper digestive tract (IR 34.8%, 95%) CI: 16.4%, 57.3%), the lower female genital tract (IR 18.3%, 95% CI: 9.5%, 30.4%), the skin (IR 13%, 95% CI: 2.8%, 33.6%) and the upper female genital tract (IR 12.0%, 95%CI: 8.2%, 16.8%).

More detailed analysis examined how the HAI incidence rate varied with the specific type of primary procedure performed. The results are presented in Appendix 11. Amongst patients who had a primary procedure involving the upper digestive tract, the incidence of HAI was highest in patients whose primary procedure involved the ileum in the form of surgery such as the excision of the ileum or creation of an artificial opening in the ileum (57.1%) and in patients who had a total or partial excision of the stomach and/or oesophagus (37.5%). Amongst those patients whose surgery involved the lower female genital tract, the highest incidence was observed in patients whose primary procedure involved the upper female genital tract, the highest incidence was observed in patients whose surgery involved the upper female genital tract, the highest incidence was observed in patients whose surgery involved the upper female genital tract, the highest incidence was observed in patients whose surgery involved the upper female genital tract, the highest incidence was observed in patients whose surgery involved the upper female genital tract, the highest incidence was observed in patients whose surgery involved the upper female genital tract, the highest incidence was observed in patients whose a hysterectomy (14.9%).

[Operation category		HAI		10 (8/)
			No	Yes	(95% Cl)
A	Nervous system	10	10	0.0	0.0 (0.0 , 30.8)
B	Endocrine system and breast	104	100	4.0	3.8 (1.1, 9.6)
С	Еуе	0	0	0.0	0.0
D	Ear	0	0	0.0	0.0
Ε	Respiratory tract	1	0	1.0	100.0 (2.5, 100.0)
F	Mouth	7	6	1.0	14.3 (0.4, 57.9)
G	Upper digestive tract	23	15	8.0	34.8 (16.4, 57.3)
Н	Lower digestive tract	225	211	14.0	6.2 (3.4, 10.2)
J	Other abdominal organs - principally digestive	80	75	5.0	6.3 (2.1, 14.0)
ĸ	Heart	0	0	0.0	0.0-
L	Arteries and Veins	56	51	5.0	8.9 (3.0, 19.6)
М	Urinary	355	334	21.0	5.9 (3.6, 8.9)
N	Male genital organs	26	26	0.0	0.0 (0.0, 13.2)
P	Lower female genital	60	49	11.0	18.3 (9.5, 30.4)
Q	Upper female genital	242	213	29.0	12.0 (8.2, 16.8)
R	Female genital tract associated with pregnancy and childbirth	224	205	19.0	8.5 (5.2, 12.9)
S	Skin	23	20	3.0	13.0 (2.8, 33.6)
T	Soft tissue	179	168	11.0	6.1 (3.1, 10.7)
V	Bones and joints of skull and spine	15	15	0.0	0.0 (0, 21.8)
W	Other bones and joints	375	352	23.0	6.1 (3.9, 9.1)
X	Miscellaneous operations	10	7	3.0	30.0 (6.7, 65.2)
Y	Subsidiary classification of methods of operation	8	8	0.0	0.0 (0.0, 36.9)
Z	Subsidiary classification of sites of operations	1	1	0.0	0.0 (0.0, 98.0)

Table 7.2: Incidence of HAI by primary operative procedure category

Note: Based on 2024 patients for whom operation data available.

IR - incidence rate; CI -confidence interval

Further analysis examined whether the incidence of HAI varied with Body Mass Index (BMI). This aspect of the analysis was limited to 1588 patients for whom data on both height and weight were available. Patients were classified into three groups: patients with a BMI of less than 20; patients with a BMI of 20-29; and patients with a BMI of greater than 30. The incidence of HAI was found to vary with BMI category as follows: 10.3% (95% CI: 5.5 to 17.4) of the 115 patients who had a BMI of less than 20 acquired one or more infections, compared to 6.0% (4.7 to 7.4) of the 74 patients who had a BMI of 20-29 and 6.4% (3.0 to 8.7) of the 280 patients with a BMI of 30 or more. The variation in rates was not found to be significant (p=0.149)

7.2.2 The number of primary HAIs/1000 patient days at risk

The number of days at risk was calculated as the number of days from admission to either the day prior to the day the primary infection was identified, or date of discharge, depending on which came first.

As previously indicated 184 patients acquired and presented with an infection during their admission. Of these seven did not have a date of onset recorded. These seven patients were excluded from this analysis. The remaining 177 infected patients had a total of 1,280 days at risk and uninfected patients had 15,149 days at risk. The overall number of days at risk was therefore 16,429 days, giving an incidence density of 10.8 (95% CI: 9.2, 12.5) infections per 1000 patient days at risk.

Table 7.3 shows how the incidence of HAI per 1000 patient days at risk varied with key patient characteristics.

		No. of d		Total Ma	Ma of the		1
		Pts with an HAI	Pts with no HAI	of days at risk	with one of more HAIs	Incidence density (95% Cl)	P value
Patient characteristic	n	(a)	(b)	(a+b) = (c)	(d)	(d/c)*1000	1
Say		1					
Male	1001	430	5636	6066	49	8.1 (5.9, 10.7)	0.0093
Female	1461	850	9513	10363	128	12.4 (10.3, 14.7)	
Age group			······································				
18-34	514	156	2921	3077	28	9.1 (6.0, 13.2)	0.7924
35-54	672	215	3477	3692	41	11.1 (8.0, 15.1)	
55-74	946	600	5981	6581	74	11.2 (8.8, 14.1)	
75+	330	309	2770	3079	34	11.0 (7.6, 15.4)	
Admission specialty							
General surgery	984	499	4481	4980	53	10.6 (8.0, 13.9)	0.0001
Orthopaedics	496	286	4768	5054	32	6.3 (4.3, 8.9)	I
Urology	472	153	2304	2457	24	9.8 (6.3, 14.5)	
Gynaecology	384	236	2068	2304	47	20.4 (15.0, 27.1)	
Obstetrics	226	106	1528	1634	21	12.9 (8.0, 19.6)	
Admission type					107	44.0 (0.0 42.2)	0.7605
Elective	1627	613	9126	9739	107	11.0 (9.0, 13.3)	0.7505
Emergency	835	667	6023	6690	70	10.5 (0.2, 13.2)	
Primary diagnosis		L				20.9 (0.5 116 1)	0.0009
Infectious and parasitic	13	2	40	48	1	20.0 (0.5, 110.1)	0.0000
Neenlaama	257	206	2102	2308	28	11 7 (7.8, 16.9)	
Endoorine autritional & metabolic	24	12	07	109	4	36.7 (10.0, 93.9)	†{
diseases & immunity disorders	67			100			
Diseases of blood & blood forming organs	3	0	13	13	0	0.0 (0.0, 283.7)	
Diseases of the nervous system & sense organs	2	0	7	7	0	0.0 (0.0, 526.8)	
Diseases of the circulatory system	150	129	706	835	8	9.6 (4.1, 18.9)	
Diseases of the respiratory	6	19	28	47	2	42.6 (5.2, 153.7)	
Diseases of the digestive system	418	127	1937	2064	21	10.2 (6.3, 15.6)	
Diseases of the genitourinary	568	260	2799	3059	47	15.4 (11.3, 20.4)	
Complications of pregnancy, childbirth & puerperium & certain conditions originating in the perinatal period	254	114	1681	1795	23	12.8 (8.1, 19.2)	
Diseases of the skin &	31	23	126	149	3	20.1 (4.1, 58.8)	
Diseases of the musculoskletal	309	91	3122	3213	12	3.7 (1.9, 6.5)	
Injury & poisoning	181	171	1616	1787	17	9.5 (5.5, 15.2)	
Symptoms, signs & ill- defined conditions; mental disorders & concenital abnormalities	146	126	779	905	11	12.2 (6.1, 21.7)	
Co-morbidities							
None	1637	550	9568	10118	99	9.8 (8.0, 11.9)	0.0042
One	576	378	3690	4068	46	11.3 (8.3, 15.1)	
Тwo	173	203	1378	1581	14	8.9 (4.8, 14.9)	
Three or more	76	149	513	662	18	27.2 (6.1, 43.0)	
Diabetes listed as a co-morbidity							
No	2371	1179	14370	15549	169	10.9 (9.3, 12.6)	0.6119
Yes	91	101	779	880	8	9.1 (3.9, 17.9)	

Table 7.3The number of patients presenting with one or more HAIs in hospital
per 1000 patient days at risk

CI -- confidence interval

7.3 Day of presentation

Table 7.4 shows the number and percentage of primary infections by the day the infection was identified, and Figure 7.1 provides a graphical presentation of the proportion of infections presenting by day of admission. Data on the day the HAI presented was available for 177 of the 184 patients who were not admitted with an HAI, but acquired one or more infections whilst in hospital which presented during the in-patient phase. The results presented in Table 7.4 indicate that over 50% of the infected patients presented with their primary infection within six days of admission and 75% within nine days of admission.

Day primary HAI	Number of HAIs	% of HAIs	Cumulative %
presented			
2	4	2.3	2.3
3	21	11.9	14.1
4	27	15.3	29.4
5	22	12.4	41.8
6	19	10.7	52.5
7	19	10.7	63.3
8	13	7.3	70.6
9	8	4.5	75.1
10	6	3.4	78.5
11	4	2.3	80.8
12	7	4.0	84.7
13	2	1.1	85.9
14	1	0.6	86.4
15	5	2.8	89.3
16	4	2.3	91.5
17	3	1.7	93.2
18	1	0.6	93.8
19	2	1.1	94.9
20	1	0.6	95.5
21	2	1.1	96.6
22	1	0.6	97.2
23	0	0.0	97.2
24	1	0.6	97.7
25-27	0	0.0	97.7
28	1	0.6	98.3
29	0	0.0	98.3
39	1	0.6	98.9
40-44	0	0.0	98.9
45	1	0.6	99.4
47-49	0	0.0	99.4
50	1	0.6	100.0
Total	177	100.0	

Table 7.4: The day the	primary	YHAI	presented*
------------------------	---------	-------------	------------

The analysis was limited to 177 or the 184 HAIs presenting during the in-patient phase. In seven cases data on day of presentation was not available.

214





7.4. Independent risk factors for HAI

The preceding sections have presented the results of the analysis that examined how the incidence of HAI varied with different incidence measures and with selected patient characteristics. In this section the results of the multi-variable logistic regression analysis that identified independent risk factors for infection are presented. The factors included in the analysis were those listed in Tables 7.5 and 7.6. Details of the methods used can be found in section 5.5.2 of Chapter 5.

Table 7.5 presents the results of the single variable analysis and Table 7.6 the results of the multivariable analysis. The results of the single variable analysis indicate that with the exception of diabetes, the odds of acquiring an HAI whilst in hospital vary significantly with all the factors included in the analysis. Females were found to be at greater risk of acquiring an infection than males; the risk increased with age and increasing number of co-morbidities; emergency admissions were at higher risk than elective admissions; the risk was greatest amongst gynaecology patients and higher amongst patients who had received antibiotic prior to the onset of an infection. The results of the multivariable analysis indicted that this pattern remained, and with the exception of sex, antibiotics and diabetes the odds of acquiring an infection varied significantly with all the factors included in the analysis.
	H/	AJ .	Odds ratio	
n	No	Yes	(98% CI)	P value
1003	952	51	1.0	0.0002
1466	1333	133	1.9 (1.3, 2.6)	
514	486	28	1.0	0.0076
673	631	42	1.2 (0.7 , 1.9)	
951	872	79	1.6 (1.0, 2.5)	+
331	296	35	2.1 (1.2, 3.4)	
1629	1519	110	1.0	0.003
840	766	74	1.4 (1.0 , 1.9)	
		· · · · · · · · · · · · · · · · · · ·		
884	831	53	1.0	0.0003
501	464	37	1.3 (0.8, 1.9)	
472	448	24	0.9 (0.5 , 1.4)	
386	337	49	2.3 (1.5, 3.4)	
226	205	21	1.6 (1.0, 2.7)	
1642	1538	104	1.0	0.0016
579	531	48	1.3 (0.9, 1.9)	
249	235	14	2.2 (1.4, 3.3)	
1002	948	54	1.0	0.0074
1467	1338	129	1.7 (1.2, 2.4)	
2378	2202	176	1.0	0.6289
91	83	8	1.2 (0.6, 2.5)	
428	409	19	1.0	0.0056
2041	1876	165	1.9 (1.2.3.1)	
	n 1003 1466 514 673 951 331 1629 840 884 501 472 386 226 1642 579 249 1002 1467 2378 91 428 2041	n No 1003 952 1466 1333 514 486 673 631 951 872 331 296 1629 1519 840 766 884 831 501 464 472 448 386 337 226 205 1 1642 1642 1538 579 531 249 235 1002 948 1467 1338 2378 2202 91 83 428 409 2041 1876	No Yes 1003 952 51 1466 1333 133 514 486 28 673 631 42 951 872 79 331 296 35 1629 1519 110 840 766 74 884 831 53 501 464 37 472 448 24 386 337 49 226 205 21 1642 1538 104 579 531 48 249 235 14 1002 948 54 1467 1338 129 2378 2202 176 91 83 8 428 409 19 2041 1876 165	HAI Odds ratio (95% Cl) n No Yes 1003 952 51 1.0 1466 1333 133 1.9 (1.3, 2.6) 514 486 28 1.0 673 631 42 1.2 (0.7, 1.9) 951 872 79 1.6 (1.0, 2.5) 331 296 35 2.1 (1.2, 3.4) 1629 1519 110 1.0 840 766 74 1.4 (1.0, 1.9) 384 831 53 1.0 1.0 884 831 53 1.0 1.4 (1.0, 1.9) 472 448 24 0.9 (0.5, 1.4) 386 337 49 2.3 (1.5, 3.4) 226 205 21 1.6 (1.0, 2.7) 1642 1538 104 1.0 579 531 48 1.3 (0.9, 1.9) 249 </td

Table 7.5: The odds of acquiring a HAI by key patient characteristics(single variable analysis)

CI - confidence interval

• · · · ·

		HA	J		
Patient characteristic	n	No	Yes	Odds ratios (95% Cl)	P value
Sex					
Male	1003	952	51	1.0	0.061
Female	1466	1333	133	1.5 (1.0 , 2.3)	
Age group					
18 34	514	486	28	1.0	<0.0001
35-54	673	631	42	1.7 (1.0 , 3.1)	
55-74	951	872	79	3.4 (1.9 , 6.3)	
75+	331	296	35	4.2 (2.1 , 8.1)	
Admission type					
Elective	1629	1519	110	1.0	<0.0001
Emergency	840	766	74	2.3 (1.6 , 3.4)	
Specialty					
General surgery	884	831	53	1.0	<0.0001
Orthopaedics	501	464	37	0.9 (0.6 , 1.5)	
Urology	472	448	24	0.9 (0.5 , 1.6)	
Gynaecology	386	337	49	3.0 (1.8 , 4.8)	
Obstetrics	226	205	21	1.9 (0.9 , 3.9)	
Number of co-morbidities					
None	1642	1538	104	1.0	0.0118
One	579	531	48	1.3 (0.9 , 2.0)	
Two or more	249	235	14	2.1 (1.3 , 3.4)	
Diabetes					
No	2378	2202	176	1.0	0.4601
Yes	91	83	8	0.7 (0.3 , 1.7)	
Antibiotics pre HAI- any route					
No	1002	948	54	1.0	0.1340
Yes	1467	1338	129	1.3 (0.9 , 1.8)	
Operation					
No	428	409	19	1.0	0.0002
Yes	2041	1876	165	2.7 (1.5 , 4.7)	

Table 7.6: The odds of acquiring a HAI by key patient characteristics(multivariable analysis)

CI - confidence interval

.

7.5 The incidence of specific types of HAI

Whilst 184 patients acquired and presented with one or more HAIs during the in-patient phase, overall 208 infections were identified during this time period. UTIs were the most frequent type of infection accounting for 48.1% of the infections observed, followed by SWIs and LRTIs (Table 7.7).

Type of HAI	Number of Infections	Proportion of infections identified (%)	Incidence of HAI (%) (95% CI)
UTI	100	48.1	4.1 (3.3 , 4.9)
LRTI	23	11.1	0.9 (0.6 , 1.4)
SWI	40	19.2	1.6 (1.2 , 2.2)
BSI	4	1.9	0.2 (0.0 , 0.4)
Skin	16	7.7	0.6 (0.4 , 1.1)
Other	25	12.1	1.0 (0.7 , 1.5)

Table 7.7: The number, percentage and incidence of HAI by site of infection

UTI - urinary tract infection; LRTI - lower respiratory tract infection; SWI - surgical wound infection; BSI - bloodstream infection CI - confidence interval

7.6 Day of presentation of different types of infection

Table 7.8 shows the number and percentage of infections by the day the infection was identified. Data on the day the HAI presented was not available for all HAIs. Details of the number of infections included for each site specific analysis are presented at the bottom of the table. The results presented in Table 7.8 indicate that for all sites of infection at least 50% of the infections identified presented within eight days of admission, and with the exception of skin infections 75% or more of each type of infection presented within two weeks of admission.

sites	% ui			1.7	6.1	7.8	0.9	9.6	9.6	8.3	8.3	2.6	2.6	2.6	7.0	1.3	1.3	1.3	1.3	1.3	5.7	5.7	5.7	5.7	5.7	
other	Ŭ 			7 2	3 2	7 4	0	7 6	0	7 7	0 7	3	0	0	3	3	6	0	0	0	3	0	6 0	6	6	0
ons at	~	-	-	21.	4	21.	13.	ø	0	ø	o	4	o	o	4	4	o	0	o	o	4	0	o	0	0	•
Infection	No			5	-	5	e	2	0	2	0.0	-	0	0	-	-	0	0	0	0	-	0	0	0	0	0
suo	Cum %	6.25	12.5	25	43.8	50.0	50.0	56.3	56.3	56.3	56.3	62.5	62.5	62.5	68.89	75.0	87.5	87.5	93.75	100.0	100.0	100.0	100.0	100.0	100.0	1000
kin infect	% of skin	6.25	6.25	12.5	18.75	6.25	0.0	6.25	0.0	0.0	0.0	6.25	0.0	0.0	6.25	6.25	12.5	0.0	6.25	6.25	0.0	0.0	0.0	0.0	0.0	0
0	No	-	-	2	3	-	0	-	0	0	0	-	0	0	-	-	2	0	-	-	0	0	0	0	0	0
	Cum %	25.0	25.0	25.0	50.0	50.0	50.0	75.0	75.0	75.0	75.0	75.0	75.0	75.0	75.0	75.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0
BSIs	% of BSIs	25.0	0.0	0.0	25.0	0.0	0.0	25.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	25.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	00
	Ŷ	-	0	0	-	0	0	-	0	0	0	0	0	0	0	0	-	0	0	0	0	0	0	0	0	00
	Cum %	0.0	0.0	5.9	17.6	20.6	41.2	52.9	58.8	61.8	64.7	70.6	76.5	76.5	79.4	85.3	88.2	88.2	91.2	91.2	91.2	94.1	94.1	94.1	94.1	1 10
SWI	% of SWIs	0.0	0.0	5.9	11.8	2.9	20.6	11.8	5.9	2.9	2.9	5.9	5.9	0	2.9	5.9	2.9	0.0	2.9	0.0	0.0	2.9	0.0	0.0	0.0	00
	Ŷ	0.0	0.0	2	4	-	2	4	2	-	-	2	2	0	-	2	-	0.0	-	0.0	0.0	-	0.0	0.0	0.0	000
	Cum %	0.0	8.7	39.1	47.8	60.9	9.69	78.3	82.6	82.6	82.6	87.0	95.7	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0
LRTIs	% of LRTIs	0.0	8.7	30.4	8.7	13.0	8.7	8.7	4.3	0.0	0.0	4.3	8.7	4.3	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	00
	No	0	e	7	2	e	2	2	-	0	0	-	2	+	0	0	0	0	0	0	0	0	0	0	0	0
	Cum %	3.0	18.2	31.3	48.5	54.5	65.7	73.7	79.8	83.8	85.9	89.9	89.9	89.9	90.9	91.9	92.9	93.9	94.9	96.0	97.0	97.0	97.0	98.0	98.0	08.0
5	% of UTIS	3.0	15.2	13.1	17.2	6.1	11.1	8.1	6.1	4.0	2.0	4.0	0.0	0.0	1.0	1.0	1.0	1.0	1.0	1.0	1.0	0.0	0.0	1.0	0.0	00
	No	3	15	13	17	9	11	80	9	4	2	4	0	0	1	-	1	1	1	1	1	0	0	1	0	0
Day HAI presented		2	3	4	5	9	7	80	6	10	11	12	13	14	15	16	17	18	19	20	21	22	23	24	25	ac

Table 7.8: The day HAIs presented by site of infection

1	
-	S
1	
4	5
	9
-	S
1	2
-	
-	ē
1	en
1	es
1	ā
-	S
-	Ĩ
	2
-	B
1	he
ł	-
-	ö
	ne
	E
1	5
	ŭ
-	
	Ð
-	Q
ł	-

	5			LRTIS			IMS			BSIs		0	cin infect	suo	Infection	s at oth	er sites
	% of UTIS	Cum %	No	% of LRTIs	Cum %	No	% of SWIs	Cum %	No	% of BSIs	Cum %	No	% of skin	Cum %	No	%	% mn)
1	0.0	98.0	0	0.0	100.0	0	0.0	94.1	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	0.0	98.0	0	0.0	100.0	-	2.9	97.1	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	0.0	98.0	0	0.0	100.0	0	0.0	97.1	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	0.0	98.0	0	0.0	100.0	•	0.0	97.1	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	0.0	98.0	0	0.0	100.0	0	0.0	97.1	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	0.0	98.0	0	0.0	100.0	0	0.0	97.1	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	0.0	98.0	0	0.0	100.0	0	0.0	97.1	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	0.0	98.0	0	0.0	100.0	0	0.0	97.1	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	0.0	98.0	0	0.0	100.0	0	0.0	97.1	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	0.0	98.0	0	0.0	100.0	0	0.0	97.1	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
	0.0	98.0	0	0.0	100.0	0	0.0	97.1	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
	0.0	98.0	0	0.0	100.0	0	0.0	97.1	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	1.0	99.0	0	0.0	100.0	-	2.9	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
	0.0	0.66	0	0.0	100.0	ö	0.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	96.7
-	1.0	100.0	0	0.0	100.0	•	0.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	1.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	1.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	1.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
1	1.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	95.7
	1.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	100.0	0	0.0	100.0	1	4.3	100.0
		100.0	24		100.0	34		100.0	4		100.0	16		100.0	23		100.0

7.7 The incidence of UTIs and identification of risk factors

7.7.1 The incidence of UTIs

Of the 2469 patients involved in this study, 100 patients acquired and presented with a UTI during the in-patient period: an incidence rate of 4.1% (95% CI: 3.3 to 4.9), with 13 patients presenting with an infection at one or more additional sites. Of these 13 patients, in eight cases the UTI was the primary infection. The incidence density was 5.7(95% CI: 4.7, 7.0) UTIs per 1000 patient days at risk, and 17.4 utis/1000 catheter days at risk.

Table 7.9 shows how the incidence of UTIs varies with selected patient characteristics. The characteristics included in this analysis included both intrinsic and extrinsic risk factors for hospital acquired UTIs identified from the literature and for which data were available (see section 2.8.3 for a discussion of risk factors for UTIs).

The incidence of UTIs was found to be significantly higher in women (p=<0.001) and varied with admission specialty (p<0.001), being highest in patients admitted to the gynaecology and obstetric specialties. The incidence of UTI was also found to vary significantly with discharge diagnosis, being highest in patients with a primary discharge diagnosis classified as 'Diseases of the genitourinary system', and was considerably higher in patients who had a catheter in situ prior to their infection. Urinary tract infection rates were 2.3 times higher in patients who had a catheter inserted compared to patients who did not have a catheter inserted. Finally, the incidence of UTI was found to be significantly higher in patients who had an operation and in those patients who received antibiotics prior to the infection, compared to those who did not, a factor which suggests that administration of antibiotics is a marker for another risk factor. That is prophylactic antibiotics are administered to patients thought to be at greater risk of acquiring an infection than those not given prophylactic antibiotics. With respect to the other variables included in this analysis, whilst the incidence of UTI was found to vary with age group, admission type and number of co-morbidities, the observed variation was not found to be statistically significant at the 5% level.

		TU	1	IR (%)	D volue
Patient characteristic	п	No	Yes	(95% Cl)	r value
Sex					
Male	1003	985	18	1.8 (1.1 , 2.8)	<0.001
Female	1466	1384	82	5.6 (4.5 , 6.9)	
Age group					
18 - 34	514	498	16	3.1 (1.8, 5.0)	0.203
35-54	673	646	27	4.0 (2.7, 5.9)	
55-74	951	914	37	3.9 (2.8, 5.3)	
75+	331	311	20	6.0 (3.7, 9.0)	
Admission type					
Elective	1629	1563	66	5.2 (4.0 , 6.6)	0.826
Emergency	840	807	34	4.0 (2.8, 5.6)	1
Registy					1
General surgery	884	869	15	1.7 (1.0 . 2.8)	<0.001
Orthonandian	501	499	13	26 (14, 44)	
	472	455	18	38 (23, 6.0)	
	296	244	41	10.6 (7.7, 14.1)	
Chatabia	300	212	41	58 (31 96)	1
	220	213	13	3.0 (0.1 , 0.0)	
Discharge diagnosis group	13	12	1	77 (0.2, 36.0)	0.005
Necesia	367	341	16	45 (26.7.2)	1
Endocrine, nutritional and metabolic diseases and	24	24	0	0.0 (0.0, 14.2)	1
immunity disorders		24		0.0 (0.0 70.8)	
Diseases of blood & blood forming organs	3		0		
Diseases of the nervous system & sense organs	2	2	0		
Diseases of the circulatory system	150	145	4	2.7 (0.9 , 7.1)	<u> </u>
Diseases of the respiratory system	6	6	0		
Diseases of the digestive system	418	410	8	1.9 (0.8, 3.7)	
Diseases of the genitourinary system	569	530	39	5.9 (4.9 , 9.3)	
Complications of pregnancy, childbirth & puerperium & certain conditions originating in the perinatal period	254	239	15	5.9 (3.3 , 9.6)	ļ
Diseases of the skin & subcutaneous tissue	31	31	0	0.0 (0.0 , 11.2)	
Diseases of the musculoskletal system & connective tissue	309	305	4	1.3 (0.4 , 3.3)	
Injury & poisoning	187	178	9	4.8 (2.4, 8.9)	
Symptoms, signs & ill- defined conditions; mental	146	142	4	2.7 (0.8 , 7.1)	
Number of co-morbidities	+				
None	1641	1578	63	3.8 (3.0 , 5.0)	0.862
One	579	553	26	4.5 (3.0 , 6.5)	
	249	238	11	4.4 (2.2.7.8)	
					1
	072	044	29	29 (21 44)	0.04
Vas	312	4405	74	47 (37 60)	
	1490	1420		4.1 (0.7 , 0.07	
Antibiotics prior to UTI - IV/M	4040	4474	20	12 (23 44)	0.083
No	1210	11/1	39	<u>J.2 (2.0, 4.4)</u>	
	1258	1190		4.0 (0.7 , 0.17	+
	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	4944	42	24 (10 34)	<0.001
Vac	1/5/	1/14	43	2.7 (1.8, 3.7)	-0.001
65 T	1 711	600	00	1.8 (0.0, 10.1)	+
		444	<u> </u>	10/00 34	0.000
NO	428	421	<u> </u>	1.0 (0.0 , 3.4)	0.002
Yes	2041	1948	93	4.5 (3.7, 5.5)	1

Table 7.9: The incidence of UTIs in surgical patients

UTI – urinary tract infection; IR – incidence rate * In the absence of data on day of operation for all 2041 patients who had an operative procedure the assumption was made that patients acquired their UTI after their first operative procedure. Note: With the exception of the antibiotic and catheter variables this analysis is based on all 2469 patients. The analysis examining how UTI rates

7.7.2 Independent risk factors for UTIs

In order to assess which factors were independently associated with an increased risk of acquiring a UTI, a logistic regression analysis was carried out. The analysis included all the variables listed in Table 7.9 with the exception of primary discharge diagnosis group, which was omitted from this analysis. The confidence intervals for the odds ratios of acquiring a UTI for the different primary discharge diagnosis group were very wide and as such the inclusion of this variable had little value. Table 7.10 presents the results of the single variable analysis, and Table 7.11 the results of the multivariable analysis.

It can be seen from Table 7.11 that after controlling for a range of factors the presence of a urinary catheter at some point prior to the infection and female sex were associated with the greatest increases in risk. The odds of acquiring a UTI were 2.6:1 in catheterised patients compared to non-catheterised patients and 2.8:1 in females compared to male patients. The odds of acquiring a UTI were also found to increase with increasing age category, and varied significantly with admission specialty. The odds of acquiring a UTI were higher for gynaecology and obstetric patients, compared to patients in the other specialty groups. The odds of acquiring a UTI were also greater in patients who had one or more co-morbidities. However there was no evidence that the odds increased with increasing number of co-morbidities. The odds of acquiring a UTI were higher for emergency patients, and were similar in patients who had antibiotics prior to a UTI and in those who had not, although the upper confidence intervals suggest that in some case patients who received prophylactic antibiotics were at greater risk of acquiring a UTI. This outcome suggests that the administration of prophylactic antibiotics is perhaps a maker for another risk factor not included in this model.

	T	UTI		Odds ratios	Buche
Patient characteristic	n	No	Yes	(95% Cl)	- value
Sex					
Male	1003	985	18	1.00	<0.0001
Female	1466	1384	82	3.24 (1.9 , 5.4)	
Age group					
18 - 34	514	498	16	1.00	0.2331
35-54	673	646	27	1.30 (0.7 , 2.4)	
55-74	951	914	37	1.26 (0.7 , 2.4)	
75+	331	311	20	2.00 (1.0 , 3.9)	
Admission type					
Elective	1629	1563	66	1.00	0.8233
Emergency	840	807	34	0.95 (0.6 , 1.5)	
Specialty					
General surgery	884	869	15	1.00	<0.0001
Orthopaedics	501	488	13	1.54 (0.7 , 3.3)	
Urology	472	454	18	2.16 (1.1 , 4.4)	
Gynaecology	386	345	41	7.07 (3.9 , 12.9)	
Obstetrics	226	213	13	3.54 (1.7 , 7.5)	
Number of co-morbidities					
None	1641	1579	63	1.00	0.8634
One	579	553	26	1.11 (0.7 , 1.8)	
Two or more	249	238	11	1.16 (0.6 , 2.2)	
Diabetes listed as a co morbidity					
No	2378	2283	95	1.0	0.4977
Yes	91	86	5	1.4 (0.6 , 3.5)	
Antibiotics pre UTI – any route					
No	972	944	28	1.00	0.0266
Yes	1496	1425	71	1.62 (1.1 , 2.3)	
Antibiotics pre UTI – IV/IM			1		
No	1210	1171	39	1.00	
Yes	1258	1198	60	1.47 (1.0 , 2.2)	0.0636
Catheter present prior to UTI					
No	1757	1714	43	1.00	<0.0001
Yes	711	655	56	3.41 (2.3 , 5.1)	
Operation			1		
No	428	421	7	1.00	0.002
Yes	2041	1948	93	2.9 (1.3 , 6.2)	

Table 7.10:The odds of acquiring a UTI key patient characteristics(single variable analysis)

UTI - urinary tract infection, CI - Confidence interval

Note: The analysis limited to 2468 patients for whom complete data sets were available

Variable	Odds Ratio	Sig of Log
	(95% Cis)	likelihood ratio
Sex		
Male	1.0	0.0027
Female	2.8 (1.4 , 5.6)	
Age group		
18-34	1.0	0.0154
35-54	1.8 (0.8 , 3.9)	
55-74	2.6 (1.1 , 6.0)	
75+	4.2 (1.7 , 10.4)	
Admission type		
Elective	1.0	0.0132
Emergency	2.0 (1.2 3 ,.5)	
Specialty		
General surgery	1.0	<0.0001
Orthopaedics	1.2 (0.5 , 2.7)	
Urology	1.9 (0.8 , 4.3)	
Gynaecology	5.2 (2.6 , 10.6)	
Obstetrics	3.1 (1.1 , 8.5)	
Number of co-morbidities		
None	1.0	
One	1.2 (0.7 , 2.0)	0.8292
Two or more	1.2 (0.5 , 2.5)	
Diabetes listed as a co-morbidity		
No	1.0	0.5242
Yes	1.4 (0.5 , 4.0)	
Antibiotics (any route) administered pre NUTI)		
No	1.0	0.8362
Yes	1.1 (0.5 , 2.3)	
Antibiotics (IV/IM) administered pre NUTI		
No	1.0	0.9205
Yes	1.0 (0.5 , 2.0)	
Catheter present prior to UTI	· · ·	
No	1.0	<0.0001
Yes	2.6 (1.6 , 4.1)	
Operation		
No	1.0	0.0191
Yes	2.7 (1.1 , 6.5)	

Table 7.11: The odds of acquiring a UTI by selected factors after controllingfor all other factors listed (multivariable analysis).

CI - Confidence interval

Note: The analysis limited to 2468 patients for whom complete data sets were available

7.8 Incidence of SWIs and the identification of risk factors

7.8.1 Overall incidence of SWIs

Of the 2469 patients recruited into the study 44 presented with a SWI with 40 (1.6%, 95% CI: 1.2, 2.2) acquiring and presenting with a SWI during their admission to hospital. Twelve patients acquired one or more infections at other sites as well. The incidence density was 2.3 (95% CI: 1.6, 3.1) SWIs per 1000 patient days at risk.

Table 7.12 shows how the incidence varied with key patient characteristics. There was little variation in the incidence of SWI by the factors listed in Table 7.12.

7.8.2 The incidence of SWIs occurring in patients who had one or more operative procedures.

Of the 2469 patients recruited into the study 2041 had one or more operative procedures. Of these 2041 patients, 38 acquired a SWI: an incidence rate of 1.9% (95% CI: 1.3, 2.5). Table 7.13 shows how the incidence varied with selected patient characteristics. As with the analysis that included all patients, there was little variation in the incidence of SWIs with the patient characteristics included in the analysis.

7.8.3 Independent risk factors for SWI

In order to assess which factors were independently associated with an increased risk of acquiring a SWI, a logistic regression analysis was carried out. Since SWIs by definition can only occur in patients who had one or more operations, the analysis was limited to patients who had one or more operative procedures (n=2041). Table 7.14 presents the results of the single variable analysis, and Table 7.15 the results of the multivariable analysis. The analysis included all the variables listed in these tables.

There was little variation in the odds of acquiring a SWI with the factors included in the analysis. The only significant variation was with wound drain status. Patients who had one or more wound drains in place were 2.2 (95% CI: 1.0, 5.0) times more likely to acquire a SWI than those who did not.

		SI	NI	IR (%)	P value
	n	No	Yes	(95% Čl)	r vaiut
Sex					
Male	1003	989	14	1.4 (0.8 , 3.3)	0.465
Female	1466	1440	26	1.8 (1.2 , 2.6)	
Age group					
18-34	514	508	6	1.2 (0.4 , 2.5)	0.236
35-54	673	666	7	1.0 (0.4 , 2.1)	
55-74	951	932	19	2.0 (1.2 , 3.1)	
75+	331	323	8	2.4 (1.0 , 4.7)	
Specialty					
General surgery	884	868	16	1.8 (1.0 , 2.9)	0.162
Orthopaedics	501	489	12	2.4 (1.2 , 4.1)	
Urology	472	470	2	0.4 (0.1 , 1.5)	
Gynaecology	386	379	7	1.8 (0.7 , 3.7)	
Obstetrics	226	223	3	1.3 (0.3 , 3.8)	
Admission type	-				
Flactive	1629	1604	25	1.5 (1.0 , 2.3)	0.555
Emergency	840	825	15	1.8 (1.0 , 2.9)	
			1		
Infactions & parasitic disassa	13	13	0	0.0 (0.0 , 24.7)	0.239
	357	350	7	2 (0.8 , 4.0)	
Endocrine, nutritional & metabolic diseases &	24	23	1	4.2 (0.1 , 21.1)	
Disasses of blood & blood forming organs	3	3	0	0.00 (0.0 , 70.8)	
Diseases of the nervous system & sense organs	2	2	0	0.0 (0.0 , 84.2)	
Diseases of the alreulatory surtem	150	147	3	2 (0.4 , 5.7)	
	6	6	0	0 (0.0 , 45.9)	
Diseases of the disease system	418	414		1.0 (0.3 , 2.4)	
	560	583	6	1.1 (0.4, 2.3)	
Complications of pregnancy, childbirth & puerperium & certain conditions originating in the	254	251	3	1.2 (0.2 , 3.4)	
perinatal period		20		3 2 (0.1 , 16.7)	
Diseases of the skin & subcutaneous tissue				0.2 (0.1 (0.0)	
	309	305	4	1.3 (0.4 , 3.3)	
Iniury & poisoning	187	178	9	4.8 (2.2 , 8.9)	
Symptoms, signs & ill- defined conditions; mental disorders & congenital abnormalities	146	144	2	1.4 (0.2 , 4.9)	
Co-morbidities					
None	1642	1622	20	1.2 (0.7 , 1.9)	0.083
One	578	564	14	2.4 (1.3 , 4.0)	
Two or more	249	243	6	2.4 (0.9 , 5.2)	
Antibiotics (any route) administered pre SWI		1			
No	968	956	12	1.2 (0.6 , 2.2)	0.18
Yes	1501	1473	28	1.9 (1.2 , 2.7)	
Antibiotics (IV/IM) administered pre SWI		1	1		
No	1213	1199	14	1.2 (0.6 , 1.9)	0,10
Yes	1256	1230	26	2.1 (1.4 , 3.1)	
• • • •		<u> </u>		فسيريا أحسب ومجرو يتأوينا والمستوي والمستوي والم	

Table 7.12: The incidence of SWIs in surgical patients

SWI - surgical wound infections, IR - Incidence rate, CI - Confidence interval

		SM	/1	IR (%)	D value
Patient characteristic	n	No	Yes	(95% CI)	r value
Sex					
Male	815	799	16	2.0 (1.1, 3.2)	0.7823
Female	1226	1204	22	1.8 (1.1, 2.7)	
Age group					0.1665
18-34	416	410	6	1.4 (0.5, 3.1)	0.1000
35-54	570	564	6	1.1 (0.4 , 2.3)	
55-74	787	769	18	2.3 (1.4 , 3.6)	
75+	268	260	8	3.0 (1.3 , 5.8)	
Specialty					
General surgery	682	666	16	2.3 (1.3 , 3.8)	0.3083
Orthopaedics	433	422	11	2.5 (1.3 , 4.5)	
Urology	373	370	3	0.8 (0.2 , 2.3)	
Gynaecology	327	322	5	1.5 (0.5 , 3.5)	
Obstetrics	226	223	3	1.3 (0.3 , 3.8)	
Admission type					
Elective	1559	1536	23	1.5 (0.9 , 2.2)	0.0202
Emergency	482	467	15	3.1 (1.8, 5.1)	
Diagnosis group					
Infectious & parasitic disease	3	3	0	0.0 (0.0 , 70.8)	0.328
Neoplasms	322	315	7	2.2 (0.9, 4.3)	
Endocrine, nutritional & metabolic diseases & immunity disorders	24	23	1	4.2 (0.1 , 21.1)	
Diseases of blood & blood forming organs	1	1	0	0.0 (0.0 , 97.5)	
Diseases of the nervous system & sense organs	0	0	0	_	
Diseases of the circulatory system	131	128	3	2.3 (0.5, 6.5)	
Diseases of the respiratory system	5	5	0	0.0 (0.0 , 52.2)	
Diseases of the digestive system	327	323	4	1.2 (0.3, 3.1)	
Diseases of the gentourinary system	493	428	5	1.0 (0.3, 2.4)	
Complications of pregnancy, childbirth & puerperium	044	229	2	12 (03 36)	
& certain conditions originating in the perinatal period	241	230	3	1.2 (0.3 , 5.0)	
Diseases of the skin & subcutaneous tissue	25	24	1	4.0 (0.1, 20.4)	
Diseases of the musculoskletal system & connective tissue	287	282	5	1.7 (0.6 , 4.0)	
Injury & poisoning	142	135	7	4.9 (2.0 , 9.9)	
Symptoms, signs & ill- defined conditions; mental disorders & congenital abnormalities	40	38	2	5.0 (0.6 , 16.9)	
Co-morbidities					0.0008
None	1396	1376	20	1.4 (0.9 , 2.2)	0.0990
One	452	440	12	2.7 (1.4, 4.6)	
Two or more	193	187	6	3.1 (1.1, 6.6)	
Diabetes listed as a co-morbidity					
No	1971	1935	36	1.8 (1.3 , 2.5)	0.8332
Yes	70	2	2	2.9 (0.3, 9.9)	
Antibiotics (any route) administered pre SWI					
No	720	708	12	1.7 (0.9 , 2.9)	0.63012
Yes	1321	1295	26	2.0 (1.3 , 2.9)	

Table 7.13:The incidence of SWI occurring in patients who had one or more operative procedures

Patient characteristic		S	W	Odds Ratio (95% Cl)	P value
	n	No	Yes		
Sex					
Male	815	799	16	1.0	0.8973
Female	1226	1204	22	1.0 (0.5 , 2.1)	
Age group	T				
18-34	416	410	6	1.0	0.2789
35-54	570	564	6	0.7 (0.2 , 2.3)	
55-74	787	769	18	1.5 (0.6 , 3.9)	
75+	268	260	8	1.8 (0.6 , 5.5)	- ·
Specialty					
General surgery	682	666	16	1.0	0.1514
Orthopaedics	433	422	11	1.0 (0.4 , 2.2)	
Urology	373	370	3	0.2 (0.1 , 1.0)	
Gynaecology	327	322	5	0.6 (0.2 , 1.8)	
Obstetrics	226	223	3	0.6 (0.2 , 1.9)	
Admission type					
Elective	1559	1536	23	1.0	0.0896
Emergency	482	467	15	1.9 (0.9, 3.7)	
Co-morbidities			1		
None	1396	1376	20	1.0	0.0659
One	452	440	12	2.1 (1.0 , 4.4)	
Two or more	193	187	6	2.5 (1.0 , 6.3)	
Diabetes listed as a co-morbidity			1		
No	1971	1935	36	1.0	0.5147
Yes	70	2	2	1.7 (0.4 , 7.1)	
Antibiotics (any route) administered pre SWI					
No	720	708	12	1.0	0.2682
Yes	1321	1295	26	1.5 (0.7, 3.1)	
Antibiotics (IV/IM) administered pre SWI					
No	904	891	13	1.0	0.0885
Yes	1137	1112	25	1.8 (0.9 , 3.7)	
Wound drain prior to SWI	1		1		
No	1921	1894	28	1.0	0.0071
Yes	548	539	12	2.5 (1.3, 4.9)	

Table 7.14:The odds of acquiring a SWI by key patient characteristics(single variable analysis)

SWI - surgical wound infections

Patient characteristic		sv	vi	Odds Ratio (95% Cl)	P value
	n	No	Yes		
Sex					
Male	815	799	16	1.0	0.6812
Female	1226	1204	22	0.8 (0.4 , 1.9)	
Age group					
18-34	416	410	6	1.0	0.3978
35-54	570	564	6	0.8 (0.2 , 2.8)	
55-74	787	769	18	1.7 (0.5 , 5.3)	
75+	268	260	8	1.9 (0.5 , 6.6)	
Specialty					
General surgery	682	666	16	1.0	0.3536
Orthopaedics	433	422	11	0.6 (0.3 , 1.5)	
Urology	373	370	3	0.3 (0.1 , 1.4)	
Gynaecology	327	322	5	1.2 (0.4 , 4.0)	
Obstetrics	226	223	3	0.7 (0.1 , 3.5)	
Admission type					
Elective	1559	1536	23	1.0	0.1043
Emergency	482	467	15	1.9 (0.9 , 4.1)	
Co-morbidities					
None	1396	1376	20	1.0	0.2693
One	452	440	12	1.9 (0.8 , 4.1)	
Two or more	193	187	6	1.8 (0.6 , 5.1)	
Diabetes listed as a co-morbidity					
No	1971	1935	36	1.0	0.820
Yes	70	2	2	0.8 (0.2 , 3.9)	
Antibiotics (any route) administered pre SWI					
No	720	708	12	1.0	0.5136
Yes	1321	1295	26	0.5 (0.1 , 4.3)	
Antibiotics (IV/IM) administered pre SWI					
No	904	891	13	1.0	0.3396
Yes	1137	1112	25	2.4 (0.3 , 18.6)	
Wound drain prior to SWI					
No	1921	1894	28	1.0	0.0501
Yes	548	539	12	2.2 (1.0 , 5.0)	

Table 7.15: The odds of acquiring a SWI by key patient characteristics (multi-variable analysis)

SWI - surgical wound infections, CI - Confidence Interval

7.9 The incidence of LRTIs and identification of independent risk factors

7.9.1 The incidence of LRTIs

Of the 2469 patients involved in this study 23 (0.9%: 95% CI: 0.6, 1.4) acquired and presented with a LRTI during the in-patient period. The incidence density was 1.3 (95%CI: 0.8,1.9) per 1000 days at risk. Table 7.16 shows how the incidence of LRTI varied with selected patient characteristics. The results indicated that the incidence was higher in males than females; increased with increasing age; was highest in surgical patients; was higher in emergency as compared to elective admissions; was highest in patients whose primary discharge diagnosis was classified as a neoplasm; increased with increasing number of co-morbidities; was higher in patients who had diabetes listed as a co-morbidity, patients who had surgery, patients with a naso-gastric tube and/or endotracheal or tracheostomy tube in place prior to the onset of infection, and higher in patients who received antibiotics prior to infection. However, the only significant variation was with age group, number of co-morbidities, presence of a naso-gastric and/or endotracheal or tracheostomy tube prior to the onset of infection and increasing age.

7.9.2 Independent risk factors for LRTIs

To assess which factors were independently associated with an increased risk of acquiring a LRTI, a logistic regression analysis was conducted. Tables 7.18 and 7.19 present the results of the single and multivariable logistic regression analysis. In the single variable analysis significant variation was observed with respect to age, the number of co-morbidities, and presence of an NG tube and/or endotracheal or tracheostomy tube. However, the results of the multivariable analysis indicated that the only significant variation in the odds of acquiring a LRTI was with the presence or absence of an endotracheal or tracheostomy tube. The odds of acquiring a LRTI were found to be 44.4 (95% CI: 6.6, 288.0) times higher in patients who had an endotracheal or tracheostomy tube than those who did not.

		LRT		IR%	
Patient characteristic	n	No	Yes	(95% Cl)	P value
Sex					0.110
Male	1003	990	13	1.3 (0.7, 2.2)	0.119
Female	1466	1456	10	0.7 (0.3, 1.3)	
Age group		E 40		02 (00 11)	0.024
18-34	514	513			0.024
35-54	673	669	4	0.6 (0.2, 1.5)	
55-74	951	940	11	1.2(0.0, 2.1)	
75+	331	324		2.1 (0.9, 4.3)	
Specialty	884	871	13	1.5 (0.8, 2.5)	0.299
General surgery	501	497		0.8 (0.2.2.0)	
Ornopaedics	472	470	2	04 (0.1.1.5)	
Urology	296	383		08 (0.2, 2.3)	
Gynaecology	200	225	$\left \begin{array}{c} \bullet \\ \bullet \end{array} \right $	04 (00, 24)	
Obstetrics	220	225		0.4 (0.0 ; 2.1)	
Admission type	1620	1616	12	0.8 (0.4 , 1.4)	0.336
Elective	1029	920	10	12 (06, 22)	
Emergency	040	030		1.2 (0.0 / 2/	
Ulagnosis group	12	13	0	0.0 (0.0 . 24.7)	0.0387
	357	351	6	1.7 (0.6 , 3.6)	
Neoplasms					
disorders	24	23	1	4.2 (0.1, 21.9)	
Diseases of blood & blood forming organs	3	3	0	0.0 (0.0 , 70.8)	
Diseases of the nervous system & sense organs	2	2	0	0.0 (0.0 , 45.9)	
Diseases of the circulatory system	150	149	1	0.7 (0.0 , 3.7)	
Diseases of the respiratory system	6	5	1	16.7 (0.4, 64.1)	
Diseases of the digestive system	418	414	4	1.0 (0.3 , 2.4)	
Diseases of the genitourinary system	569	566	3	0.5 (0.1, 1.5)	
Complications of pregnancy, childbirth & puerperium &	254	253	1	0.4 (0.0 , 2.2)	
Diseases of the skin & subcutaneous tissue	31	31	0	0.0 (0.0 , 11.2)	
Diseases of the musculoskietel system & connective tissue	309	307	2	0.6 (0.1, 2.3)	
	187	185	2	1.1 (0.1, 3.8)	
Symptoms, signs & ill- defined conditions; mental disorders	146	144	2	1.4 0.2 , 4.9)	
& congenital abnormalities		<u> </u>			·
None	1642	1632	10	0.6 (0.3, 1.1)	<0.001
None	678	573	5	0.9 (0.3, 2.0)	
	249	241	8	3.2 (1.4, 6.2)	
Disbetes listed as a co-morbidity					
No	2378	2357	21	0.9 (0.6 , 1.3)	0.214
Yes	91	89	2	2.2 (0.3 , 7.7)	ļ
NG tube present pre- LRTI					
No	2376	2360	16	0.7 (0.4 , 1.1)	<0.001
Yes	93	86	7	7.5 (3.1, 14.9	
Tracheostomy or ET tube pre LRTI					-0.001
No	2449	2433	16	0.7 (0.4, 1.1)	<0.001
Yes	20	13	7	35.0 (15.4, 59.2)	
Antibiotics pre LRTI - any route			+	07/02 14	0.467
No	1002	995	1	<u>0.7 (0.3, 1.4)</u>	0.40/
Y65	1457	1451	10	1.1 (0.0, 1.0)	1
No	1228	1220	8	0.7 (0.3, 1.5)	0.228
Yes	1241	1226	15	1.2 (0.7 , 1.9)	
Operation	1				
No	428	426	2	0.5 (0.1 , 1.7)	0.096
Yes	2041	2020	21	1.0 (0.6 , 1.6)	
and the second					

Table 7.16: The incidence of LRTIs in surgical patients

	n	LRT	1	Odds Ratio (95% Cl)	P value
		No	Yes		
Sex	1				
Male	1003	990	13	1.0	0.1231
Female	1466	1456	10	0.5 (0.2 , 1.2)	
Age group					
18-34	514	513	1	1.0	0.0223
35-54	673	669	4	3.1 (0.3 , 37.5)	
55-74	951	940	11	6.0 (0.8 , 46.6)	
75+	331	324	7	11.1 (1.4, 90.5)	
Specialty	1				
General surgery	884	871	13	1.0	0.2919
Orthopaedics	501	497	4	0.5 (0.2, 1.7)	
Urology	472	470	2	0.3 (0.1 , 1.3)	
Gynaecology	386	383	3	0.5 (0.1 , 1.9)	
Obstetrics	226	225	1	0.3 (0.0 , 2.3)	
Admission type	1				
Elective	1629	1616	13	1.0	0.3450
Emergency	840	830	10	1.5 (0.7 , 3.4)	
Co-morbidities					
None	1642	1632	10	1.0	0.0047
One	578	573	5	1.4 (0.5 , 4.2)	
Two or more	249	241	8	5.4 (2.1, 13.9)	
Diabetes listed as a co-morbidity	1				
No	2378	2357	21	1.0	0.2723
Yes	91	89	2	2.5 (0.6 , 10.9)	
NG tube pre LRTI	1 1				
No	2376	2360	16	1.0	<0.001
Yes	93	86	7	12.0 (4.8 , 29.9)	
Endo-tracheal or tracheostomy tube pre LRTI	11				
No	1			1.0	<0.001
Yes	2449	2433	16	81.9 (28.9, 232.1)	
Antibiotics pre LRTI – any route					
No	1002	995	7	1	0.3110
Yes	1467	1451	16	1.6 (0.6, 3.8)	
Antibiotics pre LRTI - intravenous route only	1 1				
No	1228	1220	8	1.0	0.1463
Yes	1241	1226	15	1.9 (0.8 , 4.4)	
Operation	<u> </u>		1		
No	428	426	2	1.0	0.2323
Yes	2041	2020	21	2.2 (0.5, 9.5)	

Table 7.17: The odds of acquiring a LRTI (single variable analysis)

LRTI - lower respiratory tract infection; CI - confidence interval

	n	LRTI		Odds Ratio	P value
	"	No	Yes	(95% CI)	
Sex	1				
Male	1003	990	13	1.0	0.1690
Female	1466	1456	10	0.5 (0.2 , 1.4)	
Age group					
18-34	514	513	1	1.0	0.0813
35-54	673	669	4	3.9 (0.3 , 47.6)	
55-74	951	940	11	5.1 (0.4, 69.3)	
75+	331	324	7	14.3 (1.0, 201.6)	
Specialty	1				
General surgery	884	871	13	1.0	0.4609
Orthopaedics	501	497	4	0.8 (0.2 , 2.9)	
Urology	472	470	2	0.4 (0.1, 2.1)	
Gynaecology	386	383	3	2.6 (0.5, 13.2)	
Obstetrics	226	225	1	2.9 (0.2, 48.1)	
Admission type	1				
Elective	1629	1616	13	1.0	0,5128
Emergency	840	830	10	1.4 (0.5 , 4.4)	
Co-morbidities	1				
None	1642	1632	10	1.0	0.0737
One	578	573	5	0.9 (0.3 , 3.4)	
Two or more	249	241	8	3.7 (1.1 , 12.0)	
Diabetes listed as a comorbidity					
No	2378	2357	21	1.0	
Yes	91	89	2	0.5 (0.1, 3.5)	
NG tube pre LRTI	†				
No	2376	2360	16	1.0	0.8084
Yes	93	86	7	1.3 (0.2 , 8.3)	
Endotracheal or tracheostomy tube					
No				1.0	<0.001
Yes	2449	2433	16	44.4 (6.6 , 288.0)	
Antibiotics pre LRTI- any route					
No	1002	995	7	1.0	0.6549
Yes	1467	1451	16	0.6 (0.1 , 5.4)	
Antibiotics pre LRTI – intravenous route only			· · · · · · · · · · · · · · · · · · ·		
No	1228	1220	8	1.0	0.8358
Yes	1241	1226	15	1.3 (0.1 , 10.9)	
Operation					
No	428	426	2	1.0	0.4022
Yes	2041	2020	21	2.0 (0.4 , 10.8)	

Table 7.18: The odds of acquiring a LRTI (multi-variable analysis)

7.10 Conclusion

In this chapter the results of the analysis that examined the incidence of HAI and how it varied with selected patient characteristics have been presented. Of the 2469 patients included in this analysis 7.5% acquired and presented with an infection during their hospital stay. UTIs were the most frequent type of infections, followed by surgical wound and lower respiratory tract infections. Independent risk factors for these infections varied with site. The following two chapters consider the impact these infections had on resource use and costs incurred by the hospital sector. Chapter 8 presents the results of the analysis that examined the impact these infections had on costs incurred by the hospital sector and Chapter 9 the results of the analysis that examined the impact these infections had on costs incurred by the hospital sector and Chapter 9 the results of the analysis that examined the impact these infections had on costs incurred by the hospital sector and Chapter 9 the results of the analysis that examined the impact these infections had on costs incurred by the hospital sector and Chapter 9 the results of the analysis that examined the impact these infections had on length of hospital stay.

CHAPTER 8

THE IMPACT OF HAIS ON HOSPITAL IN-PATIENT COSTS: RESULTS

8.1 Introduction

The results of the analysis that assessed the impact that hospital acquired infections (HAIs) occurring in surgical patients, had on hospital costs incurred during the patients' hospital stay are presented in this chapter. The results of the single variable analysis that explored how costs varied with HAI status and a range of other patient characteristics are presented first (section 8.2). This is followed by the results of the multivariable analysis and the results of the analysis that looked at the distribution of the costs incurred by infected and uninfected patients (sections 8.3 and 8.4). Further details of the methods used can be found in section 5.5 of Chapter 5.

8.2 **Results of the single variable analysis**

Table 8.1 presents the results of the single variable analysis that assessed how hospital costs varied with selected patient characteristics. The mean, standard deviation, median, 25th and 75th percentiles and minimum and maximum values are presented. The median figure in all cases is lower than the mean costs and the standard deviations, in most cases, are larger than the mean costs, indicating that the hospital cost data are highly skewed in the positive direction.

On average inpatient costs amounted to £1,845 per patient. Costs were found to vary with patient characteristics: women, on average, incurred higher hospital costs than men; hospital costs increased with increasing age group after the age of 35; hospital costs varied with admission specialty, with obstetric patients on average incurring the highest costs; mean hospital costs also varied with diagnosis group; increased with increasing number of co-morbidities; were higher for emergency admissions compared to elective admissions and were higher in infected compared to uninfected patients. Infected patients on average incurred costs that were 2.5 times those incurred by uninfected patients.

The significance of the variation in costs incurred by patients of differing sex, age groups, admission specialty, diagnosis groups, number of co-morbidities and HAI status was assessed using the non-parametric Mann-Whitney and Kruskal-Wallis tests and found to be highly significant in all cases (p=<0.001). Because of the large samples the significance of the observed variation in the mean costs incurred by patients with differing patient characteristics was also tested using the parametric t test. Similar results were obtained. The variation in arithmetic means was found to be highly significant (p=<0.001).

Table 8.1:Hospital costs incurred during the in-patient hospital stay by keypatient characteristics

		The cost of hospital in-patient care (£)						
					Per	centile		
Patient characteristic	n	Mean	Median	SD	25th	75th	Minimum	Maximum
Cav							<u> </u>	
Male	1003	1635	1143	1891	750	1847	247	21269
Female Semale	1466	1035	1773	2074	993	2243	247	39179
	.700	1909	<u> </u>	2014			<u> </u>	
18.34	514	1788	1588	1655	870	2115	247	15937
35 - 54	673	1646	1391	2043	800	1948	255	39179
55-74	951	1824	1379	1871	860	2201	247	21269
75+	331	2398	1800	2636	939	2722	292	21881
Specialty								
General surgery	884	1628	939	2475	671	1534	247	23688
Orthopaedics	501	2422	2157	2403	1262	2711	356	39179
Urology	472	1404	1159	1085	853	1573	292	12901
Gynaecology	386	1732	1815	726	1481	2043	334	5412
Obstetrics	226	2528	2121	1480	1872	2535	922	12072
Admission type								
Elective	1629	1667	1430	1460	750	1847	247	23688
Emergency	840	2191	1510	2748	993	2243	297	39179
Discharge diagnosis group						<u> </u>	L	
Infectious & parasitic diseases	13	872	845	452	514	1114	396	2074
Neoplasms	357	1992	1393	2201	952	2169	276	21158
Endocrine, nutritional & metabolic diseases & immunity disorders	24	1484	1278	764	1155	1484	594	4389
Diseases of blood & blood forming organs	3	992	922	688	342	1713	342	1713
Diseases of the nervous	2	712	712	461	386	1038	386	1039
Diseases of the circulatory system	150	1719	753	3514	597	1065	247	21881
Diseases of the respiratory system	6	5596	2098	6230	1488	12038	1086	15937
Diseases of the digestive system	418	1346	933	1781	670	1373	288	23688
Diseases of the genitourinary system	569	1568	1587	936	963	1938	292	11253
Complications of pregnancy, childbirth & puerperium & certain conditions originating in the perinatal period	254	2405	2104	1473	1857	2392	504	12072
Diseases of the skin & subcutaneous tissue	31	1389	828	2384	611	1173	452	13839
Diseases of the musculoskietal system & connective tissue	309	2215	2216	1184	1268	2681	356	8520
Injury & Poisoning	187	2634	1941	3356	1258	2781	431	39179
Symptoms, signs & ill-defined conditions; mental disorders & congenital abnormalities	146	1479	844	2018	663	1517	299	15731
Co-morbidities							ļ	
None	1642	1699	1407	1763	833	2089	247	39179
One	578	1946	1499	2197	851	2181	288	23688
Two	173	2315	1718	2338	1029	2579	398	15937
Three or more	76	3158	1916	3509	936	3628	254	18987
HAI								-
No	2276	1656	1360	1420	815	2073	247	23688
Yes	193	4081	2430	4745	1824	4014	668	39179

It should be noted the effects of HAI confound the results presented in Table 8.1. As reported in Chapter 7 the incidence of HAI varied with patient characteristics, and as such the variation in costs incurred by patients with differing patient characteristics can in part be attributed to the differing incidence of HAI in the sub-groups examined and the impact that the infections had on costs. The results presented in Table 8.2 show how hospital costs varied with both infection status (infected/uninfected) and selected patient characteristics. The mean costs for both infected and uninfected patients and how these vary with selected patient characteristics are presented, together with the ratios of the costs for infected compared to uninfected patients and the additional costs incurred by infected patients.

The mean costs for infected patients were higher than uninfected patients for all the patient characteristics examined. On average, amongst infected patients, costs were higher in males than females, increased with increasing age, were highest for surgical patients, varied with primary diagnosis group and increased with increasing number of co-morbidities.

Table 8.3 shows how mean costs varied with type of infection. The mean costs varied with site of infection and for all sites, infected patients, on average, incurred higher costs than uninfected patients. The observed variation was found to be significant (p=<0.0001). Of those patients who acquired an infection in hospital, patients who acquired a urinary tract infection (UTI), on average, incurred the lowest costs and patients who acquired more than one infection, on average, incurred the highest hospital costs. The results of the analysis which considered how costs varied with each type of infection and selected characteristics are presented in Appendix 12. The results indicated that whilst, on average, hospital costs were higher in infected than uninfected patients for all the selected characteristics, the pattern of variation varied with type of infection. However, it should be noted that the number of patients in some of the infection groups was small and the results therefore cannot be safely generalized.

	Mean cos	t of hos	pital in-p	atient		Extra in-patient	
Patient characteristic		care	(E)		Ratio of costs	costs/ patient (£)	
	NO H		H/	AJ	(95% Cl)	(95% CI)	
Roy	mean	<u> </u>	(h)	<u>n</u>	(1)	(h. n)	
Mala		040	(0)			2450 (2001 2000)	
Fomale	1939	4220	2742	120	3.4 (2.3, 4.2)	1022 (1570 2285)	
	1010	1330	3/42	130	2.1 (1.0, 2.3)	1932 (1578, 2205)	
18-34	1672	495	2744	20	22/14 20)	2060 (1473 2665)	
25.54	1672	400	2542	29	2.2 (1.4, 3.0)	2003 (14/3, 2003)	
65.74	1636	970	2027	40	2.3 (1.2, 3.5)	2003 (1400, 2007)	
751	1000	204	5597	27		3500 (2772 4408)	
Specialty	1331	204	0001	- 31	2.0 (1.0, 3.0)	5530 (2112, 400)	
General surgery	1338	820	6011	55	45 (23 57)	4673 (4071 5275)	
Orthonaedics	2157	462	5558	30	26 (16 36)	3401 (2672, 4130)	
Urology	1316	445	2846	27	22 (14 29)	1530 1130 1929)	
Gynaecology	1682	336	2073	50	12 (11, 1.4)	391 (178, 604)	
Obstetrics	2508	204	2715	22	1.1 (0.9, 1.2)	207 (-448, 863)	
Admission type	1	1					
Elective	1569	1519	3013	757	1.9 (1.5, 2.3)	1443 (1169, 1717)	
Emergency	1828	110	5496	83	3.0 (2.3. 3.7)	3667 (3095, 4240)	
Discharge diagnosis group		1	1				
Infectious & parasitic diseases	867	12	940	1	1.1 (0.8, 1.4)	73 (-1007, 1154)	
Neoplasms	1714	329	5263	28	3.1 (1.9, 4.2)	3549 (2780, 4318)	
Endocrine, nutritional & metabolic diseases & immunity disorders	1360	20	2102	4	1.5 (0.4, 2.7)	742 (-82, 1566)	
Diseases of blood & blood forming organs	992	3	-	0			
Diseases of the nervous system & sense organs	712	2	-	0			
Diseases of the circulatory system	1134	142	12111	8	10.7 (5.1, 6.2)	10977 (9180, 12774)	
Diseases of the respiratory system	1726	4	13338	2	7.7 (4.1, 11.4)	11612 (7071, 16152)	
Diseases of the digestive system	1249	396	3107	22	2.5 (1.6, 3.3)	1859 (1112, 2605)	
Diseases of the genitourinary system	1501	519	2259	50	1.5 (1.3, 1.7)	758 (493, 1023)	
Complications of pregnancy, childbirth & puerperium & certain conditions originating in the perinatal period	2382	230	2616	24	1.1 (0.9, 1.3)	234 (-389, 856)	
Diseases of the skin & subcutaneous tissue	951	28	5483	3	5.8 (-3.0, 14.5)	4532 (2060, 7005)	
Diseases of the musculoskietal system & connective tissue	2176	296	3106	13	1.4 (1.4, 10.1)	930 (1783, 7282)	
Injury & Poisoning	2136	160	5586	27	2.6 (1.2, 4.0)	3450 (2163, 4736)	
Symptoms, signs & ill- defined conditions; mental disorders & congenital abnormalities	1189	135	5050	11	4.2 (1.6, 6.9)	3861 (2779, 4943)	
Number of co-morbidities							
None	1582	1532	3332	110	2.1 (1.5, 2.7)	1750 (1419, 2081)	
One	1723	528	4304	50	2.5 (1.8, 3.2)	2582 (1979, 3185)	
Two	2012	216	6236	33	3.1 (2.3, 3.9)	4224 (3350, 5097	

Table 8.2: Mean in-patient hospital costs by HAI status and key patient characteristics

		· · · · · · · · · · · · · · · · · · ·						
					Perce	enti le	T	T
Type of HAI	n	Mean	Median	SD	25th	75 th	Minimum	Maximum
No HAI	2276	1656	1360	1420	815	2073	247	23688
UTI only	88	2856	2322	2665	1753	2718	668	19781
SWI only	32	4315	3032	3310	1849	4988	1254	12362
LRTI only	15	3182	2366	2308	1675	3652	714	9762
BSI only	3	8953	8263	6665	2660	•	2660	15937
Skin only	13	3603	2898	1954	2348	5603	953	7247
Other only*	18	3410	2138	3731	1655	3138	782	13839
Multiple	24	9774	5161	9315	2340	16253	916	39179

Table 8.3: Hospital in-patient costs by type of HAI

UTI – urinary tract infection, SWI – surgical wound infection, LRTI – lower respiratory tract infection, BSI – bloodstream infection

*single site infections at sites not classified elsewhere i.e. infections at a sites other than the urinary tract, surgical wounds, lower respiratory tract, skin, or bloodstream

8.3 **Results of the multivariable regression analysis**

This section presents the results of the multivariable regression analysis. For the purposes of this analysis it was assumed that the dependent variable 'hospital costs' had a Gamma distribution. Based on this assumption a generalised linear model of the impact of HAI on hospital costs after controlling for age, sex, specialty, admission type and number of co-morbidities was constructed using a maximum likelihood approach and a) an identity link and b) a log link. Details of the methods employed can be found in section 5.5 of Chapter 5

As detailed in Chapter 5, the identity link assumes additive effects and enables estimates of the mean additional costs incurred by infected patients to be deduced directly from the model, whereas using a log link assumes multiplicative or proportional effects and thus gives estimates of the ratio of costs incurred by infected and uninfected patients. to be taken directly from the model. The results of the generalised linear model that used an identity link are presented in Section 8.3.1 and the results of the model that used a loglink are presented in section 8.3.2.

8.3.1 Results of the generalised linear model that used an identity link

8.3.1.1 Estimates of the cost of HAI

The results of the generalised linear model, which assumed that the dependent variable 'hospital cost' had a Gamma distribution and used a maximum likelihood approach and an identity link, are presented in Table 8.4.

		Coef	9	95% Cl		
			Low	High	1	
Constant		759.02	607.02	911.01		
Sex	Males	REF			0.0055	
	Females	213.21	79.36	347.06	1	
Age Group	18-34	REF			<0.0001	
<u></u>	35 - 54	278.98	128.41	429.56	1	
	55-74	467.86	318.03	617.68	1	
	75+	650.33	424.31	876.34	1	
Specialty	General surgery	REF			<0.0001	
······································	Orthopaedics	741.46	549.64	933.27		
<u> </u>	Urology	17.20	-126.85	161.25	1	
	Gynaecology	336.82	153.37	520.28		
	Obstetrics	1253.73	935.94	1571.51	1	
Type of admission	Elective	REF			0.0118	
	Emergency	179.01	51.48	306.53		
Number of co-morbidities	None	REF			0.0203	
	One	90.02	-47.44	227.48		
	Two or more	351.26	101.80	600.73		
HAI status	No HAI	REF			<0.0001	
<u> </u>	HAI	2254.21	1738.361	2770.05		

Table 8.4: Results of the generalised linear model which assessed the impact of HAI on hospital costs using an identity link

REF - reference category

The results presented in Table 8.4 indicate that on average HAIs cost the hospital sector an additional £2,254 (95% CI: £1,738, £2,770) per case.

8.3.1.2 Testing for interactions

The above model presents a relatively simple model of the impact of HAI on hospital costs. A more detailed model was fitted allowing for the effects of interactions. All two-way interactions between HAI and the other independent variables were assessed. All interaction terms were entered into the model and the least significant term was subsequently removed and the model re-run. This process was repeated until only significant interaction terms remained (p=<0.05). Significant interactions were found between HAI and specialty.

8.3.1.3 Specialty specific estimates of the impact of HAI on hospital costs Specialty specific estimates of the additional costs incurred by infected patients derived from the generalised linear model, which incorporated an interaction term for HAI and specialty, are presented in Table 8.5 together with the mean costs incurred by infected and uninfected patients.

 Table 8.5: Specialty specific estimates of the additional costs incurred by infected patients

	Mean in-patie	nt costs (£)	Mean additional costs (£) (model estimate: 95% Cls)*
	No HAI	HAI	
General surgery	1338	6011	4673 (4368:2992, 5744)
Orthopaedics	2157	5558	3401 (3357:1821, 4894)
Urology	1316	2846	1530 (1474:534, 2414)
Gynaecology	1682	2073	391 (375:-145, 895)
Obstetrics	2508	2715	207 (185:-841, 1211)

*Estimates were derived from the generalised linear model assuming that the outcome variable 'hospital costs' has a Gamma distribution and using a maximum likelihood approach and identity link. The model controlled for the effects of age, sex, admission specialty, admission type, and number of co-morbidities and incorporated an interaction term for HAI and specialty

With the exception of HAIs occurring in gynaecology and obstetric patients the estimated additional costs incurred by infected patients were significantly higher than those incurred by uninfected patients from the same specialty (p=<0.01). Infections occurring in surgical patients on average cost the hospital sector an estimated £4,368 (95% CI: £2,992, £5,744) per case, in orthopaedic patients an estimated £3,357 (95% CI: £1,821, £4,894) per case; in urology patients an estimated £1,474 (95% CI: £534, £2,414) per case; in gynaecology patients an estimated £375 (95% CI: -£145, £895) per case and in obstetric patients an estimated £185 (95% CI: -£841, £1,211) per case.

8.3.1.4 Site specific estimates of the impact of HAI on hospital costs

The preceding tables have presented the results of the analysis that assessed the average costs of HAI. Table 8.6 presents the results of the analysis that assessed the costs of specific types of infection. As before estimates were derived using generalised linear modelling statistical techniques. The model assessed the impact of different types of infection on hospital costs after controlling for age, sex, specialty, admission type and number of co-morbidities. The impact of the following types of infection was assessed: urinary tract, lower respiratory tract, surgical wound, bloodstream, an infection at a site not Since the preceding model classified elsewhere, and multiple infections. identified an interaction between HAI and specialty, the model also included this interaction term. However, the interaction term was not found to be significant in this model and as such was dropped from the analysis. The model identified highly significant variation in costs incurred by patients with different types of infection. However, at the level of each category of infection the significance of the variation in costs incurred compared to uninfected patients varied. Bloodstream and skin infections costs were not found to be significantly different from the costs incurred by uninfected patients, whereas the cost of all other infections were (p=<0.01). However, the number of patients in some of the subgroups were small and thus the study sample is unlikely to have had sufficient power to identify a significant difference in costs in these patients if present.

Table 8.6: Estimates of the additional costs incurred by infected patients by site of infection

Infection status	Mean costs (£)	Mean additional costs incurred by infected patients (£) (model estimate: 95% Cls)			
No HAI	1656				
UTI	2856	1200 (944: 441, 1446)			
LRTI	4315	2659 (2672: 753, 4592)			
SWI	3182	1526 (1497: 548, 2447)			
BSI	8953	7297 (6953: -1652, 15558)			
Skin	3603	1947 (1567: -110, 3245)			
Other single site infection	3410	1754 (1671: 279, 3064)			
Multiple infections	9774	8118 (7930: 4551, 11310)			

UTI - urinary tract infection, LRTI - lower respiratory tract infection, SWI - surgical wound infection,

BSI - bloodstream infection, CI - confidence interval

The results presented in Table 8.6 indicate that UTIs were the least expensive infections, and multiple infections the most expensive. On average UTIs were estimated to cost the health service an additional £944 (95% CI: £441, £1,446) per case, whereas multiple infections were estimated to cost the health service an additional £7,930 (95% CI: £4,551, £11,310) per case.

8.3.2 Results of the generalised linear model that used a loglink

8.3.2.1 Estimates of the ratio of costs incurred by infected and uninfected patients

The results of the generalised linear model which assumed that the dependent variable 'hospital cost' had a Gamma distribution and used a maximum likelihood approach and a log link are presented in Table 8.7.

		Model estimates of the ratio of costs Coef (exp)	9	P value	
			Low	High	
Sex	Males	REF			0.0468
	Females	1.1	1.0	1.2	
Age Group	18-34	REF			<0.0001
	35 - 54	1.2	1.1	1.4	
	55-74	1.4	1.2	1.5	
	75+	1.6	1.4	1.8	
Specialty	General surgery	REF			<0.0001
	Orthopaedics	1.5	1.4	1.7	1
	Urology	1.0	0.9	1.1	
	Gynaecology	1.2	1.1	1.4	
· · · · · · · · · · · · · · · · · · ·	Obstetrics	2.0	1.7	2.3	
Type of admission	Elective	REF			0.002
	Emergency	1.1	1.1	1.2	
Number of co-morbidities	None	REF			
	One	1.1	1.0	1.2	0.006
	Two or more	1.2	1.1	1.4	
HAI status	No HAI	REF			<0.0001
	HAI	2.3	2.0	2.6	

Table 8.7: Results of the generalised linear model which assessed the impact ofHAI on hospital costs using a log link

REF - reference category

The results presented in Table 8.7 indicate that on average hospital costs incurred by infected patients are 2.3 (95% CI: 2.0, 2.6) times that of uninfected patients.

8.3.2.2 Testing for interactions

As with the model that used an identity link, the above model presents a relatively simple model of the impact of HAI on hospital costs. A more detailed model was fitted allowing for the effects of interactions. All two-way interactions between HAI and the other independent variables were assessed. All interaction terms were entered into the model and the least significant term was subsequently removed and the model re-run. This process was repeated until only significant interaction terms remained (p=<0.05). Significant interactions were found between HAI and specialty.

8.3.2.3 Specialty specific estimates of the impact of HAI on hospital costs Specialty specific estimates of the ratio of costs incurred by infected and uninfected patients derived from the generalised model allowing for possible interactions between HAI and admission specialty are presented in Table 8.8.

Table 8.8:	Specialty	specific	estimates	of the	ratio of	costs	incurred by	r infected
	and uninf	ected pa	tients					

	Mean in-petie	ont costs (£)	Ratio of costs incurred by infected and uninfected patients (model estimate: 95% Cls)*
	No HAI	HAI	
General surgery	1338	6011	5.5 (3.9: 3.0,4.9)
Orthopaedics	2157	5558	3.6 (2.4: 1.8, 3.3)
Urology	1316	2845	3.2 (2.1: 1.5, 2.9)
Gynaecology	1682	2073	2.2 (1.2: 0.9, 1.6)
Obstetrics	2508	2715	2.1 (1.1: 0.7,1.6)

*Estimates were derived from the generalised linear model which assessed the impact of HAI on hospital costs allowing for the effects of age, sex, admission type, admission specialty, and number of co-morbidities and allowing for possible interactions between HAI and admission specialty and which used a log link.

The results presented in Table 8.8 show that infected general surgical patients on average incurred the greatest proportional increases in cost. The results of the generalised linear model indicated that, on average, infections occurring in general surgical patients were estimated to increase hospital costs by a factor of 3.9 (95% CI: 3.0, 4.9). The model estimates of the ratio of costs were similar for orthopaedic and urology patients. On average, infected patients incurred costs that were twice those of uninfected patients. In contrast estimates of the impact of infections occurring in gynaecology and obstetric patients on hospital costs indicated that infections occurring in these patients resulted in a slight increase in costs. On average costs were 1.2 (95% CI: 0.9, 1.6) times higher in infected gynaecology patients than uninfected patients from the same specialty and 1.1 (95% CI: 03.7, 1.6) in obstetric patients compared to uninfected patients from the same specialty. However, it should be noted that the confidence intervals for these estimates included zero and as such there was some uncertainty as to whether HAls occurring in these patients increased costs.

8.3.2.4 Site specific estimates of the impact of HAI on hospital costs

The preceding tables have presented the results of the analysis that assessed the ratio of costs incurred by patients with one or more HAIs at any site compared to the costs incurred by uninfected patients. Table 8.9 presents the results of the analysis that assessed the costs of specific types of infection. As before, the generalised linear model assumed a Gamma distribution and used a log link to assess the impact of different types of infection on hospital costs after controlling for age, sex, specialty, admission type and number of co-morbidities. Since the preceding model identified an interaction between HAI and specialty, the model also included this interaction term. However, the interaction term was not found to be highly significant in this model (p = 0.054) and as such was dropped from the analysis. The results presented in Table 8.10 are limited to the estimates of the ratio of costs incurred by patients acquiring specific types of infection compared to uninfected patients derived from the generalised linear model.

Table 8.9:	Estimates of the ratio of costs incurred by infected compared
	to uninfected patients by site of infection

Infection status	Mean costs (£)	Ratio of costs incurred by infected compared to uninfected patients (model estimate: 95% Cls)*
No HAI	1656	
UTI	2856	1.7 (1.6: 1.3, 1.9)
LRTI	4315	2.6 (2.8: 1.8, 4.3)
SWI	3182	1.9 (1.9: 1.4, 2.6)
BSI	8953	5.4 (6.0: 2.2, 16.2)
Skin	3603	2.2 (2.0: 1.2, 3.2)
Other single site infection	3410	2.1 (2.0: 1.3, 2.9)
Multiple infections	9774	5.9 (5.8: 3.9, 7.9)

UTI - urinary tract infection, LRTI - lower respiratory tract infection, SWI - surgical wound infection,

BSI - bloodstream infection, CI - confidence interval

* Estimates were derived from the generalised linear model which assessed the impact of HAI on hospital costs allowing for the effects of age, sex, admission type, admission specialty, and number of co-morbidities and allowing for possible interactions between HAI and admission specialty and which used a log link. The results presented in Table 8.9 show that UTIs were estimated to have the lowest impact on hospital costs. Patients who acquired a UTI were estimated to incur hospital costs that were 1.6 (95% CI: 1.3, 1.9) times those incurred by uninfected patients. Surgical wound infections, skin infections and infections at sites not classified elsewhere were all estimated to increase hospital costs by a factor of about 2, whereas bloodstream infections were estimated to increase hospital costs by a factor of 6.0 and multiple infections by a factor 5.8

8.4 The distribution of hospital costs incurred by infected and uninfected patients

This section presents the results of the analysis that explored how hospital costs and the distribution of these costs differed between infected and uninfected patients. Table 8.10 presents the mean costs for infected and uninfected patients, the additional costs incurred by infected compared to uninfected patients, and the contribution each category of costs makes to the overall additional costs incurred by infected patients. Model estimates of the additional costs incurred by infected patients. Model estimates of the additional costs incurred by infected patients. Model estimates of the additional costs incurred by infected patients allowing for the effects of age, sex, admission specialty, admission type and number of co-morbidities, are also presented in the table. It should be noted that whilst a significant interaction between HAI and admission specialty was identified, for the purposes of this analysis this has been ignored and the average impact of HAI across all specialties on the various cost categories assessed.

aue 0.3.	o lionnainsin a	I IIOS PILAI		aun yu yu ya		IIIIIected partering	
		Mean hospi	tal costs (E)				
		No HAI	HAJ (any site(s))				*
		A7276	n=103	Ratio of costs	Additional cor	sts (£) (model estimate: 95% Cl)	contribution to additional costs
lospital overheads		392.89	855.46	2.18	462.57	(411.94: 284.35, 539.34)	19.07
Directorate management		47.55	101.69	2.14	54.13	(44.34: 29.72, 58.95)	2.23
Capital charges		208.20	439.36	2.11	231.16	(203.66: 139.95, 267.37)	9.53
Medical time		150.72	334.42	2.22	183.69	(148.12: 99.75, 196.48)	7.57
Vursing care		338.11	1319.18	3.90	981.07	(912.36: 645.95, 1178.78)	40.46
Paramedics and specialis	t nurses	11.71	52.42	4.48	40.71	•	1.68
Physiotherapy		16.01	60.06	3.75	44.05		1.82
Consumables used for sp	ecific procedures	11.06	80.77	7.30	69.71	(62.95: 16.35, 109.55)	2.87
Surgical interventions		335.63	437.73	1.30	102.10	(134.16: 81.97, 186.35)	4.21
Antimicrobials		11.80	48.60	4.12	36.80	(27.54: 6.49, 48.58)	1.52
Von-antimicrobial drugs		32.92	119.31	3.62	86.39	(73.37: 13.24, 138.51)	3.56
Microbiology tests		5.74	26.99	4.70	21.24		0.88
Other pathology tests		54.59	110.75	2.03	56.16	•	2.32
Endoscopies		1.71	2.22	1.30	0.51	4	0.02
Radiology		33.35	85.38	2.56	52.02	4	2.15
Other tests		3.53	6.29	1.78	2.76	•	0.11
Total costs		1655.51	4080.60	2.46	2425.09	(2254: 1738, 2770)	100.00
*The distrib	ution of the values	of these com	nonents of	costs were too e	xtreme for star	ndard modelling procedures	

The results present in Table 8.10 show that in all cost categories the costs incurred by infected patients were higher than in uninfected patients. The cost categories that accounted for the majority of the additional costs were those linked to time in hospital. The costs of hospital overheads, directorate management, capital charges and medical time were all assigned to individual patients on the basis of their length of hospital stay. When these categories are taken together it follows that 38.4% of the additional costs were directly linked to time in hospital. This combined category represents the second largest contributor to additional costs, the largest being nursing care accounting for 40.46% of the additional costs incurred. The cost of antimircrobials and microbiology tests were over four times higher in infected than uninfected However, anti-microbial costs only accounted for 1.52% of the patients. additional costs incurred by infected patients and the cost of microbiology tests just 0.88% of the additional costs incurred.

The results of the analysis that examined the distribution of additional costs incurred by infected patients for each type of infection are presented in Appendix 13. For all cost categories, costs incurred by infected patients were higher than uninfected patients and in most cases costs linked to LOS represented a substantial proportion of the additional costs incurred by infected cases. The exception was BSIs, where costs directly linked to time in hospital accounted for just 1.13% of the additional costs incurred. The cost of nursing care represented the greatest proportion of additional costs accounting for 59.98%, and drugs other than anti-microbials the second largest contributor accounting for 13.49%. The cost of drugs other than antibiotics was 30.9 times that of uninfected patients. However, it should be noted that there were only three patients who acquired a BSI and no other infection.

Antibiotic costs varied with site of infection. Overall, patients who acquired an infection in hospital on average incurred antibiotic costs amounting to £48.60 per patient. However, antibiotic costs ranged from a low mean cost of £13.41 per patient with a UTI to a relatively high average cost of £141.06 per case for patients with multiple infections.
Similarly, when looking at the costs of microbiology tests, a cost category which is directory relevant to the detection and management of infections, whereas on average infected patients incurred costs amounting to £26.90 (4.7 times the costs incurred by uninfected patients); the cost varied with site, from a low of £11.92 per patient with a skin infection to a relatively high average cost of £78.25 for patients with multiple infections.

8.5 Conclusion

This chapter has presented the result of the analysis that assessed the impact of HAIs on hospital costs. These results, to be discussed in detail in Chapter 11, clearly demonstrate the substantial burden HAIs place on the hospital sector. It is also clear from the distribution analysis, that whilst mean costs incurred by infected patients are in all categories higher than the mean costs incurred by uninfected patients, a large proportion of the additional costs incurred are linked to a prolonged hospital stay. This impact on length of hospital stay is explored in more detail in the next chapter.

CHAPTER 9

THE IMPACT OF HAIS ON LENGTH OF HOSPITAL STAY: RESULTS

9.1 Introduction

The results of the analysis that assessed the impact of HAIs on length of hospital stay (LOS) are presented in this chapter. Section 9.2 presents the results of the single variable analysis and sections 9.3 - 9.5 the results of the multivariable analysis. Further details of the methods used can be found in section 5.5 of Chapter 5.

9.2 Results of the single variable analysis

Table 9.1 presents the results of the single variable analysis which assessed how LOS varied with key patient characteristics. The mean, standard deviation, median, 25th and 75th percentiles and minimum and maximum values are presented. The median LOS in almost all cases is lower than the mean LOS and the standard deviations in most cases are larger than the mean LOS indicating that the LOS data is highly skewed.

The mean length of hospital stay was 7.4 days. On average women remained in hospital longer than men; LOS increased with increasing age; orthopaedic patients on average remained in hospital longer than patients in the other specialty groups; LOS varied with diagnosis group; was longer in emergency compared to elective admissions; increased with increasing number of comorbidities, and was longer in infected than uninfected patients.

The significance of the variation in LOS by patients of differing sex, age group, admission specialty, admission type, number of co-morbidities and HAI status was first assessed using the non-parametric Mann-Whitney and Kruskal-Wallis tests. The observed variation was significant (p=<0.001). The variation in median LOS by all the factors listed in Table 9.1 was also found to be highly significant (p=<0.001).

253

The significance of the observed variation in the mean LOS was also tested using the parametric t test. Parametric tests are based on the assumption that the data for each group has an approximately normal distribution.²³² In this case it clearly was not; however, the t test is fairly robust to non-normality. The variation in arithmetic means was found to be highly significant (p=<0.001).

Table 9.2 presents the results of the analysis that explored in more detail how LOS stay varied with HAI status. The mean LOS for both infected and uninfected patients are presented and how this varies with selected patient characteristics, together with the ratios of the LOS for infected compared to uninfected patients and the number of additional days infected patients remained in hospital. The mean LOS for infected patients was higher than for uninfected patients for all the patient characteristics examined. On average infected males remained in hospital longer than females, the LOS amongst infected patients increased with increasing age, was highest for infected surgical patients, varied with primary diagnosis group and increased with increasing number of co morbidities.

Table 9.3 shows how the mean LOS varied with type of infection. The mean LOS varied with site of infection and for all sites was greater than uninfected patients. The observed variation was found to be significant (p=<0.0001). Of those patients who acquired an infection in hospital, patients who acquired a UTI, on average, had the shortest length of stay and patients who acquired more than one infection on average had the longest length of stay.

Further analysis examined how for each type of infection the length of stay varied with selected characteristics. The results are presented in Appendix 14. The results showed that whilst the length of hospital stay for all the selected characteristics was higher in infected than uninfected patients the pattern of variation varied with type of infection. However, it should be noted that the number of patients in some of the infection groups was small and as such the results cannot be safely generalized.

				Leng	th of hosp	oital stay (d	ays)	
	n	Meen	Median	SD	Perc	entile	Minimum	Maximum
Patient characteristic					25th	75 th		
Sex								
Male	1003	6.5	4	7.1	2.0	8.0	1	88
Female	1466	7.9	6	9.3	4.0	9.0	1	153
Age group								
18-34	514	6.4	5	8.6	3.0	6.0	1	124
35 - 54	673	5.9	5	5.1	3.0	7.0	1	78
55-74	951	7.6	5	7.6	3.0	9.0	1	88
75+	331	11.2	8	13.7	3.0	13.0	1	153
Specialty								
Surgery	884	6.5	4	9.9	2.0	7.0	1	153
Orthopaedics	501	11.3	9	10.1	5.0	13.0	1	97
Urology	472	5.6	4	6.3	3.0	6.0	1	88
Gynaecology	386	6.4	6	3.7	5.0	8.0	1	34
Obstetrics	226	7.5	6	5.5	5.0	8.0	1	45
Admission type			1					
Elective	1629	6.4	5	7.2	3.0	8.0	1	153
Emergency	840	9.2	6	10.4	4.0	10.0	1	97
Primary discharge diagnosis group		<u> </u>	<u> </u>					
Infectious and parasitic diseases	13	3.8	3	1.6	2.0	5.0	2	6
Neoplasms	357	7.5	5	7.6	3.0	9.0	1	68
Endocrine, nutritional and metabolic	24	5.2	8	24	4.0	6.0	1	13
diseases and immunity disorders		5.2		<u> </u>	4.0			
Diseases of blood and blood forming organs	3	4.3	5	3.1	1.0	6.0	1	7
Diseases of the nervous system and sense organs	2	3.5	3.5	2.1	2.0	5.0	2	5
Diseases of the circulatory system	150	7.7	3	18.6	2.0	5.0	1	153
Diseases of the respiratory system	6	9.2	9.5	4.3	5.3	12.0	3	15
Diseases of the digestive system	418	5.3	3	5.4	2.0	6.0	1	51
Diseases of the genitourinary	569	5.7	6	3.8	3.0	7.0	1	38
Complications of pregnancy, childbirth and puerperium & certain conditions originating in the perinatal period	254	7.4	6	5.6	5.0	7.0	1	45
Diseases of the skin and subcutaneous tissue	31	7.6	4	16.9	2.0	6.0	2	97
Diseases of the musculoskletal system and connective tissue	309	10.7	10	7.7	5.0	13.0	1	52
Injury and Poisoning	187	11.6	8	11.6	5.0	14.0	1	78
Symptoms, signs & III- defined conditions; mental disorders & congenital abnormalities	146	7.2	4	10.5	3.0	7.0	1	88
Co-morbidities								
None	1642	6.6	5	6.4	3.0	8.0	1	97
One	578	8.1	5	11.5	3.0	8.0	1	153
Two	173	10.3	7	11.1	3.5	12.0	1	75
Three or more	76	11.8	8	12.0	4.0	15.0	1	80
HAI								
No	2276	6.6	5	6.6	3.0	8.0	1	124
Yes	193	16.1	10	18.1	7.0	17.0	3	153

Table 9.1: Length of hospital stay by key patient characteristics

CI - confidence interval

	Mean	length of	hospita	stay		No. of additional
		No HAI	HA		Ratio of days	days
Patient characteristic	Mean	n	Mean	n		(95% Cl)
	(a)		(b)		(b/a)	(b-a)
Sex	1	1	<u> </u>		· · · · · · · · · · · · · · · · · · ·	
Male	5.9	946	17.1	57	2.9 (2.2, 3.6)	2.9 (9.5 , 13.0)
Female	7.1	1330	15.6	136	2.2 (1.7, 2.7)	2.2 (6.9, 10.1)
Age group	1	1	1			
18 - 34	6.0	485	12.6	29	2.1 (1.0 ,3.2)	2.1 (3.4, 9.8)
35-54	5.5	627	11.1	46	2.0 (1.4 , 2.6)	2.0 (4.1 , 7.1)
55-74	6.8	870	15.6	81	2.3 (1.8, 2.7)	2.3 (7.1, 10.4)
75+	9.4	294	26.0	37	2.8 (2.0 , 3.5)	2.8 (12.3, 21.0)
Specialty		1				
Surgery	5.4	829	22.8	55	4.2 (3.0, 5.5)	4.2 (14.9 , 19.9)
Orthopaedics	10.3	462	22.9	39	2.2 (1.6, 2.8)	2.2 (9.5 , 15.7)
Urology	5.1	445	14.0	27	2.7 (1.8, 3.7)	2.7 (6.5 , 11.2)
Gynaecology	6.1	336	8.0	50	1.3 (1.1 , 1.5)	1.3 (0.7 , 2.9)
Obstetrics	7.5	204	8.3	22	1.1 (0.9 , 1.3)	1.1 (-1.6 , 3.3)
Admission type						
Elective	6.0	1519	12.4	110	2.1 (1.6 , 2.6)	2.1 (5.1, 7.8)
Emergency	7.9	757	20.9	83	2.7 (2.1, 3.2)	2.7 (10.9 , 15.2)
Primary discharge						
Infectious and parasitic						0.9* 0.9*
diseases	3.8	12	3.0	1	0.8" 0.8"	0.8 -0.0
Neoplasms	6.7	329	17.2	28	2.6 (1.9, 3.2)	2.6 (7.8, 13.3)
Endocrine, nutritional and metabolic diseases and	49	20	70	4	1.4 (0.9.2.0)	1.4 (-0.4 , 4.7)
immunity disorders				·		
Diseases of blood and blood forming organs	4.3	3	0.0	0.0		
Diseases of the nervous	25	2	0.0	0.0		
system and sense organs	3.5	<u> </u>	0.0	0.0		
system	5.0	142	55.9	8	11.2 (3.8, 18.6)	11.2 (40.4 , 51.5)
Diseases of the respiratory	7.0	4	13.5	2	1.9 (1.0 , 2.9)	1.9 (-0.6 , 13.6)
Diseases of the digestive	4.9	396	12.2	22	2.5 (1.8, 3.2)	2.5 (5.1 , 19.5)
Diseases of the						47 (29 50)
genitourinary system	5.4	519	9.3	50	1.7 (1.4 , 2.1)	1.7 (2.9, 0.0)
pregnancy, childbirth and						
puerperium & certain	7.3	230	8.1	24	1.1 (1.0 , 1.2)	1.1 (-5.4 , 7.1)
perinatal period						
Diseases of the skin and	4.5	28	37.0	3	8.2 (6.0., 10.5)	8.2 (29.7, 35.3)
Diseases of the						
musculoskietal system and	10.5	296	16.2	13	1.5 (1.4 , 1.7)	1.5 (1.5 , 10.0)
		100	22.0	~~	00 00/48 00	22 (77 186)
Symptoms, signs and ill.	9.9	100	22.0	27	2.2 2.2 (1.5 , 3.0)	∠. ∠ (1.1, 10.0)
defined conditions; mental	58	135	24 R	44	A3 (1A 71)	43 (131 246)
disorders & congenital abnormalities	0.0		≞ 7. U	11	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	T. (10.1 1 27.0)
Number of co-morbidities						
the second s	i					L

Table 9.2: The mean length of stay by HAI status and key patient characteristics

			Le	ngth of h	ospital st	ay (days	5)	
	N	Mean	Median	SD	Perce	entile	Minimum	Maximum
Type of HAI				· ·	25th	75 ^m	1	
No HAI	2276	6.6	5.0	6.6	3.0	8.0	1.0	124.0
UTI	88	11.5	8.0	10.0	6.0	12.8	3.0	56.0
SWI	32	14.6	11.0	10.8	9.0	17.0	3.0	45.0
LRTI	15	14.5	13.0	9.4	7.25	18.8	4.0	45.0
BSI	3	8.7	9.0	3.5	5.0	12	5.0	12.0
Skin infection	13	18.8	16.0	11.8	8.0	32	4.0	38.0
Other single site infection not classified elsewhere	18	20.0	8.0	27.4	6.0	21.3	4.0	97.0
Multiple infections	24	32.5	20.5	33.7	9.0	34.8	4.0	153.0

Table 9.3: Length of hospital stay by infection status

UTI – urinary tract infection, LRTI – lower respiratory tract infection, SWI – surgical wound infection, BSI – bloodstream infection

9.3 **Results of the multivariable analysis**

This section presents the results of the multivariable analysis. In this model the dependent variable 'LOS' was assumed to have a Gamma distribution and a generalised linear model of the impact of HAI on LOS after controlling for age, sex, specialty, admission type and number of co-morbidities was constructed using a maximum likelihood approach and a) an identity link and b) a log link. Details of the methods employed can be found in section 5.5 of Chapter 5.

As detailed in Chapter 5, and referred to in Chapter 8, the identity link assumes additive effects and consequently enables estimates of the mean additional costs incurred by infected patients to be taken directly from the model. In contrast the log link assumes multiplicative or proportional effects and as such gives estimates of the ratio of costs incurred by infected patients and uninfected patients. The results of the generalised linear model that used an identity link are presented in Section 9.3.1 and the results of the model that used a log link are presented in section 9.2.

9.3.1 Results of the generalised linear model that used an identity link

9.3.1.1 Estimates of the impact of HAI on length of hospital stay

Table 9.4 presents the results of the generalised linear model to assess the impact of HAI on LOS after controlling for the effects of age, sex, admission specialty, admission type and number of co-morbidities. The model assumed that the dependent variable 'hospital LOS' had a Gamma distribution and used a maximum likelihood approach and an identity link.

		Coef	95% Ci		P value
······································			Low	High	
Constant		2.64	1.99	3.29	
Sex	Males	REF			0.0001
······································	Females	1.20	0.622	1.78	
Age Group	18-34	REF			<0.0001
······	35 - 54	0.59	-0.64	1.23	
	55-74	1.72	1.045	2.40	
······································	75+	2.90	1.81	4.00	
Specialty	Surgery	REF			<0.0001
·····	Orthopaedics	4.34	3.37	5.31	
	Urology	0.10	-0.50	0.70	
	Gynaecology	0.98	0.21	1.76	
	Obstetrics	1.97	0.88	3.07	
Type of admission	Elective	REF			<0.0001
	Emergency	1.57	0.99	2.15	
Number of co-morbidities	None	REF			0.0007
	One	0.81	0.19	1.43	
	Two or more	1.64	0.51	2.78	
HAI status	No HAI	REF			<0.0001
	HAI	7.83	5.67	10.00	

Table 9.4: Results of the generalised linear model which assessed the impact ofHAI on length of hospital stay using an identity link

REF - reference category

The results indicate that on average infected patients remain in hospital 7.83 (95% CI: 5.67 to 10.00) days longer than uninfected patients.

9.3.1.2 Testing for interactions

The above model presents a relatively simple model of the impact of HAI on LOS. A more detailed model was fitted allowing for the effects of interactions. All two-way interactions between HAI and the other independent variables were assessed. All interaction terms were entered into the model and the least significant term was subsequently removed and the model re-run. This process was repeated until only significant interaction terms remained (p=<0.05). Significant interactions were found between HAI and specialty (p=<0.001).

9.3.1.3 Specialty specific estimates of the impact of HAI on length of hospital stay

Table 9.5 presents the specialty specific estimates of the average number of additional days infected patients remained in hospital, derived from the generalised linear model that incorporated an interaction term for HAI and specialty, together with the mean LOS of infected and infected patients.

Table 9.5: Specialty specific estimates of the impact of HAI on length of hospital stay

	Mean LOS	(days)	Number of ext	ra days in (model	fected p estimat	atients remained in hospital te 95% Cls)*
	No HAI	HAI				
Surgery	5.4	22.8	17.4	(15.26:	9.49,	21.03)
Orthopaedics	10.3	22.9	12.6	(11.73:	4.70,	18.75)
Urology	5.1	14	8.9	(7.88:	2.85,	12.93)
Gynaecology	6.1	8	1.9	(1.79:	-0.43,	4.01)
Obstetrics	7.5	8.3	0.8	(0.60	-2.87,	4.07)

LOS = length of stay

*The model estimates were derived from a generalised linear model that used an identity link and controlled for age, sex, admission specialty, admission type, number of co-morbidities, and included an interaction term for HAI and specialty The results presented in Tables 9.5 indicate that HAIs occurring in surgical patients had the greatest impact on LOS in terms of the number of extra days patients were estimated to remain in hospital as a result of acquiring and infection. On average HAIs occurring in surgical patients were estimated to prolong the in-patient stay by an additional 15.3 (95% CI: 9.5, 21.0) days. In contrast infections occurring in gynaecology patients were estimated to increase the patients LOS by just 1.8 days (95% CI: -.04, 4.0) days and in obstetric patients the presence of one or more HAIs was estimated to increase the LOS by just over half a day (0.6 days, 95% CI: -.3, 4.1). Estimates of the impact HAIs occurring in the other specialty groups had on LOS varied. HAIs occurring in orthopaedic patients were, on average, estimated to extend the in-patient stay by 11.7 (95% CI: 4.7, 18.8) days; and HAIs occurring in urology patients were estimated to extend the hospital stay by 7.9 (95% CI: 2.9, 12.9) days.

9.3.1.4 Site specific estimates of the impact of HAI on length of hospital stay The preceding tables have presented the results of the analysis that assessed the average costs of HAI. Table 9.6 presents the results of the analysis that assessed the costs of specific types of infection. As before the generalised linear model assumed a Gamma distribution and used a maximum likelihood approach and identity link. The model assessed the impact of different types of infection on hospital LOS after controlling for age, sex, specialty, admission type and number of co-morbidities. The impact of the following types of infection were assessed: urinary tract, lower respiratory tract, surgical wound, bloodstream, infections at a site not classified elsewhere, and multiple infections. Since the preceding model identified an interaction between HAI and specialty, the model also included this interaction term. However, the interaction term was not found to be significant in this model and was dropped from the analysis. The results of this analysis are given in Table 9.6, together with the mean costs incurred by infected and uninfected patients by infection group.

Type of HAI	n	Mean	Number of additional days (model estimate: 95% Cl)*
No HAI	2276	6.6	
UTI only	88	11.5	4.9 (3.27: 1.25, 5.28)
SWI only	32	14.6	8.0 (7.18: 2.34, 12.02)
BSI only	3	8.7	2.1 (2.21: -7.08, 11.49)
LRTI only	15	14.5	7.9 (8.01: 0.62,15.37)
Skin only	13	18.8	12.2 (9.85: 0.37, 19.33)
Other only	18	20.0	13.4 (11.59: 2.97, 20.21)
Multiple	24	32.5	25.9 (24.01: 11.84, 36.18)

Table 9.6: Estimates of the impact of specific types of HAI on length of hospital stay

CI - confidence interval

*Model estimates obtained from generalised linear modelling that controlled for the effects of age, sex, admission specialty, admission type, number of co-morbidities, and included a variable denoting type of infection.

The results presented in Table 9.6 indicate that on average multiple infections have the highest impact on LOS. Patients who acquired more than one infection in hospital were estimated to remain in hospital an estimated 24 (95% CI: 11.9, 36.2) extra days. In contrast BSIs were estimated to increase the inpatient stay by just 2.2 (95% CI: -7.1, 15.37) days. However, it should be noted that there were just three patients who acquired a BSI and no other infections, two of whom died in hospital. As evidenced by the CIs there was considerable uncertainty about this finding, and as such the results cannot be safely generalised. UTIs had the second lowest impact on LOS. On average UTIs were estimated to prolong the patients stay by 3.3 (95% CI: 1.3, 5.3) days. SWIs, LRTIs, skin infections and infections at other sites were estimated to increase varying with site.

9.3.2 Results of the generalised linear model that used a loglink

9.3.2.1 Estimates of the ratio of LOS incurred by infected and uninfected patients

The results of the generalised linear model which assumed that the dependent variable 'hospital cost' had a Gamma distribution and used a maximum likelihood approach and a log link are presented in Table 9.7.

	Model estimates	9	5% Ci	P value
	of the ratio of			
	LOS			
	Coef (exp)			
		Low	High	
Males	REF			0.0008
Females	1.2	1.1	1.3	
18-34	REF			<0.0001
35 - 54	1.1	1.1	1.3	
55-74	1.4	1.2	1.5	
75+	1.6	1.4	1.9	
General surgery	REF			<0.0001
Orthopaedics	1.7	1.6	1.9	
Urology	1.0	0.9	1.1	
Gynaecology	1.2	1.0	1.3	
Obstetrics	1.4	1.2	1.6	
Elective	REF	- <u> </u>		<0.0001
Emergency	1.3	1.2	1.4	
None	REF			
One	1.2	1.0	1.3	0.0001
Two or more	1.3	1.1	1.5	
No HAI	REF			<0.0001
HAI	2.1	1.8	2.5	
	Males Females 18-34 35 - 54 55-74 75+ General surgery Orthopaedics Urology Gynaecology Obstetrics Elective Emergency None One Two or more No HAI HAI	Model estimates of the ratio of LOS Coef (exp)MalesREFFemales1.218-34REF35 - 541.155-741.475+1.6General surgeryREFOrthopaedics1.7Urology1.0Gynaecology1.2Obstetrics1.4ElectiveREFEmergency1.3NoneREFOne1.2Two or more1.3No HAIREFHAI2.1	Model estimates of the ratio of LOS Coef (exp)9MalesREFFemales1.218-34REF35 - 541.118-34REF35 - 541.11.11.275+1.61.61.4General surgeryREFOrthopaedics1.71.60.9Gynaecology1.21.00.9Gynaecology1.21.41.2ElectiveREFEmergency1.31.31.2NoneREFOne1.21.0Two or more1.31.1No HAIREFHAI2.11.8	Model estimates of the ratio of LOS Coef (exp) 95% Cl Males REF

Table 9.7: Results of the generalised linear model which assessed the impact ofHAI on length of hospital stay using a log link

REF - reference category

The results presented in Table 9.7 indicate that on average infected patients have a hospital LOS that is 2.1 (95% CI: 1.8, 2.5) times that of uninfected patients.

9.3.2.2 Testing for interactions

As with the model that used an identity link, the above model presents a relatively simple model of the impact of HAI on LOS. A more detailed model was fitted allowing for the effects of interactions. All two-way interactions between HAI and the other independent variables were assessed. All interaction terms were entered into the model and the least significant term was subsequently removed and the model re-run. This process was repeated until only significant terms remained (p=<0.05). Significant interactions were found between HAI and specialty.

9.3.2.3 Specialty specific estimates of the impact of HAI on LOS

Specialty specific estimates of the ratio of the LOS incurred by infected and uninfected patients derived from the generalised model allowing for possible interactions between HAI and admission specialty are presented in Table 9.8.

	Mean LOS	S (days)	(mod	Ratio of lel estimate	days e: 95%	Cis)*
	No HAI	HAI	·			
Surgery	5.4	22.8	4.22	(3.4:	2.6,	4.4)
Orthopaedics	10.3	22.9	2.22	(2.0:	1.4,	2.7)
Urology	5.1	14	2.75	(2.5:	1.7,	3.6)
Gynaecology	6.1	8	1.31	(1.3:	1.0,	1.7)
Obstetrics	7.5	8.3	1.11	(1.1:	0.7,	1.6)

 Table 9.8: Specialty specific estimates of the ratio of costs

 incurred by infected and uninfected patients

CI - confidence interval

*Estimates were derived from the generalised linear model which assessed the impact of HAI on LOS allowing for the effects of age, sex, admission type, admission specialty, and number of co-morbidities and allowing for possible interactions between HAI and admission specialty and which used a log link.

The results presented in Table 9.8 show that infected general surgical patients on average incurred the greatest proportional increases in LOS. The results of the generalised linear model indicated that, on average, infections occurring in general surgical patients were estimated to increase the patients LOS by a factor of 3.4 (95% CI: 2.6, 4.4). Infections occurring in urology patients were estimated to result in an average LOS that was 2.5 times that of uninfected patients from the same specialty, and infections occurring in orthopaedic patients were estimated to result in an average LOS that was twice as long as uninfected patients from the same specialty. HAIs occurring in gynaecology and obstetric patients were not estimated to have a substantial impact on LOS. Infections occurring in gynaecology patients were estimated to increase the LOS by a factor of 1.3, and infections occurring in obstetric patients were estimated to have a LOS that was just 1.1 times that of uninfected patients from the same specialty.

9.3.2.4 Site specific estimates of the impact of HAI on LOS

The preceding tables have presented the results of the analysis that assessed the average increase in LOS incurred by patients who acquired one or more Table 9.9 presents the results of the analysis that infections in hospital. assessed the impact of specific types of infection on LOS. As before the generalised linear model assumed a Gamma distribution and used a log link to assess the impact of different types of infection on hospital LOS after controlling The for age, sex, specialty, admission type and number of co-morbidities. impact of the following types of infection were assessed: urinary tract, lower respiratory tract, surgical wound, bloodstream, an infection at a site not Since the preceding model classified elsewhere, and multiple infections. identified an interaction between HAI and specialty, the model also included this interaction term. However, the interaction term was not found to be significant in this model (p = 0.138) and as such was dropped from the analysis.

Type of HAI	n	Mean	Ratio of LOS (model estimate: 95% Cl)*
No HAI	2276	6.6	
UTI only	88	11.5	1.7(1.5:1.13, .1.9)
SWI only	32	14.6	2.2(2.0: 1.5, 2.9)
BSI only	3	8.7	1.3(1.5: 0.5, 4.4)
LRTI only	15	14.5	2.2(2.3: 1.4, 3.8)
Skin only	13	18.8	2.8(2.5: 1.5, 4.2)
Other only	18	20.0	3.3(2.7: 1.7, 4.2)
Multipie	24	32.5	4.9(4.4: 2.9, 6.5)

Table 9.9: Estimates of the ratio of LOS incurred by infected patients compared to uninfected patients by site of infection

UTI - urinary tract infection, LRTI - lower respiratory tract infection, SWI - surgical wound infection,

BSI - bloodstream infection, CI - confidence interval

* Estimates were derived from the generalised linear model which assessed the impact of HAI on hospital LOS allowing for the effects of age, sex, admission type, admission specialty, and number of co-morbidities and allowing for possible interactions between HAI and admission specialty and which used a log link.

The results presented in Table 9.9 show that BSIs were estimated to have the lowest impact on LOS. Acquisition of a BSI was estimated, on average, to increase the LOS by a factor of 0.5. However, as noted in section 9.3.1 there were just three patients in this infection group, two of whom died as in-patients thus curtailing their length of hospital stay. The acquisition of a UTI was also estimated to increase the LOS by a factor of 0.5. SWIs, LRTIs, skin infections and infection at sites not classified elsewhere were estimated to result in a LOS that on average was twice that of uninfected patients, whereas multiple infections were estimated to result in a LOS that on average was 4.4 times that of uninfected patients.

9.4 Conclusion

In this chapter the results of the analysis that explored the impact of HAI on length of hospital stay were presented. Whilst the results of this analysis will be discussed in detail in Chapter 11, it is clear from the preceding sections that HAIs have a substantial impact on hospital length of stay. The following chapter will consider how the estimates of both the impact of HAI on LOS and also the impact on hospital cost can be used in models to assess the benefits of investment in prevention.

CHAPTER 10 THE POTENTIAL BENEFITS OF INVESTMENT IN INFECTION PREVENTION AND CONTROL

10.1 Introduction

The results of this study clearly demonstrate that hospital acquired infections (HAI) place a substantial burden on health care resources. This chapter will explore how the results of this analysis, and indeed estimates taken from other studies, can be used to assess the magnitude and distribution of the potential benefits of investment in infection control. Estimates of the gross benefits of investment in prevention and control activities and a framework for assessment of net benefits are presented.

10.2 The potential benefits of prevention

The estimates of the economic burden of HAIs derived in this study and presented in Chapters 8 and 9, may be considered to represent the gross benefits of prevention, as measured by costs avoided if an infection is prevented. The net benefits will depend on the cost and effectiveness of infection control practices.

10.3 The gross benefits of prevention

As indicated above the results of this analysis represent the gross benefits of prevention, as measured by the average costs that may be avoided if an infection is prevented. The magnitude of these gross benefits within a particular clinical setting will depend on both the number and types of infections prevented.

Estimates of the proportion of HAIs that are preventable vary. As mentioned in Chapter 1, perhaps the most widely quoted estimate of the proportion of infections that could be avoided through improvements in infection prevention and control is that derived as part of the SENIC study conducted in the US in the mid 1970s and early 1980s. The study estimated that 30% of infections could be prevented.¹¹ It is not known how applicable this estimate is to the situation in NHS hospitals in England. The results of a recent National Audit Office survey of infection control teams indicated that many of the teams felt that this was an overestimate of the proportion that were preventable. Infection control teams were asked to estimate the proportion of HAIs they considered preventable. The estimates varied, ranging from a low estimate of 5% to over 35%. The bed weighted average was 15%.³ Whilst it is acknowledged that this estimate is a subjective assessment of the proportion of HAIs that can be prevented, for the purpose of exploring the potential benefits of prevential gross benefits of prevention at the level of the study hospital and also at the national level.

10.3.1 Methods used to derive estimates of the potential gross benefits of a 15% reduction in infection rates

Estimates of the potential gross benefits of a 15% reduction in rates were derived from estimates of the number of patients acquiring one or more infections and the burden these infections imposed on the hospital sector. These latter estimates were derived from data on the observed incidence of HAIs presenting during the in-patient period; the estimated ratio of the hospital costs and LOS incurred by infected compared to uninfected patients obtained from the generalised linear modelling analysis; the mean hospital costs and mean LOS incurred by uninfected patients; and data on the number of adult admissions to the specialties covered in this study at a) the study hospital in 1994/5 and b) NHS hospitals throughout England in 1994/5.

Estimates of the number of patients acquiring one or more infections were derived by applying the observed incidence of HAI to the number of patients admitted to the specialties covered in this study at a) the study hospital in 1994/5 and b) NHS hospitals throughout England in 1994/5.

Estimates of the burden these infections imposed were derived as follows. If N is the number of patients admitted to the specialties covered in this study at the study hospital in 1994/5 (or NHS hospitals throughout England in 1994/5), C the baseline cost of treating uninfected patients, *i* the estimated incidence and *r* the estimated ratio of costs incurred by infected compared to uninfected patients, NiC(r-1) provides an estimate of the national burden.

The Ninety-five per cent confidence intervals were derived as follows. estimates of the incidence of HAI and the ratio of costs incurred by infected compared to uninfected patients used to determine the burden of HAI were those derived from this study. The true population values are unknown. The sampling error of these estimates was measured by their standard errors sei (standard error - incidence) and ser (standard error - ratios of costs (LOS)). Using these values, the variance of the estimated burden of HAI was estimated as follows: $N^2C^2i^2ser^2+(r-1)^2ser^2$. This estimated variance was then used to obtain 95% confidence intervals for the estimates of the annual burden of HAIs, on the assumption that the sampling error in such an estimate would be approximately Normal. The square root of the estimated variance was derived and this figure multiplied by 1.96 and the resultant figure added to the estimate of the cost (and number of additional days) to derive an estimate of the upper confidence limit, and subtracted from this estimate to derive an estimate of the lower limit.

Having derived estimates of the number of patients who acquired one or more infections and estimates of the burden these infections imposed both in terms of costs and the number of additional days spent in hospital, estimates of the potential gross benefits of a 15% reduction in rates were derived.

The same methodology was subsequently used to derive site and specialty specific estimates of the burden of HAI and the potential benefits of a 15% reduction in rates, utilising site and specialty specific estimates of the incidence of HAI and the ratio of costs and LOS as appropriate.

10.3.2 The potential gross benefits of a 15% reduction in infection rates at the level of the study hospital

Tables 10.1 to 10.3 present the results of the analysis that estimated the potential benefits of a 15% reduction in HAI rates at the level of the study hospital. The estimates are limited to infections occurring in adult, non-day case patients admitted to the surgical specialties covered in this study. Table 10.1 provides details of the number of patients estimated to have acquired an infection in 1994/5. Table 10.2 presents the results of the analysis that estimated the cost of these infections and the gross benefits of a 15% reduction in rates, and Table 10.3 estimates of the number of bed days utilised as a result of infection and the number that would be released if infection rates had been 15% lower.

In 1994/5 (the year in which the study was conducted) an estimated 579 (95% CI: 500, 665) adult patients admitted to the general surgical, orthopaedic, urology, gynaecology and obstetric specialties acquired one or more HAIs. These infections were estimated to have cost the hospital sector, an additional \pounds 1,224,044 (95% CI: \pounds 898,408, \pounds 1,549,680) and affected patients were estimated to have utilised an estimated 4,308 bed days (95% CIs: 3,059, 5,558). If the infections rates had been 15% lower in 1994/5 then the crude estimates presented in Tables 10.2 and 10.3 suggest that resources valued at \pounds 183,607 (95% CI: \pounds 134,761, \pounds 232,452) may have been available for alternative use, and the same reduction in rates might have resulted in the release of 646 bed days (95% CI: 459, 834 bed days).

The estimated cost of HAI, and the estimated benefits of a 15% reduction in rates varied considerably with site. Whilst multiple infections were estimated to impose the greatest burden, if infections occurring at just one site are considered, the results presented in Tables 10.2 and 10.3 show that UTIs, which had the lowest cost per case, were the most expensive type of infection.

Specialty specific estimates suggested that relative to the other surgical specialties the general surgical and urology specialties incurred the greatest costs from HAI, and the benefits of a 15% reduction in rates potentially could be substantial.

It should be stressed that the estimates presented in Table 10.2 represent the gross benefits of prevention: that is they represent the level of resources that might be released for alternative use. They do not represent the value that affected individuals and society as a whole place on prevention.

Estimates of the number of adult non day-case patients admitted to five surgical specialties at the study hospital in 1994/5 who acquired an infection by site of infection and specialty **Table 10.1:**

		Mailen of UAL MVV			Е
Specialties	Types of HAI	observed in study (95% CI)	No of patients at risk	Estimated no. of patients with an HA which presented during the in-patient period (95% CI)	· · · ·
Al listed specialties*	AI HAIs	7.5(6.4 , 8.6)	7763	579 (500, 665)	T
All listed specialties*	5	3.5(2.8 , 4.3)	7763	274 (220,336)	
	Chest	0.6(0.4 , 1.1)	7763	50 (29,82)	
	SWI	1.1 (0.8 , 1.6)	7763	88 (59 , 127)	
	BSI	0.1 (0.0 , 0.3)	7763	6 (0,23)	T
	Skin	0.5(0.3 , 0.9)	7763	41 (22 , 70)	
	Other	0.6(0.3 , 1.0)	7763	47 (24,78)	
	Multiple	0.9(0.6 , 1.4)	7763	108 (46 , 72)	· · · · ·
General surgerv	All HAIs	61(46 75)	1000	103 (130 22E)	
Orthopaedics	AIHAIS	7.4(5.2 , 10.4)	1672	124 (87, 174)	
Urology	AI HAIs	5.1 (3.9 , 8.4)	1067	54 (42,90)	
Gynaecology	AI HAIS	12.7 (9.9 , 16.8)	1702	216 (168, 286)	
Obstetrics***	AI HAIS	9.3(6.2 , 14.4)	328	31 (20.47)	
 General surgery, orthopae 	sdics, urology, gy	maecology, obstetrics (caes	arean sections only)		7

** Number of adult non day-case patients admitted to the specialties listed at the study hospital in 1994/5

Estimates of the cost of HAIs occurring in adult, non day-case patients admitted to five surgical specialties at the study hospital in 1994/5 and the potential benefits of a 15% reduction in rates Table 10.2

All listed All listed 1224044 (898406, 1549660) specialties* All listed 1224044 (898406, 130683) All listed UTI 1656 1.6 (1.3, 1.9) 274 220, 336) 258927 (126961, 30083) All listed UTI 1656 1.6 (1.3, 1.9) 274 220, 336) 258927 (126961, 30083) Specialties* UTI 1656 1.9 (1.4, 2.5) 8 (2.2, 1.6) 1.17595 (5312, 2.36879) Swin 1656 2.0 (1.2, 3.2) 8 (0, 23) 5.2217 (13163, 117599) Skin 1656 2.0 (1.2, 3.2) 41 (2.7, 70) 65663 (667, 13163, 117599) Multiple 1656 2.0 (1.3, 2.9) 47 (2.4, 78) 75103 (1680, 138525) Multiple 1656 5.6 (1.3, 2.9) 47 (2.4, 78) 75103 (1680, 138525) Multiple 1656 5.6 (1.3, 2.9) 10.2 10.2 10.2 <th>Soecialities</th> <th>Types of HA</th> <th>Mean costs unimfected patients</th> <th>Mod Infection Infection</th> <th>f costi f costi ited co ffecter (95%</th> <th>mate of the s incurred by impared to 1 patients</th> <th>Estimate infecter (95</th> <th>d number of d patients % CI)</th> <th>Estimated</th> <th>costs to ector (£) 95% Ci)</th> <th>hospital</th> <th>Estimate c released if r low (95</th> <th>of resourc ates were er (£) % CI)</th> <th>es 15%</th>	Soecialities	Types of HA	Mean costs unimfected patients	Mod Infection Infection	f costi f costi ited co ffecter (95%	mate of the s incurred by impared to 1 patients	Estimate infecter (95	d number of d patients % CI)	Estimated	costs to ector (£) 95% Ci)	hospital	Estimate c released if r low (95	of resourc ates were er (£) % CI)	es 15%
All listed UT 1656 1.5 (1.3, 1.9) 274 220, 336) 258927 (126961, 390833) Specialties* UT 1656 1.6 (1.3, 1.9) 274 220, 336) 258927 (126961, 390833) Specialties* Chest 1656 1.9 (1.4, 2.6) 88 (59, 127) 131176 (524, 209227) SM 1656 1.9 (1.4, 2.6) 88 (59, 127) 131176 (524, 20927) SM 1656 2.0 (1.2, 3.2) 41 (22, 70) 65963 (6667, 12558) Stin 1656 2.0 (1.3, 2.9) 41 (24, 76) 75103 (11680, 138525) Other 1656 2.0 (1.3, 2.9) 41 (24, 76) 75103 (11680, 138525) Multiple 1656 2.6 (3.9, 7.9) 108 (46, 72) 820424 (538577, 1102320) Multiple 1536 3.3 3.0, 4.9) 108 (46, 72) 820424 (538577, 1102320)	All listed specialties*	AI HAIs	1656	2.3	(2.0,	2.6)	579	(500 , 665)	1224044	(898408	, 1549680)	183607 (134761 . 2	32452)
All listed UIT 1656 1.6 (1.3, 1.9) 274 (220, 336) 258927 (126961, 390893) specialtice* UIT 1656 1.8 (1.3, 1.9) 50 (29, 82) 147595 (58312, 236879) specialtice* SWI 1656 1.9 (1.4, 2.5) 88 (59, 127) 131176 (5244, 209827) SWI 1656 6.0 (2.2, 16.2) 88 (59, 127) 131176 (5244, 209827) Stin 1656 1.9 (1.4, 2.5) 88 (59, 127) 131176 (52424, 209827) Stin 1656 2.0 (1.2, 3.2) 41 (22, 70) 65667, 125589 Stin 1656 5.6 (1.3, 2.9) 41 (22, 70) 65667, 125558) Multiple 1656 5.6 (1.2, 3.2) 41 (22, 70) 65667, 125558) Stin 1666 16.3 41 (22, 70) 65663 (6667, 125558) Multiple 1656 5.6 (3.9, 79) 108														
Cheat 1656 2.8 (1.8, 4.3) 5.0 (29, 82) 147595 (58312, 236879) SWI 1656 1.9 (1.4, 2.5) 88 (59, 127) 131176 (52424, 209827) SWI 1656 1.9 (1.4, 2.5) 88 (59, 127) 131176 (52424, 209827) SWI 1656 2.0 (1.2, 3.2) 81 (59, 127) 6667, 125539 Other 1656 2.0 (1.2, 3.2) 41 (22, 70) 65963 (6667, 125539) Other 1656 2.0 (1.3, 2.9) 47 (24, 76) 75103 (11680, 138525) Multiple 1656 2.0 (1.3, 2.9) 47 (24, 77) 820424 (538527, 1102320) General surgery All HAls 133 3.9 (3.0, 4.9) 108 (46, 72) 820424 (538527, 1102320) Orthopaedics All HAls 133 3.3 (3.0, 4.9) 183 (31017, 962070) Orthopaedics All HAls 133 2.16 <td>All listed specialties*</td> <td>5</td> <td>1656</td> <td>1.6</td> <td>(1.3</td> <td>1.9)</td> <td>274</td> <td>(220 , 336)</td> <td>258927</td> <td>(126961</td> <td>, 390893)</td> <td>38839 (</td> <td>19044,58</td> <td>3634)</td>	All listed specialties*	5	1656	1.6	(1.3	1.9)	274	(220 , 336)	258927	(126961	, 390893)	38839 (19044,58	3634)
SWI 1556 1.9 (1.4 2.6) 88 (59, 127) 131176 (52424, 209927) BSI 1656 6.0 (2.2, 16.2) 6.0, 23) 52217 (-13163, 117599) BSI 1656 2.0 (1.2, 3.2) 41 (22, 70) 65963 6667, 125259) Chher 1656 2.0 (1.3, 2.9) 47 (24, 78) 75103 (11680, 138525) Multiple 1656 5.6 (3.9, 7.9) 108 46, 72) 820424 (538527, 1102320) General surgery All Hals 1338 3.9 (3.0, 4.9) 108 46, 72) 820424 (538527, 1102320) General surgery All Hals 1338 3.3 (3.0, 4.9) 124 (87, 174) 820424 (538527, 1102320) Unology All Hals 1338 3.3 (1.6 128, 72) 820424 (538527, 1102320) Unology All Hals 1338 138, 225) 696543 (41017, 962070) Unology All Hals		Chest	1656	2.8	(1.8	4.3)	50	(29 , 82)	147595	(58312 ,	236879)	22139 (8747,355	i32)
BSI 1656 6.0 (22, 16.2) 6.0 (23) 52217 (-13163, 117599) Stin 1656 2.0 (12, 3.2) 41 (22, 70) 65963 (6657, 125259) Other 1656 2.0 (13, 2.9) 47 (24, 78) 75103 (11680, 138525) Multiple 1656 5.6 (13, 2.9) 47 (24, 78) 75103 (11680, 138525) Multiple 1656 5.6 (3.9, 7.9) 108 (46, 72) 820424 (538527, 1102320) General surgery All Hals 1338 3.9 (3.0, 4.9) 183 (138, 225) 696543 (431017, 962070) Urology All Hals 2157 2.4 (18, 3.3) 124 (87, 174) 385745 (185714, 585777) Urology All Hals 2157 2.4 (15, 2.9) 5.4 (42, 90) 73375 (13266, 125659) Orthopactics All Hals 1682 166 168, 286) 73375 (2305, 182656) 1065659) Orthopactics All Hals 1682 161 161 210		SWI	1656	1,9	1.4	2.6)	88	(59 , 127)	131176	(52424,	209927)	19676 (7864, 314	189)
Skin 1656 2.0 (1.2 3.2 41 (22 70 65963 (6667 125258) Other 1656 2.0 (1.3 2.9) 47 (24, 78) 75103 (11680, 138525) Multiple 1655 5.6 (3.9, 7.9) 108 (46, 72) 820424 (538527, 1102320) Multiple 1656 5.6 (3.9, 7.9) 108 (46, 72) 820424 (538527, 1102320) General surgery All HAls 1338 3.9 (3.0, 4.9) 183 (138, 225) 696543 (431017, 962070) Orthopaedics All HAls 2157 2.4 (18, 3.3) 124 (87, 174) 385745 (185714, 585777) Urology All HAls 2157 2.4 (18, 3.3) 124 (87, 174) 385745 (185714, 585777) Urology All HAls 1316 2.15 54 (42, 90) 78438 (31216, 125659) Orthopaedics All HAls 168 16 54 (42		BSI	1656	6.0	(2.2 ,	16.2)	9	(0 , 23)	52217	(-13163	117598)	7833 (-1975, 176	940)
Other 1656 2.0 (1.3, 2.9) 47 (24, 76) 75103 (1680, 138525) Multiple 1656 5.5 (3.9, 7.9) 108 (46, 72) 820424 (538527, 1102320) General surgery All Hols 1338 3.9 (3.0, 4.9) 168 (46, 72) 820424 (538527, 1102320) General surgery All Hols 1338 3.9 (3.0, 4.9) 183 (138, 225) 696543 (43017, 962070) Orthopaedics All Hols 2157 2.4 (1.8, 3.3) 124 (87, 174) 385745 (185714, 585777) Urology All Hols 2157 2.4 (1.8, 3.3) 124 (87, 174) 385745 (185714, 585777) Urology All Hols 1316 2.1 (1.5, 2.9) 54 (42, 90) 78438 (1216, 125659) Outnotaction All Hols 1682 12 (0.9, 1.6) 216 (168, 286) 79375 (23065, 182656)		Skin	1656	2.0	(1.2 ,	3.2)	41	(22 , 70)	65963	(6667 ,	125258)) 1686	1000, 187	(89)
Multiple 1656 5.5 (3.9, 7.9) 108 (46, 72) 820424 (538527, 1102320) General surgery All Hals 1338 3.9 (3.0, 4.9) 183 (138, 225) 696543 (431017, 962070) Orthopaedics All Hals 2157 2.4 (1.8, 3.3) 12.4 (87, 174) 385745 (185714, 585777) Urology All Hals 2157 2.4 (1.5, 2.9) 5.4 (42, 90) 78438 (31216, 125659) Urology All Hals 1316 2.1 (1.5, 2.9) 5.4 (42, 90) 78438 (31216, 125659) Cynaecology All Hals 1682 1.2 (0.9, 1.5) 216 (168, 286) 79375 (23905, 182659)		Other	1656	2.0	(1.3,	2.9)	47	(24 , 78)	75103	(11680,	138525)	11265 (1752 , 207	(6/1
General surgery All Hols 1338 3.9 (3.0, 4.9) 183 (138, 225) 696543 (431017, 962070) Orthopaedics All Hols 2157 2.4 (1.8, 3.3) 12.4 (87, 174) 385745 (185714, 585777) Urology All Hols 2157 2.4 (1.8, 3.3) 12.4 (87, 174) 385745 (185714, 585777) Urology All Hols 1316 2.1 (1.5, 2.9) 54 (42, 90) 78438 (31216, 125659) Gynaecology All Hols 1682 1.2 (0.9, 1.5) 216 (168, 286) 79375 (-23005, 182656)		Multiple infections	1656	5.6	(3.9,	(6.7	108	(46 , 72)	820424	(538527	, 1102320)	123064 (80779 , 16	55348)
General surgery Ail HAIs 1338 3.9 (3.0, 4.9) 163 (138, 225) 696543 (431017, 962070) Orthopaedics Ail HAIs 2157 2.4 (1.8, 3.3) 124 (87, 174) 385745 (185714, 585777) Urology Ail HAIs 1316 2.1 (1.5, 2.9) 54 (42, 90) 78438 (31216, 125659) Gynaecology Ail HAIs 1682 1.2 (0.9, 1.6) 216 (168, 286) 79375 (-23905, 182656) Constructione Ail HAIs 1682 1.2 (0.9, 1.6) 216 (168, 286) 79375 (-23905, 182656)														
Orthopaedics All HAIs 2157 2.4 (1.8, 3.3) 124 (87, 174) 385745 (185714, 585777) Urology All HAIs 1316 2.1 (1.5, 2.9) 54 (42, 90) 78438 (31216, 125659) Gynaecology All HAIs 1682 1.2 (0.9, 1.6) 216 (168, 286) 79375 (-23905, 182656)	General surgery	All HAIs	1338	3.9	(3.0 ,	4.9)	183	(138, 225)	696543	(431017	, 962070)	104482 (64652,1	(111)
Urology All HAIs 1316 2.1 (1.5, 2.9) 54 (42, 90) 78438 (31216, 125659) Gynaecology All HAIs 1682 1.2 (0.9, 1.6) 216 (168, 286) 79375 (-23905, 182656) All HAIs 2508 1.1 (0.7 1.6) 3.1 (20 4.7) 4.713 (-21338 30763)	Orthopaedics	AI HAIS	2157	2.4	(1.8,	3.3)	124	(87,174)	385745	(185714	,585777)	57862 ((27857,8	7867)
Cynaecology Al HAIs 1682 1.2 (0.9, 1.6) 216 (168, 286) 79375 (-23905, 182656)	Urology	AI HAIS	1316	2.1	(1.5 ,	2.9)	25	(42,90)	78438	(31216,	125659)	11766 ((4682 , 18	849)
AN HAL 2508 11 (07 16) 31 (20 47) 4713 (21338 30763)	Gynaecology	AI HAIS	1682	1.2	(0.9	1.6)	216	(168,286)	79375	(-23905	, 182656)	11906	(-3586,27	(398)
	Obstetrics	AI HAIS	2508	1.1	(0.7 .	1.6)	31	(20,47)	4713	(-21338	, 30763)	707	-3201 .	1614)

2 5 AA' AN

Ceneral surgery, or unoparous,
 Figures taken from Table 10.1
 *Caesarean sections only

272

Table 10.3: Estimates of the number of additional days, adult non-day case patients admitted to five surgical specialties at the study hospital in 1994/5 and who acquired an HAI, remained in hospital and the number of bed days released if rates were 15% lower

Specialties	Types of HAI	Mean LOS uninfected patients	Model estimate of the ratio of costs incurred by infected compared to uninfected patients (95% CI)	Estimated number of infected patients** (95% CI)	Estimate of the number of additional days in hospital (95% Cl)	Estimates of the number of days released if rates were reduced by 15% (95% CI)
All listed specialties*	AII HAIS	6.6	2.1 (1.8 , 2.5)	579 (500 , 665)	4308 (3059 , 5558)	646 (459,834)
All listed snacialities*	E		16/13 100			
	Chest	9.9	2.3 (14, 3.8)	50 (29 82)	504 (420 , 1340) 426 (78 775)	64 (12 116)
	SWI	6.6	2.0 (1.5, 2.9)	88 (59 . 127)	606 (207.1004)	91 (31, 151)
	BSI	6.6	1.5 (0.5 , 4.4)	6 (0 , 23)	19 (-26 , 65)	3 (-4.10)
	Skin	6.6	2.5 (1.5 , 4.2)	41 (22 , 70)	401 (67 , 735)	60 (10, 110)
	Other	6.6	2.7 (1.7 , 4.2)	47 (24 . 78)	516 (119.912)	77 (18, 137)
	Muttiple infections	6.6	4.4 (2.9 , 6.5)	108 (46 . 72)	2402 (1506 . 3298)	360 (226, 495)
General surgery	AII HAIs	5.4	3.4 (2.5 , 4.4)	183 (138, 225)	2317 (1348, 3286)	348 (202,493)
Orthopaedics	AI HAIS	10.3	2.0 (1.4 , 2.7)	124 (87, 174)	1790 (1018, 2562)	268 (153, 384)
Urology	AII HAIS	5.1	2.5 (1.7 , 3.6)	54 (42,90)	937 (698 , 1176)	141 (105,176)
Gynaecology	AI HAIS	6.1	1.3 (1.0 , 1.7)	216 (168 , 286)	100 (-341,541)	15 (-51, 81)
Obstetrics**	AI HAIS	7.5	1.1 (0.7 , 1.7	31 (20 ,47)	115 (30, 201)	17 (4.30)
General surgery, ortho	poedics, urbloov	V. ovnaecology.	obstetrics (caesarean sections only)			

Figures taken from Table 10.1 *Caesarean sections only

10.3.3 National estimates of the potential gross benefits of a 15% reduction in infection rates

Tables 10.4 – 10.6 present site and specialty specific estimates of the potential benefits of a 15% reduction in infection rates occurring in adult non-day case patients admitted to the surgical specialties, covered in this study at NHS hospitals in England in 1994/5. Overall an estimated 154,920 (95% CI: 134,024, 177,943) adult non-day case patients admitted to the specialties covered in this study at other NHS hospitals in England acquired one or more HAIs in 1994/5, at a cost to the health sector of £327.78 million (95%CI: £240.58, £414.98) (in-patient costs only). Overall an estimated 1,153,726 (95% CI: 819,019, 1,488,434) additional bed days were utilised as a result of patients acquiring one or more HAIs. The magnitude of the burden imposed varied with site and specialty.

Estimates of the gross benefits of a reduction in rates indicated that a 15% reduction in rate would result in resources valued at £49.17 million (95% CI: £36.09, £62.25 million) released for alternative use. The same level of reduction in the rate would have released an estimated 173,059 bed days (95% CI: 122,853, 223,265) for alternative use. Estimates the magnitude of the level of resources released varied with site and with specialty, as indeed they did for the study hospital.

Estimates of the number of adult non day case patients admitted to five surgical specialties at NHS hospitals in England who acquired an HAI IN 1994/5 by site of infection and specialty **Table 10.4:**

Specialties	Types of HA	Incidence of HAI (%) observed in study (95% CI)	No of patients at risk**	Estimated no. of patients with an HA which presented during the in-patient period (95% Cl)
Al listed specialties*	AI HAIs	7.5 (6.4 , 8.6)	2078793	154920 (134024 , 177943)
All listed specialties*	5	3.5 (2.8, 4.3)	2078793	73250 (58867 , 89981)
	Chest	0.6 (0.4 , 1.1)	2078793	13471 (7708,21831)
	SWI	1.1 (0.8, 1.6)	2078793	23576 (15693 , 33986)
	BSI	0.1 (0.0, 0.3)	2078793	1684 (0, 6076)
	Skin	0.5 (0.3 , 0.9)	2078793	10945 (5833, 18682)
	Other	0.6 (0.3 , 1.0)	2078793	12629 (6320 , 20790)
	Multiple	0.9 (0.6 , 1.4)	2078793	19366 (12294 , 28989)
General surgery	AI HAIS	6.1 (4.6 , 7.5)	804676	49085 (37015, 60351)
Orthopaedics	AII HAIS	7.4 (5.2 , 10.4)	464948	34406 (24177, 48355)
Urology	All HAIs	5.1 (3.9, 8.4)	223685	11408 (8724, 18790)
Gynaecology	AI HAIS	12.7 (9.9 , 16.8)	501395	63677 (49638, 84234)
Obstetrics**	All HAIS	9.3 (6.2 , 14.4)	84089	7820 (5214 , 12109)
* Coneral elect	new withman	fice under masching obstatrics (carear	rean sections only)	

Centeral surgery, or mopaeures, u energy, gynaecouogy, unservice, researce excurse oury)
**Number of adult, non-day case patients admitted to the specialties listed in the table in 1994/5. Data source: Hospital episode statistics data set 1994/5²³¹
***Cassarean soctions only

Estimates of the cost of HAIs occurring in adult, non day-case patients admitted to five surgical specialties at NHS hospitals in England in 1994/5 and the potential benefits of a 15% reduction in rates **Table 10.5:**

				Contraction of infection		Estimate of resources
	Types of	costs	model esumate of use ratio of costs incurred	csurrace number of meeter patients**	Estimated costs to hospital sector £ million	released if rates were 15% lower
speciaines	infection	ummeu ed ratients	uninfected patients (95% CI)	(95% CI)	(95% CI)	£ million (95% Cl)
All listed specialties*	AI HAIS	1656	2.3 (2.0, 2.6)	154920 (134024, 177943)	327.78 (240.58 , 414.98)	49.17 (36.09, 62.25)
All listed specialties*	5	1656	1.6 (1.3, 1.9)	73250 (58867, 89981)	69.34 (34.00, 104.67)	10.40 (5.10, 15.70)
	Chest	1656	2.8 (1.8, 4.3)	13471 (7708, 21831)	39.52 (15.61 , 63.43)	5.93 (2.34, 9.51)
	SWI	1656	1.9 (1.4 , 2.6)	23576 (15693 , 33986)	35.13 (14.04, 56.21)	5.27 (2.11, 8.43)
	BSI	1656	6.0 (2.2, 16.2)	1684 (0, 6076)	13.98 (-3.52 , 31.49)	2.10 (-0.53 , 4.72)
	Skin	1656	2.0 (1.2, 3.2)	10945 (5833, 18682)	17.66 (1.79, 33.54)	2.65 (0.27 , 5.03)
	Other	1656	2.0 (1.3, 2.9)	12629 (6320 , 20790)	20.11 (3.13, 37.09)	3.02 (0.47 , 5.56)
	Multiple infections	1656	5.6 (3.9 , 7.9)	19366 (12294 , 28989)	146.77 (71.28 , 222.25)	22.02 (10.69 , 33.34)
General surgery	AIHAIS	1338	3.9 (3.0, 4.9)	49085 (37015,60351)	187.21 (115.84 , 258.57)	28.08 (17.38, 38.79)
Orthonaedics	AI HAIs	2157	2.4 (1.8, 3.3)	34406 (24177, 48355)	107.27 (51.64 , 162.89)	16.09 (7.75, 24.43)
Urolocy	AIHAIs	1316	2.1 (1.5, 2.9)	11408 (8724, 18790)	16.44 (6.54 , 26.34)	2.47 (0.98, 3.95)
Gunaecology	AI HAIS	1682	1.2 (0.9 , 1.6)	63677 (49638, 84234)	23.38 (-7.04, 53.81)	3.51 (-1.06, 8.07)
Obstetrics***	AI HAIS	2508	1.1 (0.7 , 1.6)	7820 (5214, 12109)	1.21 (-5.47 , 7.89)	0.18 (-0.82, 1.18)
 General surgery, orthogonal 	hopaedics, urolog	y, gynaecology	, obstetrics (caesarean sectic	visional or the second s		

**Figures taken from Table 10.4

Estimates of the number of additional days adult, non day case patients, admitted to selected specialties at NHS hospitals in England 1994/5 and who acquired an HAI remained in hospital and the number of bed days released if rates were 15% lower **Table 10.6:**

		Mean LOS	Model estimates of the ratio of LOS	Estimated number infected	Estimated number of	Estimated number of days released if rates were
Specialties	Types of HA	uninfect ed patients	infected and uninfected patients	patients** (95% Cf)	additional days (95% Cl)	reduced by 15% (95% Cl)
All listed energalities.	AI HAIs	6.6	2.1 (1.8, 2.5)	154920 (134024, 13261)	1153726 (819019 , 1488434)	173059 (122853 , 223265)
All listed specialties*	5	6.6	1.54 (1.3 , 1.9)	73250 (58867 , 89981)	263457 (112505 , 414409)	39519 (16876, 62161)
	Chest	9 [.] 9	2.28 (1.4, 3.8)	13471 (7708, 21831)	114149 (20846 , 207452)	17122 (3127, 31118)
	IMS	6.6	2.04 (1.5, 2.9)	23576 (15693, 33986)	162203 (55564, 268843)	24330 (8335, 40326)
	BSI	6.6	1.47 (0.5 , 4.4)	1684 (0, 6076)	5198 (-6884 , 17281)	780 (-1033, 2592)
	Skin	6.6	2.49 (1.5 , 4.2)	10945 (5833, 18682)	107291 (17856,196726)	16094 (2678, 29509)
	Other	9.9	2.66 (1.7 , 4.2)	12629 (6320,20790)	138123 (31912, 244333)	20718 (4787, 36650)
	Muttiple infections	6.6	4.36 (2.9, 6.5)	19366 (12294, 28989)	429706 (189782, 669631)	64456 (28467 , 100445)
Ganaral eirnew	AIHAIS	5.4	3.35 (2.6 , 4.4)	49085 (37015, 60351)	622795 (362403, 883187)	93419 (54360,132478)
Othmaedics	AI HAIS	10.3	1.95 (1.4 , 2.7)	34406 (24177, 48355)	337204 (122425 , 551983)	50581 (18364, 82797)
1 Irohaw	AI HAIs	5.1	2.48 (1.7, 3.6)	11408 (8724, 18790)	86385 (85809,86961)	12958 (12871, 13044)
Gunaecolony	AI HAIS	6.1	1.30 (1.0, 1.7)	63677 (49638, 84234)	116961 (-13072, 246994)	17544 (-1961, 37049)
Obstatrics***	AI HAIS	7.5	1.07 (0.7 , 1.7)	7821 (5214, 12109)	4177 (-17804, 26159)	627 (-2671, 3924)
 General surgery, orthopaedics, urok 	ogy, gynaecology,	obstetrics (cae	sarean sections only)			

Figures taken from Table 10.4 *Caesarean sections only LOS – length of stay

10.4 The net benefits of investment in infection control

The preceding sections presented estimates of the gross benefits of investment in infection control that may result if a 15% reduction in rates were achieved. No attempt was made to take into account the costs of prevention and control activities and their actual effectiveness. The above examples simply highlight the magnitude of the level of resources that might be released if a reduction at the 15% level was achieved. This section explores how the results of this study may be incorporated into simple economic models to demonstrate the net benefits of investment in infection control practices.

Models of the net benefits of investment in prevention activities can be derived from information on infection rates;

the cost of the infection control practice to be evaluated and its

• efficacy (assuming 100% compliance);

the cost of alternative strategies which would need to be introduced to maximise compliance and the level of compliance they are expected to achieve; and the magnitude, nature and distribution of the economic burdens that may have resulted had the HAIs not been prevented.

Data on infection rates can be obtained from a variety of sources including hospital records, the literature and national surveillance schemes. Data on the cost of selected infection control practices and the cost of strategies to enhance compliance with a given practice can generally be relatively easily estimated, and data on the cost of the burden that may have resulted had the HAIs not been prevented can be obtained from studies such as this. However, data on the efficacy of interventions and strategies that aim to enhance compliance and hence effectiveness are a little more difficult to obtain. Whilst a number of studies have demonstrated the effectiveness of surveillance and feedback of results to the appropriate personnel,¹⁶⁴ ²³³ with the exception of studies that have assessed the efficacy of specific antibiotics there is a marked lack of information on the efficacy of specific infection control activities. Lack of information on the effectiveness of infection control practices was highlighted by

Thames Valley University, when drafting infection control guidelines for the prevention of HAIs.²³⁴ When drafting these guidelines prevention activities were categorised according to the quality of the supporting evidence.

Category 1 included activities where there were generally consistent findings from a range of evidence derived from well designed experimental studies. Category 2 included activities for which evidence of effectiveness was based on a single acceptable study, or a weak or inconsistent finding in multiple acceptable studies. Category 3 included those activities for which there was limited scientific evidence that did not meet all the criteria of 'acceptable studies', or an absence of directly applicable studies of good quality. This included published or unpublished expert opinion. The majority of the activities fell within category three.²³⁵

In the absence of rigorous data on the efficacy of a particular intervention models can be developed which assess the potential benefits of prevention assuming different levels of effectiveness. Information on the break-even point given different levels of effectiveness can subsequently be derived from these models. That is the point at which the potential benefits cover the cost of the intervention and any further reduction in rates result in net benefits. This information can subsequently be used to inform decision marking regarding investment in the activity assessed.

The following presents a worked example of how this could be achieved. The example looks at the problem of catheter related UTIs; it represents a reworking of a study by Plowman *et al* ²¹⁵ but incorporates data from the work presented here. A copy of the paper by Plowman et al²¹⁵ can be found at the back of this thesis.

279

10.4.1 An economic model to assess the cost and benefits of the routine use of silver alloy coated urinary catheters to reduce the risk of urinary tract infections in catheterised patients

As reported in Chapter 2, prevalence studies have generally found urinary tract infections (UTIs) to be the most common type of hospital acquired infection accounting for between 21 and 45% of all HAIs identified.⁷ ¹⁹ ²⁰ ²² Studies which have assessed the incidence of urinary tract infections in patients admitted to both medical and surgical specialties suggest that between 1 and 3% acquire a urinary tract infection,⁴⁷ ⁸¹ ⁸² whilst a study that was limited to the incidence of infections occurring in surgical, urology, gynaecology and orthopaedic patients who had an operative procedure found that 6.3% acquired a UTI.¹⁴ In this study the overall incidence of UTIs in patients admitted to the general surgical, orthopaedic, urology, gynaecology and, if the patient had a caesarean section, the obstetric specialties was 4.1% (95% CI: 3.3, 4.9).

These infections may result in additional morbidity⁸² ¹²⁵ ¹³¹ ¹⁴⁴ ¹⁴⁵ and in some cases mortality,¹¹² ¹²² a prolonged hospital stay and additional costs incurred by the hospital sector as a result of additional in-patient care.¹⁴ ⁶⁹ ¹⁷² ²⁰⁷ ²³⁶ The results of this study indicated that on average UTIs cost the hospital sector an additional £944 (95% CI: £441, £1446) per case and prolonged the LOS by 3.27 (95% CI: 1.25. 5.28) days. Further costs are borne by the secondary and primary health care sector following discharge from hospital, and by patients and their carers.

A key risk factor for these infections is the presence of a urinary catheter,¹⁴ ¹³⁴ ¹³⁷ ¹³⁸ ¹⁴³ with an estimated 80% of hospital acquired UTIs being associated with the presence of this device.²³⁷ In an endeavour to reduce the infection risk associated with urinary catheters, silver alloy coated catheters have been developed. A number of trials have found these catheters to reduce the infection risk.²³⁸⁻²⁴⁴ The silver alloy coating prevents the adherence of microbes to the catheter wall and this reduces the risk of infections being established. However, whilst these trials indicate that silver alloy coated catheters appear to

be effective at reducing the patients risk of acquiring a UTI, the level of effectiveness has been found to vary, this perhaps reflecting variations in case mix and the underlying infection rate. A meta-analysis of eight clinical trials found that silver alloy coated catheters, when compared to non-coated catheters, had a preventive effect, and this effect was over and above other coatings such as silver oxide coatings,²⁴⁰ that have been removed from the market in the US. One trial demonstrated that up to 48% of hospital acquired UTIs may be prevented through the use of silver alloy coated urinary catheters.²⁴¹

Whilst studies have demonstrated the effectiveness of these catheters their routine use has cost implications. Silver alloy coated catheters are more expensive that the catheters routinely used in hospitals in England and Wales: in 1994/5 silver–alloy coated catheters cost an additional £9 each.²⁴⁵ As such before routine use can be advocated the potential costs and benefits associated with their routine use in a particular clinical setting needs to be assessed.

The following sections present a model for assessing the potential costs and benefits of the routine use of silver alloy coated catheters. This is followed by an illustrative model of the potential costs and benefits of their routine use in patients admitted to the specialties covered in this study at the study hospital in 1994/5.

10.4.2 Aims and objectives of the economic model

The model aims to assess the following:

- i. the number of hospital acquired UTIs occurring in catheterised patients admitted to the specialties of interest at one or more hospital.
- ii. the economic burden these infections impose on the hospital sector as a result of additional in-patient care
- iii. the number of extra days infected patients remain in hospital.

281

iv. the potential benefits of the routine use of silver alloy coated catheters estimated as the value of resources and bed days released for alternative use.

10.4.3 The structure of the economic model

Figure 10.1 illustrates the structure of the model. The starting point is the number of admissions to the specialties of interest (cell 1). These patients are then sub-divided into the specialty groups of interest (cells 2-6). Figure 10.1 shows 5 surgical specialties, however this could be adapted to reflect the needs of the user. The patient groups of interest are then further subdivided into those who are catheterised and those not catheterised (cells 7-16). The estimated number of hospital acquired UTIs that might occur in these catheterised patients are then determined based on the specialty specific incidence of UTIs (cells 17-26). Estimates of the economic impact these infections place on the hospital sector are then derived by multiplying the number of UTIs by an appropriate estimate of the additional cost of these infections (cells 27-31) and the number of extra days patients remain in hospital (cells 32-36).

Specialty specific estimates of the additional costs associated with the routine use of silver alloy coated catheters, as compared to non-coated catheters, are then derived by multiplying the additional cost of the catheter by the number of patients catheterised (cells 7, 9,11, 13, 15). The potential gross benefits (value of resources and the number of days released for alternative use) are then derived for varying levels of effectiveness (0%-100%), and the net benefits derived by subtracting the cost of the intervention from the estimated gross benefits. Finally, the costs of the intervention can be plotted against the benefits that might accrue assuming different levels of effectiveness and the cut off point where potential benefits are equal to the costs of the intervention identified. Any further reduction in rates would result in net benefits.

Sensitivity analysis can subsequently be conducted to assess the impact that varying the incidence of UTIs and the cost per case has on the results obtained.

Figure 10.1: The structure of an economic model to assess the costs and benefits of an intervention to prevent catheter associated urinary tract infections



283

In order to operationalise this model the following information is required:

- i. Number of admissions to the specialties of interest at one or more hospitals.
- ii. Of those patients selected, the number (or proportion) of patients within the particular group of interest. For example, the number of patients admitted to specific specialties. In Figure 10.1 the patients have been subdivided into five surgical specialties.
- iii. The number (or proportion) of patients in the selected groups of interest who are catheterised.
- iv. The number or estimated incidence of hospital acquired UTIs occurring in these catheterised patients.
- v. Estimates of the cost of hospital acquired UTIs to the hospital sector.
- vi. Estimates of the average number of additional days catheterised patients with a hospital acquired UTI remain in hospital.
- vii. The cost and estimated effectiveness of the intervention.

The model produces five main outputs:

- i. The number of catheter related UTIs that might occur in a pre-defined patient group.
- ii. The value of hospital resources used by catheterised patients who acquire a UTI

- iii. The number of additional bed days utilised as a result of patients acquiring a UTI.
- iv. The net financial benefits associated with the routine use of sliver alloy coated catheters assuming different levels of effectiveness.
- v. The number of bed days released for alternative use assuming different levels of effectiveness

10.4.4 Illustrative model

A model to assess the costs associated with catheter associated hospital acquired UTIs occurring in adult (\geq 18 years of age) non-day case admissions to five surgical specialties at the study hospitals in 1994/5 (Figure 10.2), and the potential benefits of introducing the routine use of silver alloy coated urinary catheters (Table 10.7) was developed. The model utilises data obtained from the study hospital together with the results of this work. Details of the data sources used and the output of this model are given below.

Number of admissions (cell 1)

Data on the number of adult (\geq 18 years of age), non-day case admissions to the surgical specialties of the study hospital were obtained from the study hospital's database. In 1994/5 there were 7763 admissions that met this criteria.

Number of patients admitted to the different surgical specialties (cells 2-6)

Data on the number of patients admitted to the five surgical specialties included in this model were obtained from the study hospital's database. In 1994/5 2994 (38.6%) of the 7763 adult non-day case admissions to the surgical specialties listed were general surgical admissions, 1672 (21.5%) were orthopaedic admissions, 1067 (13.7%) urology admissions, 1702 (21.9%) gynaecology admissions and 328 (4.2%) obstetric admissions who underwent a caesarean section.

The number of patients catheterised (cells 7-16)

Estimates of the number of patients catheterised were derived by applying estimates of the proportion of patients catheterised at some point during their admission to the number of admissions. Data on catheterisation rates were not routinely collected at the study hospital and as such the model utilises data obtained in this study. Of those patients recruited into this study 17.6% of surgical patients; 6.9% of orthopaedic patients; 47.0% of urology patients; 19.5% of gynaecology patients and 9.0% of obstetric patients were catheterised at some point during their admission to hospital.

Number (or percentage) of catheterised patients who acquire a UTI (cells 17-26)

Estimates of the number of patients who acquired a UTI were derived using specialty specific estimates of the incidence of UTIs in catheterised patients derived from this study. The incidence of UTIs in catheterised general surgical patients was 7.2% (95% CI: 3.3, 13.2), in orthopaedic patients 6.1% (95% CI: 1.2, 16.9), in urology patients 4.5% (95% CI: 2.5, 7.3), in gynaecology patients 18.0% (95% CI: 12.0, 25.4) and in obstetric patients 6.3 (95% CI: 1.7, 15.2).

The value of resources used as a result of additional in-patient care

Estimates of the value of resources used as a result of additional in-patient care were derived by multiplying the estimated cost per case derived in this study by the number of infections. Estimates derived from the generalised linear modelling analysis indicated that, on average, UTIs cost the hospital sector an additional £944 (95% CI: £441, £1,446) per case as a result of additional in-patient care.

The number of additional days catheterised patients with a UTI remain in hospital

Estimates of the number of additional days catheterised patients with a UTI remained in hospital were derived by applying the model estimates of the impact of UTIs on LOS derived in this study, to the number of infections estimated to have occurred. Estimates derived from the generalised linear modelling analysis indicated that, on average, patients who acquired a UTI remained in hospital an additional 3.3 days (95% CI: 1.3, 5.3).

Estimating the potential benefits of the routine use of silver alloy coated catheters

Having identified the number of patients who acquired a UTI, and the estimated impact these infections have on hospital sector resource use, the final stage was to identify the potential savings that might result from the routine use of silver alloy coated catheters. In order to estimate these benefits information on the cost of the intervention and its estimated level of effectiveness is required. As indicated above, in 1994/5 the additional cost of silver alloy coated catheters compared to non-coated catheters was $\pounds 9.^{245}$ Estimates of the cost of this intervention can therefore be derived by applying this figure to the number of catheterised patients within each specialty (cells 7,9,11,13 and 15 in Figure 10.2). Table 10.7 presents specialty specific estimates of the costs that would be incurred if silver alloy coated catheters were in routine use.

Estimates of the potential gross and net benefits of prevention were subsequently derived for varying levels of effectiveness ranging from 0%-100% (Table 10.7). Finally the costs and potential benefit were plotted against each other (Figures 10.3 - 10.7).
Figure 10.2: An economic model of the costs and benefits of the routine use of silver alloy coated catheters in adult non-day case patients admitted to the surgical specialties of one hospital in England in 1994/5



Estimated cost per case Estimated number of additional days per case

			Leve	l of effect	iveness (of cathete		wanting	Ē		
	*-	10%	20%	30%	40%	50%	en k	704	BUK	aner	1004
General surgery							2	2	2	22	2001
Gross benefits (£)	358	3582	7163	10745	14326	17908	21489	25071	28652	32224	35815
Additional cost of silver alloy coated catheters(£)	4742	4742	4742	4742	4742	4742	4742	4742	4742	4740	CATA
Net benefits(£)	4384	-1161	2421	6002	9584	13165	16747	20328	23010	27.401	24072
No. bed days released	-	13	25	88	20	8	75	88		4 4 2	100
Orthopaedics								3		2	C3
Gross benefits (£)	99	664	1329	1993	2657	3322	3986	4650	5315	5079	6643
Additional cost of silver alloy coated catheters(E)	1038	1038	1038	1038	1038	1038	1038	1038	1038	1038	1038
Net benefits(£)	-972	-374	290	955	1619	2283	2948	3612	4276	4941	5605
No. bed days released	0	2	S.	~	6	12	14	16	19	2	23
Urology									2		2
Gross benefits (£)	213	2130	4261	6391	8521	10652	12782	14912	17043	10173	21202
Additional cost of silver alloy coated catheters(£)	4513	4513	4513	4513	4513	4513	4513	4513	4513	4512	4512
Net benefits(E)	4300	-2383	-253	1878	4008	6138	8269	10300	12520	14660	16700
No. bed days released	-	F	15	22	g	37	45	5	9		24 L
Gynaecology									8	5	
Gross benefits (£)	564	5639	11279	16918	22558	28197	33837	30476	45116	50755	56205
Additional cost of silver alloy coated catheters(£)	2987	2987	2987	2987	2987	2987	2987	2987	2987	2087	2087
Net benefits(£)	-2423	2652	8292	13931	19571	25210	30850	36489	42129	47768	52408
No. bed days released	N	20	39	59	62	66	118	138	158	4 4	101
Obstetrics									3		10
Gross benefits (£)	18	176	351	527	702	878	1053	1229	1404	1580	1756
Additional cost of silver alloy coated catheters(£)	266	266	266	266	266	266	266	266	266	266	266
Net benefits(E)	-248	06-	85	261	437	612	788	963 1	1139	1314	1490
No. bed days released	0	4	-	8	N	ल	*	4	CJ I	9	9

Figure 10.3: Estimated costs and benefits of the routine use of silver alloy coated catheters in adult, non-day case patients admitted to the general surgical specialty of an NHS hospitals in England



Figure 10.4: Estimated costs and benefits of the routine use of silver alloy coated catheters in adult, non-day case patients admitted to the orthopaedic specialty of an NHS hospitals in England







Figure 10.6: Estimated costs and benefits of the routine use of silver alloy coated catheters in adult, non-day case patients admitted to the gynaecology specialty of an NHS hospitals in England





10.4.5 Results of the illustrative model

Effectiveness Proportion of UTIs prevented (%)

If the assumptions detailed above are accepted, the model suggests that 129 surgical patients acquired a catheter related UTI in 1994/5 (Figure 10.2). General surgical patients were estimated to cost the hospital sector an additional £35,815, orthopaedic patients an additional £6,643, urology patients £21,303, gynaecology patients £56,395, and obstetric patients who underwent a caesarean section an additional £1,756. In terms of the number of additional days patients remained in hospital surgical patients were estimated to remain in hospital an additional 125 days, orthopaedic patients 197 days, and obstetric patients who underwent a caesarean section 6 days.

The additional costs of the routine use of silver alloy coated catheters was balanced against these costs assuming different levels of effectiveness at preventing UTIs (Table 10.7). If the routine use of silver alloy coated catheters were adopted at an additional cost of £9 per catheter then, in order for the benefits to outweigh the costs, 13.2% of infections occurring in general surgical patients would need to be prevented to cover the costs, 15.6% in orthopaedic patients, 21.2% in urology patients, 5.3% in gynaecology patients and 14.4% in obstetric patients who underwent a caesarean section (Figures 10.3 - 10.7).

The results of one clinical trial suggest that up to 48% of hospital acquired UTIs can be prevented.²⁴¹ Clearly at this level of effectiveness the benefits of the silver alloy coated catheters significantly outweigh the costs.

10.4.6 Sensitivity analysis

Tables 10.8 and 10.9 show the results of the sensitivity analysis, which assessed the impact that varying the incidence of UTIs and the cost per case had on the results.

Table 10.8:The effect of varying UTI incidence rates on the proportion of
infections that need to be prevented to cover cost of silver alloy
coated catheters*

Incidence rate (%)	Proportion of infections that need to be prevented to cover cost of silver alloy coated catheters* (%)		
2	47.7		
4	23.8		
6	15.9		
8	11.9		
10	9.5		
12	7.9		
14	6.8		
16	6.0		
18	5.3		
20	4.8		

*All other assumptions (additional cost of catheter, cost per case) remain unchanged

Table 10.9: The effect of varying the cost per UTI on the proportion of infections that need to be prevented to cover cost of silver alloy coated catheters*

Cost per case	Proportion of i	Proportion of infections that need to be prevented to cover cost of silver						
		alloy coat	ed catheters	* (%)				
	General surgery	Orthopaedics	Urology	Gynaecology	Obstetrics			
50% of cost used	26.6	31.3	42.4	10.6	30.3			
100% of cost used	11.2	15.6	21.2	5.3	14.4			
150% of costs used	8.8	10.4	12.1	3.5	10.1			

~ All other assumptions (additional cost of catheter, cost per case) remain unchanged

As the individual parameters are decreased the intervention needs to be more effective for the benefits to cover the costs. As they are increased the intervention needs to be less effective to cover the costs. For example, if the incidence of UTIs in catheterised general surgical patients was just 2%, rather than 7.2% as assumed in the model, then providing all the other assumptions are thought to be valid, 47.7% of the expected UTIs occurring in catheterised general surgical patients would need to be prevented if the costs of the intervention were to be covered. However, if the specialty specific incidence rates are higher than that assumed in the model, then providing the other assumptions are accepted, a lower percentage of infections would need to be prevented to cover the cost of the intervention. For example, if the incidence of UTIs in catheterised general surgical patients was 12%, rather than the 7.2%, as assumed in the model, a 7.9% reduction in the incidence of UTIs in catheterised surgical patients would be needed to cover the cost of the intervention.

The effects of altering more than one parameter at a time were not explored in this model.

10.4.7 Validity of the model

The validity of this model is dependent on how realistic the structure of the model is and how accurately the estimates of the parameters used reflect what is happening in the patient group of interest. The illustrative model presented was based on information derived from this study. It should be noted that the specialty specific incidence rates derived had wide confidence intervals and this should be taken into account when interpreting the results. However, it should also be noted that this particular model is for illustrative purposes. If the results were to be used to inform decision making, the impact of varying the incidence should be taken into consideration.

10.4.8 Interpreting the results of the model

The model provides information on the number of UTIs occurring in catheterised patients; the number and cost of extra bed days utilised as a result of UTIs; and the net financial benefits associated with the routine use of silver alloy coated catheters. Studies that have assessed the effectiveness of the use of this type of catheter suggest that up to 48% of infections can be prevented.²⁴¹ However, the validity of some of the published estimates has been questioned,²⁴³ and as mentioned earlier, to some extent the level of effectiveness achieved depends on the scope for reducing the incidence of this type of infection. Nevertheless the evidence to date does appear to suggest a preventative effect, and the results of this model demonstrate that even at relatively low levels of effectiveness the cost of their routine use can be recouped.

However, the results of the sensitivity analysis demonstrate that if the specialty specific incidence rates used in this model are an overestimate of the actual incidence rates, then the routine use of these catheters may be an expensive option unless a high proportion of infections can be prevented. The routine use of silver alloy catheters in patients in which the infection rate is just 2% would necessitate a 47.7% reduction in rates if the costs were to be recouped. However, with the exception of obstetric patients the lower confidence rate was above 2% for all specialties.

When relatively conservative estimates of the additional costs resulting from HAI were introduced into the model, the results indicated that with the exception of the urology specialty, the catheters would need to be effective at preventing between 11% and 23% of infections if the cost of the catheters were to be covered, the precise level varying with specialty. The level of effectiveness needed for urology patients was somewhat higher - 42%. However, if the estimates used are underestimates of the value of resources that might be released, then the catheters would only have to be effective at preventing between 3.5 and 12.1% of infections, if the costs of the catheters were to be covered, again the actual level of effectiveness needed varying with specialty. It should be noted that the sensitivity analysis did not consider the impact of varying more than one parameter at a time. If the incidence of UTIs and the

additional cost per case used in this model are both thought to overestimate the

situation in a given setting, the benefits may no longer outweigh the cost of the intervention.

However, the model did not assess the benefits associated with a reduction in secondary bactereamias, a known sequela to UTIs. Studies indicated that between 1% and 5% of patients with a UTI will develop a secondary bacteraemia.⁸¹ ⁸² ¹⁴⁴ Acquisition of a BSI following a UTI may further prolong the patients length of hospital stay and represent an additional burden to the health sector. As such the benefits of preventing hospital acquired UTIs may have been underestimated in the illustrative model. If the use of silver alloy coated catheters are effective in preventing UTIs in patients who in the absence of this intervention would have acquired both a UTI and a bacteraemia, the benefits of preventing uTIs through the routine use of this intervention will increase.

When interpreting the benefits the user should be aware that the cost of the infection, if avoided, will not all be realised as a cash saving. Many of the costs/benefits are fixed costs and expenditure on these resources is committed at the beginning of a time period and cannot be recovered. However, the variable costs/benefits (for example drugs, and other consumable items), which represent a smaller proportion of the total costs, would show as cash savings and as such expenditure that could be avoided.

While the fixed costs avoided will not all be realised as cash savings, they do represent economic benefits as these resources could be deployed to produce other outputs (instead of treating the infection). NHS resources are severely limited and so there is almost certainly an opportunity cost associated with using fixed resources to treat hospital acquired UTIs. This implies a positive value for the fixed resources freed up by prevention control activities. It is therefore justified to use the full cost data (fixed plus variable costs) to represent the benefits.

10.4.9 Potential applications

The model described represents a flexible tool which could easily be adapted to the specific needs of the user. It may be of particular interest to infection control nurses and doctors who wish to demonstrate the magnitude of the burden of this type of infection and the benefits associated with the routine use of silver alloy coated catheters. The information derived from the model may be used to justify the additional expenditure associated with this intervention, and to change policy regarding infection control practice.

While the model presented focuses on UTIs occurring in catheterised patients and the routine use of silver alloy coated catheters as a means of reducing this type of infection, the model could equally be adapted to assess the benefits of an alternative infection prevention intervention.

10.5 Conclusion

In this chapter estimates of the gross benefits that might result if a 15% reduction in infections rates in adult non-day case patients admitted to the five surgical specialties covered in this study, at the study hospital or at the national level were achieved, have been presented. Estimates of both the value of resources that might be released for alternative use and the number of bed days released have been presented. The results demonstrate that the magnitude of these benefits, should this reduction be achieved, is likely to be substantial. However, achieving such a reduction is not cost free. As such the net benefits will be dependent on the costs of achieving such a reduction in rates. With the exception of studies that have assessed the costs and benefits of surveillance, and prophylactic antibiotics, few studies have assessed the cost and benefits of selected infection control practices. In the absence of such studies, economic models can be developed, utilising data from a variety of sources, and the information derived used to inform decision making regarding resource allocation and practice. An illustrative model that assessed the cost and benefits of the routine use of silver alloy coated catheters was presented and the results discussed. The use of models together with the findings of this study will be discussed further in the following chapter.

CHAPTER 11 DISCUSSION AND CONCLUSIONS

11.1 Introduction

In this chapter the results of this study will be discussed. The chapter begins with an overview of the main results and how they compare to those derived in earlier published studies. The strengths and limitations of this research are then discussed in sections 11.3 and 11.4, followed by a discussion of methodological issues in section 11.5. The implications of the study findings for policy and practice are discussed in section 11.6, areas for future research identified in section 11.7 and conclusions drawn in section 11.8.

11.2 Overview of results

This study provides important data on the incidence of HAIs occurring in adult non-day case patients admitted to selected surgical specialties of a district general hospital in England, independent risk factors for these infections, and estimates of the economic burden these infections place on the hospital sector as a result of additional in-patient care.

11.2.1 Incidence of HAI

Of the 2469 patients included in this study 7.5% (95% CI: 6.4%, 8.6%) acquired and presented with one or more HAIs during their in-patient stay. Twenty-nine per cent of infections presented within the first four days of admission, over 50% within the first six days, and 75% within nine days of admission.

The incidence of HAI varied with admission specialty. The incidence of HAI in gynaecology patients was 12.7% (95% CI: 9.9%, 16.8%); 9.3% (95% CI: 6.2%, 14.4%) in obstetric patients who had undergone a caesarean section; 7.4% (95% CI: 5.2%, 10.4%) in orthopaedic patients; 6.1% (95% CI: 4.6%, 7.9%) in general surgical patients and 5.1% (95% CI: 3.9%, 8.4%) in urology patients.

The incidence of specific types of infection also varied. Urinary tract infections (UTIs) were the most frequent type of infection accounting for 48.1% of infections identified; surgical wound infections (SWIs) accounted for 19.2%; lower respiratory tract infections (LRTIs) 11.1%; skin infections 7.7%; bloodstream infections (BSIs) 1.9%; and infections at other sites 12.1%.

Valid comparisons of the incidence rate observed in this study with those observed in other studies are difficult. Important differences exist between this and other studies with respect to the definitions used, surveillance methods employed, the case-mix studied and the treatment patterns followed in the different clinical settings. As discussed in Chapter 2 these differences will inevitably influence the reported incidence rates. Inability to control for these differences renders meaningful, valid comparisons difficult. A study conducted in a similar type of NHS hospital in England in 1992, using the same definitions and surveillance methods used in this study, observed a higher incidence rate amongst surgical patients than that observed in this study: 9.7% of in-patients admitted to the general surgical, gynaecology and orthopaedic specialties of an NHS hospital acquired and presented with an infection whilst in hospital.47 Without further details about the case mix and treatment patterns present it is difficult to interpret these findings. The different infection rates could reflect differences in practice, with the study hospital performing better with respect to infection control. Alternatively they could reflect important case-mix differences, with the study hospital treating patients at lower risk of infection than patients involved in the study by Glenister et al (1992).47 The most likely explanation is that there was a combination of factors at play.

Whilst precise comparisons of the incidence rates observed in the various studies cannot be made, it is interesting to note that studies have found similar patterns with respect to the frequency of the different types of HAI. As in this study, other incidence studies have reported UTIs, SWIs, and LRTIs be the most frequent types of HAI with UTIs being the most frequent.^{48 47}

11.2.2 Risk factors for HAI

The results of the multivariable logistic regression analysis indicated that the odds of acquiring an HAI were higher in females than males; increased with increasing age category and increasing number of co-morbidities; were higher in patients who had diabetes mellitus listed as a co-morbidity; were higher in emergency as compared to elective admissions; varied with admission specialty being highest in gynaecology patients; were greater in patients who had received antibiotics prior to the onset of an infection; and were higher in patients who had one or more operations. With the exception of sex, and diabetic status, the variation identified above was significant at the 5% level. The results of a separate analysis limited to 1,588 patients for whom data on body mass index (BMI) were available found that the incidence did not vary significantly with BMI.

11.2.2.1 Urinary tract infections

The results of the multivariable logistic regression analysis, undertaken to assess independent risk factors for UTIs, indicated that female patients and patients who had been catheterised prior to the onset of this type of infection were at greater risk of acquiring a UTI than male and non-catheterised patients. The odds of acquiring a UTI were 2.6 (95% CI: 1.6, 4.1) times higher in catheterised patients compared to non-catheterised patients and 2.8 (95% CI: 1.4, 5.6) times higher in female compared to male patients. The odds of acquiring a UTI were also found to increase with increasing age category, and varied significantly with admission specialty. The odds of acquiring a UTI were higher for gynaecology and obstetric patients, compared to patients in the other specialty groups and were higher in patients who had an operative procedure (Odds ratio 1.1 95% CI: 1.1, 6.5). The odds increased with increasing number of co-morbidities, and were higher in patients who had diabetes mellitus listed as a co-morbidity, however these increases were not significant at the 5% level. Similarly whilst the odds of acquiring a UTI were higher for emergency patients, this was not found to be significant at the 5% level. The odds of acquiring a UTI were similar both in patients who had antibiotics prior to a UTI and in those who

had not, although the upper confidence intervals suggest that in some case patients who received prophylactic antibiotics were at greater risk of acquiring a UTI. This outcome suggests that the administration of prophylactic antibiotics is perhaps a marker for another risk factor not included in this model.

11.2.2.2 Surgical wound infections

The results of the multivariable logistic regression analysis undertaken to assess independent risk factors for SWIs identified just one independent risk factor that was significantly associated with the risk of acquiring a SWI at the 5% level. The presence of one or more wound drains was found to increase the risk of acquiring this type of infection by a factor of 2.2:1 (95% CI:1.0, 5.0).

11.2.2.3 Lower respiratory tract infections

The results of the multivariable logistic regression analysis undertaken to assess independent risk factors for LRTIs also identified just one factor that was significant at the 5% level. The presence of an endotracheal or tracheostomy tube was found to increase the risk of acquiring a LRTI by a factor of 77:1 (95% CI: 27.4, 216.9).

To a degree the results of this study are consistent with the findings of earlier studies. For example, a number of earlier studies have identified urinary catheters and female sex to be key risk factors for UTIs,¹ ¹³⁴ ¹³⁷ ¹³⁸ ¹⁴³ ¹⁴⁴ whilst others have identified the presence of an endotracheal or tracheostomy tube as key risk factors for lower respiratory tract infections.¹ ¹⁴⁷ However, unlike some earlier studies, this study did not find a significant associated between diabetes mellitus and infection or a significant association between body mass index and infection risk.¹⁹ ⁶⁶ ⁷⁴ ¹³⁸ ¹⁴¹ ¹⁴² However, it should be noted that only 3.4% of patients studied had diabetes mellitus and data on BMI was only available for 63% of the patients studied, thus limiting the ability of this work to identify an association if present.

11.2.3 Economic burden

Generalised linear modelling statistical techniques were used to determine how much of the observed variation in costs and LOS could be attributed to the presence of an infection. The analysis controlled for the potentially confounding effects of age, sex, number of co-morbidities, admission type, and admission specialty and used both an identity and a log link. Models that incorporate an identity link assume additive effects, and as such its use enabled estimates of the mean additional costs incurred by infected patients, and the mean number of additional days infected patients remained in hospital, to be taken directly from the model. In contrast models incorporating a log link assume multiplicative or proportional effects, thus providing estimates of the ratio of costs and LOS incurred by infected compared to uninfected patients.

It is not clear which approach most accurately reflects the impact of HAI on costs and LOS. Whether HAIs on average have an additive or a proportional effect will to some extent vary with specialty and type of infection. Consequently, since it is impossible to establish that either is the correct way to assess the impact of HAI on costs, estimates of both the average costs of HAI and the average proportional increase in costs incurred by infected compared to uninfected patients, after controlling for a number of potential confounding factors, have been reported in this thesis. However, it is worth noting that the ratios of the costs incurred by infected compared to uninfected patients, derived from the additional cost estimates, taken from the identity link model, are very similar to the ratios obtained directly from the model using a log link and vice-versa.

11.2.3.1 Estimated cost to the hospital sector as a result of additional inpatient care

The results indicate that acquiring an infection, on average, increased hospital costs by a factor of 2.3 (95% CI: 2.0, 2.6). Estimates of the additional costs attributable to infection indicated that on average HAIs cost the hospital sector an additional £2,254 (95% CI: £1,738, £2,770) per case.

Estimates of the impact of HAI on hospital costs were found to vary with specialty. Estimates of the proportional increase in hospital costs incurred by infected compared to uninfected patients from the same specialty indicated that infections occurring in general surgical patients, on average, increased costs by a factor of 3.9 (95% CI: 3.0, 4.9), whereas infections occurring in orthopaedic patients were estimated to increase costs by a factor of 2.4 (95% CI: 1.8, 3.3); in urology patients costs were estimated to increase by a factor of 2.1 (95% CI: 1.5, 2.9); in gynaecology patients costs were estimated to increase by a factor of 1.2 (95% CI: 0.9, 1.6); and in patients who had a caesarean section, costs were estimated to increase by a factor of 1.1 (95% CI: 0.7, 1.6). As evidenced by the confidence intervals, there was some uncertainty about the estimates relating to both obstetric and gynaecology patients.

Estimates of the additional costs incurred, indicated that on average HAIs occurring in general surgical patients cost the hospital sector an additional \pounds 4,368 (95% CI: £2,992, £5,744) per case; this compared to an additional \pounds 3,357 (95% CI: £1,821, £4,894) per case in orthopaedic patients; an additional \pounds 1,474 (95% CI: £534, £2,414) per case in urology patients; an additional £375 (95% CI:-£145, £895) per case in gynaecology patients and an additional £207 (95% CI:-£841, £121) per case in patients who had undergone a caesarean section. Again, as shown by the confidence intervals, there was some uncertainty about the estimates relating to both obstetric and gynaecology patients.

Cost estimates also varied with site of infection. The results indicated that multiple infections were the most costly and that UTIs were the least costly. The acquisition of more than one HAI was estimated to result in hospital costs almost six times higher than uninfected patients (model estimate: 5.8 95% CI: 3.9, 7.9). In contrast, UTIs, on average, were estimated to increase costs by a factor of just 1.6 (95% CI: 1.3, 1.9). With the exception of BSIs, infections occurring at the other sites were estimated to result in a two to three fold increase in costs. Acquisition of a BSI and no other infection was estimated to

increase costs by a factor of 6.0 (95% CI: 2.2, 16.2). However, as evidenced by the confidence interval there was considerable uncertainty about this estimate. Only three patients acquired a BSI and no other infection, two of whom died as in-patients.

Estimates of the additional costs incurred indicated that on average, multiple infections cost the hospital sector an additional £8,118 (95%CI: £7,930, £11,310) per case, whereas UTIs, on average, were estimated to cost £944 (95% CI: £441, £1,446) per case. The average cost of infections occurring at other sites were as follows: SWIs £1,497 (95% CI: £548, £2,447); infections at sites not elsewhere classified £1,671 (95% CI: £279, £3,064); skin infections £1,567 (95% CI: -£110, £3,245); LRTIs £2,672 (95% CI: £753, £4,592) and BSIs £6,953 (95% CI: -£1,652, £15,558).

The estimates derived in this study are considerably higher than those derived in other studies. For example, the most recent UK based study that estimated the economic burden of HAIs occurring in surgical patients (general surgical, gynaecology and orthopaedic patients), indicated that on average HAIs costs the hospital sector £1,041 per case.¹⁴

A number of factors could account for the higher cost estimate derived in this study. Firstly the infections identified in this study may have been more resource intensive and thus more costly to treat. However, this is unlikely to explain all the cost differences observed. Other factors which might in part explain the considerable difference in the estimates derived in this study, and those reported in the study by Coello *et al* include inflation (the study by Coello *et al* was conducted in 1988), the more detailed approach to identifying and valuing resources used in this study, and differences with respect to the attribution methods employed which in turn could impact on the estimates derived.

Whilst the precise estimates derived in earlier studies differ markedly to those obtained in this study, the pattern of increase is similar. For example, as in this study, Coello *et a* ¹⁴ found UTIs to be the least costly infections and multiple infections the most resource intensive. A number of other studies that have examined the relative costs of different types of HAI have also found UTIs to be the least resource intensive.^{171 11 69}

11.2.3.2 Estimated number of additional days infected patients remain in hospital

HAIs were, on average, estimated to increase the patients LOS by a factor of 2.1 (95% CI: 1.8, 2.5). Estimates of the number of extra days patients remained in hospital indicated that on average HAIs extended the in-patient stay by 7.8 days (95% CI: 5.7 to 10.0).

As with costs, the estimates of the number of extra days patients remained in hospital varied with specialty. Infections occurring in general surgical patients were estimated to result in a three fold increase in LOS (model estimate 3.4, 95% CI: 2.6, 4.4), whereas infections occurring in orthopaedic and urology patients, on average, were estimated to double the in-patient stay. HAIs occurring in orthopaedic patients were estimated to increase the LOS by a factor of 2.0 (95% CI: 1.4, 2.7) and in urology patients by a factor of 2.5 (95% CI: 1.7, 3.6). Infections occurring in gynaecology and obstetric patients were estimated to have a relatively small impact on LOS. HAIs occurring in gynaecology patients were estimated to increase the LOS by a factor of 1.3 (95% CI: 1.0, 1.7) and infections occurring in obstetric patients were estimated to increase LOS by a factor of 1.1 (95% CI: 0.7, 1.6).

Estimates of the number of extra days attributable to HAI indicated that, on average, infections occurring in general surgical patients prolonged the inpatient stay by 15.3 (95% CI: 9.5, 21.0) days, whereas HAIs occurring in orthopaedic patients were estimated to extend the LOS by 11.7 (95% CI: 4.7, 18.8) days; in urology patients by 7.9 (95% CI: 2.9, 12.9) days, in gynaecology

305

patients by 1.8 (95% CI:-0.4, 4.0) days and in patients who had undergone a caesarean section LOS, on average, was extended by 0.6 (95% CI:-2.8, 4.1) days. As with costs, as shown by the confidence intervals obtained, there was considerable uncertainty about the estimates of the number of additional days infected obstetric and gynaecology patients remained in hospital.

Estimates of the impact HAI had on LOS also varied with site of infection. The results indicated that acquiring more than one infection resulted in the greatest proportional increase in LOS: on average, multiple infections were estimated to increase LOS by a factor of 4.4 (95% CI: 2.9, 6.5). In contrast BSIs were estimated to increase LOS by a factor of just 1.5 (95% CI: 0.5, 4.4). This might appear a rather surprising finding. However, as mentioned above only three patients acquired a BSI and no other infection, two of whom died whilst in hospital thus curtailing their length of hospital stay. As shown by the confidence It is interval, there was considerable uncertainty around this estimate. interesting to note that whilst the LOS of patients who acquired a BSI and no other infection was not markedly prolonged, as mentioned in section 11.2.3.1 the costs incurred were considerable. On average, acquiring a BSI was estimated to result in a six-fold (95% CI: 2.2, 16.2) increase in costs. Whilst, as evidenced by the wide confidence intervals, there was some uncertainty surrounding this estimate, the results suggest that whilst LOS is not markedly increased costs are, thus indicating that the time in hospital is resource intensive.

Acquisition of a UTI was also found to have a relatively small impact on LOS. UTIs were estimated to extend the LOS by a factor of 1.5 (95% CI: 1.3, 1.9). Infections occurring at the other sites on average were estimated to result in a two fold increase in LOS.

Site-specific estimates of the number of extra days patients remained in hospital indicated that, on average, multiple infections extended the in-patient stay by 24 days (95% CI:11.8, 36.4), whereas BSIs and no other infection were estimated

to extend the in-patient stay by just 2.2 (95% CI: -7.1, 11.5) days. UTIs were estimated to prolong the in-patient stay by 3.3 (95% CI: 1.3, 5.3) days, SWIs were estimated to extended the LOS by 7.2 (95% CI: 2.3, 12.0) days, LRTIs by 8.0 (95% 0.6, 15.4) days; skin infections by 9.9 (95% CI: 0.4, 19.3) days, and infections at sites not elsewhere classified by 11.6 (95%CI: 3.0, 20.2) days.

11.2.3.3 Distribution of additional costs

This study took a detailed approach to assessing the cost of resources used by both infected and uninfected patients. The resources used were categorised under a number of headings (hospital overheads, directorate management, capital charges, medical time, nursing care, paramedics and specialist nurses, physiotherapy. surgical interventions. consumables used for specific procedures, antimicrobials, non-antimicrobial drugs, microbiology tests, other pathology tests, endoscopies, radiology and other tests) and the distribution of the additional costs incurred by infected patients subsequently assessed. The results indicated that for all cost categories, the costs incurred by infected patients were higher than for uninfected patients. The cost of nursing care was the largest contributor to the additional costs incurred accounting for 40.5% of the additional costs. Cost categories that were directly linked to time in hospital, taken together, were the second largest contributor accounting for 38.4% of the additional costs incurred by infected patients. Costs that were considered to be directly linked to time in hospital included the costs of hospital overheads, directorate management, capital charges and medical time. These costs were all assigned to individual patients on the basis of their LOS.

The distribution of costs varied with site of infection reflecting the different resource requirements of different types of HAI. For example, nursing care accounted for a varying proportion of the costs attributed to infection. Nursing care accounted for 60% of the cost of BSIs; 46% of the costs of multiple infections and LRTIs; 34% of the cost of UTIs, 33% of cost of SWIs, just 25% of the cost of skin infections and 28% of the costs associated with infections at a range of other sites.

307

11.3 Strengths of study

11.3.1 Inclusion of a number of surgical specialties

One of the strengths of this study was that it assessed the incidence and economic burden of HAIs occurring in adult, non-day case patients admitted to surgical specialties common to all district general hospitals: general surgery, orthopaedics, urology, gynaecology and obstetrics (caesarean sections only). Nationally patients admitted to these specialties accounted for 35% of all adult non-day case admissions in 1994/5.²²¹ The results of this study are therefore relevant to a substantial proportion of NHS patients.

This study provides important data on the incidence of, and risk factors for HAIs occurring in adult non-day case patients admitted to the specialties listed above and the economic burden they impose on the hospital sector. Few studies have assessed the incidence of HAI occurring in such a broad case mix of patients. Three notable exceptions are the SENIC study conducted in the US in 1975 which assessed the incidence of HAI occurring in 6,449 medical and surgical patients;⁴⁹ a study by Glenister *et al*⁴⁷ which assessed the incidence of HAI occurring in 3,326 adult medical, surgical, orthopaedic and gynaecology patients; and an audit study involving 81,218 adult patients admitted to the medical, surgical, orthopaedic and gynaecology specialties of 19 hospitals in England and Wales conducted in 1993/4.¹ The majority of incidence studies have focussed on the incidence of HAIs occurring in a narrower group of patients defined by specialty or operative procedure. For example, a number of studies have focussed on the incidence of HAIs occurring in patients who had a caesarean section.⁵¹⁻⁵⁶

Studies that have assessed the economic burden of HAI also vary in scope. Many studies are limited to the economic burden of HAIs occurring in patients admitted to a particular specialty or the burden associated with a particular type of infection occurring in a selected patient group. The most recent UK study that involved a broad case-mix of patients was a study in 1988 by Coello *et al* (1993)¹⁴ that assessed the economic burden of HAIs occurring in adult patients admitted to a similar range of specialties: general surgery, orthopaedics and gynaecology. The results reported in this thesis provide a more timely assessment of costs resulting from infections occurring in adult patients admitted to specialties common to most hospitals.

11.3.2 Use of previously validated definitions of infection and surveillance methods

The study used previously validated definitions of infection and surveillance methods thus enhancing the validity of the results obtained. The definitions and surveillance methods used were developed and validated in an earlier study by Glenister *et al* (1992).⁷⁷ The surveillance method adopted involved reviewing patient records (patient notes, drug charts, observations charts and microbiology records) and liaising with ward staff to identify signs and symptoms that were indicative of a possible infection. Those patients who had signs and symptoms that met the criteria for infection as detailed in the validated definitions were classified as having an infection.

11.3.3 A detailed approach to costing resource use

In marked contrast to other studies this study took a very detailed approach to the estimation of the economic burden HAIs place on the hospital sector as a result of additional in-patient care. Studies reported in the literature have generally limited their assessment of costs to a few areas considered to be directly linked to infection. For example, some studies have limited their assessment of costs to those associated with an extended length of stay, whilst others have adopted a slightly more detailed approach, including the cost of specific resources considered directly linked to infection such as the cost of antibiotics, in addition to those costs associated with an extended stay.^{13 14 189}

In this study no assumptions were made as to what types of resources might be used in greater quantity by infected patients as a consequence of having acquired an infection in hospital. As far as possible data on all resources used by both infected and uninfected patients were collected and subsequently valued. This included the costs of all drugs prescribed, investigations undertaken, procedures performed, care administered by nurses, doctors and personnel from the professions allied to medicine, and the costs associated with an extended hospital stay. Estimates of the value of these resources were subsequently made and statistical modelling techniques employed to determine how much of the observed variation in costs between infected and uninfected patients could be attributed to the presence of an HAI.

The methods derived to estimate the value of these resources were also very Earlier studies have tended to value resources by applying an detailed. average cost to the resources used. For example, in many cases estimates of the cost of an extended hospital stay have been derived by first deriving an estimate of the average cost per bed day and then applying this to the number of extra days infected patients were estimated to remain in hospital. Such an approach fails to take into account that costs may vary markedly with day of admission. Hollingsworth et al (1993) found that patients admitted to hospital with a fractured neck of femur incurred relatively high daily costs during the first few days in hospital, after which they decreased.¹⁹⁸ Similar patterns are likely to be present for other types of patients undergoing emergency surgery, whereas for elective patients, depending on the day of surgery, costs may initially be relatively low on admission to hospital, rising on the day of surgery, and then decreasing over time as the patient recovers and becomes less dependent on nursing and medical care. The use of an average cost also fails to take into account that complications such as an infection may cause daily costs to increase, the level of increase depending on the type and severity of the infection and how it interacts with the patients other co-morbidities.

In this study, detailed data were collected on the types and quantities of resources used. As such, wherever possible, the estimates of the cost of resources used reflect consumption by individual patients. Unit costs were derived for laboratory and radiology tests, drugs, operations and procedures performed, and subsequently allocated to individual patients based on the

310

amount consumed. The methods developed to derive unit costs aimed to take into account all resources used. For example, the cost of administering an intramuscular drug not only included the cost of the drug, but also the cost of the syringe, needle and, if applicable the cost of the solution used in the preparation of the drug for administration. The cost of the time and skills involved in preparing the drug and subsequently administering it were included in the pharmacy overhead costs allocated on a cost per bed day, and the nursing care costs allocated on the basis of the amount of nursing care individual patients received on a daily basis. The algorithms used to derive nursing costs took into account the number and types of drugs administered on a daily basis.

As indicated above, estimates of the cost of nursing care took into account the level of nursing care administered to patients on a daily basis. Patients were categorised into one of seven nursing care groups based on the intensity of the nursing care received. Care group one denoted patients who needed very little nursing care, whereas at the other end of the spectrum, care group seven denoted patients requiring 24 hour nursing care. Using a previously validated care group weighting system, nursing care costs for individual wards were allocated to patients based on their daily care group.

Estimates of the costs of care administered by physiotherapists, occupational therapists, nutritionists and other health personnel from the professions allied to medicine, were derived from activity data and employment costs, and allocated on the basis of the number of contacts individual patients had with these health care professionals. Unfortunately it was not possible to adopt a similar approach for medical care. Resource constraints prohibited the assessment of the amount of care individual patients received from medical staff. As such an average cost per bed day was derived for medical costs and allocated to patients on the basis of the number of days they remained in hospital.

Similarly it was beyond the scope of this study to assess individual patient consumption of the services provided by the various overhead directorates. As such estimates of the cost of the hospital overheads were attributed to patients on the basis of an average cost per day, having first allowed for consumption of the various overhead directorates by the overheard directorates themselves. For example, the allocation model developed did not simply apportion the costs of heating to individual patients based on their LOS, but through the use of a series of simultaneous equations first allowed for consumption of heating by the 'Estates Directorate' itself and the other overhead directorates. Once this had been taken into account the remaining costs were allocated to patients on the basis of an average cost per day.

This detailed approach to identifying and estimating the cost of resources used by both infected and uninfected patients was both time consuming and complex to undertake. However, the approach offered a number of advantages over limiting estimates to the number and cost of extra days patients remained in hospital and the use and cost of a few specific resources considered to be linked to infection. The approach used enabled a more accurate assessment of the costs of resources used by individual patients, from which estimates of the ratio of costs and the additional costs incurred by infected compared to uninfected patients could be derived. Since the methods employed did not make any assumptions as to what categories of costs were likely to be higher in infected than uninfected patients, the approach enabled a detailed assessment of the distribution of additional costs to be made, thus providing information which may be of use to those involved in decision making regarding the allocation and use of scarce hospital resources.

11.3.4 Attribution methods

As discussed in Chapter 3, the majority of studies that have assessed the economic burden of HAI have adopted one of three methods to attribute resource use and costs to infection: the concurrent method which involves the subjective assessment of the resources used by infected patients as a result of

their infection, the comparative method, and the comparative method with matching of cases and controls. As detailed in Chapter 3 all three approaches have their limitations.

In this study an alternative approach to the attribution of resource use and costs to HAI was taken. A large cohort of patients was followed and patients who acquired an infection in hospital identified. Generalised linear modelling statistical techniques were then used to assess the impact of HAI on hospital resource use and costs by comparison of the HAI group and remainder of the cohort. The modelling took into account a range of factors that, in addition to the presence of an HAI, were considered to influence resource use and costs (age, sex, admission specialty, admission type, and number of co-morbidities). Two methods of describing the impact of HAI on resource use were used. Estimates of the proportional increase in costs and LOS incurred by infected patients compared to uninfected patients were derived through the use of a log link, and estimates of the additional costs and extra days in hospital attributable to HAI were derived through the use of an identity link.

This alternative approach to attribution has a number of strengths. The approach is considerably more rigorous that the concurrent approach, the validity and reliability of which has been questioned in the literature.¹⁸⁷ It has been suggested that physician reviewers may be hesitant to attribute resources to the presence of an HAI and as such the resulting estimates of costs may be underestimates of the resources and costs that are attributable to the presence of an infection.⁴⁸ The approach adopted in this study did not involve subjective assessment of resources used by infected patients. The resources used by both infected and uninfected patients were identified and costed and generalised linear modelling statistical techniques applied to determine how much of the observed variation in resource use and costs could be attributed to infection.

This approach also has a number of strengths when compared to the comparative approach. The comparative approach without matching simply involves assessing resource use and costs incurred by infected and uninfected patients and attributing any cost differences to the presence of an infection. The approach does not take into account any other factors that might differ between the two groups of patients (infected and uninfected) that in turn might influence resource use and costs. For example, infected patients even in the absence of an infection may have utilised more resources than uninfected patients and as such only a proportion of the additional costs incurred can be justly attributed to the infection. Thus failure to control for important differences between infected and uninfected patients that in turn might influence resource use and costs, may result in an overestimate of the economic burden of HAI. Haley et al (1980) compared three alternative methods for the attribution of resources and costs to HAI and concluded that studies which attributed resource use and costs to HAI using the comparative approach resulted in overestimates of the costs attributable to infection.⁴⁸ The analysis used in this study controlled for a number of factors that might influence resource use and costs.

The comparative approach with matching of cases and controls does take into account factors that may influence resource use and costs. Infected patients are matched with controls on the basis of factors such as age, sex, diagnosis and number of co-morbidities. However, studies frequently encounter difficulties finding suitable controls for infected patients, resulting in infected patients being lost from the final analysis. As detailed in Chapter 3, the proportion of infected patients for whom suitable controls can be found varies considerably with study. Haley *et al* (1980)⁴⁸ in a review of matched studies conducted between 1953 and 1975, found the proportion of infected patients who were successfully matched with uninfected 'controls' varied considerably from a low of 32% to 100%. Scheckler *et al* (1978)¹⁷¹ in a study to assess the economic burden of HAIs occurring in 104 patients admitted to a community hospital in the US between January and March 1978 were unable to find a

sufficient number of suitable controls and therefore had to abandon this approach to attribution of costs. A more recent study, conducted in the UK, successfully matched 67 (85%) of the 79 infected surgical patients for whom medical records were available for review.¹⁴ A study by Kappstein *et al* (1992),¹⁸² which assessed the excess costs and LOS resulting from ventilator associated pneumonia occurring in patients admitted to the intensive care unit of a university teaching hospital in Germany, found that after excluding cases that died during their admission, suitable controls could only be found for 34 (60%) of the 57 cases, and a study conducted in Turkey in 1994 involving general surgical patients successfully matched only 67% of their 225 infected patients.⁶¹

In those cases where suitable controls are found this approach provides valuable estimates of the economic burden imposed. However, the estimates derived represent the economic burden imposed by infections occurring in a limited set of patients. Frequently it is the sicker, more resource intensive patients for whom suitable controls cannot be found. Thus the estimates derived may underestimate the average costs of HAIs occurring in the wider population from which the infected patients were identified. The strength of the approach adopted in this study was that it included all infected patients. By utilising all the available data it was possible to control for a similar range of factors as have been used in comparative-with-matching studies, yet include all infected patients. As such a broader understanding of the average costs of infections may have imposed a relatively small burden and others may have been considerable more expensive, the approach allows for an overall average estimate to be derived.

11.3.5 Generalisability of results to other health care settings

A major strength of this work is the perceived generalisability of the results to other health care settings. The results of this study were derived from data relating to adult non-day case patients admitted to five surgical specialties of one NHS hospital in England. The inclusion of just one hospital inevitably represents a limitation of the study design. Ideally patients would have been recruited from a number of hospitals randomly selected, however funding constraints prohibited the inclusion of more than one hospital site. However, despite this limitation for a number of reasons it would appear justifiable to assume that the results may be taken to represent what occurs in patients admitted to the same clinical specialties in other health care settings.

The hospital selected was an NHS district general hospital not dissimilar to other district general hospitals in England in terms of the types of care offered, its size, and a range of financial indicators² and, as already mentioned, patients were selected from specialties common to all district general hospitals. It is acknowledged that the cost of resources will inevitably vary to some extent from one hospital to another, depending on a number of factors such as the suppliers to the hospital. However, the cost of specific resources within any given setting will be the same for both infected and uninfected patients. As such, if it can be assumed that clinical practice at the study hospital is similar to that occurring in other health care facilities, it is likely that the ratios of costs incurred by infected compared to uninfected patients derived from the statistical analysis, reflect those occurring in adult non-day case patients admitted to the same specialties at other NHS hospitals. Additionally, since the study hospital was not dissimilar to other district general hospitals it is also likely that the estimates of the additional costs of HAIs are also broadly generalisable to other health care settinas.

If it is accepted that the results are generalisable to other settings this represents a major strength of the study design facilitating the wider application and use of the study results. For example, they can be used in economic models to assess the burden of HAI occurring in a specified population of patients from similar specialties, and when used together with data on the cost and effectiveness of infection control practices, the results can be used to estimate the potential benefits of investment in infection control. Predictive

316

models can be developed and the results of such models can subsequently contribute towards decisions about the allocation of resources to infection prevention and control and the nature of infection control programmes.

11.4 Limitations

Whilst recognising the strengths of the study, the results should be interpreted in the context of the limitations of the methods used. These limitations relate to a number of factors that are discussed below.

11.4.1 The study was limited to one hospital

As acknowledged above in section 11.3.5 the study was limited to one hospital. Inevitably this potentially has implications for the generalisabily of the results. Ideally a number of randomly selected hospitals would have been included allowing for inter-hospital comparisons to be made and the generalisability of the main results to be assessed. Resource constraints prohibited such an approach. Nevertheless it should be noted that the hospital selected was not dissimilar to other district general hospitals in England and as such, whilst the inclusion of a number of hospitals would have been preferable, the limitations of just one study site are not as marked as they might have been.

11.4.2 Limited to adult non-day case patients admitted to selected surgical specialties

The estimates derived in this study reflect the incidence and economic burden of HAIs occurring in adult non-day case patients admitted to five surgical specialties of a district general hospital. The burden imposed by infections occurring in excluded patient groups may differ from these estimates. In some cases the costs may be considerably higher than those estimated here. For example, the cost of HAIs occurring in patients who have received cardiac surgery may be considerably higher. Other excluded groups include neonates; paediatrics; day cases; and patients admitted to burns, oncology, dialysis, and neurosurgical units. Whilst focussing on this patient group represents a limitation of the study, it should be noted that the specialties included are common to most NHS hospitals and that patients admitted to these specialties account for a substantial proportion of NHS admissions. In 1994/5 patients admitted to the five surgical specialties covered in this study accounted for 35% of all adult non-day case admissions.²²¹

11.4.3 Potential patient selection bias

Selection bias occurs if patients successfully recruited into the study are systematically different from those who were not recruited.²⁴⁶ In this study failure to invite all eligible patients to participate in the study and patient refusal are two areas that potentially may have introduced some bias.

As far as possible the research assistants attempted to invite all eligible patients (i.e. patients who met the inclusion criteria) to participate in the study. However, for a number of practical reasons this was not always possible. Time constraints were a key limiting factor. The research assistants were required to collect a vast amount of data on each patient successfully recruited into the study. In order to ensure a complete data set was obtained for all those recruited it was not possible to attempt to recruit all eligible patients. Another constraint was that at times it was not possible to identify a suitable time to approach eligible patients and invite them to participate.

Failure to recruit all eligible patients is not in itself a problem, providing that the cohort of patients not invited to participate, are similar to those who were invited to participate. However, the risk is that those patients not invited to participate in the study differ from those invited in some important respect. For example due to time pressures it is possible that the research assistants may have consciously, or subconsciously, elected to recruit patients who were perhaps not as acutely ill, in preference to those who were acutely ill. Such patients are likely to have been easier to recruit and, following recruitment may have resulted in less data collection that the sicker patients. Furthermore, it is likely that the less acutely ill patients are more readily available for recruitment than those who are acutely ill. Many of the sicker patients may have been engaged

with medical and nursing staff when the research assistants were trying to recruit patients, thus prohibiting recruitment at that time. However, it should be noted that if these and/or other differences were present they are not likely to have major implications unless the differences are unequally distributed between infected and uninfected patient groups. If those patients who were not invited to participate in the study differed in important respects to those who were invited, and these differences were concentrated in what would have been the 'infected' or the 'uninfected' patient group had they been recruited, failure to recruit all eligible patients will have implications for the representativeness of the results. However, if these differences are equally distributed between the intended 'infected' and 'uninfected' patient groups had all eligible patients been recruited, the implications for the representativeness of the results obtained are likely to be minimal.

It is possible that many of the anticipated differences between patients invited to participate in the study and those not invited may have had impact on the level of resource use and in turn may have been unequally distributed between infected and uninfected patients. For example, it is possible that patients who were acutely ill on admission, may in some cases have utilised more resources than the less acutely ill, and may also have been at greater risk of infection. Had they been recruited these patients may have been concentrated in what would have been the 'infected' group. Consequently, failure to recruit these patients may have introduced bias into the results obtained.

Refusal to participate in the study may also have introduced some bias. Overall 5.6% of patients refused to participate. Their reasons for doing so varied. Some patients had an inherent distrust of research studies and simply did not want to be involved. Others were concerned about the time implications for themselves as participants (the underlying study required a proportion of patients to complete a post-discharge questionnaire). Again if differences were present and these were unequally distributed between what would have been the 'infected' and 'uninfected' patient groups, then this again would have introduced some bias into the results.

319

A number of steps were taken to minimise and monitor this risk of selection bias. The importance of minimising this risk was discussed with the research assistants prior to the start of data collection; appropriate recruitment training was given and the research assistants were required to record base line data on those patient they were unable to recruit. This included details of age group, admission specialty and type of admission. The age group, admission specialty and admission type distributions of recruited patients and eligible patients who for practical reasons were not invited to participate and patients who refused participation were subsequently compared. Whilst some differences were observed, these were not marked. As such it can be concluded that on the basis of these factors the recruited sample was fairly representative of the wider eligible population group at the study hospital. However it is possible that there may be other systematic differences between those invited to participate and those who were not invited or refused participation, which are not accounted for in this analysis.

11.4.4 Limited to the assessment of the incidence of HAIs presenting during the in-patient period

This thesis was limited to the estimation of the incidence of HAIs presenting during the in-patient period. Infections acquired in hospital but presenting following discharge were not included. As such the results presented are inevitably an underestimate of the 'true' incidence of HAI. The results of the study to which this work is linked suggest that infections presenting in the community are a considerable problem.² The study found that 19% of patients who, at the time of discharge were classified as not having an infection, reported signs and symptoms suggestive of an infection occurring within four weeks of discharge from hospital. Of those patients who had an infection identified in hospital 30% reported signs and symptoms suggestive of an infection of the infection first identified in hospital. Other studies have also highlighted this problem.

Studies examining the incidence of SWIs presenting post-discharge indicate that anything from 20 -86% of SWIs present post-discharge.^{98 99 65 100-102 247 57 74} 103-105

When considering the magnitude of the problem of HAI the fact that the incidence of HAI observed in this study was limited to infections presenting during the in patient period and that more infections may have presented post-discharge should be borne in mind.

11.4.5 Strict criteria for identifying surgical wound infections

As indicated above in section 11.3.2 one of the strengths of this study was the use of previously validated definitions of infection. However, it should be noted that the definition for SWI was rather strict. The definition used stated that there needed to be a purulent discharge from the wound plus or minus a range of symptoms. Whilst this definition probably ensured that there were few false positives, it also probably led to a number of false negatives and as such an under-estimation of the incidence of SWIs. The need for a purulent discharge to be present is likely to have resulted in a number of patients being misclassified as uninfected patients. The research assistants reported that in a number of cases patients who were pyrexial, and had a red and inflamed wound area were treated as if they had an infection with initiation of antibiotic therapy. The surgical team did not wait for a purulent discharge to develop before prescribing antibiotics. This was particularly noticeable amongst orthopaedic patients where the consequences of an infected wound can be particularly severe. For example, a SWI following a total hip replacement could quickly develop into an infected prosthesis requiring revision of the hip replacement. Thus, as indicated above, the research assistants reported that antibiotics were frequently prescribed prior to the development of purulent discharge. Consequently, the incidence of SWI in this study is likely to be an underestimate of the 'true' incidence of this type of infection, and as such the overall estimate of the incidence of HAI is also likely to be an underestimate. The cost estimates may also be underestimated. If a proportion of infected patients were misclassified as uninfected, and resource use amongst these incorrectly classified patients was higher than uninfected patients, then the effect would have been to increase the average cost of uninfected patients and reduce the ratio of increase in costs amongst infected patients and the estimates of the additional costs incurred.

11.4.6 Incomplete data on the pathogens involved.

The results of the underlying study indicated that data on pathogens involved were only available for 50% of infections identified.² Given the paucity of data, it was realised at the outset that an analysis of the impact infections involving specific pathogens or groups of pathogens had on hospital costs was likely to be beyond the scope of this thesis. This was borne out by the data: less that 50% of infections occurring in surgical patients had a pathogen identified. The study's inability to estimate the incidence of specific types of infection defined by causative pathogen, and the impact they have on resources used, represents a limitation of this study, and is an area requiring further study. It is likely that the cost of infections will vary depending on the pathogen involved, with the greatest costs resulting from infections involving antibiotic resistant pathogens. If this is considered to be the case, future work should perhaps focus on this.

11.4.7 Incomplete data on risk factors for HAI

This study is linked to a wider study of the socio-economic burden of HAI in adult non-day case patients admitted to both medical and surgical specialties of a district general hospital. The underlying study, was primarily concerned with the economic burden infections imposed, and not the identification of risk factors for these infections. However, at the outset it was acknowledged that if possible it would be desirable to collect data on risk factors for HAIs. At the same time, the need not to over burden the research nurses with excessive data collection requirements was appreciated. However, since the decision was taken to adopt a very detailed approach to identifying and costing resources used by infected and uninfected patients, it transpired that the planned collection of data on resource use also represented data on risk factors. For

example, data on the use of intravenous lines and urinary catheters not only provided data on the resources used but also the presence of two important risk factors for HAI. Similarly data on case-mix (age, sex, diagnosis, number of comorbidities), which was essential for the attribution process also provided important risk factor data. However, the collection of some of the planned data caused problems for the research assistants. The volume of data required was considerable and the availability of some of the data varied. Whilst the research assistants encountered few problems when collecting data on case-mix, and key items of resource use (e.g. catheters and drains), problems were encountered when collecting operation data. The collection of data on operations undertaken did not present any problems. However more detailed information such as the length of time on the operating table, the grade of the surgeon who performed the surgery, and the ASA grade proved more difficult to obtain. The research assistants were instructed to obtain this information from the operating directorates audit data collection form. During the first few months of the data collection period no major problems were encountered. The form was readily available and generally well completed. However, over time the availability and the completeness of these forms decreased. A change of management may in part have been responsible for this decline. At the outset the manager of the Operations Directorate had an active interest in the data recorded on these forms that was subsequently used for audit purposes. However, this interest declined following a change in management. The resulting effect was incomplete data for a number of variables, which in turn reduced the ability to conduct risk factor analysis.

As such, when considering the results of the analysis that examined how the incidence of HAI and specific types of HAI varied with selected risk factors, it should be borne in mind that the range of factors included in the analysis were out of necessity limited to those for which data had been collected in the underlying study and for which there were complete data sets. Incomplete data particularly affected the analysis examining how SWIs varied with selected factors. Ideally additional factors would have been included in the analysis such
as data on the time the patient was on the operating table, grade of surgeon and ASA score. Thus whilst the results clearly show how the incidence varied with the factors included in the analysis, there may be other factors not included in the analysis which represent a more substantial risk factor for these infections.

11.4.8 The use of average costs

As detailed in section 11.3.3 this study adopted a very detailed approach to valuing resources used. As far as possible the costs allocated to individual patients reflect individual resource use. However, for some components of cost it was not possible to adopt such a detailed approach. For example, the cost of medical care did not reflect the actual amount of care administered to individual patients every day. An average cost per day was derived and allocated to patients on the basis of the number of days they remained in hospital. Resource constraints prohibited a more detailed approach. A more detailed approach to costing medical care, resulting in cost estimates which reflected the amount and type of medical care individual patients received on a daily basis would have required detailed time and motion studies: an approach that was beyond the scope of this study.

11.4.9 Attribution of resources used and costs to HAI

This study utilised statistical modelling techniques to assess how much of the observed variation in resource use and costs could be attributed to the presence of one or more HAIs. The strengths of this approach have been discussed in some detail above (section 11.3.4). In this section some of the limitations of the approach are described.

One of the immediate problems encountered is deciding what factors in addition to HAI status should be included in the analysis. The following were selected: age, sex, type of admission, admission specialty, and number of co-morbidities. These factors were considered to be influential in determining resource use. Single and multi-variable analysis indicated that total costs and LOS varied with all these factors. However, whilst these variables were considered to be key, the possibility exists that there may be other factors that were excluded from this analysis which are better predictors of LOS and costs and this may have an impact on the estimates of the impact of HAI on total costs and LOS derived from the modelling analysis. However, unless these factors existed and influenced either the 'infected' or 'uninfected group' more than the other they will not have caused systematic biases.

11.4.10 The study was limited to the estimation of the costs to the hospital sector

This thesis has focussed on the economic burden HAIs place on the hospital sector as a result of additional in-patient treatment and care. No attempt has been made to assess the economic burden placed on the health sector (primary and secondary) post-discharge; on the patient concerned and their family and friends; or on the economy as a whole. These latter areas represent important cost centres. Focusing on the hospital alone inevitably can only provide a partial picture of the full costs that result from HAIs. In some cases patients identified as having acquired an infection, may be discharged from hospital and followed up in the community, imposing costs on the primary, and in some cases secondary, health care sector, if a greater number of out-patient visits than would have been the case in the absence of infection are required, or if the patient requires re-admission. Patients and carers may also experience increased costs. In other cases patients may not present with an HAI until after they have been discharged from hospital. Over recent years patient admissions have decreased in length, with patients discharged into the community at an earlier point in their recovery. Whilst in some respects this reduces the risk of acquiring an infection (the patient is not in hospital so long) it is not clear by how much the risk is reduced. The patient may have been exposed to many of the key risk factors during their short admission. In this study 29% of patients acquired an infection within the first four days of their admission and over 50% within the first six days. Consequently, it would appear that many of the major risks are encountered during the first few days. As such a policy of early

discharge may result in an increased likelihood of the infection presenting postdischarge. In such cases this may result in additional cost to the health sector in the community, and family and friends. In other cases, where costs are not increased, it may represent unmet need.

Estimation of the burden to these other areas was beyond the scope of this work. When considering the results of this work and their implication for the health service, costs falling on these other areas should be borne in mind.

11.4.11 The study was limited to the estimation of the economic burden of HAI

This study was primarily a cost of illness study and as such represents what Drummond refers to as a partial economic evaluation.¹⁶² Estimation of the benefits of investing in activities that are directed towards preventing these infections was beyond the scope of this study. As such the results cannot be used to directly inform policy on the level of investment in infection control that should be made. However, the results do serve to highlight the magnitude of the burden these infections place on scarce hospital resources, and the gross benefits of prevention should these infections be prevented, in terms of the average value of resources that would be released if an infection was Furthermore, as detailed in Chapter 10 simple models can be prevented. developed from which estimates of the burden of HAI in a defined population, and the estimated gross benefits that would arise if a proportion were prevented, can be derived. More detailed models which utilise additional data on the cost and effectiveness of prevention activities, can subsequently be developed to estimate the net benefits of prevention. The results of such models can be used to inform and contribute to decision making regarding the allocation of resources to infection control and the nature of infection control programmes.

11.5 Methodological considerations

The methods employed in this study raised a number of practical difficulties, some of which are discussed in the following sections.

11.5.1 Recruitment

The recruitment of patients into this study presented a number of difficulties. Patients were recruited into the study by a research assistant, who adopted the following procedure. The research assistant would explain the study to the patient, ask the patient if they would be willing to participate in the study, and subsequently leave them with an information sheet and return at a convenient time to see if they were willing to participate. The main problem associated with this procedure was the fact that the research assistant had to introduce the topic of HAI at a time when frequently patients were anxious about their admission and proposed treatment. Many patients were unaware of the problem of HAI, and as such were rather alarmed to hear that whilst in hospital they might acquire an infection. This recruitment 'problem' was acknowledged at the outset and first experienced when conducting the pilot study. In order to overcome this and reduce the anxiety that patients may experience a number of steps were taken. Measures were taken to ensure that all ward staff and medical staff were aware of the study; posters were placed in all wards informing patients and relatives that a study was taking place; the study received some media coverage at its official launch, thus raising awareness of the issue; and the research assistants all received training in communication skills and recruitment procedures from myself and importantly from a trained communication expert. The aim was to equip the research assistants with the necessary skills to manage the recruitment process in a way that minimised anxiety yet kept patients informed of all relevant factors. It was important that all steps were taken to ensure that patients could make an informed choice about whether to participate in the study, whilst at the same time not causing undue anxiety and stress. These procedures appeared to be effective. Whilst it did not eliminate the difficulties associated with this procedure, the measures

were successful in helping and equipping the research assistants with the skills necessary to carry out this task.

11.5.2 Difficulties associated with conducting research in a clinical context

Conducting research in a clinical context presented a number of practical difficulties. These difficulties primarily related to the relationship between the patient and research assistant. All the research assistants were qualified nurses, a factor that whilst helping them to fulfil their role, also presented its own difficulties and conflicts.

As discussed in section 11.5.1 there were a number of difficulties associated with recruiting patients, in particular problems relating to raising the issue of the possibility that the patient might acquire an infection in hospital. The most recent UK prevalence study indicated that at any one time an estimated nine per cent of hospital in-patients have an infection that they acquired whilst in hospital.⁷ For many patients this proved to be quite an alarming statistic.

In addition to these problems, there was the issue of confidentiality. Patients were assured that any information that they provided would be treated as confidential. A standards document was drafted which outlined procedures with respect to the handling and storage of patient data which all the research assistants and other members of the team were required to read and sign up to prior to the start of data collection (see Appendix 2). However, occasionally situations arose when a patient would tell the research assistant something directly relevant to their condition and circumstances, which ideally medical and nursing staff needed to know. This placed the research assistant in a difficult position. The research assistant was obliged to maintain the patients confidentiality. However, at the same time the research assistant was acutely aware of the need to inform nursing and medical staff. Such situations had to be handled with great care and sensitivity. In all cases the research assistant successfully persuaded the patient through reasoned discussion that they

should inform a member of medical or nursing staff. Additional training was given to the research assistants to help with situations such as these.

As indicated above the research assistants were all qualified nurses, a fact which undoubtedly helped them in their role. However, from their perspective, this often gave rise to conflict between their previous nursing role and their newly acquired research role. Support was given to help them to adapt to this different role, and develop boundaries that enabled them to fulfil the requirements of their research assistant role.

11.5.3 Quantity of data required.

The data requirements of this study were large. The research assistants were required to recruit patients into the study, obtain base line case-mix and demographic data, and then collect detailed data on the resources used by individual patients each day, whilst at the same time undertaking surveillance for all types of infection. In addition to this, although not related to the analysis presented in this thesis, they were also required to collect data from primary health care records. In order to collect this data successfully and ensure complete data sets, the research assistants were unable to recruit as many patients as they had wished. This trade off between data quality and number of patients recruited was discussed on many occasions. At all times the priority was to ensure full data sets.

11.6 Implications of results for policy

The results of this study have important implications for policy and practice. The results reinforce the findings of earlier studies that these infections affect a considerable number of people and place a substantial burden on scarce hospital resources. Whilst it is acknowledged that inclusion of just one hospital may be viewed as a limitation of the study design, for the reasons discussed in section 11.3.5 the results of this study are likely to adequately reflect the ratio of increase in resource use and costs incurred by infected compared to uninfected adult patients admitted to similar specialties within other NHS facilities, the

additional costs incurred and number of extra days infected patients remain in hospital, and to some extent the incidence of HAI.

The results of this study not only provide important data on the incidence of HAIs and the burden infections place on limited health sector resources, but also the gross benefits that might result if infections were prevented: that is the value of resources that might be released for alternative use if infections are prevented. Whilst not all HAIs are likely to be preventable, there being what Ayliffe¹⁶³ described as an 'irreducible minimum' it is clear that a proportion can be prevented. The results of a National Audit Office survey provide some insights into the proportion of infections that infection control teams consider could be prevented through improvements in infection control. The proportion of infections considered preventable varied between hospitals from a low of 5% to over 35%, with a bed weighted average of 15%.³ The results presented in Chapter 10 indicate that if the results of this study are extrapolated to all adult non-day case patients admitted to the specialties covered in this study at the study hospital, it follows that a 15% reduction in rates would result in the prevention of 87 (95% CI: 75, 100) HAIs and the release of resources valued at £183,607 (95% CI: £134,761, £232,452) for alternative use. These released resources would include 646 (95% CI: 459, 834) bed days freed up for alternative use. Applying the results to a wider population of all adult non-day case patients admitted to similar specialties at other NHS hospitals throughout England in 1994/5, indicated that a 15% reduction in rates would result in the prevention of 23,238 (95% CI: 20,104, 26,691) HAIs and the release of resources valued at £49.17 million (95% CI: £36.09, £62.25 million) for alternative use. This estimate includes 173,059 (95% CI: 122,853, 223,265) bed days released for alternative use.

These results clearly demonstrate that prevention of infections would result in the release of considerable hospital resources for alternative use. These benefits may be considered to be the gross benefits of prevention, with net benefits dependent on the cost of achieving a reduction in rates. Estimation of

330

the net benefits of investment in infection control was beyond the scope of this work. However, as indicated in Chapter 10, the results of this study can be incorporated into economic models, together with data on the cost of infection control practices, to derive estimates of the net benefits of investing in infection control given various levels of effectiveness. The information generated by these models can subsequently be used to assist decision making regarding the allocation of resources to interventions to prevent and control infection. Models can be developed for a range of different activities. For example, models can be developed that assess the net benefits of investing in a comprehensive infection control programme, employing an additional infection control nurse, or investing in a particular piece of equipment thought to reduce the risk of infection. These models may relate to the prevention of all types of HAI or specific types of a HAI in a selected patient group or a wider patient population.

The results of the analysis that examined independent risk factors for HAI also have important implications for policy and practice. The results indicated that the risk of acquiring an HAI varied with both intrinsic and extrinsic risk factors. For example, the results of the analysis that examined independent risk factors for UTIs indicated that there were a number of intrinsic risk factors including sex (women had a greater risk of acquiring a UTI than men) and increasing age (the risk increased with age). The analysis also identified an extrinsic risk factor: the presence of a urinary catheter prior to the onset of infection. The presence of a urinary catheter was found to increase the risk of acquiring a UTI by almost three fold. Whilst it is acknowledged that the analysis was limited to a few selected potential risk factors for which data were available, the results still have important implications for policy and practice.

Information on independent intrinsic and extrinsic risk factors provides important data about which patients might be at greater risk of acquiring an infection. This information can be used by nursing and medical staff when planning and implementing care. For example, as indicated above the presence of a urinary catheter was found to significantly increase the risk of acquiring a UTI, thus

reinforcing the findings of earlier studies.¹ ¹³⁴ ¹³⁷ ¹³⁸ ¹⁴³ ¹⁴⁴ In terms of their implications for policy and practice, the results point to the need to question the need for catheterisation in individual patients. The results of an audit of infection control policies and practices in 19 hospitals in England and Wales indicated that catheterisation rates in patients admitted to similar specialties at these 19 hospitals varied considerably.¹ For example, amongst gynaecology patients, the catheterisation rate varied from 21% to 72%. Whilst some of this variation in rates may reflect case mix differences, such differences are unlikely to account for all the variation observed. Thus since catheterisation is a significant risk factor, the decision about whether to catheterise a patient should be taken carefully. When in use, care should be taken to minimise the risk of infection through the use of an appropriate catheter and appropriate catheter care. Where this involves additional resources, models can be developed that can provide important data on the costs and benefits of investment in a particular strategy and the results used to inform decision-making. In Chapter 10 an illustrative model was presented which demonstrated that investment in silver-alloy coated catheters resulted in positive returns at relatively low levels of effectiveness. Finally the results can be used to stimulate further research into the prevention of infections. That is the identification of key risk factors may be used to focus research into areas that can reduce the risk of these key independent factors.

The results of this study therefore have a number of implications for policy and practice. The results highlight the magnitude of the problem of infection in terms of the number of patients affected and the costs that fall on the hospital sector; they represent the potential gross benefits of prevention and when used in conjunction with data on the cost of prevention strategies they can be used to estimate the potential net benefits of investment in infection control.

These findings are of relevance to policy makers and health care professionals working at different levels in the health service. By demonstrating the magnitude of the problem, the results may help to keep the problem of HAI on the Department of Health's policy agenda and through various initiatives also on the agenda of hospital trusts and health care professionals. The results may be used to raise awareness of the problem and when used in conjunction with data on the cost of prevention strategies may be used to inform decisions about how much should be invested in infection prevention and control and the nature of the programmes themselves.

11.7 Areas for future research

Several areas for future research were identified which would enable a more comprehensive understanding of the burden imposed by HAIs occurring in surgical patients. The focus of this study was the incidence of HAIs occurring in adult non-day case patients admitted to five surgical specialties common to most hospitals, and the economic burden these infections placed on the hospital sector as a result of additional in-patient care. Further work is now needed to assess the incidence of HAIs presenting after discharge from hospital and the economic burden infections presenting during the in-patient period and/or post-discharge place on the primary and secondary health care sector following discharge from hospital.

Another area for future research concerns the estimation of the benefits of investment in infection control activities. The research reported in this thesis was limited to the assessment of the economic burden HAIs placed on the secondary health sector as a result of additional in-patient care. As such the results presented represent the gross benefits of prevention. Net benefits will be dependent on the cost of effective infection control activities. Assessment of the net benefits of investment in specific infection control activities was beyond the scope of this study. However, a framework for assessing the net benefits of investment in specific prevention activities was presented in Chapter 10. This modelling framework needs to be further developed to establish the benefits of investment in various infection control activities, the results of which may be used to inform decisions regarding the allocation of funding and the nature of infection prevention activities.

Other areas for future research include the assessment of the incidence and cost of infections involving specific pathogens or groups of pathogens and the incidence of HAIs occurring in omitted patient groups and the economic burden imposed. Omitted surgical groups include patients admitted to the renal, cardiac, plastics, paediatrics, neonates and burns specialties; and day cases. The burden imposed by HAIs occurring in some of these omitted groups is likely to be substantial.

11.8 Conclusion

The problem of HAI has attracted considerable research interest in recent years. The results of this study add to the current body of knowledge providing detailed data on the incidence of HAI occurring in adult patients admitted to five surgical specialties common to most hospitals, risk factors for these infections and the economic burden these infections place on the hospital sector as a result of additional in-patient care. Overall 7.5% (95% CI: 6.4%, 8.6%) of adult patients admitted to selected surgical specialties of a district general hospital acquired one or more HAIs that presented in hospital. Independent risk factors for these infections were found to vary with site of infection. These infections were found to increase resource use. On average infected patients had a LOS 2.1 (95% CI: 1.8, 2.5) times that of uninfected patients and utilised resources valued at 2.3 (95% CI: 2.6, 3.0) times those used by uninfected patients. HAIs were estimated to prolong the hospital stay by 7.8 days (95% CI: 5.7, 10.0) and increase hospital costs by £2,254 (95% CI: £1,738, £2,770) per case. Whilst the results of this study reflect the experience of patients admitted to just one hospital site, for reasons discussed in section 11.3.5 it is likely that the results are generalisable to other heath care settings in the UK.

The results of the analysis that estimated the burden of HAIs occurring in patients admitted to the specialties covered in this study at other NHS hospital in England, provide important data on the magnitude of the problem at the National level. An estimated 154,920 (95% CI: 134,024 to 177,943) adult non-day case patients admitted to the specialties covered in this study at other NHS hospitals in England acquired one or more HAIs in 1994/5, utilising 1,153,726

334

(95% CI: 819,019 to 1,488,434) additional bed days and, as a result of additional in-patient care, cost the health sector an additional £327.77 million (95%CI: £240.58 to £414.98) with further costs borne by the primary health care sector, patients and carers.

The results of this study have important implications for policy and practice. The estimates demonstrate both the substantial burden these infections place on limited heath sector resources and at the same time the gross benefits of prevention. It is acknowledged that not all HAI can be prevented. However a proportion can. The results of a recent NAO survey of ICTs suggested that a 15% reduction in rates could be archived through improvements in infection control. If this were achieved then at the National level resources valued at an estimated £49.17 million (95% CI: £36.09 to £62.25 million) and 173,059 bed days (95% CI: 122,853 to 223,265) may be released for alternative use. These estimates represent the potential gross benefits of prevention. The net benefits will depend on the cost and effectiveness of infection control activities. Estimation of these net benefits was beyond the scope of this study. However a framework for assessing net benefits was presented and it is concluded that after taking into account the costs of prevention activities it is likely that in many cases a considerable level of resources would be released for alternative use.

The results of this study thus demonstrate that a substantial number of surgical patients acquire an infection in hospital and that these infections place a considerable burden on limited heath sector resources. The findings are of relevance to health care professionals and policy makers. They serve to raise awareness of the magnitude of the problem of HAI, and the potential benefits of prevention, and when used in conjunction with data on the costs of prevention strategies may be used to inform decisions about how much should be invested in infection prevention and control and the shape of infection control programmes. If improvements in infection control, and a reduction in rates follow, this is likely to result in the release of considerable resources for alternative use and most importantly improved patient outcomes.

REFERENCES

- Glynn A, Ward V, Wilson J, Charlett A, Cookson B, Taylor L, et al. Hospital Acquired Infection: Surveillance Policies and Practice. London: Public Health Laboratory Service, 1997.
- Plowman R, Graves N, Griffin M, Roberts JA, Swan AV, Cookson B, et al. The socio-economic burden of hospital acquired infection. London: Public Health Laboratory Service, 1999.
- Report of the Comptroller and Auditor General. The Management and Control of Hospital Acquired Infection in Acute NHS Trusts in England. London: The Stationery Office Limited, 2000.
- Forty-second report by the Committee of Public Accounts The management and control of hospital acquired infection in Acute NHS Trusts in England: The Stationery Office Limited, 2000
- 5. Taylor K, Plowman R, Roberts J. *The challenge of hospital acquired infection*. London: The Stationery Office, 2001.
- Meers PD, Ayliffe GA, Emmerson AM, Leigh DA, Mayon-White RT, Mackintosh CA, et al. Report of the national survey of infection in hospitals.1980. Journal of Hospital Infection 1981;2(suppl):1-11.
- Emmerson AM, Entstone J, Griffin M, Kelsey MC, Smyth ETM. The second national prevalence survey of infection in hospitals - Overview of the results. *Journal of Hospital Infection* 1996(32):175-190.

- Plowman R, Graves N, Griffin M, Roberts JA, Swan AV, Cookson B, et al. The rate and cost of hospital acquired infections occurring in patients admitted to selected specialties of a district general hospital in England and the national burden imposed. *Journal of Hospital Infection* 2001;47(3):198-209.
- 9. Department of Health and Public Health Laboratory Service. Hospital Infection Control: Guidance on the control of infections in hospital. Prepared by the Hospital Infection Working Group of the Department of Health and Public Health Laboratory Service. London: Department of Health, 1995.
- Fabry J, Meynet R, Joron M, Sepetjan M, Lambert D, Guillet R. Cost of nosocomial infections: analysis of 512 digestive surgery patients. World Journal of Surgery. 1982;6(3):362-365.
- Haley RW. Managing hospital infection control for cost-effectiveness: a strategy for reducing infectious complications. Chicago: American Hospital Publishing, 1986.
- Haley R. Incidence and nature of endemic and epidemic nosocomial infections. In: JV. B, Brachman P, editors. *Hospital Infections*. Boston: Little, Brown & Co., 1985.
- 13. Davies T, Cottingham J. The cost of Hospital Infection in Orthopaedic Patients. *Journal of Infection* 1979;1:330-338.
- Coello R, Glenister H, Fereres J, Bartlett C, Leigh D, Sedgwick J, et al. The Cost of Infection in Surgical patients: A Case Control Study. *Journal* of Hospital Infection 1993;25:239-250.

- 15. DH/PHLS. Hospital Infection Control: Guidance on the control of infections in hospitals. London: PHLS, 1988.
- Bernander S, Hambraeus A, Myrback KE, Nystrom B, Sundelof B. Prevalence of hospital associated infections in five Swedish hospitals in November 1975. Scandinavian Journal Infectious Disease 1978;10(1):66-70.
- Jepsen OP, Mortensen N. Prevalence of nosocomial infections and infection control in Denmark. *Journal of Hospital Infection* 1980; 1:237-244.
- 18. Hovig B, Lystad A, Opsjon H. A prevalence survey of infections amongst hospitalised patients in Norway. *Norwegian Institute of Public Health Annals* 1981;4:49-60.
- 19. EPINE Working Group. Prevalence of hospital-acquired infections in Spain. *Journal of Hospital Infection*. 1992;1992(20):1-13.
- 20. Sartor C, Sambuc R, Birnar MC, Gulian C, De Micco P. Prevalence surveys of nosocomial infections using a random sampling method in Marseille hospitals. *Journal of Hospital Infection* 1995;29:209-216.
- 21. Vaque J, Rossello J, Arribas JL, Group. EW. Prevalence of nosocomial infections in Spain: EPINE study 1990 -1997. *Journal of Hospital Infection* 1999;43(Suppl):S105 S111.
- 22. Sramova H, Bartonova A, Bolek S, Krecmerova M, Subertova V. National prevalence survey of hospital-acquired infections in Czechoslovakia. *Journal of Hospital Infection* 1988;11:328-334.

- Centre for Disease Control. Outline for Surveillance and Control of Nosocomial Infections. US Department of Health and Human Services -Public Health Service. 1972;Appendix II.
- 24. Moro ML, Stazi MA, Marasca D, Greco D, Zampieri A. National prevalence survey of hospital-acquired infections in Italy, 1983. *Journal of Hospital Infection* 1986;8:72-85.
- 25. Mertens R, Kegels G, Stroobant A, *et al.* The national prevalence survey of nosocomial infections in Belgium. *Journal of Hospital Infection* 1987;9:219-229.
- 26. World Health Organisation Surveillance, control and prevention of hospital acquired (nosocomial) infections. Report of an Advisory Group. Geneva, 1981.
- 27. McLaws ML, Gold J, King K, Irwig LW, Berry G. The prevalence of nosocomial and community-acquired infections in Australian hospitals. *Medical Journal of Australia* 1988;149:582-590.
- 28. Garner J, Jarvis W, Emori T, Horan T, Hughes J. CDC definitions for nosocomial infections, 1988. *American Journal of Infection Control* 1988;16:128-140.
- 29. Horan TC, Haynes RP, Martone WJ, Jarvis WR, Emori TG. CDC definitions of nosocomial surgical site infections, 1992: A modification of CDC definitions of surgical wound infections. *Infection Control and Hospital Epidemiology* 1992;13(10):606-608.
- 30. Avistland P, Stormark M, Lystad A. Hospital-acquired infections in Norway: a national prevalence survey in 1991. *Scandinavian Journal of Infectious Disease* 1992;24:477-483.

- Gikas A, Pediaditis I, Roumbelaki M, Troulakis G, Romanos J, Tselentis
 Y, et al. Repeated multi-centre prevalence surveys of hospital-acquired infection in Greek hospitals. *Journal of Hospital Infection* 1999;41:11-18.
- 32. Gastmeier P, Kampf G, Wischnewski N, Hauer T, Schulgen G, Schumacher M, et al. Prevalence of nosocomial infections in representative German Hospitals. *Journal of Hospital Infection* 1998;38:37-49.
- 33. Pittet D, Harbarth S, Ruef c, Francioli P, Sudre P, Petignat C, et al. Prevalence and risk factors for nosocomial infections in four university hospitals in Switzerland. *Infection Control and hospital epidemiology* 1999;20(1):37-42.
- 34. Astagneau P, Fleury L, Leroy S, Lucet J-C, Golliot F, Brucker G. Cost of antimicrobial treatment for nosocomial infections based on a French prevalence survey. *Journal of Hospital Infection* 1999;42:303-312.
- 35. The French Prevalence Survey Study Group. Prevalence of nosocomial infections in France: results of the nationwide survey in 1996. *Journal of Hospital Infection* 2000;46:186-193.
- 36. Scheel O, Stormark M. National prevalence survey on hospital infections in Norway. *Journal of Hospital Infection* 1999;41:331-335.
- 37. Pavia M, Bianco N, Viggiani NMA, Angelillo IF. Prevalence of hospitalacquired infections in Italy. *Journal of Hospital Infection* 2000;44:135-139.

- 38. Gikas A, Pediaditis J, Papadakis J, Starakis J, Levidiotou S, Nikolaides P, et al. Prevalence study of hospital-acquired infections in 14 Greek hospitals: planning from the local to the national surveillance level. *Journal of Hospital Infection* 2002;50(4):269-275.
- 39. Christensen M, Jepsen OB. Reduced rates of hospital-acquired UTI in medical patients. Prevalence surveys indicate effect of active infection control programmes. *Journal of hospital infection* 2001;47:36-40.
- 40. Mayon-White R, Ducel G, Kereselidze T, Tikomirov E. An International Survey of the Prevalence of Hospital Acquired Infection. *Journal of Hospital Infection* 1988;11(supplement A):43-48.
- 41. Gastmeier P, Schumacher M, Daschner F, Ruden H. An analysis of two prevalence surveys of nosocomial infection in German intensive care units. *Journal of Hospital Infection* 1997;35:97-105.
- 42. Gastmeier P, Kampf G, Wischnewski N, Schumacher M, Daschner F, Ruden H. Importance of the surveillance method: National prevalence studies on nosocomial infections and the limits of comparison. *Infection Control and Hospital Epidemiology* 1998;19(9):661-667.
- 43. France G. Cross-boarder flows of Italian patients within the European Union. *European Journal of Public Health* 1997;7((suppl)):1-41.
- Vincent J-L, Bihari DJ, Suter PM, Bruining HA, White J, Nicolas-Chanoin M-H, et al. The prevalence of nosocomial infection in intensive care units in Europe: Results of the European Prevalence of Infection in Intensive Care (EPIC) study. Journal of the American Medical Association 1995;274(8):639-644.

- 45. Sax H, Hugonnet S, Harbarth S, Herrault P, Pittet D. Variation in nosocomial infection prevalence according to patient care setting: a hospital-wide survey. *Journal of Hospital Infection* 2001;48:27-32.
- 46. Tess BH, Glenister HM, Rodrigues LC, Wagner MB. Incidence of hospital-acquired infection and length of hospital stay. *European Journal Clinical Microbiology Infectious Disease*. 1993;12(2):81-86.
- 47. Glenister HM, Taylor L, Bartlett CLR, Cooke EM, Mackintosh CA. An 11 month incidence study of infections in wards of a district general hospital. *Journal of Hospital Infection* 1992;21(4):261-273.
- 48. Haley R, W., Schaberg D, Von Allmen S, McGowan J. Estimating the extra charges and prolongation of hospitalization due to nosocomial infections: A comparison of methods. *Journal of Infectious Diseases* 1980;141:2:248-257.
- 49. Haley R, W., Culver DH, White J, W., Morgan W, M., Emori T, G. The nationwide nosocomial infection rate. A need for vital new statistics. *American Journal of Epidemiology* 1985;121:159-167.
- 50. Freeman J, McGowan Jnr J. Methodological issues in hospital epidemiology 1: Rates, case-finding, and interpretation. *Review Infectious Diseases* 1981;3(4):658-667.
- 51. Moir-Bussy BR, Hutton RM, Thompson JR. Wound Infection after Caesarean Section. *Journal of Hospital Infection* 1984;5:359-370.
- 52. Leigh DA, Emmanuel FXS, Sedgwick J, Dean R. Post-operative urinary tract infection and wound infection in women undergoing Caesarean section: a comparison of two study periods in 1985 and 1987. *Journal of Hospital Infection* 1990;15:107-116.

- 53. Henderson E, Love EJ. Incidence of hospital-acquired infections associated with caesarean section. *Journal of Hospital Infection* 1995;29:245-255.
- 54. Hillan E. Postoperative morbidity following caesarean delivery. *Journal Advanced Nursing* 1995(22):1035 1042.
- 55. Nice C, Feeney A, Godwin P, Mohanraj M, Edwards A, Bladwin A, *et al.* A prospective audit of wound infection rates after caesarean section in five West Yorkshire hospitals. *Journal of Hospital Infection* 1996;33(1):55-61.
- Berland M, Brochery P, Croidieu C, Bongain A, Masson A, Minguet C, et al. Hospital-acquired infections in French maternity units: trends noted in 2-year surveillance (1997-1998). Journal of Hospital Infection 2000:334-335.
- 57. Meltomaa S, Makinen J, Taalikka M, Helenius H. Incidence, risk factors and outcome of infection in a 1-year hysterectomy cohort: a prospective follow-up study. *Journal of Hospital Infection* 2000;45:211-217.
- 58. Costantini M, Donisi PM, Turrin MG, Diana L. Hospital-acquired infections surveillance and control in intensive care services. Results of an incidence study. *European Journal of Epidemiology*. 1987;3(4):347-355.
- 59. Gravel-Tropper D, Oxley C, Memish Z, Garber GE. Underestimation of surgical site infection rates in obstetrics and gynaecology. *American Journal of Infection Control* 1995;23(1):22-26.
- 60. Leigh DA. An eight year study of post-operative wound infection in two district general hospitals. *Journal of Hospital Infection* 1981;2:207-217.

343

- 61. Erbaydar S, Akgun A, Eksik A, Erbaydar T, Bilge O, Bulut A. Estimation of increased hospital stay due to nosocomial infections in surgical patients: comparison of matched groups. *Journal of Hospital Infection* 1995;30:149-154.
- 62. Cruse P, Foord R. The epidemiology of wound infection: a 10 year prospective study of 62,939 wounds. *Surgical Clinics of North America* 1980;60:27-40.
- 63. National Nosocomial Infections Surveillance (NNIS) System report data summary from January 1990 May 1999, issued June 1999. *American Journal Infection Control* ;27:520-532.
- 64. Abussaud M, Meqdem M. A study of some risk factors associated with wound infection. *Journal of Hospital Infection* 1986;8:300-304.
- 65. Lynch W, Davey P, Malek M, Byrne D, Napier A. Cost-effectiveness analysis of the use of chlorhexidine detergent in pre-operative wholebody disinfection in wound infection prophylaxis. *Journal of Hospital Infection* 1992; 21:179 -191.
- 66. Malone D, Genuit T, Tracy J, Gannon C, Napoliyano L. Surgical site infections: reanalysis of risk factors. *Journal of Surgical Research* 2002;103(1):89-95.
- 67. Culver Dea. Surgical wound infection rates by wound class, operative procedure, and patient risk index. *American Journal of Medicine* 1991;91(suppl 3B):152S-157S.

- 68. Kirkland K, Briggs J, Trivette S, Wilkinson W, Sexton D. The impact of surgical-site infections in the 1990s: attributable mortality, excess length of hospitalization, and extra costs. *Infection Control and Hospital Epidemiology* 1999;20:725-730.
- 69. Rubenstein E, Green M, Modan M, Amit P, Bernstein L, Rubenstein A. The Effect of Nosocomial Infections on the Length and Costs of Hospital Stay. *Journal of Anti-microbial Chemotherapy* 1982;9 (suppl A):93-100.
- 70. Coles B, Van Heerden J, Keys T, Haldorson A. Incidence of wound infection for common general surgical procedures. *Surgery, Gynaecology and Obstetrics* 1982;154(557-560).
- 71. Krukowski Z, Stewart M, Alsayer H, Matheson N. Infection after abdominal surgery. *British Medical Journal* 1984;288:278 -280.
- 72. Bremmelgaard A, Raahave D, Beier-Holgersen R, Penderson J, Anderson S, Sorenson A. Computer aided surveillance of surgical infections and identification of risk factors. *Journal of Hospital Infection* 1989;13:1-18.
- 73. Garibaldi RC, D. Lerer, T. Risk Factors for Postoperative Infection. American Journal of Medicine 1991;91(suppl 3B):158S-163S.
- 74. Mishriki SF, Law DJW, Jeffrey PJ. Factors affecting the incidence of postoperative wound infection. *Journal of Hospital Infection* 1990;16:223-230.
- 75. Reilly JS, Baird D, Hill R. The importance of definitions and methods in surgical wound infection audit. *Journal of Hospital Infection* 2001;47:64-66.

- 76. Bruce J, Russell EM, Mollison J, Krukowski ZH. The quality of measurement of surgical wound infection as the basis for monitoring: a systematic review. *Journal of Hospital Infection* 2001;49:99-108.
- 77. Glenister HM, Taylor LJ, Cooke EM, Bartlett CLR. A Study of Surveillance Methods for Detecting Hospital Infection. London: Public Health Laboratory Service, 1992.
- 78. Peel AL, Taylor EW. Proposed definitions for the audit of postoperative infections: a discussion paper. Surgical Infection Study Group. *Journal of Hospital Infection* 1991;49:99-108.
- 79. Ayliffe GA, Casewell MW, Cookson BD, *et al.* National prevalence survey of hospital acquired infections: A preliminary report of the Steering Group of the second National Prevalence Survey. *Journal of Hospital Infection* 1993;24:69-76.
- 80. Horan TC, Culver DH, Gaynes RP, Jarvis WR, Edwards JR, Reid CR. Nosocomial infections in surgical patients in the United States January 1986-June 1992. Infection control and hospital epidemiology 1993;14,2:73-80.
- Haley RW, Hooton TM, Culver DH, Stanley RC, Emori TG, Hardison CD, et al. Nosocomial infections in US hospitals, 1975-1976. Estimated frequency by selected patient characteristics of patients. *American Journal of Medicine* 1981;70:947-59.
- 82. Krieger JN, Kaiser DL, Wenzel RP. Urinary tract etiology of bloodstream infections in hospitalised patients. *Journal of Infectious Diseases* 1983;148:57-62.

- Bueno-Cavanillas A, Rodriguez-Contreras R, Delgado- Rodriguez M, Moreno-Abril O, Lopez-Gigosos R, Guillen-Solvas J, et al. Preoperative stay as a risk factor for nosocomial infection. European Journal of Epidemiology 1991;7(6):670-676.
- 84. Hughes J, Culver D, White J, *et al.* Nosocomial infection surveillance. *Morbidity and Mortality Weekly Report* 1983;32(Suppl 4):1-15.
- 85. Horan T, White J, Jarvis W, et al. Nosocomial infection surveillance, 1984. Morbidity and Mortality Weekly Report 1986;35(17SS-29SS).
- 86. Kollef M. Ventilator-associated pneumonia. A multivariate analysis. Journal of American Medical Association 1993;270(1965-1970).
- Craven D, Kunches L, Kilinski V, Lichtenberg D, Make B, McBabe W. Risk factors for pneumonia and fatality in patients receiving continuous mechanical ventilation. *American Review Respiratory Disease* 1986;133:792-796.
- 88. Craven D, Driks M. Nosocomial pneumonia in the intubated patient. Semin Respir Med 1987;2(20-33).
- 89. Craven D, Kunehes L, Litchenberg D, et al. Nosocomial infection and fatality in medical surgical intensive care unit patients. Archives of Internal Medicine 1988;148:1161 1168.
- 90. Craven D, Steger K. Epidemiology of nosocomial pneumonia: new perspectives on an old disease. *Chest* 1995;108:1S-16S.
- 91. Leu H, Kaiser D, Mori M, Woolson R, Wenzel RP. Hospital acquired pneumonia. Attributable mortality and morbidity. *American Journal of Epidemiology* 1989;129(6):1258 1267.

- 92. Fagon J, Chastre J, Vaugnat A, Troillet J, Novara A, Gibert C. Nosocomial pneumonia and mortality among patients in intensive care units. *Journal of the American Medical Association* 1996;275(11):866 -869.
- 93. Torres A, Aznar R, Gatell J, al e. Incidence, risk, and prognosis factors of nosocomial pneumonia in mechanically ventilated patients. *American Review Respiratory Disease* 1990;142(523-528).
- Cellis R, Torres A, Gatell J, Almda M, Rodriguez-Rosain R, Agusti-Vidal A. Nosocomial pneumonia. A multivariate analysis of risk and prognosis. *Chest* 1988;93(318-324).
- 95. Nosocomial Infection National Surveillance Scheme. Surveillance of hospital-acquired bacteraemia in English Hospitals 1997-1999. London:
 Public Health Laboratory Service, 2000.
- 96. Nosocomial Infection National Surveillance Scheme. Surveillance of surgical site infection in English Hospitals 1997-1999. London: Public Health Laboratory Service, 2000.
- 97. Shapiro M, Munoz A, Tager I, Schoenbaum S, Polk B. Risk factors for infection at the operative site after abdominal or vaginal hysterectomy. *New England Journal of Medicine* 1982;307(1661-1666).
- Law D, Mishriki S, Jeffrey P. The importance of surveillance after discharge from hospital in the diagnosis of post-operative wound infection. *Annals of Royal College of Surgeons in England* 1990;72:207-209.

- 99. Platt R, Zaleznik D, Hopkins C, Dellinger E, Karchmer A, Bryan C, et al. Perioperative antibiotic prophylaxis for hernioraphy and breast surgery. New England Journal of Medicine 1990;322:153-60.
- Weigelt JD, D. Haley, RW. The necessity and efficacy of wound surveillance after discharge. Archives of Surgery 1992;127(January):77-82.
- 101. Bailey I, Karran S, Toyn K, Brough P, Ranaboldo C, Karran S. Community surveillance of complications after hernia surgery. *British Medical Journal* 1992;304:469-471.
- 102. Medina-Cuadros M, Sillero-Arenas M, Martinez-Gallego G, Delgado-Rodriguez M. Surgical wound infections diagnosed after discharge: Epidemiological differences with in-hospital infections. American Journal of Infection Control 1996;24(6):421-428.
- 103. Noel I, Hollyoak V, Galloway A. A survey of incidence and care of postoperative wound infections in the community. *Journal of Hospital Infection* 1997;36:267-273.
- 104. Byrne D, Lynch W, Napier A, Davey P, Malek M, Cuschieri A. Wound infection rates: the importance of definition and post-discharge wound surveillance. *Journal of Hospital Infection* 1994;26:26: 37-43.
- 105. Reimer K, Gleed C, Nicolle L. The impact of postdischarge infection on surgical wound infection rates. *Infection Control.* 1987;8:237-240.
- 106. Holtz T, Wenzel R. Postdischarge surveillance for nosocomial wound infection. A brief review and commentary. *American Journal of Infection Control* 1992;20(4):206-213.

- 107. Plowman R, Graves N, Roberts JAR. *Hospital Acquired Infection*. London: Office of Health Economics, 1997.
- 108. Rose R, Hunting K, J., Townsend T, R., Wenzel R, P. Morbidity/Mortality and Economics of Hospital-Acquired Bloodstream Infections: A Controlled Study. *Southern Medical Journal* 1977;70;11:1267-1269.
- 109. Sprengler R, Greenough III W. Hospital costs and mortality attributed to nosocomial bacteremias. *Journal of the American Medical Association* 1978;240(22):2455 2458.
- 110. Daschner F, Nadjem H, Langmaack H, Sandritter W. Surveillance, prevention and control of hospital-acquired infections. III. Nosocomial infections as cause of death: retrospective analysis of 1000 autopsy reports. *Infection* 1978;6:261-5.
- 111. Freeman J, Rosner BA, McGowan JE. Adverse Effects of Nosocomial Infection. *Journal of Infectious Diseases* 1979;140,5:732-740.
- 112. Platt R. Mortality associated with nosocomial urinary tract infection. *The New England Journal of Medicine* 1982;307:637.
- 113. Pittet D, Tarara D, Wenzel R, P. Nosocomial Bloodstream Infection in critically III Patients. Excess length of stay, extra costs and attributable mortality. *Journal of the American Medical Association* 1994;271:1598-1601.
- 114. Dinkel R, Lebok U. A survey of nosocomial infections and their influence on hospital mortality rates. *Journal of Hospital Infection* 1994;28(4):297-304.

- 115. Salemi C, Morgan J, Padilla S, Morrissey R. Association between severity of illness and mortality from nosocomial infection. *American Journal of Infection Control* 1995;23(3):188 - 193.
- 116. Garcia-Martin M, Lardelli-Claret P, Bueno-Cavanillas A, Luna-del-Castillo J, Espigares-Garcia M, Galvez-Vargas R. Proportion of hospital deaths associated with adverse events. *Journal of Clinical Epidemiology* 1997;50(12):1319-1326.
- 117. Girou E, Stephan F, Novara A, Safar M, Fagon J. Risk factors and outcome of nosocomial infections: results of a matched case-control study of ICU patients. *American Journal Respiratory Critical Care Medicine* 1998;157:1151-1158.
- 118. Delgado-Rodriguez M, Gomez-Ortega A, Llorca J, Lecuona M, Dierssen T, Sillero-Arenas M, et al. Nosocomial infection, indices of intrinsic infection risk, and in-hospital mortality in general surgery. Journal of Hospital Infection 1999;41:203-211.
- 119. Gross PA, Neu HC, Aswapokee P, Antwerpen CV, Aswapokee N. Deaths from nosocomial infections: Experience in a University Hospital and a community hospital. *The American Journal of Medicine* 1980;68:219-223.
- 120. Soufir L, Timsit J, Mahe C, Carlet J, Regnier B, Chervet S. Attributable morbidity and mortality of catheter-related septicaemia in critically ill patients: a matched, risk-adjusted, cohort study. *Infection control Hospital Epidemiology* 1999;20:396-401.
- 121. Orsi G, Di Stefano L, Noah N. Hospital-acquired laboratory confirmed bloodstream infection: increased hospital stay and direct costs. *Infection Control and Hospital Epidemiology* 2002;23(4):190-197.

- 122. Martone W, Jarvis W, Culver D, Haley RW. Incidence and Nature of endemic and epidemic nosocomial infections. In: Bennett J, Brachman P, editors. Hospital Infections. Fourth Edition. Philadelphia: Lippincott-Raven Publishers, 1998:461-476.
- 123. Astagneau P, Rioux C, Golliot F, Brucker G for the INCISCO Networj Study Group. Morbidity and mortality associated with surgical site infections: results from the 1997-1999 INCISO surveillance. *Journal of Hospital Infection* 2001;48:267-274.
- 124. Martone W, J., Jarvis W, R., Culver DH, Haley R, W. Incidence and nature of endemic and epidemic nosocomial infections. In: Bennet J, V., Brachman P, S., editors. *Hospital Infections*. First edition. Boston: Little, Brown and Company, 1992:577-596.
- 125. Bryan C, Reynolds K. Hospital acquired bacteremic urinary tract infection: Epidemiology and outcome. *Journal of Urology* 1984;132:494-498.
- 126. Krieger BE, Craven DE, Carling PC, McCabe WR. Gram-negative bacteria. Reassessment of etiology, epidemiology and ecology in 612 patients. *American Journal of Medicine* 1980;68:332-355.
- 127. Gordan D, Bune A, Griffin B, *et al.* Diagnostic criteria and natural history of catheter associated urinary tract infections after prostatectomy. *Lancet* 1983;1:1269-1271.
- 128. Kolfe, MH. Ventilator-associated pneumonia: a multivariate analysis. Journal of the American Medical Association 1993;270:1965-1970.

- 129. Hughes J, et al. Mortality associated with nosocomial infections in the United States, 1975-1981. In Abstracts of the twenty-second interscience conference on antimicrobial agents in chemotherapy, Miami Beach, Florida, October 4-6, 1982. Washington DC: American Society for Microbiology, 1982; 189, 1982.
- 130. Inglis TJJ. *Microbiology and Infection*: Churchill Livingston, 1996.
- 131. Warren JW. Nosocomial urinary tract infections. In: Mendell GL, Douglas RG, Bennett JE, editors. *Principles and practices of infectious disease*. Third edition: Churchill Livingstone., 1990.
- 132. Christie A. Infectious Diseases: Churchill and Livingstone, 1987.
- 133. Northey D. Microbial surveillance in surgical intensive care unit. *Surgery Gynaecology and Obstetrics* 1974;139:321-326.
- 134. Garibaldi RA, Burke JP, Dickman ML, *et al.* Factors predisposing to bacteriuria during indwelling urethral catheterisation. *New England Journal of Medicine* 1974;291:215.
- 135. Cruse P. Infection surveillance: Identifying the problems and the high-risk patient. *Southern Medical Journal* 1977;70(suppl 1):4-8.
- 136. Simchen E, Shapiro M, Michel J, Sacks T. Multivariate analysis of determinants of post-operative wound infection: a possible basis for intervention. *Reviews of Infectious Diseases* 1981;3(678-682).
- 137. Shapiro M, Simchen E, Izareli S, *et al.* A multivariate analysis of risk factors for acquiring bactriuria in patients with indwelling urinary catheters for longer than 24 hours. *Infection Control* 1984;5:525-532.

- 138. Platt R, Polk BF, Murdoch B, *et al.* Risk factors for nosocomial urinary tract infection. *American Journal of Epidemiology* 1986;124:977-985.
- Valentine R, Weigelt J, Dryer D, Rodgers C. Effect of remote infections on clean wound infection rates. *American Journal of Infection Control* 1986;14(64-67).
- 140. Gorse G, Messner R, Stephens N. Association of malnutrition with nosocomial infection. *Infection Control and Hospital Epidemiology* 1989;10(5):194-203.
- 141. Roberts J, Bates T. The use of body mass index in studies of abdominal wound infection. *Journal of Hospital Infection* 1992;20:217-220.
- 142. Kampf G, Gastmeier P, Wischnewski J, Schlingmann J, Schumacher M, Daschner F, et al. Analysis of risk factors for nosocomial infections results from the first national prevalence survey in Germany (NIDEP study, part I). Journal of Hospital infection 1997;37:103-112.
- 143. Hartstein AI, Garber SB, Ward TT, *et al.* Nosocomial urinary tract infections: A prospective evaluation of 108 catheterised patients. *Infection Control* 1981;2:380-386.
- 144. Stamm WE. Catheter -Associated Urinary Tract Infections: Epidemiology, Pathogenesis, and Prevention. *The American Journal of Medicine* 1991;91(Suppl 3B):65S-71S.
- 145. Garibaldi RA, Mooney BR, Epstein BJ, Britt MR. An evaluation of daily bacteriologic monitoring to identify preventable episodes of catheterassociated urinary tract infection. *Infection Control* 1982;3:466-470.

- 146. Crow R, Mulhall A. Indwelling catheterisation and related nursing practice. *Journal of Advanced Nursing* 1988;13:489-496.
- 147. Kampf G, Wishnewski N, Schulgen G, Schumacher M, Daschner F. Prevalence and risk factors for nosocomial lower respiratory tract infections in German hospitals. *Journal of Clinical Epidemiology* 1998;51(6):495-502.
- 148. Rojo D, Pinedo A, Clavijo E, Garcia-Rodriguez A, Garcia v. Analysis of risk factors associated with nosocomial bactereamias. *Journal of Hospital Infection* 1999;42:135-141.
- 149. Haley R, Culver D, Morgan W, White J, Emori T, Hooton T. Identifying patients at high risk of surgical wound infection. *American Journal of Epidemiology* 1985;121(2):206-215.
- 150. Hayward C, Griffin G. Antibiotic Resistance: The current situation and the molecular mechanisms involved. *British Journal of Hospital Medicine* 1994(52/9):473-478.
- 151. Cohen M. Epidemiology of drug resistance: implications for a postantimicrobial era. *Science* 1992;257:1050-1055.
- 152. Duckworth G. Diagnosis and management of methicillin resistant Staphylococcus aureus infection. *British Medical Journal* 1993(307):1049
 - 1052.
- 153. Cetin E, Ang O. Staphylococci resistant to methicillin ('Celbenin'). British Medical Journal 1962;II:51-52.

- 154. House of Lords Select Committee of Science and Technology. Resistance to antibiotics, and other antimicrobial agents. London Stationery Office, 1998.
- 155. Standing Medical Advisory Committee Sub-group on Antimicrobial Resistance. The Path of Least Resistance. Department of Health, 1998.
- 156. Conference EU. 'The Microbial Threat': The Copenhagen Recommendations: Danish Ministry of Health and Ministry of Food and Agriculture and Fisheries, September 1998.
- 157. World Health Organisation. *Overcoming antibiotic resistance*: World Health Organisation, 2000.
- 158. Department of Health. UK Antimicrobial Resistance Strategy and Action Plan. London: Department of Health, 2000.
- 159. House of Lords Select Committee on Science and Technology. Third Report on Antibiotic Resistance Session 2000-2001.
- 160. Department of Health. Government's response to the House of Lords Select Committee on Science and Technology Report: Resistance to antibiotics, and other antimicrobial agents CM 4172. London: Stationary Office, 1999.
- Kunin C. Problems in antibiotic use: In: Mendel G, Douglas RJ, Bennett J, editors. *Principles and practices of infectious disease*: Churchill Livingstone Inc, 1990:427-434.
- Drummond MF, Stoddart GL, Torrance GW. Methods for the Economic Evaluation of Health Care Programmes. Oxford Medical Publications, 1987.

- 163. Ayliffe GAJ. Nosocomial infection the irreducible minimum. Infection Control. 1986;7(suppl 2):92-95.
- 164. Haley R, W., White J, W., Culver DH, Meade Morgan W, Emori T, G., Munn V, P., et al. The Efficacy of infection surveillance and control programmes in preventing Nosocomial Infections in US Hospitals. American Journal of Epidemiology 1985;121:182-205.
- 165. Fein R. On measuring economic benefits of health programmes. In: Medical History and Medical Care. A symposium of perspectives. London: Nuffield Provincial Hospital Trust, 1971:183-219.
- 166. Jefferson T, Demicheli V, Mugford M. *Elementary economic evaluation in health care*. Second edition. London: BMJ Publishing Group, 2000.
- 167. Drummond M. Cost-of-illness studies. A major headache? *PharmacoEconomics* 1992;2:1-4.
- 168. Byford S, Torhgerson D, Raftery J. Cost of illness studies. *British Medical Journal* 2000;320:1335.
- 169. Rice D. Cost of Illness Studies, fact or fiction. Lancet 1995;334:1519-20.
- Haley R, W. Cost benefit Analysis of Infection Control Activities. In: Brachman P, S., Bennett J, V., editors. *Hospital Infections*. Third edition. Boston: Little, Brown, 1992:507-532.
- 171. Scheckler WE. Hospital costs of Nosocomial infections: A prospective three month study in a community hospital. *Infection Control* 1978;1(3):150-152.

- 172. Haley R, W., Schaberg D, Crossley K, Von Allmen S, McGowan J. Extra Charges and Prolongation of Stay Attributable to Nosocomial Infections: A Prospective Interhospital Comparison. *American Journal of Medicine* 1981;70:51-58.
- 173. Vegas A, Jodra V, Garcia M. Nosocomial infection in surgery wards: a controlled study of increased duration of hospital stays and direct cost of hospitalization. *European Journal of Epidemiology* 1993;9:504-510.
- 174. Tambyah P, Knasinski V, Maki D. The direct costs of nosocomial catheter-associated urinary tract infection in the era of managed care. *Infection Control and Hospital Epidemiology* 2002;23(1):27-31.
- 175. Mugford M, Kingston J, Chalmers I. Reducing the incidence of infection after caesarean section: implications of prophylaxis with antibiotics for hospital resources. *British Medical Journal* 1989;299:1003-1006.
- 176. Poulsen K, B., Bremmelgaard A, Sorensen A, I., Raahave D, Petersen J,
 V. Estimated costs of postoperative wound infections. *Epidemiology of Infection* 1994;113:283-295.
- 177. Hyryla M, L,J., Sintonen H. The Use of Health Services in the management of Wound Infection. *Journal of Hospital Infection* 1994;26:1-14.
- 178. O'Donoghue M, Allen K. Costs of an outbreak of wound infections in an orthopaedic ward. *Journal of Hospital Infection* 1992;22:73-78.
- 179. Taylor G, Kirkland T, McKenzie M, Sutherland B, Wiens R. The effect of surgical wound infection on post-operative stay. Canadian Journal of Surgery 1995;38(2):149-153.

- 180. Whitehouse J, Friedman N, Kirkland K, Richardson W, Sexton D. The impact of surgical-site infections following orthopaedic surgery at a community hospital and a university hospital: adverse quality of life, excess length of stay, and extra costs. *Infection Control and Hospital Epidemiology* 2002;23(4):183-189.
- 181. Dixon R. Effect of Infections on Hospital Care. Annals of Internal Medicine 1978;89 (part2):749-753.
- 182. Kappstein I, Schulgen G, Beyer U, Geiger K, Schumacher M, Daschner FD. Prolongation of stay and extra cost due to ventilator-associated pneumonia in an intensive care unit. *European Journal of Clinical Microbiology and Infectious Disease* 1992;11(6):504-508.
- 183. Wey S, Mori M, Pfaller M, Woolson R, Wenzel R. Hospital-acquired candidemia. The attributable mortality and excess length of stay. *Archives of Internal Medicine* 1988;148(12):2642-2645.
- 184. Daschner F. Cost-effectiveness in hospital infection control lessons for the 1990s. *Journal of Hospital Infection*. 1989;13:325-336.
- 185. Elliston PRA, Slack RCB, Humphreys H, Emmerson AM. The Cost of Postoperative Wound Infection. *Journal of Hospital Infection* 1994;28(3):241-2.
- 186. Persson U, Montgomery F, Carlsson A, Lindgren B, Ahnfelt L. How far can prophylaxis against infection in total joint replacement offset its cost? *British Medical Journal* 1988;296:99-102.
- 187. McGowan JE. Cost and benefit a critical issue for hospital infection control. *American Journal of Infection Control* 1982;100-108.
- 188. McGowan JE. Cost and benefit in control of nosocomial infection: method for analysis. *Review of Infectious Diseases* 1981;3:790-797.
- 189. Wakefield DS, Pfaller MA, Hammons GT, Massanari RM. Use of the Appropriateness Evaluation Protocol for Estimating the incremental costs Associated with Nosocomial Infections. *Medical Care* 1987;25,6:481-488.
- 190. Gertman P, Restuccia J. The appropriateness evaluation protocol: A technique for assessing unnecessary days of hospital care. *Medical Care*. 1981;19:855-871.
- 191. Rishpon S, Lubrash S, Epstein L. Reliability of a method of determining the necessity of hospitalisation days in Israel. *Medical Care* 1986;24:279.
- 192. Wakefield DS, Pfaller M, Ludke R, L., Wenzel R, P. Methods for estimating Days of Hospitalisation due to Nosocomial Infections. *Medical Care* 1992;30(4):373-376.
- 193. Haley R, W. Measuring the Costs of Nosocomial Infections: Methods for Estimating Economic Burden on the Hospital. American Journal of Medicine 1991;91 (suppl 3B):32s-38s.
- 194. LeGrand J, Robinson R, Propper C. *The economics of social problems*. Basingstoke: Macmillan, 1992.
- 195. Finkler S, A. The distinction between costs and charges. Annals of Internal Medicine 1982;96:102-109.
- 196. Dawson D. Costs and prices in the internal market. First edition. York: University of York, 1994.

- 197. Magee CC, Osmolski RJ. A comprehensive system of management information for financial planning, decision making and control in the hospital service. London: Royal Commission on the NHS, 1978.
- 198. Hollingsworth W, Todd C, Parker M, Roberts JA, Williams R. Cost analysis of early discharge after hip fracture. *British Medical Journal* 1993;307:903-906.
- 199. Losos J, Trotman M. Estimated economic burden of nosocomial infection. *Canadian Journal of Public Health* 1984;75:248-250.
- 200. Editorial. How much infection and the cost? *Journal of Hospital Infection* 1981;2:103-104.
- 201. Drummond M, Davies LF. Evaluation of the costs and benefits of reducing hospital infection. *Journal of Hospital Infection* 1991;18 (suppl18):85-93.
- 202. Weinstein M, Read J, MacKay D, Kresel J, Ashley H, Halvorsen K, *et al.* Cost-effective choice of antimicrobial therapy for serious infections. *Journal of General Internal Medicine* 1986;1(6):351-363.
- 203. Dixon R. Costs of Nosocomial Infections and benefits of infection control programmes. In: Wenzel RP, editor. *Prevention and Control of Nosocomial Infections*. Balitimore: Williams and Wilkins, 1987:19-25.
- 204. Wenzel R. The economics of nosocomial infections. *Journal of Hospital Infection* 1995;31:79-87.
- 205. Miller P, J., Farr B, M., Gwaltney J, M. Economic Benefits of an Infection Control Program: Case Study and Proposal. *Reviews of Infectious Diseases* 1989;11;2:284-288.

- 206. Currie E, Maynard A. *Economics of Hospital Acquired Infection*. Discussion paper 56. Centre for Health Economics, University of York, 1989.
- 207. Shapiro M, Schoenbaum S, C., Tager I, B., Munoz A, Polk B, F. Benefit-Cost Analysis of Antimicrobial Prophylaxis in Abdominal and Vaginal Hysterectomy. *Journal of the American Medical Association* 1983;249;10:1290-1294.
- 208. Jones JD, Chawala A. Patient costs in the prevention and treatment of post-caesarean infections. *American Journal of Obstetrics and Gynaecology.* 1984;149:363-366.
- 209. Hojer H. The effect on total anti-microbial consumption and hospitalization time after prophylactic treatment with doxycycline in colorectal surgery. *Acta chirugica Scandinavia* 1978;144:175-179.
- 210. Kaiser AB, Roach AC, Mulherin JLJ, al. e. The cost-effectiveness of antimicrobial prophylaxis in clean vascular surgery. *Journal of Infectious Diseases*. 1983;147:1103.
- 211. Davey P, G., Parker S, E., Crombie I, K., Jaderberg M. The costeffectiveness of Amoxicillin/Clavulanic Acid as Antibacterial Prophylaxis in Abdominal and Gynacological Surgery. *PharmacoEconomics* 1995;7;4:347-356.
- 212. Davey P, Lynch B, Malek M, Byrne D, Thomas P. Cost-effectiveness of single dose cefotaxime plus metronidazole compared with three doses each of cefuroxime plus metronidazole for the prevention of wound infections after colorectal surgery. *Journal of Antimicrobial Chemotherapy*. 1992;30:855-864.

- 213. McGowan JE. Cost and Benefit of peri-operative antimicrobial prophylaxis Methods for economic analysis. *Review of Infectious Diseases*. 1991;13(suppl 10):S879-89.
- 214. Slater F. Cost-effective infection control success story: a case presentation. *Emerging Infectious Diseases* 2001;7(2):293-294.
- 215. Plowman R, Graves N, Esquivel N, Roberts J. An economic model to assess the cost and benefits of an intervention to reduce the risk of catheter related nosocomial urinary tract infections. *Journal of Hospital Infection* 2001;4(1):33-42.
- 216. Lawrence V, Gafni A, Gross M. The unproven utility of the preoperative urinalysis: economic evaluation. *Journal of Clinical Epidemiology* 1989;42:1185-1192.
- 217. Daschner FD. The costs of Hospital Acquired Infection. Journal of Hospital Infection 1984;5 (suppl A):27-33.
- 218. CIPFA/HFM. Health Database: 1994 Health Services Financial Database: CIPFA, London, 1994.
- 219. Health service indicators: NHS Executive, Quarry House, Leeds., 1996.
- 220. Cookson B. Progress with establishing and implementing standards for infection control in the United Kingdom. *Journal of Hospital Infection* 1995;30(Suppl):69 75.
- 221. Hospital Episodes Statistics Database: Department of health, London, 1994/5.

- 222. Office of Population Censuses and Surveys. *Standard occupation classification*. London: HMSO, 1990.
- 223. BUPA. Working with BUPA: A guide for hospitals (schedule of procedures). London: BUPA, 1993.
- 224. NHS Supplies North Thames. *NHS Supplies catalogue*: North Thames Anglia Division, 1994.
- 225. Emberton M, Rokall T, Meridith P. Scanning for audit. *British Journal of Health Compute and Information Management* 1994;11:23-25.
- 226. *GLIM4. The statistical system for generalized linear interactive modelling.* Oxford: Clarendon Press, 1993.
- 227. McKee M, Petticrew M. Disease staging a case-mix system for purchasers? *Journal of Public Health Medicine*. 1993;15(1):25-36.
- 228. Ware J, Snow K, Kosinski M, Gandek M. SF-36 Health survey manual and interpretation guide. Boston, USA: The Health Institute, New England Medical Center, 1993.
- 229. Ware J, Kosinski M, Keller S. *SF-36 physical and mental health summary scales, A users manual.* Boston, USA: The Health Institute, New England Medical Center, 1994.
- 230. Jenkinson C, Layte R, Wright L, Coulter A. The UK SF-36: An analysis and interpretation manual. *University of Oxford Health Services Research Unit* 1996.
- 231. Office of Population Censuses and Surveys. *Fourth revision*: HMSO, 1990.

364

- 232. Barber JA, Thompson SG. Analysis and interpretation of cost data in randomised controlled trials: a review of published studies. *British Medical Journal.* 1998;317:1195-1200.
- 233. Merle V, Germain J, Bugel H, Nouvellon J, Czernichow P, Grise P. Nosocomial urinary tract infections in urologic patients: assessment of a prospective surveillance program including 10,000 patients. *European Urology* 2002;41(5):483-489.
- 234. EPIC. National evidence-based guidelines for preventing hospitalacquired infections: standard principles: Thames Valley University, 2000.
- 235. Pratt R, Pellowe C, Loveday H, Robinson N, Smith G, epic guideline development team. The epic Project: Developing National Evidencebased Guidelines for Preventing Healthcare Associated Infections. *Journal of Hospital Infection* 2001;47(suppl):S3- S66.
- 236. Scheckler WE. Septicemia and Nosocomial Infections in a Community Hospital. Annals of Internal Medicine 1978;89(part 2):754-756.
- Stamm WE. Urinary Tract Infections. In: Bennett JV, Brachman PS, editors. *Hospital Infections*. Fourth Edition. New York: Lippincott-Raven, 1998:477-485.
- 238. Liedberg H, Lundeberg T. Silver alloy coated catheters reduce catheter associated bateriuria. *British Journal of Urology* 1990;65:379-381.
- 239. Lee J, Hernandez J. Silver/hydrogel-coated catheter used to reduce incidence of nosocomial urinary tract infections. *American Journal of Infection Control.* 1996;24(2):117.

- 240. Saint S, Elmore JG, Sullivan SD, Emerson SS, Koepsell TD. The efficacy of silver alloy-coated urinary catheters in preventing urinary tract infection: a meta-analysis. *American Journal of Medicine*. 1998;105(3):236-241.
- 241. Karchmer TB, Muto CA, Cage EG, Strain BA, Farr BM. Randomized cross-over study of silver-coated urinary catheters in hospitalised patients. *Infection Control and Hospital Epidemiology*. 1998.
- 242. Karchmer TB, Muto CA, Byers KE, Dill JB, Farr BM. . Randomised cross-over study of silver-coated urinary catheters in hospitalised patients. *Infection Control and Hospital Epidemiology* 1997;18(No 5 Part 2 M50).
- 243. Darouche RO. Anti-Infective Efficacy of Silver-Coated Medical Prostheses. *Clinical Infectious Diseases* 1999;29:1371-1377.
- 244. Newton T, Still J, Law E. A comparison of the effect of early insertion of standard latex and silver-impregnated latex foley catheters on urinary tract infections in burn patients. *Infection Control and Hospital Eidemiology* 2002;23(4):217-218.
- 245. PrevNiTech Ltd, UK 1999. Personal communication
- 246. Kirkwood B. *Essentials of Medical Statistics*: Blackwell Scientific Publications, 1988.
- 247. Santos K, Bravo Neto G, Fonesca L, Gontijo Filho P. Incidence surveillance of wound infections in hernia surgery during hospitalisation and after discharge in a university hospital. *Journal of Hospital Infection* 1997;36:229-233.

- 248. Hughes J, Jarvis W. Epidemiology of nosocomial. infections. In: Lennette E, Balows A, Hauser W, Shadowy H, editors. *Manual of clinical microbiology*. Washington DC: Department of Health and Human Services, 1985.
- 249. Adler J, Shulman J. Nosocomial infections and antibiotic usage at Grady Memorial Hospital. South Med Journal 1970;63:102-105.
- 250. Latham E, Standfast S, Baltch A. The prevalence survey as an infection surveillance method in an acute and long-term care institution. *American Journal of Infection Control* 1981;32:117-126.
- 251. Wenzel R, Osterman C, Hunting K, Gwaltney J. Hospital acquired infections: 1. Surveillance in a university hospital. *American Journal of Epidemiology* 1976;103:251-260.

Research assistant training and seminar programme

A1.1 Introduction

This appendix contains details of the research assistants' training and seminar programme.

A1.2 Training programme

A1.2.1 Aim

To enable the research assistants to fulfill their role in the study by providing them with the necessary information and enabling them to develop the appropriate skills

A1. 2.2 Learning outcomes

At the end of the induction programme the research assistant will;

- 1. Understand the epidemiology of HAIs, how the cost of HAI has been studied in the past and some principles of infection control including the surveillance of HAI.
- 2. Understand the study methods and how the data collection will be organised.
- 3. Be able to discuss the project with patients and ask for their consent to be included in the study.
- 4. Be able to collect relevant information from clinical records and complete the data collection forms.

- 5. Be able to use the surveillance methods to identify patients with an infection.
- 6. Know how to enter data onto the database and validate some of the information.
- 7. Know who they should contact, and how to contact them if they encounter difficulties or need advice.
- 8. Understand the principles of the Data Protection Act and the standards of record keeping expected of them.
- 9. Understand their terms and conditions of service.
- 10. Be familiar with the areas they will be working in, and emergency procedures.

A.1.2.3 Areas covered:

Over the course of three weeks the following areas were covered:

- 1. Personnel issues: terms of service, disciplinary procedures.
- 2. The study site: the layout of the study hospital; hospital policies; library services.
- 3. Emergency procedures: fire; cardiac arrest and security procedures.
- 4. Aims and objectives of the study and envisaged benefits.

- 5. Study methods: recruitment, informed consent, patient confidentiality, the Data Protection Act and Standards for record keeping, surveillance methods, other data collection procedures (in-patient and post discharge phase), data coding, data entry, filing systems.
- 6. Basic health economics: costs arising from HAIs; valuing the cost of resources used; methods used to attribute resource use to the presence of an HAI.
- 7. Organisational issues: wards responsible for; introduction to ward areas; organisation of workload; time management; who to contact if there is a problem; where to find information of relevance to study; how to order stationary and other supplies.
- 8. Importance of accurate data collection.
- 9. Adapting to a new role and ethical issues
 - What to do if asked to help out on a ward.
 - What to do if you see something being done that you believe to be wrong or unethical.
 - What to do if a patient tells you something in confidence that you believe a member of the medical/nursing staff should be aware of.
 - What to do if a patient does not want to take part in the study how to make them feel comfortable with their decision not to participate.

10. Importance of the Research Assistant's role in the study,

The programme commenced on the 14th March 1994 and was delivered over a three-week period. Much of the training was delivered by the project coordinator, with specialist sessions given by experts in the relevant field. For example, Helen Glenister (member of the steering Group) ran a one day workshop on surveillance of HAI. The Senior Infection Control Nurse from the study site discussed infection control issues and the organistation of infection control services within the Trust. Nick Graves (research economist) gave a couple of sessions on the economics of HAI. Jennie Wilson (Member of the steering Committee) assisted in a couple, of sessions that considered the research assistant's new role and ethical issues that might arise over the course A Mental Health Services Manager ran a session on of the study. communication skills, and obtaining informed consent. The Trust's Community Liaison Nurse Manager ran a session which discussed the organisation of district nursing services in the area. Lynda Taylor (Steering Committee Member) provided an overview of the Public Health Laboratory Services. Over the course of the three week programme the research assistants visited the wards they would be working on, met with ward staff and familiarised themselves with the ward setting. The research assistants also visited a local GP practice. Finally the research assistants went on a weekend team building training programme. The latter was an outward-bound training programme that encouraged team building skills.

The programme was accompanied by a reader, which contained copies of key papers and a list of additional recommended references and journals. Papers addressing the following topics were included: incidence and prevalence of HAI; surveillance; risk factors for HAI; infection control; and the economic burden of HAI.

A1.2 Seminar Programme

The seminar programme covered the following areas: epidemiology of HAI, surveillance and infection control; financing and health care; research methodology; research and health policy; and health care evaluation. The programme was as follows.

371

Hospital Acquired Infection - Epidemiology, Surveillance and Infection Control			
14.6.94	The epidemiology of hospital acquired infection and surveillance methods	Jennie Wilson	
28.6.94	Infection Control	Lynda Taylor	
12.7.94	Hospital Acquired Infection - the pathogens involved	Dr Dianna Barry	
26.7.94	MRSA	Dr Barry Cookson	
9.8.94	The role of the Consultant in Communicable Disease Control	Dr Rachael Joce	

Financing and Managing Health Care

23.8.94	Financing health care - the alternatives	Ros Plowman
6.9.94	The NHS and the NHS reforms	Dr Jenny Roberts
20.9.94	Contracting in the NHS	Jane Bandcroft
4.10.94	Recent Reforms and Primary Care - The District Nurses Perspective	District nurse
18.10.94	The project - NHS applications	Nick Graves

Research Methodology

Research methodology				
1.11.94	Research methods	Dr Mike Rowland		
15.11.94	Costing Methodology	Nick Graves		
29.11.94	Statistics made easy - part 1	Mark Griffin		
13.12.94	Statistics made easy - part 2	Mark Griffin		

Research and Health Policy

28.2.94	Does research affect policy? If so how?	Dr A Zwi
31.1.95	Research, Policy and the DOH perspective	Dr E Meerabeau

Economics and Health Care, Health Care Evaluation

14.2.95	Supply, demand and the market	Nick Graves
28.2.95	Health care evaluation	Nick Graves

Standards for Records and Record Keeping - Socio-economic Burden of Hospital Acquired Infection

1. Introduction

- 1.1 This paper sets out the standards to be applied to ensure confidentiality of patient centred data for the Socio-economic Burden of Hospital Acquired Infection study.
- 1.2 Nurses have a responsibility to adhere to the United Kingdom Central Council for Nursing, Midwifery and Health Visiting `Code of Professional Conduct for the Nurse, Midwife and Health Visitor' Standards for records and Record Keeping and abide by the following principles:

"As a registered nurse, midwife or health visitor you are personally accountable for your practice and, in the exercise of your professional accountability, must:

- 1. act always in a manner as to promote and safeguard the interests and well-being of patients and clients;
- ensure that no action or omission on your part, or within your sphere of responsibility, is detrimental to the interests, condition or safety of patients and clients"

2. Maintaining Patient Confidentiality

- 2.1 Each patient that participates in this study will be identified by a unique study number. This number will be the only form of patient identification included in the computer held database.
- 2.2 Access to these patient data record files will be restricted to individuals named in Appendix A.
- 2.3. Personal details such as a patient's name, address, telephone number and hospital number will be stored in a separate file held on computer and on paper stored in a locked filing cabinet, within a locked room at the *study hospital*, or at the LSHTM or CPHL. CPHL has a secure perimeter fence and 24 hour security. The LSHTM has 24-hour security. Access to this file will be restricted to the individuals listed in Appendix B.
- 2.4 It will not be possible to access records held on computer without the use of a password. (Separate passwords will be required to gain access to the patients data record files and the patient personal detail files). Passwords will be changed periodically.
- 2.5 The project computers will be held in a locked room. Only individuals listed in Appendix A will be allowed access to the project computer.
- 2.6 Data exported for analysis will be entered onto designated computers at (LSHTM). Colindale has a secure perimeter fence and 24 hour security. The LSHTM has 24 hour security.
- 2.7 Access to the computers held at Colindale and the LSHTM will be restricted to individuals listed in appendix A.
- 2.8 Access to data held on these computers will be through the use of a password which will be changed periodically.

2.9 After completion of the research or its formal abandonment the key to the identities of all persons involved in the research and all personal data no longer required will be destroyed unless the Secretary of State directs otherwise in writing. The certificate of the Secretary of State to this effect shall be conclusive.

3. Patients Access to Records

- 3.1 In accordance with the Data Protection Act 1984 and the Access to Health Records Act 1990, participants in the study have the right to see their records. Information will only be withheld if in the view of the patients medical practitioner its release might cause serious harm to the physical or mental health of the patient or it would identify a third party.
- 3.2 If a patient requests access to their records, the request must be passed to the project co-ordinator, who will facilitate access.
- 3.3 When necessary advice regarding release of data will be sought from members of the Project Group.

4. Training of Research Assistants

- 4.1 The research assistants will receive training on the Data Protection Act and their professional responsibility for record keeping during their induction programme.
- 4.2 The project co-ordinator will provide additional training if the standards of record keeping described in this document are not being maintained.
- 4.3 Each research assistant will receive a copy of this document which they should read and sign.

Signature.....

Date.....

Appendix A

Individuals permitted to have access to patient records held on paper and on computer, the room where the study computer is held, the study computer and the computers to be used for analysis.

Project co-ordinator Project economist Project statistician Research assistants Project secretaries Data entry clerk Programme leader in surveillance and infection control Statisticians from Steering Committee Health economist from Steering Committee

Appendix B

Individuals allowed access to the patient personal details file

- Project co-ordinator
- Project economist
- **Project statistician**
- **Research assistants**
- Project secretaries
- Data entry clerk

Presentations concerning the Socio-economic Burden of Hospital Acquired Infection Study

Date	Presentation	
Sept 1993 – April 1994	A number of seminars were held for clinical staff at the study hospital and also for district nurses within the community. The seminars provided an opportunity to discuss the aims and objectives of the socio-economic burden of HAI study and the methods we wished to employ, and provided staff at the study site with an opportunity to ask questions about the study.	
Nov. 1994	Council of Europe, Strasbourg	
	`Economic and social aspects of hospital acquired infection.'	
March 1995	Canadian Nosocomial Infection Surveillance Programme - Semi-annual Meeting, Ottawa, Canada	
	`Estimating the costs of hospital acquired infection.'	
June 1995	University of Hertfordshire Research Leaders Forum	
	`The socio-economic burden of hospital acquired infection.'	
Sept. 1995	Public Health Laboratory Service 20th Annual Scientific Conference	
	Socio-economic burden of hospital acquired infection - An extensive follow up survey.	
March 1996	European Infection Control Seminar, London.	
	'Hospital acquired infection - how much does it cost?'	
March 1996	The London Hospital Medical College, London.	
•	'The economics of infection.'	
June 1996	University of Hertfordshire.	
	'The socio-economic burden of hospital acquired infection'	
July 1997	Diploma in infection control nursing – London.	
	'The cost of hospital acquired infection'	
Sept 1997	Infection control nurses association annual conference Swansea.	
	<i>`Hospital-acquired infection – How much does it cost and why do we need to know?'</i>	

Date	Presentation
June 1998	Infection Control Nurses Association, Wessex Regional Group Seminar.
	'An expensive business - The real cost of hospital-acquired infection'.
July 1998	The NHS Confederation 50 th Anniversary Conference: The NHS: All our tomorrows. The Public Health Laboratory Service and hospital-acquired infection.
	'The Economic challenge of hospital-acquired infection.'
Nov. 1999	Nosocomial Infection National Surveillance Scheme, Surgical Site Infection Annual Meeting.
	'The cost of surgical site infections'
April 2000	Kelsey Lecture 2000, Central Sterilising Club.
	'Hospital —acquired infection — Where is the Czar?'
June 2000	Infection Control Nurses Association – London Branch.
	`The socio-economic burden of hospital acquired infection'.
June 2000	North Thames (East) Microbiology Consultants Meeting.
	'The price of <u>not</u> washing your hand'
July 2000	The first conference of the Economics of Infectious Disease. London School of Hygiene and Tropical Medicine.
	'The burden of hospital acquired infection'
Nov. 2000	The Infection Control Nurses Association International Conference. Edinburgh
	'The economic burden of hospital acquired infection'
March 2001	Risk and Responsibilities. A conference on hospital acquired infection. North Staffordshire Hospital NHS Trust.
	'The socio-economic burden of hospital acquired infection'
March 2001	International Conference on The Economics of Infectious Disease, London School of Hygiene and Tropical Medicine.
	'Cost information informing policy and practice – A case study of hospital acquired infection'
June 2001	Infectious Diseases and Hospital Acquired Infection Guidelines - The
	implementation challenge. A conference organised by the Health Studies
	Socio-economic benefits of better control of hospital acquired infection'

Wards involved in the Socio-economic burden of Hospital Acquired Infection Study

A4.1 Introduction

The Socio-economic Burden of Hospital Acquired Infection study involved the assessment of the incidence and economic burden of hospital acquired infections (HAIs) occurring in patients admitted to eight clinical specialties, common to most general hospital. Six research assistants were responsible for recruiting patients admitted to these specialties and subsequent data collection. For practical reasons it was not possible to recruit all patients admitted under the eight clinical specialties. As such, 14 study wards were selected. Each research assistant was responsible for the recruitment of patients admitted to the selected study wards. This appendix provides details of the number of wards included, their primary clinical specialty and the number and type of wards each research assistant was responsible for.

A4.2 Wards involved in the socio-economic burden of hospital acquired infection study.

Table A4.1 provides details of the wards involved, their clinical specialty and the number of beds on each ward. It should be noted that whilst each ward had a primary clinical interest, in practice patients admitted under other specialties were admitted to these study wards. Consequently, in addition to recording the ward, the research assistants recorded the clinical specialty and the consultant patients were admitted under.

Table A4.1 Number of wards involved in the Socio-economic Burden of HAI study

Clinical specialty	Number of wards	Total number of beds
General medicine	3*	75
General surgery	2	60
Orthopaedics	2	60
Urology	1	15
Gynaecology	1	20
Elderly care	2	56
ENT	1	15
Obstetrics**	2	50

one ward included a 5 bed coronary care unit

*Only those patients who had a caesarean section were eligible for recruitment.

A4.3 Number of wards each research assistant was primarily responsible for

Table A4.2 provides details of the number of wards each research assistant was primarily responsible for. The allocation of wards took into account both the number of beds available on each ward and the types of patients admitted to each of the selected wards, and the implications this might have for recruitment and subsequent data collection.

Research assistant	Wards allocated	Total number of beds
1	2 care of the elderly wards	60
	Obstetrics*	50*
2	I surgical ward	30
	I medical ward	15
3	I gynaecology ward	20
	1 surgical ward	30
4	1 urology ward	15
	1 medical ward	30
5	2 orthopaedic wards	60
	1 ENT ward	15
6	I medical ward	30

Ta	ble	A4.2	The allocation of	i wards to	research	assistant
----	-----	------	-------------------	------------	----------	-----------

Only those patients who had a caesarean section were eligible for recruitment.

Patient information sheet, consent forms and decline to participate form

A5.1 Introduction

This appendix contains a copy of the following documents:

- a) The patient study information form.
- b) The consent form administered to patients who agreed to participate in the study.
- c) The consent form administered to relatives of patients who were unable to give their own consent to participation in the study.

INFORMATION SHEET

COSTING INFECTIONS

A small number of patients develop an infection following hospital treatment; this can occur whilst in or shortly after leaving hospital. The study hospital is involved in a study, funded by the Department of Health, looking at how many patients develop such infections and the cost of these to the health service and the patients concerned. We would be grateful if you would take part in this study.

We would like to gather information about your condition and the treatment you receive. This will involve looking at your medical and nursing records. After you have left hospital we may need to contact your GP.

You may also be contacted about a month after you have left hospital to find out how you are feeling and how you have been since leaving hospital. Any information you give us will be kept strictly confidential.

Since the information we obtain from you will be treated in total confidence, and will not be reported to your doctor or nurse, it is important that you also mention any concerns you might have about your health and treatment to them.

We would appreciate your help with this study. If you are willing, would you please sign the attached consent form. If you do not wish to participate, it will not affect the treatment you are given in any way.

If you have any questions or require any further information about the study please do not hesitate to ask <u>(insert name)</u> your research nurse or contact Rosalind Plowman the Project Co-ordinator. (Insert name of research nurse) will visit the ward daily but can also be contacted on Ext. XXX at the study hospital. Rosalind Plowman can be contacted either on Ext. XXX at the study hospital or at the Central Public Health Laboratory, London, Tel 081 200 4400 ext. 4234.

Thank you for your help.

Central Public Health Laboratory, 61 Colindale Avenue, London NW9 5HT

COSTING INFECTIONS

CONSENT FORM

I ------

agree to take part in this study looking at the extent of and cost of hospital infections. I have read the study information sheet and understand that I have given my consent to the collection of data on my condition and the treatment I receive. Whilst in hospital this information will be obtained from my medical and nursing notes. After leaving hospital my GP may be contacted to obtain this information.

I also understand that taking part in this study may involve my completing a questionnaire one month after leaving hospital.

I understand that any information obtained from my medical and nursing notes and any information given will be kept strictly confidential.

Patient's signature

Date

Research Assistant's signature

Date

Central Public Health Laboratory, 61 Colindale Avenue, London NW9 5DF

COSTING INFECTION

CONSENT FORM

-
I have ollection Vhilst in After
and any

Central Public Health Laboratory, 61 Colindale Avenue, London NW9 5DF

PATIENT DECLINES TO PARTICIPATE IN THE STUDY OR IS EXCLUDED FROM STUDY	Q1 Study Number			
Q2 Date of admission:	Q3 Day of attem	pted recruitment:		
Q5 Pt declines participation Unable to recruit pt due to lack of time Unable to find a suitable time to recruit pt Pt too sick/next of kin unavailable/inappropriate to ask next of kin Pt admitted on a non-recruiting day Weekend admission				
Q6 Age group: 18-30	Q7 Specialty: Surgery Medicine Care of the elderly Orthopaedics Urology Obs & Gynae ENT	Q8 Sex: Male Female Q9 Type of admission: Elective Emergency		
Survey : 1		Page : 1		

Definitions of infections presenting during the patient's hospital stay

A6.1 Introduction

The definitions of hospital acquired infections used in this study, with the exception of the definition of a chest infection, are those developed and used in the recent study, 'A Study of Surveillance Methods for Detecting Hospital Infections.'⁷⁷

A6.2 Hospital acquired infection (HAI)

An infection found to be active (or under active treatment at the time of survey) which was not present or incubating on admission to hospital.⁶ Where doubt exists, infections appearing at 72 hours or more after admission should be classified as hospital acquired.^{248 249} A patient readmitted with established infection resulting from an earlier admission is recorded as having a HAI.²³ Transfers admitted from another hospital with a nosocomial infection acquired there will be coded separately.

A6.3 Community Acquired Infection (CAI)

An infection found to be active (or under active treatment) at the time of survey, which was present or incubating on admission to hospital.²⁵⁰

A6.4 Criteria for Diagnosing the Presence of Infection

There must be clinical evidence of infection except in the case of central nervous system infections where laboratory evidence may suffice. Colonisation should be excluded.

A6.4.1 Clinical evidence.

This includes the cardinal signs and symptoms as defined in this document which are presented, or have been present during the patient's stay in hospital. Some signs and symptoms may include fever \geq 37.8°C.²⁵¹ where infection is the only known cause, inflammation (i.e. redness, swelling, pain, heat) and the production of pus.

A6.4.2 Laboratory evidence

This is present if the results of specimens meet the criteria defined in this document. With some infections, a clinician's diagnosis of infection based on clinical signs and symptoms with or without laboratory evidence will be accepted.

A6.5 Date of onset

This is the date the first clinical evidence of infection appeared, or if no signs and symptoms are present, the date the specimen used to make or confirm the diagnosis was collected, whichever comes first.

A6.6 Urinary tract infection

A5.6.1 Patients without a urinary catheter in situ

A urinary tract infection is present if the patient has two or more of the following signs and symptoms:

- Fever \geq 37.8°C with no other recognised cause
- Urgency
- Frequency
- Dysuria

WITH OR WITHOUT

- A positive urine culture, that has ≥10⁵ colonies per ml of urine with no more than two species of micro-organisms.
- A positive urine culture that has <10⁵ colonies per ml of urine of a single microorganism in the presence of an antibiotic being given to treat an urinary tract infection.

OR

• A positive urine culture with more than 2 species of micro-organisms identified and the presence of 10 white blood cells or more seen on high power film.

Note:

- If the patient is unable to communicate the above signs and symptoms a clinician's diagnosis of urinary tract infection will suffice.
- Surveyor should note whether clinical, or clinical with laboratory evidence is used to determine the presence of infection.
- Infections of organs of the urinary tract (kidney, ureter, bladder or urethra) following surgery to that area should be recorded as surgical wound infection.

A6.6.2 Patients with a catheter in situ

A clinician's diagnosis of a urinary tract infection will suffice.

WITH OR WITHOUT

- A positive urine culture, that has ≥10⁵ colonies per ml of urine width no more than two species of micro-organisms.
- A positive urine culture that has <10⁵ colonies per ml of urine of a single microorganism in the presence of an antibiotic being given to treat an urinary tract infection.

OR

• a positive urine culture with more than 2 species of micro-organisms identified and the presence of 10 white blood cells or more seen on high power film.

Note:

- The surveyor should note whether clinical, or clinical with laboratory, evidence is used to determine the presence of infection.
- Infection of the urethra occurring at the insertion site of a catheter should be included as an urinary tract infection. The presence of the device (i.e. catheter) should be noted.
- Infections of organs of the urinary tract (kidney, ureter, bladder or urethra) following surgery to that area should be recorded as surgical wound infection.

A6.7 Asymptomatic bacteriuria

A6.7.1 Asymptomatic bacteriuria

This is present if:

- There are two positive urine cultures that have ≥10⁵ colonies per ml of urine with repeated isolation of the same micro-organism and no more than two species of micro-organisms.
- The patient does not have any of the following signs and symptoms: fever (<37.8°C); urgency; frequency; dysuria.

A6.8 Infections of upper respiratory tract and ear

A clinician's diagnosis of one or more of the following with or without microbiological evidence of infection:

- Furuncle
- Rhinitis (infective)
- Sinusitis
- Pharyngitis
- Epiglottis
- Tonsillitis
- Otitis media

Note:

- Infection of the anterior nares surrounding the insertion site of a nasogastric tube should be included as upper respiratory tract infection. The presence of the nasogastric tube should be noted.
- Infections of the upper respiratory tract (ear, nose or throat) following surgery to that area should be recorded as surgical wound infection.

A6.9 Pneumonia

Pneumonia is present if the patient has appropriate chest signs including consolidation and/or x-ray changes showing new or progressive infiltrate and one or more of the following:

- New or increased production of sputum
- Fever (≥37.8⁰C)

If the above are not present, a clinician's diagnosis of pneumonia will suffice.

A6.10 Chest Infection

A chest infection is present if the clinician has diagnosed a chest infection and the patients' symptoms do not meet the definition of pneumonia

A6.11 Other lower respiratory tract infection

A clinician's diagnosis of one or more of the following will suffice:

- Empyema
- Lung abscess
- Tracheitis
- Bronchitis
- Mediastinitis

Note:

 Infections of any one area of the lower respiratory tract (trachea, bronchus, lung, mediastinum) following surgery to that area should be recorded as a surgical wound infection.

A6.12 Wound infection

A wound is defined as a break in the epithelial surface (skin or mucous membrane) and the underlying tissue made by some positive act such as an accident or surgical incision. Burns should be excluded. An ulcer or pressure sore is not a wound for the purposes of this definition.

All wound infections must have one of the following:

- Purulent discharge in the wound
- Purulent discharge exuding from a wound
- Purulent discharge seen on direct examination at the operative site.

A6.12.1 Major infection

This is present when the wound is broken down, gaping or completely dehisced or there is evidence of septicaemia, spreading cellulitis and lymphangitis.

A6.12.2 Minor infection

This is present when the wound is not broken down, gaping or completely dehisced and there is no evidence of septicaemia, spreading cellulitis and lymphangitis.

A6.12.3 Surgical wound infection

This is present if infection occurs at the incision site or operative site (including drains) within 30 days after surgical operation if no implant is left in place, or within one year if an implant is in place. The infection must appear to be related to the surgical procedure.

A6.12.4 Accidental wound

This is present if infection occurs at or in the accidental wound site.

Note:

 Infections occurring at the entry site of a device which has required an incision for insertion should be noted as surgical wound infection (e.g. tracheostomy, intravascular catheters, renal dialysis catheters, suprapubic catheter). The presence of the device should be noted.

A6.13 Skin infection

A skin infection is present if there is inflammation of the skin and infection is the only known cause. There may or may not be pus on the skin.

Note:

• Ulcers, pressure sores and otitis externa should be excluded.

A6.14 Burn infection

A burn infection is present if one or both of the following are present:

- Discharge of purulent material
- Graft rejection with clinical (i.e. inflammation and/or pus) evidence of infection.

A6.15 Septicaemia

Septicaemia is present if the patient has at least one of the following signs or symptoms:

- Fever (\geq 37.8^oC) with no other recognised cause
- Chills or rigors
- Hypotension

AND

• Micro-organisms are isolated from one or more blood cultures taken when the symptoms were present.

A6.16 Bacteraemia

Bacteraemia is present if:

 Micro-organisms have been isolated from one or more blood cultures taken on the one occasion, except in the isolation of a skin contaminant (e.g. diptheroids, coagulase negative staphylococci), when two or more positive blood cultures drawn on separate occasions should be obtained.

AND

• The patient does not have any clinical signs or symptoms of infection, i.e. there is no fever (<37.8⁰C), chills or hypotension.

A6.17 Eye infection

An eye infection is present if there is new purulent discharge or pus within or on the surface of the eye.

Note:

- Infections of the skin surrounding the eye, e.g. stye, should be noted as skin infections.
- Infection of the eye following surgery should be noted as a surgical wound infection.

A6.18 Central nervous system infection

A central nervous system infection must meet at least one of the following criteria:

- Micro-organisms in cerebral-spinal fluid (CSF), but excluding contaminants, with or without white blood cells.
- White blood cells in CSF in the absence of micro-organisms if the patient is receiving antibiotics.
- White blood cells in the CSF in the absence of micro-organisms if there is no other obvious cause for their presence.

A6.19 Genital tract infection

Genital tract infections can be divided into post-partum and other genital tract infections.

A6.19.1 Post-partum infection

This requires systemic evidence of infection with a new purulent discharge.

A6.19.2 Other genital infection

This is present if there is new purulent discharge with or without microbiological evidence of infection.

Note:

- Episiotomy should be classified as a surgical wound and a perineal tear classified as an accidental wound.
- Infection of any one area of the genital tract following surgery to that area should be recorded as wound infection.

A6.20 Gastrointestinal infection

A gastrointestinal infection is present if diarrhoea and/or vomiting occurs which is not as a result of any of the following:

- Diagnostic tests
- Therapeutic regimens
- Other underlying non-infectious causes

Note:

• The presence of a gastrointestinal infection should be supported whenever possible by microbiological evidence.

A6.21 Other abdominal infection

Other abdominal infections include a clinician's diagnosis, with or without microbiological evidence, of the following:

- Intra-abdominal abscess formation
- Peritonitis

Note:

- Appendicitis, cholecystitis, pancreatitis and diverticulitis should not be recorded as infections unless the presence of pus is noted.
- Infections within the abdomen following surgery to the affected area should be recorded as surgical wound infection.

A6.22 Bone and joint infections

These require a clinician's diagnosis of septic arthritis or osteomyelitis, with or without microbiological evidence.

A6.23 Systemic infection

This requires a clinician's diagnosis, with or without laboratory evidence (including serology), of generalised bacterial, viral, fungal or parasitic infection without a definable single site of infection (e.g. measles, mumps, herpes, varicella).

A6.24 Other infections

These require a clinician's diagnosis, with or without microbiological evidence of infection, which does not fall into the above categories (e.g. varicose ulcers, rectal abscesses, pressure sores, otitis externa, oral thrush and non-therapeutic related hepatitis). This includes clinical symptoms of infectious hepatitis (A,B, Non A and Non B) and serum positive for hepatitis B antigen without symptoms.
Examples of data collection forms used

A7.1 Introduction

This appendix contains examples of some of the data collection forms used. The following forms are included:

- i) General data collection form. Information relating to the type of admission, reason for admission, the patients' social circumstances, formal and informal care received prior to admission, ward transfers, discharge diagnosis and care organised on discharged were recorded on this form.
- ii) Drug data collection forms. Information on drugs and infusions administered were recorded on a number of different data collection forms. The enclosed examples were used to record infusions and drugs administered by the intravenous, intra-muscular and/or subcutaneous route during the patients' hospital stay. Similar forms were available to record, tablets, topical drugs and other drugs administered.
- iii) Investigations data collection form. Information on investigations undertaken (e.g. cardiac tests, endoscopies and x-rays) were recorded on this form, together with information on devices in place (e.g. intravenous lines, wound drains and urinary catheters); care provided by health care professionals allied to medicine (e.g. physiotherapists, occupational therapists and dieticians); and nursing care administered to patients during their hospital stay.
- iv) Operation data collection form. Operation details, including the type of anaesthetic, procedures performed and duration of surgery were recorded on this form.

- v) Laboratory tests data collection form. Information on laboratory tests performed were recorded on this form. Specimens taken either at the preassessment clinic or in the accident and emergency department were recorded in the appropriate columns. If the test was performed out of normal working hours the test was recorded in the 'o/c' (on call) column.
- vi) Infections data collection form. Information on both hospital and community acquired infections were recorded on this form. Where possible, the date of onset, site of infection, devices in situ and pathogens involved were recorded.

With the exception of the general and the laboratory tests data collection forms, the forms contained within this appendix apply to the first two weeks in hospital. The laboratory tests data collection form only applies to the first week in hospital. Additional forms were available for patients who had a length of stay that exceeded these time-periods.



Q12 Ward admitted to: _

0	1	2	3	4	6	6	7	9	
									10
\Box									1

Survey : 2

Orthopaedics Urology

Gynae c/o elderly ENT Obs

Scanning by Formic Ltd, London, (+44) 0171 924 1730

398

Page : 1

Q13 Admitting Consultant

0	1	2	3	4	5	6	7	8	9	
										10
										۱

Q16 GP_____

0	1	2	3	4	5	6	7	8	9	
										100
										10
										1

Q19 DOB

_	1	2	3									
				4	5	8	7	1	9			
					[_							
0	1	2	3	4	5	6	7		9	10	11	12
0	1	2	3	4	5	6	7	1	9			

Q14 Joint care on admission - consultant

0	1	2	3	4	6	6	7	9	
									10
									1

Q15 Sex Male Female

Q17	A	g	e 1	2	3	4	6	6	7	8	
		Τ									100
	Γ	T				Γ					10
	Ē	Τ	٦					Ē			1

Q18 Weight (kgs)

0 1	23	4 5	87	
	ΙE	Π		100
	Π	Ш		10
	TT	T		1

Q20	H	eig	ht	(ft)					
Ů	<u>+</u>			6	8	<u>,</u>	•	h	1
		Т	Т	Т	T				0.1
		Ī.	Ì	Î	Γ				0.01

1 1 1 1 0.1

100 10 1

Q22 Diagnosis on admission:

Q21 Reason for admission:

0	1	2	3	4	5	6	7	8	9	
										1000
\Box										100
\Box										10
\Box										1
										0.1

Q23 Co-morbidities 1













٨ 4 2 3 4 6 8 7 9 1000



Q26 Co-morbidities 4



	Household Size
Q27 Marital status	Q28 Number of adults in household including patient
married living with husband/wife/partner	0 1 2 3 4 5 6 7 8 9 >18 yrs
married not living withhusband/wite widowed widowed	Q29 Number of children in household
☐ divorced	0 1 2 3 4 5 6 7 8 9
	Q30 How many children have you got?
	0 1 2 3 4 5 6 7 8 9
Q31 How many children under 5 years old	Q32 How many children under 16 years old
Q33 How many children under 11 years old	Q34 How many children under 18 years old
Q35 Currently living in Residential home Nursing hom	
<u>CHILD CARE</u>	
CHILD CARE	Q37 If yes from whom? I Husband/wife/partner
CHILD CARE Q36 Have you had to arrange any addditional care for your children whilst in hospital?	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s)
CHILD CARE Q36 Have you had to arrange any additional care for your children whilst in hospital? Yes No No	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder
CHILD CARE Q36 Have you had to arrange any additional care for your children whilst in hospital? Yes No <u>900 is a half day</u>	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder other:
CHILD CARE Q36 Have you had to arrange any addditional care for your children whilst in hospital? Yes No <u>900 is a half day</u>	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder other:
CHILD CARE Q36 Have you had to arrange any addditional care for your children whilst in hospital? Yes No 900 is a half day Q38 Husband/wife partner - days caring for chn	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder other: Q39 Husband/wife partner - days off work
CHILD CARE Q36 Have you had to arrange any addditional care for your children whilst in hospital? Yes No 900 is a half day Q38 Husband/wife partner - days caring for chn 0 1 2 3 4 5 6 7 8 9 100 10	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder other: Q39 Husband/wife partner - days off work 0 1 2 3 4 5 6 7 8 9 10 10
CHILD CARE Q36 Have you had to arrange any addditional care for your children whilst in hospital? Yes No 900 is a half day Q38 Husband/wife partner - days caring for chn 0 1 2 3 4 5 6 7 8 9 100 100 100 100 100 100 100 10	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder other: Q39 Husband/wife partner - days off work 0 1 2 3 4 5 6 7 8 9 100 10 10 10 10
CHILD CARE Q36 Have you had to arrange any addditional care for your children whilst in hospital? Yes No 900 is a half day Q38 Husband/wife partner - days caring for chn 0 1 2 3 4 5 6 7 8 9 100 100 100 100 Q40 Relative 1- days caring for chn Q41 Relative 1- days caring for chn	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder other: Q39 Husband/wife partner - days off work 0 1 2 3 4 5 6 7 8 9 100 10 10 10 10 10 10 10 10 1
CHILD CARE Q36 Have you had to arrange any addditional care for your children whilst in hospital? Yes No 900 is a half day Q38 Husband/wife partner - days caring for chn 0 1 2 3 4 5 6 7 8 9 100 0 1 2 3 4 5 6 7 8 9 Q40 Relative 1- days caring for chn Q41 Relative 1- days caring for chn 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder other: Q39 Husband/wife partner - days off work 0 1 2 3 4 5 6 7 8 9 10 10 10 10 10 10 10 10 10 10
CHILD CARE Q36 Have you had to arrange any additional care for your children whilst in hospital? Yes No 900 is a half day Q38 Husband/wife partner - days caring for chn 0 1	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder Other: Q39 Husband/wife partner - days off work 0 1 2 3 4 5 6 7 8 9 10 10 10 10 10 10 10 10 10 10
CHILD CARE Q36 Have you had to arrange any additional care for your children whilst in hospital? Yes No 900 is a half day Q38 Husband/wife partner - days caring for chn 0 1	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder Other:
CHILD CARE Q36 Have you had to arrange any additional care for your children whilst in hospital? Yes No 900 is a half day Q38 Husband/wife partner - days caring for chn 0 1	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder Other: Q39 Husband/wife partner - days off work 0 1 2 3 4 5 6 7 8 9 10 10 10 10 10 10 10 10 10 10
CHILD CARE Q36 Have you had to arrange any additional care for your children whilst in hospital? Yes No 900 is a half day Q38 Husband/wife partner - days caring for chn 0 1	Q37 If yes from whom? Husband/wife/partner 0 other relative (s) friend/neighbour (s) 0 nanny/childminder other:
CHILD CARE Q36 Have you had to arrange any addditional care for your children whilst in hospital? Yes No 900 is a half day Q38 Husband/wife partner - days caring for chn Q41 Relative 1- days 0 1 2 3 4 5 0 7 8 0 0 1 2 3 4 5 0 7 8 0 0 1 2 3 4 5 0 7 8 0 0 1 2 3 4 5 0 7 8 0 Q41 Relative 1- day Q41 Relative 2- day Q44 Relative 2- day 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0 0 1 2 3 4 5 6 7 8 0	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder Other: Q39 Husband/wife partner - days off work 0 1 2 3 4 5 0 7 8 9 100 10 10 10 10 10 10 10 10 1
CHILD CARE Q36 Have you had to arrange any addditional care for your children whilst in hospital? Yes No 900 is a half day Q38 Husband/wife partner - days caring for chn 0 1 2 3 4 5 0 7 0 9 0	Q37 If yes from whom? Husband/wife/partner other relative (s) friend/neighbour (s) nanny/childminder Other:



Q61 Whilst in hospital have you had to arrange for any one to care for these dependents?

No Husband/wife/partner Other relative Ifriend/neighbour Iespite care



Idatery etc Sociaervos pets

Q62	Husband/wife/partner ca											for D
	0	1	2	3	4	6	6	7	8	9		
										Γ	100	
							Γ				10	
										Γ]1	

Q64	Re	lai	tiv	e1-	da	iys	ca	ri	ŋ	for	D
	_	1	2	3	4	5	6	7	8	<u> </u>	
				L						\Box	100
											10
											1

Q67	7	Re	lat	ive	2-	d	ays	ca	irii	١g	for D
	0	1	2	3	4	5	6	7	8	Ő	
i											100
							Π				10

Q70 Relative 3- days caring for D 0 1 2 3 4 5 8 7 8 9 -| 100 TIT 10 Т 1111 Т

Т Т Т

Q65	Rela	tiv	e 1	- d	lay		ff	wo	rk		
-	0	1	2	3	4	6	8	7	8	9	
											100
	Ē	Ĺ									10
		Γ			Γ						1

Q68 Relative 2 - days off work

				Π			100
	Ι	Τ	1				10
\Box		Ι				Ι	1

Q71 Relative 3 - days off work

U	1	2	3	4	9		7		
									100
H									10
	-	-				_		-	

Days	i of	ĩ w	/01	k						
ĥ	h	2	,	4	6	6	'n	-	h	100
									Ī	10
										1

Q63

Q66 Occupation



Q69 Occupation



Q72 Occupation

FRIENDS/NEIGHBOURS

Q73	F /1	1 1	- da	ays	car	ing	for	D
Ľ	Ļ	Ĺ	Ţ	Ţ	Ů	Í] 100
C	Ι		Τ	Τ	\square] 10
C				Ι			1	1

Q76	5	F/ľ	N 2	d	ays	ca	rit	ng	foi	r D	
	0	1	2	3	4	5	8	7		9	
											100
Ì											10
											1

Q75)	F/	N 3	- 0	lay	5 (ari	ing	fo	r E)
ĺ	Ĉ	Ļ	ľ	J	Ĺ	Ļ	Ē	ŕ		'n	100
		Γ	Γ								10
		Γ_			L						1

	τ.	ork	w	ff	75 (day	- (11	F/N	4 1	Q7
			7	8	5	4	3	2	1	0	
100										\square	
10										\Box	
1										П	

Q77	1 1	F/T	12	- d	ay	s 0	ff	wo	rk		
-	0	1	2	3	ă,	6	6	7	8	9	
										\square	100
ĺ										\Box	10
										Π	1

Q8	0 3	F/N	13	- da	ays	off	wo	rk		
	0	1	2	3	4	5 6	7	8	9	
				Ι	Τ		Γ			100
				Ι	Ι	Τ	Γ			10
				Τ	Ι					1

Q75 Occupation



Q78 Occupation



Q81 Occupation





Scanning by Formic Ltd, London, (+44) 0171 924 1730

OCCUPATION/EMPLOYMENT

Q93 Employment status prior to admission to hospital.

	Q94 Job description
employed full time employed part time employed but temporary laid off unemployed looking for work unable to work because of illhealth (disabled) Housewife/ house husband student (FT) retired	 Self employed with employees Self employed without employees Employed - manager Employed - foreman/supervisor Employed - normal employee/apprentice
other - please specify	Q95 Social Class
Present or most recent job?	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

Q97 Husband/wife/partners occupational status?

- N/A single/not living with partner/husband/wife
 - employed full time
 - employed part time
 - employed but temporary laid off unemployed looking for work
 - unable to work because of illhealth (disabled)
 - Housewife/husband
 - student (FT)
 - retired

other - please specify

Q98 Job description

- Self employed with employees
- Self employed without employees
 - Employed manager
 - Emplyed foreman/supervisor
- Employed normal employee/apprentice

Q99	So	Social Class										
		0	1	2	3	4		8	7	8	Ð	
	[1

Q100 Socio economic group 0 1 2 3 4 5 6 7 8 Present or most recent job 10 1 0.1 Survey : 2 Page : 7 \square

Scanning by Formic Ltd, London, (+44) 0171 924 1730

Consultant Transfer 1	Q101 Consultant transfer to	Q102 Day of	- Q102 Day of transfer: 0 1 2 3 4 6 6 7 8 9 10 11 10 1			
<u>Consultant transfer 2</u>	Q103 Consultant transfer to 0 1 2 3 4 5 6 7 8 9 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Q104 Day of 0 1 2	Q104 Day of transfer 0 1 2 3 4 5 6 7 8 9 10 10 10 10			
<u>Consultant transfer 3</u>	Q105 Consultant transfer to 0 1 2 3 4 5 8 7 6	Q106 Day of 10 0 1 2 1 1 1	Q106 Day of transfer 0 1 2 3 4 5 6 7 8 9 10 10 11 10 10 10 10 10 10 10			
<u>Ward transfer 1</u>	Q107 Ward transfered to: 0 1 2 3 4 5 6 7 8 9 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Q108 Day of	transfer 3 4 5 9 7 8 9 10 11 1			
Ward transfer 2	Q109 Ward transfered to: 0 1 2 3 4 5 6 7 8 9 10 10 10 11 10 10 10 10 10 10	Q110 Day of	transfer 3 4 6 6 7 8 9 10 11 10 1			
Ward transfer 3	Q111 Ward transfered to: 0 1 2 3 4 5 6 7 8 9 10 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Q112 Day of	Q112 Day of transfer 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 10 0 10 0 10 10			
Ward transfer 4	Q113 Ward transfered to: 0 1 2 3 4 6 6 7 8 9 0 1 2 3 4 7 8 9 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Q114 Day of 1 0 1 2	transfer 3 4 5 6 7 8 9 10 10 1			
<u>0115</u>	O116	0117	O118			
<u>Consultant Referral 1</u>	Consultant Referral 2	<u>Consultant Referral 3</u>	<u>Consultant Referral 4</u>			
Nos Code Day of days	Nos Code Day of days	Nos Code Day of days	Nos Code Day of days			

Survey : 2

.*i. .

Scanning by Formic Ltd, London, (+44) 0171 924 1730

•

Page : 8

.



DRUGS SUPPLIED

Q130 Drug 2

0

0

Q136 Drug 8

1

٩

Q133 Drug 5

0123456789

T

Т Т

Supply (days) 1 2 3 4 5 6

T

Supply (days) 1 2 3 4 5 8 7 8

TT T 10

Т

1000

100

1111

.

7 8

3 4 5 6 7 6 9

100

5 6 7 8

10

1000

10

7 10

10

71

1

. 10 111111111

8

100

1

-



Q132	Drug 2	14 3 4	68	78					
	TT		Π	IT	1000				
	Π	Τ			100				
	Π	T	IT		10				
	Π				1				
Supply (days)									
° r h	τŤ	3 4	5 8		9 10				
h	$\frac{1}{1}$				=				
					'				

Q135	Drug 7		. ,						
Ť	İ Ţ	ΠŤ	ΓĹ	ГТ Т	1000				
					100				
	ПТ] 10				
	Π]ı				
Supply (days)									
Г,	<u>í</u>	r n	'nń	n	10				
	ΓT				1				

Q138	Dr	ug 10				-			
г	<u> </u>	ΤŤ	Ť	ГТ	<u>_</u>	'n	100		
Ē	Î	TT			+	П	100		
	Т	TT			1	П	10		
Ē	Ť	TT		T	1	П	1		
Supply (days)									
г	12					h	10		
	1	$\overline{\uparrow}$	\pm	+	1	۲	1		

0123455789

Supply (days) 1 2 3 4 5 5

1000

100

7 . . 10 ٦ı

10

Q141 Drug 13

ł

	0 1 2 3 4 6 6 7 8 9
· · · · ·	
	Q139 Drug 11
100	

2135	1	Dru 2	1 3	1	6	6	7			
Ĺ				Ĺ	Ē		Ċ		Ď	1000
Ē										100
										10
										1
٥	Supply (days)									
Ď	Ċ			Ň	Ů	Ĵ	ń	Ē	ń	10
							-			1

Q131 1	Drug 3	• • •							
Ť	ΓĪ.	ΓĪ	ТŤ	ΠĪ	1000				
\Box	Π	Π	Π	Π	100				
	Π				10				
		ΠT	Π		1				
Supply (days)									
Ĺ	ĒĒ	ÌÌ	ΤŤ	П	10				
	IT	TT	TT	$\overline{\Box}$	1				

Q13	4 1	Daru	26							
ĥ	$\frac{1}{1}$	2	3	4	6	•	ŕ	-	ů	1000
۲	╞	-	-		[-	-			100
۲	÷									10
									Η	1
L	Supply (days)									
م	1	2	3	4	5		^	8	-	10
2			_						۲	1
										•

Q137	Dru	; ?						
<u></u>	2	3 4	1 5	-	~	-	^	1000
⊢ +-	+		+	-				100
	4	┝╾┿╴	+					10
<u> </u>	누				_			
		1						1
0 1	2	3 4	5		7		9	
· I	Π							10
\square	\Box		Γ					1

Q140	Drug 2	12 3 4	5 6	7					
	TT		Ι		Τ	1000			
	TT	TT	Τ	Π		100			
Ī	TT	ΤΙ				10			
	TT	TT	Τ	Π		1			
Supply (days)									
Ů		ΪĪ		Ĺ.		10			
	П	TT				1			

Q142	Drug 1	4							
È	ŪΓ	ГТ.	ΙŢ	TT	, 1000				
	\Box		Π	Π] 100				
		П	Π	Π] 10				
				Π	ין				
Supply (days) 0 1 2 3 4 5 6 7 8 8									
				П] 10				
				Π	ין				



Survey: 2

1

Page : 10

Q1 Study Number

<u>02</u> CRYSTALLOID

INFUSIONS

	N/saline 500mls x2 x3 x4 x5 x6	x1				400000	500000	°000000	700000		<u>ا ا ا ا ا ا</u>					
	N/saline 100mls x2	x 1														
Q3	D/saline 500mls x2 x3 x4 x5 x6	x1		2000000	3000000	400000	<u>s</u> 	°000000	7000000		°000000	2 00000			13	
Q4			1	2	3	4	5	6	7	8	9	10	11	12	13	14
	Dextrose 5% 500 x2 x3 x4 x5 x6	0mls x														
	Dextrose 10% x2	x 1														
	Dextrose 20% x2	xl														
	Dextrose 50% x2	x1							8							

Page : 1

Q5	N/saline + 20kcl 500mls x1 x2 x3 x4					s 0000		10000		, 			1 3 0000	
	D/saline + 20kcl 500mlx1 x2 x3 x4													
	5% Dextrose + 20kcl 500mls x x2 x3 x4													
	20kci x1 20kci x2 20kci x3 20kci x4 20kcix5													
Q6	Heparin/saline 500mls x1	1	2 []	3 []	4	\$ []	6 []	, 0	8	9 []	10 []	12 []	13 []	14
	Hartmanns soln 500mls x1 x2 x3 x4													

Page : 2

Survey : 19

Contraction of the local division of the loc	

07 COLLOID

Y	Haemocell 500mls x1 x2 x3 x4 x5			<u> </u>				• - - - - - - - - - - - - -			¹³ 00000	
	x6 Hespan 500mls x1 x2 x3 x4 x5 x6											
	HAS 4.5% X1 X2 X3 X4 X5 X6											
	HAS 20% X1 X2											
	Blood x1 x2 x3 x4 x5 x6											
Q8	Platelets units x1 x2 x3 x4 x5 x6 Platelets pooled bag x1 x2			<u>, 000000</u> 000		100000 00		,000000 00				
Q9	FFP X1 FFPX2	2 	3	\$	6 	7 	»	。 □		12 	13 	! ₄

sena e

Page : 3

Scanning by Formic Ltd, London, (+44) 0171 924 1730



IV, IM & SC DRUGS WEEKS 1&2

Q2 Q3	Adenosine iv Adrenaline Bolus Amoxycillin Injection Argipressin Atropine Sulphate	1-6mg 1 in 1000 250mg TDS 500mg TDS 5-20iu 600mcg	x1 x2 x1 x2 x3 x1 x2 x3 x1 x2 x3 x1 x2 x3 x4	,0000000000000000000000000000000000000							
	Benzyll-penicillin	600mcg	BD QDS								
	Betamenthazone	1.2g 4mg	BD QDS xl						ğ		
			x2 x3 x4								
Q4Ca	lcium gluconate	10mg of 10%	xl x2				******				
	Calcium Chloride	10 mg	x1 x2	ŏ						<u>ן</u> ס	
	Cefotaxime Cefuroxime	2g 750mg	TDS TDS								
	Ceflaxidime	1.5g	TDS TDS								
	Chlorpheniramine	2g 10mg	BD xl						Ë		
		Ū	x2 x3						H	<u> </u>	ΪÄ
QChi	lopromazine	25mg-50mg	x1 x2								
	Co-Amoxiclav	1.2g	x3 TDS					ğ	H	ğ	
	Cyclizine	50mg	x1 x2							Ы	
				 	******	 			 Page	:1	
		. 32							TTT	\prod	

Q6 Q7	Dexamethasone Dextrose Diazemuls Diclofenac Digoxin inj Droperidol	4-8mg 50% minijet 5-10mg 11-20mg 75mg 125mcg 250mcg 10mg	x1 x2 x3 x1 x1 x2 x4 x1 x2 x4 x1 x2 x3 x1 x1 x1	20000 00000000000000000000000000000000	• <u>م</u> ممد محمد مدمد • • • • • • • • • • • • • • • •	
Q8	Ergometrine Erythromycin Etomidate	500mcg 500mcg 1g Bolus	x1 x2 QDS BD x1 x2			
Q9	Fleconide Flucloxacillin Flumazenil Fruseimid&Omgs/2mls	150mg 250mg 500mg	x1 x2 x3 QDS QDS x1 x2 x1 x2 x1 x2 x3 x4 x5 x6			
Q10	Gentamicin Goserelin Glucagon Glycopyronium	1-80mg 81-160mg 3.6mg 1 gm 600mg	BD TDS BD TDS x1 x1 x1 x1 x2 x3			
	Survey	: 32			 	Page : 2

	-

.

QII	Haloperidol Heparin s/c Heparin Infusion Hydralazine Hydrocortisone	5mg 5000u 0-20mg 100mg 200mg	x1 x2 BD TDS x1 x2 x1 x2 x1 x2 x1 x2 x3 x4 x1 x2 x3 x4 x4	10000000000000000000000000000000000000					
Q12	Hyocine Butylbromide Busca Hyocine Hydrobromide	pan 20mg 600mcg	x1 x2 x1 x2						
	Labetolol Leuprorelin Lorazepam	50-100mg 3.75mg 4mg	x1 x2 x1 x1]]]
	Metaclopromide Methyl-prednisolone	10mg 120mg daliya	x1 x2 x3 x4 x1						
Q13	Naloxone	400mcg	x1 x2 x3]]]
	Pabrinex I&II Phenytoin Phytomenadion Piperacillin Primaxin	100-250mg 10mg 2g 4g 500mg	OD x1 x2 x3 x1 QDS QDS BD TDS QDS						
	Survey : 32]	•			 Page	: 3	

																	I
Q14	Prochlorperazine	12.5mg	1 x1 72	Q	2	3		5	6	7		8	,			3 14	
	Promazine	50mg	x3 x1 x2 x4														
	Protamine Promethazine	up to 50mg 25-50mg	xl xl x2														
	Ranitidine	50mg	TDS				0	0		٥		۵				ם כ	
	Tranexamic-acid	500mg	TDS	0								٥		0		ם כ	
	Vancomycin	500mg	xl x2 x3								****						
	Verapermil	5mg	x1 x2 x3														

Survey : 32	Page : 4
Scanning by Formic Ltd, London, (+44) 0171 924 1730	

I	NVESTIGATIONS	8				
Q2	Cardiac Tests 1 ECG 2 ECGs 3 ECGs 4 ECGs 5 ECGs more than 5 24 hour tang			7000000	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	
Q3	exercise ECG echocardiogram Endoscopies Bronchoscopy Colonoscopy ERCP Gastroscopy					
Q4	Sigmoidoscopy X Rays Chest X-ray x 1 x2 x3 X Pays	PA 0 0			8 9 10 11 12 13 14 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
¢9	Group A x 1 x2 x3 x4 x5 x6				8 9 10 11 12 13 14 14 15 14 15 14 15 16 16 16 16 12 13 14 16 16 14 15 16 16 16 15 16 16 16 16 16 16 16 16 16 17 16 16 16 16 18 16 16 16 16 19 16 16 16 16 19 16 16 16 16 19 16 16 16 16 10 16 16 16 16 10 16 16 16 16 10 16 16 16 16 11 12 16 16 16 16 12 16 16 16 16 16 13 16 16 16 16 16	
Q6	х кауз Group B x 1 x2 x3 x4 x5 x6				8 9 10 11 12 13 14 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	

Study Number

ĺ

Page : 1

Scanning by Formic Ltd, London, (+44) 0171 924 1730

Survey : 3

.

415

ł

Q7	X Rays	1 2 3 4 5 6 7	8 9 10 11 12 13 14	
	Group C x 1 x2			
Q8	X Rays	1 2 3 4 5 6 7	8 9 10 11 12 13 14	
	GroupD x 1			
	Group E x1 Group F x1			
	Nuclear Med scan x I			
	X-Rays that are not found on list Day	X-Rays that are Day	not found on list	
	Test	Test		
Q9	December of Test		8 9 10 11 12 13 14	
	Urine Flow Studies			
QI	D	Q11	Q12	
Tes	t1: 0 1 2 3 4 5 8 7 8 0 1			י
Day				
				<u>i i i i i i i i i i i i i i i i i i i </u>
	Lines/Drains/Pack	S		
Q13	Lines	_	R Q 10 11 12 13 14	
		1234507		
	Peripheral x1			
	x2 x3			
	x4			
	central line x1 x2			
	x3			
	Tripple lumen x1 x2			
	Hickman			
	Swan ganz			
	Arterial Epidural Catheter			
	Feeding line			
Q14	Packs etc. Vaginal pack			
	Nasal pack Nasal pack splints			
	Survey : 3		Page : 2	1

Corrugated drain x1x2x3x4Corrugated drain x1x2Robinson Portex Drain x1x2x3x4Chest Drain x1x2x3x4NG tube - bile drainageT tubeUrinary CatheterRe-catheterisedSuprapubic CatheterNephrostomy tube		
Q16 Other Tracheotomy Tube ET Tube Pacing Wire		
<u>Physiotherapy, OT, Sp</u>	eech Therapy etc	
Q17 Physio - Face to Face contacts x1 x2 x3 x4 x5 x6 x7 x8 x9 x10	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	
Q18 Dietician x1 Dietician x2 Social worker Speech therapy x1 x2 x3 Spec. stoma nrs Macmillan Nrs Diabetic Nrs Spec Continence Nurse	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	8 9 10 11 12 13 14 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 <t< th=""></t<>
Survey : 3	Scanning by Formic Ltd. London, (+44) 0171	Pege : 3

Q19	OT x1 x2 x3 x4 x5 x6 x7 x8								
Q20	OT Home Visit Day of Visit 1 0 1 2 3 4 5 6 7 8 9 Units supplied 1 0 1 2 3 4 5 6 7 8 9 10 1 0 1 2 3 4 5 6 7 8 9 10 1	Day of Visit 2 0 1 2 3 4 6 6 Units supplied 2 0 1 2 3 4 6 6 Units supplied 2							
Q21	Dressings/nursing procedures Wound care Number of wounds								
Q22	Wound care Wound redressed X1 X2 X3 X4 X5 >5 Stiches/clips removed	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	8 9 10 11 12 13 14 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0						

Q23 Traction/Plaster/TEDS

200000 °_____ Skin traction set up Check traction c/o new plaster Limb plaster renewed TED stockings **CPM** machine Nursing Care **O24** Personal Hygiene 1 2 3 4 5 6 7 8 9 10 11 12 13 14 Bed Bath Assisted wash Full bath Assistance with dressing Survey : 3 Page : 4

~~~~

| Q25       | Elimination<br>c/o urinary incontinence<br>c/o faecal incontinence<br>c/o illeostomy/colostomy<br>c/o bladder irrigation<br>bladder washout<br>Assistance to the toilet<br>Manual evacuation<br>Peritoneal Dialysis                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                       | 8       9       10       11       12       13       14         11       12       13       14         11       12       13       14         11       12       13       14         11       12       13       14         12       14       14       14         14       14       14       14         14       14       14       14         15       14       14       14         14       14       14       14         14       14       14       14         14       14       14       14         14       14       14       14         14       14       14       14         14       14       14       14         14       14       14       14         14       14       14       14         14       14       14       14         14       14       14       14         14       14       14       14         14       14       14       14         14       14       14 <t< th=""></t<> |
|-----------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Q26       | Prevention of presure sores,<br>DVT etc<br>2-4 hourly change of position<br>1 nurse<br>2-4 hourly change of position -<br>2 nurses +                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 1 2 3 4 5 6 7<br>0 0 0 0 0 0 0 0 0<br>0 0 0 0 0 0 0 0 | 8 9 10 11 12 13 14<br>C C C C C C C C<br>C C C C C C C C C                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| Q27       | Encouragement /education<br>deep breathing exercises,<br>leg exercises<br>Mobility<br>Bedfast/chairfast<br>Assistance needed in<br>and out of bed<br>Assistance with walking                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                       | $\begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| Q28       | <ul> <li>Observations</li> <li>Post-op obs</li> <li>Hourly TPR/BP/O2 SAT/PCA</li> <li>4 hourly TPR &amp; BP</li> <li>BD/Daily Obs.</li> <li>Neuro Obs.</li> <li>Stool Chart</li> <li>Pain Chart</li> <li>PV Loss Chart</li> <li>HOURLY BMS</li> <li>4-6 Hourly BMS</li> <li>Daily urinalysis</li> <li>4-6 Hourly Urinalysis</li> <li>6 Hourly Urinalysis</li> <li>6 Hourly Urinalysis</li> <li>Fluid Balance Chart</li> <li>Weight recorded</li> <li>Peak Flows</li> <li>Continuous cardiac monitor</li> <li>Swan ganz studies</li> <li>Hourly ventilator checks</li> <li>CVP Readings</li> <li>Blood gasses 1-5</li> <li>Blood gasses 1-5</li> <li>Blood gasses 16-20</li> <li>Blood gasses 21-25</li> <li>C/O Fall</li> </ul> |                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| - <u></u> | Survey : 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |

· •

Page : 5

| Q29 | Chest physic by nurse etc                                                                                                                                                                     |         | 1       | 2 | 3           | 4 | 5            | 6                | 7     |   | 8           | 9            | 10            | 11                                      | 12       | 13 14                                     |
|-----|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------|---------|---|-------------|---|--------------|------------------|-------|---|-------------|--------------|---------------|-----------------------------------------|----------|-------------------------------------------|
|     | ET suction<br>2-4 hourly chest physio<br>c/o oxygen therapy<br>c/o CPAP                                                                                                                       |         |         |   |             |   |              |                  |       |   |             |              |               |                                         |          |                                           |
| Q30 | Special Bed/mattress<br>Foam/Spenco mattress<br>Ripple mattress<br>Low air loss bed<br>Water/air fluidise bed                                                                                 |         |         |   | 30000       |   | <b>5</b><br> | 6<br>0<br>0<br>0 | 70000 |   |             | <b>,</b><br> | <b>°</b> 0000 |                                         |          | 13 14<br>0 0<br>0 0<br>0 0                |
| Q31 | Nutrition<br>Help with fluids and meals<br>Food chart maintained<br>c/o ng feed<br>c/o TPN<br>c/o gastrostomy feed line<br>Nutrison x1<br>x2<br>x3<br>x4<br>Teaching<br>pt/relative education |         |         |   | 400000000 D |   |              |                  |       |   | •00000000 0 | 200000000 O  | "00000000 O   | 120000000000000000000000000000000000000 | "<br>    | *<br>000000000000000000000000000000000000 |
| Q32 | Mental state/sensory<br>defects<br>c/o acute confusion                                                                                                                                        | 1<br>[] | 2<br>[] | 3 | 4           | 5 | 6<br>[]      | 7                |       | 8 | <b>°</b>    | 10<br>[]     |               | 12<br>[]                                | 13<br>[] | 14<br>□                                   |
|     | emotional support<br>needed for pt +/or relatives<br>unconcious                                                                                                                               |         |         |   |             |   |              |                  |       |   |             |              |               |                                         |          |                                           |
|     | impaired speech/sight/<br>hearing/language<br>difficulties                                                                                                                                    |         |         | ٥ |             |   |              |                  |       | ٥ | ٥           | ۵            |               |                                         |          |                                           |
| Q33 | Other<br>Cardio version<br>Cardiac arrest<br>Barrier nursing                                                                                                                                  |         |         |   |             |   |              |                  |       |   |             |              |               |                                         |          |                                           |
| Q34 | Haemofiltration/Dialysis<br>Haemofiltration<br>Set change x1<br>Set change x2<br>Set change x3<br>Dialyisis                                                                                   |         |         |   |             |   |              |                  | ]     |   |             |              |               |                                         |          |                                           |

Survey : 3

u ..., 🖛

Page : 6

Scenning by Formic Ltd, London, (+44) 0171 924 1730









| Study Number | <br> |   |  |
|--------------|------|---|--|
|              |      |   |  |
|              |      |   |  |
|              |      | 1 |  |
|              |      | 1 |  |
|              |      | 1 |  |
|              |      |   |  |
|              |      |   |  |
|              |      |   |  |

# LAB TESTS: Day1-Day7

Q3

. .

# Q2 Biochemistry, Haemotology, Histopathology & Cytopathology

|                                                                           |                                        | P/A | A&E | 1 o/c | 2 o/c | 3 o/c | 4 | o/c 5 o/c 6 o/c 7 o/c |
|---------------------------------------------------------------------------|----------------------------------------|-----|-----|-------|-------|-------|---|-----------------------|
| AIAT<br>ABHB<br>ADNA<br>ALBU<br>ALK P<br>ALT<br>AMYL<br>ANA<br>AST<br>ATA | x1<br>x2                               |     |     |       |       |       |   |                       |
| BICA<br>BILI<br>BJP<br>BM                                                 |                                        |     |     |       |       |       |   |                       |
| CALC<br>CE<br>CHOL<br>CK<br>COAGS<br>COMP<br>CORT<br>COULT                | X1<br>X2<br>X3<br>X1<br>X2<br>X3<br>X4 |     |     |       |       |       |   |                       |
| CREA<br>CREA C<br>CRP                                                     | L                                      |     |     |       |       |       |   |                       |
| DIFF                                                                      | *****                                  |     |     |       |       |       |   |                       |
| ESR                                                                       |                                        | 0   |     |       |       |       |   |                       |



Scanning by Formic Ltd, London, (+44) 0171 924 1730

Page : 1

| 1999.<br>1 |                                                           |            |  |  |                          |          |  |  |  |  |
|------------|-----------------------------------------------------------|------------|--|--|--------------------------|----------|--|--|--|--|
| Q4         | FARA<br>FBC<br>FDP<br>FE ST                               | P/A        |  |  |                          |          |  |  |  |  |
|            | FL GLU<br>FL PROT<br>FOB<br>FSH                           |            |  |  |                          |          |  |  |  |  |
|            | GASE<br>GF<br>GGT<br>GLUC<br>GLUC F<br>GLYCA<br>GTT<br>GP |            |  |  |                          |          |  |  |  |  |
| QS         | HBD<br>HEP x1<br>x2                                       |            |  |  | % <b>4</b><br>0 0<br>0 0 |          |  |  |  |  |
|            | IGA<br>IGE<br>IGG<br>IGM<br>IGS<br>INSU<br>IRON           |            |  |  |                          |          |  |  |  |  |
|            | KL                                                        |            |  |  |                          | 000000   |  |  |  |  |
|            | LAP<br>LFT<br>LH<br>LIB<br>LITH                           |            |  |  |                          |          |  |  |  |  |
| Qđ         | 4AGN                                                      |            |  |  |                          |          |  |  |  |  |
|            | NA/K.                                                     |            |  |  |                          |          |  |  |  |  |
|            | OSMO                                                      | Б          |  |  |                          |          |  |  |  |  |
|            | PHENY<br>PRC<br>PROG<br>PROL<br>PROT<br>PRS<br>PSA        |            |  |  |                          |          |  |  |  |  |
|            |                                                           | Survey : 4 |  |  |                          | Page : 2 |  |  |  |  |
|            |                                                           |            |  |  |                          |          |  |  |  |  |

| <b>4</b> Q7 | RA<br>RC<br>RCF<br>RU BILI<br>RU CREA<br>RU NA/K<br>RU OSMO<br>RU PORP<br>RU UBG<br>RU UREA                                               |          |                     |                                        |          |
|-------------|-------------------------------------------------------------------------------------------------------------------------------------------|----------|---------------------|----------------------------------------|----------|
|             | SB 12<br>SCF GLU<br>SFER<br>· SPE<br>SSP                                                                                                  |          |                     |                                        |          |
| Q8          | TEST<br>THEO<br>THY<br>TRIG                                                                                                               |          |                     |                                        |          |
|             | U CALC<br>U CALC<br>U CREA<br>U NA/K<br>U OXAL<br>U PHOS<br>U PORP<br>U PROT<br>U URAT<br>U URAT<br>U UREA<br>U VMA<br>U/E x1<br>x2<br>x3 |          | 0000000000 00<br>00 |                                        |          |
| Q9          | URAT<br>UREA                                                                                                                              |          |                     |                                        |          |
|             | V CORT                                                                                                                                    | 0        |                     |                                        |          |
|             | XM x1<br>x2                                                                                                                               |          |                     |                                        |          |
| Q10         | 1FS<br>2FS<br>2B<br>3B<br>4B<br>5B<br>6B<br>7B<br>8B<br>9B<br>10B                                                                         |          |                     |                                        |          |
|             |                                                                                                                                           | Survey : | 4                   | Sempine by Formie Ltd Lender (444) 013 | Page : 3 |

| Q11 | AFB<br>ASO<br>AWO                          | P/A | A&E |     |    |    |         |
|-----|--------------------------------------------|-----|-----|-----|----|----|---------|
|     | BC                                         | ۵   | ۵   |     |    |    |         |
|     | CAPD<br>CHLAM<br>CMV                       |     |     |     |    |    |         |
|     | DIF<br>DIS                                 | B   |     |     |    |    |         |
|     | EAR<br>EYE                                 |     |     |     |    |    |         |
|     | FLUID<br>FU                                |     |     | 888 | 88 | 88 | 8888888 |
|     | GA<br>GENT<br>GNT                          |     |     |     |    |    |         |
| Q12 | HAV<br>HBSAG<br>HEPC<br>HIV<br>HOLD<br>HSV |     |     |     |    |    |         |
|     | IUC                                        | 0   |     |     |    |    |         |
|     | LEP<br>LP                                  |     |     |     |    |    |         |
|     | MRSA<br>MS                                 |     |     |     |    |    |         |
|     | NS                                         | ٥   | ٥   |     |    |    |         |
|     | PARV<br>PEN<br>PLEUR<br>PUS                |     |     |     |    |    |         |

# Survey : 4

Scanning by Formic Ltd, London, (+44) 0171 924 1730

ł

.....



I



# **Infection 1**



# Infection 2



**Infection 3** 



# **Infection 4**







# Members of the Socio-economic Burden of Hospital Acquired Infection Project Team, Steering Committee and Advisory Group

# A8.1 Introduction

This appendix provides details of the members of the Socio-economic Burden of Hospital Acquired Infection Project Team, Steering Committee, and Advisory Committee.

# A8.2 Project Team

Rosalind Plowman BA, MSc, RN Nicholas Graves, BA, MA Mark Griffin, BA, MSc Rachael Dunk, RN Alison Franklin, BSc, RN Janette Trevarthen, RN, OHND Maggie Waters, RN Jennifer White, RN Lynda Wright, BSc, RN Barbara Ayres Christine Berry Project Co-ordinator Research Economist Statistician Research Assistant Research Assistant Research Assistant Research Assistant Research Assistant Research Assistant Project Secretary Administrative support
# A8.3 Project Steering Committee

| Jane Bandcroft RN, DN, HV, FPCert          | Director of Nursing Practice, Study<br>Hospital                                                                                                                         |
|--------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Barry Cookson MB, BDS, FRCP Path           | Director of laboratory of Hospital<br>Infection, Public Health Laboratiry<br>Service                                                                                    |
| Helen Glenister BSc, PhD, RN               | Nursing Director, Medical Devices<br>Agency, DOH                                                                                                                        |
| Bernadette Nazareth MBBS, MSc,<br>MRCPath  | Consultant in Communicable Disease<br>Control, Redbridge and Waltham<br>Health Authority                                                                                |
| Jennifer A Roberts MSc Econ, PhD,<br>HMPHM | Reader in Economics of Public Health,<br>London School of Hygiene and<br>Tropical Medicine, London                                                                      |
| Mike Rowland, MBBS, FRCP, FFPHM            | Consultant Epidemiologist,<br>Communicable Disease Surveillance<br>Centre, Anglia and Oxford RHA                                                                        |
| Tony Swan PhD, HMFPHM, C Stat              | Director, Statistical Unit, Public Health<br>Laboratory Service                                                                                                         |
| Lynda Taylor, RN, RM                       | Head of Nursing and Infection Control<br>Unit, Laboratory of Hospital Infection,<br>Public Health Laboratory Service                                                    |
| Jennie Wilson, BSc, RN                     | Programme Leader Infection Control<br>and Surveillance, Nursing and Infection<br>Control Unit, laboratory of Hospital<br>Infection, Public Health Laboratory<br>Service |

# A8.4 Advisory Committee Members

| Kay Butcher                                         | Department of Health (to Oct 1995)                                                |
|-----------------------------------------------------|-----------------------------------------------------------------------------------|
| Millie Carter                                       | Nursing Officer, Department of Health (to Sep 1994)                               |
| Sue Dewar RN, DNCert, MA                            | District Nurse, Chichester Primary<br>Care Services Trust.                        |
| Georgia Duckworth MSc, FRCP, FRC<br>Path, FRIPHH    | Regional Epidemiologist, CDSC,<br>North Thames                                    |
| Johanna Finn BSc, MHSM, Dip HSM                     | Chief Executive, West Suffolk<br>Hospitals Trust                                  |
| Paul Gillett, MRCP, FRC Path                        | Consultant Medical Microbiologist,<br>Stoke Mandeville Hospital                   |
| Simon Harding                                       | Economic Advisor, Department of Health (to Sept 1995)                             |
| Jayne Holmes RN                                     | Senior Nurse, Queen Elizabeth II<br>Hospital, Welwyn Garden City (to Nov<br>1996) |
| Rosemary Jenkins MSc, DMS, MTD,<br>RN, RM           | Nursing Officer, Department of Health (from Sept 1994 to Sept 1995)               |
| Bill Maton-Howarth PhD                              | Department of Health (from 1 April 1997)                                          |
| Jennifer McIntyre RN, MSc                           | Nursing Officer, Department of Health (from Sept 1995)                            |
| Elizabeth Meerabeau BSc, PhD, RN,<br>RHV, RNT, RHVT | Department of Health (to Mar 1997)                                                |
| Richard Murray                                      | Economic Advisor, Department of Health (from Sept 1995)                           |
| Elizabeth Tebbs MBBS, ChB                           | Senior Medical Officer, Department of<br>Health                                   |
| Ann Whittle                                         | Department of Health (to Oct 1995)                                                |

# Data variables included in the data analysis

### A9.1 Introduction

This appendix provides further details regarding the variables included in the analysis of the incidence of HAI, and the impact HAIs had on hospital costs and length of hospital stay.

# A9.2 Explanatory variables used in the regression analysis

# A9.2.1 Overview of variables used

Table A9.1 provides details of the explanatory variables used in the data analysis.

### A9.2.2 Admission type classification system

Patients were classified as being admitted via one of nine routes: elective via pre-admission, elective direct to the ward, urgent direct to the ward, emergency via the accident and emergency department, emergency via the out-patients department, emergency GP referral direct to the ward, transfer from another ward via the day hospital, transfer from another hospital and finally via another unspecified route. As indicated in Table A9.1, for the purpose of the analysis these nine admission types were compressed into three main categories: emergency admissions, elective admissions and transfers. Table A9.2 shows how the nine admission types were compressed into three main categories.

|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | _       |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------|
| Sex Male                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |         |
| Female                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |         |
| Age 18-34                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |         |
| 35 - 54                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |         |
| 55-74                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |         |
| 75 +                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |         |
| Admission specialty Surgery                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |         |
| Orthopaedics                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |         |
| Urology                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |         |
| Gynaecology                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |         |
| Obstetrics                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |         |
| Admission type* Elective                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |         |
| Bmergency                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |         |
| Transfer                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |         |
| Diagnosis group* Primary discharge diagnosis grouped into one of 14                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | ICD9    |
| categories                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |         |
| Number of co-morbidities None                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |         |
| One                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |         |
| 1 Wo or more                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | <u></u> |
| Body Mass Index BMI < 20                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |         |
| BMI 20-530                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |         |
| Diviliation Vee No                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |         |
| Diabetes Tes, No                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |         |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |         |
| Cancer present provide UTI PTI Vas No                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |         |
| Endourachean tube present prof LK11 16, 100                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |         |
| Wound drain prior to 5 wi                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |         |
| Antibiolos administered provide each type of microlon 1 res, roo                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |         |
| 1V/IM antibiolos administered prior to each type of intection 1 res, roo                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |         |
| HAI status HAI not identified during the in-                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |         |
| The net detailed that is the second s |         |
| Type of HAT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |         |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |         |
| a suprise women infections                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |         |
| lower receivator tract infections                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |         |
| blood stream infections                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |         |
| skin infections                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |         |
| infections at other sites                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |         |
| multiple infections at more that                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | one     |
| sito)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |         |

# Table A9.1: Explanatory variables as used in the data analysis

\*Further details of the classification system used are presented below

# Table A9.2: Admission type classification system

| Admission categories used in the analysis | Admission types within each category  |
|-------------------------------------------|---------------------------------------|
| Elective                                  | Elective via pre-admission clinic     |
|                                           | Elective                              |
| Emergency                                 | Urgent direct to the ward             |
|                                           | Emergency via accident and emergency  |
|                                           | Emergency via out patients department |
|                                           | GP referral                           |
|                                           | Via day hospital                      |
| Transfers                                 | Transfer from another ward            |
|                                           | Transfer from another hospital        |

# A9.2.3 Primary diagnosis classification system

The primary discharge diagnosis were coded using the 9th revision of the International Classification of Diseases system (ICD9). Each primary discharge diagnosis was allocated either a three or four digit code depending on the amount of information available. These diagnosis were initially grouped into 19 disease categories according to the ICD9 classification system and then further compressed into 14 categories. Table A9.3 provides details of the ICD9 categories included in each of the 14 study categories.

| Study disease<br>categories 1-14 | ICD9 Disease<br>categories<br>included in<br>study<br>categories | Title of Categories                                                   |
|----------------------------------|------------------------------------------------------------------|-----------------------------------------------------------------------|
| 1                                | 1                                                                | Infectious and parasitic disease                                      |
| 2                                | 11                                                               | Neoplasms                                                             |
| 3                                | 111                                                              | Endocrine, nutritional and metabolic diseases, and immunity disorders |
| 4                                | IV                                                               | Diseases of blood and blood forming organs                            |
| 5                                | VI                                                               | Diseases of the nervous system and sense organs                       |
| 6                                | VII                                                              | Diseases of the circulatory system                                    |
| 7                                | VIII                                                             | Diseases of the respiratory system                                    |
| 8                                | IX                                                               | Diseases of the digestive system                                      |
| 9                                | ×                                                                | Diseases of the genitourinary system                                  |
| 10                               | XI                                                               | Complications of pregnancy, childbirth and puerperium                 |
|                                  | xv                                                               | Certain conditions originating in the perinatal period                |
| 11                               | XII.                                                             | Diseases of the skin and subcutaneous tissue                          |
| 12                               | XIII                                                             | Diseases of the musculoskeletal system and connective tissue          |
| 13                               | XVII                                                             | Injury and Poisoning                                                  |
| 14                               | XVI                                                              | Symptoms, signs and ill- defined conditions                           |
|                                  | V                                                                | Mental disorders                                                      |
|                                  | XIV                                                              | Congenital abnormalities                                              |

### Table A9.3: Diagnosis group classification system.

# A9.2.4 HAI classification system

HAIs identified during the in-patient phase were classified into eight mutually exclusive infection groups: no HAI, infections of the urinary tract, surgical wounds, lower respiratory tract, bloodstream, skin, other sites, and in those cases where patients acquired more than one infection, multiple infections. Table A9.4 provides details of the sites of infection included in the six single site infection groups.

| HAI groups used in the analysis                   | Sites of infection included in the analysis | No. of        | Percentage    |
|---------------------------------------------------|---------------------------------------------|---------------|---------------|
|                                                   | groups*                                     | infections    | of infections |
|                                                   |                                             | identified at | identified at |
|                                                   |                                             | each site     | each site     |
| Urinary tract infections                          | UTI Clinical                                | 39            | 18.8          |
|                                                   | UTI Clinical and laboratory evidence        | 61            | 29.3          |
|                                                   | Asymptotic Bacteriuria                      | 0             | 0.0           |
| Lower respiratory tract infections                | Pneumonia                                   | 10            | 4.8           |
|                                                   | Chest Infection                             | 13            | 6.3           |
|                                                   | Lower respiratory tract                     | 0             | 0.0           |
| Surgical wound infections                         | Major SWI                                   | 6             | 2.9           |
|                                                   | Minor SWI                                   | 34            | 16.3          |
| Bloodstream infections**                          | Septicaemia                                 | 3             | 1.4           |
|                                                   | Bacteraemia                                 | 1             | 0.5           |
| Skin infections                                   | Skin                                        | 15            | 7.2           |
|                                                   | Skin Ulcers                                 |               | 0.0           |
| · · · · · · · · · · · · · · · · · · ·             | Pressure sores                              | 1             | 0.5           |
| ₩ <sup>₩</sup> ₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩ | Blistered skin                              |               | 0.0           |
| Infections at other sites                         | Major accidental wound                      | 1             | 0.5           |
|                                                   | Minor accidental wound                      |               | 0.0           |
|                                                   | Upper respiratory tract and ear             | 2             | 1.0           |
|                                                   | EYE                                         |               | 0.0           |
|                                                   | Otitis externa                              | 2             | 1.0           |
|                                                   | Oral thrush                                 | 2             | 1.0           |
| ······································            | Upper respiratory tract - unspecific        | 1             | 0.5           |
|                                                   | Mouth                                       | 3             | 1.4           |
|                                                   | Other genital tract                         | 6             | 2.9           |
|                                                   | GI                                          | 1             | 0.5           |
|                                                   | Other abdominal                             | 2             | 1.0           |
|                                                   | Other                                       | 3             | 1.4           |
|                                                   | Systemic                                    | 1             | 0.5           |
|                                                   | Pericarditis                                | 1             | 0.5           |

### Table A9.4: HAI classification system

Note: Where more than one infection had been acquired these were classified as multiple infections

\* Definitions of the infections listed can be found in Appendix 5

# The age, sex, admission type and specialty distribution of study participants and eligible patients who were not recruited into the study

### A10.1 Introduction

Between April 1994 and May 1995, 3534 patients admitted to the surgical specialties covered in this study were eligible for recruitment. Of these patients, 2477 (70.1%) patients were recruited into the study, 197 (5.6%) declined participation, 851 (24.1%) were not recruited due to practical reasons such as insufficient time, and in a further nine (0.3%) cases the reason for non-recruitment was not specified. In order to check how representative the study sample was of the wider eligible population the following data were recorded for all eligible patients who were not recruited: sex, age, admission type and admission specialty. The age, sex, admission type and specialty distributions of study participants and eligible patients who were not recruited into the study were subsequently analysed and compared. The results of this analysis are presented in this appendix.

# A10.2 The distribution of study participants and eligible patients who were not recruited by sex

Table A10.1 provides details of the sex distribution of study participants and eligible patients who either declined participation, or were not recruited due to practical reasons such as insufficient time. Table A10.2 shows the sex distribution of study participants and eligible patients who were not recruited, together with that, which would have been present if all eligible patients had been recruited.

# Table A10.1: Sex distribution of study patients and eligible patients who

| ,       |                      |       |                                   |      |                                                       |      |                                           |      |                                           |      |
|---------|----------------------|-------|-----------------------------------|------|-------------------------------------------------------|------|-------------------------------------------|------|-------------------------------------------|------|
| Sex     | Study<br>Participant |       | Patient declined<br>participation |      | Patient not<br>recruited due to<br>practical reasons* |      | Reason for non-<br>recruitment<br>unknown |      | All eligible<br>patients not<br>recruited |      |
|         | n                    | %     | n.                                | %    | n                                                     | %    | n                                         | %    | n                                         | %    |
| Male    | 1003                 | 40.62 | 98                                | 49.7 | 423                                                   | 49.7 | 4                                         | 28.6 | 525                                       | 49.4 |
| Female  | 1466                 | 59.38 | 97                                | 49.2 | 425                                                   | 49.9 | 5                                         | 35.7 | 527                                       | 49.6 |
| Unknown | 0                    | 0.0   | 2                                 | 1.0  | 3                                                     | 0.4  | 5                                         | 35.7 | 10                                        | 0.9  |
| TOTAL   | 2469                 | 100   | 197                               |      | 851                                                   |      | 14                                        |      | 1062                                      |      |

### were not recruited

\* For example insufficient time

# Table A10.2:Sex distribution of study participants, eligible patients who were<br/>not recruited, and that which would have been present if all<br/>eligible patients were recruited

| Eligible patients     | Males | Females | All patients | Proportion (%) (95% CI) of male patients | P value |
|-----------------------|-------|---------|--------------|------------------------------------------|---------|
| Recruited             | 1003  | 1466    | 2469         | 40.6 (38.7; 42.6)                        | <0.001  |
| Not - recruited*      | 525   | 527     | 1052         | 49.9 (46.8; 53.0)                        |         |
| All eligible patients | 1528  | 1993    | 3521         | 43.4 (41.8; 45.1)                        |         |

\*An additional 10 patients were not recruited however their sex was not recorded, consequently they have been excluded from this analysis

The percentage of male patients in the cohort of recruited patients was 8.3% higher than in the cohort of patients who were not recruited. This difference was found to be highly significant. However, the percentage of male patients in the recruited cohort was only 2.8% less than the intended cohort. This difference is extremely unlikely to lead to bias in estimates of how those with and without HAI compare.

# A10.3 The distribution of study participants and eligible patients who were not recruited by age group

Table A10.3 provides details of the age distribution of study participants and eligible patients who either declined participation, or were not recruited due to practical reasons such as insufficient time. Table A10.4 shows the age distribution of study participants and all eligible patients who were not recruited, together with that which would have been present if all eligible patients had been recruited.

|           | Reason for not recruiting eligible patients |       |                                |       |                                                       |       |                                           |       |                                           |       |
|-----------|---------------------------------------------|-------|--------------------------------|-------|-------------------------------------------------------|-------|-------------------------------------------|-------|-------------------------------------------|-------|
| Age Group | ge Group Study<br>Participant               |       | Patient declined participation |       | Patient not<br>recruited due to<br>practical reasons* |       | Reason for non-<br>recruitment<br>unknown |       | All eligible<br>Patients not<br>recruited |       |
|           | n                                           | %     | n.                             | %     | n                                                     | %     | n.                                        | %     | n                                         | %     |
| 18-30     | 363                                         | 14.7  | 15                             | 7.6   | 79                                                    | 9.3   | 1                                         | 11.1  | 95                                        | 9.0   |
| 31-40     | 337                                         | 13.6  | 17                             | 8.6   | 91                                                    | 10.7  | 2                                         | 22.2  | 110                                       | 10.4  |
| 41 - 50   | 336                                         | 13.6  | 13                             | 6.6   | 127                                                   | 14.9  | 1                                         | 11.1  | 141                                       | 13.3  |
| 51 - 60   | 375                                         | 15.2  | 43                             | 21.8  | 174                                                   | 20.4  | 1                                         | 11.1  | 218                                       | 20.6  |
| 61 - 80   | 915                                         | 37.1  | 80                             | 40.6  | 244                                                   | 28.7  | 3                                         | 33.3  | 327                                       | 30.9  |
| 81 - 100  | 143                                         | 5.8   | 28                             | 14.2  | 60                                                    | 7.1   | 1                                         | 11.1  | 89                                        | 8.4   |
| Unknown   |                                             | 0.0   | 1                              | 0.5   | 76                                                    | 8,9   | 0                                         | 0.0   | 77                                        | 7.3   |
| TOTAL     | 2469                                        | 100.0 | 197                            | 100.0 | 851                                                   | 100.0 | 9                                         | 100.0 | 1057                                      | 100.0 |

 Table A10.3: Distribution of study participants and eligible patients who were not recruited by age group

\*For example insufficient time

# Table A10.4: Age distribution of study participants, eligible patients who were not recruited, and that which would have been present if all eligible patients were recruited

| Eligible patients     | Age g | roup | All patients | Proportion (%) of patients aged 18-60 | P value |
|-----------------------|-------|------|--------------|---------------------------------------|---------|
|                       | 18-59 | 60+  |              | (95% CI)                              |         |
| Recruited             | 1411  | 1058 | 2469         | 57.1 (55.2; 59.1)                     | 0.8126  |
| Not - recruited*      | 564   | 418  | 980          | 57.6 (54.4; 60.7)                     |         |
| All eligible patients | 1975  | 1474 | 3449         | 57.3 (55.6; 58.9)                     |         |

\*An additional 77 patients were not recruited however their age was not recorded, consequently they have been excluded from this analysis

The results presented in Table A10.4 indicate that the proportion of study participants who were over 60 years of age was almost identical to that which would have been present if all eligible patients were recruited.

# A10.4 The distribution of study participants and eligible patients who were not recruited by admission type

Table A10.5 provides details of the admission type distribution of study participants and eligible patients who either declined participation, or were not recruited due to practical reasons such as insufficient time. Table A10.6 shows the admission type distribution of study participants and all eligible patients who were not recruited, together with that which would have been present if all eligible patients had been recruited.

 Table A10.5: Distribution of study participants and eligible patients who were not recruited by admission type

|                   |                      |       | Reason for not recruiting eligible patients |       |                                                          |       |                                           |       |                                           |      |  |
|-------------------|----------------------|-------|---------------------------------------------|-------|----------------------------------------------------------|-------|-------------------------------------------|-------|-------------------------------------------|------|--|
| Admission<br>Type | Study<br>Participant |       | Patient declined<br>participation           |       | Patient not<br>recruited due to<br>practical<br>reasons* |       | Reason for non-<br>recruitment<br>unknown |       | All eligible<br>Patients not<br>recruited |      |  |
|                   | n                    | %     | n.                                          | %     | n                                                        | %     | n                                         | %     | n                                         | %    |  |
| Elective          | 1629                 | 66.0  | 118                                         | 59.9  | 421                                                      | 49.5  | 4                                         | 44.4  | 543                                       | 51.4 |  |
| Emergency         | 840                  | 34.0  | 78                                          | 39.6  | 425                                                      | 49.9  | 5                                         | 55.6  | 508                                       | 48.1 |  |
| Unknown           | 0                    | 0.0   | 1                                           | 0.5   | 5                                                        | 0.6   | 0                                         | 0.0   | 6                                         | 0.6  |  |
| TOTAL             | 2460                 | 100.0 | 407                                         | 400.0 |                                                          | 400.0 |                                           |       | 4087                                      | 100. |  |
| TOTAL             | 2409                 | 100.0 | 197                                         | 100.0 | 851                                                      | 100.0 | 9                                         | 100.0 | 105/                                      | 0    |  |

\*For example, insufficient time

# Table A10.6: Admission type distribution of study participants, eligible patientswho were not recruited, and that which would have been present ifall eligible patients were recruited

| Eligible patients     | Admis    | sion type | All      | Proportion (%) of patients who were | P value |
|-----------------------|----------|-----------|----------|-------------------------------------|---------|
|                       | Elective | Emergency | patients | elective admissions (95% Ci)        |         |
| Recruited             | 1629     | 840       | 2469     | 68.0 (64.1; 67.8)                   | <0.001  |
| Not - recruited*      | 543      | 508       | 1051     | 51.7 (48.6; 54.7)                   |         |
| All eligible patients | 2172     | 1348      | 3520     | 61.7 (60.1; 63.3)                   |         |

The proportion of patients who were elective admissions differs significantly in the two cohorts: patients recruited into the study and eligible patients who were not recruited. However, the proportion of patients in the recruited cohort who were elective or emergency admissions only differed from the intended cohort by 4.3%. Again there is no reason to believe that differences of this magnitude will have led to substantial biases in comparisons of those with and without HAI.

# A10.5 The distribution of study participants and eligible patients who were not recruited by admission specialty

Table A10.7 provides details of the distribution of study participants and eligible patients who either declined participation, or were not recruited due to practical reasons by admission specialty. Table A10.8 shows the admission specialty distribution of study participants and all eligible patients who were not recruited, together with that which would have been present if all eligible patients had been recruited.

|                             |                     |      | Re                     | eason for r       | not recru                       | iting eligit                            | ole patier                   | rts                             |                                   |                 |
|-----------------------------|---------------------|------|------------------------|-------------------|---------------------------------|-----------------------------------------|------------------------------|---------------------------------|-----------------------------------|-----------------|
| Admission<br>Specialty      | Study<br>Participan | t    | Patient (<br>participa | declined<br>ation | Patie<br>recrui<br>to pr<br>rea | ent not<br>ited due<br>actical<br>sons* | Reas<br>nc<br>recrui<br>unkr | on for<br>on-<br>itment<br>nown | All eligi<br>patients<br>recruite | ble<br>not<br>d |
|                             | n                   | %    | n                      | %                 | n                               | %                                       | n.                           | %                               | n                                 | %               |
| Surgery                     | 884                 | 35.8 | 73                     | 37.1              | 468                             | 55.0                                    | 3                            | 33.3                            | 544                               | 51.5            |
| Orthopaedics                | 501                 | 20.3 | 45                     | 22.8              | 121                             | 14.2                                    | 5                            | 55.6                            | 171                               | 16.2            |
| Urology                     | 472                 | 19.1 | 60                     | 30.5              | 186                             | 21.9                                    | 0                            | 0.0                             | 246                               | 23.3            |
| Obstetrics &<br>Gynaecology | 612                 | 24.8 | 19                     | 9.6               | 76                              | 8.9                                     | 1                            | 11.1                            | 96                                | 9.1             |
| TOTAL                       | 2469                | 100  | 197                    | 100.0             | 851                             | 100.0                                   | 9                            | 100.0                           | 1057                              | 100.0           |

 Table A10.7: Distribution of study participants and eligible patients who were not recruited by admission specialty

\*For example, insufficient time

# Table A10.8: Admission specialty distribution of study participants, eligiblepatients who were not recruited, and that which would have beenpresent if all eligible patients were recruited

| Admission energiativ     |           | Eligible patie   | mts                   | Proportion (% | 5) recruited<br>specialty | from each             | P value                               |
|--------------------------|-----------|------------------|-----------------------|---------------|---------------------------|-----------------------|---------------------------------------|
| Putitission specially    | Recruited | Not<br>recruited | All eligible patients | Recruited     | Not<br>recruited          | All eligible patients | t Value                               |
| Surgery                  | 884       | 544              | 1428                  | 35.8          | 51.5                      | 40.5                  | <0.001                                |
| Orthopaedics             | 501       | 171              | 672                   | 20.3          | 16.2                      | 19.1                  | ₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩ |
| Urology                  | 472       | 246              | 718                   | 19.1          | 23.3                      | 20.4                  |                                       |
| Obstetrics & gynaecology | 612       | 96               | 708                   | 24.8          | 9.1                       | 20.1                  |                                       |
| All patients             | 2469      | 1057             | 3526                  | 100.0         | 100.0                     | 100.0                 |                                       |

The results presented in Table A10.8 indicate that there are significant differences in the proportion of patients from each specialty in the recruited and non-recruited patient groups. However, in no specialty was there more than a 4.7% difference between the recruited and intended cohort. There is therefore very little likelihood of these differences causing a substantial bias in the comparison between those who did and did not have HAIs.

# A10.6 Conclusion

While the recruits and non-recruits had almost the same proportion of older patients, there were significantly more female patients and less emergency patients in the recruited cohort of patients, fewer surgical and urology patients and more orthopaedic and obstetric and gynaecology patients. However, the differences between the cohort of recruited patients and the intended cohort were quite small. Since the HAI and non HAI comparisons were stratified by all of these factors it is reasonable to assume that the results obtained from those recruited are generalisable to all eligible patients in the study hospital, despite the significant differences between those recruited and those not recruited.

# The incidence of HAIs by primary operative procedure

### A11.1 Introduction

This appendix presents the results of this analysis that examined the incidence of HAIs and how this varied with primary operative procedure. Tables A11.1 – A11.11 show how the incidence of HAI varied with primary operative procedure classified according to the Office of Population Census and Surveys operation classification system (fourth edition)<sup>231</sup> three-digit code. Where appropriate similar operations have been grouped together and the incidence of HAI calculated.

The incidence was found to vary with operative procedure. However the number of patients undergoing many of the specific procedures were small and as such, as evidenced by the confidence intervals, there is considerable uncertainty about the estimate derived.

The highest incidence of HAI occurred in patients whose primary procedure involved the female genital tract: 13.2% acquired one or more HAIs. Amongst these patients the incidence was highest in patients who had a repair of a prolapse of the vagina (19.3%), and hysterectomy (14.9%).

The second highest incidence rate was observed in patients whose primary procedure involved the arteries and veins: 8.9% of patients acquired one or more infections in hospital. Of these patients 33.3% of patients whose primary procedure involved the aorta and 25.0% of patients whose primary procedure involved the aorta and 25.0% of patients whose primary procedure involved the iliac and/or femoral artery acquired one or more infections that presented during the in-patient period. However, the number of patients in these two subgroups was small.

A similar incidence rate was observed in patients whose primary procedure involved the digestive tract: 8.9% acquired one or more infections that presented during the in-patient stay. The highest incidence rate was observed in patients who had a primary procedure which involved the ileum (57.1%); and patients who had a total or partial excision of the stomach and /or oesophagus (37.5%). Again the number of patients within the various sub-groups of procedures was small and as such strong conclusions cannot be drawn.

The incidence of HAI occurring in patients whose primary operative procedure involved the nervous system, endocrine system, breast, respiratory tract or mouth Table A11.1:

| Operations<br>involving:     |      |                                       | II | ndividual | I procedure | S      |    | Combined | category |      |
|------------------------------|------|---------------------------------------|----|-----------|-------------|--------|----|----------|----------|------|
|                              | Code | Category                              | a  | No<br>HAI | HAI         | IR (%) | E  | No HAI   | HAI      | %    |
|                              | A52  | Therapeutic epidural injection        | 7  | 7         | 0           | 0.0    |    |          |          |      |
| Nervous system               | A68  | Other release of peripheral nerve     | -  | -         | 0           | 0.0    | თ  | 0        | 6        | 0.0  |
|                              | A81  | Other operations on sympathetic nerve | -  | -         | •           | 0.0    |    |          |          |      |
| Thursday and                 | B08  | Excision of thyroid gland             | 20 | 16        | 4           | 20.0   | 23 | 19       | 4        | 17.4 |
|                              | B10  | Operations on thyroglossal tissue     | 1  | -         | 0           | 0.0    |    |          |          |      |
| parathyroid gland            | B14  | Excision of parathyroid gland         | 2  | 2         | 0           | 0.0    |    |          |          |      |
|                              | B27  | Total excision of breast              | 28 | 28        | 0           | 0.0    |    |          |          | ł    |
|                              | B28  | Other excision of breast              | 40 | 40        | 0           | 0.0    |    |          |          |      |
|                              | B30  | Prosthesis for breast                 | 5  | 5         | 0           | 0.0    | 81 | 81       | C        | 0.0  |
| Breast                       | B31  | Other plastic operations on breast    | 1  | -         | 0           | 0.0    | 5  | ;        | ,        |      |
|                              | B32  | Biopsy of breast                      | 4  | 4         | 0           | 0.0    |    |          |          |      |
|                              | B33  | Incision of breast                    | 1  | 1         | 0           | 0.0    |    |          |          |      |
|                              | B34  | Operations on duct of breast          | 2  | 2         | 0           | 0.0    |    |          |          |      |
| Exteriorisation of trachea   | E42  | Exteriorisation of trachea            | -  | 0         | -           | 100.0  |    |          |          | -    |
| Excision of lesion of tongue | F23  | Excision of lesion of tongue          | -  | 0         | -           | 100.0  |    |          | •        |      |
| Excision of salivary<br>dand | F44  | Excision of salivary gland            | 9  | 9         | 0           | 0.0    |    |          |          |      |

Table A11.2: The incidence of HAI occurring in patients whose primary operative procedure involved the digestive tract and other abdominal organs principally related to the digestive tract

|                                   |      |                                                                     |    | Individual | procedures |        |    | Combined | category |      |
|-----------------------------------|------|---------------------------------------------------------------------|----|------------|------------|--------|----|----------|----------|------|
| uperations<br>involving:          | Code | Category                                                            | E  | No<br>HAI  | HAI        | IR (%) | E  | No HAI   | HAI      | %    |
| Excision of                       | G01  | Excision of oesophagus and stomach                                  | 3  | •          | 2          | 66.7   |    | 2        |          |      |
| orecombagnic                      | G27  | Total excision of stomach                                           | 3  | 2          | -          | 33.3   | 80 | 2        | m        | 37.5 |
| and/or stomach                    | G28  | Partial excision of stomach                                         | 2  | 8          | 0          | 0.0    |    |          |          |      |
| Other operations<br>involving the | G15  | Other therapeutic fibreoptic endoscopic<br>operations on oesophagus | -  | -          | 0          | 0.0    |    |          |          | 0    |
| oesophagus,                       | G24  | Antireflux operations                                               | 2  | 2          | 0          | 0.0    | 7  | e        | -        | 14.3 |
| duodenum and                      | G52  | Operations on ulcer of duodenum                                     | 3  | 3          | 0          | 0.0    |    |          |          |      |
| jejunum                           | G59  | Extirpation of lesion of jejunum                                    | -  | 0          | -          | 100.0  |    |          |          |      |
|                                   | G69  | Excision of ileum                                                   | 2  | 0          | 2          | 100.0  |    |          |          |      |
|                                   | G74  | Creation of artificial opening in ileum                             | 1  | 0          | 1          | 100.0  |    |          |          |      |
| F                                 | G75  | Attention to artificial opening into ileum                          | -  | 0          | 1          | 100.0  | 7  | e        | 4        | 57.1 |
| Ileum                             | G76  | Intraabdominal manipulation of ileum                                | -  | -          | 0          | 0.0    | -  | 1999     |          |      |
|                                   | G78  | Other open operations on ileum                                      | -  | -          | 0          | 0.0    |    |          |          |      |
|                                   | G82  | Other operations on ileum                                           | -  | -          | 0          | 0.0    |    |          |          |      |
|                                   | J08  | Therapeutic endoscopic operations on liver<br>using laprascope      | 19 | 18         | -          | 5.3    |    | ~        |          | 4    |
| Liver                             | 601  | Diagnostic endoscopic examination of liver<br>using laprascope      | +  | -          | 0          | 0.0    | 21 |          |          |      |
|                                   | J13  | Diagnostic percutaneous operations on liver                         | +  | 1          | 0          | 0.0    |    |          | _        |      |

| The incidence of HAI occurring in patients whose primary operative procedure involved the | digestive tract and other abdominal organs principally related to the digestive tract |
|-------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|
| Table A11.2 continued:                                                                    |                                                                                       |

|                          | 10.800 |                                                                                    | -  | ndividual p | procedures |        |    | Combined ci | stegory |      |
|--------------------------|--------|------------------------------------------------------------------------------------|----|-------------|------------|--------|----|-------------|---------|------|
| Operations<br>involving: | Code   | Category                                                                           | E  | No<br>HAI   | HAI        | IR (%) | c  | No HAI      | HAI     | %    |
|                          | J18    | Excision of gall bladder                                                           | 53 | 50          | e          | 5.7    | 22 | 54          | P       | 73   |
| Gall bladder and         | 119    | Connection of gall bladder                                                         | -  | -           | 0          | 0.0    | ß  | 5           | r       | 2    |
| bile duct                | J37    | Other open operations on bile duct                                                 | 1  | 0           | -          | 100.0  |    |             |         |      |
| Pancreas and             | J65    | Other open operations on pancreas                                                  | 1  | -           | 0          | 0.0    | c  | ç           | c       | 000  |
| spleen                   | J69    | Total excision of spleen                                                           | 1  | -           | 0          | 0.0    | 7  | v           | >       | 0.0  |
|                          | H01    | Emergency excision of appendix                                                     | 36 | 35          | 1          | 2.8    |    | 4U          |         |      |
| Annendix                 | H02    | Other excision of appendix                                                         | 3  | 8           | 0          | 0.0    | 41 | }           |         | 2.4  |
| unandde .                | HOA    | Other operations on appendix                                                       | 2  | 2           | 0          | 0.0    |    |             |         |      |
|                          | 90H    | Extended excision of right hemi colon                                              | 1  | 1           | 0          | 0.0    |    |             |         |      |
|                          | H07    | Other excision of right hemi colon                                                 | 7  | 7           | 0          | 0.0    |    |             |         |      |
|                          | H08    | Excision of transverse colon                                                       | •  | 1           | 0          | 0.0    |    |             |         |      |
|                          | 60H    | Excision of left hemicolon                                                         | -  | 1           | 0          | 0.0    | 38 | 25          | ~       | 10.7 |
| Colon/caecum             | H10    | Excision of sigmoid colon                                                          | œ  | 2           | 1          | 12.5   | 24 | 3           | ,       |      |
|                          | H11    | Other excision of colon                                                            | 3  | 2           | 1          | 33.3   |    |             |         |      |
|                          | H14    | Exteriorisation of caecum                                                          | -  | 1           | 0          | 0.0    |    |             |         |      |
|                          | H15    | Other exteriorisation of colon                                                     | 9  | 5           | •          | 16.7   |    |             |         |      |
|                          | H20    | Endoscopic extirpation of lesion of colon                                          | •  | -           | 0          | 0.0    |    |             |         |      |
|                          | H21    | Other therapeutic endoscopic operations on<br>colon                                | 1  | -           | 0          | 0.0    |    |             |         |      |
|                          | H22    | Diagnostic endoscopic examination of colon                                         | 3  | 3           | 0          | 0.0    |    |             |         |      |
| Endoscopic               | H25    | Diagnostic endoscopic examination of lower<br>bowel using fibreoptic sigmoidoscope | 35 | 33          | 2          | 5.7    | 99 | 63          | e       | 4.5  |
|                          | H26    | Endoscopic extirpation of lesion of sigmoid<br>colon using rigid sigmoidoscopy     | 3  | 2           | -          | 33.3   |    |             |         |      |
|                          | H28    | Diagnostic endoscopic examination of sigmoid                                       | 23 | 23          | 0          | 0.0    |    |             |         |      |

|                          | -    |                                                   |    | Individual r | procedures |        |    | Combined c | ategory |       |
|--------------------------|------|---------------------------------------------------|----|--------------|------------|--------|----|------------|---------|-------|
| Operations<br>involving: | Code | Category                                          | E  | No<br>HAI    | HAI        | IR (%) | F  | No HAI     | HAI     | %     |
|                          | H33  | Excision of rectum                                | 16 | 12           | 4          | 25.0   |    |            |         |       |
|                          | H36  | Other abdominal operations for prolapse of rectum | 2  | 2            | 0          | 0.0    |    |            |         |       |
|                          | H41  | Other operations on rectum through anus           | -  | 1            | 0          | 0.0    |    |            |         |       |
| Rectum/anus              | H42  | Perineal operations for prolapse of rectum        | 1  | 1            | 0          | 0.0    |    |            | 64      | 1.0   |
|                          | H44  | Manipulation of rectum                            | -  | 1            | 0          | 0.0    |    |            |         |       |
|                          | H48  | Excision of lesion of anus                        | -  | -            | 0          | 0.0    |    |            |         |       |
|                          | H49  | Destruction of lesion of anus                     | -  | 1            | 0          | 0.0    |    |            |         |       |
|                          | H51  | Excision of haemorrhoid                           | 17 | 17           | 0          | 0.0    | 87 | 81         | y       | 69    |
|                          | H52  | Destruction of haemornhoid                        | 14 | 14           | 0          | 0.0    | 5  | 5          | ,       | 2     |
|                          | H54  | Dilation of anal sphincter                        | -  | -            | 0          | 0.0    |    |            |         | 25.25 |
|                          | H55  | Other operations on perianal region               | 5  | 5            | 0          | 0.0    |    |            |         |       |
|                          | H56  | Other operations on anus                          | 11 | 10           | ٢          | 9.1    |    |            |         |       |
|                          | H58  | Drainage through perianal region                  | 4  | 4            | 0          | 0.0    |    |            |         |       |
|                          | H59  | Excision of pilonidal sinus                       | 6  | 80           | 1          | 11.1   |    |            |         | 1000  |
|                          | HEN  | Other onerations on nilonidal sints               | ~  | ~            | 0          | 0.0    |    |            |         |       |

ahrad tha . 9 ÷ ţ ---. . -.

| 5        |
|----------|
| 5        |
| -        |
| e        |
| ~        |
|          |
| Ē        |
| ~        |
|          |
| S        |
| <b>O</b> |
| - E      |
| 5        |
| 2        |
| Т        |
| 3        |
| -        |
| Q        |
| θ        |
| >        |
| -        |
| 2        |
| 2        |
| -        |
|          |
| e        |
| -        |
| 3        |
| D        |
| đ        |
| ü        |
| õ        |
| Ľ        |
| ō        |
| and the  |
| Φ        |
| >        |
| B        |
| 1        |
| 2        |
| 0        |
| č        |
| -        |
| 0        |
| -        |
| C        |
|          |
| Ē        |
| =        |
| -        |
| ā        |
| -        |
| e        |
| (D)      |
| õ        |
| ×        |
| -        |
| -        |
|          |
| SN .     |
| =        |
|          |
| Ð        |
| ÷        |
| 3        |
| Q.       |
| _        |
| 5        |
|          |
| 0        |
| ē        |
|          |
| -        |
|          |
| -        |
| S        |
| Q        |
| 0        |
| _        |
| •        |
| -        |
| -        |
| -        |
| 0        |
| -        |
| 9        |
| S        |
| E        |
| 0        |
| T        |
|          |
| C        |
| Ē        |
|          |
| (D)      |
| č        |
| -        |
| -        |
| 22       |
| 3        |
| ÷.       |
| -        |
| -        |
|          |
| (D)      |
| -        |
| -        |
| -        |
| ap       |
| Tab      |

| Operations                                           |             |                                                            |    | Individual | procedures |        |    | Combined c | ategory | 1     |
|------------------------------------------------------|-------------|------------------------------------------------------------|----|------------|------------|--------|----|------------|---------|-------|
| involving:                                           | Code        | Category                                                   | u  | No<br>HAI  | HAI        | IR (%) | E  | No HAI     | HAI     | %     |
| Aorta                                                | L08         | Transluminal operations on abnormality of great vessel     | -  | -          | 0          | 0.0    | 12 |            | - 24    |       |
|                                                      | L18         | Emergency replacement of aneurysmal segment of aorta       | ÷  | -          | 0          | 0.0    | 9  | 4          | 8       | 33.3  |
|                                                      | L19         | Other replacement of aneurysmal segment of aorta           | 2  | -          | -          | 50.0   |    |            |         |       |
| and the state of the second                          | 121         | Other bypass of segment of aorta                           | 1  | -          | 0          | 0.0    |    |            |         | 10.00 |
| lliac and femoral<br>artery                          | L45         | Reconstruction of other visceral branch of abdominal aorta | +  | •          | -          | 100.0  |    |            |         |       |
|                                                      | L51         | Other bypass of iliac artery                               | -  | 0          | -          | 100.0  | 80 | 9          | 2       | 25.0  |
|                                                      | L62         | Other open operations of femoral artery                    | 2  | -          | -          | 50.0   |    |            |         |       |
|                                                      | L63         | Transluminal operations on femoral artery                  | 5  | 5          | 0          | 0.0    |    |            |         |       |
| Other<br>arteriovenous<br>operations                 | L75         | Other arteriovenous operations                             | -  | 0          | -          | 100.0  | -  | -          | 0       | 100.0 |
| Varicose veins                                       | L85         | Ligation of varicose veins                                 | 25 | 25         | 0          | 0.0    |    |            |         |       |
|                                                      | L87         | Other operations on varicose vein of leg                   | 15 | 15         | 0          | 0.0    | 40 | 40         | >       | 0.0   |
| Therapeutic<br>transluminal<br>operations on<br>vein | <b>7</b> 61 | Therapeutic transluminal operations on vein                | ÷  |            | 0          | 0.0    | -  | -          | 0       | 0     |

Table A11.4: The incidence of HAI occurring in patients whose primary operative procedure involved the urinary system

|                                        |      |                                                            |      | ndividual p | procedures |        |    | Combined c | ategory |      |
|----------------------------------------|------|------------------------------------------------------------|------|-------------|------------|--------|----|------------|---------|------|
| Operations<br>involving:               | Code | Category                                                   | E    | 9 HAI       | HAI        | IR (%) | E  | No HAI     | HAI     | %    |
| Vidam                                  | MO2  | Total excision of kidnev                                   | 6.0  | 4           | 2          | 33.3   |    | 1410 BANK  |         |      |
| - Alimira                              | MO3  | Partial excision of kidnev                                 | 2.0  | 2           | 0          | 0.0    |    |            |         | 0.00 |
| procedures                             | M05  | Open repair of kidney                                      | 2.0  | 2           | 0          | 0.0    | 11 | თ          | N       | 7.97 |
| involving an<br>incision               | MOG  | Incision of kidney                                         | 1.0  | -           | 0          | 0.0    | 10 | P          |         |      |
| Kidnev - endoscopic                    | 60W  | Therapeutic endoscopic operations of calculus<br>of kidney | 5.0  | 4           | -          | 20.0   |    |            |         |      |
| operations and/or<br>percutaneous      | M10  | Other therapeutic endoscopic operations of kidney          | 1.0  | -           | 0          | 0.0    | 6  | ø          | -       | 11.1 |
| puncture of kidney                     | M11  | Diagnostic endoscopic examination of kidney                | 1.0  | 1           | 0          | 0.0    |    |            |         |      |
|                                        | M13  | Percutaneous puncture of kidney                            | 2.0  | 2           | 0          | 0.0    |    |            |         |      |
| []reter -                              | M20  | Replantation of ureter                                     | 2.0  | 2           | 0          | 0.0    |    |            |         |      |
| procedures<br>involving an<br>incision | M23  | Incision of ureter                                         | 2.0  | 2           | 0          | 0.0    | 4  | 4          | 0       | 0.0  |
|                                        | M37  | Therapeutic ureterscopic operations on ureter              | 6.0  | 9           | 0          | 0.0    |    |            |         |      |
| Ureter -                               | M38  | Other endoscopic removal of calculus from<br>ureter        | 1.0  | -           | 0          | 0.0    | 1  | 5          |         | 10   |
| endoscopic                             | M39  | Other therapeutic endoscopic operations on<br>ureter       | 7.0  | 9           | -          | 14.3   | 77 | ę          | -       | 1.0  |
| provouuce                              | M320 | Diagnostic endoscopic examination of ureter                | 11.0 | 11          | 0          | 0.0    |    |            |         |      |
|                                        | M320 | Operations of ureteric orifice                             | 2.0  | 2           | 0          | 0.0    |    |            | _       |      |

Table A11.4 continued: Incidence of HAI occurring in patients whose primary operative procedure involved the urinary system

|                            |            |                                                                               | 5     | dividual  | procedures |        |     | Combined o | ategory |      |
|----------------------------|------------|-------------------------------------------------------------------------------|-------|-----------|------------|--------|-----|------------|---------|------|
| Operations                 | Code       | Category                                                                      | E     | No<br>HAI | HAI        | IR (%) | c   | No HAI     | HAI     | %    |
| :BUINIOAU                  | M2A        | Total excision of bladder                                                     | 2.0   | 2         | 0          | 0.0    |     |            |         |      |
|                            | SCM        | Dartial avrieion of hladder                                                   | 1.0   | 1         | 0          | 0.0    |     |            |         |      |
| biadoer -                  | COM        | Enternament of hladdar                                                        | 1.0   | 0         | -          | 100.0  | 80  | 1          | -       | C.21 |
| procedures                 | OCM DCM    | Chan drainant of bladder                                                      | 3.0   | 3         | 0          | 0.0    |     |            |         |      |
| INNOVA IN BURNING          | M30        | Other onerations on contents of bladder                                       | 1.0   | 1         | 0          | 0.0    |     |            |         |      |
|                            | M42        | Endoscopic extirpation of lesion of bladder                                   | 60.09 | 25        | 3          | 5.0    |     | 10         |         |      |
| Bladder -                  | M43        | Endoscopic operations to increase capacity of<br>bladder                      | 1.0   | -         | 0          | 0.0    | 167 | 140        | ~       | 5.1  |
| endoscopic                 | M44        | Other therapeutic endoscopic operations on<br>bladder                         | 2.0   | 3         | 0          | 0.0    | 201 | <u>P</u>   | ,       | ;    |
|                            | M45        | Diagnostic endoscopic examination of bladder                                  | 92.0  | 87        | 2          | 5.4    |     |            |         |      |
|                            | M49        | Other operations on bladder                                                   | 2.0   | 2         | D          | 0.0    |     |            |         |      |
| Outlet of female           | M51        | Combined abdominal and vaginal operations to support outlet of female bladder | 1.0   | -         | 0          | 0.0    | S   | 4          | -       | 20.0 |
| bladder                    | M52        | Abdominal operations to support our let of<br>female hladder                  | 4.0   | 3         | -          | 25.0   |     |            |         |      |
| Drastate and nitlet        | M61        | Onen excision of prostate                                                     | 4.0   | 3         | -          | 25.0   | •   | u          | •       | 16.7 |
| of male bladder -          | M64        | Other open operations on outlet of male                                       | 2.0   | 3         | 0          | 0.0    | ٥   | 0          | -       |      |
| Uperi processio            | M65        | Endoscopic resection of outlet of male bladder                                | 82.0  | 76        | 9          | 7.3    |     |            |         |      |
| operations on              | 901        | Other therapeutic endoscopic operations on                                    | 9.0   | 6         | 0          | 0.0    | 111 | 104        | 7       |      |
| prostate anuror<br>hladder | <b>B</b> M | Other therapeutic endoscopic operations on                                    |       |           |            |        | -   |            |         |      |
|                            | M67        | prostate                                                                      | 4.0   | 4         | 0          | 0.0    | _   |            |         | 6.9  |
|                            | M70        | Other operations on outlet of male bladder                                    | 16.0  | 15        | -          | 6.3    |     |            |         | 2.2  |
| Onerations involving       | M73        | Repair of urethra                                                             | 1.0   | -         |            | 0.0    | -   |            |         |      |
| urethra                    | M76        | Therapeutic endoscopic operations on urethra                                  | 7.0   |           |            | 0.0    | 16  | 15         | -       | 6.3  |
|                            | M77        | Diagnostic endoscopic examination of urethra                                  | 1.0   |           |            | 100.0  | -   |            |         |      |
|                            | M79        | Other operations on urethra                                                   | 7.0   |           |            | 0.0    |     |            |         |      |

The incidence of HAI occurring in patients whose primary operative procedure involved the male genital organs Table A11.5:

| Operations     |      |                                          |             | Individual p | procedures |        |    | Combined c                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ategory |    |
|----------------|------|------------------------------------------|-------------|--------------|------------|--------|----|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------|----|
| involving:     | Code | Category                                 | E           | No<br>HAI    | HAI        | IR (%) | =  | No HAI                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | IAI     | %  |
| Scrotum testes | N03  | Other operations of scrotum              | 3           | 3            | 0          | 0.0    |    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |         |    |
| and notice     | 800  | Bilateral excision of testes             | -           | 1            | 0          | 0.0    |    | and the second se |         |    |
| and penis      | 90N  | Other excision of testes                 | 6           | 6            | 0          | 0.0    |    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |         | 26 |
|                | N08  | Bilateral placement of testis in scrotum | 1           | 1            | 0          | 0.0    |    | and the decision of the decision                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |         |    |
|                | N11  | Operations on hydrocele sac              | -           | 1            | 0          | 0.0    | 26 | 25                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 0       | 00 |
|                | N13  | Other operations on testes               | 2           | 2            | 0          | 0.0    | 2  | 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ,       |    |
|                | N15  | Operations on epididymis                 | 2           | 2            | 0          | 0.0    |    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |         |    |
|                | N28  | Plastic operations on penis              | 1           | 1            | 0          | 0.0    |    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |         |    |
|                | N29  | Prosthesis of penis                      | 1 - 17 S.S. | 1            | 0          | 0.0    |    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |         |    |
|                | N30  | Operations on prepuce                    | 4           | 4            | 0          | 0.0    |    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |         |    |

The incidence of HAI occurring in patients whose primary operative procedure involved the female genital organs Table A11.6:

| Operations                         |      |                                                                |       | Individual | procedures |        |      | Combined | category |      |
|------------------------------------|------|----------------------------------------------------------------|-------|------------|------------|--------|------|----------|----------|------|
| involving:                         | Code | Category                                                       | E     | No HAI     | HAI        | IR (%) | L    | No HAI   | HAI      | %    |
|                                    | P05  | Excision of vulva                                              | -     | -          | 0          | 0.0    |      |          |          |      |
| Vulva                              | 60d  | Other operations on vulva                                      | T     | -          | •          | 0.0    | 2    | 2        | 0        | 0.0  |
| Extirpation of lesion<br>of vagina | P20  | Extirpation of lesion of vagina                                | -     | -          | 0          | 0.0    | -    | 0        | -        | 0.0  |
| Vagina - repair of                 | P22  | Repair of prolapse of vagina and amputation<br>of cervix uteri | 2     | 2          | 0          | 0.0    | 57   | 46       | #        | 19.3 |
| prolapse                           | P23  | Other repair of prolapse of vagina                             | 55    | 44         | 11         | 20.0   |      |          |          |      |
| Hysterectomy                       | 207  | Abdominal excision of uterus                                   | 105.0 | 89         | 16         | 15.2   | 17.4 | 148      | 26       | 14.9 |
|                                    | 008  | Vaginal excision of uterus                                     | 69.0  | 59         | 10         | 14.5   |      |          |          |      |
|                                    | 002  | Destruction of lesion of cervix uteri                          | -     | 0          | 1          | 0.0    |      |          |          |      |
|                                    | 003  | Biopsy of cervix                                               | 0     | 3.0        | 9          | 0.0    |      |          |          |      |
|                                    | 005  | Other operations on cervix uteri                               | 0     | 1.0        | -          | 0.0    |      |          |          |      |
|                                    | 000  | Other open operations on uterus                                | 0     | 2.0        | 2          | 0.0    |      |          | 10       | 1.12 |
|                                    | 010  | Curettage of uterus                                            | 0     | 6.0        | 9          | 0.0    |      |          |          |      |
|                                    | 012  | Removal of intrauterine contraceptive device                   | 0     | 1.0        | 1          | 0.0    |      |          |          |      |
| Other operations                   | 022  | Bilateral excision of adnexa of uterus                         | 0     | 6.0        | 9          | 0.0    |      |          |          |      |
| involving uterus,                  | 023  | Unitateral excision of adnexa of uterus                        | 0     | 11.0       | 11         | 0.0    | 53   | 20       | ო        | 5.7  |
| tallopian tubes,                   | 025  | Partial excision of fallopian tube                             | 0     | 1.0        | 1          | 0.0    |      |          |          |      |
| cervix and ovaries                 | 029  | Open reversal of female sterilisation                          | 0     | 3.0        | 3          | 0.0    |      |          |          |      |
|                                    | 030  | Salpingostomy                                                  | -     | 2.0        | 1          | 50.0   |      |          |          |      |
|                                    | 034  | Other open operations on fallopian tubes                       | 0     | 1.0        | -          | 0.0    |      |          |          |      |
|                                    | 041  | Other operations on fallopian tubes                            | 0     | 2.0        | 2          | 0.0    |      |          |          |      |
|                                    | Q43  | Partial excision of ovary                                      | -     | 5.0        | 4          | 20.0   |      |          |          |      |
|                                    | 047  | Other open operations on ovary                                 | -     | 8.0        | 2          | 12.5   |      |          |          |      |

Table A11.6 continued: The incidence of HAI occurring in patients whose primary operative procedure involved the female genital organs

| Operations         |      |                                                | -     | ndividual p | procedures |        |     | Combined c | ategory  |      |
|--------------------|------|------------------------------------------------|-------|-------------|------------|--------|-----|------------|----------|------|
| involving:         | Code | Category                                       | c     | NA<br>HAI   | HAI        | IR (%) | E   | No HAI     | HAI      | %    |
| Diagnostic         | 018  | Diagnostic endoscopic examination of uterus    | 11.0  | 11          | 0          | 0.0    |     |            |          |      |
| endoscopic         | 000  | Diagnostic endoscopic examination of fallopian | 00    | c           | c          | 00     | 4   | 14         | •        | 0.0  |
| examinauoris       | 200  | Abdominal excision of Itlerie                  | 105.0 | 200         | 16         | 15.2   | 174 | 148        | 26       |      |
| Hvsterectomv       | 008  | Vacinal excision of uterus                     | 69.0  | 59          | 10         | 14.5   |     |            |          | D.41 |
| Other onerations   | 002  | Destruction of lesion of cervix uteri          | 1     | -           | 0          | 0.0    |     |            |          |      |
| involving uterus.  | 003  | Biopsv of cervix                               | 3.0   | 3           | 0          | 0.0    |     |            |          |      |
| fallopian tubes.   | 005  | Other operations on cervix uteri               | 1.0   | -           | 0          | 0.0    |     |            |          |      |
| cervix and ovaries | 600  | Other open operations on uterus                | 2.0   | 2           | 0          | 0.0    |     |            |          |      |
|                    | 010  | Curettage of uterus                            | 6.0   | 9           | 0          | 0.0    |     |            |          |      |
|                    | 012  | Removal of intrauterine contraceptive device   | 1.0   | -           | 0          | 0.0    |     |            |          |      |
|                    | 022  | Bilateral excision of adnexa of uterus         | 6.0   | 9           | 0          | 0.0    |     |            |          |      |
|                    | 023  | Unilateral excision of adnexa of uterus        | 11.0  | 11          | 0          | 0.0    |     |            |          |      |
|                    | 025  | Partial excision of fallopian tube             | 1.0   | -           | 0          | 0.0    | 53  | 20         | <b>с</b> | 5.7  |
|                    | 029  | Open reversal of female sterilisation          | 3.0   | 3           | 0          | 0.0    |     |            |          |      |
|                    | 030  | Salpinoostomv                                  | 2.0   | -           | •          | 50.0   |     |            |          |      |
|                    | 034  | Other open operations on fallopian tubes       | 1.0   | -           | 0          | 0.0    |     |            |          |      |
|                    | 041  | Other operations on fallopian tubes            | 2.0   | 2           | 0          | 0.0    |     |            |          |      |
|                    | 043  | Partial excision of ovarv                      | 5.0   | 4           | 1          | 20.0   |     |            | Sector - |      |
|                    | 047  | Other open operations on ovarv                 | 8.0   | 7           | -          | 12.5   |     |            |          |      |

Table A11.7: The incidence of HAIs occurring in patients whose primary operative procedure code involved the female genital tract associated with pregnancy and birth

| Operations         |       |                                                                        |     | ndividual | procedures |        |     | Combined c | ategory |     |
|--------------------|-------|------------------------------------------------------------------------|-----|-----------|------------|--------|-----|------------|---------|-----|
| involving:         | Code  | Category                                                               | E   | No<br>HAI | HAI        | IR (%) | E   | No HAI     | HAI     | %   |
| Caesarean sections | R17.2 | Elective caesarean defivery                                            | 91  | 83        | 80         | 8.8    | 242 | 107        | 00      | 44  |
|                    | R18.2 | Other caesarean delivery                                               | 121 | 109       | 12         | 9.9    | 212 | 761        | 2       |     |
|                    | R28.2 | Instrumental removal o products of conception<br>from delivered uterus | •   | 0         | -          | 0.0    | 1   | 0          | Ŧ       | 0.0 |
| Other operations   | R12.1 | Operations on gravid uterus                                            | -   | 0         | -          | 0.0    |     |            |         |     |

# The incidence of HAI occurring in patients whose primary operative procedure involved the skin Table A11.8:

|                          |      |                                            |   | Individual | procedures |        |    | Combined | category | -  |
|--------------------------|------|--------------------------------------------|---|------------|------------|--------|----|----------|----------|----|
| Operations<br>involving: | Code | Category                                   | E | No<br>HAI  | HAI        | IR (%) | c  | No HAI   | HAI      | %  |
|                          | S04  | Other excision of skin                     | - | -          | 0          | 0      |    |          |          |    |
|                          | 806  | Other excision of lesion of skin           | 5 | 4          | -          | 20     |    |          |          | _  |
|                          | S15  | Other biopsy of skin                       | e | e          | 0          | 0      |    |          |          |    |
|                          | S18  | Distant flap of skin and fascia            | - | -          | 0          | 0      | 24 | 10       | ~        | 50 |
| SKIN                     | S42  | Suture of skin of other site               | 2 | 2          | 0          | 0      |    | 2        | •        |    |
|                          | S45  | Removal of other substance of skin         | - | -          | 0          | 0      |    |          |          |    |
|                          | S47  | Opening of skin                            | 5 | 4          | -          | 20     |    |          |          |    |
|                          | S57  | Exploration of other skin of other site9.5 | 3 | 3          | 0          | 0      |    |          |          | _  |

The incidence of HAI occurring in patients whose primary operative procedure code involved soft tissue Table A11.9:

| Operations                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |             |                                                    | ų  | dividual p | rocedures |        |     | Combined c | ategory |      |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------|----------------------------------------------------|----|------------|-----------|--------|-----|------------|---------|------|
| involving:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Code        | Category                                           | E  | N IAI      | HAI       | IR (%) | E   | No HAI     | HAI     | %    |
| Pleura                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | T12         | Puncture of pleura                                 | 5  | 0          | 5         | 0.0    | 5   | 0          | 5       | 0    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | T19         | Simple excision of inguinal hernia                 | -  | -          | 0         | 0.0    |     |            |         |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | T20         | Primary repair of inguinal hernia                  | 8  | 95         | 1         | 1.0    |     |            |         |      |
| Inguinal, femoral                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | T21         | Repair of recurrent inguinal hernia                | 9  | 9          | 0         | 0.0    |     |            |         |      |
| and umbilical                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 122         | Primary repair of femoral hernia                   | 2  | -          | -         | 50.0   | 147 | 139        | 80      | 5.5  |
| hernia                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | T24         | Repair of umbilical hemia                          | 13 | 13         | 0         | 0.0    |     |            |         |      |
| and the second se | 125         | Primary repair of incisional hernia                | 7  | 2          | 0         | 0.0    |     |            |         |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 127         | Repair of other hernia of abdominal wall           | 5  | 5          | 0         | 0.0    |     |            |         |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | T30         | Opening of abdomen                                 | 17 | 11         | 9         | 35.3   |     |            |         |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | T34         | Open drainage of peritoneum                        | -  | 1          | 0         | 0.0    |     |            |         |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | T36         | Operations on omentum                              | -  | 0          | 1         | 100.0  |     |            |         |      |
| Abdomen                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | T41         | Other open operations on peritoneum                | 2  | 1          | 1         | 50.0   |     | ;          | 1       | i    |
| peritoneum,                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | T43         | Diagnostic endoscopic examination of<br>peritoneum | 7  | 7          | 0         | 0.0    | 33  | 54         | 55      | 21.3 |
| omentum.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | T52         | Excision of fascia                                 | 2  | 1          | 1         | 50.0   |     |            |         |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | T55         | Release of fascia                                  | -  | -          | 0         | 0.0    |     |            |         |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | T62         | Operations on bursa                                | 1  | -          | 0         | 0.0    |     |            |         |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 179         | Repair of muscle                                   | -  | -          | 0         | 0.0    |     |            |         |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | T85         | Block dissection of lymph nodes                    | 4  | 4          | 0         | 0.0    |     |            |         |      |
| Other onerstione                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | T87         | excision of biopsy of lymph node                   | 3  | 3          | 0         | 0.0    | ç   |            | 4       |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | T94         | Operations on bronchial cleft                      | -  | 1          | 0         | 0.0    | 2   | >          | 2       | 0.0  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | <b>T</b> 96 | Other operations on soft tissue                    | 2  | 2          | 0         | 0.0    |     | -          | -       | 1.2  |

 Table A11.10:
 The incidence of HAI occurring in patients whose primary operative procedure involved bones and joints

|                            |      |                                                                           |    | ndividual | procedures |        |    | Combined | category |     |
|----------------------------|------|---------------------------------------------------------------------------|----|-----------|------------|--------|----|----------|----------|-----|
| Operations involving:      | Code | Category                                                                  | c  | No<br>HAI | HAI        | IR (%) | E  | No HAI   | HAI      | %   |
|                            | V25  | Primary decompression operations on<br>lumbar spine                       | 3  | 3         | 0          | 0      | 15 | 0        | 15       | 0   |
| Lumbar spine               | V33  | Primary excision of lumbar intervertebral disc                            | 12 | 12        | 0          | 0      | 2  |          | 2        |     |
|                            | WG   | Total excision of bone                                                    | 3  | 3         | 0          | 0.0    |    |          |          |     |
|                            | W8   | Other excision of bone                                                    | 4  | 4         | 0          | 0.0    |    |          |          |     |
|                            | 6M   | Extirpation of lesion of bone                                             | 2  | 2         | 0          | 0.0    |    |          |          |     |
|                            | W10  | Open surgical fracture of bone                                            | 1  | 0         | 1          | 100.0  |    |          |          |     |
| Excision/division/iracture | W11  | Other surgical fracture of bone                                           | -  | 1         | 0          | 0.0    | 50 | ~        | 27       | 69  |
| of bone                    | W12  | Angulation periarticular division of bone                                 | 3  | 2         | -          | 33.3   | 1  |          | 1        |     |
|                            | W14  | Diaphyseal division of bone                                               | -  | -         | 0          | 0.0    |    |          |          |     |
|                            | W15  | Division of bone of foot                                                  | 13 | 13        | 0          | 0.0    |    |          |          |     |
|                            | W16  | Other division of bone                                                    | -  | -         | 0          | 0.0    |    |          |          |     |
|                            | W19  | Primary open reduction of fracture of bone<br>and intramedullary fixation | 22 | 20        | 2          | 9.1    |    |          |          |     |
|                            | W20  | Primary open reduction of fracture of bone<br>and extramedullary fixation | 26 | 25        | -          | 3.8    |    |          |          |     |
|                            | W21  | Primary open reduction of intraarticular<br>fracture of bone              | 9  | 9         | 0          | 0.0    |    |          |          |     |
| Reduction of fracture      | W22  | Other primary open reduction of fracture of bone                          | 2  | 2         | 0          | 0.0    | 87 | 82       | 2        | 5.7 |
|                            | W24  | Closed reduction of fracture of bone and<br>internal fixation             | 17 | 17        | 0          | 0.0    |    |          |          |     |
|                            | W26  | Other closed reduction of fracture of bone                                | 3  | 3         | 0          | 0.0    |    |          |          |     |
|                            | W28  | Other internal fixation of bone                                           | 7  | 9         | -          | 14.3   |    |          |          |     |
|                            | W32  | Other graft of bone                                                       | 1  | -         | 0          | 0.0    |    |          |          |     |
|                            | W35  | Therapeutic puncture of bone                                              | 3  | 2         | -          | 33.3   |    | _        | _        |     |

| Oneratione                              |      |                                                                | E  | dividual p | rocedures |        |     | Combined c | ategory |      |
|-----------------------------------------|------|----------------------------------------------------------------|----|------------|-----------|--------|-----|------------|---------|------|
| involving:                              | Code | Category                                                       | c  | No<br>HAI  | HAI       | IR (%) | c   | No HAI     | HAI     | %    |
| Hip replacement                         | W37  | Total prosthetic replacement of hip joint using -<br>cement    | 85 | 7          | 92        | 7.6    |     |            |         |      |
|                                         | W38  | Total prosthetic replacement of hip joint not<br>using cement  | 10 | 0          | 10        | 0.0    | 103 | 96         | 7       | 6.8  |
|                                         | W39  | Other total prosthetic replacement of hip joint                | 74 | 69         | 5         | 6.8    |     |            |         |      |
|                                         | W40  | Total; prostitetic replacement of knee joint using cement      | 4  | 4          | 0         | 0.0    |     |            |         |      |
|                                         | W41  | Total prosthetic replacement of knee joint not<br>using cement | 4  | 3          | 1         | 25.0   | 82  | 26         | Ø       | 7.3  |
| Knee replacement                        | W42  | Other prosthetic replacement of knee joint                     |    |            |           |        |     |            |         |      |
| Replacement of other ioint              | W44  | Total prosthetic replacement other joint not<br>using cement   | 2  | -          | ٢         | 50.0   | e   | ~          |         | 33.3 |
| L                                       | W45  | Other total prosthetic replacement of other initial            | -  | -          | 0         | 0.0    | >   |            |         |      |
| Replacement of head<br>of femurchuments | W46  | Prosthetic replacement of head of femur using<br>cement        | 2  | 0          | 2         | 100.0  |     |            | 19-1    |      |
|                                         | W47  | Prosthetic replacement of head of femur not<br>using cement    | 6  | 6          | 0         | 0.0    |     |            |         |      |
|                                         | W49  | Prosthetic replacement of head of humerus<br>using cement      | 2  | 2          | 0         | 0.0    | 19  | 17         | 2       | 10.5 |
|                                         | W50  | Prosthetic replacement of head of humerus<br>not using cement  | 4  | 4          | 0         | 0.0    | 2   | :          |         |      |
|                                         | W51  | Other prosthetic replacement of head of<br>humens              | -  | -          | 0         | 0.0    |     |            |         |      |
|                                         | CAIN | Prosthetic replacement of articulation of other                | -  |            | 0         | 0.0    |     |            |         |      |
|                                         | W55  | Prosthetic interposition reconstruction of joint               |    | -          | 0         | 0.0    |     |            |         |      |
|                                         | W57  | Other interposition reconstruction of joint                    | 8  | 80         | 0         | 0.0    |     |            | :       |      |
|                                         | W59  | Fusion of joint of toe                                         | 1  | -          | 0         | 0.0    | 11  | 0          | 11      | 0    |
| Reconstruction/fusion                   | WED  | Fusion of other joint and extraarticular bone                  | -  | -          | 0         | 0.0    |     |            |         |      |
| Reduction of                            | -    | Primary open reduction of traumatic                            | -  | 0          | -         | 0.0    |     |            | c       | c    |
| traumatic dislocation                   | COW  | discolution of traumatic                                       |    | 0          |           | 0.0    | ø   | 5          | Ø       | 5    |

The incidence of HAI occurring in patients whose primary operative procedure involved

 Table A11.10 continued: The incidence of HAI occurring in patients whose primary operative procedure code involved bones and joints

| Operations                         |      |                                                              |    | Individua | procedures |       |    | Combined | category |     |
|------------------------------------|------|--------------------------------------------------------------|----|-----------|------------|-------|----|----------|----------|-----|
| involving:                         | Code | Category                                                     | E  | No<br>HAI | HAI        | IR(%) | E  | No HAI   | HAI      | %   |
| Open operations on                 | W70  | Open operations on semilunar cartilage                       | -  | -         | 0          | 0.0   |    |          |          |     |
| ligaments of joints                | W72  | Prosthetic replacement of ligament                           | 5  | 5         | 0          | 0.0   | 6  | 9        | c        | 007 |
|                                    | V77  | Stabalising operations of joint                              | 3  | e         | 0          | 18.2  | NZ | <u>e</u> | N        | 2.0 |
|                                    | W81  | Other open operations on joint                               | 11 | 6         | 2          | 0.0   |    |          |          |     |
| Endoscopic<br>operations on joints | W82  | Therapeutic endoscopic operations on semilunar cartilage     | 3  | 8         | 0          | 0.0   |    |          |          |     |
|                                    | W84  | Therapeutic endoscopic operations on other joint structure   | -  | -         | 0          | 0.0   |    |          |          |     |
|                                    | W85  | Therapeutic endoscopic operations on cavity<br>of knee joint | 2  | 2         | 0          | 0.0   | =  | 0        | 11       | 0.0 |
|                                    | W87  | Diagnostic endoscopic examination of knee joint              | 3  | ო         | 0          | 0.0   |    |          |          |     |
|                                    | W88  | Diagnostic endoscopic examination of other joint             | -  | -         | 0          | 0.0   |    |          |          |     |
|                                    | 06M  | Puncture of joint                                            | -  | -         | 0          |       |    | _        |          |     |
| Manipulation of joint              | W91  | Other manipulation on joint                                  | -  | 0         | -          | 0.0   | -  | 0        | -        | 0.0 |

The incidence of HAI occurring in patients whose primary operative procedure involved other miscellaneous operations not classified elsewhere Table A11.11:

| Onerstinne      |      |                                           |   | Individual p | procedures | 240    |   | Combined c | ategory |       |
|-----------------|------|-------------------------------------------|---|--------------|------------|--------|---|------------|---------|-------|
| involving:      | Code | Category                                  | E | No<br>HAI    | HAI        | IR (%) | E | No HAI     | HA      | %     |
| Amputation of   |      |                                           |   | nn           |            | -36    |   | d.         | 20      |       |
| hand leg or toe | X08  | Amputation of hand                        | 2 | 2            | 0          | 0.0    |   |            | c       | 37.6  |
|                 | 60X  | Amputation of leg                         | 2 | 0            | 8          | 100    | æ | 0          | 2       | C. 10 |
|                 | X11  | Amputation of toe                         | 3 | 2            | 1          | 33.3   |   |            | g)      |       |
|                 | X23  | Correction of congenital deformity of leg | - | -            | 0          | 0.0    |   |            | 3       |       |
| Miscellaneous   | X03  | Replacement of other organ                | 1 | 1            | 0          | 0.0    |   | 1          |         |       |
|                 | Y06  | Excision of lesion of organ nec           | 1 | 1            | 0          | 0.0    |   |            |         |       |
|                 | Y22  | Drainage or organ nec                     | - | 1            | 0          | 0.0    | თ | 0          | 6       | 0.0   |
|                 | Y30  | Incision of organ noc                     | 1 | -            | 0          | 0.0    |   |            | 54      |       |
|                 | V50  | Approach through abdominal cavity         | 5 | 5            | 0          | 0.0    |   | 2          | in.i    |       |

Results of the analysis that assessed how hospital costs varied with site of infection and selected patient characteristics

# A12.1 Introduction

The results of the analysis that assessed how hospital costs varied with site of infection and selected patient characteristics are presented in this appendix.

The results presented in tables A12.1 - A12.7 show that whilst for all sites of infection hospital costs, on average, are higher in infected than uninfected patients for all the patient characteristics examined, the level of increase varied with type of infection. However, it should be noted that in some cases the number of patients in the infected groups was small and as such the results cannot be generalized.

# Table A12.1: Mean in-patient hospital costs for patients with a urinary tract infection compared with those incurred by uninfected patients by key patient characteristics

|                                                                                                                          | Me   | an hospit | al costs (£ | E)  |                            |                                  |
|--------------------------------------------------------------------------------------------------------------------------|------|-----------|-------------|-----|----------------------------|----------------------------------|
| Petient oberneteristic                                                                                                   | N    | o HAI     | UTI or      | nly | Ratio of costs<br>(95% Cl) | Additional Costs (£)<br>(95% Cl) |
| Patient characteristic                                                                                                   | Mean | n         | Mean        | n   |                            | (00% 01)                         |
|                                                                                                                          | (a)  | -         | (b)         |     | (b/a)                      | (b-a)                            |
| Sex                                                                                                                      |      |           |             |     |                            |                                  |
| Male                                                                                                                     | 1439 | 946       | 3148        | 18  | 2.2 (1.0, 3.4)             | 1709 (1034, 2383)                |
| Female                                                                                                                   | 1810 | 1330      | 2781        | 70  | 1.5 (1.2, 1.8)             | 971 (612, 1331)                  |
| Age group                                                                                                                |      |           |             |     |                            |                                  |
| 18 –34                                                                                                                   | 1672 | 485       | 2614        | 15  | 1.6 (1.3, 1.8)             | 942 (234, 1651)                  |
| 35-54                                                                                                                    | 1509 | 627       | 2146        | 26  | 1.4 (1.2, 1.6)             | 636 (122, 1150)                  |
| 55-74                                                                                                                    | 1636 | 870       | 2908        | 32  | 1.8 (1.2, 2.4)             | 1272 (736, 1807)                 |
| 75+                                                                                                                      | 1997 | 294       | 4218        | 15  | 2.1 (0.9, 3.3)             | 2221 (1284, 3159)                |
| Specialty                                                                                                                |      |           |             |     |                            |                                  |
| General surgery                                                                                                          | 1338 | 829       | 5440        | 10  | 4.1 (1.0, 7.1)             | 4102 (2934, 5271)                |
| Orthopaedics                                                                                                             | 2157 | 462       | 4330        | 12  | 2.0 (1.4, 2.6)             | 2173 ( 1421, 2926)               |
| Urology                                                                                                                  | 1316 | 445       | 1952        | 17  | 1.5 (1.2, 1.8)             | 636 (222, 1051)                  |
| Gynaecology                                                                                                              | 1682 | 336       | 2102        | 36  | 1.2 (1.1, 1.4)             | 420 (176, 664)                   |
| Obstetrics                                                                                                               | 2508 | 204       | 2779        | 13  | 1.1 (0.5, 1.7)             | 271 (-648, 1190)                 |
| Admission type                                                                                                           |      |           |             |     |                            |                                  |
| Elective                                                                                                                 | 1569 | 1519      | 2061        | 58  | 1.3 (1.2, 1.4)             | 492 (178, 806)                   |
| Emergency                                                                                                                | 1828 | 757       | 4392        | 30  | 2.4 (1.6, 3.2)             | 2564 (1873, 3255)                |
| Discharge diagnosis group                                                                                                |      |           |             |     |                            |                                  |
| Infectious & parasitic diseases                                                                                          | 867  | 12        | 940         | 1   | 1.1 *                      | 73 *                             |
| Neoplasms                                                                                                                | 1714 | 329       | 2563        | 14  | 1.5 (1.1, 1.9)             | 850 (85, 1615)                   |
| Endocrine, nutritional & metabolic<br>diseases & immunity disorders                                                      | 1360 | 20        | at          | 0   |                            |                                  |
| Diseases of blood & blood forming<br>organs                                                                              | 992  | 3         | -           | 0   |                            | ar malaja                        |
| Diseases of the nervous system & sense organs                                                                            | 712  | 2         |             | 0   |                            |                                  |
| Diseases of the circulatory system                                                                                       | 1134 | 142       | 19781       | 1   | 17.4 *                     | 18647 *                          |
| Diseases of the respiratory system                                                                                       | 1726 | 4         |             | 0   |                            |                                  |
| Diseases of the digestive system                                                                                         | 1249 | 396       | 2007        | 7   | 1.6 (1.0, 2.2)             | 759 (-505, 2023)                 |
| Diseases of the genitourinary system                                                                                     | 1501 | 519       | 2061        | 34  | 14(1215)                   | 560 (255, 856)                   |
| Complications of pregnancy,<br>childbirth & puerperium & certain<br>conditions originating in the<br>perinatal period    | 2382 | 230       | 2612        | 15  | 1 1(0 8 1 2)               | 230 (-854, 713)                  |
| Diseases of the skin &                                                                                                   | LUUL | 200       | LUIL        |     | 1.1(0.0, 1.2)              | 200 ( 001, 110)                  |
| subcutaneous tissue                                                                                                      | 951  | 28        |             | 0   |                            |                                  |
| system & connective tissue                                                                                               | 2176 | 296       | 2904        | 3   | 13(10,17)                  | 728 (-613, 2070)                 |
| Injury & Poisoning                                                                                                       | 2136 | 160       | 4806        | 9   | 22(1431)                   | 2670 (1674, 3666)                |
| Symptoms, signs & ill- defined<br>conditions; mental disorders &<br>congenital abnormalities<br>sumber of co-morbidities | 1189 | 135       | 4865        | 4   | 4.1(1.9, 10.1)             | 3676 (2127, 5225)                |
| None                                                                                                                     | 1582 | 1532      | 247         | 56  | 16/12 20                   | 894 (552 1222)                   |
| One                                                                                                                      | 1722 | E20       | 2660        | 24  | 15(12 10)                  | 046 (255 1620)                   |
| Two                                                                                                                      | 2012 | 216       | 6082        | 8   | 3.0 (1.4, 4.7)             | 4069 (2590, 5548)                |

UTI - urinary tract infection. \*Numerator n=1 consequently standard deviation not estimable so confidence limits could not be estimated.

|                                                                                                                       | Mea  | an hospi | ital costs (£ | )    |                 | Additional costs  |
|-----------------------------------------------------------------------------------------------------------------------|------|----------|---------------|------|-----------------|-------------------|
|                                                                                                                       | Not  | IAI      | SWI on        | lv   | Ratio of costs  | (£)               |
| Patient characteristic                                                                                                | Mean | n        | Mean          | n    | (95% CI)        | (95% CI)          |
|                                                                                                                       | (2)  | <u> </u> | (b)           |      | (b/a)           | (h_a)             |
| Sex                                                                                                                   | (a)  |          | (0)           | -    | (bra)           | (b-a)             |
| Male                                                                                                                  | 1430 | 946      | 4105          | 1 16 | 29/1146)        | 2669(974 4364)    |
| - Male                                                                                                                | 4040 | 1220     | 2266          |      | 12(00 18)       | 556 ( 224, 1247)  |
| Female                                                                                                                | 1010 | 1330     | 2300          | 1/   | 1.3 (0.9, 1.6)  | 556 (-254, 1547)  |
| Age group                                                                                                             |      |          |               |      |                 |                   |
| 18-34                                                                                                                 | 1672 | 485      | 3788          | 8 7  | 2.3 (0.8, 3.7)  | 2116 (1050, 1868) |
| 35-54                                                                                                                 | 1509 | 627      | 2070          | 4    | 1.4 (0.7, 2.1)  | 561 (-747, 1868)  |
| 55-74                                                                                                                 | 1636 | 870      | 2865          | 16   | 1.8 (1.2, 2.3)  | 1228 (506, 1951)  |
| 75+                                                                                                                   | 1997 | 294      | 4242          | 5    | 2.1 (1.0, 3.30  | 2245 (876, 3613)  |
| Specialty                                                                                                             |      |          |               |      |                 |                   |
| General surgery                                                                                                       | 1338 | 829      | 3325          | 12   | 2.5 (1.6, 3.4)  | 1988 (1000, 2796) |
| Orthopaedics                                                                                                          | 2157 | 462      | 3612          | 11   | 1.7 (0.9, 2.5)  | 1455 (658, 2252)  |
| Urology                                                                                                               | 1316 | 445      | 4525          | 2    | 3.4 (0.1, 6.9)  | 3208(2000, 4416)  |
| Gynaecology                                                                                                           | 1682 | 336      | 1473          | 4    | 09(04 13)       | -208 (-921, 504)  |
| Obstatzics                                                                                                            | 2508 | 204      | 2421          | 3    | 10(07 12)       | -88 (-1840 1665)  |
|                                                                                                                       | 2500 | 204      | 2421          |      | 1.0 (0.7, 1.2)  | -00 (-1040, 1000) |
| Flective                                                                                                              | 1569 | 1519     | 2715          | 16   | 17(1322)        | 1146 (547, 1744)  |
| Emorgonou                                                                                                             | 1929 | 757      | 2650          | 10   | 20(12 28)       | 1822 (036, 2707)  |
| Emergency                                                                                                             | 1020 | 151      | 3050          | 10   | 2.0 (1.2, 2.8)  | 1022 (930, 2101)  |
| Discharge diagnosis group                                                                                             | 867  | 12       |               | - 0  |                 |                   |
| Neoplasms                                                                                                             | 1714 | 329      | 3929          | 3    | 2.3 (1.6, 3.0)  | 2216 (579, 3833)  |
| Endocrine, nutritional & metabolic                                                                                    |      |          |               |      |                 |                   |
| diseases & immunity disorders                                                                                         | 1360 | 20       | 1164          | 1    | 0.9 *           | -197 *            |
| Diseases of blood & blood forming                                                                                     | 992  | 3        |               |      |                 |                   |
| Diseases of the nervous system &                                                                                      | 002  |          |               |      |                 |                   |
| sense organs                                                                                                          | 712  | 2        |               | 0    |                 |                   |
| Diseases of the circulatory system                                                                                    | 1134 | 142      | 3451          | 1    | 3.0 *           | 2317 *            |
| Diseases of the respiratory system                                                                                    | 1726 | 4        |               | 0    |                 |                   |
| Diseases of the digestive system                                                                                      | 1249 | 396      | 3073          | 4    | 2.5 (1.0, 3.9)  | 1825 (149, 3501)  |
| Diseases of the genitourinary                                                                                         |      |          |               |      | 2.0 (           |                   |
| system                                                                                                                | 1501 | 519      | 4339          | 2    | 2.9 (0.4, 6.2)  | 2838 (1571, 4104) |
| Complications of pregnancy,<br>childbirth & puerperium & certain<br>conditions originating in the perinatal<br>period | 2382 | 230      | 2421          | 3    | 1.0 (0.7, 1.3)  | 38 (-699, 1775)   |
| Diseases of the skin &                                                                                                | 054  | 20       | 1050          |      | 17(12.22)       | 706 ( 556 1069)   |
| Diseases of the musculoskletal                                                                                        | 951  | 28       | 1656          | 1    | 1.7 (1.3, 2.2)  | 706 (-556, 1968)  |
| system & connective tissue                                                                                            | 2176 | 296      | 2762          | 5    | 1.3 (0.9, 1.6)  | 586 (-4456, 1628) |
| Injury & Poisoning                                                                                                    | 2136 | 160      | 3108          | 10   | 1.5 (0.5, 2.4)  | 972 (-30, 1973)   |
| Symptoms, signs & ill- defined<br>conditions; mental disorders &<br>congenital abnormalities                          | 1189 | 135      | 5331          | 2    | 4.5 (04.4, 9.4) | 4142 (2489, 5795) |
| None                                                                                                                  | 4500 | 1500     | 0004          | 10   | 15/10 20        | 740 (187 1313)    |
| One                                                                                                                   | 1582 | 528      | 2331          | 10   | 2.2 (1.3, 3.0)  | 2027 (958, 3095)  |
| Two                                                                                                                   | 2012 | 216      | 5596          | 4    | 2.8 (1.3, 4.2)  | 3584 (1655, 5512) |

# Table A12.2: Mean in-patient hospital costs for patients with a surgical wound infection compared with those incurred by uninfected patients by key patient characteristics

SWI - surgical wound infection

|                                                                                                                       | Me   | an hospit | al costs (£) | )     |                |                         |
|-----------------------------------------------------------------------------------------------------------------------|------|-----------|--------------|-------|----------------|-------------------------|
|                                                                                                                       | No   | HAI       | LR           | ТІ    | Ratio of costs | Additional Costs<br>(£) |
| Patient characteristic                                                                                                | Mean | n         | Mean         | n     | (95% CI)       | (95% Cl)                |
|                                                                                                                       | (a)  |           | (b)          |       | (b/a)          | (b-a)                   |
| Sex                                                                                                                   |      | -         |              |       |                |                         |
| Male                                                                                                                  | 1439 | 946       | 5450         | 9     | 3.8 (1.3, 6.3) | 4011 (1778, 6244)       |
| Female                                                                                                                | 1810 | 1330      | 2613         | 6     | 1.4 (0.8, 2.1) | 804 (-435, 2042)        |
| Age group                                                                                                             |      |           |              |       |                |                         |
| 18 34                                                                                                                 | 1672 | 485       | 4693         | 1     | 2.8 *          | 3022 *                  |
| 35-54                                                                                                                 | 1509 | 627       | 2081         | 2     | 1.4 (1.1. 1.7) | 572 (-1274, 2418)       |
| 55-74                                                                                                                 | 1636 | 870       | 5553         | 7     | 3.4 (1.4. 5.3) | 3917 (2809, 5024)       |
| 75+                                                                                                                   | 1997 | 204       | 3400         | 5     | 17(0827)       | 1403 (41 2764)          |
| Specialty                                                                                                             | 1007 | 204       | 0400         | -     | 1.7 (0.0, 2.1) | 1100 (11, 210.)         |
| General surgery                                                                                                       | 1338 | 829       | 5068         | 10    | 3.8 (2.0, 5.6) | 3730 (2630, 4830)       |
| Orthonaedics                                                                                                          | 2157 | 462       | 2825         | 2     | 13(1115)       | 668 (-1106, 2442)       |
| Urology                                                                                                               | 1316 | 445       | 1392         | 1     | 1.1 *          | 76 *                    |
| Gynaecology                                                                                                           | 1682 | 336       | 2313         | 1     | 1.4 *          | 632 *                   |
| Obstetrics                                                                                                            | 2508 | 204       | 4693         | 1     | 1.9*           | 2185 *                  |
| Admission type                                                                                                        |      |           |              | · · · |                |                         |
| Elective                                                                                                              | 1569 | 1519      | 4820         | 8     | 3.1 (1.5,4.6)  | 3251 (2395, 4107)       |
| Emergency                                                                                                             | 1828 | 757       | 3738         | 7     | 2.0 (0.7.3.4)  | 1910 (589, 3230)        |
| Discharge diagnosis group                                                                                             | TOLO | 1         | 0100         |       | 2.0 (0.7, 0.1) |                         |
| Infectious & parasitic diseases                                                                                       |      |           |              |       |                |                         |
| Neerleeme                                                                                                             | 1714 | 12        | 7143         | 0     | 42(1172)       | 5430 (3746 7113)        |
| Endocrine nutritional & metabolic                                                                                     | 1/14 | 329       | 7145         |       | 4.2 (1.1, 7.2) | 5450 (5740, 7115)       |
| diseases & immunity disorders                                                                                         | 1360 | 20        | 4389         | 1     | 3.2 *          | 3028 *                  |
| Diseases of blood & blood forming                                                                                     |      |           |              |       |                |                         |
| organs                                                                                                                | 992  | 3         |              | 0     |                |                         |
| sense organs                                                                                                          | 712  | 2         | _            | 0     | _              |                         |
| Diseases of the circulatory system                                                                                    | 1134 | 142       |              | 0     |                |                         |
| Diseases of the respiratory system                                                                                    | 1726 |           | 10738        | 4     | 62*            | 9013 *                  |
| Diseases of the digestive system                                                                                      | 1240 | 206       | 10750        |       | 1 = (1 0 2 0)  | 642 ( 1027 2212)        |
| Diseases of the genitourinary                                                                                         | 1249 | 390       | 1891         | 4     | 1.5 (1.0, 2.0) | 043 (-1027, 2312)       |
| system                                                                                                                | 1501 | 519       | 2313         | 1     | 1.5*           | 813*                    |
| Complications of pregnancy,<br>childbirth & puerperium & certain<br>conditions originating in the<br>perinatal period | 2382 | 230       | 4693         | 1     | 2.0*           | 2311*                   |
| Diseases of the skin &                                                                                                |      |           |              |       |                |                         |
| Diseases of the musculocklotal                                                                                        | 951  | 28        |              | 0     |                |                         |
| system & connective tissue                                                                                            | 2176 | 296       | 2825         | 2     | 1.3 (1.1, 1.5) | 649 (-992, 2290)        |
| Injury & Poisoning                                                                                                    | 2136 | 160       |              | 0     |                |                         |
| Symptoms, signs & ill- defined<br>conditions; mental disorders &<br>congenital abnormalities                          | 1189 | 135       | 3974         | 2     | 3.3 (0.9, 7.6) | 2785 (1150, 4421)       |
| Number of co-morbidities                                                                                              | 4500 | 4800      | 0700         | -     | 10/10 05       | 1000 (110 0000)         |
| One                                                                                                                   | 1582 | 1532      | 2788         | 5     | 1.8 (1.0, 2.5) | 1206 (149, 2263)        |
| Two                                                                                                                   | 2012 | 216       | 5215         | 7     | 2.6 (0.9, 4.3) | 3203 (1704, 4701)       |

# Table A12.3: Mean in-patient hospital costs for patients with a lower respiratory tract infection compared with those incurred by uninfected patients by key patient characteristics

LRTI - lower respiratory tract infection

|                                                                                                             | Many kany last and 101  |      |       |   | Ratio of costs  | Additional Costs (£)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|-------------------------------------------------------------------------------------------------------------|-------------------------|------|-------|---|-----------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Patient characteristic                                                                                      | Mean hospital costs (£) |      |       |   |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|                                                                                                             | No HAI                  |      | BSI   |   | (95% CI)        | (95% Ci)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|                                                                                                             | Mean                    | n    | Mean  | n |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|                                                                                                             | (a)                     |      | (b)   |   | (b/a)           | (b-a)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| Sex                                                                                                         |                         |      |       |   |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Male                                                                                                        | 1439                    | 946  | 8953  | 3 | 6.2 (0.2, 12.3) | 7514 *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| Female                                                                                                      | 1810                    | 1330 |       | 0 | -               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Age group                                                                                                   |                         |      |       |   |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 18 –34                                                                                                      | 1672                    | 485  | 15937 | 1 | 9.5 *           | 14265 *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 35-54                                                                                                       | 1509                    | 627  | 8263  | 1 | 5.5 *           | 6753 *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| 55-74                                                                                                       | 1636                    | 870  | 2660  | 1 | 1.6 *           | 1024 *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| 75+                                                                                                         | 1997                    | 294  |       | 0 | -               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Specialty                                                                                                   |                         |      |       |   |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| General surgery                                                                                             | 1338                    | 829  | 8263  | 1 | 6.2 *           | 6925 *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| Orthopaedics                                                                                                | 2157                    | 462  | 15937 | 1 | 7.4 *           | 13779 *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| Urology                                                                                                     | 1316                    | 445  | 2660  | 1 | 2.0 *           | 1344 *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| Gynaecology                                                                                                 | 1682                    | 336  | _     | 0 |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Obstatrics                                                                                                  | 2508                    | 204  |       | 0 |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Admission type                                                                                              | 2000                    | 201  |       | - |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Elective                                                                                                    | 1569                    | 1519 | 2660  | 1 | 1.7 *           | 1091 *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| Electric                                                                                                    | 1828                    | 757  | 12100 | 2 | 66 (2 5 10 8)   | 10271 (7818, 12724)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| Discharge diagnosis group                                                                                   | 1020                    | 101  | 12100 | - | 0.0 (2.0, 10.0) |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Infectious & parasitic diseases                                                                             | 967                     | 12   |       | 0 |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Neoplasms                                                                                                   | 1714                    | 220  |       | 0 |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Endocrine putritional & metabolic                                                                           | 1/14                    | 329  |       | 0 |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| diseases & immunity disorders                                                                               | 1360                    | 20   |       | 0 |                 | ****                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Diseases of blood & blood forming                                                                           | 000                     | 2    |       |   |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Organs                                                                                                      | 992                     | 3    |       | 0 |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| sense organs                                                                                                | 712                     | 2    |       | 0 |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Diseases of the circulatory system                                                                          | 1134                    | 142  |       | 0 |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Diseases of the respiratory system                                                                          | 1726                    | 4    | 15937 | 1 | 9.2 *           | 14211 *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| Diseases of the digestive system                                                                            | 1249                    | 396  | 8263  | 1 | 6.6 *           | 7014 *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| Diseases of the genitourinary                                                                               |                         |      |       |   |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| system                                                                                                      | 1501                    | 519  | 2660  | 1 | 1.8 *           | 1160 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| complications of pregnancy,<br>childbirth & puerperium & certain<br>conditions originating in the perinatal |                         |      |       |   |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Diseases of the skin &                                                                                      | 2382                    | 230  |       | 0 |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| subcutaneous tissue                                                                                         | 951                     | 28   |       | 0 |                 | 19197155                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| Diseases of the musculoskletal                                                                              | 2176                    | 296  |       | 0 |                 | and an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| Injury & Poisoning                                                                                          | 2136                    | 160  |       | 0 |                 | North Contraction of the second secon |
| Symptoms, signs & ill- defined<br>conditions; mental disorders &<br>congenital abnormalities                | 1189                    | 135  |       | 0 | Ighter          | 4855                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Number of co-morbidities                                                                                    | in the second           |      |       |   |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| None                                                                                                        | 1582                    | 1532 | 2660  | 1 | 1.7*            | 1078*                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| One                                                                                                         | 1723                    | 528  |       | 0 |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Two                                                                                                         | 2012                    | 216  | 12100 | 2 | 6.0 (2.2, 9.8)  | 10087(7357, 12818)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |

# Table A12.4: Mean in-patient hospital costs for patients with a bloodstream infection compared with those incurred by uninfected patients by key patient characteristics

BSI - bloodstream infection

|                                                                                                             | Mea                   | n hospi | tal costs ( | £) |                  | T                                |
|-------------------------------------------------------------------------------------------------------------|-----------------------|---------|-------------|----|------------------|----------------------------------|
| Patient characteristic                                                                                      | No HAI                |         | Skin        |    | Ratio of costs   | Additional Costs (£)<br>(95% Cl) |
|                                                                                                             | Mean n                |         | Mean n      |    | (95% CI)         |                                  |
|                                                                                                             | (a)                   |         | (b)         |    | (b/a)            | (b-a)                            |
| Sex                                                                                                         | 1                     |         |             |    |                  | <u> </u>                         |
| Male                                                                                                        | 1439                  | 946     | 3298        | 5  | 2.3 (0.8, 3.8)   | 1859 (276, 3441)                 |
| Eemale                                                                                                      | 1810                  | 1330    | 3793        | 8  | 21(1131)         | 1983 (774, 3193)                 |
|                                                                                                             | 1010                  | 1000    |             |    | 2 (, 0)          |                                  |
| 18-34                                                                                                       | 1672                  | 485     | 2344        | 1  | 1.4 *            | 673 *                            |
| 35.54                                                                                                       | 1509                  | 627     | 2797        | 3  | 19(0631)         | 1287 (-223, 2798)                |
| 65-54<br>EE 74                                                                                              | 1636                  | 870     | 4031        | 7  | 25 (1 4 3 5)     | 2394 (1309 3479)                 |
| 55-74                                                                                                       | 1000                  | 070     | 4031        | 1  | 2.0 (1.4, 3.5)   | 1045 (1910 6075)                 |
| 75+<br>Enecialty                                                                                            | 1997                  | 294     | 3942        | 2  | 2.0 (0.4, 3.5)   | 1945 (1810, 8075)                |
|                                                                                                             | 4000                  | 000     | 2570        | 5  | 27/11 4 2)       | 2222 (710 3754)                  |
| General surgery                                                                                             | 1338                  | 829     | 3570        | 5  | 2.7 (1.1, 4.2)   | 2232 (710, 3734)                 |
| Orthopaedics                                                                                                | 2157                  | 462     | 3565        | 5  | 1.7 (0.8, 2.5)   | 1407 (274, 2541)                 |
| Urology                                                                                                     | 1316                  | 445     | 5919        | 1  | 4.5 *            | 4603 *                           |
| Gynaecology                                                                                                 | 1682                  | 336     |             | 0  |                  |                                  |
| Obstetrics                                                                                                  | 2508                  | 204     | 2621        | 2  | 1.0 (0.8, 1.3)   | 113 (-2033, 2259)                |
| Admission type                                                                                              |                       |         |             | -  |                  |                                  |
| Elective                                                                                                    | 1569                  | 1519    | 3354        | 7  | 2.1 (1.3, 3.0)   | 1784 (882, 2686)                 |
| Emergency                                                                                                   | 1828                  | 757     | 3893        | 6  | 2.1 (0.4, 3.8)   | 2065 (634, 3495)                 |
| Discharge diagnosis group                                                                                   | and the second second |         |             |    |                  |                                  |
| Infectious & parasitic diseases                                                                             | 867                   | 12      |             | 0  |                  |                                  |
| Neoplasms                                                                                                   | 1714                  | 329     | 4339        | 1  | 2.5 *            | 2626 *                           |
| Endocrine, nutritional & metabolic                                                                          | 3,3995                |         |             |    |                  |                                  |
| diseases & immunity disorders                                                                               | 1360                  | 20      | 1152        | 1  | 0.8 *            | -208 *                           |
| Diseases of blood & blood forming<br>organs                                                                 | 992                   | 3       |             | 0  |                  |                                  |
| Diseases of the nervous system &                                                                            |                       |         |             |    |                  |                                  |
| sense organs                                                                                                | 712                   | 2       |             | 0  |                  |                                  |
| Diseases of the circulatory system                                                                          | 1134                  | 142     | 2352        | 1  | 2.1 *            | 1218 *                           |
| Diseases of the respiratory system                                                                          | 1726                  | 4       |             | 0  | -                |                                  |
| Diseases of the digestive system                                                                            | 1249                  | 396     | 7247        | 1  | 5.8 *            | 5999 *                           |
| Diseases of the genitourinary system                                                                        | 1501                  | 519     | 4338        | 2  | 2.9 (0.8, 5.0)   | 2838 (1582, 4093)                |
| Complications of pregnancy,<br>childbirth & puerperium & certain<br>conditions originating in the perinatal | arata 1               |         | 21.2        |    |                  |                                  |
| period                                                                                                      | 2382                  | 230     | 2621        | 2  | 1.1 (0.9, 1.3)   | 239 (-1888, 2365)                |
| Diseases of the skin &<br>subcutaneous tissue                                                               | 951                   | 28      | 953         | 1  | 1.0 *            | 3 *                              |
| Diseases of the musculoskletal                                                                              |                       |         |             |    |                  | 1050 (000 0001)                  |
| system & connective tissue                                                                                  | 2176                  | 296     | 4128        | 2  | 1.9 0.6, 3.2)    | 1952 (302, 3601)                 |
|                                                                                                             | 2136                  | 160     | 4307        | 2  | 2.0 (0.7, 3.3)   | 2171 (201, 4141)                 |
| Symptoms, signs & ill- defined<br>conditions; mental disorders &<br>congenital abnormalities                | 1189                  | 135     | _           | 0  |                  | -                                |
| Number of co-morbidities                                                                                    |                       |         |             |    |                  |                                  |
| None                                                                                                        | 1582                  | 1532    | 1942        | 2  | 4 1.2 (0.6, 1.9) | 360 (-820, 1541)                 |
| One                                                                                                         | 1723                  | 528     | 3282        | 2  | 4 1.9 (1.0, 2.8) | 1559 (-114, 3233)                |
| Тwo                                                                                                         | 2012                  | 216     | 5187        | 1  | 52.6 (1.8, 3.3)  | 3175 (1458, 4892)                |

### Table A12.5: Mean in-patient hospital costs for patients with a skin infection compared with those incurred by uninfected patients by selected characteristics
# Table A12.6: Mean in-patient hospital costs for patients with a single site infection at a site not classified elsewhere compared with those incurred by uninfected patients by key patient characteristics

|                                                                                                             | Mean hospital costs (£) |            |             |    | D dia at a star | Additional Costs (C) |  |
|-------------------------------------------------------------------------------------------------------------|-------------------------|------------|-------------|----|-----------------|----------------------|--|
| Potiont characteristic                                                                                      | No                      | HAI        | Other site* |    | (95% CI)        | (95% CI)             |  |
| Patient characteristic                                                                                      | Mean                    | n          | Mean        | n  |                 |                      |  |
|                                                                                                             | (a)                     |            | (b)         |    | (b/a)           | (b-a)                |  |
| Sex                                                                                                         |                         |            |             |    |                 |                      |  |
| Male                                                                                                        | 1439                    | 946        | 5278        | 3  | 3.7 (1.8, 9.2)  | 3839 (255, 7424)     |  |
| Female                                                                                                      | 1810                    | 1330       | 3037        | 15 | 1.7 (0.7, 2.6)  | 1227 (29, 2425)      |  |
| Age group                                                                                                   |                         | 1.000      |             |    |                 |                      |  |
| 18_34                                                                                                       | 1672                    | 485        | 6040        | 3  | 3.6 (1.0, 8.2)  | 4369 (2717, 6020)    |  |
| 25.54                                                                                                       | 1500                    | 627        | 2410        | 5  | 16(08 24)       | 901 (-271 2072)      |  |
| 30-04                                                                                                       | 1509                    | 021        | 2410        |    | 1.0 (0.0, 2.4)  | 001 (2/1, 20/2)      |  |
| 55-74                                                                                                       | 1636                    | 870        | 3681        | 7  | 2.2 (0.4, 4.1)  | 2040 (940, 3150)     |  |
| 75+                                                                                                         | 1997                    | 294        | 1816        | 3  | 0.9 (0.2, 1.6)  | -181 (78, 3554)      |  |
| Specialty                                                                                                   |                         |            |             |    |                 |                      |  |
| General surgery                                                                                             | 1338                    | 829        | 2474        | 6  | 1.8 (1.0, 2.7)  | 1136 (-250, 2522)    |  |
| Orthopaedics                                                                                                | 2157                    | 462        | 5319        | 4  | 2.5 (0.1, 5.1)  | 3162 (1829, 4495)    |  |
| Urology                                                                                                     | 1316                    | 445        | 4983        | 3  | 3.8 (2.1, 9.7)  | 3667 92564, 4769)    |  |
| Gynaecology                                                                                                 | 1682                    | 336        | 1957        | 2  | 1.2 (0.8, 1.5)  | 276 (-728, 1279)     |  |
| Obstetrics                                                                                                  | 2508                    | 204        | 2136        | 3  | 0.9 (0.8, 0.9)  | -373 (-2124, 1379)   |  |
| Admission type                                                                                              | 2000                    |            |             |    |                 |                      |  |
| Elective                                                                                                    | 1569                    | 1519       | 2440        | 6  | 1.6 (0.9, 2.2)  | 871 (-101, 1843)     |  |
| Emergency                                                                                                   | 1828                    | 757        | 3896        | 12 | 2.1 (0.7, 3.5)  | 2067 (1026, 3109)    |  |
| Discharge diagnosis group                                                                                   | and the second          | 1.12231    |             |    |                 |                      |  |
| Infectious & parasitic diseases                                                                             | 867                     | 12         |             | 0  |                 |                      |  |
| Neoplasms                                                                                                   | 1714                    | 329        | 1607        | 1  | 0. 0**          | -106**               |  |
| Endocrine, nutritional & metabolic                                                                          |                         |            |             |    |                 |                      |  |
| diseases & immunity disorders                                                                               | 1360                    | 20         | 1705        | 1  | 1.3 **          | 345**                |  |
| Diseases of blood & blood forming<br>organs                                                                 | 992                     | 3          |             | 0  |                 |                      |  |
| Diseases of the nervous system &                                                                            |                         |            |             |    |                 |                      |  |
| sense organs                                                                                                | 712                     | 2          |             | 0  |                 | *****                |  |
| Diseases of the circulatory system                                                                          | 1134                    | 142        | 4804        | 1  | 4.2 **          | 3670**               |  |
| Diseases of the respiratory system                                                                          | 1726                    | 4          |             | 0  |                 |                      |  |
| Diseases of the digestive system                                                                            | 1249                    | 396        | 1707        | 2  | 1.4 (1.1, 1.6)  | 458 (-1901, 2818)    |  |
| Diseases of the genitourinary<br>system                                                                     | 1501                    | 519        | 1566        | 3  | 1.0 (0.5, 1.6)  | 65 (-955, 1085)      |  |
| Complications of pregnancy,<br>childbirth & puerperium & certain<br>conditions originating in the perinatal |                         | 000        | 0100        |    | 0.0 (0.8, 1.0)  | 247 ( 1083 1480)     |  |
| Discasso of the chin f                                                                                      | 2382                    | 230        | 2136        | 3  | 0.9 (0.8, 1.0)  | -247 (-1903, 1408)   |  |
| subcutaneous tissue                                                                                         | 951                     | 28         | 13839       | 1  | 14.6 **         | 12888 **             |  |
| Diseases of the musculoskletal                                                                              | 2176                    | 206        | 1           | 0  |                 |                      |  |
| loiury & Poisoning                                                                                          | 2170                    | 160        | 2480        | 3  | 12(09 15)       | 344 (-1257, 1944)    |  |
| Symptoms signs & ill_defined                                                                                | 2150                    | 100        | 2400        | 5  | 1.2 (0.0, 1.0)  |                      |  |
| conditions; mental disorders & congenital abnormalities                                                     | 1189                    | 135        | 5826        | 3  | 4.9 (1.1, 10.1) | 4637 (3085, 6189)    |  |
| Number of co-morbidities                                                                                    |                         | ( interest |             |    |                 |                      |  |
| None                                                                                                        | 1582                    | 1532       | 3021        | 11 | 1.9 (0.5, 3.3)  | 1439 (706, 2171)     |  |
| One                                                                                                         | 1723                    | 528        | 4981        | 4  | 2.9 (0.1, 5.9)  | 3258 (1555, 4962)    |  |
| Two                                                                                                         | 2012                    | 216        | 2746        | 3  | 1.4 (0.2, 2.9)  | 734 (1486, 2953)     |  |

\*single site infections at sites not classified elsewhere i.e. infections at a sites other than the urinary tract surgical wound, lower respiratory tract, skin, or bloodstream

\*Numerator n=1 consequently standard deviation not estimable so confidence limits could not be estimated.

# Table A12.7: Mean in-patient hospital costs for patients with more than one HAI compared with those incurred by uninfected patients by key patient characteristics

| the description of heats                                                                                              | Mean hospital costs (£) |      |        |        | astis-china v    |                      |  |
|-----------------------------------------------------------------------------------------------------------------------|-------------------------|------|--------|--------|------------------|----------------------|--|
|                                                                                                                       | No                      | HAI  | Multi  | ple    | Ratio of costs   | Additional Costs(£)  |  |
| Patient characteristic                                                                                                | Mean                    | n    | Mean   | n      | (35% CI)         | (0070 01)            |  |
|                                                                                                                       | (a)                     |      | (b)    |        | (b/a)            | (b-a)                |  |
| Sex                                                                                                                   |                         |      |        |        |                  |                      |  |
| Male                                                                                                                  | 1439                    | 946  | 13044  | 4      | 9.1 (4.7, 13.5)  | 11605 (10215, 12995) |  |
| Female                                                                                                                | 1810                    | 1330 | 9120   | 20     | 5.0 (2.7, 7.4)   | 7310 (6497, 8127)    |  |
| Age group                                                                                                             |                         |      |        |        |                  |                      |  |
| 18 - 34                                                                                                               | 1672                    | 485  | 1659   | 1      | 1.0*             | -12*                 |  |
| 35-54                                                                                                                 | 1509                    | 627  | 12923  | 5      | 8.6 (0.0, 17.6)  | 11413 (9814, 13012)  |  |
| 55-74                                                                                                                 | 1636                    | 870  | 6944   | 11     | 4.2 (1.9, 6.6)   | 5307 (4354, 6260)    |  |
| 75+                                                                                                                   | 1997                    | 294  | 13131  | 7      | 6.6              | 11134                |  |
| Specialty                                                                                                             | 1337                    | 204  | 10101  |        | 0.0              |                      |  |
| General surgery                                                                                                       | 1338                    | 829  | 13150  | 11     | 9.8 (6.8, 12.9)  | 11812 (10703, 12921) |  |
| Orthomodia                                                                                                            | 2157                    | 462  | 16096  | 4      | 75(00,150)       | 13939 (12125, 1573)  |  |
| Orthopaedics                                                                                                          | 1010                    | 402  | 4942   | 2      | 37 (22 5 1)      | 3526 (2341 4724)     |  |
| Urology                                                                                                               | 1316                    | 440  | 4042   |        | 3.7 (2.2, 5.1)   | 502 (2541, 4724)     |  |
| Gynaecology                                                                                                           | 1682                    | 336  | 2265   | 1      | 1.3 (0.8, 1.8)   | 583 (36, 1130)       |  |
| Obstetrics                                                                                                            | 2508                    | 204  |        | 0      |                  | 9.85 B               |  |
| Admission type                                                                                                        |                         |      |        |        |                  |                      |  |
| Elective                                                                                                              | 1569                    | 1519 | 6361   | 14     | 4.1(1.4, 6.7)    | 4792 (4001, 5581)    |  |
| Emergency                                                                                                             | 1828                    | 757  | 14552  | 10     | 8.0 (4.4, 11.5)  | 12723 (11423, 1402)  |  |
| Discharge diagnosis group                                                                                             |                         | 0.02 | 38     | 1.56.5 |                  |                      |  |
| Infectious & parasitic diseases                                                                                       | 867                     | 12   |        | 0      |                  |                      |  |
| Neoplasms                                                                                                             | 1714                    | 329  | 12051  | 6      | 7.0 (3.6, 10.4)  | 10338 (8982, 11693)  |  |
| Endocrine, nutritional & metabolic<br>diseases & immunity disorders                                                   | 1360                    | 20   | _      | 0      | -                |                      |  |
| Diseases of blood & blood forming<br>organs                                                                           | 992                     | 3    | _      | 0      |                  |                      |  |
| Diseases of the nervous system &                                                                                      | 712                     | 2    |        | 0      |                  |                      |  |
| Diseases of the circulatory system                                                                                    | 1124                    | 142  | 16625  | 4      | 147 (9 2 20 1)   | 15491 (13509 17472)  |  |
| Diseases of the respiratory system                                                                                    | 4700                    | 142  | 10025  | -      | 14.7 (0.2, 20.1) |                      |  |
| Diseases of the digestive system                                                                                      | 1/26                    | 4    |        | 0      |                  | 2006 (1078 5872)     |  |
| Diseases of the geniteurings                                                                                          | 1249                    | 396  | 51/4   | 3      | 4.1 (0.9, 7.4)   | 3920 (1978, 5675)    |  |
| system                                                                                                                | 1501                    | 519  | 2265   | 7      | 1.5 (0.9, 2.1)   | 764 (91,1436)        |  |
| Complications of pregnancy,<br>childbirth & puerperium & certain<br>conditions originating in the<br>perinatal period | 2382                    | 230  | 2.<br> | 0      |                  |                      |  |
| Diseases of the skin &                                                                                                | 2302                    | 2.50 |        | -      |                  |                      |  |
| subcutaneous tissue                                                                                                   | 951                     | 28   |        | 0      |                  |                      |  |
| Diseases of the musculoskletal                                                                                        | 2176                    | 206  | 3950   | 1      | 1.8*             | 1774*                |  |
| Injury & Poisoning                                                                                                    | 2170                    | 200  | 20145  | 1      | 0.4 (0.0. 19.9)  | 18000 (15230 20770)  |  |
| Symptoms, signs & ill- defined<br>conditions; mental disorders &<br>congenital abnormalities                          | 1189                    | 135  |        | 0      |                  |                      |  |
| Number of co-morbidities                                                                                              | 1                       |      |        |        |                  |                      |  |
| None                                                                                                                  | 1592                    | 1532 | 8556   | 15     | 54(1989)         | 6974 (4936 9012)     |  |
| One                                                                                                                   | 1723                    | 528  | 13268  | 5      | 7.7 (3.3, 12.1)  | 11546 (9913, 13178)  |  |
| Two                                                                                                                   | 2012                    | 216  | 9971   | 4      | 5.0 (2.7, 7.2)   | 7958 (5971, 9946)    |  |

\*Numerator n=1 consequently standard deviation not estimable so confidence limits could not be estimated.

The distribution of hospital costs incurred by patients with specific types of infection

## A13.1 Introduction

The results of the analysis that assessed the distribution of the additional costs incurred by patients acquiring specific types of infection are presented in this appendix. The results presented in tables A13.1 – A13.7 show the distribution of the costs incurred by patients acquiring specific types of infection and how these costs compare to those incurred by uninfected patients.

| Paik or                                  | Mean hospit | tal costs (£) |          |            | % contribution |
|------------------------------------------|-------------|---------------|----------|------------|----------------|
| Onie                                     | No HAI      | UTI           | Ratio of | Additional |                |
|                                          | n=2276      | n=88          | costs    | costs (£)  | costs          |
| Hospital overheads                       | 392.89      | 700.0         | 1.8      | 307.08     | 25.58          |
| Directorate management                   | 47.55       | 74.9          | 1.6      | 27.30      | 2.27           |
| Capital charges                          | 208.20      | 350.0         | 1.7      | 141.76     | 11.81          |
| Medical time                             | 150.72      | 242.1         | 1.6      | 91.42      | 7.61           |
| Nursing care                             | 338.11      | 746.7         | 2.2      | 408.60     | 34.03          |
| Paramedics & specialist nurses           | 11.71       | 24.0          | 2.0      | 12.29      | 1.02           |
| Physiotherapy                            | 16.01       | 39.7          | 2.5      | 23.65      | 1.97           |
| Consumables used for specific procedures | 11.06       | 33.1          | 3.0      | 22.09      | 1.84           |
| Surgical interventions                   | 335.63      | 441.8         | 1.3      | 106.16     | 8.84           |
| Antimicrobials                           | 11.80       | 13.4          | 1.1      | 1.60       | 0.13           |
| Non-antimicrobial drugs                  | 32.92       | 46.8          | 1.4      | 13.86      | 1.15           |
| Microbiology tests                       | 5.74        | 15.2          | 2.7      | 9.48       | 0.79           |
| Other pathology tests                    | 54.59       | 104.9         | 1.9      | 50.33      | 4.19           |
| Endoscopies                              | 1.71        | 1.4           | 0.8      | -0.27      | -0.02          |
| Radiology                                | 33.35       | 19.1          | 0.6      | -14.22     | -1.18          |
| Other tests                              | 3.53        | 3.0           | 0.9      | -0.51      | -0.04          |
| Total costs                              | 1655.51     | 2856.1        | 1.7      | 1200.62    | 100.00         |

| Table A13.1: | The distribution of hospital costs incurred by patients who      |     |
|--------------|------------------------------------------------------------------|-----|
|              | acquired a urinary tract infection compared to uninfected patien | Its |

UTI - urinary tract infection

| pettorito                                |                         |         |          | 1          | 1             |  |
|------------------------------------------|-------------------------|---------|----------|------------|---------------|--|
|                                          | Mean hospital costs (£) |         |          |            | %             |  |
|                                          | No HAI                  | SWI     | Ratio of | Additional | to additional |  |
|                                          | n=2276                  | n=32    | costs    | costs (£)  | costs         |  |
| Hospital overheads                       | 392.89                  | 724.87  | 1.84     | 331.98     | 21.74         |  |
| Directorate management                   | 47.55                   | 91.31   | 1.92     | 43.75      | 2.87          |  |
| Capital charges                          | 208.20                  | 386.17  | 1.85     | 177.96     | 11.65         |  |
| Medical time                             | 150.72                  | 306.47  | 2.03     | 155.75     | 10.20         |  |
| Nursing care                             | 338.11                  | 837.77  | 2.48     | 499.67     | 32.72         |  |
| Paramedics & specialist nurses           | 11.71                   | 28.41   | 2.43     | 16.70      | 1.09          |  |
| Physiotherapy                            | 16.01                   | 40.46   | 2.53     | 24.45      | 1.60          |  |
| Surgical interventions                   | 335.63                  | 412.12  | 1.23     | 76.50      | 5.01          |  |
| Consumables used for specific procedures | 11.06                   | 45.85   | 4.15     | 34.80      | 2.28          |  |
| Antimicrobials                           | 11.80                   | 58.82   | 4.99     | 47.02      | 3.08          |  |
| Non-antimicrobial drugs                  | 32.92                   | 109.25  | 3.32     | 76.33      | 5.00          |  |
| Microbiology tests                       | 5.74                    | 21.66   | 3.77     | 15.92      | 1.04          |  |
| Other pathology tests                    | 54.59                   | 81.27   | 1.49     | 26.68      | 1.75          |  |
| Endoscopies                              | 1.71                    | 4.09    | 2.40     | 2.38       | 0.16          |  |
| Radiology                                | 33.35                   | 28.57   | 0.86     | -4.78      | -0.31         |  |
| Other tests                              | 3.53                    | 5.40    | 1.53     | 1.87       | 0.12          |  |
| Total costs                              | 1655.51                 | 3182.49 | 1.92     | 1526.98    | 100.00        |  |

## Table A13.2: The distribution of hospital costs incurred by patients who acquired a surgical wound infection compared to uninfected patients

SWI - surgical wound infection

## Table A13.3: The distribution of hospital costs incurred by patients with a hospital acquired lower respiratory tract infection compared to uninfected patients

|                                          | Mean hospital costs(£) |         |          |            | %                             |
|------------------------------------------|------------------------|---------|----------|------------|-------------------------------|
|                                          | No HAI                 | LRTI    | Ratio of | Additional | contribution<br>to additional |
|                                          | n=2276                 | n=15    | costs    | costs(£)   | costs                         |
| Hospital overheads                       | 392.89                 | 754.16  | 1.92     | 361.27     | 13.58                         |
| Directorate management                   | 47.55                  | 88.09   | 1.85     | 40.54      | 1.52                          |
| Capital charges                          | 208.20                 | 398.99  | 1.92     | 190.79     | 7.17                          |
| Medical time                             | 150.72                 | 312.48  | 2.07     | 161.76     | 6.08                          |
| Nursing care                             | 338.11                 | 1563.98 | 4.63     | 1225.87    | 46.09                         |
| Paramedics & specialist nurses           | 11.71                  | 55.63   | 4.75     | 43.92      | 1.65                          |
| Physiotherapy                            | 16.01                  | 86.31   | 5.39     | 70.31      | 2.64                          |
| Surgical interventions                   | 335.63                 | 538.20  | 1.60     | 202.57     | 7.62                          |
| Consumables used for specific procedures | 11.06                  | 34.67   | 3.14     | 23.61      | 0.89                          |
| Antimicrobials                           | 11.80                  | 93.90   | 7.96     | 82.10      | 3.09                          |
| Non-antimicrobial drugs                  | 32.92                  | 161.75  | 4.91     | 128.83     | 4.84                          |
| Microbiology tests                       | 5.74                   | 30.82   | 5.37     | 25.08      | 0.94                          |
| Other pathology tests                    | 54.59                  | 127.64  | 2.34     | 73.05      | 2.75                          |
| Endoscopies                              | 1.71                   | 0.00    | 0.00     | -1.71      | -0.06                         |
| Radiology                                | 33.35                  | 52.77   | 1.58     | 19.42      | 0.73                          |
| Other tests                              | 3.53                   | 15.69   | 4.44     | 12.16      | 0.46                          |
| Total costs                              | 1655.51                | 4315.07 | 2.61     | 2659.56    | 100.00                        |

LRTI - lower respiratory tract infection

| Table A13.4: | The distribution of hospital costs incurred by patients who |
|--------------|-------------------------------------------------------------|
|              | acquired a skin infection compared to uninfected patients   |

|                                          | Mean hospital costs (£) |                   |          |            |                               |  |
|------------------------------------------|-------------------------|-------------------|----------|------------|-------------------------------|--|
|                                          | No HAI                  | Skin<br>infection | Ratio of | Additional | contribution<br>to additional |  |
|                                          | n=2276                  | n=13              | costs    | costs (£)  | costs                         |  |
| Hospital overheads                       | 392.89                  | 893.77            | 2.27     | 500.88     | 25.73                         |  |
| Directorate management                   | 47.55                   | 121.90            | 2.56     | 74.35      | 3.82                          |  |
| Capital charges                          | 208.20                  | 497.92            | 2.39     | 289.72     | 14.88                         |  |
| Medical time                             | 150.72                  | 385.09            | 2.55     | 234.36     | 12.04                         |  |
| Nursing care                             | 338.11                  | 833.30            | 2.46     | 495.20     | 25.43                         |  |
| Paramedics & specialist nurses           | 11.71                   | 39.62             | 3.38     | 27.90      | 1.43                          |  |
| Physiotherapy                            | 16.01                   | 59.41             | 3.71     | 43.40      | 2.23                          |  |
| Surgical interventions                   | 335.63                  | 478.17            | 1.42     | 142.54     | 7.32                          |  |
| Consumables used for specific procedures | 11.06                   | 105.49            | 9.54     | 94.43      | 4.85                          |  |
| Antimicrobials                           | 11.80                   | 50.17             | 4.25     | 38.38      | 1.97                          |  |
| Non-antimicrobial drugs                  | 32.92                   | 54.92             | 1.67     | 22.00      | 1.13                          |  |
| Microbiology tests                       | 5.74                    | 11.92             | 2.07     | 6.17       | 0.32                          |  |
| Other pathology tests                    | 54.59                   | 45.17             | 0.83     | -9.42      | -0.48                         |  |
| Endoscopies                              | 1.71                    | 0.00              | 0.00     | -1.71      | -0.09                         |  |
| Radiology                                | 33.35                   | 19.73             | 0.59     | -13.63     | -0.70                         |  |
| Other tests                              | 3.53                    | 5.94              | 1.68     | 2.41       | 0.12                          |  |
| Total costs                              | 1655.51                 | 3602.52           | 2.18     | 1947.00    | 100.00                        |  |

## Table A13.5: The distribution of hospital costs incurred by patients who acquired an infection at a site other than surgical wound, bloodstream, skin, or urinary or respiratory tract

|                                          | Mean hospital costs (£) |                                                      |          |            |                                    |
|------------------------------------------|-------------------------|------------------------------------------------------|----------|------------|------------------------------------|
|                                          | No HAI                  | HAI at a site<br>not<br>classified in<br>this study* | Ratio of | Additional | %<br>contribution<br>to additional |
|                                          | n=2276                  | n=18                                                 | costs    | costs      | costs                              |
| Hospital overheads                       | 392.89                  | 962.35                                               | 2.45     | 569.46     | 32.45                              |
| Directorate management                   | 47.55                   | 137.73                                               | 2.90     | 90.18      | 5.14                               |
| Capital charges                          | 208.20                  | 526.88                                               | 2.53     | 318.68     | 18.16                              |
| Medical time                             | 150.72                  | 397.19                                               | 2.64     | 246.46     | 14.04                              |
| Nursing care                             | 338.11                  | 837.59                                               | 2.48     | 499.49     | 28.46                              |
| Paramedics & specialist nurses           | 11.71                   | 30.85                                                | 2.63     | 19.14      | 1.09                               |
| Physiotherapy                            | 16.01                   | 52.58                                                | 3.28     | 36.57      | 2.08                               |
| Surgical interventions                   | 335.63                  | 216.73                                               | 0.65     | -118.90    | -6.77                              |
| Consumables used for specific procedures | 11.06                   | 25.63                                                | 2.32     | 14.57      | 0.83                               |
| Antimicrobials                           | 11.80                   | 28.43                                                | 2.41     | 16.64      | 0.95                               |
| Non-antimicrobial drugs                  | 32.92                   | 80.55                                                | 2.45     | 47.63      | 2.71                               |
| Microbiology tests                       | 5.74                    | 19.46                                                | 3.39     | 13.71      | 0.78                               |
| Other pathology tests                    | 54.59                   | 66.50                                                | 1.22     | 11.91      | 0.68                               |
| Endoscopies                              | 1.71                    | 0.00                                                 | 0.00     | -1.71      | -0.10                              |
| Radiology                                | 33.35                   | 24.16                                                | 0.72     | -9.20      | -0.52                              |
| Other tests                              | 3.53                    | 3.86                                                 | 1.09     | 0.33       | 0.02                               |
| Total costs                              | 1655.51                 | 3410.48                                              | 2.06     | 1754.96    | 100.00                             |

\* single site infection at a site other than the following: urinary tract, surgical wound, skin, lower respiratory, or bloodstream

# Table A13.6 The distribution of hospital costs incurred by patients who acquired a bloodstream infection compared to uninfected patients

|                                          | Mean hospit | al costs (£) |          |            | %<br>contribution |
|------------------------------------------|-------------|--------------|----------|------------|-------------------|
|                                          | No HAI      | BSI          | Ratio of | Additional |                   |
|                                          | n=2276      | n=3          | costs    | costs (£)  | costs             |
| Hospital overheads                       | 392.89      | 411.54       | 1.05     | 18.65      | 0.26              |
| Directorate management                   | 47.55       | 65.85        | 1.38     | 18.30      | 0.25              |
| Capital charges                          | 208.20      | 223.98       | 1.08     | 15.78      | 0.22              |
| Medical time                             | 150.72      | 180.52       | 1.20     | 29.79      | 0.41              |
| Nursing care                             | 338.11      | 4715.04      | 13.95    | 4376.93    | 59.98             |
| Paramedics & specialist nurses           | 11.71       | 0.00         | 0.00     | -11.71     | -0.16             |
| Physiotherapy                            | 16.01       | 83.28        | 5.20     | 67.28      | 0.92              |
| Surgical interventions                   | 335.63      | 849.06       | 2.53     | 513.44     | 7.04              |
| Consumables used for specific procedures | 11.06       | 785.73       | 71.06    | 774.67     | 10.62             |
| Antimicrobials                           | 11.80       | 120.31       | 10.20    | 108.51     | 1.49              |
| Non-antimicrobial drugs                  | 32.92       | 1017.26      | 30.90    | 984.34     | 13.49             |
| Microbiology tests                       | 5.74        | 110.14       | 19.18    | 104.40     | 1.43              |
| Other pathology tests                    | 54.59       | 247.40       | 4.53     | 192.81     | 2.64              |
| Endoscopies                              | 1.71        | 0.00         | 0.00     | -1.71      | -0.02             |
| Radiology                                | 33.35       | 130.06       | 3.90     | 96.71      | 1.33              |
| Other tests                              | 3.53        | 13.13        | 3.72     | 9.60       | 0.13              |
| Total costs                              | 1655.51     | 8953.31      | 5.41     | 7297.80    | 100.00            |

BSI - bloodstream infection

# Table A13.7: The distribution of hospital costs incurred by patients who acquired more than one infection compared to uninfected patients

|                                          | Mean hosp | ital costs (£)         |          |            | 94                            |
|------------------------------------------|-----------|------------------------|----------|------------|-------------------------------|
|                                          | No HAI    | Multiple<br>infections | Ratio of | Additional | contribution<br>to additional |
|                                          | n=2276    | n=24                   | costs    | costs (£)  | costs                         |
| Hospital overheads                       | 392.89    | 1617.61                | 4.12     | 1224.72    | 15.09                         |
| Directorate management                   | 47.55     | 188.92                 | 3.97     | 141.37     | 1.74                          |
| Capital charges                          | 208.20    | 792.85                 | 3.81     | 584.65     | 7.20                          |
| Medical time                             | 150.72    | 668.45                 | 4.43     | 517.73     | 6.38                          |
| Nursing care                             | 338.11    | 4107.03                | 12.15    | 3768.92    | 46.43                         |
| Paramedics & specialist nurses           | 11.71     | 216.26                 | 18.46    | 204.55     | 2.52                          |
| Physiotherapy                            | 16.01     | 147.64                 | 9.22     | 131.63     | 1.62                          |
| Surgical interventions                   | 335.63    | 486.59                 | 1.45     | 150.96     | 1.86                          |
| Consumables used for specific procedures | 11.06     | 270.60                 | 24.47    | 259.54     | 3.20                          |
| Antimicrobials                           | 11.80     | 141.06                 | 11.96    | 129.26     | 1.59                          |
| Non-antimicrobial drugs                  | 32.92     | 323.83                 | 9.84     | 290.91     | 3.58                          |
| Microbiology tests                       | 5.74      | 78.25                  | 13.62    | 72.50      | 0.89                          |
| Other pathology tests                    | 54.59     | 212.52                 | 3.89     | 157.93     | 1.95                          |
| Endoscopies                              | 1.71      | 7.12                   | 4.18     | 5.41       | 0.07                          |
| Radiology                                | 33.35     | 500.28                 | 15.00    | 466.93     | 5.75                          |
| Other tests                              | 3.53      | 14.72                  | 4.17     | 11.18      | 0.14                          |
| Total costs                              | 1655.51   | 9773.73                | 5.90     | 8118.22    | 100.00                        |

Results of the analysis that assessed how length of hospital stay varied with site of infection and selected patient characteristics costs and site of HAI

## A14.1 Introduction

The results of the analysis that assessed how length of hospital stay varied with site of infection and selected patient characteristics are presented in this appendix.

The results presented in tables A14.1-A14.7 show that whilst for all sites of infection the mean length of stay was higher in infected than uninfected patients for all the patient characteristics listed, the level of increase varied with type of infection. However, it should be noted that in some cases the number of patients in the infected groups were small and as such the results cannot be generalized.

# Table A14.1 Mean length of stay for patients with a urinary tract infection and uninfected patients by key patient characteristics

|                                    | Mean length of hospital stay (days) |      |       |     | 61. <sup>9</sup>         |                   |  |
|------------------------------------|-------------------------------------|------|-------|-----|--------------------------|-------------------|--|
|                                    | No                                  | HAI  | UTI o | nly | Ratio of LOS<br>(95% CI) | Additional days   |  |
| Patient characteristic             | Mean                                | n    | Mean  | n   | (00,00,0)                | (0010 01)         |  |
|                                    | (a)                                 |      | (b)   |     | (b/a)                    | (b-a)             |  |
| Sex                                |                                     |      |       |     |                          |                   |  |
| Male                               | 5.9                                 | 946  | 12.4  | 18  | 2.1(1.0, 3.2)            | 6.5 (3.7, 9.3)    |  |
| Female                             | 7.1                                 | 1330 | 11.2  | 70  | 1.6 (1.3, 1.9)           | 4.1 (2.3, 5.8)    |  |
|                                    |                                     |      |       |     |                          |                   |  |
| 18-34                              | 6.0                                 | 485  | 8.2   | 15  | 1.4 (1.0, 1.7)           | 2.2 (-1.7, 6.1)   |  |
| 35-54                              | 5.5                                 | 627  | 8.2   | 26  | 1.5 (1.3, 1.7)           | 2.7 (1.2, 4.3)    |  |
| 55-74                              | 6.8                                 | 870  | 12.0  | 32  | 1.8 (1.2. 2.3)           | 5.2 (2.8, 7.5)    |  |
| 75+                                | 9.4                                 | 294  | 19.2  | 15  | 2.1 (1.2. 2.9)           | 9.8 (4.9, 14.7)   |  |
| Specialty                          | 1                                   |      |       |     |                          |                   |  |
| General surgery                    | 54                                  | 829  | 18.7  | 10  | 3.5 (1.8, 5.2)           | 13.3 (8.9, 17.7)  |  |
| Orthonaedics                       | 10.3                                | 462  | 24.6  | 12  | 24(1.5.3.3)              | 14.3 (9.4, 19.2)  |  |
| Licology                           | 51                                  | 445  | 7.8   | 17  | 1.5 (1.1, 1.9)           | 2.6 (0.5, 4.8)    |  |
| Currencelory                       | 61                                  | 336  | 8.0   | 36  | 13(1115)                 | 19(06.3.1)        |  |
| Obstatzias                         | 7.5                                 | 204  | 8.2   | 13  | 11(0814)                 | 08(-24 39)        |  |
|                                    | 1.5                                 | 204  | 0.2   | 10  | 1.1 (0.0, 1.4)           | 0.0 ( 2.4, 0.0)   |  |
| Admission type                     | 60                                  | 1510 |       | 5.9 | 12/12 15)                | 20/04 35)         |  |
| Elective                           | 7.0                                 | 757  | 10.0  | 20  | 23/1730)                 | 10 4 (7 4 13 3)   |  |
| Emergency                          | 7.9                                 | 15/  | 10.2  |     | 2.3 (1.7, 3.0)           | 10.4 (7.4, 13.3)  |  |
| Discharge diagnosis group          | 2.0                                 | 12   | 2.0   |     | 0.8*                     | 0.8*              |  |
| Infectious & parasitic diseases    | 3.6                                 | 12   | 3.0   |     | 0.0                      | -0.0              |  |
| Endocrine nutritional & metabolic  | 6.7                                 | 329  | 10.7  | 14  | 1.6 (1.0, 2.2)           | 4.1 (0.4, 7.7)    |  |
| diseases & immunity disorders      | 4.9                                 | 20   | 0.0   | 0   |                          |                   |  |
| Diseases of blood & blood forming  | 43                                  | 3    | 0.0   | 0   |                          |                   |  |
| Diseases of the nervous system &   | 4.5                                 | 3    | 0.0   | 0   |                          |                   |  |
| sense organs                       | 3.5                                 | 2    | 0.0   | 0   |                          |                   |  |
| Diseases of the circulatory system | 5.0                                 | 142  | 45.0  | 1   | 9.1*                     | 40.0*             |  |
| Diseases of the respiratory system | 7.0                                 | 4    | 0.0   | 0   | -                        | -                 |  |
| Diseases of the digestive system   | 4.9                                 | 396  | 10.1  | 7   | 2.1 (1.3, 2.8)           | 5.3 (1.6, 9.0)    |  |
| Diseases of the genitourinary      | 54                                  | 510  | 8.0   | 34  | 15(1317)                 | 27 (15 38)        |  |
| Complications of pregnancy,        | 0.4                                 | 010  | 0.0   |     | 1.0 (1.0, 1.1)           | 2.7 (1.0, 0.0)    |  |
| childbirth & puerperium & certain  |                                     |      |       |     |                          |                   |  |
| perinatal period                   | 7.3                                 | 230  | 8.0   | 15  | 1.1 (0.9, 1.3)           | 0.7 (-2.2, 3.7)   |  |
| Diseases of the skin &             |                                     |      |       |     |                          |                   |  |
| Subcutaneous tissue                | 4.5                                 | 28   | 0.0   | 0   | 0.0                      |                   |  |
| system & connective tissue         | 10.5                                | 296  | 15.0  | 3   | 1.4 (0.8, 2.0)           | 4.5 (-4.3, 13.3)  |  |
| Injury & Poisoning                 | 9.9                                 | 160  | 27.8  | 9   | 2.8 (1.6, 4.0_           | 17.9 (0.6, 13.9)  |  |
| Symptoms, signs & ill- defined     | 1.5                                 |      |       |     |                          |                   |  |
| congenital abnormalities           | 5.8                                 | 135  | 13.0  | 4   | 2.3 (→0.0, 5.2)          | 7.2 (0.6, 13.9)   |  |
| Number of co-morbidities           |                                     |      |       |     |                          |                   |  |
| None                               | 6.2                                 | 1532 | 9.0   | 56  | 1.5 (1.2, 1.7)           | 2.8 (1.3, 4.3)    |  |
| One                                | 7.0                                 | 528  | 12.4  | 24  | 1.8 (1.2, 2.4)           | 5.5 (1.9, 9.0)    |  |
| Two                                | 87                                  | 216  | 25.8  | 8   | 2.9 (1.4, 4.5)           | 17.0 (10.9. 23.2) |  |

UTI – urinary tract infection; LOS – length of stay, \*Numerator n=1 consequently standard deviation not estimable so confidence limits could not be estimated.

|                                                                                                                       | Mean I | ength of h | ospital stay ( | days) |                  |                  |  |
|-----------------------------------------------------------------------------------------------------------------------|--------|------------|----------------|-------|------------------|------------------|--|
|                                                                                                                       | No HAI |            | LRTI o         | nly   | Ratio of LOS     | Additional days  |  |
| Patient characteristic                                                                                                | Mean   | n          | Mean           | n     | (0010 01)        | (0070 0.4        |  |
|                                                                                                                       | (a)    |            | (b)            |       | (b/a)            | (b-a)            |  |
| Sex                                                                                                                   |        |            |                |       |                  |                  |  |
| Male                                                                                                                  | 5.9    | 946        | 16.8           | 9     | 2.8 (1.6, 4.1)   | 10.9 (7.1, 14.7) |  |
| Female                                                                                                                | 7.1    | 1330       | 11.2           | 6     | 1.6 (1.0, 2.1)   | 4.0 (-1.7, 9.8)  |  |
| Age group                                                                                                             |        |            |                |       |                  |                  |  |
| 18 –34                                                                                                                | 6.0    | 485        | 17.0           | 1     | 2.8*             | 11.0*            |  |
| 35-54                                                                                                                 | 5.5    | 627        | 9.5            | 2     | 1.7 (0.3, 5.3    | 4.0 (-1.6, 9.6)  |  |
| 55-74                                                                                                                 | 6.8    | 870        | 13.7           | 7     | 2.0 (1.5, 2.5)   | 6.9 (2.1, 11.6)  |  |
| 75+                                                                                                                   | 9.4    | 294        | 17.2           | 5     | 1.8 (0.3, 3.4)   | 7.8 (-0.2, 15.9) |  |
| Specialty                                                                                                             |        |            |                |       |                  |                  |  |
| General surgery                                                                                                       | 5.4    | 829        | 15.9           | 10    | 3.0 (1.7, 4.2)   | 10.5 (6.1, 14.9) |  |
| Orthopaedics                                                                                                          | 10.3   | 462        | 14.0           | 2     | 1.4 (1.1, 1.6)   | 3.7 (-7.7, 15.2) |  |
| Urology                                                                                                               | 5.1    | 445        | 5.0            | 1     | 1.0*             | -0.1*            |  |
| Gynaecology                                                                                                           | 6.1    | 336        | 9.0            | 1     | 1.5*             | 2.9*             |  |
| Obstetrics                                                                                                            | 7.5    | 204        | 17.0           | 1     | 2.3*             | 9.5*             |  |
| Admission type                                                                                                        |        |            |                |       |                  |                  |  |
| Elective                                                                                                              | 6.0    | 1519       | 16.9           | 8     | 2.8 (1.3, 4.2)   | 10.9 (6.7, 15.0) |  |
| Emergency                                                                                                             | 7.9    | 757        | 11.9           | 7     | 1.5 (1.1, 1.9)   | 4.0 (-1.7, 9.7)  |  |
| Discharge diagnosis group                                                                                             |        |            |                |       |                  |                  |  |
| Infectious & parasitic diseases                                                                                       | 3.8    | 12         | 0.0            | 0     |                  | -                |  |
| Neoplasms                                                                                                             | 6.7    | 329        | 17.0           | 3     | 2.6 (1.4, 3.7)   | 10.3 (2.7, 18.0) |  |
| Endocrine, nutritional & metabolic<br>diseases & immunity disorders                                                   | 4.9    | 20         | 13.0           | 1     | 2.7*             | 8.2*             |  |
| Diseases of blood & blood forming                                                                                     | 4.3    | 3          | 0.0            | 0     |                  |                  |  |
| Diseases of the nervous system & sense organs                                                                         | 3.5    | 2          | 0.0            | 0     |                  |                  |  |
| Diseases of the circulatory system                                                                                    | 5.0    | 142        | 0.0            | 0     |                  |                  |  |
| Diseases of the respiratory system                                                                                    | 7.0    | 4          | 15.0           | 1     | 2.1*             | 8.0*             |  |
| Diseases of the digestive system                                                                                      | 4.9    | 396        | 8.8            | 4     | 1.8 (1.1, 2.5)   | 3.9 (-1.0, 8.7)  |  |
| Diseases of the genitourinary system                                                                                  | 5.4    | 519        | 9.0            | 1     | 1.7*             | 3.6*             |  |
| Complications of pregnancy,<br>childbirth & puerperium & certain<br>conditions originating in the perinatal<br>period | 7.3    | 230        | 17.0           | 1     | 2.3*             | 9.7*             |  |
| Diseases of the skin & subcutaneous tissue                                                                            | 4.5    | 28         | 0.0            | 0     |                  |                  |  |
| Diseases of the musculoskletal<br>system & connective tissue                                                          | 10.5   | 296        | 14.0           | 2     | 1.3 (1.1, 1.6)   | 3.5 (-7.2, 14.2_ |  |
| Injury & Poisoning                                                                                                    | 9.9    | 160        | 0.0            | 0     |                  |                  |  |
| Symptoms, signs & ill- defined<br>conditions; mental disorders &<br>congenital abnormalities                          | 5.8    | 135        | 25.0           | 2     | 4.3 (→0.0, 11.2) | 19.2 (9.9, 28.6) |  |
| Number of co-morbidities                                                                                              |        |            |                |       |                  |                  |  |
| None                                                                                                                  | 6.2    | 1532       | 10.6           | 5     | 1.7 (1.1, 2.4)   | 4.4 (-0.4, 9.2)  |  |
| One                                                                                                                   | 7.0    | 528        | 25.7           | 3     | 3.7 (0.9, 6.4)   | 18.7 (8.9, 28.6) |  |

# Table A14.2 Mean length of stay for patients with a lower respiratory tract infection patients and uninfected patients by key patient characteristics

LRTI - lower respiratory tract infection, LOS - length of stay, \*numerator n=1 consequently standard deviation not estimable so confidence limits could not be estimated.

216

8.7

Two

3.8 (-2.3, 9.9)

1.4 (1.0, 1.9)

7

12.6

|                                                      | Mean length of hospital stay (days) |      |        |       |                          | Additional days   |  |
|------------------------------------------------------|-------------------------------------|------|--------|-------|--------------------------|-------------------|--|
|                                                      | No                                  | HAI  | sw     | lonly | Ratio of LOS<br>(95% CI) | (95% Cl)          |  |
| Patient characteristic                               | Mean                                | n    | Mean   | n     |                          | (,                |  |
|                                                      | (a)                                 |      | (b)    |       | (b/a)                    | (b-a)             |  |
| Sex                                                  |                                     |      |        |       |                          |                   |  |
| Male                                                 | 5.9                                 | 946  | 19.33  | 15    | 3.3 (2.2, 4.4)           | 13.4 (10.4, 16.4) |  |
| Female                                               | 7.1                                 | 1330 | 10.41  | 17    | 1.5 (1.0, 1.9)           | 3.3 (-0.2, 6.7)   |  |
|                                                      |                                     |      |        |       |                          |                   |  |
| 18 - 34                                              | 6.0                                 | 485  | 13.14  | 7     | 2.2(0.9, 3.5)            | 7.1 (1.4, 12.8)   |  |
| 35-54                                                | 5.5                                 | 627  | 8.25   | 4     | 1.5 90.9, 2.1)           | 2.7 (-1.2, 6.7)   |  |
| 55-74                                                | 6.8                                 | 870  | 14.38  | 16    | 2.1 (1.3, 2.9)           | 7.5 (4.3, 10.7)   |  |
| 75+                                                  | 9.4                                 | 294  | 22.40  | 5     | 2.4 (1.2, 3.6)           | 13.0 (5.0, 21.0)  |  |
| Specialty                                            |                                     |      |        |       |                          |                   |  |
| General surgery                                      | 5.4                                 | 829  | 16.67  | 12    | 3.1 (1.8, 4.4)           | 11.3 (7.2, 15.3)  |  |
| Orthopaedics                                         | 10.3                                | 462  | 16.27  | 11    | 1.6 (1.0, 2.2)           | 6.0 (1.0, 11.0)   |  |
| Urology                                              | 5.1                                 | 445  | 5.00   | 1     | 1.0*                     | -0.1*             |  |
| Gynaecology                                          | 6.1                                 | 336  | 5.25   | 4     | 0.9 (0.5, 1.2)           | -0.9 (-4.5, 2.7)  |  |
| Obstatrics                                           | 7.5                                 | 204  | 8.00   | 3     | 1.1 (0.6, 1.5)           | 0.5 (6.0, 7.1)    |  |
| Admission type                                       | 1.0                                 |      |        |       |                          |                   |  |
| Elective                                             | 60                                  | 1519 | 12.69  | 16    | 2.1(1.5, 2.7)            | 6.7 (3.7, 9.6)    |  |
| Emergency                                            | 7.9                                 | 757  | 16.50  | 16    | 2.1(1.2.3.0)             | 8.6 (4.7, 12.5)   |  |
|                                                      | 1.0                                 |      | 1.0.00 | 1     |                          |                   |  |
| Infectious & parasitic diseases                      | 3.8                                 | 12   | 0.00   | 0     |                          |                   |  |
| Neonlasme                                            | 6.7                                 | 329  | 21.00  | 3     | 3.2 (1.9, 4.4)           | 14.3 (6.6, 27.0)  |  |
| Endocrine, nutritional & metabolic                   |                                     |      |        |       |                          |                   |  |
| diseases & immunity disorders                        | 4.9                                 | 20   | 4.00   | 1     | 0.8*                     | -0.9*             |  |
| Diseases of blood & blood forming<br>organs          | 4.3                                 | 3    | 0.00   | 0     |                          |                   |  |
| Diseases of the nervous system &                     |                                     |      | 0.00   |       |                          |                   |  |
| sense organs                                         | 3.5                                 | 2    | 0.00   | 0     |                          | 40.01             |  |
| Diseases of the circulatory system                   | 5.0                                 | 142  | 17.00  | 1     | 3.4"                     | 12.0*             |  |
| Diseases of the respiratory system                   | 7.0                                 | 4    | 0.00   | 0     |                          |                   |  |
| Diseases of the digestive system                     | 4.9                                 | 396  | 13.75  | 4     | 2.8 (0.7, 4.9)           | 8.9 (3.9, 13.8)   |  |
| system                                               | 5.4                                 | 519  | 18.50  | 2     | 3.4 (→0.0, 7.6)          | 13.1 (8.4, 17.9)  |  |
| Complications of pregnancy,                          |                                     |      |        |       |                          |                   |  |
| conditions originating in the                        |                                     |      |        |       |                          |                   |  |
| perinatal period                                     | 7.3                                 | 230  | 8.00   | 3     | 1.1 (0.6, 1.6)           | 0.7 (-5.9, 7.3)   |  |
| Diseases of the skin & subcutaneous tissue           | 4.5                                 | 28   | 8.00   | 1     | 1.8*                     | 3.5*              |  |
| Diseases of the musculoskletal                       |                                     |      |        | _     |                          |                   |  |
| system & connective tissue                           | 10.5                                | 296  | 14.00  | 5     | 1.3 (0.9, 1.7)           | 3.5 (-3.3, 10.3)  |  |
| Injury & Poisoning<br>Symptoms, signs & ill, defined | 9.9                                 | 160  | 13.70  | 10    | 1.4 (0.6, 2.2)           | 3.8 (-2.2, 9.8)   |  |
| conditions; mental disorders &                       |                                     |      |        |       |                          |                   |  |
| congenital abnormalities                             | 5.8                                 | 135  | 26.00  | 2     | 4.5 (→0.0, 11.0)         | 20.2 (10.9, 29.5) |  |
| Number of co-morbidities                             |                                     |      |        |       |                          |                   |  |
| None                                                 | 6.2                                 | 1532 | 11.00  | 18    | 1.8 (1.1, 2.5)           | 4.8 (2.2, 7.4)    |  |
| One                                                  | 7.0                                 | 528  | 18.00  | 10    | 2.6 (1.4, 3.7)           | 11.0 (5.6, 16.5)  |  |
| Two                                                  | 8.7                                 | 216  | 22.25  | 4     | 2.5 (1.6, 3.5)           | 13.5 (5.4, 21.6)  |  |

# Table A14.3: Mean length of stay for patients with a surgical wound infection and uninfected patients by key patient characteristics

SWI – surgical wound infection, LOS – length of stay, \*numerator n=1 consequently standard deviation not estimable so confidence limits could not be estimated.

# Table A14.4 Mean length of stay for patients with a blood stream infection and uninfected patients by key patient characteristics

|                                                                    | Mean length of hospital st |      | spital stay | (days) |                          |                             |  |
|--------------------------------------------------------------------|----------------------------|------|-------------|--------|--------------------------|-----------------------------|--|
|                                                                    | No                         | HAI  | BSI         | only   | Ratio of LOS<br>(95% Cl) | Additional days<br>(95% Cl) |  |
| Patient characteristic                                             | Mean                       | n    | Mean        | n      | (                        | (0000 00)                   |  |
|                                                                    | (a)                        |      | (b)         |        | (b/a)                    | (b-a)                       |  |
| Sex                                                                |                            |      |             |        |                          |                             |  |
| Male                                                               | 5.9                        | 946  | 8.7         | 3      | 1.5 (0.8, 2.1)           | 2.8 (-3.7, 9.2)             |  |
| Female                                                             | 7.1                        | 1330 | 0.0         | 0      |                          |                             |  |
|                                                                    |                            | 1    |             |        |                          |                             |  |
| 18 - 34                                                            | 6.0                        | 485  | 12.0        | 1      | 2.0*                     | 6.0*                        |  |
| 25.54                                                              | 5.5                        | 627  | 5.0         | 1      | 0.9*                     | -0.5*                       |  |
| 55-54                                                              | 6.8                        | 870  | 9.0         | 1      | 1.3*                     | 2.2*                        |  |
| 75+                                                                | 0.0                        | 204  | 0.0         | 0      | -                        |                             |  |
| /5+                                                                | 9.4                        | 294  | 0.0         |        |                          |                             |  |
| Specialty                                                          | 5.4                        | 000  | 60          | -      | 0.0*                     | 0.4*                        |  |
| General surgery                                                    | 5.4                        | 829  | 5.0         | -      | 1.0*                     | 4.7*                        |  |
| Orthopaedics                                                       | 10.3                       | 462  | 12.0        | 1      | 1.2                      | 1.7                         |  |
| Urology                                                            | 5.1                        | 445  | 9.0         | 1      | 1.8-                     | 3.9                         |  |
| Gynaecology                                                        | 6.1                        | 336  | 0.0         | 0      |                          |                             |  |
| Obstetrics                                                         | 7.5                        | 204  | 0.0         | 0      |                          |                             |  |
| Admission type                                                     |                            |      |             |        |                          |                             |  |
| Elective                                                           | 6.0                        | 1519 | 9.0         | 1      | 1.5*                     | 3.0*                        |  |
| Emergency                                                          | 7.9                        | 757  | 8.5         | 2      | 1.1 (0.3, 2.6)           | 0.6 (-5.8, 10.8)            |  |
| Discharge diagnosis group                                          |                            |      |             |        |                          |                             |  |
| Infectious & parasitic diseases                                    | 3.8                        | 12   | 0.0         | 0      |                          |                             |  |
| Neoplasms                                                          | 6.7                        | 329  | 0.0         | 0      |                          |                             |  |
| Endocrine, nutritional & metabolic                                 | 10                         | 20   | 0.0         | 0      |                          |                             |  |
| diseases & immunity disorders                                      | 4.9                        | 20   | 0.0         | 0      |                          |                             |  |
| organs                                                             | 4.3                        | 3    | 0.0         | 0      |                          |                             |  |
| Diseases of the nervous system &                                   | 2.5                        | 2    | 0.0         |        |                          |                             |  |
| sense organs                                                       | 3.5                        | 2    | 0.0         | 0      |                          |                             |  |
| Diseases of the circulatory system                                 | 5.0                        | 142  | 0.0         | 0      | 4.74                     | E 0*                        |  |
| Diseases of the respiratory system                                 | 7.0                        | 4    | 12.0        | 1      | 1.7-                     | 5.0-                        |  |
| Diseases of the digestive system                                   | 4.9                        | 396  | 5.0         | 1      | 1.0*                     | 0.1*                        |  |
| system                                                             | 5.4                        | 519  | 9.0         | 1      | 1.7*                     | 3.6*                        |  |
| Complications of pregnancy,                                        |                            |      |             |        |                          |                             |  |
| childbirth & puerperium & certain<br>conditions originating in the |                            |      |             |        |                          |                             |  |
| perinatal period                                                   | 7.3                        | 230  | 0.0         | 0      |                          |                             |  |
| Diseases of the skin &                                             | 4.5                        | 28   | 0.0         | 0      |                          |                             |  |
| Diseases of the musculoskletal                                     | 4.5                        | 20   | 0.0         |        |                          |                             |  |
| system & connective tissue                                         | 10.5                       | 296  | 0.0         | 0      |                          |                             |  |
| Injury & Poisoning                                                 | 9.9                        | 160  | 0.0         | 0      |                          |                             |  |
| Symptoms, signs & ill- defined                                     |                            |      |             |        |                          |                             |  |
| congenital abnormalities                                           | 5.8                        | 135  | 0.0         | 0      |                          |                             |  |
| Number of co-morbidities                                           |                            |      |             |        |                          |                             |  |
| None                                                               | 6.2                        | 1532 | 9.0         | 1      | 1.4*                     | 2.8*                        |  |
| One                                                                | 7.0                        | 528  | 0.0         | 0      |                          |                             |  |
| Two                                                                | 87                         | 216  | 8.5         | 2      | 1.0 (0.2, 1.8)           | -0.2 (-11.6, 11.1           |  |

BSI - bloodstream infection, LOS - length of stay, \*numerator n=1 consequently standard deviation not estimable so confidence limits could not be estimated.

|                                                                                                | Mean le | ngth of h | ospital stay |       |                |                             |  |
|------------------------------------------------------------------------------------------------|---------|-----------|--------------|-------|----------------|-----------------------------|--|
|                                                                                                | No      | HAI       | Skir         | only  | Ratio of LOS   | Additional days<br>(95% Cl) |  |
| Patient characteristic                                                                         | Mean    | n         | Mean         | n     | (00% 01)       | (0070 01)                   |  |
|                                                                                                | (a)     |           | (b)          |       | (b/a)          | (b-a)                       |  |
| Sex                                                                                            |         |           |              |       |                |                             |  |
| Male                                                                                           | 5.9     | 946       | 18.4         | 5     | 3.1 (1.5, 4.7) | 12.5 (7.5, 17.5)            |  |
| Female                                                                                         | 7.1     | 1330      | 19.1         | 8     | 2.7 (1.4, 4.0) | 12.0 (6.9, 17.0)            |  |
| Age group                                                                                      |         |           |              |       |                |                             |  |
| 18-34                                                                                          | 6.0     | 485       | 7.0          | 1     | 1.2*           | 1.0                         |  |
| 35-54                                                                                          | 5.5     | 627       | 11.3         | 3     | 2.1 (0.3, 3.9) | 5.8 (1.2, 10.4)             |  |
| 55-74                                                                                          | 6.8     | 870       | 22.0         | 7     | 3.2 (1.9, 4.5) | 15.2 (10.3, 20.0)           |  |
| 75+                                                                                            | 9.4     | 294       | 25.0         | 2     | 2.7 (0.6, 4.8) | 15.6 (3.1, 28.2)            |  |
| Specialty                                                                                      |         |           |              |       |                |                             |  |
| General surgery                                                                                | 5.4     | 829       | 17.8         | 5     | 3.3 (1.6, 5.0) | 12.4 (6.2, 18.6)            |  |
| Orthopaedics                                                                                   | 10.3    | 462       | 20.4         | 5     | 2.0 (0.9, 3.0) | 10.1 (2.8, 17.4)            |  |
| Urology                                                                                        | 5.1     | 445       | 38.0         | 1     | 7.4*           | 32.9*                       |  |
| Gynaecology                                                                                    | 6.1     | 336       | 0.0          | 0     |                |                             |  |
| Obstatrics                                                                                     | 7.5     | 204       | 8.0          | 2     | 1.1(1.2, 2.0)  | 0.5 (-3.4, 9.2)             |  |
| Admission type                                                                                 | 1.0     |           |              |       | 1              |                             |  |
| Elective                                                                                       | 6.0     | 1519      | 18.0         | 7     | 3.0 (1.5, 4.5) | 12.0 (7.5, 16.4)            |  |
| Emergency                                                                                      | 7.9     | 757       | 19.8         | 6     | 2.5 (1.2, 3.8) | 12.0 (5.7, 18.2)            |  |
|                                                                                                | 1.0     |           |              | -     |                |                             |  |
| Infectious & parasitic diseases                                                                | 3.8     | 12        | 0.0          | 0     |                |                             |  |
| Mechous a parasite diseases                                                                    | 67      | 329       | 21.0         | 1     | 3.2*           | 14.3*                       |  |
| Endocrine, nutritional & metabolic                                                             | 0.1     |           |              | · · · |                |                             |  |
| diseases & immunity disorders                                                                  | 4.9     | 20        | 4.0          | 1     | 0.8*           | -0.9*                       |  |
| Diseases of blood & blood forming<br>organs                                                    | 4.3     | 3         | 0.0          | 0     | -              | -                           |  |
| Diseases of the nervous system &                                                               |         |           |              |       |                |                             |  |
| sense organs                                                                                   | 3.5     | 2         | 0.0          | 0     |                |                             |  |
| Diseases of the circulatory system                                                             | 5.0     | 142       | 15.0         | 1     | 3.0*           | 10.0*                       |  |
| Diseases of the respiratory system                                                             | 7.0     | 4         | 0.0          | 0     |                |                             |  |
| Diseases of the digestive system                                                               | 4.9     | 396       | 33.0         | 1     | 6.8*           | 28.1*                       |  |
| System                                                                                         | 5.4     | 519       | 27.0         | 2     | 5.0 (1.0, 9.1) | 21.6 (16.9, 26.4)           |  |
| Complications of pregnancy,                                                                    |         | 4.8       |              |       |                |                             |  |
| childbirth & puerperium & certain<br>conditions originating in the                             |         | 1.1.1     | 1981.0       |       |                |                             |  |
| perinatal period                                                                               | 7.3     | 230       | 8.0          | 2     | 1.1 (0.8, 1.4) | 0.7 (-7.3, 8.8)             |  |
| Diseases of the skin &                                                                         | 4.5     | 28        | 6.0          | 1     | 1.3*           | 1.5*                        |  |
| Diseases of the musculoskletal                                                                 | 4.0     |           |              |       |                |                             |  |
| system & connective tissue                                                                     | 10.5    | 296       | 24.0         | 2     | 2.3 (0.2, 4.3) | 13.5 (2.7, 24.3)            |  |
| Injury & Poisoning                                                                             | 9.9     | 160       | 24.0         | 2     | 2.4 (1.0, 3.9) | 14.1 (1.2, 27.0)            |  |
| conditions; signs & iii- defined<br>conditions; mental disorders &<br>congenital abnormalities | 5.8     | 135       | 0.0          | 0     |                |                             |  |
| Number of comorbidities                                                                        |         |           |              |       |                |                             |  |
| None                                                                                           |         | 4800      |              | -     | 16/07 20       | 38/16 03                    |  |
|                                                                                                | 6.2     | 1532      | 10.0         | 4     | 25 (10.20)     | 3.0 (-1.0, 9.2)             |  |
| Ture                                                                                           | 7.0     | 528       | 17.3         | -     | 2.0 (1.0, 3.9) | 10.5 (1.6, 16.8)            |  |
| IWO                                                                                            | 8.7     | 216       | 27.2         | 5     | 3.1(1.0, 4.4)  | 18.5 (11.1, 25.8)           |  |

# Table A14.5Mean length of stay for patients with a skin infection and<br/>uninfected patients by key patient characteristics

LOS - length of stay, \*numerator n=1 consequently standard deviation not estimable so confidence limits could not be estimated.

# Table A14.6: Mean length of stay for uninfected patients and patients with a single infection at a site other than the urinary tract, bloodstream, surgical wound, skin or lower respiratory tract by key patient characteristics

|                                                                                                                    | Mean le | ngth o<br>(da | f hospita<br>ys) | l stay | lingelik<br>Al 1971      |                   |  |
|--------------------------------------------------------------------------------------------------------------------|---------|---------------|------------------|--------|--------------------------|-------------------|--|
| Patient characteristic                                                                                             | No H    | AI            | HA               | ۴      | Ratio of LOS<br>(95% Cl) | Additional days   |  |
| r auent characteristic                                                                                             | Mean    | n             | Mean             | n      | (00% 0.)                 | (95% CI)          |  |
|                                                                                                                    | (a)     |               | (b)              |        | (b/a)                    | (b-a)             |  |
| Sex                                                                                                                |         |               |                  |        |                          |                   |  |
| Male                                                                                                               | 5.9     | 946           | 33.7             | 3      | 5.7 (→, 14.7)            | 27.8 (20.9, 34.7) |  |
| Female                                                                                                             | 7.1     | 133<br>0      | 17.2             | 15     | 2.4 (0.7, 4.1)           | 10.1 (6.2, 13.9)  |  |
| Age group                                                                                                          |         |               |                  |        |                          |                   |  |
| 18-34                                                                                                              | 6.0     | 485           | 36.3             | 3      | 6.0 (→0.0, 15.9)         | 30.3 (20.9, 39.7) |  |
| 35-54                                                                                                              | 5.5     | 627           | 12.2             | 5      | 2.2 (0.4, 4.0)           | 6.7 (3.1, 10.3)   |  |
| 55-74                                                                                                              | 6.8     | 870           | 22.4             | 7      | 3.3 (0.6, 6.0)           | 15.6 (10.6, 20.6) |  |
| 75+                                                                                                                | 9.4     | 294           | 10.7             | 3      | 1.1(0.0, 2.2)            | 1.3 (-8.9, 11.5)  |  |
| Specialty                                                                                                          |         | 1.1           |                  |        |                          |                   |  |
| General surgery                                                                                                    | 5.4     | 829           | 14.0             | 6      | 2.6 (1.0, 4.2)           | 8.6 (3.0, 14.2)   |  |
| Orthonaedics                                                                                                       | 10.3    | 462           | 35.8             | 4      | 3.5 (-→0.0, 7.4)         | 25.5 (16.7, 34.2) |  |
|                                                                                                                    | 5.1     | 445           | 33.0             | 3      | 6.4 (-→0.0,<br>17.0)     | 27.9 (21.6, 34.2) |  |
| Gynaecology                                                                                                        | 6.1     | 336           | 7.5              | 2      | 1.2 (0.7, 1.7)           | 1.4 (-3.7, 6.4)   |  |
| Obstatrics                                                                                                         | 7.5     | 204           | 6.0              | 3      | 0.8 (-→0.0, 1.7)         | -1.5 (-8.0, 5.1)  |  |
| Admission type                                                                                                     |         |               |                  |        |                          |                   |  |
| Elective                                                                                                           | 6.0     | 151<br>9      | 11.0             | 6      | 1.8 (0.4, 3.2)           | 5.0 (0.2, 9.8)    |  |
| Emergency                                                                                                          | 7.9     | 757           | 24.4             | 12     | 3.1 (0.8, 5.4)           | 16.5 (11.6, 21.4) |  |
| Discharge diagnosis group                                                                                          |         | 1             |                  |        |                          |                   |  |
| Infectious & parasitic diseases                                                                                    | 3.8     | 12            | 0.0              | 0      | -                        |                   |  |
| Neonlasms                                                                                                          | 6.7     | 329           | 9.0              | 1      | 1.4*                     | 2.3*              |  |
| Endocrine, nutritional & metabolic diseases &<br>immunity disorders                                                | 4.9     | 20            | 7.0              | 1      | 1.4*                     | 2.2*              |  |
| Diseases of blood & blood forming organs                                                                           | 4.3     | 3             | 0.0              | 0      |                          |                   |  |
| Diseases of the nervous system & sense<br>organs                                                                   | 3.5     | 2             | 0.0              | 0      |                          |                   |  |
| Diseases of the circulatory system                                                                                 | 5.0     | 142           | 32.0             | 1      | 6.4*                     | 27.0*             |  |
| Diseases of the respiratory system                                                                                 | 7.0     | 4             | 0.0              | 0      |                          |                   |  |
| Diseases of the digestive system                                                                                   | 4.9     | 396           | 7.0              | 2      | 1.4 (0.0, 2.8)           | 2.1 (-4.7, 9.0)   |  |
| Diseases of the genitourinary system                                                                               | 5.4     | 519           | 6.3              | 3      | 1.2 (0.6, 1.7)           | 1.0 (-2.8, 4.8)   |  |
| Complications of pregnancy, childbirth &<br>puerperium & certain conditions originating in<br>the perinatal period | 7.3     | 230           | 6.0              | 3      | 0.8 (->0.0, 1.7)         | -1.3 (-7.9, 5.3)  |  |
| Diseases of the skin & subcutaneous tissue                                                                         | 4.5     | 28            | 97.0             | 1      | 21.6*                    | 92.5*             |  |
| Diseases of the musculoskletal system &<br>connective tissue                                                       | 10.5    | 296           | 0.0              | 0      | <u> </u>                 |                   |  |
| Injury & Poisoning                                                                                                 | 9.9     | 160           | 15.3             | 3      | 1.6 (0.8, 2.3)           | 5.4 (-5.1, 16.0)  |  |
| Symptoms, signs & ill- defined conditions;<br>mental disorders & congenital abnormalities                          | 5.8     | 135           | 39.0             | 3      | 6.8 (0.0, 15.3)          | 33.2 (23.9, 42.6) |  |
| Number of co-morbidities                                                                                           |         |               |                  |        |                          |                   |  |
| None                                                                                                               | 6.2     | 153<br>2      | 17.1             | 11     | 2.8 (0.2, 5.3)           | 10.9 (7.4, 14.4)  |  |
| One                                                                                                                | 7.0     | 528           | 30.8             | 4      | 4.4 (→0.0, 9.9)          | 23.8 (14.8, 32.7) |  |
| Two                                                                                                                | 87      | 216           | 16.0             | 3      | 1.8 (0.0, 3.6)           | 7.3 (-2.1, 16.7)  |  |

\*single infection at a site other than the urinary tract, bloodstream, surgical wound, skin or lower respiratory tract

LOS - length of stay, \*numerator n=1 consequently standard deviation not estimable so confidence limits could not be estimated.

# Table A14.7: Mean length of stay for patients with more than one infection (multiple infections) and uninfected patients by key patient characteristics

| No<br>Mean | HAI                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Multiple in                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | fections                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Ratio of LOS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Additional days                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Mean       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | No HAI Multiple infections                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|            | n                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Mean                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | n                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | (00% 01)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | (0070 01)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| (a)        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | (b)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | (b/a)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | (b-a)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 5.9        | 946                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 23.5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 4                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 4.0 (2.8, 5.2)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 17.6 (12.0, 23.2)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 7.1        | 1330                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 34.3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 20                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 4.8 (2.5, 7.1)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 27.1 (23.4, 30.8)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 6.0        | 485                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 6.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 1.0 *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 0.0*                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| 5.5        | 627                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 29.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 5.3 (0.6, 10.0)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 23.5 (19.4, 27.6)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 6.8        | 870                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 21.3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 3.1 (2.2, 4.0)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 14.4 (10.6, 18.3)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 9.4        | 294                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 56.3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 0.0 (1.9, 10.1)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 46.9 (38.3, 55.5)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 5.4        | 829                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 48.1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 8.9 (4.5, 13.4)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 42.7 (37.8, 47.6)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 10.3       | 462                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 33.5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 4                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 3.3 (0.4, 6.1)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 23.2 (14.8, 31.7)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 5.1        | 445                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 25.5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 5.0 (→0.0, 13.0)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 20.4 (13.8, 27.0)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 6.1        | 336                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 93                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 1.5 (0.4. 2.6)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 3.1 (0.4, 2.6)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 7.5        | 204                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 0.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 1.5        | 204                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 0.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|            | 4510                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 26.2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 14                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 44(1077)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 20.2 (16.6, 23.8)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 0.0        | 1518                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 41.2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 10                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 52(3273)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 33 3 (28.2, 38.4)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 7.9        | 151                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 41.2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 5.2 (0.2, 7.0)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 0.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 3.8        | 12                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 0.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 47/27 56)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 24.5 (19.0.30.0)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| 6.7        | 329                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 31.2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 4.7 (3.7, 3.0)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 24.0 (13.0, 50.0)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 4.9        | 20                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 0.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 0 0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | )                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 0.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 4.3        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 0.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 3.5        | 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 0.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 0 0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | )                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 5.0        | 142                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 84.5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 4                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 17.0 (5.2, 28.8)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 79.5 (66.4, 92.7)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 7.0        | 4                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 0.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 0 0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | )                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 4.9        | 396                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 18.3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 3.8 (1.6, 5.9)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 13.5 (7.8, 19.1)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|            | E10                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 0.3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 17(1025)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 3.9 (1.4. 6.4)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 5.4        | 518                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 5.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 1.1 (1.0, 2.0)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 1.2.15                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 7.3        | 230                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 0.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 4.5        | 28                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 0.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | )                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 10.5       | 296                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 20.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 1.9*                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 9.5*                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| 9.9        | 160                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 38.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 3.8 (→0.0, 7.9)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 28.1 (16.7, 39.5)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 5.8        | 135                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 0.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | )                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 62         | 1532                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 21.5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 15                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 3.5 (1.9, 5.0)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 15.3 (12.3, 18.2)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 7.0        | 529                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 49 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 7.1 (→0.0, 14.5)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 42.2 (33.4.51.0)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| 7.0        | 020                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 50.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 60(27.94)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 44.0 (26.2 52.7)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|            | (a)<br>5.9<br>7.1<br>6.0<br>5.5<br>6.8<br>9.4<br>10.3<br>5.1<br>6.1<br>7.5<br>6.0<br>7.9<br>3.8<br>6.7<br>4.9<br>4.3<br>3.5<br>5.0<br>7.0<br>4.9<br>4.3<br>3.5<br>5.0<br>7.0<br>4.9<br>5.4<br>5.4<br>10.3<br>5.1<br>6.1<br>7.5<br>6.0<br>7.9<br>5.4<br>6.0<br>7.9<br>5.4<br>6.0<br>7.9<br>5.4<br>7.5<br>6.0<br>7.9<br>5.5<br>6.0<br>7.9<br>5.5<br>6.0<br>7.9<br>6.0<br>7.9<br>5.4<br>7.5<br>6.0<br>7.9<br>5.5<br>6.0<br>7.9<br>5.4<br>7.5<br>6.0<br>7.9<br>5.4<br>7.5<br>6.0<br>7.9<br>5.4<br>7.5<br>6.0<br>7.9<br>5.4<br>7.5<br>6.0<br>7.9<br>5.4<br>7.5<br>6.0<br>7.9<br>5.4<br>7.5<br>6.0<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.9<br>5.4<br>7.0<br>7.0<br>7.0<br>7.0<br>7.0<br>7.0<br>7.0<br>7.0 | (a)         5.9       946         7.1       1330         6.0       485         5.5       627         6.8       870         9.4       294         5.5       627         6.8       870         9.4       294         5.4       829         10.3       462         5.1       445         6.1       336         7.5       204         6.0       1519         7.9       757         3.8       12         6.7       329         4.9       200         4.3       3         3.5       2         5.0       142         7.0       4         4.9       396         5.4       519         7.3       230         4.5       28         10.5       296         9.9       160         5.8       135         6.2       1532         7.0       528         8.7       216 | (a)         (b) $5.9$ 946         23.5           7.1         1330         34.3           6.0         485         6.0           5.5         627         29.0           6.8         870         21.3           9.4         294         56.3           9.4         294         56.3           5.4         829         48.1           10.3         462         33.5           5.1         445         25.5           6.1         336         9.3           7.5         204         0.0           6.0         1519         26.2           7.9         757         41.2           3.8         12         0.0           6.7         329         31.2           4.9         20         0.0           4.3         3         0.0           3.5         2         0.0           5.0         142         84.5           7.0         4         0.0           4.9         396         18.3           5.4         519         9.3           7.3         230         0.0 | (a)       (b) $5.9$ 946       23.5       4         7.1       1330       34.3       20         6.0       485       6.0       1         5.5       627       29.0       5         6.8       870       21.3       11         9.4       294       56.3       7         5.4       829       48.1       11         10.3       462       33.5       4         5.1       445       25.5       2         6.1       336       9.3       7         7.5       204       0.0       0         6.0       1519       26.2       14         7.9       757       41.2       10         3.8       12       0.0       0         3.8       12       0.0       0         4.9       20       0.0       0         3.5       2       0.0       0         4.9       20       0.0       0         3.5       2       0.0       0         4.9       396       18.3       3         5.4       519       9.3       7 | (a)         (b)         (b)           5.9         946         23.5 $44.0$ (2.8, 5.2)           7.1         1330         34.3         20 $4.8$ (2.5, 7.1)           6.0         485         6.0         1         1.0           5.5         627         29.0         5         5.3 (0.6, 10.0)           6.8         870         21.3         11         3.1 (2.2, 4.0)           9.4         294         56.3         7         0.0 (1.9, 10.1)           5.4         829         48.1         11         8.9 (4.5, 13.4)           10.3         462         33.5         4         3.3 (0.4, 6.1)           5.1         445         25.5         2         5.0 ( $\rightarrow 0.0, 13.0$ )           6.1         336         9.3         7         1.5 (0.4, 2.6)           7.5         204         0.0         0            6.0         1519         26.2         14         4.4 (1.0, 7.7)           7.9         757         41.2         10         5.2 (3.2, 7.3)           4.9         20         0.0            4.3         3         0.0            5.0 <td< td=""></td<> |

LOS - length of stay, \*numerator n=1 consequently standard deviation not estimable so confidence limits could not be estimated.



# An economic model to assess the cost and benefits of the routine use of silver alloy coated urinary catheters to reduce the risk of urinary tract infections in catheterized patients

R. Plowman\*, N. Graves\*, J. Esquivel<sup>+</sup> and J.A. Roberts\*

\*London School of Hygiene & Tropical Medicine, Keppel Street, London WCIE 7HT, UK and †PrevNiTech Ltd, Surrey, UK

**Summary**: Prevalence studies generally find nosocomial urinary tract infections to be the most common type of nosocomial infection, accounting for between 21% and 45% of all HAIs. The main risk factor appears to be the presence of a urinary catheter, with an estimated 80% of these infections being associated with their use. This paper describes a model which quantifies the extent of the burden of these infections in terms of the number of patients affected and the costs incurred by the hospital sector; and identifies the potential benefits of the routine use of silver alloy coated catheters, as a means of reducing the incidence of this type of infection. An illustrative model of the annual costs and benefits associated with the routine use of this intervention in adult, non-day case patients admitted to the medical and surgical specialties of NHS hospitals throughout England is presented. The results suggest that a 14.6% reduction in the incidence of urinary tract infections in catheterized medical patients, and a 11.4% reduction in catheterized surgical patients, would cover the cost of the intervention. Any further reduction in incidence would result in net positive benefits.

© 2001 The Hospital Infection Society

Keywords: Nosocomial urinary tract infections; urinary catheter; modelling; costs and benefits; infection control; silver alloy coated urinary catheter.

### Introduction

1

Studies of urinary tract infections in patients admitted to both medical and surgical specialties suggest that between 1% and 3% acquire a urinary tract infection (UTI),<sup>1-4</sup> whilst a study that was limited to infections occurring in surgical, urology, gynaecology and orthopaedic patients who had an

operative procedure, found that 6.3% acquired a UTI.<sup>5</sup> Studies suggest that between 1% and 5% of patients with UTI will develop a secondary bacteraemia.<sup>1,3,6</sup> Prevalence studies generally find UTIs to be the most common type of nosocomial infection accounting for between 21% and 45% of all HAIs.<sup>7-11</sup> A key risk factor for these infections is the presence of a urinary catheter<sup>6,12-15</sup> with an estimated 80% of nosocomial UTIs (NUTIs) being associated with the presence of this device.<sup>16</sup> NUTIs result in additional morbidity<sup>3,6,17-19</sup> and mortality,<sup>20,21</sup> and represent a considerable economic burden to the health care sector, patients and their carers.<sup>4,5,22-26</sup> This paper presents a model which quantifies the extent of the burden in terms

Received 6 January 2000; revised manuscript accepted 5 January 2001.

Author for correspondence: Rosalind Plowman, Lecturer, Health Services Research Unit, London School of Hygiene & Tropical Medicine, Keppel Street, London WC1E 7HT, UK Tel.: 020 7927 2255; Fax: 020 7580 8183; E-mail: R.Plowman@lshtm.ac.uk

of the number of catheterized patients who acquire a UTI and the costs incurred by the hospital sector, and identifies the potential benefits of introducing the routine use of silver alloy coated catheters, to reduce them. An illustrative model of the annual costs and benefits associated with the routine use of this intervention in adult, non-day case patients admitted to the medical and surgical specialties of National Health Service (NHS) hospitals throughout England is also presented.

## Aims and objectives of the economic model

The model aimed to assess the following:

- (1) the number of NUTIs occurring in catheterized patients admitted to specialties of interest at one or more hospitals
- (2) the economic burden these infections impose on the hospital sector in terms of the number of extra days patients remain in hospital and their associated value
- (3) the potential benefits of an intervention which aims to reduce the incidence of this type of infection.

#### Methods

The literature on the incidence of NUTIs, risk factors for them, their impact on mortality, and the economic burden imposed was reviewed. Relevant papers were identified from the Medline database and the information obtained used to devise the model, which was subsequently developed using the spreadsheet computer package Microsoft Excel 97. The model was developed as a flexible tool, which the user could adapt to reflect the patient group of interest within a particular setting. An illustrative model of the annual costs and benefits of the routine use of silver alloy coated catheters in adult, non-day case patients, admitted to the medical and surgical specialties of NHS hospitals throughout England was subsequently constructed.

#### The structure of the economic model

Figure 1 illustrates the structure of the model. The starting point for this model is the number of admissions to the specialties of interest. These patients may then be sub-divided into the groups of interest, for example categorized by admission specialty. Figure 1 shows the patients subdivided into two categories however the model allows subdivision into as many sub-groups as required. The patient groups of interest are then further subdivided into those who are catheterized and those not catheterized. The number of NUTIs that might occur in these catheterized patients is then determined. Estimates of the economic impact these infections have on the hospital sector are then derived by



Figure 1 The structure of the economic model.

multiplying the number of NUTIs by an appropriate estimate of the number of extra days patients have to remain in hospital, and multiplying this latter figure by the relevant estimate of a cost per day in hospital. Estimates of the benefits of a specific intervention which aims to reduce the incidence of NUTIs are subsequently derived. The intervention considered in this model is the routine use of silver alloy coated urinary catheters. Estimates of the number of bed days released for alternative use and their associated value are derived.

In order to use this model the following information is required:

- (1) Number of admissions to the specialties of interest at one or more hospitals.
- (2) Of those patients selected, the number (or proportion) of patients within the group of interest, for example, the number of patients admitted to relevant specialties. In Figure 1 the patients have been subdivided into medical admissions and surgical admissions.
- (3) The number (or proportion) of patients in the selected groups of interest who are catheterized.
- (4) The number or estimated incidence of NUTIs occurring in these catheterized patients.
- (5) The average number of additional days that catheterized patients with a NUTI remain in hospital.
- (6) The cost of an additional day in hospital.

(7) The cost and estimated effectiveness of the intervention.

A facility, which enabled sensitivity analysis to be conducted, was also built into the model. Sensitivity analysis reveals how the varying of one or more assumptions affects the model. In this model the sensitivity analysis was used to assess the effect of changing the estimates of the incidence of NUTIs in catheterised patients, the effectiveness of the intervention, the number of extra days patients remained in hospital as a result of the infection, and the cost per additional bed day.

The model produces four main outputs:

- (1) The number of catheter related NUTIs that would occur in a pre-defined patient group.
- (2) The additional bed days utilized as a result of a NUTI.
- (3) The value of the resources used by catheterized patients who acquire a NUTI.
- (4) The net financial benefits associated with the routine use of silver alloy coated catheters.

#### Results of the illustrative model

A model to assess the costs associated with catheterassociated NUTIs occurring in adult ( $\geq 18$  years of age), non-day case admissions to the medical and surgical specialties of NHS hospitals throughout England (Figure 2), and the potential benefits of



Figure 2 The economic model of the costs of catheter associated NUTIs occurring in adult, non-day case patients admitted to the medical and surgical specialties of NHS hospitals throughout England.

|                                                                                    |             |        | Effec         | ctiveness o       | f catheter  | at prevent | ing NUT | ls      |        |          |        |          |
|------------------------------------------------------------------------------------|-------------|--------|---------------|-------------------|-------------|------------|---------|---------|--------|----------|--------|----------|
| Surgical patients                                                                  |             | ١%     | 10%           | 20%               | 30%         | 40%        | 50%     | 60%     | 70%    | 80%      | 90%    | 100%     |
| available                                                                          |             | 1756   | 17 562        | 35   24           | 52 686      | 70 248     | 87811   | 105 373 | 122935 | 140 497  | 158059 | 175 62 1 |
| Gross benefits<br>(value of bed days)<br>Additional cost of<br>silver alloy costed | (£ Million) | 0.53   | 5.25          | 10.50             | 15.75       | 21.00      | 26.26   | 31.51   | 36.76  | 42.01    | 47.26  | 52.51    |
| catheter                                                                           | (£ Million) | 5.99   | 5.99          | 5.99              | 5.99        | 5.99       | 5.99    | 5.99    | 5.99   | 5.99     | 5.99   | 5.99     |
| Net benefits                                                                       | (£ Million) | - 5.46 | -0.7 <b>4</b> | <del>4</del> .5 l | 9.77        | 15.02      | 20.27   | 25.52   | 30.77  | 36.02    | 41.27  | 46.52    |
|                                                                                    |             |        | Effe          | ctiveness c       | of catheter | at prevent | ing NUT | ls      |        |          |        | ·        |
| Medical patients<br>Bed days                                                       |             | ١%     | 10%           | 20%               | 30%         | 40%        | 50%     | 60%     | 70%    | 80%      | 90%    | 100%     |
| made available                                                                     |             | 626    | 6255          | 12 511            | 18 766      | 25 021     | 31 276  | 37 532  | 43 78  | 7 50 042 | 56 297 | 62 553   |
| Gross benefits<br>(value of bed days)<br>Additional cost                           | (£ Million) | 0.15   | 1.46          | 2.91              | 4.37        | 5.83       | 7.29    | 8.74    | 10.20  | 11.66    | 13.12  | 14.57    |
| of silver alloy<br>coated catheter                                                 | (£ Million) | 2.13   | 2.13          | 2.13              | 2.13        | 2.13       | 2.13    | 2.13    | 2.13   | 2.13     | 2.13   | 2.13     |
| Net benefits                                                                       | (£ Million) | - 1.99 | - 0.68        | 0.78              | 2.24        | 3.70       | 5.15    | 6.61    | 8.07   | 9.53     | 10.98  | 12.44    |

Table I The estimated costs and benefits of the routine use of silver alloy coated catheters in adult patients, excluding day cases, admitted to the medical and surgical specilaties of NHS hospitals throughout England at different levels of effectiveness

introducing the routine use of silver alloy coated urinary catheters (Table I) was developed. This model uses real data obtained from a variety of sources, details of which are given below together with the output of the model.

#### Number of admissions (cell 1)

The number of adult ( $\geq$ 18 years of age), non-day case admissions to the medical and surgical specialties of all NHS hospitals throughout England was obtained from the hospital episodes statistics database.<sup>27</sup> In 1994/5 there were 4181670 admissions that met this criterion. Appendix 1 provides details of the specialties included in the illustrative model.

## Number of patients who were medical and surgical admissions (cell 2 and 3)

The number of patients admitted to medical and to surgical specialties was also obtained from the hospital episodes statistics database.<sup>27</sup> In 1994/5 2042633 (49%) of the 4181670 adult, non-day case admissions to the medical and surgical specialties of NHS hospital in England were surgical admissions and 2139037 (51%) were medical admissions. Appendix 1 contains details of the specialties included in these two categories.

# Number of patients who are catheterized (cells 4 and 6)

Information on catheterization rates is not routinely collected and as such the model utilizes estimates taken from the literature. For the purposes of this illustrative model estimates taken from the results of an audit of infection control policies and practices in 19 hospitals in England and Wales have been applied.<sup>28</sup> The audit study provided information on the proportion of medical, surgical, orthopaedic, and gynaecology patients who were catheterized at some point during their in-patient stay. The study was restricted to adult patients who had a minimum hospital stay of three days. The median values for the 19 sites for each speciality were as follows: medical 11.6% (range: 5.0-17.0%); general surgery 34.4% (range: 16.2-50.0%); orthopaedic 17.3% (range: 10.1-26.0%) and gynaecology 40.4% (range: 20.9-72.0%). This model utilizes the median value for the medical specialties (11.6%: range 5.0-17.0%) and a weighted value for all surgical specialties (31.1%: range 16-49%). This latter estimate was derived from the individual surgical speciality estimates and data on the number of admissions to these specialties. Although the case-mix of patients studied in this earlier study is not directly comparable with the patient group described in this model, and therefore it is not clear how accurately these estimates reflect the actual catheterization rate in the selected patient group, it was felt that these were the best estimates currently available.

# Number (or percentage) of catheterized patients who acquire a NUTI (cells 8 and 10)

Data on the incidence of NUTIs in catheterized patients were not available. In their absence, estimates were derived as follows. A study of the incidence of hospital acquired infections occurring in 3326 adult patients admitted to the general surgical, general medical, orthopaedic and gynaecology specialties of a district general hospital, indicated that 83 (2.5%) acquired a UTI.<sup>2</sup> Data from the US suggests that 80% of NUTIs occur in catheterized patients.<sup>16</sup> If this assumption is applied to the number of NUTIs observed it follows that 66.4 of the 83 NUTIs observed occurred in catheterized patients.

In order to obtain an estimate of the incidence of NUTIs in catheterized patients, this figure was expressed as a percentage of the number of catheterized patients. In the absence of information on how many of the 3326 patients involved in the incidence study<sup>2</sup> were catheterized, an estimate was derived by applying the specialty - specific median catheterization rates observed in the audit of infection control at 19 hospitals in England and Wales<sup>28</sup> to data on the number of patients within each specialty, and the number of catheterized patients was summed. Overall an estimated 905.47 of the 3326 patients involved in the incidence study<sup>2</sup> were catheterized at some stage during their admission. If the estimated number of catheter-related NUTIs (66.4 NUTIs) is expressed as a percentage of these 905.47 catheterized patients, then the incidence of NUTIs in catheterized patients is estimated as 7.3%, (66.4/905.47) and this figure was used in this model.

# The number of additional days that catheterized patients with a NUTI remain in hospital

The assumption was made that NUTIs occurring in surgical patients, on average, result in patients

remaining in hospital an additional 3.6 days. This estimate was taken from the results of a study which estimated the cost of NUTIs occurring in adult patients admitted to the gynaecology, orthopaedic, general surgery and urology specialties of a district general hospital.<sup>5</sup> In the absence of any literature that exclusively considered the impact of NUTIs in medical patients on hospital sector resources use, the same estimate was applied to NUTIs occurring in medical patients.

#### The cost of an additional days stay in hospital

Estimates of the cost per bed day were derived as follows. Data were retrieved from the Chartered Institute of Public Finance Accountants' Health Service Database<sup>29</sup> on total bed days supplied, by speciality, for as many hospitals as provided data. Corresponding data were also extracted on expenditure for these bed days. These specialties were organised into medical and surgical specialties and the average cost of providing a bed day was derived for each speciality. To provide an overall figure for surgical and medical specialties, the sum of the expenditures was divided by the sum of the bed days (Table II).

| lable II | Average | value | of a | bed | day | by | speciality |
|----------|---------|-------|------|-----|-----|----|------------|
|----------|---------|-------|------|-----|-----|----|------------|

| Medical/surgical category | Specialties               | Mean (£) |
|---------------------------|---------------------------|----------|
| Surgical specialties      | General surgery           | 266      |
|                           | Urology                   | 263      |
|                           | Orthopaedics              | 264      |
|                           | Gynaecology               | 323      |
|                           | Neuro surgery             | 410      |
|                           | Plastic surgery           | 358      |
|                           | Cardiothoracic            | 522      |
|                           | All surgical specialities | 299      |
| Medical specialties       | Geriatrics                | 125      |
|                           | Cardiology                | 386      |
|                           | Infectious diseases       | 275      |
|                           | Medical oncology          | 327      |
|                           | Neurology                 | 258      |
|                           | Rheumatology              | 248      |
|                           | Gastroentorology          | 241      |
|                           | Haematology               | 299      |
|                           | Thoracic medicine         | 244      |
|                           | GU medicine               | 295      |
|                           | Nephrology                | 276      |
|                           | Rehabilitation            | 188      |
|                           | General medicine          | 171      |
|                           | All medical specialties   | 233      |

Data source: Chartered Institute of Public Finance Accounts' Health Service Database, London (1998).

# Estimating the potential benefits of the routine use of silver alloy coated catheters

Having identified the number of patients who acquire a NUTI, and the estimated impact these infections have on hospital sector resource use, the final stage of this model is to identify the potential savings that might result from the routine use of silver alloy coated catheters. In order to estimate these benefits information on the cost of the intervention and its estimated level of effectiveness is required. The estimated additional cost of silver alloy coated catheters compared to non-coated catheters is  $f.9.^{30}$  If this figure is applied to all catheterized patients it follows that the routine use of this type of catheter in medical patients would cost an estimated £2132509 and in surgical patients £5987165. Estimates of the effectiveness of the intervention may be obtained from the results of clinical trials.<sup>31-34</sup> However, it should be noted that while these trials demonstrate that the intervention is effective at reducing the incidence of NUTIs, the impact that their routine use has on the incidence of NUTIs in other settings will depend on the infection rate and the case-mix of patients involved, and as such a range of estimates of the effectiveness of this intervention, (0-100%) has been incorporated into this model.

#### **Results of the illustrative model**

If the above assumptions are accepted, the model suggests that 17376 medical and 48784 surgical patients acquire a catheter related NUTI (Figure 2). Medical patients were estimated to remain in hospital for an additional 62553 days, valued at £15 million, and surgical patients were estimated to remain in hospital for additional 175621 bed days, valued at £53 million.

The additional cost of the routine use of silver alloy coated catheters was balanced against these costs assuming different levels of effectiveness at preventing NUTIs (Table I). If the routine use of silver alloy coated catheters was adopted at an additional cost of  $\pounds 9$  per catheter then in order for the benefits to outweigh the costs 14.6% of infections occurring in medical patients need to be prevented (Figure 3) and 11.4% of infections occurring in surgical patients need to be prevented (Figure 4). The results of some clinical trials suggest that the proportion of infections that could be prevented may be higher than this.<sup>31-34</sup> For example, one study found that the routine use of silver alloy coated catheters reduced the risk of infection by 32%.34 Clearly at this level of effectiveness the benefits of the use of silver alloy coated catheters significantly outweigh the costs.



Figure 3 Costs and benefits of the routine use of silver alloy coated catheters in adult, non-day case patients admitted to the medical specialities of NHS hospitals throughout England at different levels of effectiveness.

ţ



Figure 4 Costs and benefits of the routine use of silver alloy coated catheters in adult non-day case patients admitted to the surgical specialties of NHS hospitals throughout England at different levels of effectiveness.

| Parameter changed  | Percentage of infections that need to be prevented if the benefits<br>of the intervention are to cover the cost of the intervention |                            |                             |  |  |  |  |
|--------------------|-------------------------------------------------------------------------------------------------------------------------------------|----------------------------|-----------------------------|--|--|--|--|
|                    | ٢                                                                                                                                   | 1edical specialties<br>(%) | Surgical specialties<br>(%) |  |  |  |  |
| Incidence of NUTIs | 4.0%                                                                                                                                | 26.82                      | 20.90                       |  |  |  |  |
| in catheterized    | 5.0%                                                                                                                                | 21.46                      | 16.72                       |  |  |  |  |
| patients           | 5.0%                                                                                                                                | 21.46                      | 16.72                       |  |  |  |  |
|                    | 6.0%                                                                                                                                | 17.88                      | 13.94                       |  |  |  |  |
|                    | 7.0%                                                                                                                                | 15.33                      | 11.94                       |  |  |  |  |
|                    | 7.3%*                                                                                                                               | 14.63                      | 11.40                       |  |  |  |  |
|                    | 8.0%                                                                                                                                | 13.41                      | 10.45                       |  |  |  |  |
|                    | 9.0%                                                                                                                                | 11.92                      | 9.29                        |  |  |  |  |
|                    | 10.0%                                                                                                                               | 10.73                      | 8.36                        |  |  |  |  |
| Average number     | 2                                                                                                                                   | 26.34                      | 20.52                       |  |  |  |  |
| of extra days      | 3                                                                                                                                   | 17.56                      | 13.68                       |  |  |  |  |
| patients remain    | 3.6*                                                                                                                                | 14.63                      | 11.40                       |  |  |  |  |
| in hospital        | 4                                                                                                                                   | 13.17                      | 10.26                       |  |  |  |  |
|                    | 5                                                                                                                                   | 10.53                      | 8.21                        |  |  |  |  |
|                    | 6                                                                                                                                   | 8.78                       | 6.84                        |  |  |  |  |
| Cost per bed day   | 50% of cost used                                                                                                                    | 29.26                      | 22.80                       |  |  |  |  |
|                    | 100% of cost use                                                                                                                    | ed* 14.63                  | 11.40                       |  |  |  |  |
|                    | 150% of cost used                                                                                                                   | 9.75                       | 7.60                        |  |  |  |  |

 
 Table III
 The percentage of infections that need to be prevented if the benefits of the routine use of silver alloy coated urinary catheters are to cover the cost of the intervention and how this varies when the model parameters are changed

\* estimates used in the illustrative model.

#### Sensitivity analysis

Table III shows the results of the sensitivity analysis, which assessed the impact that varying the key assumptions used in the model had on its results. The impact of varying the incidence of NUTIs in catheterized patients, the number of additional bed days that patients remained in hospital, and the value of extra bed days were assessed.

As the values of individual parameters are decreased the intervention needs to be more effective for the benefits to cover the costs and vice versa. For example, if the incidence of NUTI in catheterized patients is only 4% rather than 7.3% as assumed in the model, then providing all the other assumptions are valid, 20.9% of NUTIs occurring in catheterized surgical patients, and 26.8% in medical patients would need to be prevented if the costs of the intervention are to be covered. However, if the incidence of NUTI is higher, then providing the other assumptions are accepted, a lower percentage of infections would need to be prevented to cover the cost of the intervention. For example, if the incidence of NUTIs in catheterized medical and surgical patients was 10%, an 8.4% reduction in the incidence of NUTIs in catheterized surgical patients would be needed to cover the cost of the intervention, and a 10.7% reduction in medical patients.

The effects of altering more than one parameter at a time were not explored in this model. Nor was the effect of varying the proportion of patients who are catheterized. Changing the proportion of patients catheterized will inevitably alter the number acquiring an infection, the number of extra days utilised by infected patients and the cost of this prolonged hospital stay, but will not affect the proportion of infections that needs to be prevented to cover the cost of the intervention.

#### Discussion

#### Validity of the model

The validity of the model is dependent on how realistic the structure of the model is and how accurately the estimates of the parameters used reflect what is happening in the patient group of interest. The illustrative model presented in this paper was based on information derived from the literature and a number of assumptions. However, the model has been developed so that at the level of the individual hospital, more accurate estimates, which reflect the patient group of interest in a particular setting, may be used. For example, the user may have access to local information on the number, or proportion of patients catheterized, the incidence of NUTIs in catheterized patients, and the number and cost of the extra days infected patients remain in hospital. Alternatively, the estimates used in the illustrative model may be replaced by those taken from other studies, where case mix more closely reflects the patient group of interest. Equally, the user may consider the routine use of the more expensive silver alloy coated catheter to be inappropriate and wish to limit it to patients considered to be at high risk of acquiring a urinary tract infection. The user would simply need to substitute data on the number of admissions, with data on the number of patients in the high-risk category, and replace the incidence figures with estimates which reflect the incidence of NUTIs in the high risk group.

#### Interpreting the results of the model

While a number of studies suggest that the use of silver alloy coated catheters is an effective means of reducing the incidence of NUTIs,<sup>31-34</sup> it should be noted that the validity of some of the estimates has been questioned,<sup>35</sup> and a study by Thiobon et al.<sup>36</sup> failed to identify any significant preventative effect. Concerns relate to factors such as the small number of patients involved in some of the studies, and effectiveness being assessed in terms of ability to prevent bacteriuria rather than symptomatic NUTIs.<sup>35,36</sup> However, more recently a relatively large study<sup>34</sup> assessed how effective these catheters were at reducing NUTIs as defined by Centre for Disease Control and Prevention criteria.<sup>37</sup> The results of this latter study confirmed those of earlier studies that demonstrated that silver alloy catheters exerted a preventative effect. The use of silver alloy coated catheters was found to reduce the risk of acquiring an infection by 32%.<sup>34</sup> While it may not be possible to achieve similar levels of reduction in the incidence of this type of infection in all settings, the level of effectiveness achieved to some extent depending on the scope for reducing the incidence of NUTIs, the results of this model suggest that the additional costs associated with the routine use of silver alloy coated catheters can be recouped at relatively low levels of effectiveness and any further reductions result in net benefits.

The results of the sensitivity analysis demonstrate that even when a relatively conservative estimates of the incidence of NUTIs, or the number and value of extra days patients remain in hospital are applied, the benefits of the routine use of silver alloy coated catheters continue to outweigh the additional expenditure. However, it should be noted that the sensitivity analysis did not consider the impact of varying more than one parameter at a time.

The model did not assess the benefits associated with a reduction in secondary bacteraemias and as such the benefits of preventing NUTIs may have been underestimated in the illustrative model. An estimated 1–5% of patients with catheter associated bacteriuria will develop a bacteraemia<sup>1,3</sup> which may further prolong the length of hospital stay and represent an additional burden to the health sector. If the use of silver alloy coated catheters is effective in preventing NUTIs in patients, who in the absence of this intervention would have acquired both a NUTI and a bacteraemia, the benefits of preventing NUTIs through the routine use of this intervention will increase.

It should be understood that the cost of an infection, if avoided, will not all be realised as a cash saving. Many of the costs/benefits are fixed and it is principally the variable costs/benefits (for example drugs, and other consumable items), which represent a smaller proportion of the total costs, that would show as cash savings and as such expenditure that could be avoided.

While the fixed costs avoided will not all be realised as cash savings, they do represent economic benefits as these costs could be deployed to produce other outputs other than treating infection. It is therefore justified to use the full cost data (fixed and variable costs) to represent the benefits. These methodological issues are discussed in more detail elsewhere.<sup>26</sup>

#### **Potential applications**

The model described in this paper can be adapted to the particular needs of the user. It may be of particular interest to infection control workers who wish to demonstrate the magnitude of the burden of this type of infection and the benefits associated with the routine use of silver alloy coated catheters. The information derived from the model may be used to justify the additional expenditure associated with this intervention, and to change policy regarding infection control practice.

While the model presented in this paper focuses on NUTIs occurring in catheterized patients and the routine use of silver alloy coated catheters as a means of reducing this type of infection, the model could also be adapted to assess the benefits of an alternative intervention and/or the prevention of infections at other sites.

#### Acknowledgements

The authors would like to thank Professor G. French for commenting on this manuscript. The work was funded by PrevNiTech Ltd.

#### References

- Haley RW, Hooton TM, Culver DH et al. Nosocomial infections in U.S. hospitals, 1975-1976. Estimated frequency by selected characteristics of patients. Am J Med 1981; 70: 947-959.
- Glenister HM, Taylor L, Bartlett CLR, Cooke EM, Mackintosh CA. An 11 month incidence study of infections in wards of a district general hospital. *J Hosp Infect* 1992; 21: 261-273.
- Krieger JN, Kaiser DL, Wenzel RP. Urinary tract etiology of bloodstream infections in hospitalised patients. *J Infect Dis* 1983; 148: 57-62.
- Plowman R, Graves N, Griffin M et al. The socioeconomic burden of hospital acquired infection. London: Public Health Laboratory Service, 1999.
- Coello R, Glenister H, Fereres J et al. The cost of infection in surgical patients: A case control study. J Hosp Infect 1993; 25: 239-250.
- Stamm WE. Catheter-Associated Urinary Tract Infections: Epidemiology, Pathogenesis, and Prevention. Am J Med 1991; 91(Suppl 3B): 65S-71S.
- EPINE Working Group. Prevalence of hospitalacquired infections in Spain. J Hosp Infect 1992; 20: 1-13.
- Kam KM, Mak WP. Territory-wide survey of hospital infection in Hong Kong. J Hosp Infect 1993; 23: 143-151.
- Sartor C, Sambuc R, Bimar MC, Gulian C, De Micco P. Prevalence surveys of nosocomial infections using a random sampling method in Marseille hospitals. J Hosp Infect 1995; 29: 209-216.
- Emmerson AM, Enstone JE, Griffin M, Kelsey MC, Smyth ETM. Second National prevalence survey of infections in hospitals - Overview of the results. *J Hosp Infect* 1996; 32: 175-190.
- Sramova H, Bartonova A, Bolek S, Krecmerova M, Subertova V. National prevalence survey of hospitalacquired infections in Czechoslovakia. *J Hosp Infect* 1988; 11: 328-334.
- Garibaldi RA, Burke JP, Dickman ML et al. Factors predisposing to bacteriuria during indwelling urethral catheterisation. NEJM 1974; 291: 215.
- Hartstein AI, Garber SB, Ward TT et al. Nosocomial urinary tract infections: A prospective evaluation of 108 catheterized patients. *Infect Cont* 1981; 2: 380–386.
- 14. Shapiro M, Simchen E, Izareli S et al. A multivariate analysis of risk factors for acquiring bacteriuria in

patients with indwelling urinary catheters for longer than 24 hours. Infect Control 1984; 5: 525-532.

- Platt R, Polk BF, Murdoch B et al. Risk factors for nosocomial urinary tract infection. Am J Epidemiol: 1986; 124: 977-985.
- Stamm WE. Urinary Tract Infections. In: Bennett JV, Brachman PS, Eds. Hospital Infections. 4th edn. New York: Lippincott-Raven, 1998: 477-485.
- Garibaldi RA, Mooney BR, Epstein BJ, Britt MR. An evaluation of daily bacteriologic monitoring to identify preventable episodes of catheter-associated urinary tract infection. *Infect Control* 1982; 3: 466-70.
- Bryan C, Reynolds K. Hospital acquired bacteremic urinary tract infection: Epidemiology and outcome. *J Urol* 1984; 132: 494–498.
- Warren JW. Nosocomial urinary tract infections. In: Mendell GL, Douglas RG, Bennett JE, Eds. *Principles and practices of infectious disease*. 3rd edn: Churchill Livingstone, 1990.
- Martone JM, Jarvis WR, Edwards JR, Culver DH, Haley RW. Incidence and nature of endemic and epidemic nosocomial infections. In: Bennett JV, Brachman PS, Eds. Hospital Infections. 4th edn. New York: Lippincott-Raven, 1998: 461-476.
- Platt R, Polk BF, Murdock B, Rosner R. Mortality associated with nococomial urinary tract infection. NEJM 1982; 307: 637-642.
- 22. Scheckler WE. Hospital costs of nosocomial infections: A prospective three month study in a community hospital. *Infect Control* 1978; 1: 150–152.
- Haley RW, Schaberg D, Crossley K, Von Allmen S, McGowan J. Extra Charges and Prolongation of Stay Attributable to Nosocomial Infections: A Prospec-tive Interhospital Comparison. Am J Med 1981; 70: 51-58.
- 24. Rubenstein E, Green M, Modan M, Amit P, Bernstein L, Rubenstein A. The effect of nosocomial infections on the length and costs of hospital stay. *J Antimicrob Chemother* 1982; 9(suppl A): 93-100.
- Shapiro M, Schoenbaum SC, Tager IB, Munoz A, Polk BF. Benefit-cost analysis of antimicrobial prophylaxis in abdominal and vaginal hysterectomy. JAMA 1983; 249 (10) 1290-4.
- 26. Plowman RM, Graves N, Roberts JA. (1997) Hospital Acquired Infection. Office of Health Economics, London.
- Department of Health. Hospital Episodes Statistics database 1994/5. Department of Health, London 1994/95.
- Glynn A, Ward V, Wilson J et al. Hospital-acquired infection: Surveillance, policies and practice. 1997; Public Health Laboratory Service. London.
- 29. Chartered Institute of Public Finance and Accounts. The Health Service Database. London 1998.
- 30. Personal communication. PrevNiTech Ltd.
- Liedberg H., Lundeberg T. Silver alloy coated catheters reduce catheter associated bateriuria. Brit J Urol 1990; 65: 379-381.
- 32. Lee J, Hernandez J. Silver/hydrogel-coated catheter used to reduce incidence of nosocomial urinary tract infections. *AJIC* 1996; **224**: 2, 117.
- 33. Saint S., Elmore JG., Sullivan SD., Emerson SS., Koepsell TD. The efficacy of silver alloy-coated

urinary catheters in preventing urinary tract infection: a meta-analysis. Am J Med 1998; **105**: (3) 236–41.

- 34. Karchmer TB, Giannetta ET, Muto CA, Strain BA, Farr BM. A randomized crossover study of silvercoated urinary catheters in hospitalized patients. *Arch Intern Med* 2000; 160: 3294–3298.
- Darouiche RO. Anti-Infective Efficacy of Silver-Coated Medical Prostheses. Clin Infect Dis 1999; 29: 1371-1377.
- 36. Thibon P, Le Coutour X, Leroyer R, Fabry J. Randomised multi-centre trial of the effects of a catheter coated with hydrogel and silver slats on the incidence of hospital-acquired urinary tract infections. J Hosp Infect 2000; 45: 117-124.
- Garner JS, Jarvis WR, Emori TG, Horan TC, Hughes JM. CDC definitions for nosocomial infections. Am J Infect Control 1988; 16: 128-140.

Appendix I Number of adult non-day case admissions to selected specialties of NHS hospitals throughout England in 1994/5

|                      | Speciality                | Number of adult<br>(≥18), (non-day<br>case admissions) |
|----------------------|---------------------------|--------------------------------------------------------|
| Medical specialties  | General medicine          | 1090716                                                |
| ·                    | Gastroentorology          | 17839                                                  |
|                      | Endocrinology             | 5038                                                   |
|                      | Haematology (clinical)    | 46 626                                                 |
|                      | Clinical pharmacology     | 5 578                                                  |
|                      | Rehabilitation            | 3 706                                                  |
|                      | Palliative medicine       | 4772                                                   |
|                      | Cardiology                | 96 488                                                 |
|                      | Dermatology               | 13739                                                  |
|                      | Thoracic medicine         | 42 485                                                 |
|                      | Infectious diseases       | 12206                                                  |
|                      | Genito-urinary medicine   | 1 985                                                  |
|                      | Nephrology                | 43 53 1                                                |
|                      | Medical oncology          | 37 572                                                 |
|                      | Nuclear medicine          | 151                                                    |
|                      | Neurology                 | 43316                                                  |
|                      | Clinical neuro-physiology | 307                                                    |
|                      | Rheumatology              | 38165                                                  |
|                      | Geriatric medicine        | 431310                                                 |
|                      | Old age psychiatry        | 38620                                                  |
|                      | Radiotherapy              | 56325                                                  |
|                      | Radiology                 | 1 575                                                  |
|                      | General pathology         | 8                                                      |
| . '                  | Blood transfusion         | 54                                                     |
|                      | Chemical pathology        | 495                                                    |
|                      | Haematology               | 9 970                                                  |
|                      | Histopathology            | 2                                                      |
|                      | Immunopathology           | 53                                                     |
|                      | Medical microbiology      | I.                                                     |
| Surgical specialties | General surgery           | 804 676                                                |
|                      | Urology                   | 223 685                                                |
|                      | Trauma & orthopaedics     | 464 948                                                |
|                      | Neurosurgery              | 37 587                                                 |
|                      | Plastic surgery           | 59 060                                                 |
|                      | Cardiothoracic surgery    | 47 686                                                 |
|                      | Gynaecology               | 501 395                                                |

Data source: Hospital Statistics Database 1994/5.



# The rate and cost of hospital-acquired infections occurring in patients admitted to selected specialties of a district general hospital in England and the national burden imposed

R. Plowman, N. Graves\*, M.A. S. Griffin†, J.A. Roberts\*, A.V. Swan†, B. Cookson‡ and L. Taylor

Infection Control Unit, Laboratory of Hospital Infection, Central Public Health Laboratory, \*Department of Public Health and Policy, London School of Hygiene & Tropical Medicine, †Statistics Unit, Public Health Laboratory Service, and ‡Laboratory of Hospital Infection, Central Public Health Laboratory, London, UK

Summary: Between April 1994 and May 1995 4000 adult patients admitted to selected specialties of a district general hospital were recruited to this study. Hospital-acquired infections presenting during the inpatient stay were identified using previously validated methods of surveillance, and information on daily resource use by both infected and uninfected patients was recorded and estimates of their cost derived. Linear regression modelling techniques were used to estimate how much of the observed variation in resource use and costs could be explained by the presence of an infection. Complete in-patient data sets were obtained for 3980 patients. Of these, 309 patients (7.8%; 95% C1; 7.0, 8.6) presented with one or more hospital-acquired infections during the in-patient period. Infected patients, on average, incurred hospital costs 2.9 (regression model estimate: 2.8; 95% CI; 2.6, 3.0) times higher than uninfected patients, equivalent to an additional £3154 (regression model estimate £2917). Both the incidence and the economic impact varied with site of infection and with admission specialty. Estimates of the burden of hospital-acquired infections occurring in adult patients admitted to similar specialties at NHS hospitals in England were derived from the results of this study. An estimated 320 994 (95% CI; 288 071, 353 916) patients per annum acquire one or more infections which present during the in-patient period, and these infections cost the hospital sector an estimated £930.62 million (95% CI; £780.26; £1080.97 million) per annum. The results presented represent the gross economic benefits that might accrue if these infections are prevented. Further research is required to establish the net benefits of prevention.

© 2001 The Hospital Infection Society

Keywords: Hospital-acquired infection; incidence; economic burden.

#### Introduction

At any one time approximately 10% of hospitalized patients have an infection acquired after admission to hospital.<sup>1</sup> These infections result in additional costs to the healthcare sector, patients and those who care for them.<sup>2</sup> Few studies have estimated the incidence of these infections occurring in a wide range of patients.<sup>3,4</sup> While a number of studies have assessed the economic burden imposed by these

Received 19 January 2000; revised manuscript accepted 18 October 2000.

Author for correspondence: Rosalind Plowman, Lecturer, Health Services Research Unit, London School of Hygiene & Tropical Medicine, Keppel Street, London WC1E 7HT, UK. Fax: 020 7580-8183; E-msil: R. Plowman@ishtm.ac.uk Funding: Department of Health.

#### The rate and cost of hospital-acquired infections

infections, the estimates derived have generally been limited to the costs incurred by the hospital sector, and little information on the distribution of the additional costs incurred between the budgetary centres of the hospital has been provided.<sup>3-7</sup> This study provides a more comprehensive assessment of the incidence of these infections and their impact on resource use. The results of the analysis examining the incidence of hospital-acquired infection presenting during the in-patient period and their impact on hospital sector resource use and costs are presented in this paper.

#### Methods

#### Study site

Patients were recruited from a district general hospital within close proximity of the two institutions involved in the research. This type of hospital was selected in preference to tertiary referral centres in order to facilitate the generalizability of the study results to other healthcare settings. The hospital selected was part of an English NHS Trust providing general acute services, selected regional tertiary specialist services and primary care services. At the time of recruitment it had 579 beds.

A retrospective assessment of how representative the study hospital was of others in England was undertaken in terms of the number of bed days produced, number of staff employed, the average cost per bed day, the average length of stay and the expected length of stay given the case mix of patients. Data on the number of bed days produced, number of staff employed and average cost per bed day at the study hospital and for hospitals throughout England were retrieved from the Chartered Institute of Public Finance Accountants/Healthcare Financial Managers database.8 The study hospital values for these three variables were found to lie within the interquartile range of the distributions. The Health Services Indicators database<sup>9</sup> provided information on the average length of stay, and expected length of stay given the case mix of patients for some of the specialties involved in this study, at both the study site and at other hospitals throughout England. The study hospital values for these variables fell within the interquartile range of the distributions. These findings suggest that for the variables considered, the study hospital was not atypical of other hospitals throughout England.

At the time the study was undertaken the study hospital's infection control team consisted of two infection control nurses and a microbiologist who dedicated approximately 3h to infection control team activities per week. This infection control team was responsible for infection control at the study hospital and 60 non-acute beds at two other hospitals, equivalent to one ICN per 289 acute beds and 30 non-acute beds. This ratio of infection control nurses to number of beds is higher than the national average of one ICN to 535 beds.<sup>10</sup>

#### Patients

Adult (>18 years of age) medical; surgical; orthopaedic; urology; gynaecology; ear, nose and throat; elderly care; and, if they had had a Caesarean section, obstetric patients admitted to selected wards at the study hospital who had a minimum in-patient stay of 30h were eligible for recruitment. Day case patients and patients who remained in hospital overnight and were discharged early the following day were therefore omitted from this study. Resource constraints prohibited the inclusion of these patients.

#### Recruitment

Having obtained ethical committee approval, recruitment began in April 1994 and continued until May 1995. The informed written consent of eligible patients was obtained by one of six research assistants, all of whom were experienced registered nurses and trained in recruitment, surveillance and data collection methods.

From the outset it was clear that the research assistants would not be able to invite all eligible patients to participate in this study. While recruitment itself did not take up a lot of the available time, the identification of hospital-scquired infections (HAIs) and the collection of detailed data on resource use was time consuming. The research assistants were instructed to give priority to the collection of complete data sets rather than recruiting all eligible patients and were given appropriate training and supervision to minimize any selection bias.

Information on the reason eligible patients were not recruited (e.g., the patient declined participation, or was not recruited due to practical reasons such as insufficient time) and the age, sex, admission type and admission specialty was recorded. These data were subsequently analysed to check for any recruitment bias.

#### Data collection

Baseline information including age, sex, diagnosis, co-morbidities and social circumstances was obtained from the patients themselves and their medical and nursing records.

Hospital-acquired infections were identified using the reference method of surveillance and previously validated definitions of infections.<sup>11</sup> The reference method of surveillance sims to identify all HAIs presenting during the in-patient period. It involves consulting relevant healthcare records on a regular basis to obtain evidence of the presence of one or more infections. If this evidence meets the criteria detailed in the definitions used, an HAI is recorded as being present. The site of the infection, date of onset, and, if known, the causative pathogen are also recorded.

Data on daily resource use, including contacts with professionals allied to medicine, laboratory radiological and other diagnostic tests undertaken, surgical and medical procedures performed, drugs prescribed, and care administered by nursing staff were obtained from the patients' medical, nursing, laboratory and paramedic records.

# Development of monetary valuations of resource use

Estimates of the value of the hospital resources used by individual patients were made and the methods used are reported in detail elsewhere.<sup>12</sup> The methods were designed to identify the likely opportunity costs of the resources used.

The hospital costs incurred by patients recruited into this study were categorized as either costs associated with occupying a hospital bed or the cost of specific care and treatment administered to individual patients.

The cost of occupying a hospital bed included the cost of hospital overheads, directorate management and capital charges. The methods used to attribute these costs were developed from interviews with the relevant managers and health care professionals. A cost per bed day was derived for each clinical specialty and this estimate multiplied by the patient's length of hospital stay.

Costs associated with care and treatment administered included the cost of medical time, nursing time, the time of other health care professionals such as physiotherapists, and the cost of diagnostic investigations, procedures carried out and consumables used. Estimates of the daily cost of medical care were derived for each specialty and this was allocated according to the patient's length of hospital stay. Nursing costs were allocated to patients based on the amount of nursing care patients received each day and unit costs were derived for contacts with professionals allied to medicine and allocated in accordance with the number of contacts supplied. Unit costs were also derived for laboratory investigations, radiological investigations, electrocardiograms, endoscopic procedures, surgical procedures, pharmaceuticals, dressings and other consumable products supplied to the patient. These costs were allocated in accordance with the individual patient's consumption of these resources.

All data on the cost of resources were retrieved from the finance department of the hospital and the clinical specialties themselves.

#### Data analysis

#### Checking for recruitment/selection bias

The age, sex, admission type and admission specialty distributions of patients recruited into the study were compared to the distributions that would have been present if all eligible patients were recruited.

#### Incidence of hospital-acquired infections presenting during the in-patient period

The number of patients with one or more HAIs presenting during the in-patient period was expressed as a percentage of the number of patients discharged and 95% confidence intervals derived. Site and specialty specific estimates were also derived.

#### Attribution of costs to HAI

Factors other than infection may have an impact on resource use.<sup>2</sup> Linear regression modelling techniques were used to control for a range of factors that could potentially influence the level of resource use: age, sex, diagnosis, number of comorbidities, admission specialty and admission type. Since the cost and length of hospital stay distributions included a few very high values they were skewed to the right. For this reason the analysis was performed assuming that the underlying distribution was Gamma in form (this distribution, very similar to the log-Normal, is appropriate for skewed data<sup>13</sup>). Estimates of the impact of one or

#### The rate and cost of hospital-acquired infections

more HAIs, and of specific types of infection on resource use, adjusted for the effects of confounding variables were thus obtained from the simplest regression model that adequstely represented the variation in the costs incurred by infected compared with uninfected patients.

#### Deriving national estimates of the number of patients who acquire one or more HAIs

National estimates of the number of adult patients who acquired one or more infections in hospital, which presented during the in-patient stay, were derived by applying the observed in-patient incidence rate and 95% confidence interval to data on the number of adult patients (aged ≥ 18), excluding day cases, admitted to similar specialties, in NHS provider units throughout England in 1994/5 (Table I). The same approach was used to derive specialty specific estimates of the number of patients acquiring one or more infections and estimates of the number of patients acquiring specific types of infection.

# Deriving national estimates of the cost of HAIs to the hospital sector

Estimates of the burden these infections imposed on the hospital sector were derived from data on the observed incidence of hospital-acquired infections presenting during the in-patient period (Tables II, III); the estimated ratio of the hospital costs incurred by infected compared to uninfected patients

Table 1 The number of adult, non-day case, admitsions to selected specialities of NHS Hospitals in England in 1994/5

| Specialty             | Number of adult admissions,<br>excluding day cases, in 1994/5* |
|-----------------------|----------------------------------------------------------------|
| Gynaecology           | 501 395                                                        |
| Elderly care          | 431 310                                                        |
| Obstetrics (caesarean |                                                                |
| sections only)        | 84 089                                                         |
| Orthopaedics          | 464 948                                                        |
| Surgery               | 804 676                                                        |
| Urology               | 223 685                                                        |
| Medicine              | 1431 984                                                       |
| Ear. nose and throat  | 173 216                                                        |
| All specialties       | 4115303                                                        |

Source: Hospital episodes statics data set 1994/5.14

\*Number of adult patients, excluding day cases, admitted to the specialties listed in Appendix 1 at NHS provider units in England In 1994/5. obtained from the linear modelling analysis (Tables II, III); the mean hospital costs incurred by uninfected patients (Tables II, III) and data on the number of adult admissions (Table I). If N is the number of patients admitted nationally, C the baseline cost of treating uninfected patients, *i* the estimated incidence and r the estimated ratio of costs incurred by infected compared to uninfected patients, then NiC(r-1) provides an estimate of the national burden.

The variance of this estimate was derived from the standard deviations of the estimates of the incidence and ratio of costs, sdi and sdr respectively, as follows:  $N^2C^2[i^2sdr^2+(r-1)^2sdi^2]$ .

This estimated variance was subsequently used to obtain 95% confidence intervals for the estimates of the national burden of infection, it being assumed that the sampling error in such an estimate would be approximately normal. Estimates of the number of additional days patients remained in hospital and site and specialty specific estimates of the burden imposed were made using the same approach.

#### Results

#### Sample characteristics

Between April 1994 and May 1995, 5909 patients were eligible for recruitment. Of these, 4000 (67.7%) were recruited, 343 (5.8%) declined participation, 1553 (26.3%) were not recruited due to insufficient time, and in a further 13 cases (0.2%) information on the reason for non-recruitment was not recorded. The age, sex, admission type and admission specialty of patients recruited into the study was broadly similar to that which would have been present if all eligible patients had been recruited (Table IV).

#### Incidence of HAIs

During the in-patient period, 309 (7.8%; 95% CI; 7.0, 8.6) patients presented with one or more HAIs. Site specific estimates indicated that urinary tract infections had the highest incidence, followed by multiple infections (Table II). Specialty specific estimates indicated that the incidence varied considerably with specialty, being lowest amongst ear, nose and throat patients and highest in gynaecology, elderly care and obsterric patients (Table III).

|                        |      | Incidence        |            | Cost (£)                                     |                                       |           | Length of stay (days)                       |                                       |
|------------------------|------|------------------|------------|----------------------------------------------|---------------------------------------|-----------|---------------------------------------------|---------------------------------------|
| ite of HAI             | z    | IR% (95% CI)     | Mean costs | Ratio of costs*<br>(model estimate: 95% CI)‡ | Additional cost†<br>(model estimate)‡ | Means LoS | Ratio of days*<br>(model estimate: 95% CI)‡ | Additional days†<br>(model estimate)‡ |
| Vo HAI                 | 3671 |                  | 1628       |                                              |                                       | 7.6       |                                             |                                       |
| Irinary tract          | 107  | 27 (22 to 3.2)   | 2955       | 1.8 (1.7:1.5 to 1.9)                         | 1327 (1122)                           | 13.7      | 1.8 (1.7: 1.5 to 1.9)                       | 6.1 (5.1)                             |
| Jultiple               | 57   | 1.4 (I.I to 1.9) | 10780      | 6.6 (6.3: 5.4 to 7.4)                        | 9152 (8631)                           | 45.4      | 6.0 (4.8.4.0 to 5.8)                        | 37.8 (29.1)                           |
| ower respiratory tract | 48   | 1.2 (0.9 to 1.6) | 4027       | 2.5 (2.3: 1.9 to 2.7)                        | 2398 (2080)                           | 20.1      | 2.6 (2.1: 1.7 to 2.6)                       | 12.5 (8.4)                            |
| urreical wound         | 38   | 1.0 (0.7 to 1.3) | 3246       | 2.0 (2.0: 1.6 to 2.4)                        | 1618 (1594)                           | 14.1      | 1.9 (1.9:1.6 to 2.4)                        | 6.5 (7.1)                             |
| Other                  | 30   | 0.8 (0.5 to 1.1) | 3892       | 2.4 (2.5: 2.0 to 3.1)                        | 2263 (2465)                           | 21.0      | 2.8 (2.6:2.1 to 3.4)                        | 13.4 (12.4)                           |
| kin                    | 25   | 0.6 (0.4 to 0.9) | 3418       | 2.1 (2.0: 1.6 to 2.5)                        | 1790 (1615)                           | 19.6      | 2.6 (2.4:1.8 to 3.1)                        | 12.0 (10.6)                           |
| loodstream             | 4    | 0.1 (0.0 to 0.3) | 7026       | 4.3 (4:8:2.6 to 8.8)                         | 5397 (6209)                           | 9.5       | 1.2 (1.5:0.8 to 3.0)                        | 1.9 (4.0)                             |
| Any site               | 309  | 7.8 (7.0 to 8.6) | 4782       | 2.9 (2.8:2.6 to 3.0)                         | 3154 (2917)                           | 21.7      | 2.9 (2.5:2.3 to 2.7)                        | 14.1 (11.3)                           |

Table 111 The incidence of hospital ocquired infection by admission specialty and its impact on the length and cost of hospital stay

| Admission         No. of parients         RX% (95% CI)         Mean Cost         Ratio of cost*         Additional cost+         Means LoS         Ratio of LoS*           specialty         No. HAI         HAI         No. HAI         HAI         For oddel estimate;         PSx CI)         Sy CI = 13         PSx                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                              |                      | Incie         | dence               |               |               | Cost(£)                                       | R. S. S. S. S.                         |                 | 10         | Length of stay (days)                         |                                       |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------|----------------------|---------------|---------------------|---------------|---------------|-----------------------------------------------|----------------------------------------|-----------------|------------|-----------------------------------------------|---------------------------------------|
| Gynaecology         339         51         13.1         (9) to 16.6         13.6         13.6         13.1         10.15.0         16.61         13.66         13.6         13.6         13.1         10.15.0         10.16         15.1         10.16         15.1         10.16         15.1         10.15.0         10.16         15.1         12.46         23.1         13.1         24.7         31.3         34.7         31.3         34.7         31.3         34.7         31.3         34.7         31.3         34.7         31.3         34.7         31.3         34.7         31.3         34.7         31.3         34.7         31.3         34.7         31.3         31.3         32.6         53.6         35.7         30.3         35.2         35.7         35.7         30.3         13.2         60.0         13.1         24.7         31.3         31.2         60.0         31.2         34.7         31.3         31.2         26.0         31.2         34.7         31.3         31.2         60.0         31.2         60.0         31.2         60.0         31.2         60.0         31.2         60.0         31.2         60.0         31.2         60.0         31.2         60.0         33.6         31.1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Admission<br>specialty                       | No. of par<br>No HAI | tients<br>HAI | IR% (95% CI)        | Mean<br>No HA | Cost<br>I HAI | Ratio of cost*<br>(model estimate:<br>95% CI) | Additional costs†<br>(model estimate)‡ | Means<br>No HAI | LoS<br>HAI | Ratio of LoS*<br>(model estimate:<br>95% CI)‡ | Additional days†<br>(model estimate)‡ |
| Elderly care         508         74         12.7         (10.1 to 15.7)         1748         5277         30 (3.1:2.6 to 3.5)         3529 (3578)         11.3         34.7         31. (3.1:2.6 to 3.7)           Obsterrics         204         23         10.1         (6.5 to 14.8)         2481         2761         1.1         (1.1:0.8 to 1.4)         2300 (118)         7.5         8.6         1.2         (1.1:0.8 to 1.4)           Contropretics         204         23         10.1         (6.5 to 14.8)         2481         2761         1.1         (1.1:0.8 to 1.4)         236         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2         1.2 <td>Gvnaecology</td> <td>339</td> <td>51</td> <td>13.1 (9.9 to 16.8)</td> <td>1661</td> <td>2196</td> <td>1.3 (1.3: 1.1 to 1.5)</td> <td>535 (470)</td> <td>6.1</td> <td>8.3</td> <td>1.4 (1.3: 1.1 to 1.6)</td> <td>2.2 (2.0)</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Gvnaecology                                  | 339                  | 51            | 13.1 (9.9 to 16.8)  | 1661          | 2196          | 1.3 (1.3: 1.1 to 1.5)                         | 535 (470)                              | 6.1             | 8.3        | 1.4 (1.3: 1.1 to 1.6)                         | 2.2 (2.0)                             |
| Obstancies         204         23         10.1 (6,5 to 14.8)         2481         2761         1.1 (1,1:0.8 to 14)         280 (118)         7.5         8.6         1.2 (1,1:0.8 to 14)           Carearian sections only)         473         40         7.8         (5,5 to 14.8)         2481         2761         1.1 (0,1:0.8 to 14)         280 (118)         7.5         8.6         1.2 (1,1:0.8 to 14)           Carearian sections only)         473         40         7.8         (5,6 to 10.5)         2089         588         2.6 (2,6.2.1 to 3.1)         3296 (3385)         10.4         22.6         2.2 (2,1:17 to 2.6)           Surgery         844         5.4         6.0 (4,5 to 7.8)         17.90         6.189         4.8 (3,5.3.3 to 4.7)         4898 (3795)         5.4         2.3         1.4.3 (3.3.2.7 to 4.0)           Urology         319         227         228         2.2 (2.2.1 (7 to 2.8))         1.4808 (3795)         5.4         2.3         1.3.2.7 to 4.0         2.7         2.4         2.3         1.3.1         2.3.2.2.7 to 4.0         2.7         2.4         2.3         2.3         2.7         2.4         2.3         2.3         2.4         2.3         2.4         2.3         2.3         2.7         2.4         2.3         2.7         2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Elderly care                                 | 508                  | 74            | 12.7 (10.1 to 15.7) | 1748          | 5277          | 3.0 (3.1:2.6 to 3.5)                          | 3529 (3578)                            | 11.3            | 34.7       | 3.1 (3.1: 2.6 to 3.7)                         | 23.5 (23.9)                           |
| Carearian sections only)         473         40         7.8         (5.6 to 10.5)         2069         5.88         2.6         (2.6.2.1 to 3.1)         32.96         (3.285)         10.4         22.6         2.2         (2.1:17)         to 2.6           Orthoppedics         844         5.4         6.0         (4.5 to 7.8)         12.90         6.189         4.8         (3.9.3.13 to 4.7)         4998         (3.795)         5.4         2.31         4.3         (3.3.2.7) to 4.6           Unobaction         844         5.4         6.0         (4.5 to 7.8)         12.70         6.88         2.3         1.3         1.3         2.7         1.0         4.3         3.3         2.7         1.4         3.3         2.7         2.0         0.4         4.3         (3.3.2.7) to 4.6         0.4         1.3         1.3         2.7         2.0         1.4         2.7         2.2         1.2         1.3         2.7         0.4         1.3         1.3         2.7         0.4         1.3         1.3         2.7         0.4         1.3         1.3         2.7         0.4         1.3         1.3         2.7         0.4         1.3         1.3         2.7         2.2         2.7         1.2         1.2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Obstetrics                                   | 204                  | 23            | 10.1 (6.5 to 14.8)  | 2481          | 2761          | 1.1 (1.1:0.8 to 1.4)                          | 280 (118)                              | 7.5             | 8.6        | 1.2 (1.1: 0.8 to 1.4)                         | 1.2 (0.5)                             |
| Orthopaedics         473         40         78         (5.6 to 10.5)         2089         5385         2.6 (2.6.2.1 to 3.1)         3296 (3285)         10.4         22.6         2.2 (2.1:17 to 2.6)           Surgery         844         5.4         6.0 (4.5 to 7.8)         1290         6189         4.8 (3.9.3.3 to 4.7)         4998 (3795)         5.4         2.3 (1.17 to 2.6)           Unology         844         5.4         6.0 (4.5 to 7.8)         1270         610         4.8 (3.9.5 to 3.1)         4308 (3795)         5.4         2.31         4.3 (3.3.2.7) to 4.0           Unology         439         2.77         2.4 (4.6.18 to 2.7)         1.480 (15.44)         5.1         1.4.0         2.7 (2.6.2.0 to 3.4)           Orology         38         4.5 (3.2.10 6.2)         1559         7.71         4.7 (4.6.380 5.6)         5.71 (2.6.2.1)         8.3         2.5.7 (2.4.0.3.4)           For noise, Athener         6.4         2         3.0 (0.4 to 10.5)         2.127 (4.6.180 5.6)         5.71 (2.6.2.1)         4.0 (2.7.2.4 to 3.1)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | (caesarian sections only)                    |                      |               |                     |               |               |                                               |                                        |                 |            |                                               |                                       |
| Surgery         844         54         6.0 (45 to 7.8)         1290         6.189         4.8 (3.9:3.13 to 4.7)         4898 (3795)         5.4         2.3.1         4.3 (3.3:2.7 to 4.0           Unology         139         27         5.8 (3.9 to 6.3)         1276         2758         2.2 (1.2:1.7 to 2.8)         1482 (1544)         5.1         14.0         2.7 (24:2.00 to 3.4           Unology         38         4.5 (1.3 to 6.2)         1526         7271         4.7 (44:3.90 to 5.4)         5.11 (4.0         5.1         14.0         2.7 (24:2.40 to 3.4)           For mode/me         800         38         4.5 (1.3 to 6.2)         1559         7271 (44:2.90 to 5.4)         5.71 (26:2.10 to 3.4)         13.0 (2.2.4 to 3.4)           For mode/me         6.4         2         3.0 (0.4 to 10.5)         2.17 (24:2.90 to 5.4)         3516 (2007)         4.0         10.0         2.5 (2.2: (27:0.5' to 2.5)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Orthopaedics                                 | 473                  | 40            | 7.8 (5.6 to 10.5)   | 2089          | 5385          | 2.6 (2.6: 2.1 to 3.1)                         | 3296 (3285)                            | 10.4            | 22.6       | 2.2 (2.1:1.7 to 2.6)                          | 12.3 (11.4)                           |
| Urology 439 27 5.8 (39 to 8.3) 1276 2758 22 (2.2: 1.7 to 2.8) 1482 (1544) 5.1 (4.0 2.7 (2.6: 2.0 to 3.4 to 3.8 to 3.2 to 3.2 to 3.1 (3.0: 2.4 to 3.6 to 3.4 to 3. | Surgery                                      | 844                  | 5             | 6.0 (4.5 to 7.8)    | 1290          | 6189          | 4.8 (3.9: 3.3 to 4.7)                         | 4898 (3795)                            | 5.4             | 23.1       | 4.3 (3.3: 2.7 to 4.0)                         | 17.7 (12.4)                           |
| Medicine 800 38 4.5 (3.2 to 6.2) 1559 7271 4.7 (4.6.3.8 to 5.6) 5712 (56.21) 8.3 2.5.7 3.1 (3.0:2.4 to 3.6<br>Far moticine 64 2 3.0 (0.4 to 10.5) 2.127 56.44 2.7 (1.9.0.8 to 4.6) 3516 (2007) 4.0 10.0 2.5 (2.2.0.9 to 5.7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Urolony                                      | 439                  | 27            | 5.8 (3.9 to 8.3)    | 1276          | 2758          | 2.2 (2.2: 1.7 to 2.8)                         | 1482 (1544)                            | 5.1             | 14.0       | 2.7 (2.6: 2.0 to 3.4)                         | 8.8 (8.2)                             |
| Ear more & throat 64 2 30 (04 to 105) 2127 5644 27 (1.9:08 to 4.6) 3516 (2007) 4.0 10.0 2.5 (2.2:0.9 to 5.7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Medicine                                     | 800                  | 38            | 4.5 (3.2 to 6.2)    | 1559          | 7271          | 4.7 (4.6: 3.8 to 5.6)                         | 5712 (5621)                            | 8.3             | 25.7       | 3.1 (3.0: 2.4 to 3.8)                         | 17.3 (16.8)                           |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Ear nose & throat                            | 64                   | 2             | 3.0 (0.4 to 10.5)   | 2127          | 5644          | 2.7 (1.9:0.8 to 4.6)                          | 3516 (2007)                            | 4.0             | 10.0       | 2.5 (2.2: 0.9 to 5.7)                         | 6.0 (4.8)                             |
| All specialities 3671 309 7.8 (7.0 to 8.6) 1628 4782 2.9 (2.8: 2.6 to 3.0) 3154 (2917) 7.6 21.7 2.9 (2.5: 2.3 to 2.7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | All specialties                              | 3671                 | 309           | 7.8 (7.0 to 8.6)    | 1628          | 4782          | 2.9 (2.8: 2.6 to 3.0)                         | 3154 (2917)                            | 7.6             | 21.7       | 2.9 (2.5: 2.3 to 2.7)                         | 14.1 (11.3)                           |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | See Table II for key.<br>*LoS=Leneth of stav |                      |               |                     |               |               |                                               |                                        |                 |            |                                               |                                       |

R. Plowman et al.

202

|                     |                          | Recruite | d patients | All eligible | patients* |
|---------------------|--------------------------|----------|------------|--------------|-----------|
|                     |                          | N        | ×          | N            | *         |
| Sex                 | Males                    | 1697     | 42.6       | 2669         | 45.5      |
|                     | Females                  | 2283     | 57.4       | 3199         | 54.5      |
|                     | All patients             | 3980     | 100        | 5868†        | 100       |
| Age group           | 18-59                    | 1874     | 47.1       | 2720         | 47.2      |
|                     | 60 +                     | 2106     | 52.9       | 3040         | 52.8      |
|                     | All patients             | 3980     | 100        | 5760‡        | 100       |
| Admission type      | Elective                 | 1743     | 43.8       | 2332         | 39.6      |
|                     | Emergency                | 2212     | 55.6       | 3499         | 59.4      |
|                     | Unknown                  | 255      | 0.6        | 58           | 1.0       |
|                     | All patients             | 3980     | 100        | 5889         | 100       |
| Admission specialty | Medicine                 | 838      | 21.1       | 1438         | 24.5      |
|                     | Surgery                  | 898      | 22.6       | 1442         | 24.6      |
|                     | Orthopaedics             | 513      | 12.9       | 684          | 11.7      |
|                     | Urology                  | 466      | 11.7       | 712          | 12.1      |
|                     | Obstetrics & gynaecology | 617      | 15.5       | 713          | 12.2      |
|                     | Elderly care             | 582      | 14.6       | 787          | 13.4      |
|                     | ENT                      | 66       | 1.7        | 90           | 1.5       |
|                     | All patients             | 3980     | 100        | 5866¶        | 100       |

Table IV The age, sex, admission type and admission specialty of patients recruited into the study compared to that which would have been obtained if all eligible patients had been recruited

\*This includes patients recruited into the study, patients who declined participation and patients who were eligible for recruitment but for reasons such as insufficient time the research assistant was unable to invite them to participate.

† An additional 21 patients were not recruited however their sex was not recorded, consequently they have been excluded from this analysis.

‡ An additional 129 patients were not recruited however their age was not recorded, consequently they have been excluded from this analysis.

§ These 25 in-patients were transfers from another ward.

¶An additional 23 patients were not recruited however their admission specialty was not recorded, consequently they have been excluded from this analysis.

#### Estimates of the impact of HAI on hospital costs and length of hospital stay

Estimates of the impact HAI had on length of stay and hospital costs, and how these varied with site and specialty are also presented in Tables II and III. Patients presenting with one or more HAIs during the in-patient stay, on average, remained in hospital and incurred costs almost three times greater than uninfected patients. The distribution of the additional costs incurred between the budgetary centres of the provider unit is presented in Table V.

Estimates which only considered the costs of consumables, such as drugs, dressings and other pharmaceuticals, indicated that patients who acquired one or more infections which presented during the in-patient stay incurred consumable costs that were 3.7 (regression model estimate 3.8; 95% CI; 3.3, 4.2) times higher than those incurred by uninfected patients, equivalent to an increase of 2.315 (regression model estimate £325) per patient.

The estimated average increase in costs was lowest for urinary tract infections, with costs almost twice as high as in uninfected patients. The greatest increases were associated with multiple infections: on average costs were over six times higher than uninfected patients. With the exception of bloodstream infections, the impact of specific types of infection on length of hospital stay followed a similar pattern. The four patients who acquired a blood-stream infection and no other infection, had the lowest increase in length of stay; however, two of these patients died during the in-patient period. The results suggest that while these four patients were not found to have a marked increase in their length of stay, their hospital stay was highly resource-intensive. Specialty specific estimates indicated that HAIs occurring in obstetric patients who had had a caesarean section resulted in the lowest increase in length of stay and hospital costs, whereas infections occurring in surgical patients resulted in the highest increases.

| Cost category                  | Mean             | costs (£)      | Ratio of costs | Additional | Percentage contribution to |
|--------------------------------|------------------|----------------|----------------|------------|----------------------------|
|                                | No HAI<br>N=3671 | HAI<br>N = 309 |                | (£)        | additional costs (%)       |
| Hospital overheads             | 436.86           | 1112.58        | 2.5            | 675.72     | 21.43                      |
| Directorate management         | 49.90            | 115.07         | 2.3            | 65.18      | 2.07                       |
| Capital charges                | 207.80           | 499.68         | 2.4            | 291.88     | 9.26                       |
| Medical time                   | 153.69           | 338.09         | 2.2            | 184.40     | 5.85                       |
| Nursing care                   | 385.40           | 1721.25        | 4.5            | 1335.85    | 42.36                      |
| Paramedics & specialist nurses | 17.46            | 63.70          | 3.6            | 46.24      | 1.47                       |
| Physiotherapy                  | 19.02            | 93.35          | 4.9            | 74.34      | 2.36                       |
| Surgical interventions         | 195.20           | 263.44         | 1.3            | 68.24      | 2.16                       |
| Consumables used specific      |                  |                |                |            |                            |
| procedures*                    | 9.95             | 118.35         | 11.9           | 108.40     | 3.44                       |
| Antimicrobials                 | 13.40            | 71.07          | 5.3            | 57.67      | 1.83                       |
| All other drugs                | 40.99            | 150.46         | 3.7            | 109.47     | 3.47                       |
| Microbiology tests             | 6.97             | 33.13          | 4.8            | 26.16      | 0.83                       |
| Other pathology tests          | 48.26            | 113.34         | 2.3            | 65.08      | 2.06                       |
| Endoscopies                    | 2.59             | 6.88           | 2.7            | 4.29       | 0.14                       |
| Radiology                      | 35.19            | 73.13          | 2.1            | 37.94      | 1.20                       |
| Other tests                    | 5.70             | 8.50           | 1.5            | 2.80       | 0.09                       |
| Total costs                    | 1628.38          | 4782.03        | 2.9            | 3153.66    | 100                        |

Table V Distribution of the hospital costs incurred by infected and uninfected patients during the in-patient phase

\* This includes the cost of items such as dressings, drains, lines.

#### The national burden of HAI

Estimates of the number of patients who acquired an infection in hospital which presented during the in-patient period and of the economic burden these infections impose on the hospital sector are presented in Table VI.

An estimated 320 994 adult, non-day case admissions to the specialties covered in the underlying study in 1994/5 acquired one or more infections that presented during the in-patient period. These infections were estimated to cost the hospital sector  $\pounds$ 930.62 million (95% CI;  $\pounds$ 780.26,  $\pounds$ 1080.97 million).

The estimated burden varied considerably with the site of infection. While multiple infections, imposed the highest costs, urinary tract infections, which on average had a relatively low cost per case (model estimate £1122), due to their high incidence, were the most costly single site infection (£123.89 million; 95% CI; £80.96, £166.83 million). In contrast, bloodstream infections, which had a relatively high average cost per case (model estimate, £6209), imposed relatively low costs nationally (£25.53 million; 95% CI; -£6.86, £57.91 million). However, the confidence interval around this estimate is wide, reflecting the degree of uncertainty in the cost estimates derived. Specialty specific estimates indicated that HAIs occurring in medical patients resulted in considerably greater costs than infections occurring in patients admitted to the other specialty groups.

#### Discussion

This study provides the most comprehensive UK estimate to date of the incidence and economic burden imposed by HAIs occurring in adult patients admitted to specialties common to most district general hospitals. At the national level, patients admitted to these specialties accounted for approximately 70% of all adult, non-day case admissions in 1994/5.<sup>14</sup> The remaining 30% were admitted to specialties not included in this study. The incidence and cost of infections occurring in these patients was not explored in this study.

Estimates of the incidence of HAIs indicated that 7.8% of patients who were recruited into the study and for whom complete data sets were obtained acquired one or more infections which presented during the in-patient period. As would be expected, this is less than the most recent prevalence figure of  $9\%^{15}$  and interestingly is similar to the 7.2% incidence rate observed in the only other similar recent UK study.<sup>3</sup>

| ecialties at NHS provider units throughout England, who acquired an infection in hospital in 1994/5. by |                                                       |
|---------------------------------------------------------------------------------------------------------|-------------------------------------------------------|
| Table VI Estimates of the number of odult potients, excluding day coses, admitted to selected sp        | site and by specialty and the economic burden imposed |

|                                    |                      | Number of pati<br>presented durin | ients with an HAI which<br>rg the in-patient period (95% CI) | Estimated cost<br>£ million | to hospital sector (95% CI) | Estimated nui<br>days (95% CI) | mber of additional bed<br>) days million |
|------------------------------------|----------------------|-----------------------------------|--------------------------------------------------------------|-----------------------------|-----------------------------|--------------------------------|------------------------------------------|
| All specialties and sites          | All infections       | 320 994                           | (288 071-353 916)                                            | 930.62                      | (780.26-1080.97)            | 3.64                           | (3.01-4.27)                              |
| Site estimates - All               | Urinary tract        | 111113                            | (90 537-131 690)                                             | 123.89                      | (80.96 –166.83)             | 0.59                           | (0.39-0.79)                              |
| shermines                          | Multiple infections  | 57614                             | (45 268-78 191)                                              | 507.77                      | (348.89-666.65)             | 1.70                           | (11.2-2.28)                              |
|                                    | Lower respiratory    | 49 384                            | (37 038-65 845)                                              | 103.77                      | (59.41-148.12)              | 0.42                           | (0.21-0.63)                              |
|                                    | tract                |                                   |                                                              |                             |                             |                                |                                          |
|                                    | Surgical wound       | 41 153                            | (28 807-53 499)                                              | 62.37                       | (30.93-93.82)               | 0.27                           | (0.12-0.41)                              |
|                                    | Other sites          | 32922                             | (20 577-45 268                                               | 75.87                       | (36.52-115.23)              | 0.38                           | (0.17-0.58)                              |
|                                    | Skin                 | 24692                             | (16461-37038)                                                | 41.79                       | (15.40-68.17)               | 0.28                           | (0.11-0.45)                              |
|                                    | Bloodstream          | 4115                              | (0-12 346)                                                   | 25.53                       | (-6.86-57.91)               | 0.02                           | (-0.02-0.05)                             |
| Specialty estimates –<br>All sites | Gynaecology          | 65 683                            | (49 63 <del>8 - 8</del> 4 234)                               | 30.55                       | (3.68- 57.42)               | 0.13                           | (0.02-0.25)                              |
|                                    | Elderly care         | 54776                             | (43 562-67716)                                               | 196.29                      | (134.51-258.06)             | 1.31                           | (0.88-1.74)                              |
|                                    | Obstetrics C/S only  | 8 493                             | (5 466-12 445)                                               | 1.05                        | (-4.97-7.07)                | 00.0                           | (-0.02-0.02)                             |
|                                    | Orthopaedics         | 36266                             | (26 037-48 820)                                              | 118.94                      | (64.65-173.23)              | 0.41                           | (0.19-0.63)                              |
|                                    | Surgery              | 48 281                            | (36210-62765)                                                | 183.11                      | (117.40-248.82)             | 09:0                           | (0.37-0.83)                              |
|                                    | Urology              | 12974                             | (8724-18566)                                                 | 20.03                       | (8.37–31.69)                | 0.11                           | (0.05-0.17)                              |
|                                    | Medicine             | 64 439                            | (45 823-88 783)                                              | 362.66                      | (209.26-516.07)             | 1.08                           | (0.57-1.60)                              |
|                                    | Ear, nose and throat | 5 196                             | (693-18 188)                                                 | <b>6E.01</b>                | (- 17.00-37.38)             | 0.02                           | (-0.04-0.09)                             |
|                                    |                      |                                   |                                                              |                             |                             |                                |                                          |

\* The estimates are restricted to adult patients, excluding day cases, admitted to the specialties covered in the underlying study in 1994/5.

#### The rate and cost of hospital-acquired infections

The incidence of HAIs varied with the site of infection. Urinary tract infections, as in other studies,<sup>3,4,15</sup> were found to be the most frequent single site infections followed by respiratory tract infections and surgical wound infections. Bloodstream infections were the least frequent type of infection with only four patients (0.1%) acquiring an infection at this site in the absence of other infections. With the exception of the multiple infection category, the incidence figures presented relate to the incidence of infections at a single site. If the site-specific incidence figures included both patients who had a single infection and those who also had infections at one or more other sites, the incidence figures would be higher in some cases. For example, whilst only four patients solely acquired a bloodstream infection, a further 11 patients acquired a bloodstream infection and one or more infections at other sites, giving an overall incidence of 0.4%.

Estimates of the economic burden imposed by HAIs indicate that they are a substantial drain on the hospital sector. Infected patients, on average, incurred hospital costs that were almost three times higher than those of uninfected patients and they remained in hospital 2.5 times longer. The majority of the additional costs incurred were linked to a prolonged hospital stay. This finding is consistent with that found in other studies.<sup>5</sup> Unlike earlier studies, however, we have presented detailed information on how costs were distributed amongst the budgetary centres of the hospital.

The additional costs imposed on the health sector varied considerably with the site of infection. Patients who acquired a urinary tract infection, and no other infection, as in other studies,<sup>5-7</sup> had the lowest increases in costs when compared with patients having infection at other sites, with costs, on average, 1.8 times higher than those of the uninfected patients. Patients who acquired more than one infection on average incurred the highest costs with hospital costs, on average, over six times higher than for uninfected patients.

In this study, regression analysis was used to control for a range of factors other than the presence of an HAI that may have had an impact on resource use. These were age, sex, admission specialty, admission type, diagnosis and number of comorbidities. This approach to attributing costs to HAI was selected in preference to the frequently used 'case-control' method.<sup>2,5-7</sup> This latter approach involves matching infected patients with one or more uninfected patients, on the basis of factors thought to influence resource use, and attributing the difference in costs incurred to the presence of an infection. It encounters a number of difficulties, in particular selecting appropriate matching factors and finding suitable control patients for all infected patients.<sup>14-18</sup>

The approach taken in this study to attribute resource use to the presence of an infection is also associated with difficulties, in particular whether the factors included in the regression modelling covered the major confounding factors. The possibility exists that factors other than those included in the analysis may have had an impact on the level of resource use. Additional analysis explored the impact of including three additional variables: social class; the average level of nursing care required during the period extending from admission to the time of infection; and a severity of illness measure (disease stage). This latter variable was derived from information about the patients' age, sex, diagnosis, co-morbidities, operation codes, admission type, length of in-patient stay and discharge destination. Previously validated algorithms<sup>19</sup> were used to allocate patients to one of three disease stage groups (low, medium and high) depending on the severity of their illness. This work was undertaken by CHKS Ltd using a specifically designed computer software programme. The inclusion of these additional variables was found to have very little effect on the estimated impact of HAI on hospital costs, once the other explanatory variables used, had been taken into account.12

The generalizability of the results of this study to other hospitals is dependent upon whether the case mix included in this study was typical of that which might be found in similar specialties at other hospitals, and how closely clinical practice, and the resource use and associated costs that this implies, reflects that occurring elsewhere.

A retrospective comparison of the study hospitals with other English NHS hospitals indicated that the study hospital was broadly representative of other hospitals in terms of bed days produced, number of staff employed and average length of stay. It was not possible to assess the representativeness of the case mix in any detail, or how typical clinical practice was of that found in other healthcare settings. It should be noted, however, that the specialties included in this study are common to most district general hospitals, and there was nothing to suggest that clinical practice would be markedly different elsewhere.

#### The rate and cost of hospital-acquired infections

It was not possible to recruit all eligible patients. There is no reason to believe this may have biased the sample since a comparison of the age, sex, admission type and admission specialty distributions of the recruited cohort of patients, with that which would have been present if all eligible patients had been recruited, showed that there were very few differences. Due to this, and the fact that the infection group comparisons were made within the strata defined by these factors, and that these were not found to vary between strata, it is reasonable to assume that the results obtained are generalizable to all patients which this sample was intended to represent.

The estimates of the national burden of HAI are based on the observed incidence of HAI and the estimated economic burden imposed. The validity of the aggregated model is dependent upon the generalizability of these results. The analysis allowed for some variability in the incidence rates and ratios of costs incurred by infected compared to uninfected patients, based on the variance observed in the study data. The confidence limits around the national estimates reflects this. No adjustment was made for variation in the baseline cost associated with uninfected patients, nor for inflation. Although it is unclear how accurately the cost estimates associated with uninfected patients reflect costs at other hospitals, the study hospital, as indicated above, was found to be broadly representative of others in terms of the number of bed days produced, the number of staff employed and the average cost per bed day. Consequently, it is likely that the baseline estimates are a reasonable reflection of the costs elsewhere in 1994/5, when the study was conducted, but are likely to be higher today as a result of inflation.

The results presented indicate that a substantial number of adult, non-day case patients admitted to specialties common to most hospitals acquire one or more HAIs and that these infections are a considerable drain on limited hospital resources. In 1994/5 an estimated 320 994 patients acquired one or more HAIs costing the hospital sector an estimated  $\pounds$ 930.62 million, representing 9.1% of the inpatient acute, obstetric and geriatric programme budget in 1994/5,<sup>20</sup> and utilizing an estimated 3.64 million bed days, equivalent to an estimated 478 947 finished consultant episodes.

Given that these estimates are limited to adult, non-day case patients admitted to selected specialties, the actual burden is likely to be considerably greater than that indicated here. The incidence of HAIs occurring in some of the excluded specialties may be higher than that observed in this study, and in some cases are likely to impose a considerable additional burden. For example, infections occurring in patients undergoing renal transplant or cardiac surgery are likely to be particularly demanding of resources.

The estimates of both the number of patients who acquire one or more infections and the economic burden imposed, presented in this paper, are considerably higher than previously published estimates. For example in 1997, Glyn *et al.*<sup>21</sup> estimated that there were at least 100000 HAIs annually, and in 1993 Coello *et al.*<sup>5</sup> estimated that HAIs occurring in surgical patients alone cost the NHS £170 million per annum. In this study HAIs occurring in surgical patients (surgical, orthopaedic, gynaecology, urology and ear, nose and throat patients) were estimated to cost the hospital sector £363 million.

The cost estimates derived in this study represent the average value of the resources that might be released if an infection is prevented, and the national estimates indicate the value of released resources if all infections were prevented. The estimates represent the gross benefits of prevention. Whilst not all HAIs can be prevented<sup>22</sup> the literature suggests that many could be prevented through improved infection control strategies.<sup>2,10,23</sup> Data from the United States suggests that a third of infections may be preventable.23 However, the results of a recent census conducted by the National Audit Office reveal that whilst some infection control practitioners view this frequently quoted figure to be a realistic target, with some considering that a higher level of reduction could be achieved, a substantial proportion consider this to be rather ambitious. The average level of reduction thought to be achievable was 15%.10

If this modified target were schieved, considerable resources would be released for alternative use. For example, crude national estimates of the gross benefits of prevention, which take no account of which infections are prevented, suggest that a 15% reduction in HAIs would result in the release of hospital resources in the order of  $\pounds$ 140 million (95% CI;  $\pounds$ 117,  $\pounds$ 162 million). The majority of these released resources would be fixed in the cost structure of the hospital, with only 11% in the form of cash savings. Any benefits would therefore lie in their alternative use.

The same level of reduction would result in the release of an estimated 546 000 bed days (95% CI;

451 500, 640 500) In 1994/5 the average length of a finished consultant episode for the specialties covered in these estimates was 7.6 days.<sup>14</sup> It therefore follows that this is equivalent to an estimated 71 842 (95% CI; 59407; 84276) finished consultant episodes. Given the current length of waiting lists it appears that the value of these resources, if made available, would be great.

These models, whilst being gross simplifications of the complexities of determining the benefits of prevention, demonstrate that the prevention of HAIs should release considerable resources for alternative use. However, there are costs associated with prevention. Further work is now required to assess the cost and effectiveness of prevention activities and subsequently the net benefits of . infection control.

#### Conclusion

The results presented in this paper indicate that HAI is a substantial problem both in terms of the incidence of this type of infection and the economic burden imposed on the hospital sector. Additional costs fall on the health sector postdischarge, patients and their carers. Estimates of the magnitude of these costs are reported elsewhere.<sup>12</sup>

The cost estimates presented represent the average value of resources that might be released if infections are prevented, and the national estimates, the gross benefits of preventing all infections. Not all infections are avoidable, however, and there are costs associated with prevention activities. The estimation of the net benefits of alternative prevention strategies was beyond the scope of this study. Further work is now required to establish the cost and effectiveness of these activities. This information may then be used together with the information presented in this paper to model the net benefits of infection control practices, the results of which may be used to inform infection control programmes.

#### Acknowledgements

The authors would like to thank the following people: the Board and staff of the NHS Trust in which the research was conducted for their help and support; the patients concerned who gave up their time to take part in the study; Rachael Dunk, Alison Franklin, Janette Trevarthern, Maggie Watera,

.= 7

....

Jennifer White and Linda Wright who were responsible for collecting much of the patient data; Barbara Avres and Christine Berry for administrative support; Jane Bandcroft, Helen Glenister, Mike Rowland, Bernadette Nazareth and Jennie Wilson who were members of the steering group for their comments and support; Kay Butcher, Millie Carter, Sue Dewar, Georgia Duckworth, Johanna Finn, Paul Gillett, Simon Harding, Jayne Holmes, Rosemary Jenkins, William Maton-Howarth, Jennifer McIntyre, Elizabeth Meerabeau, Richard Murray, Elizabeth Tebbs and Ann Whittle who for varied periods of time were members of the Advisory Group; and finally Philip James at CHKS Ltd, UK who allocated patients to one of three severity of illness groups using disease staging software developed by the company.

#### References

- DH/PHL. Hospital infection control. Guidance on the control of infection in hospitals. Prepared by the Hospital Infection Working Group of the Department of Health and Public Health Laboratory Service, London, 1995.
- Plowman RM, Graves N, Roberts JA. Hospital acquired infection. Office of Health Economics, London, 1997.
- Glenister HM, Taylor LJ, Bartlett CLR, Cooke EM, Mackintosh CA, Leigh DA. An 11 month incidence study of infections in wards of a district general hospital. J Hasp Infect 1992; 21: 261-273
- Haley RW, Culver DH, White JW, Morgan WM, Emori TG. The nation-wide nosocomial infection rate. A need for vital new statistics. Am J Epidemiol 1985; 121:159-167
- Coello R, Glenister H, Fereres J, Bartlett C, Leigh D, Sedgwick J, Cooke E. The cost of infection in surgical patients: a case control study. *J Hosp Infect* 1993; 25:239-250.
- Rubenstein E, Green M, Modan M, Amit P, Bernstein L, Rubenstein A. The effect of nosocomial infections on the length and cost of hospital stay. J Antimicrob Chemother 1982; 9: 93-100.
- Haley RW, Schaberg D, Crossley K, Von Allmen S, McGowan J. Extra charges and prolongation of stay attributable to nosocomial infections: a prospective inter-hospital comparison. Am. J Med 1981; 70: 51-58.
- Chartered Institute of Public Finance Accountants/ Healthcare Financial Managers Database. Health Services Financial Database, 1994, London CIPFA.
- Health Service Indicators. NHS Executive, Leeds, 1996.
- National Audit Office The Management and Control of Hospital Acquired Infection in Acute NHS Trusts in England. National Audit Office, London, 2000.

#### The rate and cost of hospital-acquired infections

- Glenister HM, Taylor LJ, Cooke EM, Bartlett CLR. A study of surveillance methods for detecting hospital infection. Public Health Laboratory Service, London, 1992.
- Plowman RM, Graves N, Griffin M, Roberts JA, Swan AV, Cookson B, Taylor L. The socio-economic burden of hospital acquired infection. Public Health Laboratory Service, London, 1999.
- Francis B, Green M, Payne C, eds. GLIM4. The Statistical System for Generalized Linear Interactive Modelling. Oxford: Clarendon Press, 1993.
- 14. Hospital Episodes Statistics Database 1994/5. Department of Health, London, 1995.
- Emmerson AM, Enstone JE, Griffin M, Kelsey MC, Smyth ETM. Second National prevalence survey of infections in hospitals - Overview of the results. J Hosp Infect 1996; 32:175-190
- Freeman J, Rosner BA, McGowan JE. Adverse effects of nosocomial infections. J Infect Dis 1979; 140: 732-740
- 17. Haley RW, Schaberg DR, Von All men S, McGowan JE Jr. Estimating the extra charges and prolongation

of hospitalization due to nosocomial infections: a comparison of methods. J Infect Dis 1980; 141: 248-257.

- Haley RW. Measuring the costs of nosocomial infections: methods for estimating the economic burden on the hospital. Am J Med 1991; 32s-38s.
- McKee M, Petticrew M. Disease staging a casemix system for purchasers? J Pub Health Med. 1993; 15: 25-36.
- Hospital and Community Health Services Programme Budget 1994/5. Department of Health, London, 1995.
- Glynn A, Ward V, Wilson J et al. Hospital Acquired Infections: surveillance, policies and practice. Health Laboratory Service, London, 1997.
- Ayliffe GAJ. Nosocomial infection the irreducible minimum. Infect-Cont 1986; 92-95.
- Haley RW, Culver DH, White JW, Morgan WM, Emori TG. The efficacy of infection surveillance and control programs in preventing nosocomial infections in U.S. hospitals. Am J of Epid 1985; 121: 183-205.




# The Socio-economic Burden of Hospital Acquired Infection

## **EXECUTIVE SUMMARY**

Rosalind Plowman Nicholas Graves Mark Griffin Jennifer A Roberts Anthony V Swan Barry Cookson Lynda Taylor

#### Socio-economic burden of hospital-acquired infection

#### © Public Health Laboratory Service 1999

This report was submitted to the Department of Health in August 1997 and distributed for internal and external review. A full set of reviewers' comments was received in November 1997. A formal response was submitted to the Department of Health in February 1998, following which there was a period of discussion and debate. Further editing work and additional analysis was then undertaken. Amendments to the original text, together with the results of the additional analysis were submitted to the Department of Health in September 1998 and the report accepted for publication in September 1999.

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form or means, electronic, mechanical, photocopying, recording or otherwise, without the written prior permission of the publisher,

Public Health Laboratory Service 61 Colindale Avenue London NW9 5DF

#### ISBN 0 901144 48 7 Printed in England by MWL Print Group Ltd

#### Acknowledgements

The project steering committee wish to thank the Board and staff of the study hospital for their willingness to take part in this study, and for their help and support throughout. We would especially like to thank the Director of Nursing Practice for her invaluable contribution to the study design, and for dealing with practical issues at the study site. We would also like to thank the Information Officer, who extracted information from the hospital dataset, often at very short notice; the Laboratory Manager, who gave up considerable time to assist with the costing of laboratory tests: the Medical Coding Manager and Medical Records Supervisor; the Systems and Information Manager and the Assistant Manager for the Primary Health Care Directorate who had the daunting task of extracting the relevant data from the primary health care database; general practitioners and practice managers for their help and support with accessing medical records post-discharge; the nursing staff and consultants on the study wards and, most important of all, the patients concerned who gave up their time to take part in the study. We would also like to thank Tracy Cowan who edited the document, Stewart McGregor from the Communications Unit, Public Health Laboratory Service for designing the cover and assisting in laying out the document, and Tola Awogbamiye from the Communications Unit for her help and support in preparing this document for publication.

The project was funded by the Department of Health, and we thank Dr Elizabeth Meerabeau, Dr Elizabeth Tebbs and Dr William Maton-Howarth for their support.

#### **Project Team**

Rosalind Plowman BA, MSc, RGN Nicholas Graves BA, MA Mark Griffin BA, MSc Rachael Dunk RN Alison Franklin BSc, RN Janette Trevarthen RN, OHND Maggie Waters RN Jennifer White RN Linda Wright BSc, RN Barbara Ayres Christine Berry Project Co-ordinator Research Economist Statistician Research Assistant Research Assistant Research Assistant Research Assistant Research Assistant Research Assistant Project Secretary Administrative support

#### **Project Steering Committee**

| Jane Bandcroft RGN, DN, HV, FPCert                          | Director of Nursing Practice, Study Hospital                                                             |
|-------------------------------------------------------------|----------------------------------------------------------------------------------------------------------|
| Barry Cookson MBBS, BDS, MSc, Hon Dip.HIC,<br>MRCP, FRCPath | Director of Laboratory of Hospital Infection,<br>Public Health Laboratory Service                        |
| Helen Glenister BSc, MBA, PhD, RGN                          | Nursing Director, Medical Devices Agency,<br>Department of Health                                        |
| Bernadette Nazareth MBBS, MSc, MRCPath                      | Consultant in Communicable Disease Control,<br>Redbridge and Waltham Health Authority                    |
| Jennifer A Roberts MSc Econ, PhD, HMPHM                     | Reader in Economics of Public Health, London School of Hygiene and Tropical Medicine, London             |
| Mike Rowland MBBS, FRCP, FFPHM                              | Consultant Epidemiologist, Communicable Disease<br>Surveillance Centre, Anglia and Oxford RHA            |
| Anthony Swan PhD, HMFPHM, C Stat                            | Director, Statistics Unit, Public Health Laboratory<br>Service                                           |
| Lynda Taylor MBA, RGN, RM                                   | Head of Infection Control Unit,<br>Laboratory of Hospital Infection, Public Health<br>Laboratory Service |
| Jennie Wilson BSc, RGN                                      | Programme Leader Infection Control and Surveillance,<br>Infection Control Unit, Laboratory of            |

Hospital Infection, Public Health Laboratory Service

#### **Advisory Committee Members**

Sue Dewar RN, DNCert, MA Georgia Duckworth MSc, FRCP, FRC Path, FRIPHH Johanna Finn BSc, MHSM, Dip HSM Paul Gillett MRCP, FRC Path

Jayne Holmes RGN

Rosemary Jenkins MSc, DMS, MTD, RGN, RM

William Maton-Howarth PhD

Jennifer McIntyre MSc, RGN

Elizabeth Meerabeau BSc, PhD, MBA, RGN, RHV, RNT, RHVT

**Richard Murray MA** 

Elizabeth Tebbs MBBS, ChB

Ann Whittle

District nurse, Chichester Priority Care Services Trust

Regional Epidemiologist, CDSC, North Thames

Chief Executive, West Suffolk Hospitals Trust

Consultant Medical Microbiologist, Stoke Mandeville Hospital

Senior Nurse, Queen Elizabeth II Hospital, Welwyn Garden City (to 23 November 1996)

Nursing Officer, Department of Health (to 30 September 1995)

Senior Principal Scientific Officer Department of Health (from 1 April 1997)

Nursing Officer, Department of Health (from 30 September 1995)

Research Liaison Officer Department of Health (to 27th March 1997)

Economic Advisor, Department of Health

Senior Medical Officer, Department of Health

Department of Health (to 31st October 1995)

#### Introduction

At any one time approximately one in 10 patients in acute hospitals have a hospital-acquired infection (HAI) (DoH/PHLS, 1995). At the same time, an unquantified number of patients, discharged from hospital into the community, have an infection related to their recent hospital admission. These infections impose a burden on the secondary, tertiary and primary health-care sectors, community care services, the patients themselves and those who care for them. These burdens may be both financial and non-financial.

Studies that have estimated the cost of HAI generally focus on the burden to the hospital sector. Little is known about the costs incurred by the primary health-care sector, community care services, individual patients and their family and friends. These costs become increasingly relevant as the length of hospital stay becomes shorter and patients are discharged home at an earlier point in their recovery. This change in discharge pattern is likely to result in some treatment costs being shifted from the secondary healthcare sector to the primary health-care sector and community care services, and may result in an increase in the costs borne by patients, their family and friends.

The aim of this research was to provide a more comprehensive assessment of the nature, distribution and magnitude of the costs resulting from HAIs. To achieve this, a detailed analysis of the resources used in hospital and post-discharge was undertaken.

The results of this research should be of use to both purchasers and providers of health care, in particular those involved in the planning and management of infection prevention and control programmes.

The research was commissioned by the Department of Health to the Central Public Health Laboratory and the London School of Hygiene and Tropical Medicine, and forms part of the Department of Health's Research and Development Programme.

#### Aims and objectives

The aim of the study was to assess the burden of HAI in terms of the costs to the public sector, patients, informal carers and society as a whole. Specific objectives were to:

- 1. Determine the overall burden of HAI in terms of the:
- Costs to the secondary and primary health-care sectors and community care services.
- Impact on the health status of patients.
- Costs to patients, informal carers and the economy.
- 2. Establish the relative costs of different types of HAI.
- Determine the type of patients who incur the highest costs for specific infections.
- Use the data obtained to construct models to predict the effects of HAI on the cost categories described above.

#### **Research** methods

Adult patients with a minimum in-patient stay of 30 hours were recruited from the general wards of a district general hospital over a 13-month period between April 1994 and May 1995. Information on daily resource use was recorded for each patient for the duration of their hospital stay. Patients who presented with signs and symptoms of infection which met the definitions of infection used in this study, and a sample of patients who did not, were followed up post-discharge using a structured questionnaire. This questionnaire provided information on possible surgical wound, chest and urinary tract infections experienced after discharge from hospital; care received from health and community care services, family and friends; personal expenditure on items such as drugs and dressings; time of return to normal activities and, if applicable, employment; and information on the patients' health status following discharge from hospital. Information about care received post-discharge was also obtained from the patients' health-care records. Estimates of the cost of the resources used were made and analysed to determine the extent to which observed variations in costs incurred by infected and uninfected patients could be explained by the presence of an HAI.

The in-patient analysis considered how resource use and associated costs varied between patients with and without an HAI, and how these outcome measures varied with site of infection. The post-discharge analysis considered how costs varied between four patient groups:

 Patients who did not have an HAI identified during the in-patient phase or an infection identified postdischarge (Group 1).

- Patients who did not have an HAI identified during the in-patient phase, but reported symptoms and treatment that met the study criteria for one or more infections present post-discharge (Group 2).
- Patients who had one or more HAIs identified during the in-patient phase, but did not report symptoms and treatment that met the study criteria for one or more infections present post-discharge (Group 3).
- Patients who had one or more HAIs identified during the in-patient phase, and reported symptoms and treatment that met the study criteria for one or more infections present post-discharge (Group 4).

Patients were classified as having a possible infection postdischarge if they reported symptoms and treatment which met the criteria for surgical wound, chest or urinary tract infections used in this study. It was not possible to determine whether in all cases an infection was present, or whether it was acquired in hospital. Furthermore, where patients presented with an HAI in hospital, it was not clear whether the symptoms reported post-discharge represented a new infection or a continuation of a previously diagnosed infection.

Since factors other than the presence of an HAI may have accounted for some of the additional resource use and costs incurred by infected patients, resource and cost outcome measures were analysed using regression modelling which controlled for a range of potential confounders (age, sex, diagnosis, number of co-morbidities, admission specialty, admission type and, where appropriate, time of return of questionnaire). Estimates allowing for the effects of these confounders were subsequently derived from this modelling process.

#### Results

#### Recruitment and post-discharge response rates

- Four thousand adult patients were recruited into the study from the medical, surgical, orthopaedic, urology, gynaecology, ear, nose and throat (ENT), elderly care and, if they had a caesarean section, obstetric specialties.
- Complete in-patient data sets were obtained for 3980 patients.

- A total of 1449 patients were selected for follow-up into the community: 215 had an infection identified during the in-patient phase.
- Of those patients selected for follow-up, 41 died either before the first questionnaire was sent at four weeks post-discharge, or between the distribution of the first and second questionnaires at eight weeks post-discharge. Four of these patients had an HAI identified during the in-patient phase. All 41 patients were excluded from the response rate.
- Seventy-one per cent of patients returned the questionnaire after a maximum of two reminders.
- The response rate was similar for patients with and without an HAI identified during the in-patient phase.

#### **Incidence of HAIs**

- In-patient phase: 7.8% (95% CI: 7.0; 8.6) of patients were identified during the in-patient phase as having acquired one or more HAIs.
- Post-discharge phase: 19.1% (95% CI: 16.5; 21.9) of those patients who returned the questionnaire and who did not have an HAI identified during the inpatient phase and 30% (95% CI: 22.8; 38.0) of patients who had an HAI identified during the in-patient phase reported symptoms and treatment that met the criteria for a urinary tract, chest and/or surgical wound infection used in this study.

### Impact of HAI on hospital costs incurred during the in-patient phase

Patients who presented with one or more HAIs during their in-patient stay were found to incur costs that were, on average, 2.9 times greater than those for uninfected patients. In these study patients, this represented an absolute increase of £3154 per case. After adjusting for the effects of potential confounders the ratio was almost identical (2.8, 95% CI: 2.6; 3.0), suggesting that confounding had relatively little effect.

Hospital overheads, capital charges and the cost of management time accounted for 33% of the additional costs incurred, while nursing care accounted for 42%, medical care 6%, operations and consumables 6%, paramedics and specialist nurses 4%, antimicrobials 2%, other drugs 3%, microbiology tests 1%, and other tests and investigations 3% (see Figure 1). Figure 1. The distribution of the additional costs incurred by patients with one or more HAIs compared with uninfected patients during the in-patient hospital stay



The mean costs of treating infected and uninfected patients varied with specialty. Table 1 (page 4) shows the mean costs incurred, the ratio of the costs incurred by infected patients compared with uninfected patients, and the additional costs incurred by infected patients in this study. The figures in parentheses are the estimates obtained from the regression analysis.

### Impact of specific types of HAI on hospital costs incurred during the in-patient phase

The impact that HAIs had on hospital costs varied with the site of infection. Table 2 (page 4) shows the mean costs incurred, the ratio of the costs incurred by infected patients compared with uninfected patients, and the additional costs incurred by infected patients. The figures in parentheses are the estimates obtained from the regression analysis.

Infections of the urinary tract were found to be the least expensive, with costs, on average, 1.8 times higher than those for uninfected patients. In these patients, this is, on average, equivalent to an additional £1327 per patient.

Patients who acquired infections of the lower respiratory tract, skin, surgical wound or 'other' sites experienced similar patterns of increase in costs. Costs were, on average, two to 2.5 times greater than those incurred by uninfected patients, equivalent to an average increase of between £1618 and £2398 per patient.

The four patients who acquired bloodstream infections incurred costs that were, on average, over four times those for uninfected patients. In these patients, this is equivalent to an additional £5397 per patient. However, since there were only four patients in this infection group, two of whom died, general conclusions based on these results must be treated with caution.

Patients who acquired more than one HAI incurred the highest expenses, with costs, on average, 6.6 times greater than those incurred by uninfected patients. In these patients, this is equivalent to an additional £9152 per patient.

For all sites of infection, adjustment for potential confounders made little difference and the relative magnitudes of effects were almost entirely unchanged.

#### Impact of HAI on length of hospital stay

Patients who acquired an infection in hospital remained in hospital, on average, 2.9 times longer than uninfected patients, equivalent to an extra 14 days. After adjusting for other factors that might influence length of stay, the ratio of increase was modified to 2.5 (95% CI: 2.3; 2.7), which is, on average, equivalent to an extra 11 days.

### Impact of specific types of HAI on length of hospital stay

The extended hospital stay experienced by patients with an HAI varied with site of infection (Table 3, see page 5). Patients who acquired more than one infection were observed to have the greatest increase in mean length of stay. Patients with bloodstream infections had the lowest increase. However, as mentioned above, there were only four patients in this group, two of whom died while still in hospital.

### Impact of HAI on the health-care sector post-discharge

With the exception of patients who presented with an HAI as an in-patient and did not have an infection identified post-discharge, who on average incurred lower GP costs than patients in the other infection groups, patients who had an HAI identified during the in-patient phase, and/or an infection identified post-discharge, on average, had greater contact with their GP, visited the hospital more frequently for outpatient appointments and received more visits from district nurses compared with uninfected patients. Patients who acquired an infection therefore imposed an additional economic burden on these services. Acquiring an infection was not found to have a positive impact on the number or cost of health

| Table 1. Mean costs incurred duri | ig the in-patient phase b | y patients with and without an HAI and b | y admission special | ty |
|-----------------------------------|---------------------------|------------------------------------------|---------------------|----|
|-----------------------------------|---------------------------|------------------------------------------|---------------------|----|

| Specialty    | Mean costs (£) |                |         | Ratio of costs           | Additional costs (£) |
|--------------|----------------|----------------|---------|--------------------------|----------------------|
|              | No HAI n       | No HAI n HAI n |         | (model estimate; 95% Cl) | (model estimate)     |
|              | (a)            | (b)            | · · · · | (b/a)                    | (b-a)                |
| Medicine     | 1559 800       | <b>7</b> 271   | 38      | 4.7 (4.6; 3.8, 5.6)      | 5712 (5621)          |
| Surgery      | 1290 844       | 6189           | 54      | 4.8 (3.9; 3.3, 4.7)      | 4898 (3795)          |
| Orthopaedics | 2089 473       | 5385           | 40      | 2.6 (2.6; 2.1, 3.1)      | 3296 (3285)          |
| Urology      | 1276 439       | 2758           | 27      | 2.2 (2.2; 1.7, 2.8)      | 1482 (1544)          |
| Gynaecology  | 1661 339       | 2196           | 51      | 1.3 (1.3; 1.1, 1.5)      | 535 (470)            |
| Elderly care | 1748 508       | 5277           | 74      | 3.0 (3.1; 2.6, 3.5)      | 3529 (3578)          |
| ENT          | 2127 64        | 5644           | 2       | 2.7 (1.9; 0.8, 4.6)      | 3516 (2007)          |
| Obstetrics*  | 2481 204       | 2761           | 23      | 1.1 (1.1; 0.8, 1.4)      | 280 (118)            |
| Overall      | 1628 3671      | 4782           | 309     | 2.9 (2.8; 2.6, 3.0)      | 3154 (2917)          |

\*Caesarean sections only

|  | Table | 2. Mean | costs | incurred | during | the in | -patient | ohase b | y site | of | HAI |
|--|-------|---------|-------|----------|--------|--------|----------|---------|--------|----|-----|
|--|-------|---------|-------|----------|--------|--------|----------|---------|--------|----|-----|

| Site of infection | Mean costs (£) | n    | Ratio of costs<br>(model estimate; 95% Cl) | Additional costs (£)<br>(model estimate) |  |  |
|-------------------|----------------|------|--------------------------------------------|------------------------------------------|--|--|
| No HAI            | 1628           | 3671 | ······································     |                                          |  |  |
| UTI               | 2955           | 107  | 1.8 (1.7; 1.5, 1.9)                        | 1327 (1122)                              |  |  |
| LRTI              | 4027           | 48   | 2.5 (2.3; 1.9, 2.7)                        | 2398 (2080)                              |  |  |
| SWI               | 3246           | 38   | 2.0 (2.0; 1.6, 2.4)                        | 1618 (1594)                              |  |  |
| BSI               | 7026           | 4    | 4.3 (4.8; 2.6, 8.8)                        | 5397 (6209)                              |  |  |
| Skin              | 34+8           | 25   | 2.1 (2.0, 1.6, 2.5)                        | 1790 (1615)                              |  |  |
| Other             | 3892           | 30   | 2.4 (2.5; 2.0, 3.1)                        | 2263 (2465)                              |  |  |
| Multiple          | 10780          | 57   | 6.6 (6.3; 5.4, 7.4)                        | 9152 (8631)                              |  |  |
| Any infection     | 4782           | 309  | 2.9 (2.8; 2.6, 3.0)                        | 3154 (2917)                              |  |  |

UTI=urinary tract infection; LRTI=lower respiratory tract infection; SWI=surgical wound infection; BSI=bloodstream infection

visitor and community midwife visits. Table 4 (page 5) summarises the impact of HAI on health sector costs postdischarge. The mean costs incurred by patients in the four HAI groups are presented, together with the ratio of the costs incurred by infected patients compared with uninfected patients and the additional costs incurred by infected patients. The figures in parentheses are the estimates obtained from the regression analysis.

#### General practitioners

Patients who did not present with an HAI while in hospital but reported symptoms and treatment that met the study criteria for an infection post-discharge, and patients who developed an HAI while in hospital and had an infection identified post-discharge, on average, incurred proportionally greater costs than patients in the other two groups. However, the average increases in the absolute costs observed were minimal.

#### Table 3. Mean length of hospital stay by site of HAI

| Site of infection | Mean LoS<br>(days) | n    | Ratio<br>(model estimate: 95% CI) | Additional days<br>(model estimate) |
|-------------------|--------------------|------|-----------------------------------|-------------------------------------|
| No HAI            | 8                  | 3671 |                                   |                                     |
| UTI               | 14                 | 107  | 1.8 (1.7; 1.5, 1.9)               | 6 (5)                               |
| LRTI              | 20                 | 48   | 2.6 (2.1; 1.7, 2.6)               | 12 (8)                              |
| SWI               | 14                 | 38   | 1.9 (1.9; 1.6, 2.4)               | 7 (7)                               |
| BSI               | 10                 | 4    | 1.2 (1.5; 0.8, 3.0)               | 2 (4)                               |
| Skin              | 20                 | 25   | 2.6 (2.4; 1.8, 3.1)               | 12 (11)                             |
| Other             | 21                 | 30   | 2.8 (2.6; 2.1, 3.4)               | 13 (12)                             |
| Multiple          | 45                 | 57   | 6.0 (4.8; 4.0, 5.8)               | 38 (29)                             |
| Any infection     | 22                 | 309  | 2.9 (2.5; 2.3, 2.7)               | 14 (11)                             |

LoS=length of stay; UTI=urinary tract infection; LRTI=lower respiratory tract infection; SWI=surgical wound infection; BSI=bloodstream infection

| Table 4. Impact of HAI on health sector costs incurred post-disc | harge |
|------------------------------------------------------------------|-------|
|------------------------------------------------------------------|-------|

| One or more HAIs<br>identified during the<br>in-patient phase | One or more<br>infections identified<br>post-discharge | n             | Health-care<br>professional visited* | Mean observed<br>costs (£) | Ra<br>(model | tio of costs<br>estimate; 95% CI) | Addition<br>(mode | al costs (£)<br>estimate) |
|---------------------------------------------------------------|--------------------------------------------------------|---------------|--------------------------------------|----------------------------|--------------|-----------------------------------|-------------------|---------------------------|
| No                                                            | No                                                     | 664           | GP                                   | 18                         | _            |                                   |                   |                           |
|                                                               |                                                        | <b>658</b>    | HD/HN                                | 32                         |              |                                   |                   |                           |
|                                                               |                                                        | 558           | DN                                   | 34                         |              |                                   |                   |                           |
| No                                                            | Yes                                                    | 159           | GP                                   | 28                         | l.6          | (1.7; 1.3, 2.3)                   | 10                | (12)                      |
|                                                               |                                                        | 160           | HD/HN                                | 39                         | 1.2          | (1.9; 1.3, 2.6)                   | 7                 | (28)                      |
|                                                               |                                                        | 130           | DN                                   | 39                         | 1.2          | (1.5; 1.0, 2.1)                   | 6                 | (16)                      |
| Yes                                                           | No                                                     | <del>99</del> | GP                                   | 14                         | 0.8          | (0.8; 0.5, 1.1)                   | -4                | (-4)                      |
|                                                               |                                                        | 102           | HD/HN                                | 36                         | 1.1          | (1.3, 0.9, 2.0)                   | 4                 | (11)                      |
|                                                               |                                                        | 89            | DN                                   | 59                         | 1.8          | (1.6; 1.0, 2.3)                   | 25                | (19)                      |
| Yes                                                           | Yes                                                    | 43            | GP                                   | 24                         | 1.4          | (1.5; 0.9, 2.6)                   | 6                 | (10)                      |
|                                                               |                                                        | 43            | HD/HN                                | 40                         | 1.3          | (2.7; 1.5, 4.7)                   | 8                 | (53)                      |
|                                                               |                                                        | 39            | DN                                   | 78                         | 2.3          | (2.6; 1.4, 4.7)                   | 44                | (53)                      |

\*Sources: GP and HD/HN (the post-discharge questionnaire); DN (the DN database)

GP=general practitioner; HD/HN=hospital doctor/hospital nurse; DN=district nurse

#### Hospital doctor/nurse

Patients who did not present with an HAI while in hospital but had an infection identified post-discharge, and patients who presented with an HAI while in hospital and had an infection identified post-discharge incurred slightly higher costs than patients in the other two categories.

#### District nurses

Patients who had an HAI identified during the in-patient phase, and/or an infection identified post-discharge, on average, had a greater impact on district nursing costs compared with uninfected patients. Patients who presented with an HAI as an in-patient and had an infection identified post-discharge had the greatest impact on district nursing costs. The results from the regression analysis suggest there was some confounding and that the effects of HAI on GP, district nursing and hospital costs in a number of cases were probably larger than those observed.

#### Impact of HAI on costs incurred by patients

Personal expenditure on items such as drugs and dressings was found to vary with HAI group. The mean costs incurred by patients in the four HAI groups are presented in Table 5, together with the ratio of the costs incurred by infected patients compared with uninfected patients and the additional costs incurred by infected patients. The figures in parentheses are the estimates obtained from the regression analysis.

Increases in personal expenditure were greatest for patients who presented with an HAI as an in-patient and had an infection identified post-discharge. These patients experienced costs that were 3.2 times greater than those incurred by uninfected patients. Adjustment for potential confounders made little difference and the relative magnitudes of effect remained similar to the observed effects.

### Impact of HAI on the number of days from admission to return to normal daily activities

The number of days from admission to resuming normal daily activities varied with HAI group. The mean number of days from admission to resuming normal daily activities for patients in the four infection groups are presented in Table 6, together with the ratio of the number of days infected patients were away from normal daily activities, compared with uninfected patients. The additional number of days that infected patients took to resume normal daily activities, compared with uninfected patients, are also presented.

Patients who had an HAI identified during the in-patient phase and/or an infection identified post-discharge, on average, took longer to resume normal daily activities than patients in the uninfected group. Patients who had an HAI identified during the in-patient period and reported symptoms and treatment that met the criteria for an infection post-discharge took longer to resume normal daily activities than patients in the other infection groups.

#### Table 5. Impact of HAI on personal costs incurred by patients

| One or more HAIs<br>identified during the<br>in-patient phase | One or more<br>infections identified<br>post-discharge | n   | Mean observed<br>costs (£) | Ratio of costs<br>(model estimate: 95% Cl) | Additional costs (£)<br>(model estimate) |
|---------------------------------------------------------------|--------------------------------------------------------|-----|----------------------------|--------------------------------------------|------------------------------------------|
| No                                                            | No                                                     | 691 | 9                          |                                            |                                          |
| No                                                            | Yes                                                    | 163 | 15                         | 1.7 (1.5; 1.1, 1.9)                        | 6 (4)                                    |
| Yes                                                           | No                                                     | 105 | 5                          | 0.5 (0.9; 0.6, 1.3)                        | -4 (í)                                   |
| Yes                                                           | Yes                                                    | 45  | 30                         | 3.2 (3.2; 2.0, 5.0)                        | 20 (20)                                  |

Table 6. Mean number of days from admission to return to normal daily activities by HAI status

| One or more HAIs<br>identified during the<br>in-patient phase | more HAIs One or more<br>I during the infections identified<br>t phase post-discharge |     | Mean no.<br>of days | Ratio of days<br>(model estimate: 95% Ci) | Additional days<br>(model estimate) |  |  |
|---------------------------------------------------------------|---------------------------------------------------------------------------------------|-----|---------------------|-------------------------------------------|-------------------------------------|--|--|
| No                                                            | No                                                                                    | 642 | 29                  | · · · · · · · · · · · · · · · · · · ·     |                                     |  |  |
| No                                                            | Yes                                                                                   | 155 | 35                  | 1.2 (1.2; 1.1, 1.4)                       | 6 (6)                               |  |  |
| Yes                                                           | No                                                                                    | 94  | 41                  | 1.4 (1.4; 1.3, 1.6)                       | 12 (13)                             |  |  |
| Yes                                                           | Yes                                                                                   | 43  | 43                  | 1.5 (1.6; 1.3,1.9)                        | 13 (17)                             |  |  |

6

#### Impact of HAI on the number and value of days employed patients were away from paid employment

The number and value of days from admission to return to paid employment varied with HAI group. The mean number and value of days from admission to return to paid employment for patients in the four infection groups are presented in Tables 7–8, together with the ratio of the number and value of days infected patients were away from employment, compared with uninfected patients. The additional number of days infected patients were away from paid employment, compared with uninfected patients, are also presented.

Patients who had an HAI identified during the in-patient phase and/or an infection identified post-discharge had a greater number of days away from employment than uninfected patients.

#### Impact of HAI on the number of days informal carers spent caring for patients and their dependants

The number and value of days informal carers spent caring for the patient's dependants during the in-patient period and the patient post-discharge varied with HAI group. The mean number of days of care provided by informal carers for patients in the four infection groups are presented in Table 9 (page 8) and the estimated value of this time is presented in Table 10 (page 8). The ratio of the number of days of care received by infected compared with uninfected patients and the associated value, together with the number of additional days of care received by infected compared with uninfected patients, are also presented in these tables.

Patients who reported symptoms and treatment that met the study criteria for one or more infections present postdischarge, regardless of whether they presented with an infection in hospital, on average, received more care from informal carers than patients who had not acquired an infection, or who presented with an infection in hospital but did not have an infection identified post-discharge.

#### Impact of HAI on health status

The responses given to the general health status questionnaire, the SF-36, administered four weeks post-discharge, provided information on eight dimensions of health. Two summary measures relating to physical and mental well-being were derived from these data. Patients

Table 7. Mean number of days from admission to return to employment by HAI status

| One or more HAIs<br>identified during the<br>in-patient phase | One or more<br>infections identified<br>post-discharge | n   | Mean no.<br>of days | Ratio of days<br>(model estimate; 95% Cl) | Additional days<br>(model estimate) |
|---------------------------------------------------------------|--------------------------------------------------------|-----|---------------------|-------------------------------------------|-------------------------------------|
| No                                                            | No                                                     | 267 | 23                  |                                           |                                     |
| No                                                            | Yes                                                    | 66  | 29                  | 1.2 (1.1; 1.0, 1.3)                       | 6 (2)                               |
| Yes                                                           | No                                                     | 30  | 29                  | 1.3 (1.2; 1.0, 1.5)                       | 6 (6)                               |
| Yes                                                           | Yes                                                    | п   | 28                  | 1.2 (1.3; 0.9, 1.7)                       | 5 (6)                               |

Table 8. Mean value of days from admission to return to employment by HAI status

| One or more HAIs<br>Identified during the<br>in-patient phase | One or more<br>infections identified<br>post-discharge | n   | Mean value<br>of days | Ratio of costs<br>(model estimate; 95% CI) | Additional costs (£)<br>(model estimate) |
|---------------------------------------------------------------|--------------------------------------------------------|-----|-----------------------|--------------------------------------------|------------------------------------------|
| No                                                            | No                                                     | 267 | 1429                  |                                            |                                          |
| No                                                            | Yes                                                    | 66  | 1724                  | 1.2 (1.1; 1.0, 1.3)                        | 295 (200)                                |
| Yes                                                           | No                                                     | 30  | 1649                  | 1.2 (1.2; 1.0, ±.5)                        | 220 (300)                                |
| Yes                                                           | Yes                                                    | 14  | 1889                  | 1.3 (1.6; 1.1, 2.2)                        | 460 (801)                                |

Table 9. Mean number days informal carers spent caring for dependants and patients by HAI status

| One or more HAIs<br>identified during the<br>in-patient phase | One or more<br>infections identified<br>post-discharge | <b>.</b> . | Mean no.<br>of days | Ratio of days<br>(model estimate; 95% Cl) | Additional days<br>(model estimate) |
|---------------------------------------------------------------|--------------------------------------------------------|------------|---------------------|-------------------------------------------|-------------------------------------|
| No                                                            | No                                                     | 691        | 10.3                |                                           |                                     |
| No                                                            | Yes                                                    | 163        | 14.4                | 1.4 (1.2; ±.0, 1.6)                       | 4.1 (2.1)                           |
| Yes                                                           | No                                                     | 105        | 10.5                | 1.0 (0.9; 0.6, 1.2)                       | 0.2 (-1.3)                          |
| Yes                                                           | Yes                                                    | 45         | 20.9                | 2.0 (1.6; 1.0,2.5)                        | 10.6 (6.1)                          |

Table 10. Mean value of days informal carers spent caring for dependants and patients by HAI status

| One or more HAIs<br>identified during the<br>in-patient phase | One or more<br>infections identified<br>post-discharge | n   | Mean value<br>of days (£) | Ratio of costs<br>(model estimate; 95% CI) | Additional costs (£)<br>(model estimate) |  |
|---------------------------------------------------------------|--------------------------------------------------------|-----|---------------------------|--------------------------------------------|------------------------------------------|--|
| No                                                            | No                                                     | 691 | 348                       |                                            |                                          |  |
| No                                                            | Yes                                                    | 163 | 488                       | 1.4 (1.3; 0.8, 2.1)                        | 140 (96)                                 |  |
| Yes                                                           | No                                                     | 105 | 355                       | 1.0 (0.7; 0.4, 1.3)                        | 7 (-100)                                 |  |
| Yes                                                           | Yes                                                    | 45  | 707                       | 2.0 (2.3; 0.9, 5.6)                        | 359 (454)                                |  |

with an HAI, on average, obtained lower scores for these two measures than patients who did not acquire an infection, indicating a poorer outcome as determined by these health measures. Patients who presented with an HAI as an in-patient and reported symptoms and treatment which met the study criteria for an infection present postdischarge, on average, reported the lowest health status.

#### Impact of HAI on in-patient mortality

The in-patient death rate was found to be considerably higher in patients with an HAI which presented during the hospital stay: 13% of patients with an HAI died compared with 2% of patients who did not present with an HAI in hospital. After adjustment for the effects of age, sex, diagnosis, number of co-morbidities, admission specialty and admission type, patients with an HAI were found to be 7.1 (95% CI: 4.3; 11.7) times more likely to die in hospital than uninfected patients.

Estimates were made of the number of years of life lost by infected patients who died. Patients aged 25–44 years who acquired an infection in hospital and subsequently died, on average, lost 44 years; patients aged 45–64 lost 19 years, patients aged over 65–84 years lost 11 years, and patients aged 85 years and over lost 4 years. Since it was not pos-

sible to determine for each individual case whether the HAI was the primary cause of death, a contributing factor, or whether it made no contribution to the death, neither the number nor value of the years of life lost as a result of an HAI could be determined. However, it is important to acknowledge that years of life lost do have a value and represent an important cost associated with HAI.

#### National estimates

The study results were used to estimate the economic burden of HAIs occurring in adult (218 years) patients, excluding day cases, admitted to the specialties covered in this study throughout England. Patients admitted to these specialties accounted for approximately 70% of adult, nonday case NHS admissions in England in 1994–1995.

The results presented are based on the assumption that the incidence of HAI, the ratio of increase in costs incurred by infected compared with uninfected patients and the mean cost of treating uninfected patients observed in this study are representative of the incidence and costs incurred by patients admitted to the specialties covered in this study throughout England.

### Estimates of the economic burden of HAI to the NHS in England

HAIs were estimated to cost the NHS in England £986.36 million annually. Of this aggregate cost, £930.62 million (95% CI: £780.26; £1080.97 million) was estimated to have been incurred during the patients' hospital stay and £55.74 million post-discharge. These post-discharge costs were distributed between GPs (£8.4 million), hospitals (outpatient consultations) (£26.83 million) and district nursing services (£20.51 million). The estimates of the effect of HAI on health sector costs incurred post-discharge varied considerably, depending on whether the HAI presented during the in-patient and/or post-discharge phase. The 95% confidence intervals obtained for the different infection groups were wide and this should be taken into account when using these estimates.

The in-patient hospital estimates represent 9.1% of the acute, geriatric and obstetric programme budget for 1994-95, and estimates of the cost to the hospital sector postdischarge 0.9% of the outpatient acute, geriatric and obstetric programme budget for the same year (data from Department of Health). The estimated burden to GPs represents 0.3% of the general medical services budget for 1994-95 (data from the Department of Health) and the estimated burden to district nursing services represents 2.4% of their budget for the same year (data from the Department of Health).

Table 11 presents estimates of the impact of specific types of infection on in-patient costs. The cost estimates are limited to those incurred by the hospital sector during the inpatient stay. Nationally, infections of the urinary tract were estimated to be the most expensive single-site infection, costing an estimated £123.89 million per annum (95% CI: £80.96; £166.83). These infections were relatively inexpensive to treat (the additional cost per case observed in this study was £1327, model estimate £1122), but their relatively high incidence means that, nationally, they impose a substantial burden on the NHS. No attempt was made to derive site-specific estimates of the impact of HAI on health sector costs incurred post-discharge. Table 11. National estimates of the burden of HAI to the hospital sector in England by site of infection (in-patient costs only\*)

| Site of infection | Estimates of the national burden of HAI<br>Figures expressed in f (millions) |        |                |  |  |
|-------------------|------------------------------------------------------------------------------|--------|----------------|--|--|
|                   | Estimate                                                                     | 95% CI |                |  |  |
|                   |                                                                              | Low    | High           |  |  |
| UTI               | 123.89                                                                       | 80.96  | 166.83         |  |  |
| LRTI              | 103.77                                                                       | 59.41  | 148.12         |  |  |
| SWI               | 62.37                                                                        | 30.93  | 93.82          |  |  |
| BSI               | 25.53                                                                        | -6.86  | 57.91          |  |  |
| Skin              | 41.79                                                                        | 15.40  | 68. <b>1</b> 7 |  |  |
| Other             | 75.87                                                                        | 36.52  | 115.23         |  |  |
| Multiple          | 507.77                                                                       | 348.89 | 666.65         |  |  |

UTI=urinary tract infection; LRTI=lower respiratory tract infection; SWI=surgical wound infection; BSI=bloodstream infection

\*Estimates are limited to the additional costs incurred as a result of HAIs occurring in adult patients, excluding day cases, admitted to the specialties covered in this study: approximately 70% of all adult, nonday case NHS admissions

### Estimates of the economic burden of HAI to patients

Personal expenditure on items such as drugs and dressings incurred by patients who acquire an infection in hospital are estimated to amount to £4.74 million annually. The estimates derived varied considerably depending on whether the patient presented with an HAI in hospital and/or had an infection identified post-discharge. The confidence intervals derived for each HAI group were wide and this should be taken into account when using these estimates.

# Estimates of the number of extra days patients took to resume normal daily activities

Nationally, patients who acquire an infection in hospital, when compared with uninfected patients, were estimated to take an additional 8.7 million days to resume normal daily activities. The estimates varied considerably with HAI group and the 95% confidence intervals were wide. These factors should be taken into account when considering these estimates.

#### The benefits of prevention

This study was not directly concerned with estimating the benefits of prevention. However, the estimates presented provide important information on the value of resources that might be released for alternative use if a proportion of infections are prevented. These may be viewed as the gross benefits of prevention. Net benefits will depend on the cost and effectiveness of prevention activities.

Estimates of the gross benefits which may result from a 10% reduction in the observed incidence rate, both in terms of the benefits to the study hospital and to provider units throughout England, are presented in the report. In addition to estimates of the value of resources released for alternative use, the value of consumables released and the number of bed days released are presented.

At the level of the study hospital, a 10% reduction in the observed incidence rate was estimated to result in the release of resources valued at £361 297 (95% CI: 302 924; 419 670). A similar reduction at the national level was estimated to result in the release of resources valued at £93.06 million (95% CI: 78.03; 108.10 million).

In the short term, only a relatively small proportion of these benefits are likely to be in the form of cash savings. However, over a longer period of time it is possible that some of the fixed costs might be avoided and, as such, the proportion of benefits that may accrue as cash benefits may increase.

In terms of the number of bed days released for alternative use, at the level of the study hospital a similar level of reduction may result in an estimated 1413 (95% CI: 1168; 1659) bed days released for alternative use; equivalent to an estimated 191 finished consultant episodes (95% CI: 158; 224). At the national level, 364 056 (95% CI: 300 880; 427 223) bed days may be released; equivalent to an estimated 47 902 finished consultant episodes (95% CI: 39589; 56214).

These estimates, although considerable, may be conservative estimates of the value of resources that might be released. They are limited to the benefits that may result from a reduction in the incidence of HAI occurring in adult patients admitted to the specialties covered in this study, and are based on a 10% reduction in the incidence rate. The literature suggests that up to 30% of HAIs may be prevented through effective infection control programmes (Haley, 1986).

#### Discussion

The results of this study clearly indicate that HAIs impose a substantial burden on the secondary and primary health-care sectors, on infected patients and their informal carers. A detailed analysis of the effect of HAI on resource use and costs was undertaken, the results of which provide important information on the nature, magnitude and distribution of the economic burden. The approach taken is considerably more detailed than earlier studies which have generally limited the analysis of costs to those incurred by the hospital sector and have not attempted to determine the distribution of these costs in any great detail.

Three main points should be borne in mind when interpreting these findings.

First, attributing costs to the presence of an HAI is extremely complex. The characteristics of patients with an HAI may differ systematically from those of uninfected patients. If these differences result in additional resource use, this would bias the estimates of the effects of HAIs. The in-patient regression analysis showed this was not the case for age, sex, admission type, specialty, diagnosis and co-morbidities. Nonetheless, the possibility that there may be some other confounding factors cannot be completely ruled out. For example, due to factors not included in the regression analysis, patients with an HAI may have remained in hospital longer than similar patients who did not acquire an infection, regardless of whether they acquired an infection or not. An analysis investigating this possibility revealed some evidence that the difference in length of stay between patients with and without an HAI was not due entirely to the infection. Consequently, the estimates of the effect of HAI on length of stay and the associated costs may be biased. However, estimates of the magnitude of this bias were very sensitive to the strong simplifying assumptions on which they were based and, as such, it would be unwise to conclude more than that the estimated effects of HAI on length of hospital stay presented may include an upward bias. The post-discharge regression analysis indicated that there was some confounding and that the effects in a number of cases were probably larger than those observed in the unadjusted figures.

Second, the study was restricted to patients admitted to one NHS trust over a 13-month period. Future patients admitted to this and other NHS trusts might differ in various ways. In addition, estimates of the costs of resources used were, in most cases, specific to this NHS trust, and clinical practice affecting resource use might differ with time and with provider unit. However, it seems reasonable to assume that any differences that occur will be the same for patients with and without an HAI. On this assumption, the proportion by which an HAI increases resource use will not be affected and, consequently, the proportional effects estimated from this study will be generalisable. Absolute increases in costs incurred by infected patients may differ with time and with provider unit. However, since the study hospital was found to be broadly similar to other provider units in terms of factors such as average length of stay and average cost per bed day, it is reasonable to assume that the estimated effects of HAI on absolute costs are also fairly generalisable.

Third, when considering both the gross and net benefits of prevention, it is important to realise that any savings represent a reduction in individual treatment costs and not necessarily an overall saving to the health sector. This will depend on how released resources are utilised and this will, to some extent, depend on the structure of the contracts and agreements in place. If, for example, the prevention of infection results in a reduction in length of hospital stay, bed days will be released for alternative use. If these released bed days are utilised by more expensive patients then, rather than resulting in a cost saving for the NHS, overall expenditure will increase. However, this will be offset by benefits gained by the extra patients treated.

#### Conclusion

The results of this study provide a detailed account of the socio-economic burden imposed by HAIs occurring in adult patients admitted to selected specialties common to most NHS provider units. It represents the first comprehensive attempt to estimate these costs. The results provide valuable information that might be used at national and local level to inform the management of HAI and, when used alongside effectiveness studies of infection prevention and control measures, will facilitate the development of effective policies to control HAI.

#### Recommendations

### Specific recommendations arising from this research

Commissioners of health care (purchasing agencies) should:

- Be aware of the magnitude of the overall burden imposed by HAI and how it is distributed.
- Ensure adequate details on infection control arrangements and ongoing strategies for the prevention of infection are in place in all provider units with which they contract.
- Recognise that, in a number of cases, HAIs present after discharge from hospital, and that these infections should be monitored and the needs of affected patients met.

#### Providers of health care should:

- Use the findings of this study, together with information on the effectiveness of different infection control activities, to inform infection prevention and control strategies within their provider unit.
- Ensure appropriate arrangements are in place to monitor infections presenting post-discharge and the needs of affected patients are met.

### Educational institutions involved in the education of health care personnel should:

 Include the socio-economic burden imposed by HAI in their educational programmes on HAI and in so doing raise awareness of the issues relating to HAI and the importance of infection prevention and control strategies.

#### Further research and development

During the course of this study a number of areas which would benefit from further research and development were identified. These are briefly presented below.

The first area that requires some further work relates to how generalisable the results of this study are to future patients in other health-care settings. For reasons discussed above, it seems reasonable to assume that the results are generalisable, but further work will be carried out to assess in greater detail whether the pattern of resource use observed in this study is broadly similar to that found in other provider units. It is also recommended that further methodological work be undertaken to increase knowledge of how best to estimate the cost of hospital services.

Attributing costs to the presence of an HAI presented a number of methodological difficulties. In this study, regression analysis was used to control for a range of factors. However, as discussed above, factors not included may have had an impact on resource use and costs. For example, patients with an HAI may have remained in hospital longer than uninfected patients due to factors other than those included in the regression analysis. An analysis has been undertaken to investigate this, but the results were sensitive to the strong simplifying assumptions on which the analysis was based. It is therefore recommended that further work on the complex relationship between length of stay and HAIs be undertaken to assess more precisely what part of the length of stay can be ascribed to the effect of HAI and the associated costs.

Following discharge from hospital, patients with an HAI were found to make more visits to their GP and/or doctor or nurse at the hospital than uninfected patients. Consequently, infected patients had a greater economic impact on these health-care services than uninfected patients. The analysis to date has not taken into account the resource intensity of these visits. It is possible that visits made by patients with an HAI were more resourceintensive, and thus the economic impact was greater, than that estimated in this study. It is therefore recommended that the data obtained in this study be further analysed to determine the resource intensity of visits made to GPs and hospital doctors/nurses, and how this varies between patients with and without an HAI. It is also recommended that further work be conducted to determine whether the health needs of patients experiencing HAIs in the community are being met.

Acquiring an HAI in hospital was associated with a reduction in mental and physical well-being, as measured by the SF-36. It is recommended that further work be carried out to explore the nature and reasons for the apparent reduction in mental and physical well-being observed in patients with an HAI compared with uninfected patients, and that the results of this work are used, where possible, to inform clinical practice.

As part of this study, a decision support system to model and predict the effects of HAI on components of resource use and their costs within different provider units was developed. It is recommended that this system be further developed to create a user-friendly decision support mechanism which meets the information needs of both purchasers and providers of health care.

The results of this study provide information on the nature, distribution and magnitude of the burdens imposed by HAI. These burdens represent the potential gross benefits of prevention. Further work is required to determine the cost-effectiveness of selected infection control practices. The information derived may then be used to inform infection control practice and the overall allocation of resources to infection control.

Finally, the results of this study relate to adult patients, excluding day cases, admitted to the general specialties of a district general hospital. Patients admitted to these specialties accounted for approximately 70% of adult, nonday case NHS admissions in 1994–95. It is recommended that future work examine the socio-economic burden of HAIs occurring in the other patient groups, in particular in patients at high risk of acquiring an infection in hospital (e.g. babies cared for in special care baby units) and patients undergoing major and specialised surgery (e.g. cardiothoracic surgery and organ transplantation).

#### References

DoH/PHLS (1995) Hospital Infection Control. Guidance on the control of infection in hospitals. The Hospital Infection Working Group of the Department of Health and Public Health Laboratory Service, London.

Haley, R.W. (1986) Munaging Hospital Infection Control for Cost-Effectiveness: A Strategy for Reducing Infectious Complications. Chicago: American Hospital Publishing.